BIRLA CENTRAL LIBRARY PILANI [RAJASTHAN] Class No. 517. 306 Book No. 7581 Accession No. 79198

Acc No					
Not	ISSUE LABEL Not later the latest date stamped below				
		\$			
	,				

INTRODUCTION TO THE THEORY OF FOURIER INTEGRALS

 $\mathbf{B}\mathbf{Y}$

E. C. TITCHMARSH F.R.S.

SAVILIAN PROFESSOR OF GEOMETRY IN THE UNIVERSITY OF OXFORD

SECOND EDITION

OXFORD AT THE CLARENDON PRESS

Oxford University Press, Amen House, London E.C. 4
GLASGOW NEW YORK TORONTO MELBOURNE WELLINGTON
BOMBAY CALCUTTA MADRAS CAPE TOWN
Geoffrey Cumberlege, Publisher to the University

FIRST EDITION 1937
SECOND EDITION 1948
Reprinted lithographically in Great Britain at the UNIVERSITY PRESS, OXFORD, 1950 from corrected sheets of the first edition

PREFACE

THE object of this book is to give a more systematic account of the elements of the theory of Fourier integrals than has hitherto been given. I have, however, not attempted to deal with a number of important topics of recent growth: Wiener's Tauberian theorems; applications to almost periodic functions, quasi-analytic functions, and integral functions; Stieltjes integrals; harmonic analysis in general; and Bochner's generalized integrals, and the theory for functions of several variables, of which an account is given in Bochner's book.

The reader requires only a general knowledge of analysis, though he will presumably be familiar with the elements of the theory of Fourier series. The book may be read as a sequel to my *Theory of Functions*.

A great variety of applications of Fourier integrals are to be found in the literature, often in the form of 'operators', and often in the works of authors who are evidently not specially interested in analysis. As exercises in the theory I have written out a few of these applications as it seemed to me that an analyst should. I have retained, as having a certain picturesqueness, some references to 'heat', 'radiation', and so forth; but the interest is purely analytical, and the reader need not know whether such things exist.

E. C. T.

NEW COLLEGE, OXFORD, 1937.

CONTENTS

I. CONVERGENCE AND SUMMABILITY

1.1.	Fourier's formulae .						
1.2.	Fourier transforms .						:
1.3.	Generalized Fourier integrals						4
1.4.	The formulae of Laplace	•	•				(
1.5-6	5. The formulae of Mellin						•
1.7.	Notation		•	•			9
1.8-9	. Fundamental theorems						11
1.10.	Monotonic functions .						18
1.11.	Functions containing a period	lic factor					17
1.12.	Oscillating functions .						22
	The constant in Fourier's for	mula					24
1.14.	Fourier's single-integral forms	ula					25
	Summability of integrals			•			26
	Summability of Fourier integration	rals					28
	Cauchy's singular integral			•			30
	Weierstrass's singular integral	١.	•				31
	20. General summability						32
	3. Further summability theore	ems					36
	The integrated form of Fourie		ıla				41
	The complex form of Fourier'						42
	Perron's formula .	•					43
1.27.	Fourier's theorem for analytic	function	ıs				44
	Summability of the complex i						45
	Mellin's inversion formula			•			46
	The Laplace formulae						48
		-		-		-	
	II. AUXILIA	RY FOR	MULA	E			
2.1.	Formalities						50
2.2-5	. Conditions for validity	•			•		54
2.6.	Transform of a resultant						59
2.7.	Mellin transforms .	•					60
2.8-9.	Poisson's formula .						60
2.10.	Examples	•					64
2.11.	Sine transforms .						66
2.12.	More general conditions						66
	C						
	III. TRANSFORM	is of t	HE CL	ASS L2			
3.1.	Plancherel's theory of Fourier	transfor	ms		•		69
	Fourier transforms, first meth		•				70
	Fourier transforms, second me						73
3.4.	Fourier transforms, third met	hod			•	•	75
	The Hermite polynomials	•					76
	Fourier transforms, fourth me	thod			•		82

vi

3.10. Convergence and summability .				. 83
3.11-12. Convergence almost everywhere				. 85
3.13. Theorems on resultants .				. 90
3.14–15. Special theorems				. 92
3.16. Another case of Parseval's formula				. 94
3.17. Mellin transforms				. 94
IV. TRANSFORMS OF CTE	1121D <i>1</i>	r OT AC	e ne	
	I LYIV T	J-C LAO	BEB	
4.1-2. Transforms of functions of L^p	•	•	•	. 96
4.3. Proof of Theorem 74 for $p = 2k/(2k - 2k)$	1)	•	•	. 98
4.4-5. Extension to general p	•	•	•	. 100
4.6. The Parseval formula .	•	•	•	. 105
4.7. Theorems on resultants	•	•	•	. 106
4.8-9. Another extension of Plancherel's the	eorem	•	•	. 107
4.10. Another case of the Parseval formula		•	•	. 110
4.11. Failure of Theorems 75 and 79 for $p >$	> 2	•	•	. 111
4.12. Special conditions	•	•	•	. 113
4.13. Lipschitz conditions	•	•	•	. 115
4.14. Mellin transforms of the class L^p	•	•	•	. 118
V. CONJUGATE INTEGRALS: HI	LBER	T TRA	NSFOR	MS
5.1. Conjugate integrals				. 119
5.2-9. Hilbert transforms of the class L^2				. 121
5.10-13. Hilbert transforms of the class L^p				. 132
5.14. The case $p = 1$. 143
5.15. Lipschitz conditions				. 145
5.16. The allied integral				. 147
5.17. Application to Fourier transforms				. 148
5.18. Further cases of Parseval's formula				. 150
VI. UNIQUENESS AND MISCELL	ANEC	из тн	LEOREN	rs
6.1-6. Uniqueness of trigonometrical integra			213()1(13)	. 152
6.7. Integrals in the complex form .	2.1.7	•	•	104
6.8. Parseval's formula	•	•	•	. 164
6.9. Another uniqueness theorem .	•	•	•	3.00
6.10–12. Special properties of transforms	•	•	•	. 166
6.13. Order of magnitude of transforms	•	-	•	. 174
6.13. Order of magnitude of transforms	•	•	•	. 174
VII. EXAMPLES AND AI	PPLIC	ATION	īS	
7.1. Cosine transforms	•	•	•	. 177
7.2. Sine transforms	•	•	•	. 179
7.3. The Parseval formulae	•	•		. 180
7.4. Some Bessel-function examples				. 182
7.5. Some integrals of Ramanujan .		•	•	. 184
7.6. Some Γ-function formulae .	•	•		. 186
7.7. Mellin transforms	•	•	•	. 190
7.8. Further gamma-function formulae		•		. 193
70 Parel for ations				106

CONT	ENTS					v ii
7.10. Products of Bessel functions						198
7.11. Integrals involving Bessel func	tions	•	•	•	•	200
7.12. Some non-absolutely convergen		ra.la	•	•	•	204
7.13-14. Laplace transforms.	io micogi	Wis	•	•	•	206
	•	•	•	•	•	200
VIII. GENERAL T	RANSF	ORMA	TIONS			
8.1-3. Generalization of Fourier's for	mulae			•		212
8.4. Examples				•		214
8.5. L^2 theory						220
8.6. Proof of Theorems 129, 130				•		222
8.7. Proof of Theorem 131 .						223
8.8. Necessity of the conditions			•	•		225
8.9. The unsymmetrical formulae				•		226
8.10. A convergence theorem				•		227
8.11. The resultant of two Fourier ke	ernels					229
8.12–16. Convergence of k-integrals		•		•		232
8.17. Proof of Theorem 134 .			•			239
8.18. Hankel's theorem .						240
8.19. Formulae derived from Hankel	's theor	em				242
1X. SELF-RECIPR	OCAL	FUNC	TIONS			
9.1–3. Formalities						245
9.4. Functions of L^2	•	•	•	•	•	248
9.5-6. Functions of L^p	•	•	•	•	•	249
9.7. Analytic functions .	•	•	•	•	•	252
9.8. More general conditions	•	•	•	•	•	252
9.9. A general theorem .	•	•	•	•	•	255
9.10. Application	•	•	•	•	•	256
9.11. The second solution .	•	•	•	•	•	259
	•	•	•	•	•	260
9.12. Examples9.13. Lattice-point formulae .	•	•	•	•	•	265
9.14-17. Formulae connecting diffe	· ront c	·	of self-	· reciproca	. i	200
functions	aciio c	100000	or sen-		2.1	267
	•	•	•	•	•	
X. DIFFERENTIAL AND I	DIFFE	RENCE	E EQUA	TIONS		
10.1. Introduction						275
10.2-5. Ordinary differential equations	ţ.		•			276
10.6-15. Partial differential equations			•			281
10.16-17. Differential-difference equation						298
10.18. Difference equations .						301
-	E EOT	ATTON	re			
XI. INTEGRA	LEQU	ALIUN	io			
11.1. Introduction	•	•	•	•	•	303
11.2. The homogeneous equation	•	•	•	•	•	305
11.3. Examples	•	•	•	•	•	307
11.4. Various forms	•	•	•	•	•	311
11.5. The equation with finite limits	•	•	•	•	•	311
11.6. Another type	•	•	•	•	٠	314

viii

CONTENTS

11.7. Laplace's integral equation	•			316
11.8. Stieltjes's integral equation				317
11.9. Stieltjes's moment problem	•			320
11.10–11. Finite limits				322
11.12-13. Examples				328
11.14. Abel's integral equation		•		331
11.15. An equation of Fox		•		332
11.16. 'Dual' integral equations				334
11.17. The method of Hopf and Wiener .	•			339
11.18. An equation of A. C. Dixon				342
11.19. A problem of radiative equilibrium .	•			344
11.20. The limiting form of Milne's equation .	•			346
11.21. Bateman's equation				349
11.22-3. Kapteyn's equation				352
11.24. Solution of Kapteyn's equation .				355
11.25. A differential equation of fractional order				358
11.26. A probability problem				363
11.27. A problem in statistical dynamics .	•			367
BOOKS AND MONOGRAPHS				370
ORIGINAL MEMOIRS REFERRED TO IN T	не тех	ΚΤ .	•	371
INDEX				393

CONVERGENCE AND SUMMABILITY

1.1. Fourier's formulae. The origin of the theory of Fourier integrals is to be found in Fourier's Analytical Theory of Heat.† Fourier's argument, which would not now be called a proof, is substantially as follows. Suppose that a function f(x), of period $2\pi\lambda$, is represented by the Fourier series

$$f(x) = \frac{1}{2}a_0 + \sum_{n=1}^{\infty} \left(a_n \cos \frac{nx}{\lambda} + b_n \sin \frac{nx}{\lambda} \right).$$

The coefficients a_m , b_m are obtained formally by multiplying by $\cos(mx/\lambda)$ or $\sin(mx/\lambda)$, and integrating term-by-term over $(-\pi\lambda, \pi\lambda)$. This gives

$$a_m = rac{1}{\pi \lambda} \int\limits_{-\pi \lambda}^{\pi \lambda} f(t) ext{cos} rac{mt}{\lambda} \, dt, \qquad b_m = rac{1}{\pi \lambda} \int\limits_{-\pi \lambda}^{\pi \lambda} f(t) ext{sin} rac{mt}{\lambda} \, dt,$$

and the formula may be written

$$f(x) = \frac{1}{2\pi\lambda} \int_{-\pi\lambda}^{\pi\lambda} f(t) dt + \sum_{n=1}^{\infty} \frac{1}{\pi\lambda} \int_{-\pi\lambda}^{\pi\lambda} f(t) \cos \frac{n(x-t)}{\lambda} dt.$$

Putting $n/\lambda = u$, $1/\lambda = \delta u$, and making $\lambda \to \infty$, the sum passes formally into an integral, and we obtain

$$f(x) = \frac{1}{\pi} \int_{0}^{\infty} du \int_{-\infty}^{\infty} f(t) \cos u(x-t) dt.$$
 (1.1.1)

This is Fourier's integral formula.

It may also be written in the form (analogous to that of the Fourier series)

$$f(x) = \int_0^\infty \left\{ a(u)\cos xu + b(u)\sin xu \right\} du, \qquad (1.1.2)$$

where

$$a(u) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(t)\cos ut \, dt, \qquad b(u) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(t)\sin ut \, dt. \quad (1.1.3)$$

If f(t) is an even function, then

$$a(u) = \frac{2}{\pi} \int_{0}^{\infty} f(t) \cos ut \ dt,$$

† See list of books and monographs, pp. 370-1.

while b(u) vanishes; and the formula becomes

$$f(x) = \frac{2}{\pi} \int_{0}^{\infty} \cos xu \ du \int_{0}^{\infty} f(t) \cos ut \ dt. \tag{1.1.4}$$

This is Fourier's cosine formula. Similarly, if f(x) is odd, a(u) vanishes, and we obtain Fourier's sine formula,

$$f(x) = \frac{2}{\pi} \int_{0}^{\infty} \sin xu \ du \int_{0}^{\infty} f(t) \sin ut \ dt.$$
 (1.1.5)

We can also regard (1.1.1) as merely a combination of (1.1.4) and (1.1.5); for write

$$f(x) = \frac{1}{2} \{ f(x) + f(-x) \} + \frac{1}{2} \{ f(x) - f(-x) \} = g(x) + h(x),$$

so that g(x) is even and h(x) is odd. Then

$$\int_{-\infty}^{\infty} f(t)\cos u(x-t) dt$$

$$= 2\cos ux \int_{0}^{\infty} g(t)\cos ut dt + 2\sin ux \int_{0}^{\infty} h(t)\sin ut dt,$$

and (1.1.1) gives

$$g(x)+h(x)$$

$$=\frac{2}{\pi}\int_{0}^{\infty}\cos xu\ du\int_{0}^{\infty}g(t)\cos ut\ dt+\frac{2}{\pi}\int_{0}^{\infty}\sin xu\ du\int_{0}^{\infty}h(t)\sin ut\ dt,$$

i.e. the cosine formula for g(x) added to the sine formula for h(x).

The above formulae were discovered independently by Cauchy† in his researches on the propagation of waves. The formal basis given by Cauchy is as follows. The right-hand side of (1.1.1) is, formally, the limit as $\delta \to 0$ of

$$\frac{1}{\pi} \int_{0}^{\infty} e^{-\delta u} du \int_{-\infty}^{\infty} f(t) \cos u(x-t) dt = \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) dt \int_{0}^{\infty} e^{-\delta u} \cos u(x-t) du$$
$$= \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \frac{\delta}{\delta^{2} + (x-t)^{2}} dt.$$

The factor multiplying f(t) tends to 0 except when t = x. We should † Cauchy (1), (2); see list of original memoirs, pp. 371-87.

therefore expect the value of the integral to be unaltered if we replace f(t) by f(x); and this would give

$$\frac{f(x)}{\pi} \int_{-\infty}^{\infty} \frac{\delta}{\delta^2 + (x-t)^2} dt = f(x),$$

again verifying (1.1.1).

Another equivalent formula, given by Cauchy, is

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{-ixu} du \int_{-\infty}^{\infty} f(t)e^{iut} dt.$$
 (1.1.6)

Putting f(x) = g(x) + h(x), where g is even and h odd, as before,

$$\int_{-\infty}^{\infty} f(t)e^{iut} dt = 2\int_{0}^{\infty} g(t)\cos ut dt + 2i\int_{0}^{\infty} h(t)\sin ut dt,$$

and the right-hand side of (1.1.6) is

$$\frac{2}{\pi} \int_{0}^{\infty} \cos xu \, du \int_{0}^{\infty} g(t) \cos ut \, dt + \frac{2}{\pi} \int_{0}^{\infty} \sin xu \, du \int_{0}^{\infty} h(t) \sin ut \, dt$$
$$= g(x) + h(x) = f(x).$$

We shall call (1.1.6) the exponential form of Fourier's formula.

A formula of a slightly different type is obtained by expressing the outer integral in (1.1.1) as the limit of an integral over $(0,\lambda)$, and inverting the order of integration. The result is

$$f(x) := \lim_{\lambda \to \infty} \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \frac{\sin \lambda(x-t)}{x-t} dt.$$
 (1.1.7)

The same result may be obtained in the same way from (1.1.6). This formula is known as Fourier's single-integral formula.

1.2. Fourier transforms. It was pointed out by Cauchy that these formulae lead to reciprocal relations between pairs of functions. If we write

$$F_c(u) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(t) \cos ut \, dt, \qquad (1.2.1)$$

then (1.1.4) is

$$f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} F_c(u) \cos xu \ du, \qquad (1.2.2)$$

and the relation between f(x) and $F_c(x)$ is reciprocal. Such functions

were called by Cauchy reciprocal functions of the first kind. We shall call functions so related *Fourier cosine transforms* of each other. Thus

$$e^{-x}$$
, $\sqrt{\left(\frac{2}{\pi}\right)\frac{1}{1+x^2}}$

are a pair of Fourier cosine transforms.

Similarly, from Fourier's sine formula, we obtain

$$F_s(u) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty f(t) \sin ut \ dt, \qquad (1.2.3)$$

$$f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} F_s(u) \sin xu \ du. \tag{1.2.4}$$

These were called by Cauchy reciprocal functions of the second kind. We shall call them Fourier sine transforms of each other.

Thus e^{-x} , $\sqrt{\left(\frac{2}{\pi}\right)} \frac{x}{1+x^2}$ are Fourier sine transforms.

The formula (1.1.6) leads similarly to the unsymmetrical formulae

$$F(u) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(t)e^{iut} dt, \qquad (1.2.5)$$

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(u)e^{-ixu} du.$$
 (1.2.6)

We shall call such functions simply Fourier transforms of each other.

Thus

 $f(x) = e^{-|x|}, \qquad F(x) = \sqrt{\left(\frac{2}{\pi}\right)\frac{1}{1+x^2}}$

are Fourier transforms of each other.

If f(x) is even, $F(x) = F_c(x)$; if f(x) is odd, $F(x) = iF_s(x)$.

1.3. Generalized Fourier integrals. The existence of the integral defining F(u) implies a certain restriction on f(x) at infinity. Even if F(u) does not exist, the functions

$$F_{+}(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} f(t)e^{iwt} dt,$$
 (1.3.1)

$$F_{-}(w) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{0} f(t)e^{iwt} dt, \qquad (1.3.2)$$

where w = u+iv, may exist, the former for sufficiently large positive v, the latter for sufficiently large negative v. For

$$F_{+}(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} f(t)e^{-vt}e^{iut} dt,$$
 (1.3.3)

so that $F_{+}(w)$ is the transform of the function equal to $f(t)e^{-vt}$ for t>0, and to 0 for t<0. The formula reciprocal to (1.3.3) is

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F_{+}(u+iv)e^{-ixu} du = \begin{cases} f(x)e^{-vx} & (x>0) \\ 0 & (x<0), \end{cases}$$

or

$$\frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty}F_{+}(u+iv)e^{-ix(u+iv)}\,du=\left\{\begin{array}{ll}f(x)&(x>0)\\0&(x<0).\end{array}\right.$$

There is a similar formula involving F_{-} . Adding, we may write

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w)e^{-ixw} dw + \frac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} F_{-}(w)e^{-ixw} dw,$$
(1.3.4)

where a is a sufficiently large positive number, b a sufficiently large negative number.

For example, if $f(x) = e^x$, then

$$F_+(w) = -rac{1}{\sqrt{(2\pi)}}rac{1}{1+iw}, \qquad F_-(w) = rac{1}{\sqrt{(2\pi)}}rac{1}{1+iw}.$$

In this case (1.3.4) is at once verified by the calculus of residues.

In this form Fourier's integral formula may be applied to a periodic function. Let f(x) have the period 2π . Then for v > 0

$$\begin{split} F_+(w) &= \frac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty f(x) e^{ixw} \; dx = \frac{1}{\sqrt{(2\pi)}} \sum_{n=0}^\infty \int\limits_{2n\pi}^{2(n+1)\pi} f(x) e^{ixw} \; dx \\ &= \frac{1}{\sqrt{(2\pi)}} \sum_{n=0}^\infty \int\limits_0^{2\pi} f(\xi) e^{i(\xi+2n\pi)w} \; d\xi = \frac{1}{\sqrt{(2\pi)}} \int\limits_0^{2\pi} f(\xi) \frac{e^{i\xi w}}{1-e^{2\pi i w}} \; d\xi \\ &= \frac{1}{\sqrt{(2\pi)}} \frac{\phi(w)}{1-e^{2\pi i w}}, \\ \text{where} \qquad \qquad \phi(w) &= \int\limits_0^{2\pi} f(\xi) e^{i\xi w} \; d\xi. \end{split}$$

$$\dot{F}_{-}(w) = -\frac{1}{\sqrt{(2\pi)}} \frac{\phi(w)}{1 - e^{2\pi i w}} \quad (v < 0).$$

The reciprocal formula is therefore

$$f(x) = \frac{1}{2\pi} \int_{ia-\infty}^{ia+\infty} \frac{\phi(w)}{1 - e^{2\pi i w}} e^{-ixw} dw - \frac{1}{2\pi} \int_{ib-\infty}^{ib+\infty} \frac{\phi(w)}{1 - e^{2\pi i w}} e^{-ixw} dw.$$

Here $\phi(w)$ is an integral function. If it behaves at infinity so that we can evaluate the right-hand side by the calculus of residues in the obvious way, we obtain

$$f(x) = \frac{1}{2\pi} \sum_{n=-\infty}^{\infty} \phi(n) e^{-inx}.$$

We have thus returned to the Fourier series for f(x).

1.4. The formulae of Laplace. The formula

$$\phi(s) = \int_0^\infty f(x)e^{-sx} dx \qquad (1.4.1)$$

is known as Laplace's integral. If f(x) is the given function, $\phi(s)$ is in general analytic for $\mathbf{R}(s) > 0$. The reciprocal formula is

$$\frac{1}{2\pi i} \int_{s}^{k+i\infty} \phi(s)e^{sx} \, ds = \begin{cases} f(x) & (x>0) \\ 0 & (x<0). \end{cases}$$
 (1.4.2)

From a formal point of view the formulae are a particular case of those of § 1.2, as is seen on putting $s = \sigma + it$.

As a still more special case we obtain a reciprocity between two analytic functions. Let

$$f(x) = \sum_{n=0}^{\infty} a_n x^n,$$

and suppose that the integral (1.4.1) can be evaluated by term-by-term integration. Then

$$\phi(s) = \sum_{n=0}^{\infty} a_n \int_0^{\infty} e^{-sx} x^n dx = \sum_{n=0}^{\infty} \frac{n! a_n}{s^{n+1}},$$

$$\frac{1}{s} \phi\left(\frac{1}{s}\right) = \sum_{n=0}^{\infty} n! a_n s^n.$$

or

If f(x) is suitably restricted, $\phi(s)$ will be an analytic function regular in the neighbourhood of $s = \infty$; and, if C is a closed curve

surrounding the origin, but lying sufficiently far from it,

$$\frac{1}{2\pi i} \int_{C} \phi(w) e^{zw} dw = \frac{1}{2\pi i} \sum_{n=0}^{\infty} n! a_{n} \int_{C} \frac{e^{zn}}{w^{n+1}} dw$$

$$= \sum_{n=0}^{\infty} a_{n} z^{n} = f(z). \tag{1.4.3}$$

The function f(z) may therefore be represented as a trigonometrical integral, but now along a closed curve.

1.5. The formulae of Mellin. Still another pair of formulae embodying the same formal idea is given by

$$\mathfrak{F}(s) = \int_{0}^{\infty} f(x) x^{s-1} \, dx, \tag{1.5.1}$$

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \mathfrak{F}(s) x^{-s} ds. \tag{1.5.2}$$

The idea of such a reciprocity occurs in Riemann's famous memoir† on prime numbers. It was formulated explicitly by Cahen,‡ and the first accurate discussion was given by Mellin.|| We shall call the formulae *Mellin's inversion formulae*.

These formulae arise naturally in the theory of Dirichlet series in the following way. The particular case

$$\Gamma(s) = \int_{0}^{\infty} e^{-x} x^{s-1} dx, \qquad e^{-x} = \frac{1}{2\pi i} \int_{-i\pi}^{c+i\infty} \Gamma(s) x^{-s} ds \quad (c > 0)$$

is well known. Now let $\phi(s)$ be a function expressible as a Dirichlet series,

 $\phi(s) = \sum_{n=1}^{\infty} \frac{a_n}{n^s}.$

Then we have formally

$$\phi(s) = \sum_{n=1}^{\infty} \frac{a_n}{\Gamma(s)} \int_{0}^{\infty} e^{-nx} x^{s-1} dx = \frac{1}{\Gamma(s)} \int_{0}^{\infty} f(x) x^{s-1} dx,$$

where

$$f(x) = \sum_{n=0}^{\infty} a_n e^{-nx};$$

and reciprocally

$$\frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \phi(s) \Gamma(s) x^{-s} ds = \sum_{n=1}^{\infty} \frac{a_n}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Gamma(s) (nx)^{-s} ds$$
$$= \sum_{n=1}^{\infty} a_n e^{-nx} = f(x).$$

The forms (1.5.1), (1.5.2) are obtained by putting $\phi(s)\Gamma(s) = \mathfrak{F}(s)$.

Mellin's formulae may also be obtained by a substitution from the exponential form of Fourier's formula. In fact, putting $x = e^{\xi}$ and s = c + it, (1.5.1) becomes

$$\mathfrak{F}(c+it) = \int_{-\infty}^{\infty} f(e^{\xi}) e^{\xi(c+it)} d\xi,$$

and (1.5.2) becomes

$$f(e^{\xi}) = rac{1}{2\pi} \int\limits_{-\infty}^{\infty} \mathfrak{F}(c+it)e^{-\xi(c+it)} dt.$$

The functions

$$\sqrt{(2\pi)}e^{c\xi}f(e^{\xi}), \quad \mathfrak{F}(c+it)$$

are thus Fourier transforms of each other.

Suppose that, in Mellin's formulae, the function f(x) is analytic at the origin and in a region containing the positive real axis. Consider the integral

 $\int_{\Gamma} f(z)(-z)^{s-1} dz,$

where Γ is a loop coming from infinity on the positive real axis, encircling the origin in the positive direction, and returning to infinity. We define $(-z)^{s-1}$ as $e^{(s-1)\log(-z)}$, where $\log(-z)$ is real on the negative real axis.

Suppose Γ compressed into the real axis on both sides. The part of the integral above the real axis gives

$$-\int_{0}^{\infty} f(x)e^{(s-1)(\log x - i\pi)} dx = e^{-is\pi} \int_{0}^{\infty} f(x)x^{s-1} dx,$$

and that below the real axis gives

$$\int_0^\infty f(x)e^{(s-1)(\log x+i\pi)}\,dx = -e^{is\pi}\int_0^\infty f(x)x^{s-1}\,dx.$$
$$\int f(z)(-z)^{s-1}\,dz = -2i\sin s\pi\,\mathfrak{F}(s).$$

Hence

Let

$$\pi\chi(s) = \mathfrak{F}(s)\sin s\pi.$$

Then we obtain the reciprocal formulae

$$\chi(s) = -\frac{1}{2i\pi} \int_{\Gamma} f(z)(-z)^{s-1} dz, \qquad (1.5.3)$$

$$f(z) = \frac{1}{2i} \int_{c-i\infty}^{c+i\infty} \frac{\chi(s)z^{-s}}{\sin \pi s} ds.$$
 (1.5.4)

A simple example is $f(z) = e^{-z}$, $\chi(s) = 1/\Gamma(1-s)$. Such formulae have important applications in the theory of functions of a complex variable,† but we cannot consider them further here.

1.6. For the early history of the Fourier-Cauchy formulae we may refer to the article by Burkhardt in the *Encyklopädie*.

The theorems of this chapter are in the main analogous to classical theorems in the theory of Fourier series. We do not actually assume a knowledge of the theory of Fourier series, though the reader will presumably be familiar with it. Almost all theorems on Fourier series have some sort of analogue for integrals. In some cases the theorems are so similar that the extension from series to integrals is hardly worth making. In other cases there are new points of interest in the integral case, which is even sometimes the simpler.

1.7. Notation. We use

$$\int_{0}^{\infty} f(x) \ dx$$

to denote the Lebesgue integral of f(x) over $(0,\infty)$ in the strict sense, implying that the integral is absolutely convergent, i.e. that

$$\int_{0}^{\infty} |f(x)| \ dx$$

also exists. If f(x) is integrable over (0, X) for every X, and

$$\lim_{X\to\infty}\int\limits_0^X f(x)\ dx$$

exists, we denote the limit by

$$\int_{0}^{\infty} f(x) \ dx.$$

† Carlson (1).

Such an integral is known as a Cauchy integral. A similar notation is used in the case of other limits. Thus

$$\int_{-\infty}^{1} f(x) dx$$

$$\int_{x}^{1} f(x) dx$$

denotes the limit of

as $\delta \to 0$ through positive values.

In 'formal' analysis we use $\int_0^\infty f(x) dx$ to denote that the integral exists in some sense or other. There is generally little risk of confusion between this and the Lebesgue sense.

We say that f(x) belongs to, or is, $L^p(a, b)$ if f(x) is measurable and

$$\int_a^b |f(x)|^p dx < \infty.$$

We write L for L^1 .

 $\mathbf{B}\mathbf{y}$

l.i.m.
$$\int_{X\to\infty}^X f(x,\alpha) dx$$

(limit in mean) we denote a function $\phi(\alpha)$ such that

$$\lim_{X\to\infty}\int_{a}^{b}\left|\phi(\alpha)-\int_{0}^{X}f(x,\alpha)\;dx\right|^{p}d\alpha=0,$$

a, b, and p having prescribed values.

As complex variables we use

$$z = x+iy$$
, $w = u+iv$, $s = \sigma+it$, $\zeta = \xi+i\eta$.

If f(x) is a given function, we denote by

$$F(x)$$
, $F_c(x)$, $F_s(x)$, $F_+(w)$, $F_-(w)$, $\mathfrak{F}(s)$,

the functions defined in (1.2.5), (1.2.1), (1.2.3), (1.3.1), (1.3.2), (1.5.1) respectively. In each case it is assumed that the integral referred to exists in some sense or other. The ambiguity of the expression 'a Fourier transform', arising from the asymmetry of the formulae (1.2.5), (1.2.6), is avoided by standardizing the use of small and capital letters as in these formulae.

Similarly with other letters $(g, G, G_+, G_-, \text{ etc.})$.

We denote by A an absolute constant, not necessarily the same one at each occurrence; K is used in a similar way for a constant depending on the data of the problem in hand.

We say that the convergence of a sequence $f_n(x)$ to a limit f(x) is bounded if $|f_n(x)| \leq K$ for all n and x; and that it is dominated if $|f_n(x)| \leq \phi(x)$, where $\phi(x)$ is L over a prescribed set. It is known that $\lim_{n \to \infty} \int f_n(x) \, dx = \int f(x) \, dx$

if the convergence is bounded or dominated.

1.8. Fundamental theorems. The theorem of Riemann-Lebesgue is fundamental in the theory of Fourier integrals, as it is in the theory of Fourier series. We shall state it as follows.

THEOREM 1. Let f(x) belong to $L(-\infty,\infty)$. Then the integrals

$$\int_{-\infty}^{\infty} f(x)\cos \lambda x \, dx, \qquad \int_{-\infty}^{\infty} f(x)\sin \lambda x \, dx, \qquad (1.8.1)$$

tend to zero as $\lambda \to \infty$.

Consider the cosine integral. Let ϵ be a given positive number. Then we can choose X so large that

$$\int_{X}^{\infty} |f(x)| dx < \epsilon, \qquad \int_{-\infty}^{-X} |f(x)| dx < \epsilon.$$

$$\left| \int_{Y}^{\infty} f(x) \cos \lambda x \, dx \right| < \epsilon, \qquad \left| \int_{-\infty}^{-X} f(x) \cos \lambda x \, dx \right| < \epsilon$$

for all values of λ .

Next, we can define a function $\phi(x)$, absolutely continuous in the interval (-X, X), such that

$$\int_{-X}^{X} |f(x) - \phi(x)| \, dx < \epsilon.$$

$$\left| \int_{X}^{X} \{f(x) - \phi(x)\} \cos \lambda x \, dx \right| < \epsilon.$$

Then

for all values of λ . Finally

$$\int_{-X}^{X} \phi(x) \cos \lambda x \, dx$$

$$= \frac{\phi(X) \sin \lambda X}{\lambda} + \frac{\phi(-X) \sin \lambda X}{\lambda} - \frac{1}{\lambda} \int_{-X}^{X} \phi'(x) \sin \lambda x \, dx,$$

and (for a fixed X) we can choose λ_0 so large that the modulus of this † Titchmarsh, Theory of Functions, §§ 10.5, 10.8.

is less than ϵ for $\lambda > \lambda_0$. Then

$$\left| \int_{-\infty}^{\infty} f(x) \cos \lambda x \, dx \right| < 4\epsilon \quad (\lambda > \lambda_0).$$

This proves the theorem for the cosine integral; a similar proof applies to the sine integral.

Theorem 2. Let f(x) belong to $L(-\infty,\infty)$. Then a necessary and sufficient condition that

$$\frac{1}{\pi} \int_{0}^{\infty} du \int_{-\infty}^{\infty} f(t) \cos u(x-t) dt = a$$
 (1.8.2)

is that, for any fixed δ ,

$$\lim_{\lambda \to \infty} \int_{0}^{\delta} \{f(x+y) + f(x-y) - 2a\} \frac{\sin \lambda y}{y} \, dy = 0.$$
 (1.8.3)

Since $|f(t)\cos u(x-t)| \leq |f(t)|$, the integral

$$\int_{-\infty}^{\infty} f(t) \cos u(x-t) \ dt$$

converges uniformly with respect to u over any finite interval. Hence

$$\int_{0}^{\lambda} du \int_{-\infty}^{\infty} f(t) \cos u(x-t) dt = \int_{-\infty}^{\infty} f(t) dt \int_{0}^{\lambda} \cos u(x-t) du$$

$$= \int_{-\infty}^{\infty} f(t) \frac{\sin \lambda(x-t)}{x-t} dt.$$

Since f(t)/(x-t) is integrable over $(-\infty, x-\delta)$ and $(x+\delta, \infty)$, it follows from the Riemann-Lebesgue theorem that, for a fixed δ ,

$$\lim_{\lambda \to \infty} \int_{-\infty}^{x-\delta} f(t) \frac{\sin \lambda(x-t)}{x-t} dt = 0, \qquad \lim_{\lambda \to \infty} \int_{x+\delta}^{\infty} f(t) \frac{\sin \lambda(x-t)}{x-t} dt = 0.$$
Also
$$\int_{x-\delta}^{x+\delta} f(t) \frac{\sin \lambda(x-t)}{x-t} dt = \int_{0}^{\delta} \{f(x+y) + f(x-y)\} \frac{\sin \lambda y}{y} dy,$$
and
$$\lim_{\lambda \to \infty} \int_{0}^{\delta} 2a \frac{\sin \lambda y}{y} dy = \lim_{\lambda \to \infty} 2a \int_{0}^{\lambda \delta} \frac{\sin v}{v} dv = 2a \int_{0}^{\infty} \frac{\sin v}{v} dv = a\pi.$$

These equations together show that (1.8.2) and (1.8.3) are equivalent.

1.9. We are now in a position to extend all the ordinary convergence tests for Fourier series to Fourier integrals. We shall, however, content ourselves with proving the two following theorems, corresponding to the tests of Jordan and Dini respectively.

THEOREM 3. Let f(t) belong to $L(-\infty,\infty)$. If f(t) is of bounded variation in an interval including the point x, then

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \frac{1}{\pi} \int_{0}^{+\infty} du \int_{-\infty}^{\infty} f(t)\cos u(x-t) dt. \quad (1.9.1)$$

If f(t) is continuous and of bounded variation in an interval (a,b), then

$$f(x) = \frac{1}{\pi} \int_{0}^{+\infty} du \int_{-\infty}^{\infty} f(t) \cos u(x-t) dt, \qquad (1.9.2)$$

the integral converging uniformly in any interval interior to (a, b).

Let
$$\psi(y) = f(x+y) + f(x-y) - f(x+0) - f(x-0)$$
.

Then $\psi(y)$ is of bounded variation over $(0, \delta)$, if δ is small enough, and $\psi(y) \to 0$ as $y \to 0$. We may therefore write

$$\psi(y) = \psi_1(y) - \psi_2(y),$$

where $\psi_1(y)$ and $\psi_2(y)$ are positive non-decreasing bounded functions in $(0, \delta)$, which tend to 0 as $y \to 0$.

Given any positive number ϵ , there is a number η such that $\psi_1(y) \leqslant \epsilon$ for $y \leqslant \eta$. Let

$$\int_0^\delta \psi_1(y) \frac{\sin \lambda y}{y} \, dy = \int_0^\eta \psi_1(y) \frac{\sin \lambda y}{y} \, dy + \int_\eta^\delta \psi_1(y) \frac{\sin \lambda y}{y} \, dy.$$

By the second mean-value theorem, the first part is equal to

$$\psi_1(\eta) \int_{\xi}^{\eta} \frac{\sin \lambda y}{y} dy = \psi_1(\eta) \int_{\lambda \xi}^{\lambda \eta} \frac{\sin v}{v} dv \quad (0 < \xi < \eta),$$

and the last integral is bounded for all λ and ξ . Hence

$$\left|\int\limits_0^{\eta}\psi_1(y)\frac{\sin\lambda y}{y}\,dy\right|< A\epsilon,$$

for all values of λ . Having fixed η , $\psi_1(y)/y$ is integrable over (η, δ) , so that

$$\lim_{\lambda\to\infty}\int_{y}^{b}\psi_{1}(y)\frac{\sin\lambda y}{y}\,dy=0.$$

Since ϵ is arbitrary, it follows that

$$\lim_{\lambda\to\infty}\int\limits_0^\delta\psi_1(y)\frac{\sin\lambda y}{y}\,dy=0.$$

Similarly, the integral involving $\psi_2(y)$ tends to 0. This proves the first clause of the theorem.

If f(x) is continuous in (a,b), $\frac{1}{2}\{f(x+0)+f(x-0)\}=f(x)$; and, the function being uniformly continuous in any interval interior to (a,b), the conditions used in the proof hold uniformly, and so the convergence is uniform.

THEOREM 4. Let f(t) belong to $L(-\infty,\infty)$. Then, for a given x, (1.9.2) is true if

$$\int_{x}^{\delta} \left| \frac{f(x+y) + f(x-y) - 2f(x)}{y} \right| dy$$
 (1.9.3)

exists for some positive δ ; in particular it holds if f(x) is differentiable at the point x.

This follows at once from Theorems 1 and 2, with a = f(x). If f(x) is differentiable, the integrand in (1.9.3) is bounded, so that the condition is plainly satisfied.

THEOREM 5.† Let f(t)/(1+|t|) belong to $L(-\infty,\infty)$; let

$$a_1(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(y) \frac{\sin xy}{y} dy,$$
 (1.9.4)

$$b_1(x) = \frac{1}{\pi} \int_{-1}^{1} f(y) \frac{1 - \cos xy}{y} \, dy - \frac{1}{\pi} \left(\int_{-\infty}^{-1} + \int_{1}^{\infty} \right) f(y) \frac{\cos xy}{y} \, dy$$
(1.9.5)

be absolutely continuous over any finite interval $0 < \delta \le x \le \Delta$, and let a(x), b(x) be their respective derivatives. Let f(t) satisfy the conditions of Theorem 3 or Theorem 4 in the neighbourhood of t = x. Then

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \int_{-\infty}^{+\infty} \{a(u)\cos xu + b(u)\sin xu\} du.$$

Suppose first that f(x) = 0 for $|x| \ge 1$. Then

$$\frac{1}{\pi} \int_{0}^{x} d\xi \int_{-1}^{1} f(y) \cos \xi y \, dy = \frac{1}{\pi} \int_{-1}^{1} f(y) \frac{\sin xy}{y} \, dy = a_{1}(x),$$

† Hahn (2).

so that

$$a(x) = \frac{1}{\pi} \int_{-1}^{1} f(y) \cos xy \ dy$$

almost everywhere. Similarly,

$$b(x) = \frac{1}{\pi} \int_{-1}^{1} f(y) \sin xy \ dy$$

almost everywhere. The result then follows from Theorem 3 or Theorem 4.

Suppose next that f(x) = 0 for |x| < 1. Then f(x)/x belongs to $L(-\infty,\infty)$. Hence, by Theorem 3 or 4,

$$\frac{1}{2x}\{f(x+0)+f(x-0)\} = \frac{1}{\pi} \int_{0}^{+\infty} \{-b_1(u)\cos xu + a_1(u)\sin xu\} du.$$

Now

$$\begin{split} \int\limits_0^\infty b_1(u) \cos x u \; du &= \left[b_1(u) \frac{\sin x u}{x}\right]_0^\infty - \frac{1}{x} \int\limits_{-0}^\infty b(u) \sin x u \; du \\ &= -\frac{1}{x} \int\limits_{-0}^\infty b(u) \sin x u \; du, \end{split}$$

since $b_1(u) \to 0$ as $u \to \infty$.

Also

$$\int_{0}^{\infty} a_{1}(u)\sin xu \ du = \left[-a_{1}(u)\frac{\cos xu}{x}\right]_{0}^{\infty} + \frac{1}{x}\int_{-0}^{\infty} a(u)\cos xu \ du$$
$$= \frac{1}{x}\int_{-0}^{+\infty} a(u)\cos xu \ du,$$

since $a_1(u)$ tends to 0 as $u \to 0$ or $u \to \infty$. The result in this case thus follows.

The general result now follows by adding functions of the two classes considered.

1.10. Monotonic functions.† The next theorem is based on the fact that, even if $\int_0^\infty f(t) dt$ does not exist, the integrals

$$\int_{0}^{\infty} f(t)\cos ut \ dt, \qquad \int_{0}^{\infty} f(t)\sin ut \ dt$$

exist for u > 0 provided that $f(t) \to 0$ steadily as $t \to \infty$. Here it † Pringsheim (1).

seems slightly more convenient to take the cosine and sine integrals separately.

THEOREM 6. Let f(t) be non-increasing over $(0, \infty)$, integrable over any finite interval beginning at 0, and let $f(t) \to 0$ as $t \to \infty$. Then for any positive x

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \frac{2}{\pi} \int_{-\infty}^{\infty} \cos xu \ du \int_{0}^{\infty} f(t)\cos ut \ dt.$$

We have, by the second mean-value theorem,

$$\left|\int_{T}^{T'} f(t) \cos ut \ dt\right| = \left|f(T+0) \int_{T}^{T'} \cos ut \ dt\right| \leqslant \frac{2f(T+0)}{u}.$$

Hence the t-integral converges uniformly with respect to u over $0 < \lambda \leq u \leq \mu$. Hence

$$\int_{\lambda}^{\mu} \cos xu \ du \int_{0}^{+\infty} f(t)\cos ut \ dt = \int_{0}^{+\infty} f(t) \ dt \int_{\lambda}^{\mu} \cos xu \cos ut \ du$$

$$= \frac{1}{2} \int_{0}^{+\infty} f(t) \left\{ \frac{\sin \mu(x-t)}{x-t} - \frac{\sin \lambda(x-t)}{x-t} + \frac{\sin \mu(x+t)}{x+t} - \frac{\sin \lambda(x+t)}{x+t} \right\} dt.$$

Now

$$\left|\int_{T}^{\infty} f(t) \frac{\sin \mu(x-t)}{x-t} dt\right| = \left|f(T+0) \int_{T}^{T'} \frac{\sin \mu(x-t)}{x-t} dt\right| < Af(T+0),$$

and similarly for the integrals involving λ and x+t. We can therefore choose T so large that

$$\left|\int_{a}^{\infty} f(t) \left\{ \frac{\sin \mu(x-t)}{x-t} - \ldots \right\} dt \right| < \epsilon$$

for $T > T_0(\epsilon)$, for all values of λ and μ . Having fixed T > x,

$$\lim_{\mu \to \infty} \int_{0}^{T} f(t) \frac{\sin \mu(x-t)}{x-t} dt = \frac{1}{2} \pi \{ f(x+0) + f(x-0) \}$$

by the analysis of Theorem 3, and

$$\lim_{\mu \to \infty} \int_{0}^{T} f(t) \frac{\sin \mu(x+t)}{x+t} dt = 0$$

by the Riemann-Lebesgue theorem. Also

$$\left| \int_{0}^{T} f(t) \frac{\sin \lambda(x-t)}{x-t} dt \right| \leqslant \lambda \int_{0}^{T} f(t) dt \to 0$$

as $\lambda \to 0$; and similarly for the remaining part.

THEOREM 7. If f(t) satisfies the conditions of Theorem 6, then for any positive $x \rightarrow \infty$

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \frac{2}{\pi} \int_{0}^{\infty} \sin xu \ du \int_{0}^{\infty} f(t)\sin ut \ dt.$$
As $u \to 0$

$$\left| \int_{1}^{+\infty} f(t)\sin ut \ dt \right| = \left| f(1+0) \int_{1}^{T} \sin ut \ dt \right|$$

$$= f(1+0) \left| \frac{\cos u - \cos u}{u} \right| \leqslant \frac{2f(1+0)}{u},$$
and
$$\int_{1}^{1} f(t)\sin ut \ dt = O(1).$$

Hence the u-integral is absolutely convergent at the lower limit. Apart from this, the proof is the same as that of Theorem 6.

Fourier's formulae may be established under still more general conditions by adding a function of the type of Theorem 3 to one of the type of Theorem 6. The results of this process are sufficiently obvious.

1.11. Functions containing a periodic factor.†

THEOREM 8. Let $f(t) = g(t)\cos at$ (a > 0), where g(t) is non-increasing, integrable over (0,1), and $g(t) \to 0$ as $t \to \infty$. Then for any positive x

$$\frac{1}{2}\left\{f(x+0)+f(x-0)\right\} = \frac{2}{\pi}\left(\int_{0}^{\pi} + \int_{-\pi}^{\infty}\right)\cos xu \ du \int_{0}^{\pi} f(t)\cos ut \ dt.$$

The inner integral is

$$\int_{0}^{\infty} g(t)\cos at \cos ut \ dt,$$

which is uniformly convergent over any finite interval not including or ending at u = a. We may therefore invert the integral

$$\int_{0}^{a-\delta} \cos xu \ du \int_{0}^{+\infty} g(t) \cos at \cos ut \ dt$$
† Pringsheim (1).

4362

for every $\delta > 0$. To prove that

$$\int_{0}^{a} \cos xu \ du \int_{0}^{\infty} g(t) \cos at \cos ut \ dt = \int_{0}^{\infty} ... \int_{0}^{a} ...$$

it is therefore sufficient to prove that

$$\lim_{\delta \to 0} \int_{0}^{+\infty} g(t) \cos at \ dt \int_{a-\delta}^{a} \cos xu \cos ut \ du = 0.$$

This is clearly true for the part $0 \le t \le T$, with any finite T. It is therefore sufficient to prove that

$$\lim_{\delta \to 0} \int_{T}^{-\infty} g(t) \cos at \ dt \int_{a-\delta}^{a} \cos xu \cos ut \ du = 0,$$

or

$$\lim_{\delta \to 0} \int_{T}^{\infty} g(t) \cos at \left\{ \frac{\sin a(x-t) - \sin(a-\delta)(x-t)}{x-t} + \frac{\sin a(x+t) - \sin(a-\delta)(x+t)}{x+t} \right\} dt = 0.$$

Clearly, if T > x,

$$\int_{T}^{\infty} g(t)\cos at \left\{\sin a(x-t) - \sin(a-\delta)(x-t)\right\} \left(\frac{1}{x-t} + \frac{1}{t}\right) dt \to 0,$$

since this integral converges uniformly with respect to δ . Similarly for the integral involving x+t. Hence it is sufficient to consider

$$\int_{T}^{\infty} \frac{g(t)\cos at}{t} \{\sin a(x+t) - \sin(a-\delta)(x+t) - \sin a(x-t) + \sin(a-\delta)(x-t)\} dt$$

$$= 2 \int_{T}^{\infty} \frac{g(t)\cos at}{t} \{\cos ax \sin at - \cos(a-\delta)x \sin(a-\delta)t\} dt.$$

Now

$$2\int_{T}^{\infty} \frac{g(t)}{t} \cos at \sin(a-\delta)t \ dt = \int_{T}^{\infty} \frac{g(t)}{t} \{\sin(2a-\delta)t - \sin \delta t\} \ dt,$$

which converges uniformly with respect to δ as $\delta \to 0$, since

$$\int_{T_1}^{T_2} \frac{g(t)}{t} \sin \delta t = O\left\{g(T_1) \int_{T_1}^{T_2} \frac{\sin \delta t}{t} dt\right\} = O\left\{g(T_1)\right\}.$$

Hence

$$\int_{T}^{\infty} \frac{g(t)}{t} \cos at \sin(a-\delta)t \ dt \to \int_{T}^{\infty} \frac{g(t)}{t} \cos at \sin at \ dt,$$

and the result follows.

A similar argument applies to the integral over $(a+\delta,\lambda)$. It therefore follows that

$$\left(\int_{0}^{-a} + \int_{-a}^{\lambda}\right) \cos xu \, du \int_{0}^{-\infty} f(t) \cos ut \, dt$$

$$= \frac{1}{2} \int_{0}^{+\infty} f(t) \left\{ \frac{\sin \lambda(x-t)}{x-t} + \frac{\sin \lambda(x+t)}{x+t} \right\} dt.$$

Finally

I many
$$\int_{T}^{T'} \cos at \frac{\sin \lambda (x-t)}{x-t} dt = \int_{T-x}^{T'-x} \cos a(x+y) \frac{\sin \lambda y}{y} dy$$

$$= \cos ax \int_{T-x}^{T'-x} \frac{\cos ay \sin \lambda y}{y} dy - \sin ax \int_{T-x}^{T'-x} \frac{\sin ay \sin \lambda y}{y} dy$$

is bounded for fixed a and x, T > 2x, $\lambda > 2a$. Hence, by the second mean-value theorem,

$$\int_{T}^{+\infty} g(t)\cos at \frac{\sin \lambda(x-t)}{x-t} dt = O\{g(T)\},$$

and the proof concludes as in Theorem 6.

THEOREM 9. Let $f(t) = g(t)\sin at$ (a > 0), where g(t) satisfies the same conditions as in Theorem 8. Then for any positive x

$$\frac{1}{2}\left\{f(x+0)+f(x-0)\right\} = \frac{2}{\pi}\lim_{\delta\to 0}\left(\int_{0}^{a-\delta}+\int_{a+\delta}^{-\infty}\right)\cos xu\ du\int_{0}^{-\infty}f(t)\cos ut\ dt.$$

If, in addition,
$$\int_{x}^{\infty} \frac{g(x)}{x} dx \text{ exists, then}$$

$$\frac{1}{2} \{ f(x+0) + f(x-0) \} = \frac{2}{\pi} \left(\int_{0}^{\pi} dx + \int_{-\pi}^{+\infty} dx \right) \cos xu du \int_{0}^{+\infty} f(t) \cos ut dt.$$

Proceeding as in Theorem 8, we find that, in the first repeated integral, the integral over $0 \le u \le \lambda$ ($\lambda > a$) may be inverted if

$$\int_{T}^{\infty} \frac{g(t)\sin at}{t} \{\cos(a+\delta)x\sin(a+\delta)t - \cos(a-\delta)x\sin(a-\delta)t\} dt \to 0$$

as $\delta \to 0$, i.e.

$$\cos ax \cos \delta x \int_{T}^{+\infty} \frac{g(t)}{t} \sin at \cos at \sin \delta t dt - \\ -\sin ax \sin \delta x \int_{T}^{+\infty} \frac{g(t)}{t} \sin^{2}at \cos \delta t dt \to 0.$$

The first integral $\rightarrow 0$ by uniform convergence. In the second,

$$\int\limits_{T}^{\infty}\frac{g(t)}{t}\cos 2at\cos \delta t\ dt$$

is uniformly convergent, and so tends to a finite limit; and

$$\int_{T}^{+\infty} \frac{g(t)}{t} \cos \delta t \ dt = O\left(\int_{T}^{1/\delta} \frac{dt}{t}\right) + O\left\{\delta g\left(\frac{1}{\delta}\right) \int_{1/\delta}^{\xi} \cos \delta t \ dt\right\}$$
$$= O(\log 1/\delta) + O\{g(1/\delta)\} = O(\log 1/\delta),$$

and $\sin \delta x \log 1/\delta \to 0$. This proves the first part.

In the second part of the theorem we have to consider

$$\int_{T}^{\infty} \frac{g(t)\sin at}{t} \{\cos ax \sin at - \cos(a-\delta)x \sin(a-\delta)t\} dt,$$

and this involves uniformly convergent terms, as before, and terms involving $_{\infty}$

 $\int_{T}^{\infty} \frac{g(t)}{t} dt.$

This proves the second part.

There is also a similar pair of theorems in which sines and cosines are interchanged.

Theorem 10.† Let f(t) = g(t)h(t), where g(t) is ultimately steadily decreasing to zero, g(t)/(1+|t|) belongs to $L(-\infty,\infty)$, and h(t) is periodic (with period a) and integrable over a period. Let f(t) be of bounded variation, or satisfy one of the alternative conditions in the neighbourhood of the point t=x. Then Fourier's formula holds in the sense that

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \sum_{n=0}^{\infty} \frac{1}{\pi} \int_{-2n\pi/a}^{-(2n+2)\pi/a} du \int_{--\infty}^{-\infty} f(t)\cos u(x-t) dt.$$
† Hahn (2).

If g(t) is steadily decreasing,

$$\int_{na}^{(n+1)a} \frac{|f(t)|}{1+t} dt = \int_{na}^{(n+1)a} \frac{|g(t)h(t)|}{1+t} dt$$

$$\leq \frac{g(na)}{1+na} \int_{na}^{(n+1)a} |h(t)| dt < K \frac{g(na)}{1+na} < K \int_{(n-1)a}^{na} \frac{g(t)}{1+t} dt.$$

Hence f(t)/(1+|t|) belongs to $L(-\infty,\infty)$. Hence, as in Theorem 5,

$$\frac{1}{2x}\{f(x+0)+f(x+0)\} = \frac{1}{\pi} \int_{0}^{+\infty} \{-b_1(u)\cos xu + a_1(u)\sin xu\} du,$$

where $a_1(u)$, $b_1(u)$ are defined by (1.9.4), (1.9.5).

Also
$$\sum_{\nu=m}^{n} \int_{\nu a}^{(\nu+1)a} h(x)e^{ixy} dx = \sum_{\nu=m}^{n} \int_{0}^{a} h(x)e^{i(\nu a + x)y} dx$$
$$= \frac{e^{imay} - e^{i(n+1)ay}}{1 - e^{iay}} \int_{0}^{a} h(x)e^{ixy} dx,$$

which is bounded in any interval not containing one of the points $y = 0, \pm \frac{2\pi}{a}, \pm \frac{4\pi}{a}, \dots$ Hence the integrals

$$\int_{x}^{x_{1}} h(x) \frac{\cos}{\sin} xy \ dx$$

are bounded, for all x_1 and x_2 , in any such interval of values of y.

It then follows from the second mean-value theorem that the integrals $\rightarrow \infty$

$$\frac{1}{\pi} \int_{-\infty}^{+\infty} g(x)h(x) \cos xy \ dx$$

are uniformly convergent in any such interval, to a(y), b(y), say; $a_1(y)$, $b_1(y)$ are the integrals of a(y), b(y) in the interior of such an interval; and

$$\int_{2n\pi/a}^{(2n+2)\pi/a} b_1(u)\cos xu \ du$$

$$= \left[b_1(u)\frac{\sin xu}{x}\right]_{2n\pi/a}^{(2n+2)\pi/a} - \frac{1}{x} \int_{-2n\pi/a}^{-(2n+2)\pi/a} b(u)\sin xu \ du,$$

and similarly for a(u).

Also a_1 and b_1 are continuous, so that, on summation, all the integrated terms cancel, and the result follows.

1.12. Oscillating functions. In the above theorems our conditions on f(x) are mostly restrictions on its oscillations. We shall next obtain a case of Fourier's theorem which depends on the oscillations of f(x) being sufficiently rapid, provided they are of a regular kind.

THEOREM 11.† Let $f(t) = \phi(t)\cos\psi(t)$ or $\phi(t)\sin\psi(t)$, where $\phi(t)$ is integrable over any finite interval, continuous and of bounded variation in any interval not containing the origin, and ultimately monotonic. Let $\psi(t)$ be twice differentiable, $\psi'(t)$ and $\psi'(t)/\phi(t)$ ultimately increasing steadily to infinity, and

$$\phi(t) = o\{t \sqrt{\psi''(t)}\}. \tag{1.12.1}$$

Then

$$f(x) = \frac{2}{\pi} \int_{0}^{+\infty} \frac{\cos xu}{\sin xu} \, du \int_{0}^{+\infty} f(t) \frac{\cos ut}{\sin ut} \, dt$$
 (1.12.2)

if (i) $\psi''(t)$ is non-decreasing, $\psi''(t+1) = O\{\psi''(t)\}, \ \phi(t+1) = O\{\phi(t)\},\ or (ii) \ \psi''(t)$ is decreasing, $t\psi''(t) > K, \ \phi(2t) = O\{\phi(t)\}.$

We use the following lemma.

Lemma. If $\phi(t)/h'(t)$ is monotonic, and g(t) steadily decreasing, then

$$\int_{a}^{b} \phi(t)g(t) \cos h(t) dt = O\left\{g(a) \max \frac{\phi(t)}{h'(t)}\right\}.$$

Using the second mean-value theorem repeatedly, we have, if $\phi(t)/h'(t)$ increases,

$$\begin{split} \int_a^b \phi(t)g(t)\cos h(t) \; dt &= \int_a^b \frac{\phi(t)}{h'(t)} g(t)h'(t)\cos h(t) \; dt \\ &= \frac{\phi \; (b)}{h'(b)} \int_\alpha^b g(t)h'(t)\cos h(t) \; dt \quad (a < \alpha < b) \\ &= \frac{\phi \; (b)}{h'(b)} g(\alpha) \int_\alpha^\beta h'(t)\cos h(t) \; dt \quad (\alpha < \beta < b) \\ &= \frac{\phi \; (b)}{h'(b)} g(\alpha) \{\sin h(\beta) - \sin h(\alpha)\}, \end{split}$$

and similarly for the other cases. Hence the result.

† Suggested by Landau, Vorlesungen über Zahlentheorie, Satz 413.

The inner integral in (1.12.2) is convergent if

$$\lim_{T,T\to\infty}\int_{T}^{T'}\phi(t)\cos\{\psi(t)\pm ut\}\,dt=0. \tag{1.12.3}$$

Now

$$\frac{\phi(t)}{\psi'(t)\pm u} = \frac{\phi(t)}{\psi'(t)} \frac{\psi'(t)}{\psi'(t)\pm u},$$

and the first factor tends steadily to 0, while the second factor is

$$1\pm\frac{u}{\psi'(t)\pm u}$$

and the last term is steadily decreasing in absolute value. Hence (1.12.3) follows from the lemma; and the convergence is plainly uniform over any finite range of values of u. Hence

$$\int_{0}^{\lambda} \cos xu \ du \int_{0}^{\infty} f(t) \cos ut \ dt = \int_{0}^{\infty} f(t) \ dt \int_{0}^{\lambda} \cos xu \cos ut \ du.$$

As in previous cases, it is now sufficient to prove that

$$\lim_{\lambda \to \infty} \int_{T}^{\infty} f(t) \frac{\sin \lambda (x-t)}{x-t} dt = 0$$

for a sufficiently large T.

Take Case (i), and suppose that $\phi(t)$ is non-decreasing for $t \geqslant T$, and consider, e.g.

$$\int_{T}^{\infty} \frac{\phi(t)}{t-x} \cos\{\psi(t) + \lambda(x-t)\} dt = \int_{T}^{\frac{1}{2}t_{0}} + \int_{\frac{1}{2}t_{0}}^{t_{0}-\delta} + \int_{t_{0}-\delta}^{t_{0}+\delta} + \int_{t_{0}+\delta}^{\infty} = J_{1} + J_{2} + J_{3} + J_{4},$$

where $\psi'(t_0) = \lambda$. Now $\phi(t)/\{\lambda - \psi'(t)\}$ is steadily increasing for $t < t_0$; hence, by the lemma,

$$J_{1} = O\left\{\frac{\phi(\frac{1}{2}t_{0})}{\psi'(t_{0}) - \psi'(\frac{1}{2}t_{0})}\right\} = O\left\{\frac{\phi(\frac{1}{2}t_{0})}{t_{0}\psi''(\frac{1}{2}t_{0})}\right\} = o(1),$$

$$J_{2} = O\left\{\frac{\phi(t_{0} - \delta)}{t_{0}\{\psi'(t_{0}) - \psi'(t_{0} - \delta)\}}\right\} = O\left\{\frac{\phi(t_{0})}{t_{0}\delta\psi''(t_{0} - \delta)}\right\} = O\left\{\frac{\phi(t_{0})}{t_{0}\delta\psi''(t_{0} - \delta)}\right\}$$

$$(1.12.4)$$

provided that $\delta = O(1)$. For $t > t_0$

$$\frac{\phi(t)}{\psi'(t) - \lambda} = \frac{\phi(t)}{\psi'(t)} \left\{ 1 + \frac{\lambda}{\psi'(t) - \lambda} \right\}$$

is decreasing, and J_4 also satisfies (1.12.4).

Lastly,
$$J_3 = O\left(\int\limits_{t_0-\delta}^{t_0+\delta} \left|\frac{\phi(t)}{x-t}\right| dt\right) = O\left(\frac{\delta\phi(t_0)}{t_0}\right).$$

Taking $\delta = \{\psi''(t_0)\}^{-\frac{1}{4}}$, the required result follows from (1.12.1). The corresponding integral with $-\lambda$ instead of λ is simpler, there being now no need to introduce t_0 . The result therefore follows in this case.

If $\phi(t)$ is decreasing, we obtain instead of (1.12.4)

$$J_2 = O\left\{\frac{\phi(T)}{t_0 \delta \psi''(t_0)}\right\}$$
,

and the result follows with $\delta = 1$.

The argument in Case (ii) is substantially the same. Examples are $\phi(t) = e^{it}$, $\psi(t) = e^{t}$; $\phi(t) = 1$, $\psi(t) = t \log t$.

1.13. The constant in Fourier's formula. The constant π enters into Fourier's formula, according to our proof, through the formula

$$\int_{0}^{+\infty} \frac{\sin x}{x} \, dx = \frac{1}{2}\pi.$$

If we take the value of this integral as our fundamental constant, and denote it by C, Fourier's cosine formula, for example, is

$$f(x) = \frac{1}{C} \int_{0}^{\infty} \cos xu \ du \int_{0}^{\infty} \cos ut f(t) \ dt.$$

The values of other familiar integrals are then obtained in terms of C; for example, taking $f(t) = e^{-t}$, and x = 0, we obtain

$$1 = \frac{1}{C} \int_{0}^{+\infty} du \int_{0}^{\infty} \cos ut \, e^{-t} \, dt = \frac{1}{C} \int_{0}^{\infty} \frac{du}{1+u^2},$$
$$\int_{0}^{\infty} \frac{du}{1+u^2} = C.$$

so that

Taking $f(x) = x^{-\frac{1}{6}}$ (x > 0), we obtain (by Theorem 6)

$$\frac{1}{\sqrt{x}} = \frac{1}{C} \int_{0}^{+\infty} \cos xu \ du \int_{0}^{+\infty} \frac{\cos ut}{\sqrt{t}} \ dt$$

$$= \frac{1}{C} \int_{0}^{+\infty} \frac{\cos xu}{\sqrt{u}} \ du \int_{0}^{+\infty} \frac{\cos y}{\sqrt{y}} \ dy = \frac{1}{C\sqrt{x}} \left(\int_{0}^{+\infty} \frac{\cos y}{\sqrt{y}} \ dy \right)^{2},$$

so that

$$\int\limits_{0}^{+\infty} \frac{\cos y}{\sqrt{y}} \, dy = \sqrt{C}.$$

Many other such examples may be derived from the formulae of Chapter VII.

Later, §1.27, we shall give another proof of a case of Fourier's formula, in which the constant π comes from the theorem of residues.

1.14. Fourier's single-integral formula. This is the formula (1.1.7). Conditions for its validity are suggested by several of the foregoing theorems; but it holds still more generally, since now it is not necessary for the Fourier transform of f(x) to exist.

THEOREM 12.† The formula

$$\frac{1}{2}\left\{f(x+0)+f(x-0)\right\} = \lim_{\lambda \to \infty} \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \frac{\sin \lambda(x-t)}{x-t} dt$$

holds if

- i (a) f(x)/(1+|x|) belongs to $L(-\infty,\infty)$,
- or i (b) f(x)/x is of bounded variation in (a,∞) and $(-\infty, -a)$ for some positive a, and tends to 0 at infinity,

or i (c) $\frac{1}{x} \int_{1}^{x} f(t) dt$ is of bounded variation in (a, ∞) and tends to 0 at infinity, and a similar condition holds in $(-\infty, -a)$;

and

(ii) in an interval including x, f(t) is of bounded variation, or satisfies one of the other conditions for the validity of Fourier's series or integral.

After the analysis of § 1.9 it is sufficient to prove that we can choose T so large that

$$\left| \int_{T}^{\infty} f(t) \frac{\sin \lambda(x-t)}{x-t} dt \right| < \epsilon$$

for all values of $\lambda > \lambda_0$, with a similar condition for $(-\infty, -T)$. This is clearly true if i(a) holds. It follows from the second mean-value theorem as in § 1.10 if i(b) holds.

To prove that i (c) is sufficient, let

$$\phi(x) = \frac{1}{x} \int_{1}^{x} f(t) dt.$$

† Prasad (1), Pringsheim (1), Hobson (1).

Then

$$f(x) = x\phi'(x) + \phi(x),$$

and $x\phi'(x)$ satisfies i (a), while $\phi(x)$ satisfies i (b). Hence the result.

Condition i (c) includes i (a); for

$$\phi'(x) = \frac{f(x)}{x} - \frac{1}{x^2} \int_{1}^{x} f(t) dt.$$

The first term belongs to $L(1,\infty)$ if i(a) holds; and so does the second, since

$$\int_{1}^{\xi} \frac{dx}{x^{2}} \int_{1}^{x} |f(t)| dt = -\frac{1}{\xi} \int_{1}^{\xi} |f(t)| dt + \int_{1}^{\xi} \frac{1}{x} |f(x)| dx$$

$$\leq \int_{1}^{\xi} \frac{1}{x} |f(x)| dx < K$$

as $\xi \to \infty$. Hence $\phi'(x)$ belongs to $L(1,\infty)$, and hence i(c) holds. On the other hand, i(c) does not include i(b).

1.15. Summability of integrals. We say that the integral $\int_{0}^{\infty} f(x) dx$ is summable (C, α) , where $\alpha \geqslant 0$, to the sum I, if

$$\lim_{\lambda\to\infty}\int\limits_0^\lambda \left(1-\frac{x}{\lambda}\right)^\alpha f(x)\ dx=I.$$

The case $\alpha=0$ is ordinary convergence. In the case $\alpha=1$ we have

$$\int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) f(x) \ dx = \frac{1}{\lambda} \int_{0}^{\lambda} dx \int_{0}^{x} f(y) \ dy,$$

a form analogous to the sum

$$s_1 + s_2 + \dots + s_n$$

in the definition of (C, 1) summability of a series. The whole process is analogous to the C-summation of series, which is too well known to need much discussion here. The main points are (i) the process is more general than ordinary convergence; for example,

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \sin ax \, dx \quad (a > 0)$$

does not exist, but

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda} \right) \sin ax \, dx = \lim_{\lambda \to \infty} \left(\frac{1}{a} - \frac{\sin a\lambda}{a^2 \lambda} \right) = \frac{1}{a};$$

and (ii) that it is consistent with ordinary convergence, in the sense that, if an integral is convergent, it is summable (C, α) to the same value for every $\alpha > 0$. This is a particular case of the following theorem.

If an integral is summable (C, α) , where $\alpha \geqslant 0$, it is summable (C, β) , where $\beta > \alpha$, to the same value.

Let
$$\phi(\lambda, \alpha) = \int_{x}^{\lambda} \left(1 - \frac{x}{\lambda}\right)^{\alpha} f(x) dx.$$

Then if $\beta > \alpha$,

$$\begin{split} \int\limits_0^\mu \left(1-\frac{\lambda}{\mu}\right)^{\beta-\alpha-1} \lambda^\alpha \phi(\lambda,\alpha) \; d\lambda &= \int\limits_0^\mu \left(1-\frac{\lambda}{\mu}\right)^{\beta-\alpha-1} \lambda^\alpha \; d\lambda \int\limits_0^\lambda \left(1-\frac{x}{\lambda}\right)^\alpha f(x) \; dx \\ &= \int\limits_0^\mu f(x) \; dx \int\limits_x^\mu \left(1-\frac{\lambda}{\mu}\right)^{\beta-\alpha-1} \left(1-\frac{x}{\lambda}\right)^\alpha \lambda^\alpha \; d\lambda \\ &= \frac{\Gamma(\beta-\alpha)\Gamma(\alpha+1)}{\Gamma(\beta+1)} \mu^{\alpha+1} \int\limits_1^\mu f(x) \left(1-\frac{x}{\mu}\right)^\beta \; dx, \end{split}$$

i.e.

$$\phi(\mu,\beta) = \frac{\Gamma(\beta+1)}{\Gamma(\beta-\alpha)\Gamma(\alpha+1)} \frac{1}{\mu^{\alpha+1}} \int_{0}^{\mu} \left(1 - \frac{\lambda}{\mu}\right)^{\beta-\alpha-1} \lambda^{\alpha} \phi(\lambda,\alpha) d\lambda.$$

Hence

$$\phi(\mu,\beta)-I=\frac{\Gamma(\beta+1)}{\Gamma(\beta-\alpha)\Gamma(\alpha+1)}\frac{1}{\mu^{\alpha+1}}\int\limits_{0}^{\mu}\left(1-\frac{\lambda}{\mu}\right)^{\beta-\alpha-1}\lambda^{\alpha}\{\phi(\lambda,\alpha)-I\}\,d\lambda.$$

If $\phi(\lambda, \alpha) \to I$, suppose that $|\phi(\lambda, \alpha) - I| \le M$ for all λ , and $\le \epsilon$ for $\lambda \ge \Delta$. Then

$$\begin{split} |\phi(\mu,\beta)-I| \leqslant \frac{\Gamma(\beta+1)}{\Gamma(\beta-\alpha)\Gamma(\alpha+1)} & \Big(\frac{\epsilon}{\mu^{\alpha+1}} \int_{\Delta}^{\mu} \Big(1-\frac{\lambda}{\mu}\Big)^{\beta-\alpha-1} \lambda^{\alpha} d\lambda + \\ & + \frac{M}{\mu^{\alpha+1}} \int_{0}^{\Delta} \Big(1-\frac{\lambda}{\mu}\Big)^{\beta-\alpha-1} \lambda^{\alpha} d\lambda \Big). \end{split}$$

The first term is $\leqslant \epsilon$ (it is ϵ if $\Delta = 0$), and the second is $O(\mu^{-\alpha-1})$ for a fixed Δ . It follows by choosing first Δ and then μ that $\phi(\mu, \beta) \to I$, the required result.

1.16. Summability of Fourier integrals. We have formally

$$\frac{1}{\pi} \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) du \int_{-\infty}^{\infty} f(t) \cos u(x - t) dt$$

$$= \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) dt \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) \cos u(x - t) du$$

$$= \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \frac{2 \sin^{2} \frac{1}{2} \lambda(x - t)}{\lambda(x - t)^{2}} dt. \tag{1.16.1}$$

This integral is analogous to Fejér's integral in the theory of Fourier series. We shall deal with it as a particular case of the following theorem.

THEOREM 13. Let

$$K(x,y,\delta) = O\left(\frac{1}{\delta}\right)$$
 $(|x-y| \leq \delta)$ (1.16.2)

$$=O\left(\frac{\delta^{\alpha}}{|x-y|^{\alpha+1}}\right) \quad (|x-y|>\delta) \tag{1.16.3}$$

for some positive α , and let

$$\lim_{\delta \to 0} \int_{-\infty}^{\infty} K(x, y, \delta) \, dy = \frac{1}{2}, \qquad \lim_{\delta \to 0} \int_{-\infty}^{x} K(x, y, \delta) \, dy = \frac{1}{2}.$$

Let $f(x)/(1+|x|^{\alpha+1})$ belong to $L(-\infty,\infty)$. Then

$$\lim_{\delta \to 0} \int_{-\infty}^{\infty} K(x, y, \delta) f(y) \, dy = \frac{1}{2} \{ \phi(x) + \psi(x) \}$$
 (1.16.4)

wherever

$$\int_{0}^{h} |f(x+t) - \phi(x)| dt = o(h), \qquad \int_{0}^{h} |f(x-t) - \psi(x)| dt = o(h)$$
(1.16.5)

as $h \to +0$. The result therefore holds (i) with $\phi(x) = f(x+0)$, $\psi(x) = f(x-0)$ wherever these expressions have a meaning, (ii) with $\phi(x) = \psi(x) = f(x)$ wherever f(x) is continuous, and (iii) with

$$\phi(x) = \psi(x) = f(x)$$

for almost all values of x.

It is sufficient to prove that

$$\lim_{\delta\to 0}\int_x^\infty K(x,y,\delta)\{f(y)-\phi(x)\}\,dy=0,$$

together with a similar result with $\psi(x)$; and by (1.16.2) and (1.16.3) this is true if

$$\lim_{\delta\to 0}\frac{1}{\delta}\int_{x}^{x+\delta}|f(y)-\phi(x)|\,dy=0$$

and

$$\lim_{\delta\to 0}\delta^{\alpha}\int_{x+\delta}^{\infty}\frac{|f(y)-\phi(x)|}{|x-y|^{\alpha+1}}\,dy=0.$$

The first part follows at once from (1.16.5). Next, let

$$\chi(h) = \int_{0}^{h} |f(x+t) - \phi(x)| dt \leqslant \epsilon h$$

for $h \leqslant \eta$. Then

$$\delta^{\alpha} \int_{x+\delta}^{x+\eta} \frac{|f(y) - \phi(x)|}{|x - y|^{\alpha + 1}} dy = \delta^{\alpha} \int_{\delta}^{\eta} \frac{|f(x + t) - \phi(x)|}{t^{\alpha + 1}} dt$$

$$= \delta^{\alpha} \left[\chi(t) \atop t^{\alpha + 1} \right]_{\delta}^{\eta} + (\alpha + 1) \delta^{\alpha} \int_{\delta}^{\eta} \chi(t) \atop t^{\alpha + 2} dt$$

$$\leq \epsilon + \epsilon (\alpha + 1) \delta^{\alpha} \int_{\delta}^{\eta} \frac{dt}{t^{\alpha + 1}} < \epsilon \left(1 + \frac{\alpha + 1}{\alpha} \right).$$

Having fixed η , plainly

$$\lim_{\delta\to 0}\delta^{\alpha}\int_{x+\eta}^{\infty}\frac{|f(y)-f(x)|}{|x-y|^{\alpha+1}}dy=0.$$

This proves the theorem.

As a particular case, let $\delta = 1/\lambda$, and

$$K(x,y,\delta) = \frac{2\sin^2\frac{1}{2}\lambda(x-y)}{\pi\lambda(x-y)^2}.$$

The conditions of the above theorem are satisfied (with $\alpha=1$). We therefore deduce† the analogue of Fejér's theorem on Fourier series:

THEOREM 14. Let f(t) belong to $L(-\infty,\infty)$. Then the integral

$$\frac{1}{\pi} \int_{0}^{\infty} du \int_{-\infty}^{\infty} f(t) \cos u(x-t) dt$$
+ Hardy (5).

is summable (C, 1) to $\frac{1}{2}\{f(x+0)+f(x-0)\}$ wherever this expression has a meaning; to f(x) wherever f(x) is continuous; and to f(x) for almost all values of x.

An obvious corollary is that if f(t) is $L(-\infty, \infty)$, and a(u), b(u), defined by (1.1.3), are 0 for all u, then f(x) = 0 for almost all x.

As another particular case, let

$$\begin{split} K\!\left(x,y,\frac{1}{\lambda}\right) &= \frac{1}{\pi} \int\limits_0^{\lambda} \left(1 - \frac{u}{\lambda}\right)^{\alpha} \cos u(x-y) \, du \\ &= \frac{\lambda}{\pi} \int\limits_0^1 (1 - v)^{\alpha} \cos \lambda v(x-y) \, dv \\ &= \frac{\alpha}{\pi (x-y)} \int\limits_0^1 (1 - v)^{\alpha - 1} \sin \lambda v(x-y) \, dv \\ &= \frac{\alpha}{\pi \lambda^{\alpha} |x-y|^{\alpha + 1}} \int\limits_0^{\lambda |x-y|} \frac{\sin\{\lambda |x-y| - w\}}{w^{1 - \alpha}} \, dw. \end{split}$$

The second formula shows that $K(x, y, 1/\lambda)$ is $O(\lambda)$, and the fourth that it is $O(\lambda^{-\alpha}|x-y|^{-\alpha-1})$. Also

$$\int_{x}^{\infty} K\left(x, y, \frac{1}{\lambda}\right) dy = \frac{\alpha}{\pi} \int_{x}^{\infty} \frac{dy}{y - x} \int_{0}^{1} (1 - v)^{\alpha - 1} \sin \lambda v (y - x) dv$$

$$= \frac{\alpha}{\pi} \int_{0}^{1} (1 - v)^{\alpha - 1} dv \int_{0}^{\infty} \frac{\sin \lambda v t}{t} dt = \frac{1}{2} \alpha \int_{0}^{1} (1 - v)^{\alpha - 1} dv = \frac{1}{2}.$$

Hence

Theorem 15. Theorem 14 is still true if (C,1) is replaced by (C,α) , where $0 < \alpha < 1$.

1.17. Cauchy's singular integral.† In the theorem of the previous section we have replaced Fourier's formula by a limit of the form

$$\lim_{\delta \to 0} \frac{1}{\pi} \int_{0}^{\infty} \phi(\delta u) \, du \int_{-\infty}^{\infty} f(t) \cos u(x-t) \, dt. \tag{1.17.1}$$

For summability (C, α) we take

$$\phi(u) = \begin{cases} (1-u)^{\alpha} & (0 < u < 1), \\ 0 & (u \geqslant 1). \end{cases}$$
 (1.17.2)

† Cauchy (1), Sommerfeld (1), Hardy (4), (5).

$$\phi(u) = e^{-u}. (1.17.3)$$

The integral in (1.17.1) is then

$$\frac{1}{\pi} \int_{-\infty}^{\infty} f(t) dt \int_{0}^{\infty} e^{-\delta u} \cos u(x-t) du = \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \frac{\delta}{\delta^2 + (x-t)^2} dt.$$
(1.17.4)

Here

$$K(x, y, \delta) = \frac{1}{\pi} \frac{\delta}{\delta^2 + (x - y)^2},$$
 (1.17.5)

and the conditions of Theorem 13 are again satisfied, with $\alpha = 1$. In particular it follows that (1.17.4) tends to f(x) almost everywhere.

This result is the rigorous form of Cauchy's argument given in § 1.1, and the integral (1.17.4) may be called *Cauchy's singular integral*. The type of summability obtained is analogous to 'summability A' for series.

1.18. Weierstrass's singular integral. Now let

$$\phi(u)=e^{-u^2}.$$

The integral (1.17.1) is then

$$\frac{1}{\pi} \int_{-\infty}^{\infty} f(t) dt \int_{0}^{\infty} e^{-\delta^{2}u^{2}} \cos u(x-t) du = \frac{1}{2\delta\sqrt{\pi}} \int_{-\infty}^{\infty} f(t) \exp\left\{-\frac{(x-t)^{2}}{4\delta^{2}}\right\} dt.$$

Here

$$K(x, y, \delta) = \frac{1}{2\delta\sqrt{\pi}}\exp\left\{-\frac{(x-y)^2}{4\delta^2}\right\},$$
 (1.18.1)

and the integral is known as Weierstrass's singular integral.†

The conditions of Theorem 13 are satisfied for any positive α ; but in fact the result holds still more generally.

THEOREM 16. If $K(x,y,\delta)$ is defined by (1.18.1), the results of Theorem 13 hold if $e^{-Cx^2}f(x)$ belongs to $L(-\infty,\infty)$ for some positive value of C (and so for all greater values).

We argue as in § 1.16, with $\alpha = 1$ say, for the integrals over $(x, x+\delta)$ and $(x+\delta, x+\eta)$. It then remains to prove that, for fixed x and η ,

$$\lim_{\delta\to 0}\frac{1}{\delta}\int_{x+\eta}^{\infty}\exp\left\{-\frac{(x-y)^2}{4\delta^2}+Cy^2\right\}g(y)\,dy=0,$$

where g(y) is L. Now

$$-\frac{(x-y)^2}{4\delta^2} + Cy^2 = -\frac{(x-y)^2}{4\delta^2} + C(x-y)^2 \frac{y^2}{(x-y)^2} \leqslant -\frac{(x-y)^2}{4\delta^2} + \frac{(x-y)^2}{8\delta^2}$$

† Weierstrass (1), Hobson (1), Lebesgue (1), Hardy (6).

if

$$C\leqslant rac{1}{8\delta^2}rac{\eta^2}{(x+\eta)^2}.$$

Hence the modulus of the above expression does not exceed

$$\frac{1}{\delta}e^{-\eta^{\bullet}/(8\delta^{\bullet})}\int_{x+\eta}^{\infty}|g(y)|\ dy,$$

and this tends to 0.

1.19. General summability. If we merely require to deal with the case in which f(x+0) and f(x-0) exist, we can use the following simpler theorem.

Theorem 17. Let $K(x, y, \delta) \geqslant 0$,

$$\lim_{\delta \to 0} \int_{x}^{b} K(x, y, \delta) dy = \frac{1}{2}, \qquad \lim_{\delta \to 0} \int_{a}^{x} K(x, y, \delta) dy = \frac{1}{2}; (1.19.1)$$

and let

$$\lim_{\delta \to 0} K(x, y, \delta) = 0 \tag{1.19.2}$$

uniformly for all x and y for which $|x-y| \ge \epsilon > 0$, and also, in the case $a = -\infty$, $b = \infty$,

$$\lim_{\delta \to 0} \int_{-\infty}^{x-\epsilon} K(x,y,\delta) \, dy = 0, \qquad \lim_{\delta \to 0} \int_{x+\epsilon}^{\infty} K(x,y,\delta) \, dy = 0 \quad (1.19.3)$$

for any fixed positive ϵ .

Let f(x) belong to L(a,b). Then

$$\lim_{\delta \to 0} \int_{a}^{b} K(x, y, \delta) f(y) dy = \frac{1}{2} \{ f(x+0) + f(x-0) \}$$
 (1.19.4)

wherever the right-hand side exists.

If f is continuous at the point x, (1.19.1) can be replaced by

$$\lim_{\delta \to 0} \int_{-\infty}^{\infty} K(x, y, \delta) dy = 1.$$
 (1.19.5)

We have to prove that

$$\int_{a}^{b} K(x,y,\delta)\{f(y)-f(x+0)\} dy \to 0,$$

with a similar result for (a, x). This integral does not exceed in absolute value

$$\max_{x \leqslant y \leqslant x+\epsilon} |f(y)-f(x+0)| \int_{x}^{b} K(x,y,\delta) dy + \\ + \max_{y>x+\epsilon} K(x,y,\delta) \int_{x+\epsilon}^{b} |f(y)| dy + |f(x+0)| \int_{x+\epsilon}^{b} K(x,y,\delta) dy,$$

which tends to 0 by choosing first ϵ and then δ . Similarly for the other part.

The relevant parts of the summability theorems are clearly cases of this theorem. They may, however, be exhibited as direct consequences of the form of the summability factor; the general result is as follows.

THEOREM 18. Let $\phi(x)$ belong to $L(0,\infty)$ and have only a finite number of maxima and minima in $(0,\infty)$; let $\phi(+0)=1$; and let $\phi(x)$ be the integral of $\phi'(x)$, which is ultimately negative non-decreasing. Let f(x) belong to $L(-\infty,\infty)$. Then

$$\lim_{\delta \to 0} \frac{1}{\pi} \int_{0}^{\infty} \phi(\delta u) \, du \int_{-\infty}^{\infty} f(t) \cos u(x-t) \, dt = \frac{1}{2} \{ f(x+0) + f(x-0) \}$$

wherever the right-hand side exists.

This follows from the previous theorem if

$$K(x,y,\delta) = \frac{1}{\pi} \int_{0}^{\infty} \phi(\delta u) \cos u(x-y) \ du$$

has the properties stated.

Suppose first that $\phi'(x)$ is negative non-decreasing for all x. Then

$$\begin{split} K(x,y,\delta) &= \frac{\delta}{\pi(x-y)} \int\limits_0^\infty \{-\phi'(\delta u)\} \sin u(x-y) \ du \\ &= \frac{\delta}{\pi|x-y|} \sum\limits_{n=0}^\infty \int\limits_{\substack{n\pi\\|x-y|}}^{\frac{(n+1)\pi}{|x-y|}} \{-\phi'(\delta u)\} \sin u|x-y| \ du. \end{split}$$

The sum is positive, and its value does not exceed that of the first term. Hence $\pi/|x-y|$

which tends to 0 as $\delta \to 0$, uniformly for $|x-y| \ge \epsilon$. Hence (1.19.2) holds. Also

$$\int_{x}^{\infty} K(x, y, \delta) dy = -\frac{\delta}{\pi} \int_{x}^{\infty} \frac{dy}{x - y} \int_{0}^{\infty} \phi'(\delta u) \sin u(x - y) du$$

$$= -\frac{\delta}{\pi} \int_{0}^{\infty} \phi'(\delta u) du \int_{x}^{\infty} \frac{\sin u(x - y)}{x - y} dy$$

$$= -\frac{1}{2} \delta \int_{0}^{\infty} \phi'(\delta u) du = \frac{1}{2} \phi(+0) = \frac{1}{2},$$

the inversion being justified by dominated convergence, since $\phi'(\delta u)$ belongs to $L(0,\infty)$. Similarly for $(-\infty,x)$, and (1.19.1) follows. Similarly,

$$\begin{split} \int\limits_{Y}^{\infty} K(x,y,\delta) \; dy &= -\frac{\delta}{\pi} \int\limits_{0}^{\infty} \phi'(\delta u) \; du \int\limits_{Y}^{\infty} \frac{\sin u(x-y)}{x-y} \; dy \\ &= O\bigg(\delta \int\limits_{0}^{1} \left\{-\phi'(\delta u)\right\} du \bigg) + O\bigg(\frac{\delta}{Y} \int\limits_{1}^{\infty} \left\{-\phi'(\delta u)\right\} du \bigg) \\ &= O\{\phi(+0) - \phi(\delta)\} + O(Y^{-1}), \end{split}$$

and (1.19.3) follows from (1.19.2) on choosing first Y sufficiently large, and then δ sufficiently small.

In the more general case, we can write $\phi(x) = \phi_1(x) - \phi_2(x)$, where ϕ_1' and ϕ_2' are negative non-decreasing. Let K_1 , K_2 be the corresponding K-functions. Then K_1 and K_2 are positive, and satisfy (1.19.2); the integrals

 $\int_{-\infty}^{\infty} K_1(x,y,\delta) \ dy, \qquad \int_{-\infty}^{\infty} K_2(x,y,\delta) \ dy$

are bounded; and (1.19.1) holds as before. These conditions are clearly sufficient for the theorem.

1.20. In all the particular cases considered, we have

$$K(x, y, \delta) = K(x-y, \delta),$$

where $K(u, \delta)$ is an even function of u,

$$\int_{0}^{\infty} K(u,\delta) du = \frac{1}{2},$$

$$\lim_{\delta \to 0} \int_{n}^{\infty} K(u,\delta) du = 0$$

and

for every positive η . With these conditions, not only does

$$\chi(x,\delta) = \int_{-\infty}^{\infty} K(x-y,\delta)f(y) dy$$

tend to f(x) at particular points, but in the sense of the following theorem it tends to f(x) on the average.

THEOREM 19. If $K(u, \delta)$ is positive and satisfies the above conditions and f(x) is $L(-\infty, \infty)$, then

$$\lim_{\delta\to 0}\int_{-\infty}^{\infty}|\chi(x,\delta)-f(x)|\ dx=0.$$

For

$$\chi(x,\delta)-f(x) = \int_{-\infty}^{\infty} K(u,\delta)\{f(x-u)-f(x)\} du,$$

$$\int_{-\infty}^{\infty} |\chi(x,\delta)-f(x)| dx \leqslant \int_{-\infty}^{\infty} dx \int_{-\infty}^{\infty} K(u,\delta)|f(x-u)-f(x)| du$$

$$= \int_{-\infty}^{\infty} K(u,\delta) du \int_{-\infty}^{\infty} |f(x-u)-f(x)| dx.$$

$$\psi(u) = \int_{-\infty}^{\infty} |f(x-u)-f(x)| dx$$

Now

is bounded for all u, and tends to 0 with u. Let $|\psi(u)| \le \epsilon$ for $|u| \le \eta$. Then

$$\left|\int_{-n}^{\eta} K(u,\delta)\psi(u) du\right| \leqslant \epsilon \int_{-\infty}^{\infty} K(u,\delta) du = \epsilon,$$

and, if $|\psi(u)| \leqslant M$,

$$\left|\int_{\eta}^{\infty} K(u,\delta)\psi(u) \ du\right| \leqslant M \int_{\eta}^{\infty} K(u,\delta) \ du,$$

which by hypothesis tends to 0 with δ . Similarly for $(-\infty, -\eta)$. Hence the result.

For example, in the case of (C, 1) summability,

$$\frac{1}{\pi} \int_{-\infty}^{\infty} f(y) \frac{1 - \cos \lambda (x - y)}{\lambda (x - y)^2} dy$$

converges to f(x) on the average, in the above sense, as $\lambda \to \infty$. (Take $\lambda = 1/\delta$.)

1.21. Further summability theorems. In all the foregoing theorems we have imposed on f(t) conditions which ensure the existence of the inner integral in Fourier's formula in some sense or other. We shall next prove a theorem in which, without imposing any particular condition, we merely assume that the inner integral exists.

Here we are not particularly concerned with the behaviour of f(t) in finite intervals, and for the sake of simplicity we shall suppose that it is continuous.

THEOREM 20.† Let f(t) be integrable over any finite interval, continuous at t = x, and let

$$\int_{-\infty}^{\infty} f(t) \cos u(x-t) \ dt$$

converge uniformly over any finite interval of values of u. If the limit is g(x, u), then

 $\lim_{\lambda\to\infty}\frac{1}{\pi}\int_{0}^{\lambda}\left(1-\frac{u}{\lambda}\right)g(x,u)\ du=f(x).$

We have

$$\int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) g(x, u) \ du = \int_{-\infty}^{\infty} f(t) \ dt \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) \cos u(x - t) \ du$$

$$= \int_{-\infty}^{\infty} f(t) \frac{2 \sin^{2} \frac{1}{2} \lambda(x - t)}{\lambda(x - t)^{2}} \ dt = \int_{-\infty}^{-T} + \int_{-T}^{T} + \int_{T}^{\infty} = J_{1} + J_{2} + J_{3},$$

say, where T > |x|. The inversion is justified by uniform convergence.

Let
$$f_1(t) = \int\limits_0^t f(v) \ dv.$$

By the case u=0 of the data, $f_1(t)$ is bounded, say $|f_1(t)| \leq M$. Now

$$J_{3} = -f_{1}(T) \frac{2 \sin^{2} \frac{1}{2} \lambda(x - T)}{\lambda(x - T)^{2}} - \int_{T}^{\infty} f_{1}(t) \frac{\sin \lambda(x - t)}{(x - t)^{2}} dt - \int_{T}^{\infty} f_{1}(t) \frac{4 \sin^{2} \frac{1}{2} \lambda(x - t)}{\lambda(x - t)^{3}} dt$$

[†] The analogue for Mellin integrals was proved by Hardy (8). See Macphail and Titchmarsh (1).

on integrating by parts. Hence

$$|J_3| \leqslant rac{2M}{\lambda (T-x)^2} + M \int\limits_{T}^{\infty} rac{dt}{(t-x)^2} + M \int\limits_{T}^{\infty} rac{dt}{\lambda (t-x)^3} = rac{5M}{2\lambda (T-x)^2} + rac{M}{T-x}.$$

This can be made arbitrarily small by choice of T, for all $\lambda > \lambda_0$; and a similar argument applies to J_1 . Having fixed $T, J_2 \to \pi f(x)$ as in § 1.16. This proves the theorem.

1.22. The general result of the above type appears to be that, if the inner integral is summable (C, k), the outer integral is summable (C, k+1). The above theorem is the case k=0, and we shall next prove the case k=1. The proof of the general case does not seem to require any new idea, but it would be rather laborious to write out.

THEOREM 21. Let f(t) be integrable over any finite interval, continuous at t = x, and let

$$\lim_{\mu \to \infty} \int_{0}^{\mu} \left(1 - \frac{t}{\mu}\right) f(t) \cos u(x - t) dt$$

and

$$\lim_{\mu \to \infty} \int_{-\mu}^{0} \left(1 - \frac{|t|}{\mu}\right) f(t) \cos u(x - t) dt$$

exist, uniformly over any finite interval of values of u. If the sum of these limits is g(x, u),

$$\lim_{\lambda \to \infty} \frac{1}{\pi} \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda} \right)^{2} g(x, u) \ du = f(x).$$

It will be sufficient to consider the case where f(t) = 0 for t < 0. By uniform convergence

$$\int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right)^{2} g(x, u) \ du = \lim_{\mu \to \infty} \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right)^{2} du \int_{0}^{\mu} \left(1 - \frac{t}{\mu}\right) f(t) \cos u(x - t) \ dt.$$

The repeated integral is equal to

$$\int_{0}^{\mu} \left(1 - \frac{t}{\mu}\right) f(t) dt \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right)^{2} \cos u(x - t) du$$

$$= \int_{0}^{\mu} \left(1 - \frac{t}{\mu}\right) f(t) \left(\frac{2}{\lambda(x - t)^{2}} - \frac{2 \sin \lambda(x - t)}{\lambda^{2}(x - t)^{3}}\right) dt = \int_{0}^{T} + \int_{T}^{\mu} = J_{1} + J_{2},$$

say, where $|x| < T < \mu$. Define $f_1(t)$ as before, and let

$$f_2(t) = \int_0^t f_1(v) \ dv = t \int_0^t \left(1 - \frac{v}{t}\right) f(v) \ dv.$$

Then $f_2(t)/t$ is bounded, as a particular case of the data. On integrating by parts twice, we obtain

$$\begin{split} J_2 &= - \left(1 - \frac{T}{\mu} \right) \! f_1(T) \phi(T) + \frac{1}{\mu} \{ f_2(\mu) \phi(\mu) - f_2(T) \phi(T) \} + \\ &\quad + \left(1 - \frac{T}{\mu} \right) \! f_2(T) \phi'(T) - \frac{2}{\mu} \int_T^\mu f_2(t) \phi'(t) \; dt \; + \int_T^\mu \left(1 - \frac{t}{\mu} \right) \! f_2(t) \phi''(t) \; dt, \end{split}$$
 where
$$\phi(t) &= \frac{2}{\lambda (x - t)^2} - \frac{2 \sin \lambda (x - t)}{\lambda^2 (x - t)^3} \; ,$$

$$\phi'(t) &= \frac{4}{\lambda (x - t)^3} + \frac{2 \cos \lambda (x - t)}{\lambda (x - t)^3} - \frac{6 \sin \lambda (x - t)}{\lambda^2 (x - t)^4} \; ,$$

$$\phi''(t) &= \frac{12}{\lambda (x - t)^4} + \frac{2 \sin \lambda (x - t)}{(x - t)^3} + \frac{12 \cos \lambda (x - t)}{\lambda (x - t)^4} - \frac{24 \sin \lambda (x - t)}{\lambda^2 (x - t)^5} \; .$$

Making $\mu \to \infty$, we obtain

$$J_1
ightarrow \int\limits_0^T f(t)\phi(t)\ dt,$$

$$J_2
ightarrow -f_1(T)\phi(T) + f_2(T)\phi'(T) + \int\limits_T^\infty f_2(t)\phi''(t)\ dt.$$

We can choose T so large that the last two terms are arbitrarily small for all $\lambda > \lambda_0$. Having fixed T, $f_1(T)\phi(T) \to 0$ as $\lambda \to \infty$, and

$$\int_{0}^{T} f(t)\phi(t) dt \to \pi f(x)$$

by the theory of Fejér's integral, § 1.16, and the consistency theorem for C-summability, § 1.15. This proves the theorem.

1.23. We have seen that the (C, 1) of Theorem 14 can be replaced by (C, α) , where α is arbitrarily small. This is not true of Theorem 20; in neither Theorem 20 nor Theorem 21 can the order of summability of the outer integral be reduced. We shall now prove this by means of examples.

Let
$$I_{lpha}(\lambda,t)=\int\limits_{-\lambda}^{\lambda}\left(1-rac{u}{\lambda}
ight)^{lpha}\cos ut\ du.$$

As in § 1.16, if $0 < \alpha < 1$,

$$I_{\alpha}(\lambda,t) = \frac{\alpha}{\lambda^{\alpha}t^{\alpha+1}} \int_{0}^{\lambda t} v^{\alpha-1} \sin(\lambda t - v) \, dv = O\left(\frac{1}{\lambda^{\alpha}t^{\alpha+1}}\right)$$

as $\lambda \to \infty$, $t \to \infty$. Also

$$\frac{\partial I}{\partial t} = \frac{\alpha \lambda^{1-\alpha}}{t^{\alpha+1}} \int_{0}^{\lambda t} v^{\alpha-1} \cos(\lambda t - v) \, dv - \frac{\alpha(\alpha+1)}{\lambda^{\alpha} t^{\alpha+2}} \int_{0}^{\lambda t} v^{\alpha-1} \sin(\lambda t - v) \, dv$$

$$= \frac{\alpha \lambda^{1-\alpha}}{t^{\alpha+1}} \int_{0}^{+\infty} v^{\alpha-1} \cos(\lambda t - v) \, dv + O\left(\frac{1}{t^2}\right) + O\left(\frac{1}{\lambda^{\alpha} t^{\alpha+2}}\right)$$

$$= \Gamma(\alpha+1)\lambda^{1-\alpha} t^{-1-\alpha} \cos(\lambda t - \frac{1}{2}\pi\alpha) + O(t^{-2}).$$

Suppose that we try to prove Theorem 20 with (C, α) , where $0 < \alpha < 1$, instead of (C, 1). We encounter a term

$$egin{aligned} J_{3} &= \int\limits_{T}^{\infty} f(t) I_{lpha}(\lambda, x - t) \; dt \ \\ &= -f_{1}(T) I_{lpha}(\lambda, x - T) + \int\limits_{T}^{\infty} f_{1}(t) rac{\partial}{\partial t} I_{lpha}(\lambda, x - t) \; dt. \end{aligned}$$

Take T fixed (>|x|). Then everything is bounded except possibly the term

$$\Gamma(\alpha+1)\lambda^{1-\alpha}\int_{T}^{\infty}f_1(t)t^{-1-\alpha}\cos(\lambda x-\lambda t-\frac{1}{2}\pi\alpha)\ dt.$$

Let

$$f(t) = 2^{\nu} \nu^{-2} \sin 2^{\nu} t \quad (\nu \pi \leqslant t < (\nu + 1)\pi),$$

for $\nu = 1, 2, ...,$ and f(t) = 0 elsewhere. Then

$$f_1(t) = \nu^{-2}(1-\cos 2^{\nu}t) \quad (\nu\pi \leqslant t < (\nu+1)\pi).$$

Clearly

$$\int\limits_0^T f(t) \cos u(x-t) \; dt = f_1(T) \cos u(x-T) - u \int\limits_0^T f_1(t) \sin u(x-t) \; dt \to \text{limit}$$

as $T \to \infty$, uniformly with respect to u, so that the conditions of Theorem 20 are satisfied.

Let $\lambda = 2^{\rho}$. Then

$$\int_{-\infty}^{\infty} f_1(t)t^{-1-\alpha}\cos \lambda t \ dt = \sum_{\nu=1}^{\infty} \frac{1}{\nu^2} \int_{\nu\pi}^{(\nu+1)\pi} \frac{(1-\cos 2^{\nu}t)\cos 2^{\rho}t}{t^{\alpha+1}} dt.$$

The term $\nu = \rho$ is

$$\frac{1}{\rho^{2}} \int_{\rho\pi}^{(\rho+1)\pi} \frac{\cos 2^{\rho}t}{t^{\alpha+1}} dt - \frac{1}{2\rho^{2}} \int_{\rho\pi}^{(\rho+1)\pi} \frac{1 + \cos 2^{\rho+1}t}{t^{\alpha+1}} dt \\
< -\frac{1}{2\rho^{2}(\rho+1)^{\alpha}\pi^{\alpha}} + O\left(\frac{1}{2^{\rho}}\right).$$

The remaining terms are

$$\sum_{\nu \neq \rho} \frac{1}{\nu^2} \int_{\nu\pi}^{(\nu+1)\pi} \frac{\cos 2^{\rho}t - \frac{1}{2}\cos(2^{\nu} - 2^{\rho})t - \frac{1}{2}\cos(2^{\nu} + 2^{\rho})t}{t^{\alpha+1}} dt$$

$$= \sum_{\nu \neq \rho} O\left(\frac{1}{\nu^{\alpha+3}|2^{\rho} - 2^{\nu}|}\right) = O\left(\frac{1}{2^{\rho}}\right).$$
Similarly,
$$\int_{-\infty}^{\infty} \frac{f_1(t)\sin \lambda t}{t^{\alpha+1}} dt = O\left(\frac{1}{2^{\rho}}\right).$$

Hence

$$|J_3| > A|\cos(\lambda x - \frac{1}{2}\pi\alpha)|\lambda^{1-\alpha}(\log\lambda)^{-\alpha-2} + O(1),$$

for $\lambda=2^p$. Finally, the sequence $\cos(2^px-\frac{1}{2}\pi\alpha)$ does not tend to 0, since, if one term is small, the next is approximately $-\cos\frac{1}{2}\pi\alpha$. Hence J_3 is unbounded.

Also, by Theorem 15, J_2 tends to a limit. It follows that, in Theorem 20, (C, 1) cannot be replaced by (C, α) , if $\alpha < 1$.

If $1 < \alpha < 2$, we can write

$$I_{\alpha}(\lambda,t) = \frac{\alpha}{\lambda t^2} - \frac{\alpha(\alpha-1)}{\lambda^{\alpha}t^{\alpha+1}} \int_{0}^{\lambda t} v^{\alpha-2}\cos(\lambda t - v) dv.$$

Hence $\partial^2 I/\partial t^2$ contains a term

$$-\frac{\alpha(\alpha-1)\lambda^{2-\alpha}}{t^{\alpha+1}} \int_{0}^{\lambda t} v^{\alpha-2}\cos(\lambda t - v) dv$$

$$= \frac{\Gamma(\alpha+1)\lambda^{2-\alpha}}{t^{\alpha+1}} \sin(\lambda t - \frac{1}{2}\pi\alpha) + O\left(\frac{1}{t^{3}}\right),$$

and, in the argument with (C, α) analogous to that of § 1.22, we obtain the term

$$\Gamma(\alpha+1)\lambda^{2-\alpha}\int_{T}^{\infty}f_{2}(t)t^{-\alpha-1}\sin(\lambda x-\lambda t-\frac{1}{2}\pi\alpha)\ dt.$$

$$f(t) = 2^{2\nu}\nu^{-2}\cos 2^{\nu}t \quad (\nu\pi \leqslant t < (\nu+1)\pi)$$

Let

for $\nu = 1, 2, ..., \text{ and } f(t) = 0$ elsewhere. Then

$$f_2(t) = \nu^{-2}(1 - \cos 2^{\nu}t) \quad (\nu \pi \leqslant t < (\nu + 1)\pi).$$

Since

$$\int_{0}^{\mu} \left(1 - \frac{t}{\mu}\right) f(t) \cos u(x - t) dt = f_{2}(\mu) \frac{\cos u(x - \mu)}{\mu} - \int_{0}^{\mu} f_{2}(t) \left\{ \frac{2u}{\mu} \sin u(x - t) + \left(1 - \frac{t}{\mu}\right) u^{2} \cos u(x - t) \right\} dt,$$

which clearly tends to a limit as $\mu \to \infty$, the conditions of Theorem 21 are fulfilled.

The proof that the selected term is unbounded now proceeds as before. The remaining terms are easily seen to be bounded, and the desired result follows.

1.24. The integrated form of Fourier's formula. It is well known that the result of formally integrating a Fourier series termby-term is true, whether the original series is convergent or not. The corresponding theorem for integrals is as follows.

THEOREM 22. If f(x) belongs to $L(-\infty,\infty)$, then

$$\int_{0}^{\xi} f(x) dx = \frac{1}{\pi} \int_{0}^{+\infty} \frac{du}{u} \int_{-\infty}^{\infty} f(t) \{ \sin u(\xi - t) + \sin ut \} dt, \quad (1.24.1)$$

$$\int_{0}^{\xi} f(x) dx = \frac{1}{2\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \frac{e^{-i\xi u} - 1}{-iu} du \int_{-\infty}^{\infty} f(t)e^{iut} dt \qquad (1.24.2)$$

for all values of ξ ; and

$$\int_{0}^{\xi} f(x) dx = \frac{2}{\pi} \int_{0}^{\infty} \frac{\sin \xi u}{u} du \int_{0}^{\infty} f(t) \cos ut dt, \qquad (1.24.3)$$

$$\int_{0}^{\xi} f(x) dx = \frac{2}{\pi} \int_{0}^{\infty} \frac{1 - \cos \xi u}{u} du \int_{0}^{\infty} f(t) \sin ut dt \qquad (1.24.4)$$

for $\xi \geqslant 0$.

The formulae correspond to (1.1.1), (1.1.6), (1.1.4), (1.1.5) respectively. Consider for example (1.24.2). We have

$$\int_{-iu}^{\lambda} \frac{e^{-i\xi u}-1}{-iu} du \int_{-iu}^{\infty} f(t)e^{iut} dt = \int_{-iu}^{\infty} f(t) dt \int_{-iu}^{\lambda} \frac{e^{iu(t-\xi)}-e^{iut}}{-iu} du$$

by uniform convergence. The inner integral is

$$2\int_{0}^{\lambda}\frac{\sin tu-\sin(t-\xi)u}{u}\,du,$$

which is bounded for all t and λ , and, as $\lambda \to \infty$, tends to 2π ($0 < t < \xi$), 0 (t < 0 or $t > \xi$). The result therefore follows by dominated convergence. The other formulae are easily deduced from this, or proved in a similar way.

1.25. The complex form of Fourier's integral. The theory of the complex form of Fourier's integral is substantially the same as that of the forms already considered. We shall state briefly the most important results.

We have so far supposed that all the functions are real. There is, however, no additional difficulty in dealing with complex functions of a real variable, and it is natural to apply the complex form of the theorem to these. The extension of all the definitions is immediate; a complex function f(x) is integrable, of bounded variation, etc., if its real and imaginary parts separately have these properties.

THEOREM 23. Let f(t) belong to $L(-\infty,\infty)$, and let it be of bounded variation in the neighbourhood of t=x. Then

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \frac{1}{2\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} e^{-ixu} \, du \int_{-\infty}^{\infty} f(t)e^{iut} \, dt. \quad (1.25.1)$$

If f(t) satisfies the conditions of Theorem 4 in the neighbourhood of t = x, the left-hand side may be replaced by f(x).

We may invert by uniform convergence, and obtain

$$\int_{-\lambda}^{\lambda} e^{-ixu} du \int_{-\infty}^{\infty} f(t)e^{iut} dt = \int_{-\infty}^{\infty} f(t) dt \int_{-\lambda}^{\lambda} e^{-iu(x-t)} du$$

$$= 2 \int_{-\infty}^{\infty} f(t) \frac{\sin \lambda(x-t)}{x-t} dt,$$

and the result follows as in the proof of Theorem 3.

As a particular case, suppose in addition that f(z) is analytic for $y \ge 0$, and $f(z) \to 0$ as $|z| \to \infty$ uniformly for $0 \le \arg z \le \pi$. Then by Jordan's lemma (Whittaker and Watson, *Modern Analysis*, § 6.222) F(u) = 0 for u > 0. The formulae reduce by a change of variable to those of Laplace, (1.4.1), (1.4.2).

If we use the functions $F_{+}(w)$ and $F_{-}(w)$, we obtain a theorem in which f(x) is less restricted at infinity.

Theorem 24. Let $e^{-c|t|}f(t)$ belong to $L(-\infty,\infty)$ for some positive c, so that $F_+(w)$, $F_-(w)$, defined by (1.3.1), (1.3.2), exist for $v \ge c$, $v \le -c$, respectively. Then, if f(t) satisfies conditions corresponding to those of Theorems 3 or 4 in the neighbourhood of t = x,

$$\begin{split} \frac{1}{2}\{f(x+0)+f(x-0)\} &= \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ia-\lambda}^{ia+\lambda} F_+(w) e^{-ixw} \, dw \, + \\ &\quad + \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ib-\lambda}^{ib+\lambda} F_-(w) e^{-ixw} \, dw, \end{split}$$

where $a \geqslant c$, $b \leqslant -c$.

Let $g(x) = e^{-ax}f(x)$ (x > 0), 0 (x < 0). Then by the previous theorem

$$\begin{split} \frac{1}{2}\{g(x+0)+g(x-0)\} &= \frac{1}{2\pi}\lim_{\lambda\to\infty}\int\limits_{-\lambda}^{\lambda} e^{-ixu}\,du\int\limits_{-\infty}^{\infty} g(t)e^{iut}\,dt \\ &= \frac{1}{2\pi}\lim_{\lambda\to\infty}\int\limits_{-\lambda}^{\lambda} e^{-ixu}\,du\int\limits_{0}^{\infty} f(t)e^{i(u+ia)t}\,dt \\ &= \frac{1}{\sqrt{(2\pi)}}\lim_{\lambda\to\infty}\int\limits_{-\lambda}^{\lambda} e^{-ixu}F_{+}(u+ia)\,du, \end{split}$$

or $\frac{1}{2}e^{ax}\left\{g(x+0)+g(x-0)\right\} = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ia-\lambda}^{ia+\lambda} F_{+}(w)e^{-ixw} dw.$

Similarly, if $h(x) = e^{bx}f(x)$ (x < 0), 0 (x > 0), then

$$\frac{1}{2}e^{-bx}\{h(x+0)+h(x-0)\} = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ib-\lambda}^{ib+\lambda} F_{-}(w)e^{-ixw} dw.$$

The result stated follows on addition.

1.26. Perron's formula.† The formula known as Perron's formula in the theory of Dirichlet series can be deduced from Theorem 24.

Theorem 25. Let
$$f(s) = \sum_{n=1}^{\infty} a_n e^{-\lambda_n s}$$

be convergent for $\sigma > \sigma_0$, where the λ_n are real and steadily increasing to infinity, and let $A(x) = \sum_{n=0}^{\infty} a_n.$

[†] See Hardy and Riesz, The General Theory of Dirichlet's Series, 12-14.

Then if k > 0, $k > \sigma_0$,

$$\frac{1}{2}\{A(x+0)+A(x-0)\} = \frac{1}{2\pi i} \lim_{T\to\infty} \int_{k-iT}^{k+iT} \frac{f(s)}{s} e^{sx} ds.$$

Let $A_0 = 0$, $A_n = a_1 + a_2 + ... + a_n$.

If $\beta > \sigma_0$, $\sum_{1}^{m} a_{\nu} e^{-\lambda_{\nu} \beta}$ is bounded for all m, and hence, if also $\beta > 0$,

$$A_n = \sum_{\nu=1}^n a_{\nu} e^{-\lambda_{\nu} \beta} \cdot e^{\lambda_{\nu} \beta} = O(e^{\lambda_{\mathbf{n}} \beta}).$$

Hence, if N is the greatest n for which $\lambda_n \leqslant x$,

$$A(x) = A_N = O(e^{\lambda_N \beta}) = O(e^{\beta x}).$$

Hence for $\sigma > \beta$,

$$f(s) = \sum_{n=1}^{\infty} (A_n - A_{n-1})e^{-\lambda_n s} = \sum_{n=1}^{\infty} A_n (e^{-\lambda_n s} - e^{-\lambda_{n+1} s})$$

$$= \sum_{n=1}^{\infty} A_n s \int_{\lambda_n}^{\lambda_{n+1}} e^{-sy} dy = s \int_{0}^{\infty} A(y)e^{-sy} dy.$$

Since A(y) is of bounded variation in any finite interval, the result follows from Theorem 24.

1.27. Fourier's theorem for analytic functions. The following form of Fourier's theorem applies to a class of analytic functions.

THEOREM 26. Let f(z) be an analytic function, regular for

$$-a < y < b$$
,

where a > 0, b > 0. In any strip interior to -a < y < b, let

$$f(z) = \begin{cases} O(e^{-(\lambda - \epsilon)x}) & (x \to \infty) \\ O(e^{(\mu - \epsilon)x}) & (x \to -\infty) \end{cases}$$
 (1.27.1)

for every positive ϵ , where $\lambda > 0$, $\mu > 0$. Then F(w), defined by (1.2.5), satisfies conditions similar to those imposed on f(z), with a, b, λ , μ replaced by λ , μ , b, a; and

$$f(z) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(w)e^{-izw} dw$$
 (1.27.2)

for every z in the strip -a < y < b.

We have
$$F(w) = \frac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} f(\zeta) e^{i\zeta w} \, d\zeta,$$

and the integral converges uniformly for $-\lambda < v < \mu$. Hence F(w) is analytic in this strip. By an obvious application of Cauchy's theorem we may take the integral along any line of the strip parallel to the real axis. Thus

$$F(w) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} f(\xi+i\eta) e^{i(\xi+i\eta)(u+iv)} d\xi = O(e^{-\eta u}),$$

and by taking η arbitrarily near to -a or b the order-results for F(w) follow.

The reciprocal formula (1.27.2) can be deduced from Theorem 23; it can also be proved directly by the theorem of residues. Let $-a < -\alpha < y < \beta < b$. Then

$$\frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} F(w)e^{-izw} dw = \frac{1}{2\pi} \int_{0}^{\infty} e^{-izw} dw \int_{i\beta-\infty}^{i\beta+\infty} f(\zeta)e^{i\zeta w} d\zeta$$

$$= \frac{1}{2\pi} \int_{i\beta-\infty}^{i\beta+\infty} f(\zeta) d\zeta \int_{0}^{\infty} e^{-i(z-\zeta)w} dw$$

$$= \frac{1}{2\pi i} \int_{i\beta-\infty}^{i\beta+\infty} f(\zeta) d\zeta,$$

the inversion being justified by absolute convergence. Similarly,

$$\frac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^{0}F(w)e^{-izw}\,dw=\frac{1}{2\pi i}\int\limits_{-i\alpha-\infty}^{-i\alpha+\infty}\frac{f(\zeta)}{\zeta-z}\,d\zeta,$$

and, by an obvious application of the theorem of residues, the sum of the right-hand sides is f(z).

1.28. Summability of the complex form. The various summability theorems have obvious extensions to the complex form of the theorem. It will be sufficient to state one of them.

THEOREM 27. Let f(t) belong to $L(-\infty,\infty)$, or, more generally, let

$$\int_{-\infty}^{+\infty} f(t)e^{iut} dt$$

converge uniformly in any finite interval. Then

$$\frac{1}{2\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|u|}{\lambda} \right) e^{-ixu} du \int_{-\infty}^{+\infty} f(t) e^{iut} dt$$

is equal to $\frac{1}{2}\{f(x+0)+f(x-0)\}$ wherever this expression has a meaning; to f(x) wherever f(x) is continuous; and to f(x) for almost all values of x.

On inverting the order of integration the integral reduces to (1.16.1), and the result then follows from Theorems 14 and 20.

1.29. Mellin's inversion formula. Theorems on Mellin's formula may be obtained from theorems on Fourier's formula by a change of variable, as the formula itself was obtained in §1.5; and of course there is no difficulty in adapting the arguments to give a direct proof in each case.

We shall state only the most important theorems.

THEOREM 28. Let $y^{k-1}f(y)$ belong to $L(0,\infty)$, and let f(y) be of bounded variation in the neighbourhood of the point y=x. Let

$$\mathfrak{F}(s) = \int_{0}^{\infty} f(x)x^{s-1} dx \quad (s = k+it). \tag{1.29.1}$$

 $T^{h}en$

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \frac{1}{2\pi i} \lim_{\tau \to \infty} \int_{k-iT}^{k+iT} \mathfrak{F}(s)x^{-s} \, ds. \qquad (1.29.2)$$

THEOREM 29. Let $\mathfrak{F}(k+iu)$ belong to $L(-\infty,\infty)$, and let it be of bounded variation in the neighbourhood of the point u=t. Let

$$f(x) = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s) x^{-s} \, ds. \tag{1.29.3}$$

Then

$$\frac{1}{2} [\mathfrak{F}\{k+i(t+0)\} + \mathfrak{F}\{k+i(t-0)\}] = \lim_{\lambda \to \infty} \int_{1/\lambda}^{\lambda} f(x) x^{k+it-1} \, dx. \tag{1.29.4}$$

Both theorems are obtained by changes of the variable in Theorem 23.

In some examples the following theorem is required.

THEOREM 30. Let

$$\Re(k+it) = \phi(t)e^{i\psi(t)}.$$

where $\phi(t)$ and $\psi(t)$ satisfy the conditions of Theorem 11, both as $t \to \infty$ and $t \to -\infty$; or let

$$e^{kx}f(e^x) = \phi(x)e^{i\psi(x)},$$

where ϕ and ψ satisfy such conditions. Then Mellin's formulae hold, the integrals being non-absolutely convergent.

This follows from Theorem 11 by the usual substitutions.

THEOREM 31. Let f(z) be an analytic function of $z = re^{i\theta}$, regular for $-\alpha < \theta < \beta$, where $0 < \alpha \leqslant \pi$, $0 < \beta \leqslant \pi$; and let f(z) be $O(|z|^{-a-\epsilon})$ for small z, and $O(|z|^{-b+\epsilon})$ for large z, where a < b, uniformly in any angle interior to the above.

Then $\mathfrak{F}(s)$, defined by (1.29.1), is an analytic function of s, regular for $a < \sigma < b$; and $\mathfrak{F}(s) = \begin{cases} O(e^{-(\beta - \epsilon)t}) & (t \to \infty) \\ O(e^{(\alpha - \epsilon)t}) & (t \to -\infty) \end{cases}$

for every positive ϵ , uniformly in any strip interior to $a < \sigma < b$; and (1.29.3) holds for a < k < b.

Conversely, if $\mathfrak{F}(s)$ is a given function satisfying the above conditions and f(x) is defined by (1.29.3), then f(x) satisfies the conditions previously imposed on it, and (1.29.1) holds.

This follows from Theorem 26, or it may easily be proved by an analogous argument.

THEOREM 32. Let $f(x)x^{k-1}$ be $L(0,\infty)$; or, more generally, let

$$\int_{-\infty}^{\infty} f(x)x^{s-1} dx = \mathfrak{F}(s)$$
 (1.29.5)

be uniformly convergent for s = k+it, t in any finite interval. Then

$$\frac{1}{2\pi i} \lim_{\lambda \to \infty} \int_{k-i\lambda}^{k+i\lambda} \left(1 - \frac{|t|}{\lambda}\right) \mathfrak{F}(s) x^{-s} \, ds$$

is equal to $\frac{1}{2}\{f(x+0)+f(x-0)\}$ wherever this expression has a meaning, and in particular to f(x) wherever f(x) is continuous; and to f(x) for almost all x.

In the inverse form the assumption is that $\mathfrak{F}(k+it)$ is L, and the conclusion

$$\lim_{\mu\to\infty}\int_{1/\mu}^{\mu}\left(1-\frac{|\log x|}{\log\mu}\right)f(x)x^{s-1}\,dx=\mathfrak{F}(s)$$

almost everywhere.

This follows from Theorem 27 by the usual changes of variable. A particular case† is that, if

$$\int_{\longrightarrow 0} f(x)x^{a-1} dx, \qquad \int_{\longrightarrow 0}^{\infty} f(x)x^{b-1} dx,$$

where a < b, converge, then the result holds for a < k < b; for then † Hardy (8).

(1.29.5) converges uniformly in any finite region interior to the strip $a < \sigma < b$. In this case $\mathfrak{F}(s)$ is analytic in the strip.

1.30. The Laplace formulae. The simplest theorem on the formula (1.4.3) is

THEOREM 33. A necessary and sufficient condition that f(z) should be of the form $f(z) = \frac{1}{2\pi i} \int \phi(w) e^{ziv} dw, \qquad (1.30.1)$

where Γ is a closed contour surrounding the origin, is that it should be an integral function of exponential type, i.e. such that $f(z) = O(e^{c|z|})$ for some c.

The formula (1.30.1) plainly defines an integral function of z; and, if $|w| \leq c$ on the contour, $f(z) = O(e^{c|z|})$. Hence the condition is necessary.

Conversely, suppose that it is satisfied, and let

$$f(z) = \sum_{n=0}^{\infty} a_n z^n.$$

Then by Cauchy's inequality

$$|a_n| \leqslant \frac{1}{r^n} \max_{|z|=r} |f(z)| < \frac{Ke^{cr}}{r^n}$$

for all values of r. Taking r = n,

$$|a_n| < Ke^{cn}n^{-n}.$$

Hence the series

$$\phi(w) = \sum_{n=0}^{\infty} \frac{n! \, a_n}{w^{n+1}}$$

is convergent if w is sufficiently large, say for |w| > M. Let Γ be a simple closed curve surrounding the origin, and lying entirely outside the circle |w| = M. Then by uniform convergence

$$\frac{1}{2\pi i} \int_{\Gamma} \phi(w) e^{zw} dw = \sum_{n=0}^{\infty} \frac{n! \, a_n}{2\pi i} \int_{\Gamma} \frac{e^{zw}}{w^{n+1}} dw = \sum_{n=0}^{\infty} a_n z^n = f(z),$$

the required result.

The reciprocal formula is

$$\phi(w) = \int_0^\infty f(x)e^{-xw} dx$$

as in § 1.4, but in general this holds for R(w) > c only.

For f(z) in (1.30.1) to vanish identically, it is plainly not necessary for $\phi(w)$ to vanish; it is sufficient for $\phi(w)$ to be regular within Γ . Hence, if we are given f(z) and Γ , (1.30.1) does not determine $\phi(w)$ uniquely. It does so, however, if $\phi(w)$ is given to be regular and zero at infinity; and there is a more general result of the same kind.

Theorem 34. Let $\phi(w)$ be regular for sufficiently large w, except for a pole of order n at infinity, and let

$$\int\limits_{V} \phi(w)e^{wt}\,dw=0$$

for all t, Γ being a simple closed contour surrounding the origin. Then

$$\phi(w) = a_0 + a_1 w + ... + a_n w^n.$$

Let

$$\psi(w) = \phi(w) - a_0 - \dots - a_n w^n,$$

where $a_0 + ... + a_n w^n$ is the principal part of $\phi(w)$ at infinity. Then

$$\int_{\Gamma} \psi(w)e^{wt}\,dw=0,$$

and $\psi(w) \to 0$ as $|w| \to \infty$.

Multiply by e^{-zt} , where $\mathbf{R}(z) > \max_{\Gamma} \mathbf{R}(w)$, and integrate over

 $(0,\infty)$. We obtain

$$\int\limits_{\Gamma} \psi(w) \frac{1}{z-w} \, dw = 0,$$

and this holds by analytic continuation for any z outside Γ .

Hence, by the calculus of residues, if Γ' is a circle of radius R > |z|,

 $\psi(z) = \frac{1}{2\pi i} \int_{V} \frac{\psi(w)}{w-z} dw,$

and, making $R \to \infty$, the right-hand side tends to 0. Hence $\psi(z) = 0$.

AUXILIARY FORMULAE

2.1. Formalities. If F(x) and G(x) are the Fourier transforms of f(x) and g(x), we have formally

$$\int_{-\infty}^{\infty} F(x)G(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} G(x) dx \int_{-\infty}^{\infty} f(t)e^{ixt} dt$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(t) dt \int_{-\infty}^{\infty} G(x)e^{ixt} dx = \int_{-\infty}^{\infty} f(t)g(-t) dt. \quad (2.1.1)$$

If g(t) is replaced by $\bar{g}(-t)$, G(x) is replaced by $\bar{G}(x)$, so that an equivalent formula is

$$\int_{-\infty}^{\infty} F(x)\overline{G}(x) dx = \int_{-\infty}^{\infty} f(x)\overline{g}(x) dx. \qquad (2.1.2)$$

If g = f, we obtain

$$\int_{-\infty}^{\infty} |F(x)|^2 dx = \int_{-\infty}^{\infty} |f(x)|^2 dx.$$
 (2.1.3)

For even functions the formulae reduce to

$$\int_{0}^{\infty} F_{c}(x)G_{c}(x) dx = \int_{0}^{\infty} f(x)g(x) dx, \qquad (2.1.4)$$

and

$$\int_{0}^{\infty} \{F_{c}(x)\}^{2} dx = \int_{0}^{\infty} \{f(x)\}^{2} dx; \qquad (2.1.5)$$

for odd functions they reduce to

$$\int_{0}^{\infty} F_s(x)G_s(x) dx = \int_{0}^{\infty} f(x)g(x) dx \qquad (2.1.6)$$

and

$$\int_{0}^{\infty} \{F_{s}(x)\}^{2} dx = \int_{0}^{\infty} \{f(x)\}^{2} dx.$$
 (2.1.7)

These formulae are analogous to Parseval's formula

$$\frac{1}{\pi} \int_{0}^{2\pi} \{f(x)\}^{2} dx = \frac{1}{2}a_{0}^{2} + \sum_{n=1}^{\infty} (a_{n}^{2} + b_{n}^{2})$$

in the theory of Fourier series. They will be known generally as the Parseval formulae.† Again,

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(t)G(t)e^{-ixt} dt = \frac{1}{2\pi} \int_{-\infty}^{\infty} F(t)e^{-ixt} dt \int_{-\infty}^{\infty} g(u)e^{itu} du$$

$$= \frac{1}{2\pi} \int_{-\infty}^{\infty} g(u) du \int_{-\infty}^{\infty} F(t)e^{-it(x-u)} dt$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} g(u)f(x-u) du. \qquad (2.1.8)$$

Thus the functions

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} g(u)f(x-u) du, \qquad F(x)G(x), \qquad (2.1.9)$$

are Fourier transforms. The integral obtained is called the *resultant* of f(x) and g(x).

The process may clearly be repeated. The functions

$$\frac{1}{2\pi} \int_{-\infty}^{\infty} h(v) dv \int_{-\infty}^{\infty} g(u) f(x-u-v) du, \qquad F(x) G(x) H(x)$$
(2.1.10)

are transforms. So generally are

$$\frac{1}{(2\pi)^{\frac{1}{2}n}} \int_{-\infty}^{\infty} f_n(u_n) du_n \int_{-\infty}^{\infty} f_{n-1}(u_{n-1}) du_{n-1} \dots \times \\ \times \int_{-\infty}^{\infty} f_1(u_1) f(x-u_1-\dots-u_n) du_1 \\ F(x) F_1(x) \dots F_n(x).$$

There are analogous formulae for Mellin transforms, which may be obtained by transformation from the above, or directly as follows. If $\mathfrak{F}(s)$, $\mathfrak{G}(s)$ are the Mellin transforms of f(x) and g(x),

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)\mathfrak{G}(1-s) \, ds = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{G}(1-s) \, ds \int_{0}^{\infty} f(x)x^{s-1} \, dx$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} f(x) \, dx \int_{k-i\infty}^{k+i\infty} \mathfrak{G}(1-s)x^{s-1} \, ds = \int_{0}^{\infty} f(x)g(x) \, dx, \quad (2.1.12)$$

† The earliest reference to the formulae of which I know is in Rayleigh (1). See also Hardy (3-5), Ramanujan (1).

or, alternatively,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)\mathfrak{G}(1-s) \ ds = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s) \ ds \int_{0}^{\infty} g(x)x^{-s} \ dx$$

$$= \frac{1}{2\pi i} \int_{0}^{\infty} g(x) \ dx \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)x^{-s} \ ds = \int_{0}^{\infty} g(x)f(x) \ dx.$$

Similarly,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)\mathfrak{G}(s) ds = \frac{1}{2\pi i} \int_{0}^{\infty} g(x) dx \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s) x^{s-1} ds$$

$$= \int_{0}^{\infty} g(x) f\left(\frac{1}{x}\right) \frac{dx}{x}. \tag{2.1.13}$$

If g = f, and both are real, (2.1.12) with $k = \frac{1}{2}$ gives

$$\frac{1}{2\pi} \int_{-\infty}^{\infty} |\mathfrak{F}(\frac{1}{2} + it)|^2 dt = \int_{0}^{\infty} \{f(x)\}^2 dx. \tag{2.1.14}$$

Also

$$\int_{0}^{\infty} f(x)g(x)x^{s-1} dx = \frac{1}{2\pi i} \int_{0}^{\infty} g(x)x^{s-1} dx \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(w)x^{-w} dw$$

$$= \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(w) dw \int_{0}^{\infty} g(x)x^{s-w-1} dx$$

$$= \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(w)\mathfrak{G}(s-w) dw, \qquad (2.1.15)$$

i.e.

$$f(x)g(x), \qquad \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(w)\mathfrak{G}(s-w) \ dw \qquad (2.1.16)$$

are Mellin transforms.

Again,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)\mathfrak{G}(s)x^{-s} ds = \frac{1}{2\pi i} \int_{0}^{\infty} g(u) du \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)u^{s-1}x^{-s} ds$$

$$= \int_{0}^{\infty} g(u)f\left(\frac{x}{u}\right) \frac{du}{u}, \qquad (2.1.17)$$

another sort of resultant.

Hence we obtain as Mellin transforms

$$\int_{0}^{\infty} g(u) f\left(\frac{x}{u}\right) \frac{du}{u}, \qquad \mathfrak{F}(s)\mathfrak{G}(s). \tag{2.1.18}$$

Repeating the process, we obtain as Mellin transforms

$$\int_{0}^{\infty} f_{n}(u_{n}) \frac{du_{n}}{u_{n}} \int_{0}^{\infty} f_{n-1}(u_{n-1}) \frac{du_{n-1}}{u_{n-1}} \dots \int_{0}^{\infty} f_{1}(u_{1}) f\left(\frac{x}{u_{1} u_{2} \dots u_{n}}\right) \frac{du_{1}}{u_{1}},$$

$$\Re(s) \Re_{1}(s) \dots \Re_{n}(s).$$
(2.1.19)

From the Laplace integral formulae we derive similarly the formulae

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \phi(s)\psi(s)e^{sx} ds = \frac{1}{2\pi i} \int_{0}^{\infty} \psi(s)e^{sx} ds \int_{0}^{\infty} f(y)e^{-sy} dy
= \frac{1}{2\pi i} \int_{0}^{\infty} f(y) dy \int_{0}^{\infty} \psi(s)e^{s(x-y)} ds
= \int_{0}^{x} f(y)g(x-y) dy,$$
(2.1.20)

and

$$\frac{1}{2\pi i} \int \phi(s) \psi(-s) e^{sx} \, ds = \frac{1}{2\pi i} \int_{0}^{\infty} f(y) \, dy \int \psi(-s) e^{s(x-y)} \, ds$$

$$= \int_{\max(x,0)}^{\infty} f(y) g(y-x) \, dy. \qquad (2.1.21)$$

We can also introduce parameters into the formulae without altering them essentially. Since

$$\frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty}f(ay)e^{ixy}\,dy=\frac{1}{a\sqrt{(2\pi)}}\int_{-\infty}^{\infty}f(u)e^{ixu|a}\,du=\frac{1}{a}F\left(\frac{x}{a}\right),$$

the transform of f(ay) is $\frac{1}{a}F(\frac{x}{a})$. Thus e.g.

$$\int_{-\infty}^{\infty} f(at)g(-bt) dt = \frac{1}{ab} \int_{-\infty}^{\infty} F\left(\frac{x}{a}\right) G\left(\frac{x}{b}\right) dx.$$
 (2.1.22)

Similar changes may be made in the other formulae; e.g.

$$\int_{0}^{\infty} f(ax)x^{s-1} dx = a^{-s} \int_{0}^{\infty} f(\xi)\xi^{s-1} d\xi,$$

so that the Mellin transform of f(ax) is $a^{-s}\mathfrak{F}(s)$. Thus

$$\int_{0}^{\infty} f(ax)g(bx) \ dx = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)\mathfrak{G}(1-s)a^{-s}b^{s-1} \ ds, \quad (2.1.23)$$

and similarly in the other formulae.

2.2. Conditions for validity. We shall now give some sets of conditions for the validity of the above formulae. Some of the most important conditions depend on the theory of mean convergence, and must be postponed until later chapters. The conditions which we give here depend on analysis resembling that of Chapter I.

We begin with (2.1.1) and its special cases.

THEOREM 35. If f(x) and G(x) belong to $L(-\infty,\infty)$, and F(x) and g(x) are their transforms, then (2.1.1) holds.

For the inversions used in obtaining (2.1.1) are justified by absolute convergence. The theorem implies that f and G are the given functions, and F and g defined in terms of them.

The theorem of course includes the corresponding theorems for cosine and sine transforms.

It follows also that, if f(x) and g(x) are $L(-\infty,\infty)$, and G(x), defined as the transform of g(x), is $L(-\infty,\infty)$, then (2.1.1) holds. For, by Theorem 27, g(x) is the transform of G(x).

2.3. We next take some cases of Parseval's formula suggested by Theorem 6. Here the conditions are more appropriate to the half-line $(0,\infty)$, and we consider cosine and sine transforms separately.

THEOREM 36.† Let f(x) belong to $L(0,\infty)$, and, in some interval ending at 0, tend steadily to a limit as $x \to 0$. Let g(x) be the cosine transform of $G_c(x)$, which is integrable over any finite interval, and tends steadily to 0 as $x \to \infty$. Then

$$\int_{0}^{\infty} F_c(x)G_c(x) dx = \int_{-\infty}^{\infty} f(x)g(x) dx. \qquad (2.3.1)$$

We have to justify the inversion

$$\int_{0}^{\infty} G_{c}(y) dy \int_{0}^{\infty} f(x) \cos xy dx = \int_{-\infty}^{\infty} f(x) dx \int_{0}^{\infty} G_{c}(y) \cos xy dy.$$
(2.3.2)

Now
$$\int_{0}^{\lambda} G_{c}(y) dy \int_{\delta}^{\infty} f(x) \cos xy dx = \int_{\delta}^{\infty} f(x) dx \int_{0}^{\lambda} G_{c}(y) \cos xy dy,$$
 (2.3.3)

for every finite λ , by uniform convergence. By the second mean-value theorem

$$\int_{\lambda}^{\infty} G_c(y) \cos xy \, dy = G_c(\lambda) \int_{\lambda}^{\lambda'} \cos xy \, dy = O\left\{\frac{G_c(\lambda)}{x}\right\}.$$

$$\lim_{\lambda \to \infty} \int_{\lambda}^{\infty} f(x) \, dx \int_{\lambda}^{\infty} G_c(y) \cos xy \, dy = 0, \qquad (2.3.4)$$

Hence

and (2.3.3), (2.3.4) give

$$\int_{0}^{+\infty} G_c(y) \, dy \int_{\delta}^{\infty} f(x) \cos xy \, dx = \int_{\delta}^{\infty} f(x) \, dx \int_{0}^{+\infty} G_c(y) \cos xy \, dy$$
(2.3.5)

for every $\delta > 0$. It is now sufficient to prove that

$$\lim_{\delta \to 0} \int_0^{+\infty} G_c(y) \, dy \int_0^{\delta} f(x) \cos xy \, dx = 0. \tag{2.3.6}$$

If, e.g., f(x) is steadily decreasing in $(0, \delta)$,

$$\int_{y_{1}}^{y} G_{c}(y) \, dy \int_{0}^{\delta} f(x) \cos xy \, dx = \int_{0}^{\delta} f(x) \, dx \int_{y_{1}}^{y_{2}} G_{c}(y) \cos xy \, dy$$

$$= f(+0) \int_{0}^{\xi} dx \int_{y_{1}}^{y_{2}} G_{c}(y) \cos xy \, dy$$

$$= f(+0) \int_{y_{1}}^{y_{2}} G_{c}(y) \frac{\sin \xi y}{y} \, dy,$$

where $0 < \xi < \delta$; and

$$\int_{Y}^{y_{s}} G_{c}(y) \frac{\sin \xi y}{y} \, dy = G_{c}(Y) \int_{Y}^{\eta} \frac{\sin \xi y}{y} \, dy = O\{G_{c}(Y)\}$$

for all ξ , while, for a fixed Y,

$$\int_{y_1}^{Y} G_c(y) \frac{\sin \xi y}{y} \, dy \to 0$$

as $\xi \to 0$. The result therefore follows on choosing first Y sufficiently large and then δ sufficiently small.

THEOREM 37. The corresponding theorem for sine transforms holds provided that, in addition, $G_s(x)/x$ belongs to $L(1,\infty)$.

In this case we encounter

$$\int_{y_s}^{y_s} G_s(y) \frac{1-\cos \xi y}{y} \, dy$$

at the last stage of the proof, and the extra condition is required here.

2.4. In the above theorems the functions on which the conditions bear are on opposite sides of the Parseval formula. We next prove a theorem in which they are on the same side. †

Theorem 38. Let f(x) belong to $L(0,\infty)$. Let g(x) be positive, nonincreasing, and tend to 0 as $x \to \infty$, and let

$$\int_{0}^{1} \frac{|f(t)|}{t} dt \int_{0}^{t} g(u) du < \infty.$$

$$\int_{0}^{\infty} F_{c}(x)G_{c}(x) dx = \int_{0}^{\infty} f(x)g(x) dx,$$

$$(2.4.2)$$

Then

and similarly for sine transforms.

We have

$$\int_{\xi}^{X} F_{c}(x)G_{c}(x) dx = \sqrt{\left(\frac{2}{\pi}\right)} \int_{\xi}^{X} G_{c}(x) dx \int_{0}^{\infty} f(t)\cos xt dt$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(t) dt \int_{\xi}^{X} G_{c}(x)\cos xt dx$$

$$= \frac{2}{\pi} \int_{0}^{\infty} f(t) dt \int_{\xi}^{X} \cos xt dx \int_{0}^{\infty} g(u)\cos xu du$$

$$= \frac{2}{\pi} \int_{0}^{\infty} f(t) dt \int_{0}^{X} g(u) du \int_{\xi}^{X} \cos xt \cos xu dx$$

$$= 2 \int_{0}^{\infty} f(t) \{g(t, X) - g(t, \xi) + g(-t, X) - g(-t, \xi)\} dt,$$
where
$$g(t, x) = \frac{1}{\pi} \int_{0}^{\infty} g(u) \frac{\sin x(u - t)}{u - t} du.$$
(2.4.5)

where

† See Hardy and Titchmarsh (5).

The inversion of the x and t integrations is justified by the uniform convergence of the t-integral, and that of the x and u integrations by that of the u-integral.

Since g(t) is non-increasing, $g(t, X) \rightarrow g(t)$ as $X \rightarrow \infty$ for almost all positive t, and $g(-t, X) \to 0$. Also as $\xi \to 0$

$$g(t,\xi) = \int_{0}^{U} g(u) \frac{\sin \xi(u-t)}{u-t} du + O\{g(U)\}$$
$$= O\left(\xi \int_{0}^{U} g(u) du\right) + O\{g(U)\} = o(1)$$

by choosing first U and then ξ .

Now

$$\left|\int\limits_{\frac{1}{2}t}^{t}g(u)\frac{\sin x(u-t)}{u-t}\,du\right| = \left|g(\frac{1}{2}t+0)\int\limits_{\frac{1}{2}t}^{T}\frac{\sin x(u-t)}{u-t}\,du\right|$$

$$< Ag(\frac{1}{2}t) < \frac{A}{t}\int\limits_{0}^{\frac{1}{2}t}g(u)\,du < \frac{A}{t}\int\limits_{0}^{t}g(u)\,du,$$
and
$$\left|\int\limits_{0}^{\frac{1}{2}t}g(u)\frac{\sin x(u-t)}{u-t}\,du\right| \leqslant \int\limits_{0}^{\frac{1}{2}t}\frac{g(u)}{t-u}\,du \leqslant \frac{2}{t}\int\limits_{0}^{\frac{1}{2}t}g(u)\,du$$

$$\leqslant \frac{2}{t}\int\limits_{0}^{t}g(u)\,du.$$
Hence
$$|f(t)g(t,x)| \leqslant A\frac{|f(t)|}{t}\int\limits_{0}^{t}g(u)\,du,$$

Hence

which belongs to L(0,1), by hypothesis; and it belongs to $L(1,\infty)$, since f(t) belongs to $L(1,\infty)$, and

$$\frac{1}{t}\int_{0}^{t}g(u)\ du\to 0.$$

The result now follows from (2.4.5) on making $X \to \infty$, $\xi \to 0$, by dominated convergence.

Immediate corollaries are

- (i) If f(x) belongs to $L(0,\infty)$, and g(x) is of bounded variation in $(0,\infty)$, and tends to 0 at infinity, then (2.4.3) holds.
- (ii) If f is L and g bounded in (0,1), then (2.4.2) holds and the theorem follows.

(iii) If f is bounded and $g(x)\log(1/x)$ is L in (0, 1), (2.4.2) holds and the theorem follows.

Apparently f bounded and g L in (0,1) is not sufficient.

2.5. In a later chapter we find some examples of Parseval's formula which evade all the above theorems. These are cases where the existence of the transforms and the convergence of the integrals involved is obvious enough, and all that is needed is to prove the equality of the two sides of Parseval's formula. We can deal with some such cases by means of the following theorem.

THEOREM 39. Let f(x) and g(x) be integrable over any finite interval. Let

$$F(x,a) = rac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(t)e^{ixt} dt, \qquad G(x,a) = rac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} g(t)e^{ixt} dt, \ \chi(x,a,b) = \int_{-a}^{b} g(u)f(x-u) du$$

and

be all $O(e^{c|x|})$ for some positive c, independently of a and b, and tend to F(x), G(x), and $\chi(x)$ as $a \to \infty$, $b \to \infty$, for almost all x. Then

$$\lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} F(x)G(x) dx = \frac{1}{2} \{ \chi(+0) + \chi(-0) \}$$

provided that the limits indicated exist.

Let $\lambda > 0$. Then by dominated convergence

$$\int_{-\infty}^{\infty} F(t)e^{-\frac{1}{4}t^{a}|\lambda+iut|} dt = \lim_{a \to \infty} \int_{-\infty}^{\infty} F(t,a)e^{-\frac{1}{4}t^{a}|\lambda+iut|} dt$$

$$= \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(x) dx \int_{-\infty}^{\infty} e^{-\frac{1}{4}t^{a}|\lambda+iut+ixt|} dt$$

$$= \lim_{a \to \infty} \sqrt{(2\lambda)} \int_{-a}^{a} f(x)e^{-\lambda(x+u)^{a}} dx,$$

and the convergence is uniform over a finite u-interval. Hence

$$\int_{-b}^{b} g(u) du \int_{-\infty}^{\infty} F(t)e^{-\frac{1}{2}t^{a}/\lambda + iut} dt = \lim_{a \to \infty} \sqrt{(2\lambda)} \int_{-b}^{b} g(u) du \int_{-a}^{a} f(x)e^{-\lambda(x+u)^{a}} dx$$

$$= \lim_{a \to \infty} \sqrt{(2\lambda)} \int_{-b}^{b} g(u) du \int_{-a+u}^{a+u} f(x-u)e^{-\lambda x^{a}} dx$$

For

$$= \lim_{a \to \infty} \sqrt{(2\lambda)} \int_{-a-b}^{a+b} e^{-\lambda x^{2}} dx \int_{\max(x-a,-b)}^{\min(x+a,b)} g(u) f(x-u) du$$

$$= \sqrt{(2\lambda)} \int_{-\infty}^{\infty} e^{-\lambda x^{2}} \chi(x,b,b) dx,$$

by dominated convergence. Also we may invert the left-hand side, by uniform convergence, and obtain

$$\sqrt{(2\pi)}\int_{-\infty}^{\infty}F(t)G(t,b)e^{-\frac{1}{2}t^{2}/\lambda}dt.$$

Hence, making $b \to \infty$, and using dominated convergence,

$$\int_{-\infty}^{\infty} F(t)G(t)e^{-\frac{1}{2}t^{\mu}/\lambda} dt = \sqrt{\left(\frac{\lambda}{\pi}\right)} \int_{-\infty}^{\infty} e^{-\lambda x^{\mu}} \chi(x) dx.$$

Making $\lambda \to \infty$, the result now follows from Theorem 16.

In particular, the result holds if f and g belong to $L(-\infty,\infty)$, and one of them is bounded.

2.6. Transform of a resultant. We now turn to (2.1.8), giving the Fourier transform of a product, or of a resultant. From one point of view this is merely a case of Parseval's formula, since f(x-u) is the transform of $F(t)e^{-ixt}$. A new problem arises, however, when we consider all values of x at once. We then ask whether (2.1.9) are transforms belonging to one of the general classes already considered.

THEOREM 40. Let f(x) be the transform of a function F(x) of $L(-\infty,\infty)$, and let g(x) belong to $L(-\infty,\infty)$ (so that its transform G(x) is bounded). Then $\sqrt{(2\pi)}F(x)G(x)$ belongs to $L(-\infty,\infty)$, and its transform is

 $k(x) = \int\limits_{-\infty}^{\infty} g(u)f(x-u) du.$

For the inversion in (2.1.8) is justified by absolute convergence.

THEOREM 41. Let f(x) and g(x) belong to $L(-\infty,\infty)$. Then so does k(x), and its transform is $\sqrt{(2\pi)}F(x)G(x)$.

 $\int_{-a}^{a} k(u)e^{ixu} du = \int_{-a}^{a} e^{ixu} du \int_{-\infty}^{\infty} f(v)g(u-v) dv$ $= \int_{-\infty}^{\infty} f(v) dv \int_{-a}^{a} g(u-v)e^{iux} du$ $= \int_{-\infty}^{\infty} f(v)e^{ivx} dv \int_{-a-v}^{a-v} g(t)e^{ixt} dt.$

The inner integral converges boundedly to $\sqrt{(2\pi)G(x)}$. Hence

$$\int_{-\infty}^{\infty} k(u)e^{ixu} du = \sqrt{(2\pi)} \int_{-\infty}^{\infty} f(v)e^{ivx}G(x) dv = 2\pi F(x)G(x).$$

2.7. Mellin transforms.

THEOREM 42. Let $x^{k-1}f(x)$ be $L(0,\infty)$, and $\mathfrak{G}(1-k-it)$ be $L(-\infty,\infty)$, or alternatively let $\mathfrak{F}(k+it)$ be $L(-\infty,\infty)$, and $x^{-k}g(x)$ be $L(0,\infty)$. Then (2.1.12) holds.

For the inversion which gives the formula is justified by absolute convergence.

THEOREM 43. Let f(x) and g(x) be integrable over any finite interval not ending at x = 0. Let

$$\mathfrak{F}(s,a) = \int_{1/a}^{a} f(x)x^{s-1} dx, \qquad \mathfrak{G}(s,a) = \int_{1/a}^{a} g(x)x^{s-1} dx$$

tend to $\mathfrak{F}(s)$, $\mathfrak{G}(s)$ for $\sigma = k$, $\sigma = 1-k$ respectively, for almost all t, in such a way that $e^{-c|s|}\mathfrak{F}(s,a)$, $e^{-c|s|}\mathfrak{G}(s,a)$ are, for some positive c, bounded independently of a. Let

$$\frac{\xi^{1-k}}{\xi^c + \xi^{-c}} \int_a^b f(x) g(\xi x) \ dx.$$

be bounded for all a, b, ξ , and, as $a \to 0$, $b \to \infty$, converge to a continuous limit in the neighbourhood of $\xi = 1$. Then

$$\frac{1}{2\pi i} \lim_{\lambda \to \infty} \int_{k-i\lambda}^{k+i\lambda} \mathfrak{F}(s) \mathfrak{G}(1-s) \, ds = \int_{-\infty}^{+\infty} f(x) g(x) \, dx,$$

provided the left-hand side exists.

This follows at once from Theorem 39.

The analogue of Theorem 41 is

THEOREM 44. Let $x^k f(x)$ and $x^k g(x)$ belong to $L(0,\infty)$, and let

$$h(x) = \int_{0}^{\infty} f(y)g\left(\frac{x}{y}\right)\frac{dy}{y}.$$

Then $x^kh(x)$ belongs to $L(0,\infty)$, and its Mellin transform is $\mathfrak{F}(s)\mathfrak{G}(s)$, with $\sigma = k+1$.

2.8. Poisson's formula. This is

$$\sqrt{\beta}\left\{\frac{1}{2}F_c(0) + \sum_{n=1}^{\infty} F_c(n\beta)\right\} = \sqrt{\alpha}\left\{\frac{1}{2}f(0) + \sum_{n=1}^{\infty} f(n\alpha)\right\}, \qquad (2.8.1)$$

where $\alpha\beta = 2\pi$, $\alpha > 0$.

We shall provet

THEOREM 45. Let f(x) be of bounded variation in $(0,\infty)$, and tend to 0 as $x \to \infty$. Then

$$\sqrt{\beta} \sum_{n=1}^{\infty} F_c(n\beta) = \sqrt{\alpha} \lim_{M \to \infty} \left[\frac{1}{2} f(0+0) + \sum_{m=1}^{M} \frac{1}{2} \{ f(m\alpha - 0) + f(m\alpha + 0) \} - \frac{1}{\alpha} \int_{0}^{(M+\frac{1}{2})\alpha} f(t) dt \right].$$
(2.8.2)

$$\sqrt{\beta} \left\{ \frac{1}{2} F_c(0) + \sum_{n=1}^{\infty} F_c(n\beta) \right\} \\
= \sqrt{\alpha} \left[\frac{1}{2} f(0+0) + \sum_{m=1}^{\infty} \frac{1}{2} \left\{ f(m\alpha - 0) + f(m\alpha + 0) \right\} \right]. \quad (2.8.3)$$

If also f(x) is continuous, then (2.8.1) holds.

Since f(t) is the difference between two non-increasing functions, each of which $\to 0$ as $x \to \infty$, we may take it to be one such function.

The integral

$$F_c(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{-\infty} f(t) \cos xt \ dt$$

exists for x > 0, and

$$\sqrt{\beta} \sum_{m=1}^{n} F_{c}(m\beta)
= \sqrt{\left(\frac{2\beta}{\pi}\right)} \int_{0}^{\infty} f(t) \sum_{m=1}^{n} \cos m\beta t = \sqrt{\left(\frac{2\beta}{\pi}\right)} \int_{0}^{\infty} f(t) \left\{\frac{\sin(n+\frac{1}{2})\beta t}{2\sin\frac{1}{2}\beta t} - \frac{1}{2}\right\} dt
= \sqrt{\left(\frac{2\beta}{\pi}\right)} \sum_{m=0}^{M} \int_{2m\pi/\beta}^{(2m+1)\pi/\beta} \left\{f(t) - f\left(\frac{2m\pi}{\beta} + 0\right)\right\} \frac{\sin(n+\frac{1}{2})\beta t}{2\sin\frac{1}{2}\beta t} dt +
+ \sqrt{\left(\frac{2\beta}{\pi}\right)} \sum_{m=1}^{M} \int_{(2m-1)\pi/\beta}^{2m\pi/\beta} \left\{f(t) - f\left(\frac{2m\pi}{\beta} - 0\right)\right\} \frac{\sin(n+\frac{1}{2})\beta t}{2\sin\frac{1}{2}\beta t} dt -
- \sqrt{\left(\frac{\pi}{2\beta}\right)} \left[f(0+0) + \sum_{m=1}^{M} \left\{f\left(\frac{2m\pi}{\beta} + 0\right) + f\left(\frac{2m\pi}{\beta} - 0\right)\right\}\right] -
- \sqrt{\left(\frac{\beta}{2\pi}\right)} \int_{0}^{(2M+1)\pi/\beta} f(t) dt. \quad (2.8.4)$$

† I do not know whether this version of the theorem has been published previously. I obtained it by combining one of my own with one communicated to me by Mr. W. L. Ferrar. For other methods see Linfoot (1), Mordell (1).

Now

$$\int_{2m\pi/\beta}^{(2m+1)\pi/\beta} \left\{ f(t) - f\left(\frac{2m\pi}{\beta} + 0\right) \right\} \frac{\sin(n + \frac{1}{2})\beta t}{2\sin\frac{1}{2}\beta t} dt$$

$$= \int_{0}^{\pi/\beta} \left\{ f\left(\frac{2m\pi}{\beta} + t\right) - f\left(\frac{2m\pi}{\beta} + 0\right) \right\} \frac{\sin(n + \frac{1}{2})\beta t}{2\sin\frac{1}{2}\beta t} dt$$

$$= \left\{ f\left(\frac{(2m+1)\pi}{\beta} - 0\right) - f\left(\frac{2m\pi}{\beta} + 0\right) \right\} \int_{\xi}^{\pi/\beta} \frac{\sin(n + \frac{1}{2})\beta t}{2\sin\frac{1}{2}\beta t} dt,$$

by the second mean-value theorem. The last integral is bounded for all n and ξ , e.g. as

$$\int_{\xi}^{\pi/\beta} \sin(n+\frac{1}{2})\beta t \left(\frac{1}{2\sin\frac{1}{2}\beta t} - \frac{1}{\beta t}\right) dt + \frac{1}{\beta} \int_{(n+\frac{1}{2})\beta\xi}^{(n+\frac{1}{2})\pi} \frac{\sin u}{u} du,$$

$$\sum \left| f\left(\frac{(2m+1)\pi}{\beta} - 0\right) - f\left(\frac{2m\pi}{\beta} + 0\right) \right|$$

and

is convergent. Hence the first series on the right-hand side of (2.8.4) is convergent as $M \to \infty$, uniformly with respect to n; and each term tends to 0 as $n \to \infty$. Hence the limit of the sum is 0. Similarly for the second series. This proves (2.8.2); and (2.8.3) and (2.8.1) clearly follow from (2.8.2) in the cases stated.

There are also more complicated formulae of the same type. For example, Ramanujan† gives

$$\sqrt{\beta} \{ F_c(\beta) - F_c(3\beta) - F_c(5\beta) + F_c(7\beta) + \ldots \}
= \sqrt{\alpha} \{ f(\alpha) - f(3\alpha) - f(5\alpha) + \ldots \},$$

where $\alpha\beta = \frac{1}{4}\pi$; and

$$\sqrt{\beta} \{ F_c(\beta) - F_c(5\beta) - F_c(7\beta) + F_c(11\beta) + F_c(13\beta) - ... \}$$

$$= \sqrt{\alpha} \{ f(\alpha) - f(5\alpha) - ... \},$$

where $\alpha\beta = \frac{1}{6}\pi$, and 1, 5, 7, 11, 13,... are the numbers prime to 6. These formulae are easily verified by the above method.

2.9. There is another interesting formal method of procedure.‡ Suppose that f(x) is represented by Mellin's integral

$$f(x) = \frac{1}{2\pi i} \int_{c-i\pi}^{c+i\infty} \mathfrak{F}(s) x^{-s} ds.$$

† Ramanujan (2).

† Ferrar (2).

Then formally

$$\sum_{n=1}^{\infty} f(n\alpha) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \mathfrak{F}(s) \sum_{n=1}^{\infty} (n\alpha)^{-s} ds \quad (c>1)$$

$$= \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \mathfrak{F}(s) \zeta(s) \alpha^{-s} ds.$$

Move the line of integration from $\sigma = c$ to $\sigma = -b$, where b > 0; $\zeta(s)$ has a simple pole at s = 1, with residue 1; and

$$\mathfrak{F}(s) = \frac{f(0)}{s} + \int_{0}^{1} \{f(x) - f(0)\} x^{s-1} dx + \int_{1}^{\infty} f(x) x^{s-1} dx$$

has in general a simple pole at s = 0, with residue f(0). Since $\zeta(0) = -\frac{1}{2}$ we obtain

$$\begin{split} &\frac{1}{2}f(0) + \sum_{n=1}^{\infty} f(n\alpha) - \alpha^{-1}\mathfrak{F}(1) \\ &= \frac{1}{2\pi i} \int_{-b-i\infty}^{-b+i\infty} \mathfrak{F}(s)\zeta(s)\alpha^{-s} \, ds \\ &= \frac{1}{2\pi i} \int_{1+b-i\infty}^{1+b+i\infty} \mathfrak{F}(1-s)\zeta(1-s)\alpha^{s-1} \, ds \\ &= \frac{1}{\alpha\pi i} \int_{1+b-i\infty}^{1+b+i\infty} \mathfrak{F}(1-s)\Gamma(s)\cos\frac{1}{2}s\pi \, \zeta(s) \left(\frac{2\pi}{\alpha}\right)^{-s} \, ds \\ &= \frac{1}{\alpha\pi i} \sum_{n=1}^{\infty} \int_{1+b-i\infty}^{1+b+i\infty} \mathfrak{F}(1-s)\Gamma(s)\cos\frac{1}{2}s\pi \left(\frac{2n\pi}{\alpha}\right)^{-s} \, ds. \end{split}$$

But by (2.1.23), with f and g interchanged, and

$$g(x) = \cos x,$$
 $\mathfrak{G}(s) = \Gamma(s)\cos \frac{1}{2}s\pi,$ $b = 1,$ $F_c(a) = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(1-s)\Gamma(s)\cos \frac{1}{2}s\pi a^{-s} ds.$

We have therefore obtained (2.8.1) again.

We shall not attempt to justify this process here. The main interest of it is that it suggests a method of dealing with sums such as

$$\sum_{n=1}^{\infty} d(n) f(n),$$

where d(n) is the number of divisors of n. This sum, for example, gives

$$\frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \mathfrak{F}(s)\zeta^{2}(s) ds$$

$$= \frac{1}{4}f(0) + \mathfrak{F}'(1) - 2\gamma\mathfrak{F}(1) + \frac{1}{2\pi i} \int_{-b-i\infty}^{-b+i\infty} \mathfrak{F}(s)\zeta^{2}(s) ds,$$

and the last term is

$$\frac{1}{2\pi i} \int_{1+b-i\infty}^{1+b+i\infty} \mathfrak{F}(1-s)\zeta^{2}(1-s) ds$$

$$= \frac{2}{\pi i} \int_{1+b-i\infty}^{1+b+i\infty} \mathfrak{F}(1-s)\Gamma^{2}(s)\cos^{2}\frac{1}{2}s\pi 2^{-2s}\pi^{-2s}\zeta^{2}(s) ds$$

$$= \sum_{n=1}^{\infty} \frac{2d(n)}{\pi i} \int_{1+b+i\infty}^{1+b+i\infty} \mathfrak{F}(1-s)\Gamma^{2}(s)\cos^{2}\frac{1}{2}s\pi (4\pi^{2}n)^{-s} ds.$$

From (7.9.7) and (7.9.11) we deduce

$$\frac{2}{\pi}K_0(x) - Y_0(x) = \frac{1}{2\pi^2i} \int_{k-i\infty}^{k+i\infty} \Gamma^2(\frac{1}{2}s)\cos^2\frac{1}{4}s\pi \, 2^s x^{-s} \, ds \quad (k > 1),$$

and, proceeding as before, the result is

$$\begin{split} \sum_{n=1}^{\infty} d(n)f(n) &= \frac{1}{4}f(0) + \mathfrak{F}'(1) - 2\gamma \mathfrak{F}(1) + \\ &+ \sum_{n=1}^{\infty} d(n) \int_{0}^{\infty} f(x) [K_0 \{4\pi \sqrt{(nx)}\} - \frac{1}{2}\pi Y_0 \{4\pi \sqrt{(nx)}\}] dx. \end{split}$$

2.10. Examples. (i) Let $f(x) = e^{-x}$, $F_c(x) = \sqrt{\left(\frac{2}{\pi}\right)\frac{1}{1+x^2}}$. Then

$$\sqrt{\alpha}\left(\frac{1}{2} + \sum_{n=1}^{\infty} e^{-n\alpha}\right) = \sqrt{\left(\frac{2\beta}{\pi}\right)\left(\frac{1}{2} + \sum_{n=1}^{\infty} \frac{1}{1 + n^2\beta^2}\right)}$$

(ii) Let $f(x) = e^{-\frac{1}{2}x^2}$, $F_c(x) = e^{-\frac{1}{2}x^2}$. Then

$$\sqrt{\alpha}\left(\frac{1}{2} + \sum_{n=1}^{\infty} e^{-\frac{1}{2}\alpha^2 n^2}\right) = \sqrt{\beta}\left(\frac{1}{2} + \sum_{n=1}^{\infty} e^{-\frac{1}{2}\beta^2 n^2}\right).$$

(iii) Let $f(x) = e^{-\frac{1}{2}x^2}\cos kx$. Then $F_c(x) = e^{-\frac{1}{2}(k^2+x^2)}\cosh kx$. Hence

$$\sqrt{\alpha} \left(\frac{1}{2} + \sum_{n=1}^{\infty} e^{-\frac{1}{2}\alpha^{n}n} \cos k\alpha n \right) = \sqrt{\beta} e^{-\frac{1}{2}k^{n}} \left(\frac{1}{2} + \sum_{n=1}^{\infty} e^{-\frac{1}{2}\beta^{n}n} \cosh k\beta n \right).$$

(iv) The function $f(x) = \frac{\sin \sqrt{x}}{1+x}$ satisfies the conditions of Theorem 45, but does not belong to $L(0,\infty)$.

(v) Let
$$f(x) = x^{-s} \sin^2 x$$
 (1 < σ < 2). Then

$$F_c(x) = \frac{1}{\sqrt{(2\pi)}}\Gamma(1-s)\sin{\frac{1}{2}s\pi} (x^{s-1}-\frac{1}{2}|x-2|^{s-1}-\frac{1}{2}|x+2|^{s-1}) \quad (x>2),$$

$$F_c(0) = \frac{1}{\sqrt{(2\pi)}}\Gamma(1-s)\sin{\frac{1}{2}s\pi}(-2^{s-1}).$$

Hence, taking $\alpha = \frac{1}{2}\pi$, $\beta = 4$,

$$\begin{split} \sqrt{\left(\frac{\pi}{2}\right)} & \left\{ \left(\frac{\pi}{2}\right)^{-s} + \left(\frac{3\pi}{2}\right)^{-s} + \ldots \right\} \\ &= \frac{\Gamma(1-s)\sin\frac{1}{2}s\pi}{\sqrt{(2\pi)}} \left\{ -2^{s-2} + \sum_{n=1}^{\infty} \left((4n)^{s-1} - \frac{1}{2}(4n-2)^{s-1} - \frac{1}{2}(4n+2)^{s-1} \right) \right\}, \end{split}$$

σr

$$1 + \frac{1}{3^{s}} + \frac{1}{5^{s}} + \dots = \frac{1}{2}\Gamma(1-s)\sin\frac{1}{2}s\pi \,\pi^{s-1} \times \left\{ -\frac{1}{2} + \sum_{s=1}^{\infty} \left(\frac{1}{(2n)^{1-s}} - \frac{1}{2} \frac{1}{(2n-1)^{1-s}} - \frac{1}{2} \frac{1}{(2n+1)^{1-s}} \right) \right\}.$$

This is the functional equation for $(1-2^{-s})\zeta(s)$.

(vi) Let

$$f(x) = \frac{2^{\frac{1}{4}-\nu}}{\Gamma(\nu+\frac{1}{6})} (1-x^2)^{\nu-\frac{1}{4}} \quad (0 < x < 1), \qquad 0 \quad (x \geqslant 1).$$

Then†

$$F_c(x) = x^{-\nu}J_{\nu}(x)$$
 $(x > 0),$ $F_c(0) = 2^{-\nu}/\Gamma(\nu+1).$

Hence

$$\frac{1}{2^{\nu+1}\Gamma(\nu+1)} + \sum_{n=1}^{\infty} \frac{J_{\nu}(n\beta)}{(n\beta)^{\nu}} = \sqrt{\left(\frac{\alpha}{\beta}\right)} \frac{2^{\frac{1}{2}-\nu}}{\Gamma(\nu+\frac{1}{2})} \left(\frac{1}{2} + \sum_{n \leq 1/\alpha} (1-n^2\alpha^2)^{\nu-\frac{1}{2}}\right),$$

where, in the case $\nu = \frac{1}{2}$, the term $n = 1/\alpha$, if it occurs, is to be halved.

This is a case of Theorem 45 if $\nu \geqslant \frac{1}{2}$. Actually it is easy to see that the same proof applies if $-\frac{1}{2} < \nu < \frac{1}{2}$, provided that α is not the reciprocal of an integer.

2.11. Sine transforms. The corresponding theorem is

THEOREM 46. Let f(x) be integrable over $(0, \delta)$, of bounded variation over (δ, ∞) , where $0 < \delta < \frac{1}{2}\pi$, and tend to 0 at infinity. Then

$$\sqrt{\beta} \{ F_s(\dot{\beta}) - F_s(3\beta) + \ldots \}
= \sqrt{\alpha} \left[\frac{1}{2} \{ f(\alpha + 0) + f(\alpha - 0) \} - \frac{1}{2} \{ f(3\alpha + 0) + f(3\alpha - 0) \} + \ldots \right], \quad (2.11.1)$$

 $= \sqrt{\alpha} \left[\frac{1}{2} \{ J(\alpha + 0) + J(\alpha - 0) \} - \frac{1}{2} \{ J(3\alpha + 0) + J(3\alpha - 0) \} + \dots \right], \quad (2.11.1)$ where $\alpha \beta = \frac{1}{2} \pi$.

In this case the right-hand side is necessarily convergent. Proceeding as before, we obtain

$$\begin{split} \sqrt{\beta} [F_s(\beta) - F_s(3\beta) + ... + (-1)^n F_s \{ (2n+1)\beta \}] \\ &= \frac{(-1)^n \sqrt{\beta}}{\sqrt{(2\pi)}} \int_0^\infty f(t) \frac{\sin(2n+2)\beta t}{\cos \beta t} dt \\ &= \frac{(-1)^n \sqrt{\beta}}{\sqrt{(2\pi)}} \sum_{m=1}^\infty \int_{(m-1)\pi/\beta}^{m\pi/\beta} f(t) \frac{\sin(2n+2)\beta t}{\cos \beta t} dt \\ &= \frac{1}{\sqrt{(2\pi\beta)}} \sum_{m=1}^\infty (-1)^{m-1} \int_{-\frac{1}{2}\pi}^{\frac{1}{2}\pi} f\left\{ \frac{(m-\frac{1}{2})\pi + v}{\beta} \right\} \frac{\sin(2n+2)v}{\sin v} dv. \end{split}$$

This differs from the right-hand side of (2.11.1) by

$$\frac{\sqrt{\alpha}}{\pi} \sum_{m=1}^{\infty} (-1)^{m-1} \int_{0}^{\frac{1}{2}\pi} \left[f\left(\frac{(m-\frac{1}{2})\pi+v}{\beta}\right) - f\left(\frac{(m-\frac{1}{2})\pi}{\beta}+0\right) \right] \frac{\sin(2n+2)v}{\sin v} \, dv + \\ + (-1)^{m-1} \int_{-\frac{1}{2}\pi}^{0} \left[f\left(\frac{(m-\frac{1}{2})\pi+v}{\beta}\right) - f\left(\frac{(m-\frac{1}{2})\pi}{\beta}-0\right) \right] \frac{\sin(2n+2)v}{\sin v} \, dv,$$

and the result follows as before.

Example. Let
$$f(x)=x^{-s}$$
 $(0<\sigma<1)$. Then
$$F_s(x)=\sqrt{\binom{2}{\pi}}\Gamma(1-s)\cos{\frac{1}{2}s\pi}\,x^{s-1},$$

and

$$\begin{split} \sqrt{\left(\frac{2\beta}{\pi}\right)}\Gamma(1-s)\cos\tfrac{1}{2}s\pi\{\beta^{s-1}-(3\beta)^{s-1}+\ldots\} &= \sqrt{\alpha}\{\alpha^{-s}-(3\alpha)^{-s}+\ldots\},\\ \text{or } \left(\frac{\pi}{2}\right)^s\Gamma(1-s)\cos\tfrac{1}{2}s\pi\,L(1-s) &= L(s), \qquad L(s) &= 1 - \frac{1}{3^s} + \frac{1}{5^s} - \ldots. \end{split}$$

2.12. More general conditions. The next theorem is a more general one, in which f(x) is not necessarily of bounded variation.

THEOREM 47. Let f(x) be integrable over any finite interval, let $\alpha\beta = 2\pi$, and

 $\chi_N(y) = \sum_{n=1}^{N} f(y + n\alpha) \to \chi(y),$ (2.12.1)

and $|\chi_N(y)| \leqslant \phi(y)$, where $\phi(y)$ is $L(-\frac{1}{2}\alpha, \frac{1}{2}\alpha)$. Then

$$F_c(n\beta) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^{\infty} f(x) \cos n\beta x \, dx \qquad (2.12.2)$$

exists for every n, and

$$\lim_{\delta \to 0} \sqrt{\beta} \left\{ \frac{1}{2} F_c(0) + \sum_{n=1}^{\infty} F_c(n\beta) e^{-\delta n\beta} \right\} = \frac{1}{2} \sqrt{\alpha} \{ f(+0) + \chi(+0) + \chi(-0) \}$$
(2.12.3)

provided that the right-hand side exists.+

We have

$$\int_{\frac{1}{2}\alpha}^{(N+\frac{1}{2})\alpha} f(x)\cos n\beta x \, dx = \sum_{m=1}^{N} \int_{(m-\frac{1}{2})\alpha}^{(m+\frac{1}{2})\alpha} f(x)\cos n\beta x \, dx$$

$$= \sum_{m=1}^{N} \int_{-\frac{1}{2}\alpha}^{\frac{1}{2}\alpha} f(y+m\alpha)\cos n\beta y \, dy = \int_{-\frac{1}{2}\alpha}^{\frac{1}{2}\alpha} \chi_{N}(y)\cos n\beta y \, dy.$$

$$\cot \left| \int_{-\frac{1}{2}\alpha}^{(N+\frac{1}{2})\alpha} f(x)\cos n\beta x \, dx \right| \leq \int_{-\frac{1}{2}\alpha}^{\frac{1}{2}\alpha} \phi(y) \, dy.$$

Hence

Also, if $(N+\frac{1}{2})\alpha \leqslant X < (N+\frac{3}{2})\alpha$,

$$\begin{vmatrix} \int\limits_{(N+\frac{1}{4})\alpha}^X f(x) \cos n\beta x \ dx \end{vmatrix} \leqslant \int\limits_{(N+\frac{1}{4})\alpha}^{(N+\frac{1}{4})\alpha} |f(x)| \ dx$$

$$= \int\limits_{-\frac{1}{4}\alpha}^{\frac{1}{4}\alpha} |\chi_{N+1}(x) - \chi_{N}(x)| \ dx \leqslant 2 \int\limits_{-\frac{1}{4}\alpha}^{\frac{1}{4}\alpha} \phi(y) \ dy.$$
where
$$\int\limits_{-\frac{1}{4}\alpha}^X f(x) \cos n\beta x \ dx$$

Hence

is bounded for all n and X. Also

$$\int_{(N+\frac{1}{2})\alpha}^{(N'+\frac{1}{2})\alpha} f(x)\cos n\beta x \ dx = \int_{-\frac{1}{2}\alpha}^{\frac{1}{2}\alpha} \{\chi_{N'}(y) - \chi_N(y)\}\cos n\beta y \ dy$$

tends to 0 as $N \to \infty$, $N' \to \infty$, by Lebesgue's convergence theorem; and similarly

$$\int_{T}^{(N+\frac{1}{2})\alpha} f(x) \cos n\beta x \, dx \to 0, \qquad \int_{(N'+\frac{1}{2})\alpha}^{T'} f(x) \cos n\beta x \, dx \to 0$$

† Borgen (1).

if $(N-\frac{1}{2})\alpha < T \leqslant (N+\frac{1}{2})\alpha$, $(N'+\frac{1}{2})\alpha \leqslant T' < (N'+\frac{3}{2})\alpha$, $T \to \infty$, $T' \to \infty$. Hence (2.12.2) converges boundedly for every n.

In (2.8.1) with α and β interchanged, take $f(x) = e^{-\delta x} \cos xy$. Then

$$F_c(t) = rac{1}{\sqrt{(2\pi)}} \left\{ rac{\delta}{\delta^2 + (y-t)^2} + rac{\delta}{\delta^2 + (y+t)^2}
ight\},$$

and we obtain

$$egin{aligned} \sqrt{eta} \left\{ rac{1}{2} + \sum_{n=1}^{\infty} e^{-\delta n eta} \cos n eta y
ight\} &= \sqrt{\left(rac{lpha}{2\pi}\right)} \sum_{n=-\infty}^{\infty} \delta^2 + rac{\delta}{(y-nlpha)^2} \ &= \sqrt{\left(rac{1}{2}lpha\pi\right) K(y,\delta)}, \end{aligned}$$

say. Hence

$$\sqrt{\beta} \left\{ \frac{1}{\sqrt{(2\pi)}} \int_{0}^{Y} f(y) \, dy + \sqrt{\binom{2}{\pi}} \sum_{n=1}^{\infty} e^{-\delta n\beta} \int_{0}^{Y} \cos n\beta y f(y) \, dy \right\} \\
= \sqrt{\alpha} \int_{0}^{Y} f(y) K(y, \delta) \, dy.$$

By the bounded convergence of (2.12.2), the left-hand side tends to that of (2.12.3) as $Y \to \infty$. Also since $K(y, \delta)$ is periodic, with period α ,

$$\int_{0}^{\infty} f(y)K(y,\delta) dy = \left\{ \int_{0}^{\frac{1}{1}\alpha} + \sum_{m=1}^{\infty} \int_{(m-\frac{1}{2})\alpha}^{(m+\frac{1}{2})\alpha} f(y)K(y,\delta) dy \right\}$$

$$= \int_{0}^{\frac{1}{1}\alpha} f(y)K(y,\delta) dy + \sum_{m=1}^{\infty} \int_{-\frac{1}{2}\alpha}^{\frac{1}{2}\alpha} f(y+m\alpha)K(y,\delta) dy$$

$$= \int_{0}^{\frac{1}{2}\alpha} f(y)K(y,\delta) dy + \int_{-\frac{1}{2}\alpha}^{\frac{1}{2}\alpha} \chi(y)K(y,\delta) dy,$$

by the dominated convergence of the series, $K(y, \delta)$ being bounded for a fixed δ . The result now follows from Theorem 17, with x = 0, $a = -\frac{1}{2}\alpha$, $b = \frac{1}{2}\alpha$, and $K(0, y, \delta) = K(y, \delta)$.

TRANSFORMS OF THE CLASS L2

3.1. Plancherel's theory of Fourier transforms. The formulae (1.2.1), (1.2.2), connecting a pair of Fourier cosine transforms f(x), $F_c(x)$, express a relation between these functions which is formally symmetrical. But in all the theorems which we have proved so far, the two functions satisfy quite different conditions, so that the symmetry is only formal.

A theory of the reciprocity which is completely symmetrical was first given by Plancherel.† It depends, not on ordinary convergence or summability, but on mean convergence.

For complex transforms Plancherel's theorem is

THEOREM 48. Let f(x) be a (real or complex) function of the class $L^2(-\infty,\infty)$, and let

$$F(x,a) = \frac{1}{\sqrt{(2\pi)}} \int_{-\pi}^{a} f(y)e^{ixy} dy.$$
 (3.1.1)

Then, as $a \to \infty$, F(x, a) converges in mean over $(-\infty, \infty)$ to a function F(x) of $L^2(-\infty, \infty)$; and reciprocally

$$f(x,a) = \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} F(y)e^{-ixy} dy$$
 (3.1.2)

converges in mean to f(x).

The transforms f(x), F(x) are connected by the formulae

$$F(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} f(y) \frac{e^{ixy} - 1}{iy} dy,$$
 (3.1.3)

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} F(y) \frac{e^{-ixy} - 1}{-iy} dy,$$
 (3.1.4)

for almost all values of x.

It will be seen from the proof that we might replace F(x,a) by

$$F(x,a,b) = \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{b} f(y)e^{ixy} dy,$$

where $a \to \infty$, $b \to \infty$, in any manner.

† Plancherel (1), (2), (3), (4).

At the same time we obtain

THEOREM 49. If f(x), F(x) and g(x), G(x) are Fourier transforms as in the above theorem, (2.1.1.), (2.1.2), and (2.1.3) hold.

For cosine and sine transforms the theory is as follows.

THEOREM 50. Let f(x) belong to $L^2(0,\infty)$, and let

$$F_c(x,a) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^a f(y) \cos xy \ dy.$$

Then, as $a \to \infty$, $F_c(x, a)$ converges in mean over $(0, \infty)$ to a function $F_c(x)$ of $L^2(0, \infty)$; and reciprocally

$$f_c(x,a) = \sqrt{\left(rac{2}{\pi}
ight)} \int\limits_0^a F_c(y) \cos xy \ dy$$

converges in mean to f(x). We have almost everywhere

$$F_c(x) = \sqrt{\left(\frac{2}{\pi}\right)} \frac{d}{dx} \int_0^\infty f(y) \frac{\sin xy}{y} dy, \quad f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \frac{d}{dx} \int_0^\infty F_c(y) \frac{\sin xy}{y} dy.$$

THEOREM 51. The analogue of Theorem 50 for sine transforms holds, with $\cos xy$ replaced by $\sin xy$, and $\sin xy$ by $1-\cos xy$.

THEOREM 52. (2.1.4), (2.1.5), (2.1.6), and (2.1.7) hold for transforms of L^2 .

The cosine and sine theorems may be obtained by taking f(x) even or odd in the 'complex' theorem.

We shall give several different proofs of these theorems.

3.2. Fourier transforms, first method.† This is suggested by Fourier's formal process (§ 1.1). Let

$$a_{\nu} = \int_{
u/\lambda}^{(\nu+1)/\lambda} f(x) dx \quad (\nu = 0, \pm 1, ...),$$

$$\Phi_{n}(x) = \sum_{\nu=1}^{n} a_{\nu} e^{i\nu x/\lambda}.$$

and

Then if b > 0 and $n = \lceil \lambda b \rceil - 1$,

$$\lim_{\lambda \to \infty} \Phi_n(x) = \int_{-b}^b f(y) e^{ixy} \, dy$$

† Titchmarsh (1), (2).

uniformly in any finite interval. For

$$egin{aligned} \left|\Phi_n(x)-\int\limits_{-b}^bf(y)e^{ixy}\,dy
ight| &=\left|\sum\limits_{
u=-n}^n\int\limits_{
u/\lambda}^{(
u+1)/\lambda}f(y)(e^{i
u x/\lambda}-e^{ixy})\,dy-
ight. \ &\left.-\int\limits_{(n+1)/\lambda}^bf(y)e^{ixy}\,dy-\int\limits_{-b}^{-n/\lambda}f(y)e^{ixy}\,dy
ight| \ &\leqslantrac{x}{\lambda}\int\limits_{-b}^b|f(y)|\,dy+\int\limits_{(n+1)/\lambda}^b|f(y)|\,dy+\int\limits_{-b}^{-n/\lambda}|f(y)|\,dy o 0, \end{aligned}$$

since $|e^{i\nu x/\lambda}-e^{ixy}| \leqslant x/\lambda$ in each integral.

Also
$$|a_{\nu}|^2\leqslant \int\limits_{\nu/\lambda}^{(\nu+1)/\lambda}|f(x)|^2\,dx\int\limits_{\nu/\lambda}^{(\nu+1)/\lambda}dx=\frac{1}{\lambda}\int\limits_{\nu/\lambda}^{(\nu+1)/\lambda}|f(x)|^2\,dx.$$

Hence, if $X \leqslant \pi \lambda$,

$$\begin{split} \int_{-X}^{X} |\Phi_n(x)|^2 \, dx & \leqslant \int_{-\pi\lambda}^{\pi\lambda} |\Phi_n(x)|^2 \, dx = \int_{-\pi\lambda}^{\pi\lambda} \left(\sum_{\nu=-n}^n a_{\nu} e^{i\nu x/\lambda} \sum_{\mu=-n}^n \bar{a}_{\mu} e^{-i\mu x/\lambda} \right) dx \\ & = 2\pi\lambda \sum_{\nu=-n}^n |a_{\nu}|^2 \leqslant 2\pi \int_{-\pi/\lambda}^{(n+1)/\lambda} |f(x)|^2 \, dx \leqslant 2\pi \int_{-h}^{h} |f(x)|^2 \, dx. \end{split}$$

Keeping X fixed and making $\lambda \to \infty$, it follows that

$$\int\limits_{-X}^{X} igg| \int\limits_{-b}^{b} f(y) e^{ixy} \ dy igg|^2 dx \leqslant 2\pi \int\limits_{-b}^{b} |f(x)|^2 \ dx.$$

Making $X \to \infty$,

$$\int_{-\infty}^{\infty} \left| \int_{-b}^{b} f(y)e^{ixy} dy \right|^{2} dx \leqslant 2\pi \int_{-b}^{b} |f(x)|^{2} dx. \tag{3.2.1}$$

If we take f(y) = 0 for -a < y < a, this gives

$$\int_{-\infty}^{\infty} |F(x,b) - F(x,a)|^2 dx \leqslant \left(\int_{-b}^{-a} + \int_{a}^{b}\right) |f(x)|^2 dx,$$

which tends to 0 as $a \to \infty$, $b \to \infty$. Hence F(x,a) converges in mean, to a function F(x), say, of $L^2(-\infty,\infty)$; and, making $b \to \infty$ in (3.2.1),

$$\int_{-\infty}^{\infty} |F(x)|^2 dx \leqslant \int_{-\infty}^{\infty} |f(x)|^2 dx. \tag{3.2.2}$$

A similar argument now shows that f(x,a) converges in mean, to

 $\phi(x)$ say. We have to prove that $\phi(x) = f(x)$ almost everywhere, and for this it is sufficient to show that

$$\int_{0}^{\xi} \phi(x) \ dx = \int_{0}^{\xi} f(x) \ dx$$

for all values of ξ . Now

$$\int_{0}^{\xi} \phi(x) \, dx = \lim_{a \to \infty} \int_{0}^{\xi} f(x, a) \, dx = \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\xi} dx \int_{-a}^{a} F(y) e^{-ixy} \, dy$$

$$= \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} F(y) \frac{e^{-i\xi y} - 1}{-iy} \, dy = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(y) \frac{e^{-i\xi y} - 1}{-iy} \, dy.$$

On the other hand, Theorem 22 (1.24.2), with f(x) = 0 for |x| > a, gives

 $\int_{0}^{\xi} f(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \frac{e^{-i\xi u} - 1}{u} F(u, a) du \quad (|\xi| < a).$

But $\lim_{a\to\infty}\int_{-\infty}^{\infty}\frac{e^{-i\xi u}-1}{u}F(u,a)\,du=\int_{-\infty}^{\infty}\frac{e^{-i\xi u}-1}{u}F(u)\,du,$

since $(e^{-i\xi u}-1)/u$ belongs to $L^2(-\infty,\infty)$. The result stated therefore follows.

Incidentally we have proved (3.1.4); since we may now argue similarly with (3.1.2) instead of (3.1.1), (3.1.3) also follows.

Also we may interchange f and F in (3.2.2). Hence in fact

$$\int_{-\infty}^{\infty} |F(x)|^2 dx = \int_{-\infty}^{\infty} |f(x)|^2 dx.$$

If G(x) is the transform of g(x) in the same sense, F+G is the transform of f+g; hence

$$\int_{-\infty}^{\infty} |F(x) + G(x)|^2 dx = \int_{-\infty}^{\infty} |f(x) + g(x)|^2 dx,$$

i.e.

$$\int_{-\infty}^{\infty} \{|F(x)|^2 + |G(x)|^2 + 2\mathbf{R}F(x)\overline{G}(x)\} dx$$

$$= \int_{-\infty}^{\infty} \{|f(x)|^2 + |g(x)|^2 + 2\mathbf{R}f(x)\overline{g}(x)\} dx.$$

Hence

$$\mathbf{R} \int_{-\infty}^{\infty} F(x) \overline{G}(x) \ dx = \mathbf{R} \int_{-\infty}^{\infty} f(x) \overline{g}(x) \ dx.$$

Arguing in the same way with f+ig, we see that the imaginary parts are also equal. Hence

$$\int_{-\infty}^{\infty} F(x)\overline{G}(x) \ dx = \int_{-\infty}^{\infty} f(x)\overline{g}(x) \ dx.$$

3.3. Fourier transforms, second method.† Let f(x) belong to $L^2(-\infty,\infty)$. Then we can construct a sequence of functions $f_n(x)$, each of which is continuous and of bounded variation over a finite interval, and zero outside this interval, and such that

$$\int_{-\infty}^{\infty} |f(x)-f_n(x)|^2 dx \to 0.$$

$$F_n(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f_n(u)e^{ixu} du.$$

Let

Then

$$\begin{split} \int_{-\lambda}^{\lambda} |F_n(x)|^2 \, dx &= \frac{1}{2\pi} \int_{-\lambda}^{\lambda} dx \int_{-\infty}^{\infty} f_n(u) e^{ixu} \, du \int_{-\infty}^{\infty} \bar{f}_n(v) e^{-ixv} \, dv \\ &= \frac{1}{2\pi} \int_{-\infty}^{\infty} f_n(u) \, du \int_{-\infty}^{\infty} \bar{f}_n(v) \, dv \int_{-\lambda}^{\lambda} e^{ix(u-v)} \, dx \\ &= \frac{1}{2\pi} \int_{-\infty}^{\infty} f_n(u) \, du \int_{-\infty}^{\infty} \bar{f}_n(v) \frac{2 \sin \lambda (u-v)}{u-v} \, dv. \end{split}$$

By the theory of § 1.9, the inner integral tends to $2\pi \bar{f}_n(u)$ uniformly over any finite range, as $\lambda \to \infty$; and hence

$$\int_{-\infty}^{\infty} |F_n(x)|^2 dx = \int_{-\infty}^{\infty} |f_n(u)|^2 du.$$

Similarly

$$\int_{-\infty}^{\infty} |F_m(x) - F_n(x)|^2 dx = \int_{-\infty}^{\infty} |f_m(u) - f_n(u)|^2 du,$$

† Bochner, Vorlesungen über Fouriersche Integrale, § 41.

and the right-hand side tends to 0 as m and n tend to infinity. Hence $F_n(x)$ converges in mean, to F(x) say; and

$$\int_{-\infty}^{\infty} |F(x)|^2 dx = \lim_{n \to \infty} \int_{-\infty}^{\infty} |F_n(x)|^2 dx$$

$$= \lim_{n \to \infty} \int_{-\infty}^{\infty} |f_n(x)|^2 dx = \int_{-\infty}^{\infty} |f(x)|^2 dx.$$

This function F(x) is the Fourier transform of f(x). It is of course not yet obvious that it is equivalent to the transform obtained before, or even that it is unique, since the sequence $f_n(x)$ is not unique. However, we have

$$\int_{0}^{\xi} F_{n}(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\xi} dx \int_{-\infty}^{\infty} f_{n}(u) e^{ixu} du = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f_{n}(u) \frac{e^{i\xi u} - 1}{iu} du$$

(the range of integration being really finite). Making $n \to \infty$,

$$\int_{0}^{\xi} F(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(u) \frac{e^{i\xi u} - 1}{iu} du,$$

since $(e^{i\xi u}-1)/(iu)$ belongs to L^2 . Hence

$$F(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} f(u) \frac{e^{ixu} - 1}{iu} du$$

almost everywhere. Hence F(x) is unique (apart from sets of measure zero), and is equivalent to the transform obtained by the first method.

In the first method we deduced the Parseval formula from the reciprocity; in this method we have proved the Parseval formula already, and we deduce the reciprocity from it. As before, the Parseval formula gives

$$\int_{-\infty}^{\infty} F(x)\overline{G}(x) dx = \int_{-\infty}^{\infty} f(x)\overline{g}(x) dx.$$
Let $g(x) = 1$ $(0 \leqslant x \leqslant \xi)$, $g(x) = 0$ $(x < 0 \text{ or } x > \xi)$. Then
$$G(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{0}^{\xi} \frac{e^{ixu} - 1}{iu} du$$

$$= \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{0}^{\xi x} \frac{e^{iu} - 1}{iu} du = \frac{1}{\sqrt{(2\pi)}} \frac{e^{i\xi x} - 1}{ix}.$$

Hence

$$\frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty}F(x)\frac{e^{-i\xi x}-1}{-ix}\,dx=\int_{0}^{\xi}f(x)\,dx,$$

so that f(x) is the transform of F(x).

Again, let h(x) = f(x) $(-a \leqslant x \leqslant a)$, 0 (|x| > a). Then

$$H(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-a}^{a} f(u) \frac{e^{ixu} - 1}{iu} du = \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(u) e^{ixu} du = F(x, a),$$

with the usual notation. Hence the transform of F(x)-F(x,a) is f(x)-h(x), i.e. it is 0 ($|x| \le a$), f(x) (|x| > a). Hence

$$\int_{-\infty}^{\infty} |F(x)-F(x,a)|^2 dx = \left(\int_{-\infty}^{-a} + \int_{a}^{\infty}\right) |f(x)|^2 dx,$$

which tends to 0 as $a \to \infty$. Hence

$$F(x) = \lim_{a \to \infty} F(x, a).$$

3.4. Fourier transforms, third method.† Suppose first that f(x) belongs to both L and L^2 , and let

$$F(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(t)e^{ixt} dt.$$

Then

$$\int_{-\infty}^{\infty} e^{-\frac{1}{2}\delta^{2}x^{2}} |F(x)|^{2} dx = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{-\frac{1}{2}\delta^{2}x^{2}} dx \int_{-\infty}^{\infty} f(u)e^{ixu} du \int_{-\infty}^{\infty} \bar{f}(v)e^{-ixv} dv$$

$$= \frac{1}{2\pi} \int_{-\infty}^{\infty} f(u) du \int_{-\infty}^{\infty} \bar{f}(v) dv \int_{-\infty}^{\infty} e^{-\frac{1}{2}\delta^{2}x^{2} + ix(u-v)} dx$$

$$= \frac{1}{\delta\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(u) du \int_{-\infty}^{\infty} \bar{f}(v)e^{-\frac{1}{2}(u-v)^{2}/\delta^{2}} dv, \quad (3.4.1)$$

and by Schwarz's inequality for double integrals

$$\left| \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} f(u) \overline{f}(v) e^{-\frac{1}{2}(u-v)^{3}/\delta^{2}} du dv \right|$$

$$= \left| \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} f(u) e^{-\frac{1}{2}(u-v)^{3}/\delta^{2}} \overline{f}(v) e^{-\frac{1}{2}(u-v)^{3}/\delta^{2}} du dv \right|$$
† F. Riesz (2).

$$\leqslant \left\{ \int\limits_{-\infty}^{\infty} \int\limits_{-\infty}^{\infty} |f(u)|^2 e^{-\frac{1}{4}(u-v)^2/\delta^2} \, du dv \int\limits_{-\infty}^{\infty} \int\limits_{-\infty}^{\infty} |\bar{f}(v)|^2 e^{-\frac{1}{4}(u-v)^2/\delta^2} \, du dv \right\}^{\frac{1}{2}}$$

$$= \int\limits_{-\infty}^{\infty} \int\limits_{-\infty}^{\infty} |f(u)|^2 e^{-\frac{1}{4}(u-v)^2/\delta^2} \, du dv$$

$$= \int\limits_{-\infty}^{\infty} |f(u)|^2 \, du \int\limits_{-\infty}^{\infty} e^{-\frac{1}{4}t^2/\delta^2} \, dt = \delta \sqrt{(2\pi)} \int\limits_{-\infty}^{\infty} |f(u)|^2 \, du.$$
ace
$$\int\limits_{-\infty}^{\infty} e^{-\frac{1}{4}\delta^2 x^2} |F(x)|^2 \, dx \leqslant \int\limits_{-\infty}^{\infty} |f(u)|^2 \, du,$$

Hence

and, making $\delta \to 0$, it follows that F(x) belongs to $L^2(-\infty,\infty)$. Also (3.4.1) is equal to

$$\begin{split} \frac{1}{\delta\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} f(u) \ du \int\limits_{-\infty}^{\infty} \bar{f}(u+t) e^{-\frac{i}{\hbar} t^{i}/\delta^{i}} \ dt \\ &= \frac{1}{\delta\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{-\frac{i}{\hbar} t^{i}/\delta^{i}} \ dt \int\limits_{-\infty}^{\infty} f(u) \bar{f}(u+t) \ du = \frac{1}{\delta\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{-\frac{i}{\hbar} t^{i}/\delta^{i}} \psi(t) \ dt \end{split}$$

say. Since f belongs to L^2 , ψ is bounded and continuous. Hence

$$\int_{-\infty}^{\infty} |F(x)|^2 dx = \lim_{\delta \to 0} \int_{-\infty}^{\infty} e^{-\frac{1}{2}\delta^2 x^2} |F(x)|^2 dx = \lim_{\delta \to 0} \frac{1}{\delta \sqrt{(2\pi)}} \int_{-\infty}^{\infty} e^{-\frac{1}{2}t^2/\delta^2} \psi(t) dt$$
$$= \psi(0) = \int_{-\infty}^{\infty} |f(u)|^2 du,$$

by the theory of Weierstrass's singular integral (§ 1.18).

The existence of F for any f of L^2 , and the reciprocity, may now be proved as in the previous method.

3.5. The Hermite polynomials.[†] The Hermite polynomial of degree n is defined by

$$H_n(x) = (-1)^n e^{x^i} \left(\frac{d}{dx}\right)^n e^{-x^i},$$
 (3.5.1)

and we write

$$\phi_n(x) = e^{-\frac{i}{2}x^2} H_n(x) = (-1)^n e^{\frac{i}{2}x^2} \left(\frac{d}{dx}\right)^n e^{-x^2}.$$
 (3.5.2)

The interest of these functions for our theory is that they form an orthogonal sequence, each member of which is, apart from a trivial factor, its own Fourier transform.

† See Wiener, The Fourier Integral, 51-71.

We have

$$\phi_n''(x) = (-1)^n \left\{ (x^2 + 1) \left(\frac{d}{dx} \right)^n e^{-x^2} + 2x \left(\frac{d}{dx} \right)^{n+1} e^{-x^2} + \left(\frac{d}{dx} \right)^{n+2} e^{-x^4} \right\} e^{\frac{1}{2}x^2},$$
and

$$\left(\frac{d}{dx}\right)^{n+2}e^{-x^{2}} = \left(\frac{d}{dx}\right)^{n+1}(-2xe^{-x^{2}}) = -2x\left(\frac{d}{dx}\right)^{n+1}e^{-x^{2}} - 2(n+1)\left(\frac{d}{dx}\right)^{n}e^{-x^{2}}.$$

Hence

$$\phi_n''(x) = (-1)^n \left\{ (x^2 + 1) \left(\frac{d}{dx} \right)^n e^{-x^2} - 2(n+1) \left(\frac{d}{dx} \right)^n e^{-x^2} \right\} e^{\frac{i}{2}x^2}$$

$$= (x^2 - 2n - 1) \dot{\phi}_n(x). \tag{3.5.3}$$

Thus $y = \phi_n(x)$ is a solution of the differential equation

$$\frac{d^2y}{dx^2} - x^2y = -(2n+1)y. (3.5.4)$$

Putting $y = e^{-\frac{1}{2}x^2}u$, we obtain

$$\frac{d^2u}{dx^2} - 2x\frac{du}{dx} = -2nu, (3.5.5)$$

so that $H_n(x)$ is a solution of this equation.

Further, it is the only polynomial solution. For let

$$u = a_0 + a_1 x + ...$$

be a solution. Then

$$\sum a_r r(r-1) x^{r-2} - 2 \sum a_r r x^r = -2n \sum a_r x^r.$$

$$(r+1)(r+2) a_{r+2} = 2(r-n) a_r,$$

Hence

and the general solution is the sum of two series, of which the one in which r has the same parity as n terminates, and the other does not.

The Hermite functions $\phi_n(x)$ form an orthogonal set. For by (3.5.3)

 $\phi_m''(x)\phi_n(x)-\phi_n''(x)\phi_m(x) = 2(n-m)\phi_m(x)\phi_n(x).$

Hence

$$\int_{-\infty}^{\infty} \phi_m(x)\phi_n(x) dx = \frac{1}{2(n-m)} [\phi'_m(x)\phi_n(x) - \phi'_n(x)\phi_m(x)]_{-\infty}^{\infty}$$

$$= 0$$

if $m \neq n$.

3.6. Theorem 53.† If |t| < 1,

$$\sum_{n=0}^{\infty} \frac{e^{-\mathbf{i}(x^{2}+y^{2})}}{2^{n}n!} t^{n} H_{n}(x) H_{n}(y) = \frac{1}{\sqrt{(1-t^{2})}} \exp\left\{\frac{x^{2}-y^{2}}{2} - \frac{(x-yt)^{2}}{1-t^{2}}\right\}. \quad (3.6.1)$$

We have

$$e^{-x^2} = \frac{1}{\sqrt{\pi}} \int_{-\infty}^{\infty} e^{-u^2+2ixu} du,$$

† See Watson (3).

and hence

$$H_n(x) = \frac{(-2i)^n e^{x^n}}{\sqrt{\pi}} \int_{-\infty}^{\infty} u^n e^{-u^n + 2ixu} du.$$

Hence

$$\begin{split} \sum_{n=0}^{\infty} \frac{e^{-\frac{\mathbf{i}(x^{3}+y^{3})}{2^{n}n!}} t^{n} H_{n}(x) H_{n}(y) \\ &= \frac{e^{\frac{\mathbf{i}(x^{3}+y^{3})}{\pi}} \sum_{n=0}^{\infty} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \frac{(-2tuv)^{n}}{n!} e^{-u^{3}-v^{4}+2ixu+2iyv} \, du dv \\ &= \frac{e^{\frac{\mathbf{i}(x^{3}+y^{3})}{\pi}} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} e^{-u^{3}-v^{4}+2ixu+2iyv-2tuv} \, du dv \\ &= \frac{e^{\frac{\mathbf{i}(x^{3}-y^{3})}{\sqrt{\pi}}} \int_{-\infty}^{\infty} e^{-(1-t^{3})u^{4}+2i(x-yt)u} \, du \\ &= \frac{1}{\sqrt{(1-t^{2})}} \exp\left\{\frac{1}{2}(x^{2}-y^{2}) - \frac{(x-yt)^{2}}{1-t^{2}}\right\}. \end{split}$$

The inversion is justified by the convergence of

$$\int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \sum_{n=0}^{\infty} \frac{(2\rho uv)^n}{n!} e^{-u^1-v^1+2Au+2Bv} dudv$$

if $\rho < 1$.

THEOREM 54. The functions

$$\psi_n(x) = \frac{\phi_n(x)}{(2^n n! \sqrt{\pi})^{\frac{1}{2}}}$$

form a normal orthogonal set over $(-\infty,\infty)$, i.e.

$$\int_{-\infty}^{\infty} \psi_m(x)\psi_n(x) \ dx = \begin{cases} 1 & (m=n) \\ 0 & (m \neq n). \end{cases}$$

For $m \neq n$ the result follows from § 3.5. Also, putting x = y in (3.6.1),

$$\sum_{n=0}^{\infty} \frac{t^n e^{-x^1} \{H_n(x)\}^2}{2^n n! \sqrt{\pi}} = \frac{1}{\sqrt{\{\pi(1-t^2)\}}} \exp\left(-x^2 \frac{1-t}{1+t}\right).$$

Hence

$$\begin{split} \sum_{n=0}^{\infty} \frac{t^n}{2^n n! \sqrt{\pi}} \int_{-\infty}^{\infty} e^{-x^t} \{H_n(x)\}^2 \, dx &= \frac{1}{\sqrt{\{\pi(1-t^2)\}}} \int_{-\infty}^{\infty} \exp\left(-x^2 \frac{1-t}{1+t}\right) \, dx \\ &= \frac{1}{\sqrt{\{\pi(1-t^2)\}}} \sqrt{\left(\pi \frac{1+t}{1-t}\right)} = \frac{1}{1-t} = \sum_{n=0}^{\infty} t^n. \end{split}$$

Equating coefficients of t^n ,

$$\int_{-\infty}^{\infty} {\{\phi_n(x)\}^2 dx} = \int_{-\infty}^{\infty} e^{-x^2} {\{H_n(x)\}^2 dx} = 2^n n! \sqrt{\pi},$$

and the result follows.

3.7. THEOREM 55. If f(x) is any function of $L^2(-\infty,\infty)$, and

$$a_m = \int_{-\infty}^{\infty} f(x)\psi_m(x) dx \quad (m = 0, 1, 2, ...),$$
 (3.7.1)

then

$$\lim_{n \to \infty} \int_{-\infty}^{\infty} \left| f(x) - \sum_{m=0}^{n} a_m \psi_m(x) \right|^2 dx = 0.$$
 (3.7.2)

We can write (3.6.1) as

$$\sum_{n=0}^{\infty} t^n \psi_n(x) \psi_n(y) = \frac{1}{\sqrt{\{\pi(1-t^2)\}}} \exp\left\{\frac{x^2 - y^2}{2} - \frac{(x-yt)^2}{1-t^2}\right\}. \quad (3.7.3)$$

Denote the right-hand side by K(x, y, t). Then $K(x, y, t) \ge 0$, and (putting $y = 2xt/(1+t^2)+u$)

$$\int_{-\infty}^{\infty} K(x, y, t) dy = \frac{e^{-\frac{1}{2} \frac{1-t^{2}}{1+t^{2}} x^{2}}}{\sqrt{\pi(1-t^{2})}} \int_{-\infty}^{\infty} e^{-\frac{1+t^{2}}{2(1-t^{2})} u^{2}} du$$

$$= \sqrt{\left(\frac{2}{1+t^{2}}\right)} e^{-\frac{1}{2} \frac{1-t^{2}}{1+t^{2}} x^{2}} \to 1$$

as $t \to 1$. The conditions (1.16.2), (1.16.3) are also plainly satisfied. Hence, if f(x) is any continuous function, vanishing outside a finite range, by Theorem 17

$$\lim_{t \to 1} \int_{-\infty}^{\infty} K(x, y, t) f(y) \, dy = f(x). \tag{3.7.4}$$

Hence, multiplying (3.7.3) by f(y) and integrating over $(-\infty,\infty)$,

$$\sum_{n=0}^{\infty} a_n t^n \psi_n(x) = \int_{-\infty}^{\infty} K(x, y, t) f(y) \ dy \to f(x)$$

as $t \to 1$, and the convergence is plainly bounded. Hence, multiplying by f(x) and integrating again,

$$\sum_{n=0}^{\infty} a_n^2 t^n \to \int_{-\infty}^{\infty} \{f(x)\}^2 dx,$$
$$\sum_{n=0}^{\infty} a_n^2 = \int_{-\infty}^{\infty} \{f(x)\}^2 dx.$$

so that

Now let f(x) be any function of $L^2(-\infty,\infty)$. Let $f_{\nu}(x)$ be a continuous function, vanishing outside a finite range, such that

$$\int_{-\infty}^{\infty} |f(x)-f_{\nu}(x)|^2 dx < \epsilon,$$

$$a_{n,\nu} = \int_{-\infty}^{\infty} f_{\nu}(x)\psi_n(x) dx.$$

and let

Then

$$\int_{-\infty}^{\infty} \{f(x) - a_{1,\nu} \psi_1(x) - \dots - a_{n,\nu} \psi_n(x)\}^2 dx$$

$$= \int_{-\infty}^{\infty} \{f(x)\}^2 dx + \sum_{1}^{n} a_{m,\nu}^2 - 2 \sum_{1}^{n} a_{m,\nu} a_m$$

$$\geqslant \int_{-\infty}^{\infty} \{f(x)\}^2 dx - \sum_{1}^{n} a_m^2 = \int_{-\infty}^{\infty} \{f(x) - a_1 \psi_1(x) - \dots - a_n \psi_n(x)\}^2 dx,$$

and also

$$\leqslant 2 \int_{-\infty}^{\infty} \{f(x) - f_{\nu}(x)\}^{2} dx + 2 \int_{-\infty}^{\infty} \{f_{\nu}(x) - a_{1,\nu} \psi_{1}(x) - \dots - a_{n,\nu} \psi_{n}(x)\}^{2} dx$$

$$< 2\epsilon + 2 \int_{-\infty}^{\infty} \{f_{\nu}(x)\}^{2} dx - 2a_{1,\nu}^{2} - \dots - 2a_{n,\nu}^{2} < 3\epsilon$$

for n sufficiently large. This proves (3.7.2).

THEOREM 56. If a_1 , a_2 ,... are given numbers such that $\sum a_n^2$ is convergent, there is a function f(x) of $L^2(-\infty,\infty)$ such that (3.7.1) holds.

This is the Riesz-Fischer theorem for the set $\psi_n(x)$. By Theorem 54

$$\int_{-\infty}^{\infty} \left| \sum_{m=n}^{N} a_m \psi_m(x) \right|^2 dx = \sum_{m=n}^{N} a_m^2,$$

which tends to 0 as n and N tend to infinity. Hence, as $n \to \infty$,

$$\sum_{m=0}^{n} a_m \psi_m(x)$$

converges in mean, to f(x) say; and

$$\int_{-\infty}^{\infty} f(x)\psi_r(x) dx = \lim_{n \to \infty} \int_{-\infty}^{\infty} \sum_{m=0}^{n} a_m \psi_m(x)\psi_r(x) dx$$
$$= a_r.$$

3.8. Theorem 57. The Fourier transform of $\phi_n(x)$ is $i^n\phi_n(x)$.

For

$$\int_{-\infty}^{\infty} \phi_n(x)e^{ixy} dx = (-1)^n \int_{-\infty}^{\infty} e^{ixy + \frac{1}{2}x^2} \left(\frac{d}{dx}\right)^n e^{-x^2} dx$$

$$= \int_{-\infty}^{\infty} e^{-x^2} \left(\frac{d}{dx}\right)^n e^{ixy + \frac{1}{2}x^2} dx$$

$$= e^{\frac{1}{2}y^2} \int_{-\infty}^{\infty} e^{-x^2} \left(\frac{d}{dx}\right)^n e^{\frac{1}{2}(x+iy)^2} dx$$

$$= (-i)^n e^{\frac{1}{2}y^2} \int_{-\infty}^{\infty} e^{-x^2} \left(\frac{d}{dy}\right)^n e^{\frac{1}{2}(x+iy)^2} dx$$

$$= (-i)^n e^{\frac{1}{2}y^2} \left(\frac{d}{dy}\right)^n \int_{-\infty}^{\infty} e^{-\frac{1}{2}x^2 + ixy - \frac{1}{2}y^2} dx$$

$$= (-i)^n e^{\frac{1}{2}y^2} \left(\frac{d}{dy}\right)^n \sqrt{(2\pi)} e^{-y^2}$$

$$= i^n \sqrt{(2\pi)} \phi_n(y).$$

Alternatively, let

$$\Phi_n(y) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} \phi_n(x) e^{ixy} dx.$$

Then

$$\Phi_n''(y) = -\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} x^2 \phi_n(x) e^{ixy} dx.$$

Also, integrating by parts twice,

$$\Phi_n(y) = -\frac{1}{y^2\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \phi_n''(x)e^{ixy} dx.$$

Hence

$$\begin{split} \Phi_n''(y) - y^2 & \Phi_n(y) + (2n+1)\Phi_n(y) \\ &= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \{\phi_n''(x) - x^2 \phi_n(x) + (2n+1)\phi_n(x)\} e^{ixy} \ dx \\ &= 0 \end{split}$$

Thus $\Phi_n(x)$ satisfies the same differential equation as $\phi_n(x)$; and it is easily seen that, if $e^{ix}\phi_n(x)$ is a polynomial, so is $e^{ix}\Phi_n(x)$.

Hence

$$\Phi_n(x) = c_n \phi_n(x).$$

Now

$$\begin{split} e^{-\frac{1}{2}x^2} &= \frac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{-\frac{1}{2}y^2 + ixy} \, dy, \\ \left(\frac{d}{dx}\right)^n e^{-\frac{1}{2}x^2} &= \frac{i^n}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{-\frac{1}{2}y^2} y^n e^{ixy} \, dy, \\ \left(\frac{d}{dx}\right)^n e^{-\frac{1}{2}x^2} &= \{(-1)^n x^n + ...\} e^{-\frac{1}{2}x^2}. \end{split}$$

and also

Hence
$$\frac{i^n}{\sqrt{2}}$$

$$\frac{i^n}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} e^{-\frac{1}{2}y^2} y^n e^{ixy} dy = \{(-1)^n x^n + \ldots\} e^{-\frac{1}{2}x^2},$$

and hence

$$rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} \phi_n(y) e^{ixy} \, dy = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} (2^n y^n + ...) e^{-\frac{1}{4}y^4} e^{ixy} \, dy$$

$$= i^n (2^n x^n + ...) e^{-\frac{1}{4}x^4}.$$

$$c_n = i^n.$$

Hence

3.9. Fourier transforms, fourth method. Take, for example, an even function f(x) of $L^2(0,\infty)$, and let

Then

$$a_n = \int\limits_{-\infty}^{\infty} f(x)\psi_n(x) \, dx.$$
 $a_1 = a_3 = \dots = 0,$
 $a_{2n} = 2\int\limits_{0}^{\infty} f(x)\psi_{2n}(x) \, dx,$
 $\sum_{n=0}^{\infty} a_{2n}^2 = 2\int\limits_{0}^{\infty} \{f(x)\}^2 \, dx.$

and

By Theorem 56 there is an even function g(x) such that

$$(-1)^n a_{2n} = 2 \int_0^\infty g(x) \psi_{2n}(x) \, dx \quad (n = 0, 1, ...);$$

$$\int_0^\infty (g(x))^2 \, dx \quad \sum_{n=0}^\infty x^2 \quad \int_0^\infty |f(x)|^2 \, dx$$

and

$$\int_{0}^{\infty} \{g(x)\}^{2} dx = \sum_{n=0}^{\infty} a_{2n}^{2} = \int_{0}^{\infty} \{f(x)\}^{2} dx.$$

The relation between f and g is plainly reciprocal.

We now identify g(x) with the Fourier cosine transform of f(x) previously obtained.

We have

$$\int_{0}^{\infty} \frac{\sin xy}{y} \psi_{2n}(y) \, dy = (-1)^{n} \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \frac{\sin xy}{y} \, dy \int_{0}^{\infty} \psi_{2n}(t) \cos yt \, dt$$

$$= (-1)^{n} \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \psi_{2n}(t) \, dt \int_{0}^{\infty} \frac{\sin xy \cos yt}{y} \, dy = (-1)^{n} \sqrt{\left(\frac{\pi}{2}\right)} \int_{0}^{x} \psi_{2n}(t) \, dt,$$

since $\psi_{2n}(t)$ belongs to $L(0,\infty)$ and the y-integral converges boundedly. Hence

$$\int_{0}^{\infty} g(y) \frac{\sin xy}{y} dy = \lim_{N \to \infty} \int_{0}^{\infty} \frac{\sin xy}{y} \sum_{n=0}^{N} (-1)^{n} a_{2n} \psi_{2n}(y) dy$$

$$= \lim_{N \to \infty} \sqrt{\left(\frac{\pi}{2}\right)} \sum_{n=0}^{N} a_{2n} \int_{0}^{x} \psi_{2n}(t) dt = \sqrt{\left(\frac{\pi}{2}\right)} \int_{0}^{x} f(t) dt,$$

so that f and g are Fourier transforms in the ordinary sense.

Similarly, by taking f(x) odd, we obtain the theory of sine transforms.

3.10. Convergence and summability. We can now prove theorems for L^2 functions corresponding to Theorems 3 and 14.

THEOREM 58. If f(t) belongs to $L^2(-\infty,\infty)$, and is of bounded variation in the neighbourhood of t=x, then

$$\frac{1}{2}\left\{f(x+0)+f(x-0)\right\}=\frac{1}{\sqrt{(2\pi)}}\lim_{\lambda\to\infty}\int_{-\lambda}^{\lambda}F(u)e^{-ixu}\,du.$$

The transform of $G(u) = e^{-ixu}$ ($|u| < \lambda$), 0 ($|u| > \lambda$), is

$$g(v) = rac{1}{\sqrt{(2\pi)}} \int_{-\lambda}^{\lambda} e^{-ixu-iuv} du = \sqrt{\left(rac{2}{\pi}
ight)} rac{\sin\lambda(x+v)}{x+v}.$$

Hence, by Parseval's formula,

$$\int_{-\lambda}^{\lambda} F(u)e^{-ixu} du = \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} f(v) \frac{\sin \lambda(x-v)}{x-v} dv.$$

The result now follows from the theory of Fourier's single integral (Theorem 12, case i (a)).

Theorem 59.† If f(t) belongs to $L^2(-\infty,\infty)$, then

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|u|}{\lambda} \right) F(u) e^{-ixu} du$$
$$\int_{-\lambda}^{h} |f(x+t) + f(x-t) - 2f(x)| dt = o(h)$$

wherever

as $h \to 0$, and so for almost all values of x; and Fourier's repeated integral for f(x) holds almost everywhere, if both integrals are taken in the (C, 1) sense.

The transform of
$$G(u) = \left(1 - \frac{|u|}{\lambda}\right)e^{-ixu}$$
 ($|u| < \lambda$), 0 ($|u| > \lambda$), is

$$g(v) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\lambda}^{\lambda} \left(1 - rac{|u|}{\lambda}
ight) e^{-ixu - iuv} \, du = \sqrt{\left(rac{2}{\pi}
ight)} rac{\sin^2 rac{1}{2}\lambda(x+v)}{\lambda(x+v)^2}.$$

Hence by Parseval's formula

$$\int_{-\lambda}^{\lambda} \left(1 - \frac{|u|}{\lambda}\right) F(u) e^{-ixu} du = \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} f(v) \frac{\sin^2 \frac{1}{2} \lambda (x - v)}{\lambda (x - v)^2} dv,$$

and, as in § 1.16, the result follows from Theorem 13.

The result also holds with f and F interchanged, and this gives the second part of the theorem.

We also deduce

THEOREM 60. If

$$f(x) = \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} \psi(t)e^{-ixt} dt,$$

where ψ belongs to L^2 , and also

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \chi(t)e^{-ixt} dt,$$

where χ belongs to L, then $\psi \equiv \chi$.

For by the above theorem

$$\lim_{\lambda \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-\lambda}^{\lambda} \left(1 - \frac{|t|}{\lambda}\right) f(t) e^{ixt} dt$$

is equal to $\psi(x)$ almost everywhere; and by Theorem 14 it is equal to $\chi(x)$ almost everywhere.

† Plancherel (3).

3.11. Convergence almost everywhere. If f(t) belongs to L^2 , the integral

 $\int\limits_{-\infty}^{\infty}f(t)e^{ixt}\,dt$

converges in mean; it is also summable (C, 1) almost everywhere, by Theorem 59, since in this theorem f and F are interchangeable.

It is not known whether the integral necessarily converges in the ordinary sense almost everywhere. As in Theorem 58 it would be easy to make it converge almost everywhere by imposing extra conditions on F(x). The object of the next sections is to state simple additional conditions on f(x) itself which make the integral converge almost everywhere.

THEOREM 61.† If f(t) belongs to $L^2(-\infty,\infty)$, then

$$\int_{-\lambda}^{\lambda} f(t)e^{ixt} dt = o(\log \lambda)$$
 (3.11.1)

wherever

$$\chi(h) = \int_{0}^{h} |F(x+y) + F(x-y) - 2F(x)| \, dy = o(h)$$

as $h \to 0$; and so almost everywhere.

As in the proof of Theorem 58

$$\int_{-\lambda}^{\lambda} f(t)e^{ixt} dt = \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} F(x+y) \frac{\sin \lambda y}{y} dy.$$

Now

$$\left|\int_{1}^{\infty} F(x+y) \frac{\sin \lambda y}{y} \, dy\right|^{2} \leqslant \int_{1}^{\infty} |F(x+y)|^{2} \, dy \int_{1}^{\infty} \frac{dy}{y^{2}} \leqslant \int_{-\infty}^{\infty} |F(t)|^{2} \, dt$$

and similarly for the integral over $(-\infty, -1)$. Also

$$\int_{-1}^{1} F(x+y) \frac{\sin \lambda y}{y} \, dy = \int_{0}^{1} \{F(x+y) + F(x-y)\} \frac{\sin \lambda y}{y} \, dy$$

$$= \int_{0}^{1} \{F(x+y) + F(x-y) - 2F(x)\} \frac{\sin \lambda y}{y} \, dy + O(1)$$

† Plancherel (3).

if F(x) is finite. The modulus of this integral does not exceed

$$\lambda \int_{0}^{1/\lambda} |F(x+y)+F(x-y)-2F(x)| dy + \int_{1/\lambda}^{1} |F(x+y)+F(x-y)-2F(x)| \frac{dy}{y}.$$

If $\chi(h) = o(h)$, the first term is o(1), and the second is

$$\int_{1/\lambda}^1 \chi'(y) \frac{dy}{y} = \left[\frac{\chi(y)}{y}\right]_{1/\lambda}^1 + \int_{1/\lambda}^1 \frac{\chi(y)}{y^2} dy = O(1) + o \int_{1/\lambda}^1 \frac{dy}{y} = o(\log \lambda).$$

This proves the theorem.

THEOREM 62.† If f(t) and also $f(t)\log(|t|+2)$ belong to $L^2(-\infty,\infty)$, then

$$\lim_{\lambda \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-\lambda}^{\Lambda} f(t)e^{ixt} dt = F(x)$$
 (3.11.2)

almost everywhere.

We have

$$\int_{-\lambda}^{\lambda} f(t)e^{ixt} dt = \int_{-\lambda}^{\lambda} \left(1 - \frac{|t|}{\lambda}\right) f(t)e^{ixt} dt + \frac{1}{\lambda} \int_{-\lambda}^{\lambda} |t| f(t)e^{ixt} dt.$$

The first term tends to $\sqrt{(2\pi)}F(x)$ almost everywhere, by Theorem 59. Hence it is sufficient to prove that

$$\int_{-\lambda}^{\lambda} |t| f(t) e^{ixt} dt = o(\lambda)$$

almost everywhere. By Theorem 61

$$\phi(\lambda) = \int_{0}^{\lambda} f(t) \log(t+2) e^{ixt} dt = o(\log \lambda)$$

almost everywhere. If x is a point where this is true, then

$$\int_{0}^{\lambda} t f(t)e^{ixt} dt = \int_{0}^{\lambda} \frac{t \phi'(t)}{\log(t+2)} dt$$

$$= \left[\frac{t \phi(t)}{\log(t+2)}\right]_{0}^{\lambda} - \int_{0}^{\lambda} \left\{\frac{1}{\log(t+2)} - \frac{t}{(t+2)\log^{2}(t+2)}\right\} \phi(t) dt$$

$$= \frac{\lambda o(\log \lambda)}{\log(\lambda+2)} + \int_{0}^{\lambda} o(1) dt = o(\lambda),$$

and the theorem follows.

† Plancherel (3).

3.12. In spite of the satisfactory appearance of the above analysis, it is possible to improve on the result.

THEOREM 63. If f(t) and also $f(t)\sqrt{\log(|t|+2)}$ belong to $L^2(-\infty,\infty)$, then (3.11.2) holds almost everywhere, †

Let
$$\Phi(x) = \int_{-\lambda(x)}^{\lambda(x)} f(t)e^{ixt} dt = \int_{-\infty}^{\infty} f(t)\omega(x,t)e^{ixt} dt,$$

where $\lambda(x)$ is any function of x such that $\lambda(x) \leqslant a$, and where $\omega(x,t) = 1$ $(|t| \leqslant \lambda(x))$, 0 $(|t| > \lambda(x))$, so that $\omega(x,t) = 0$ for |t| > aand every x. Then

$$\int_{0}^{\xi} \Phi(x) dx = \int_{0}^{\xi} dx \int_{-\infty}^{\infty} f(t)\omega(x,t)e^{ixt} dt = \int_{-\infty}^{\infty} f(t)\sqrt{\{\log(|t|+2)\}}\chi(t) dt,$$
where
$$\chi(t) = \frac{1}{\sqrt{\{\log(|t|+2)\}}} \int_{0}^{\xi} \omega(x,t)e^{ixt} dx,$$

(the t-range being really finite). Hence

$$\left|\int_{0}^{\xi} \Phi(x) dx\right|^{2} \leqslant \int_{-\infty}^{\infty} |f(t)|^{2} \log(|t|+2) dt \int_{-\infty}^{\infty} |\chi(t)|^{2} dt.$$

Now

$$\int_{-\infty}^{\infty} |\chi(t)|^2 dt = \int_{-\infty}^{\infty} \frac{dt}{\log(|t|+2)} \int_{0}^{\xi} \omega(x,t)e^{ixt} dx \int_{0}^{\xi} \omega(y,t)e^{-iyt} dy$$

$$= \int_{0}^{\xi} dx \int_{0}^{\xi} dy \int_{-\infty}^{\infty} \frac{\omega(x,t)\omega(y,t)}{\log(|t|+2)} e^{i(x-y)t} dt$$

$$= \int_{0}^{\xi} dx \int_{0}^{\xi} dy \int_{-\lambda(x,y)}^{-\lambda(x,y)} \frac{e^{i(x-y)t}}{\log(|t|+2)} dt,$$

where $\lambda(x,y) = \min\{\lambda(x),\lambda(y)\}$. Writing $\lambda(x,y) = \lambda$, we have, on integrating by parts twice,

$$\int_{0}^{\lambda} \frac{\cos(x-y)t}{\log(t+2)} dt = \frac{\sin(x-y)\lambda}{(x-y)\log(\lambda+2)} + \frac{1-\cos(x-y)\lambda}{(x-y)^{2}(\lambda+2)\log^{2}(\lambda+2)} + \frac{1}{(x-y)^{2}} \int_{0}^{\lambda} \left\{1-\cos(x-y)t\right\} \frac{\log(t+2)+2}{(t+2)^{2}\log^{3}(t+2)} dt$$

$$= J_{1} + J_{2} + J_{3},$$

† The theorem for series is given by Plessner (2).

say. We now observe that if F(x, y, t) = F(y, x, t), then

$$\left|\int\limits_0^\xi\int\limits_0^\xi F\{x,y,\lambda(x,y)\}\,dxdy
ight|\leqslant 2\int\limits_0^\xi dx\int\limits_0^\xi |F\{x,y,\lambda(x)\}|\,dy.$$

For let Q be the square $0\leqslant x\leqslant \xi,\, 0\leqslant y\leqslant \xi$, and let

$$Q_1 = Q\{\lambda(x) \leqslant \lambda(y)\}, \qquad Q_2 = Q\{\lambda(x) > \lambda(y)\}.$$

Then in Q_1 , $\lambda(x,y) = \lambda(x)$, and in Q_2 , $\lambda(x,y) = \lambda(y)$. Hence

$$\left| \iint_{Q} F\{x, y, \lambda(x, y)\} dxdy \right|$$

$$\leqslant \iint_{Q_{1}} |F\{x, y, \lambda(x)\}| dxdy + \iint_{Q_{2}} |F\{x, y, \lambda(y)\}| dxdy$$

$$\leqslant 2 \int_{0}^{\xi} dx \int_{0}^{\xi} |F\{x, y, \lambda(x)\}| dy$$

by symmetry.

It follows that

$$\left|\int\limits_0^\xi \int\limits_0^\xi J_1 dx dy\right| \leqslant 2\int\limits_0^\xi dx \int\limits_0^\xi \left|\frac{\sin\{(x-y)\lambda(x)\}}{(x-y)\log\{\lambda(x)+2\}}\right| dy.$$

Now if $0 \leqslant x \leqslant \xi$,

$$\int_{0}^{\xi} \left| \frac{\sin\{(x-y)\lambda(x)\}}{x-y} \right| dy \leqslant 2 \int_{0}^{\xi\lambda(x)} \left| \frac{\sin u}{u} \right| du \leqslant 2 \left(\int_{1}^{\xi\{\lambda(x)+2\}} \frac{du}{u} + 1 \right)$$

$$= 2[1 + \log \xi + \log\{\lambda(x) + 2\}].$$

Hence

$$\left|\int\limits_0^{\xi}\int\limits_0^{\xi}J_1\,dxdy\right|\leqslant 4\int\limits_0^{\xi}\frac{1+\log\xi+\log\{\lambda(x)+2\}}{\log\{\lambda(x)+2\}}\,dx< K(\xi).$$

Similarly

$$igg| \int\limits_0^{\xi} \int\limits_0^{\xi} J_2 \, dx dy igg| \leqslant 2 \int\limits_0^{\xi} rac{dx}{\{\lambda(x) + 2\} \log^2 \{\lambda(x) + 2\}} \int\limits_0^{\xi} rac{1 - \cos\{(x - y)\lambda(x)\}}{(x - y)^2} \, dy \ < A \int\limits_0^{\xi} rac{\lambda(x)}{\{\lambda'(x) + 2\} \log^2 \{\lambda(x) + 2\}} \, dx < K(\xi).$$

Also

$$\begin{split} \left| \int\limits_0^\xi \int\limits_0^\xi J_3 \, dx dy \, \right| & \leq \int\limits_0^\infty \frac{\log(t+2)+2}{(t+2)^2 \log^3(t+2)} \, dt \int\limits_0^\xi dx \int\limits_0^\xi \frac{1-\cos(x-y)t}{(x-y)^2} \, dy \\ & < K(\xi) \int\limits_0^\infty \frac{t\{\log(t+2)+2\}}{(t+2)^2 \log^3(t+2)} \, dt = K(\xi). \end{split}$$

Hence, for every $\lambda(x)$,

$$\left|\int_{0}^{\xi} \Phi(x) dx\right|^{2} < K(\xi) \int_{-\infty}^{\infty} \{f(t)\}^{2} \log(|t|+2) dt.$$

$$\phi(x, T, T') = \max_{T \leq \lambda(x) \leq T'} \int_{x}^{\lambda(x)} f(t) \cos xt dt.$$

Let

Then $\phi(x, T, T')$ is the difference between the real parts of two integrals of type Φ , in which f(t) = 0 for t < T and t > T'. Hence

$$\left| \int_{0}^{\xi} \phi(x,T,T') \, dx \right|^{2}$$

$$< K(\xi) \int_{T}^{T'} \{f(t)\}^{2} \log(t+2) \, dt < K(\xi) \int_{T}^{\infty} \{f(t)\}^{2} \log(t+2) \, dt.$$
As $T' \to \infty$,
$$\phi(x,T,T') \to \phi(x,T) = \max_{T \leqslant \lambda(x)} \int_{T}^{\lambda(x)} f(t) \cos xt \, dt,$$
and $\phi(x,T,T') \geqslant 0$, since $\int_{T}^{\lambda(x)} f(t) \cos xt \, dt = 0$ if $\lambda(x) = T$. Hence†
$$\left| \int_{0}^{\xi} \phi(x,T) \, dx \right|^{2} < K(\xi) \int_{T}^{\infty} |f(t)|^{2} \log(t+2) \, dt.$$

It is then clear that, given ϵ , we can choose a sequence $T_1, T_2,...$ such that $\phi(x, T_n) \to 0$ except when x lies in a part of the interval $(0, \xi)$ of measure less than ϵ . A similar argument applies to the function $\psi(x, T)$ defined with 'min' instead of 'max'. Since

$$\left| \int_{u}^{u'} f(t) \cos xt \ dt \right| = \left| \int_{T_n}^{u'} - \int_{T_n}^{u} \right|$$

$$\leq \phi(x, T_n) - \psi(x, T_n)$$

if $T_n \leqslant u < u'$, it follows that $\int f(t)\cos xt \, dt$ converges for $0 \leqslant x \leqslant \xi$ except for a set of measure less than ϵ .

† By Fatou's theorem; Titchmarsh, Theory of Functions, § 10.81.

$$\int_{0}^{\infty} f(t) \cos xt \ dt$$

exists for almost all x. Similarly, the sine integral converges for almost all x. Hence the limit (3.11.2) exists almost everywhere; since the limit and limit in mean of a sequence are equal almost everywhere, by Theorem 48 its value is almost everywhere F(x).

3.13. Theorems on resultants.

THEOREM 64. If f(x) and g(x) belong to $L^2(-\infty,\infty)$, then (2.1.9) are transforms in the sense that (2.1.8) holds for all values of x.

For a fixed t, the transform of f(u+t) is

$$\lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(u+t)e^{ixu} du = \lim_{a \to \infty} \frac{e^{-ixt}}{\sqrt{(2\pi)}} \int_{-a+t}^{a+t} f(v)e^{ixv} dv$$

$$= F(x)e^{-ixt}.$$

The result therefore follows from Parseval's formula, Theorem 49.

THEOREM 65. If f(x) belongs to $L^2(-\infty,\infty)$, and g(x) to $L(-\infty,\infty)$, then (2.1.9) are transforms of the class L^2 .

Since F belongs to L^2 and G is bounded, FG belongs to L^2 . The integral of its transform is

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(u)G(u) \frac{e^{-ixu}-1}{-iu} du.$$
 (3.13.1)

Now the transform of $G(u)(e^{-ixu}-1)/(-iu)$ is

$$\begin{split} & \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} G(u) \frac{e^{-ixu} - 1}{-iu} e^{-iyu} \, du \\ &= \lim_{a \to \infty} \frac{1}{2\pi} \int_{-a}^{a} \frac{e^{-ixu} - 1}{-iu} e^{-iyu} \, du \int_{-\infty}^{\infty} g(\xi) e^{i\xi u} \, d\xi \\ &= \lim_{a \to \infty} \frac{1}{\pi} \int_{-\infty}^{\infty} g(\xi) \, d\xi \int_{0}^{a} \frac{\sin(x + y - \xi)u - \sin(y - \xi)u}{u} \, du \\ &= \int_{0}^{x + y} g(\xi) \, d\xi \quad (x > 0), \end{split}$$

the ordinary limit existing by dominated convergence. Hence (3.13.1) is equal to

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) \, dy \int_{-y}^{x-y} g(\xi) \, d\xi$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) \, dy \int_{0}^{x} g(u-y) \, du$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{0}^{x} du \int_{-\infty}^{\infty} f(y)g(u-y) \, dy,$$

this inversion being justified by absolute convergence. The theorem now follows on differentiating with respect to x.

The direct proof that, if f is L^2 and g is L, then

$$h(x) = \int_{-\infty}^{\infty} f(y)g(x-y) \ dy$$

is L^2 , follows from the inequalities

$$|h(x)|^{2} \leqslant \int_{-\infty}^{\infty} |f(y)|^{2} |g(x-y)| dy \int_{-\infty}^{\infty} |g(x-y)| dy$$

$$= \int_{-\infty}^{\infty} |f(y)|^{2} |g(x-y)| dy \int_{-\infty}^{\infty} |g(u)| du,$$

$$\int_{-\infty}^{\infty} |h(x)|^{2} dx \leqslant \int_{-\infty}^{\infty} |g(u)| du \int_{-\infty}^{\infty} |f(y)|^{2} dy \int_{-\infty}^{\infty} |g(x-y)| dx$$

$$= \int_{-\infty}^{\infty} |f(y)|^{2} dy \left(\int_{-\infty}^{\infty} |g(u)| du \right)^{2}.$$

THEOREM 66. If f(x) is positive, even, and $L(-\infty,\infty)$, and F(x) is its transform, then F(x) is of the form

$$F(x) = \int_{-\infty}^{\infty} \phi(t)\phi(x-t) dt, \qquad (3.13.2)$$

where ϕ is $L^2(-\infty,\infty)$; and conversely, if F(x) is of this form, then it is the transform of a function f(x) which is positive and L.

If f(x) is positive and L, $\sqrt{f(x)}$ is L^2 ; let G(x) be its transform. Then G is L^2 , and by Theorem 64

$$\int_{-\infty}^{\infty} G(t)G(x-t) dt$$

is the transform of \sqrt{f} . $\sqrt{f} = f$.

Conversely, if (3.13.2) holds, then f is the square of the transform of ϕ , and so is positive and L.

THEOREM 67. If f is L, then $F = \int_{-\infty}^{\infty} \phi(t)\psi(x-t) dt$, where ϕ and ψ are L^2 ; and conversely.

For $f = \sqrt{|f|} \times \sqrt{|f|} \operatorname{sgn} f$.

3.14. Special theorems. Theorem 68. If both f(x) and f'(x) belong to L^2 , then both F(x) and xF(x) belong to L^2 ; and conversely.

We have
$${f(x)}^2-{f(0)}^2=2\int\limits_0^x f(t)f'(t)\ dt$$
,

which tends to a limit as $x \to \infty$. Since $\{f(x)\}^2$ belongs to L, it cannot tend to a limit other than 0, and so tends to 0 as $x \to \infty$. Now

$$\int_{-a}^{a} f'(u)e^{ixu} du = [f(u)e^{ixu}]_{-a}^{a} - ix \int_{-a}^{a} f(u)e^{ixu} du.$$

As $a \to \infty$, the left-hand side converges in mean square, to $\sqrt{(2\pi)}\Phi(x)$ say. The first term on the right tends to 0, uniformly in x. The second term on the right converges in mean square to $-\sqrt{(2\pi)ix}F(x)$, at any rate over a finite interval. Hence

$$\Phi(x) = -ixF(x),$$

and since $\Phi(x)$ belongs to L^2 the result follows.

Conversely, if xF(x) is L^2 , let $\phi(x)$ be its transform. Then

$$\int_{0}^{x} \phi(u) du = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} u F(u) \frac{e^{-ixu} - 1}{-iu} du$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(u) e^{-ixu} du - C$$

say, F(u) being $L(-\infty,\infty)$, since F and xF are L^2 . But the first term

on the right is if(x) almost everywhere, and we may take it as the definition of if(x) everywhere. Then

$$\int_{0}^{x} \phi(u) \ du = if(x) - C$$

everywhere, and the result follows.

The result can obviously be extended to any number of derivatives.

3.15. THEOREM 69. If f(x), $F_c(x)$ are cosine transforms of L^2 , so are

$$\frac{1}{x}\int_{0}^{x}f(t)\ dt, \qquad \int_{t}^{\infty}\frac{F_{c}(t)}{t}\ dt;$$

and similarly for sine transforms.

That the second pair of functions belong to L^2 is a theorem of Hardy. (See Titchmarsh, *Theory of Functions*, p. 396.)

To prove that they are transforms, we have

$$\frac{1}{x}\int_{0}^{x}f(t)\ dt=\sqrt{\left(\frac{2}{\pi}\right)}\int_{0}^{\infty}F_{c}(y)\frac{\sin xy}{xy}\ dy.$$

The cosine transform of this is

$$\frac{2}{\pi} \frac{d}{du} \int_{0}^{\infty} \frac{\sin xu}{x} dx \int_{0}^{\infty} F_{c}(y) \frac{\sin xy}{xy} dy$$

$$= \frac{2}{\pi} \frac{d}{du} \int_{0}^{\infty} \frac{F_{c}(y)}{y} dy \int_{0}^{\infty} \frac{\sin xu \sin xy}{x^{2}} dx$$

$$= \frac{d}{du} \int_{0}^{\infty} \frac{F_{c}(y)}{y} \min(u, y) dy$$

$$= \frac{d}{du} \left\{ \int_{0}^{u} F_{c}(y) dy + u \int_{u}^{\infty} \frac{F_{c}(y)}{y} dy \right\} = \int_{u}^{\infty} \frac{F_{c}(y)}{y} dy,$$

almost everywhere.

The inversion is justified by absolute convergence. In fact, if y < u,

$$\int_{0}^{\infty} \left| \frac{\sin x u \sin x y}{x^2} \right| dx \leqslant \int_{0}^{1/u} u y dx + \int_{1/u}^{1/y} \frac{y}{x} dx + \int_{1/y}^{\infty} \frac{dx}{x^2}$$

$$= y + y \log \frac{u}{y} + y = \left(2 + \log \frac{u}{y} \right) y,$$

and, similarly, if u < y; and the integral

$$\int\limits_{0}^{u}|F_{c}(y)|\left(2+\log\frac{u}{y}\right)dy\,+u\int\limits_{u}^{\infty}\frac{|F_{c}(y)|}{y}\left(2+\log\frac{y}{u}\right)dy$$

is convergent if F_c belongs to L^2 .

3.16. Another case of Parseval's formula.

THEOREM 70. If f is L, and g is L^2 and bounded, then

$$\lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda} \right) F(x) G(x) \ dx = \int_{-\infty}^{\infty} f(x) g(-x) \ dx.$$

We have

$$\int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda}\right) F(x) G(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda}\right) G(x) dx \int_{-\infty}^{\infty} f(t) e^{ixt} dt$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(t) dt \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda}\right) G(x) e^{ixt} dx,$$

inverting by uniform convergence. As in the proof of Theorem 59, the inner integral is equal to

$$\sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} g(u) \frac{\sin^2 \frac{1}{2} \lambda(x+u)}{\lambda(x+u)^2} du.$$

This is bounded if g(u) is bounded, and tends almost everywhere to $\sqrt{(2\pi)g(-x)}$. The result therefore follows by dominated convergence.

3.17. Mellin transforms. We shall say that f(x) belongs to \mathfrak{L}^2 if

$$\int_{0}^{\infty} |f(x)|^{2} \frac{dx}{x} < \infty.$$

Theorem 71. Let $x^k f(x)$ belong to \mathfrak{L}^2 . Then

$$\mathfrak{F}(s,a) = \int_{1/a}^{a} f(x)x^{s-1} dx \quad (\mathbf{R}(s) = k)$$

converges in mean square over $(k-i\infty, k+i\infty)$, to $\mathfrak{F}(s)$ say;

$$f(x,a) = \frac{1}{2\pi i} \int_{k-ia}^{k+ia} \mathfrak{F}(s) x^{-s} ds$$

converges in mean to f(x), in the sense that

$$\lim_{a\to\infty}\int_{0}^{\infty}|f(x)-f(x,a)|^{2}x^{2k-1}\,dx=0;$$

and

$$\int_{0}^{\infty} |f(x)|^{2} x^{2k-1} dx = \frac{1}{2\pi} \int_{-\infty}^{\infty} |\mathfrak{F}(k+it)|^{2} dt.$$

This follows from Plancherel's theorem by the usual transformation.

THEOREM 72. Let $x^k f(x)$ and $x^{1-k} g(x)$ belong to \mathfrak{L}^2 . Then

$$\int_{0}^{\infty} f(x)g(x) dx = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s)\mathfrak{G}(1-s) ds.$$

This is the corresponding transformation of Theorem 49.

THEOREM 73. Let $x^k f(x)$ and $x^{\sigma-k} g(x)$ belong to \mathfrak{L}^2 . Then

$$f(x)g(x), \qquad \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(w)\mathfrak{G}(s-w) \ dw$$

are Mellin transforms in the sense that

$$\int_{0}^{\infty} f(x)g(x)x^{s-1} dx = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(w)\mathfrak{G}(s-w) dw$$

for all values of t.

This is obtained by replacing g(x) by $g(x)x^{s-1}$ in the previous theorem.

TRANSFORMS OF OTHER L-CLASSES

4.1. Transforms of functions of L^p . Plancherel's theorem can be extended from the exponent 2 to a general exponent p. Throughout the chapter we write p' = p/(p-1), and similarly for other letters.

Theorem 74.† Let f(x) belong to $L^p(-\infty,\infty)$, where 1 . $Then, as <math>a \to \infty$,

$$F(x,a) = \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(t)e^{ixt} dt$$
 (4.1.1)

converges in mean with exponent p'. The mean limit F(x), called the transform of f(x), satisfies

$$\int_{-\infty}^{\infty} |F(x)|^{p'} dx \leqslant \frac{1}{(2\pi)^{\frac{1}{2}p'-1}} \left\{ \int_{-\infty}^{\infty} |f(x)|^{p} dx \right\}^{\frac{1}{(p-1)}}$$
(4.1.2)

The Fourier reciprocity holds in the sense that

$$F(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} f(t) \frac{e^{ixt} - 1}{it} dt, \qquad (4.1.3)$$

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} F(t) \frac{e^{-ixt} - 1}{-it} dt, \qquad (4.1.4)$$

almost everywhere.

As in the L^2 case, we might replace F(x,a) by

$$F(x,a,b) = \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{b} f(t)e^{ixt} dt,$$

where $a \to \infty$, $b \to \infty$, in any manner.

4.2. The most obvious source of such results is in the formulae (2.1.11). If k is an integer, the transform of $\{F(x)\}^k$ is formally $(2\pi)^{-\frac{1}{2}k+\frac{1}{2}}\phi_k(x)$, where

$$\phi_k(x) = \int_{-\infty}^{\infty} f(u_{k-1}) du_{k-1} \dots \int_{-\infty}^{\infty} f(u_1) f(x - u_1 - \dots - u_{k-1})' du_1.$$

We can deal with such integrals by means of the following lemmas. † Titchmarsh (2). † See Hardy, Littlewood, and Pólya, Inequalities, pp. 198-203.

LEMMA α . (Young's inequality.) If f(x) and g(x) belong to $L^{1/(1-\lambda)}$ and $L^{1/(1-\mu)}$ respectively, where $\lambda > 0$, $\mu > 0$, $\lambda + \mu < 1$, then

$$\left| \int_{-\infty}^{\infty} fg \, dx \right|$$

$$\leq \left(\int_{-\infty}^{\infty} |f|^{1/(1-\lambda)} |g|^{1/(1-\mu)} \, dx \right)^{1-\lambda-\mu} \left(\int_{-\infty}^{\infty} |f|^{1/(1-\lambda)} \, dx \right)^{\mu} \left(\int_{-\infty}^{\infty} |g|^{1/(1-\mu)} \, dx \right)^{\lambda}.$$

Hölder's inequality for three functions is

$$\bigg|\int\limits_{-\infty}^{\infty}\phi\psi\chi\;dx\bigg|\leqslant \bigg(\int\limits_{-\infty}^{\infty}\;|\phi|^{1/\alpha}\;dx\bigg)^{\alpha}\bigg(\int\limits_{-\infty}^{\infty}\;|\psi|^{1/\beta}\;dx\bigg)^{\beta}\bigg(\int\limits_{-\infty}^{\infty}\;|\chi|^{1/\gamma}\;dx\bigg)^{\gamma},$$

where $\alpha + \beta + \gamma = 1$, $\alpha > 0$, $\beta > 0$, $\gamma > 0$. Putting

$$|\phi|^{1/lpha} = |\psi|^{1/eta} |\chi|^{1/\gamma}, \qquad |\psi|^{1+lpha/eta} = |f|, \qquad |\chi|^{1+lpha/\gamma} = |g|,$$

and $\gamma = \lambda$, $\beta = \mu$, the result follows.

Lemma
$$\beta$$
. Let $\mathfrak{I}_p(f)=\Bigl(\int\limits_{-\infty}^\infty|f(x)|^p\ dx\Bigr)^{1/p}.$ $\phi(x)=\int\limits_{-\infty}^\infty f(t)g(x-t)\ dt,$

then

If

$$\mathfrak{I}_{1/(1-\lambda-\mu)}(\phi) \leqslant \mathfrak{I}_{1/(1-\lambda)}(f)\mathfrak{I}_{1/(1-\mu)}(g).$$

Young's inequality gives

$$|\phi(x)| \leqslant \left(\int_{-\infty}^{\infty} |f(t)|^{1/(1-\lambda)} |g(x-t)|^{1/(1-\mu)} dt\right)^{1-\lambda-\mu} \times \\ \cdot \times \{\mathfrak{I}_{1/(1-\lambda)}(f)\}^{\mu/(1-\lambda)} \{\mathfrak{I}_{1/(1-\mu)}(g)\}^{\lambda/(1-\mu)}.$$

Hence

$$\begin{split} \int\limits_{-\infty}^{\infty} |\phi(x)|^{1/(1-\lambda-\mu)} \, dx & \leqslant \int\limits_{-\infty}^{\infty} |f(t)|^{1/(1-\lambda)} \, dt \int\limits_{-\infty}^{\infty} |g(x-t)|^{1/(1-\mu)} \, dx \times \\ & \times \{\mathfrak{I}_{1/(1-\lambda)}(f)\}^{\mu/(1-\lambda)(1-\lambda-\mu)} \{\mathfrak{I}_{1/(1-\mu)}(g)\}^{\lambda/(1-\mu)(1-\lambda-\mu)} \\ & = \{\mathfrak{I}_{1/(1-\lambda)}(f)\}^{\frac{1}{1-\lambda}\left\{1+\frac{\mu}{1-\lambda-\mu}\right\}} \{\mathfrak{I}_{1/(1-\mu)}(g)\}^{\frac{1}{1-\mu}\left\{1+\frac{\lambda}{1-\lambda-\mu}\right\}}, \end{split}$$

and the result follows.

LEMMA γ . If f(x) belongs to $L^{2k/(2k-1)}$, then $\phi_k(x)$ belongs to L^2 , and

$$\int\limits_{-\infty}^{\infty} \left\{\phi_k(x)\right\}^2 \, dx \leqslant \left(\int\limits_{-\infty}^{\infty} |f(x)|^{2k/(2k-1)} \, dx\right)^{2k-1}$$

4362 F

We have

$$\phi_k(x) = \int_{-\infty}^{\infty} f(t)\phi_{k-1}(x-t) dt,$$

and, applying the previous lemma repeatedly,

$$\begin{split} \mathfrak{I}_{1/(1-\frac{k}{2k})}(\phi_{k}) &\leqslant \mathfrak{I}_{1/(1-\frac{k-1}{2k})}(\phi_{k-1})\mathfrak{I}_{1/(1-\frac{1}{2k})}(f) \\ &\leqslant \mathfrak{I}_{1/(1-\frac{k-2}{2k})}(\phi_{k-2}) \left\{ \mathfrak{I}_{1/(1-\frac{1}{2k})}(f) \right\}^{2} \leqslant \ldots \leqslant \left\{ \mathfrak{I}_{1/(1-\frac{1}{2k})}(f) \right\}^{k}, \end{split}$$

the result stated.

4.3. Proof† of Theorem 74 for p = 2k/(2k-1). Suppose first that f(x) and g(x) belong to L^2 , and are zero outside a finite interval. Then

$$\frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty}g(u_1)f(x-u_1)\ du_1$$

satisfies the same conditions, and (e.g. by Theorem 64) its transform is F(x)G(x).

Repeating the argument k-1 times, and making all the functions equal, we see that the functions $\{F(x)\}^k$ and $(2\pi)^{-\frac{1}{2}k+\frac{1}{2}}\phi_k(x)$ are transforms of the class L^2 . Hence

$$\int_{-\infty}^{\infty} |F(x)|^{2k} dx = \frac{1}{(2\pi)^{k-1}} \int_{-\infty}^{\infty} |\phi_k(x)|^2 dx$$

$$\leq \frac{1}{(2\pi)^{k-1}} \left(\int_{-\infty}^{\infty} |f(x)|^{2k/(2k-1)} dx \right)^{2k-1}$$

This proves (4.1.2) for the special class of functions considered, and p = 2k/(2k-1).

Now let f(x) be any function of the class $L^{2k/(2k-1)}$. The function equal to f(x) if a < |x| < b and $|f(x)| \le n$, and to 0 elsewhere, belongs to the special class. Applying the above result to it, and making $n \to \infty$, we obtain

$$\int_{-\infty}^{\infty} |F(x,b) - F(x,a)|^{2k} dx \leqslant \frac{1}{(2\pi)^{k-1}} \left\{ \left(\int_{-b}^{-a} + \int_{a}^{b} \right) |f(x)|^{2k/(2k-1)} dx \right\}^{2k-1}.$$

The right-hand side tends to 0 as $a \to \infty$, $b \to \infty$. Hence F(x, a) con-

[†] The argument is analogous to that of W. H. Young for Fourier series. For a list of Young's papers see Zygmund's bibliography.

verges in mean, to F(x) say. We have

$$\int_{-\infty}^{\infty} |F(x)|^{2k} dx = \lim_{a \to \infty} \int_{-\infty}^{\infty} |F(x, a)|^{2k} dx$$

$$\leq \lim_{a \to \infty} \frac{1}{(2\pi)^{k-1}} \left(\int_{-a}^{a} |f(x)|^{2k/(2k-1)} dx \right)^{2k-1}$$

$$= \frac{1}{(2\pi)^{k-1}} \left(\int_{-\infty}^{\infty} |f(x)|^{2k/(2k-1)} dx \right)^{2k-1}.$$

Also

$$\int_{0}^{\xi} F(x) dx = \lim_{a \to \infty} \int_{0}^{\xi} F(x, a) dx = \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\xi} dx \int_{-a}^{a} f(t)e^{ixt} dt$$

$$= \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(t) \frac{e^{i\xi t} - 1}{it} dt = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(t) \frac{e^{i\xi t} - 1}{it} dt,$$
so that
$$F(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{0}^{\infty} f(t) \frac{e^{ixt} - 1}{it} dt$$

almost everywhere. Again, if $0 < \xi < a$,

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(u,a) \frac{e^{-i\xi u} - 1}{-iu} du = \frac{1}{2\pi} \int_{-a}^{a} f(t) dt \int_{-\infty}^{\infty} \frac{e^{-i\xi u} - 1}{-iu} e^{iut} du$$
$$= \int_{0}^{\xi} f(t) dt,$$

the inversion being justified by the bounded convergence of the *u*-integral. Making $a \to \infty$, we obtain

$$\frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty}F(u)\frac{e^{-i\xi u}-1}{-iu}\,du=\int_{0}^{\xi}f(t)\,dt,$$

since $(e^{-i\xi u}-1)/(-iu)$ belongs to $L^{2k/(2k-1)}$. Hence

$$\frac{1}{\sqrt{(2\pi)}}\frac{d}{dx}\int_{-\infty}^{\infty}F(u)\frac{e^{-ixu}-1}{-iu}du=f(x)$$

almost everywhere (x > 0). Similarly for x < 0.

4.4. Extension to general p. To extend the theorem to other values of p we first prove the corresponding theorem for trigonometrical polynomials.

LEMMA δ . For any given numbers c_m $(-n \leq m \leq n)$,

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \left| \sum_{m=-n}^{n} c_m e^{imx} \right|^{p'} dx \leqslant \left(\sum_{m=-n}^{n} |c_m|^p \right)^{1/(p-1)}. \tag{4.4.1}$$

We give two proofs, the original proof of Young and Hausdorff and a later one of Hardy and Littlewood.

(i) \dagger Let f(t) be a function of $L(-\pi, \pi)$, and let

$$c_m = \frac{1}{2\pi} \int_{-\pi}^{\pi} f(t)e^{-imt} dt \quad (m = 0, \pm 1, ...).$$
 (4.4.2)

We write

$$J_p(f) = \left(\frac{1}{2\pi} \int_{-\pi}^{\pi} |f(t)|^p dt\right)^{1/p} \tag{4.4.3}$$

and

$$S_p(f) = \left(\sum_{-\infty}^{\infty} |c_m|^p\right)^{1/p}$$
 (4.4.4)

We have to prove that, for any trigonometrical polynomial f(t),

$$J_{p'} \leqslant S_p \quad (1 (4.4.5)$$

If p is of the form 2k/(2k-1), this follows from the argument for sums parallel to that just given for integrals. If

 $egin{aligned} f(x) &= \sum c_m e^{imx}, & g(x) &= \sum \gamma_m e^{imx}, \ f(x)g(x) &= \sum d_m e^{imx}, \ d_m &= \sum c_r \gamma_{m-r}. \end{aligned}$

then where

The analogue of Lemma β is therefore

$$S_{1/(1-\lambda-\mu)}(fg) \leqslant S_{1/(1-\lambda)}(f)S_{1/(1-\mu)}(g),$$
 (4.4.6)

and, as in Lemma γ , it follows that

$$S_2(f^k) \leqslant \left\{ S_1 / (1 - \frac{1}{2k})(f) \right\}^k.$$
 (4.4.7)

But for any trigonometrical polynomial ϕ

$$S_2(\phi) = J_2(\phi).$$
 (4.4.8)

Thus

$$S_2(f^k) = J_2(f^k) = \{J_{2k}(f)\}^k,$$

and (4.4.5) with p = 2k/(2k-1) follows.

To extend it to other values of p we consider maximal polynomials, viz. those for which $J_{p'}^{p'}$ is a maximum for a given value of S_{p}^{p} , and † Hausdorff (1).

a given n. Since S_p^p , $J_{p'}^{p'}$ are continuous functions with continuous partial derivatives with respect to the components x_m , y_m of the coefficients $c_m = x_m + iy_m$, maximal polynomials exist; and we can determine them by the ordinary method of the differential calculus. Let

$$\phi = \sum_{m=-n}^{n} (x_m^2 + y_m^2)^{\frac{1}{2}p}, \quad \psi = \frac{1}{2\pi} \int_{-\pi}^{\pi} |f(t)|^{p'} dt.$$

Then the condition for a maximum is

$$rac{\partial \psi}{\partial x_m} = rac{\partial \psi}{\partial y_m} = \lambda \quad (m = -n, ..., n).$$
 $rac{\partial \psi}{\partial x_m} + i rac{\partial \psi}{\partial y_m} = \lambda \quad (m = -n, ..., n).$

Hence

Now†

$$\frac{\partial \phi}{\partial x_m} + i \frac{\partial \phi}{\partial y_m} = p(x_m + i y_m)(x_m^2 + y_m^2)^{\lfloor p-1} = p|c_m|^{p-1} \operatorname{sgn} c_m.$$

Also

$$\begin{split} |f(t)|^2 &= f(t)\bar{f}(t), \\ 2|f(t)|\frac{\partial}{\partial x_m}|f(t)| &= \bar{f}(t)\frac{\partial}{\partial x_m}f(t) + f(t)\frac{\partial}{\partial x_m}\bar{f}(t) \\ &= \bar{f}(t)e^{imt} + f(t)e^{-imt}, \end{split}$$

and similarly $2|f(t)|\frac{\partial}{\partial y_m}|f(t)| = \bar{f}(t)ie^{imt} - f(t)ie^{-imt}$.

Hence

$$2|f(t)|\left(\frac{\partial}{\partial x_m}+i\frac{\partial}{\partial y_m}\right)|f(t)| = 2f(t)e^{-imt},$$

$$\left(\frac{\partial}{\partial x_m}+i\frac{\partial}{\partial y_m}\right)|f(t)| = e^{-imt}\operatorname{sgn}f(t).$$

Hence

$$\frac{\partial \psi}{\partial x_m} + i \frac{\partial \psi}{\partial y_m} = \frac{p'}{2\pi} \int_{-\pi}^{\pi} |f(t)|^{p'-1} \operatorname{sgn} f(t) e^{-imt} dt.$$

Hence

$$\frac{p'}{2\pi} \int_{-\pi}^{\pi} |f(t)|^{p'-1} \operatorname{sgn} f(t) e^{-imt} dt = \lambda p |c_m|^{p-1} \operatorname{sgn} c_m \quad (m = -n, ..., n).$$

$$\dagger \operatorname{sgn} z = \frac{z}{|z|} \quad (z \neq 0), \quad \operatorname{sgn} 0 = 0.$$
(4.4.9)

To find λ , multiply by $\overline{c_m}$ and sum. We obtain

$$\frac{p'}{2\pi} \int_{-\pi}^{\pi} |f(t)|^{p'} dt = \lambda p \sum |c_m|^p,$$

$$p'J_{e'}^p = \lambda p S_{e'}^p. \tag{4.4.10}$$

i.e.

$$p'J_{p'}^{p'} = \lambda p S_p^p. \tag{4.4.10}$$

Now (4.4.9) gives the first 2n+1 Fourier coefficients of the function $|f(t)|^{p'-1}\operatorname{sgn} f(t)$. Hence Bessel's inequality for this function gives

$$\sum_{m=-n}^{n} \left| \frac{\lambda p}{p'} |c_m|^{p-1} \operatorname{sgn} c_m \right|^2 \leqslant \frac{1}{2\pi} \int_{-\pi}^{\pi} |f(t)|^{p'-1} \operatorname{sgn} f(t)|^2 dt,$$

i.e.

$$\frac{\lambda^2 p^2}{p'^2} \sum_{m=-n}^n |c_m|^{2p-2} \leqslant \frac{1}{2\pi} \int_{-\pi}^{\pi} |f(t)|^{2p'-2} dt,$$

i.e.

$$\lambda^2 p^2 S_{2p-2}^{2p-2} \leqslant p'^2 J_{2p'-2}^{2p'-2}. \tag{4.4.11}$$

From (4.4.10) and (4.4.11) it follows that

$$\frac{J_{p'}^{p'}}{S_p^p} \leqslant \frac{J_{2p'-2}^{p'-1}}{S_{2p-2}^{p-1}} \tag{4.4.12}$$

for every maximal polynomial.

Let

$$r'=2p'-2, \qquad r=2/(3-p).$$

Then it follows from Hölder's inequality that

$$S_r^{p'-1} \leqslant S_{2p-2}^{p-1} S_p^{p'-p},$$
 (4.4.13)

and (4.4.12) and (4.4.13) give

$$\left(\frac{J_{p'}}{S_p}\right)^{p'} \leqslant \left(\frac{J_{r'}}{S_r}\right)^{p'-1}.$$
 (4.4.14)

Now suppose that (4.4.5) holds for p' = r and all polynomials. Then it follows from (4.4.14) that it holds for $p' = \frac{1}{2}r' + 1$ for maximal polynomials, and so a fortiori for all polynomials. We have already proved it for p'=2k. Hence we deduce it in succession for

$$p'=k+1, \frac{k+3}{2}, \frac{k+7}{4}, ...,$$

i.e. for all rational numbers whose denominators are powers of 2. Since these numbers are everywhere dense, the general result now follows from the continuity of S_n and $J_{n'}$ as functions of p.

(ii) We again consider maximal polynomials; but, instead of the general condition for a maximum of a function of many variables,

† Hardy and Littlewood (1). See also F. Riesz (1).

we use the theorem that, in Hölder's inequality

$$\sum a_m b_m \le (\sum |a_m|^p)^{1/p} (\sum |b_m|^{p'})^{1/p'},$$

the case of equality occurs only if the $|a_m|^p$ and $|b_m|^{p'}$ are proportional. Also this proof is independent of the lemmas of § 4.2.

We define S_p and J_p as before, and write

$$f_n(x) = \sum_{m=-n}^n c_m e^{imx}.$$

For given n and p, let the upper bound of $S_{p'}(f_n)/J_p(f)$ for all f be denoted by $M = M(n) = M_p(n)$; and let the upper bound of $J_{p'}(f_n)/S_p(f_n)$ for all sets of c_m be $M'=M'(n)=M'_p(n)$. We first show that these bounds exist for every n.

We may suppose on grounds of homogeneity that $S_{n'}(f_n) = 1$. Then $|c_m|^{p'} \geqslant 1/(2n+1)$ for some value of m. Hence

$$(2n+1)^{-1/p'}\leqslant |c_m|\leqslant rac{1}{2\pi}\int\limits_{-\pi}^{\pi}|f(x)|\;dx\leqslant J_p,$$
 and so $M_p(n)\leqslant (2n+1)^{1/p'}.$ Again, let $g(x)=|f_n(x)|^{p'-1}{
m sgn}f_n(x),$ and let $\gamma_m=rac{1}{2\pi}\int\limits_{-\pi}^{\pi}g(x)e^{-imx}\;dx.$

and let

and so

$$\begin{split} J_{p'}^{n(e)}(f_n) &= \frac{1}{2\pi} \int_{-\pi}^{\pi} f_n(x) g(x) \ dx = \sum_{m=-n}^{n} c_m \gamma_m \leqslant \sum_{m=-n}^{n} |c_m \gamma_m| \\ &\leqslant S_p(f_n) S_{p'}(g_n) \leqslant M S_p(f_n) J_p(g) = M S_p(f_n) J_{p'}^{1/(p-1)}(f_n), \\ &(4.4.15) \end{split}$$

and, dividing by $J_{n'}^{1/(p-1)}(f_n)$, it follows that M' is finite, and $M' \leq M$.

Again, if
$$h_n(x) = \sum_{m=-n}^n |c_m|^{p'-1} \operatorname{sgn} c_m e^{imx},$$

we have (by an obvious term-by-term integration)

$$S_{p'}^{p'}(f_n) = \frac{1}{2\pi} \int_{-\pi}^{\pi} f(x)h_n(x) dx \leqslant J_p(f)J_{p'}(h_n) \leqslant M'J_p(f)S_p(h_n)$$

$$= M'J_n(f)S_{p'}^{p'-1}(f_n),$$

and hence $M \leq M'$. Hence in fact M = M'. The example f(x) = 1shows that $M \geqslant 1$.

Suppose now that $f_n(x)$ is a polynomial for which the maximum M'

of $J_{p'}(f_n)/S_p(f_n)$ is attained (since it is a continuous function of the variables c_m , there is such a polynomial). Since M=M', the extreme terms of the chain (4.4.15) are then equal, and so all the terms are equal. The case of Hölder's inequality used is therefore an equality, and hence

and hence $|c_m|^p = \lambda |\gamma_m|^{p'} \quad (-n \leqslant m \leqslant n),$

where λ is independent of m. Hence

$$S_p^p(f_n) = \lambda S_p^{p'}(g_n).$$

But, since f_n is a maximal polynomial,

$$S_p(f_n) = \frac{1}{M} J_{p'}(f_n) = \frac{1}{M} J_p^{p-1}(g) = M^{-p} S_p^{p-1}(g_n),$$

the last step depending on the equality of the 5th and 6th terms of (4.4.15). Hence $\lambda = M^{-p^*} S_p^{p^* - p - p'}(q_n).$

Let r' = 2p' - 2, r = 2/(3-p).

Then by Bessel's inequality

$$S_2^2(g_n) \leqslant rac{1}{2\pi} \int\limits_{-\pi}^{\pi} |g|^2 dx = J_r^r(f_n) \leqslant M_r^r S_r^{r'}(f_n)$$

$$=M_r^{r'}\lambda_p^{\frac{r'}{p}}\{S_{r|(p-1)}(g_n)\}^{\frac{r'}{p-1}}=M_r^{r'}M_p^{-pr'}\{S_{p'}(g_n)\}^{r'\binom{p-1-1}{p-1}}\{S_{r|(p-1)}(g_n)\}^{\frac{r'}{p-1}}.$$

Since
$$\sum |\gamma_m|^{\frac{r}{p-1}} \leqslant (\sum |\gamma_m|^2)^{\frac{p-1}{3-p}} \left(\sum |\gamma_m|^{\frac{p}{p-1}}\right)^{\frac{2(2-p)}{3-p}},$$

the product of these S-terms on the right-hand side does not exceed $S_2^2(g_n)$. Hence $1 \leq M_r^r M_n^{-pr}$.

and so $M_r\geqslant M_p^p\geqslant M_p.$

We can now repeat the argument with p replaced by r, and r by s=2/(3-r); and so on indefinitely. We thus obtain a sequence of values of p tending to 1 (since r'-2=2(p'-2), etc.) through which M_p is non-decreasing. But

$$M_p(n) \leqslant (2n+1)^{1/p'} \rightarrow 1$$

as $p \to 1$, $p' \to \infty$, for a fixed n. Hence $M_p(n) = 1$.

4.5. We can now prove Theorem 74 by the method used in § 3.2. Let f(x) belong to L^p , $1 , and define <math>a_{\nu}$ and $\Phi_n(x)$ as in § 3.2. Then

$$\int\limits_{-\pi\lambda}^{\pi\lambda}|\Phi_n(x)|^{p'}\,dx=\lambda\int\limits_{-\pi}^{\pi}\left|\sum\limits_{\nu=-n}^na_{\nu}\,e^{i\nu x}\right|^{p'}\,dx\leqslant 2\pi\lambda\Big(\sum\limits_{\nu=-n}^n|a_{\nu}|^p\Big)^{1/(p-1)}$$

by the above lemma; and

$$|a_{\nu}|^{p}\leqslant \int\limits_{\nu/\lambda}^{(\nu+1)/\lambda}|f(x)|^{p}~dx \bigg(\int\limits_{\nu/\lambda}^{(\nu+1)/\lambda}dx\bigg)^{p-1}= \frac{1}{\lambda^{p-1}}\int\limits_{\nu/\lambda}^{(\nu+1)/\lambda}|f(x)|^{p}~dx.$$
 Hence
$$\int\limits_{-\pi\lambda}^{\pi\lambda}|\Phi_{n}(x)|^{p'}~dx\leqslant 2\pi\bigg(\int\limits_{-b}^{b}|f(x)|^{p}~dx\bigg)^{1/(p-1)}$$

It follows as in § 3.2 that

$$\int_{-\infty}^{\infty} |F(x,b) - F(x,a)|^{p'} dx \leqslant (2\pi)^{1-\frac{1}{2}p'} \left\{ \left(\int_{-b}^{-a} + \int_{a}^{b} \right) |f(x)|^{p} dx \right\}^{1/(p-1)}$$

Hence F(x,a) converges in mean as $a \to \infty$, to F(x) say, with exponent p'. The remainder of the proof is the same as in the special case where p' = 2k.

Still another proof of Theorem 74 can be obtained from a general theorem of M. Riesz on functional operations. See Zygmund, § 9.2.

4.6. The Parseval formula.

Theorem 75. If f(x) and G(x) belong to $L^{p}(-\infty,\infty)$, 1 ,and F(x) and g(x) are their transforms, then (2.1.1) holds.

We know that if $\phi(x)$ is L^p , and $\psi(x,a)$ converges in mean to $\psi(x)$ with exponent p', then

$$\lim_{a\to\infty} \int \{\psi(x) - \psi(x,a)\} \phi(x) \ dx = 0. \tag{4.6.1}$$

Now

$$\int_{-b}^{b} F(x,a)G(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{-b}^{b} G(x) dx \int_{-a}^{a} f(t)e^{ixt} dt$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(t) dt \int_{-b}^{b} G(x)e^{ixt} dx = \int_{-a}^{a} f(t)g(-t,b) dt.$$

Making $a \to \infty$, and applying (4.6.1) to the left-hand side, we obtain

$$\int_{-b}^{b} F(x)G(x) dx = \int_{-\infty}^{\infty} f(t)g(-t,b) dt.$$

Making $b \to \infty$, and applying (4.6.1) to the right-hand side, we obtain (2.1.1).

There are also obvious extensions of Theorems 58-62.

† Titchmarsh, Theory of Functions, § 12.53.

4.7. Theorems on resultants.

THEOREM 76. If f(x), F(x) are transforms of L^p , $L^{p'}$, and g(x), G(x) of $L^{p'}$, L^p , then (2.1.9) are transforms in the sense that (2.1.8) holds for all values of x.

Proof similar to that of Theorem 64.

THEOREM 77. If f(x), F(x) are transforms of L^p , $L^{p'}$, and g(x) is L, then (2.1.9) are transforms of L^p , $L^{p'}$.

Proof similar to that of Theorem 65.

THEOREM 78. Let f(x), F(x) be transforms of L^{ν} , $L^{\nu'}$, and g(x), G(x) of L^{q} , $L^{q'}$, where

$$\frac{1}{p} + \frac{1}{q} > 1. \tag{4.7.1}$$

Then

$$F(x)G(x), \qquad \frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty} f(y)g(x-y) \ dy$$

are transforms of classes $L^{P'}$, L^{P} respectively, where

$$P = \frac{pq}{p + q - pq}.$$

That the resultant of f and g belongs to L^P follows from Lemma β of § 4.2, with $1-\lambda=1/p$, $1-\mu=1/q$. That FG belongs to $L^{P'}$ follows at once from Hölder's inequality in the form

$$\int |FG|^{P'} dx \leqslant \Big(\int |F|^{p'} dx\Big)^{P'/p'} \Big(\int |G|^{q'} dx\Big)^{P'/q'}.$$

The condition (4.7.1) implies that p < q', q < p'. Suppose that $p \leqslant q$. Then p < p', i.e. p < 2.

Suppose that $1 < P' \le 2$. Then FG has a transform, the integral of which is

$$\frac{1}{\sqrt{(2\pi)}}\int_{-\infty}^{\infty}F(u)G(u)\frac{e^{-ixu}-1}{-iu}\,du.$$

Now $G(u)(e^{-ixu}-1)/(-iu)$ belongs to L and to $L^{q'}$, and so to L^{p} ; and, by Theorem 74,

$$\int\limits_{y}^{x+y}g(\xi)\;d\xi=\frac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^{\infty}G(u)\frac{e^{-ixu}-1}{-iu}\,e^{-iyu}\;du,$$

i.e. it is the transform of $G(u)(e^{-ixu}-1)/(-iu)$. Hence, by Theorem 76,

$$\int_{-\infty}^{\infty} F(u)G(u) \frac{e^{-ixu}-1}{-iu} du = \int_{-\infty}^{\infty} f(y) dy \int_{-y}^{x-y} g(\xi) d\xi$$

$$= \int_{-\infty}^{\infty} f(y) dy \int_{0}^{x} g(u-y) du = \int_{0}^{x} du \int_{-\infty}^{\infty} f(y)g(u-y) dy,$$

and the result follows on differentiating.

Suppose next that 1 < P < 2. Then

$$\frac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^{\infty}f(y)g(x-y)\;dy$$

has a transform, the integral of which is

$$\frac{1}{2\pi}\int_{-\infty}^{\infty}\frac{e^{ixu}-1}{iu}\,du\int_{-\infty}^{\infty}f(y)g(u-y)\,dy.$$

This is the limit as $a \to \infty$ of

$$\begin{split} \frac{1}{2\pi} \int\limits_{-\infty}^{\infty} \frac{e^{ixu}-1}{iu} \; du \; \int\limits_{a}^{a} \; f(y)g(u-y) \; dy \\ \left(\text{since} \int\limits_{-\infty}^{\infty} \dots dy = \text{l.i.m.} \int\limits_{-a}^{a} \right) \\ &= \frac{1}{2\pi} \int\limits_{-a}^{a} f(y) \; dy \int\limits_{-\infty}^{\infty} g(u-y) \frac{e^{ixu}-1}{iu} \; du, \end{split}$$

and by the Parseval formula (for q, q') the inner integral is equal to

$$\sqrt{(2\pi)}\int\limits_0^xG(v)e^{ivy}\;dv.$$

Hence we obtain

$$\frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} f(y) \, dy \int_{0}^{x} G(v)e^{ivy} \, dv = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{x} G(v) \, dv \int_{-a}^{a} f(y)e^{ivy} \, dy$$

$$\rightarrow \int_{0}^{x} G(v)F(v) \, dv,$$

since G(v) belongs to L^p over (0,x). Hence the result.

If p = 1 and q = 1, then P = 1; see Theorem 41.

4.8. Another extension of Plancherel's theorem. We shall next obtain a generalization of Plancherel's theorem in a different direction, due to Hardy and Littlewood.†

† Hardy and Littlewood (1).

THEOREM 79. If $|f(x)|^{qx^{q-2}}$ (q > 2) belongs to $L(-\infty, \infty)$, then F(x), the transform of f(x), exists, and belongs to L^q ; and

$$\int_{-\infty}^{\infty} |F(x)|^q dx \leqslant K(q) \int_{-\infty}^{\infty} |f(x)|^q |x|^{q-2} dx.$$

(i) Consider the case q=4.

Suppose first that f(x) belongs to L^2 , and vanishes outside a finite interval. Then F(x) is L^2 and bounded, and $\sqrt{(2\pi)}\{F(x)\}^2$ is the transform of

$$\phi(x) = \int_{-\infty}^{\infty} f(y)f(x-y) dy,$$

which also belongs to L^2 . Hence

$$2\pi \int_{-\infty}^{\infty} |F(x)|^4 dx = \int_{-\infty}^{\infty} |\phi(x)|^2 dx.$$

Now $f(x) = |x|^{-\frac{1}{2}}g(x)$, where g(x) belongs to L^4 . Hence

$$\begin{aligned} \phi(x) &= \int_{-\infty}^{\infty} \frac{g(y)}{|y|^{\frac{1}{4}}} \frac{g(x-y)}{|x-y|^{\frac{1}{4}}} \, dy = \int_{-\infty}^{\infty} \frac{g(y)}{|y|^{\frac{1}{4}}} \frac{g(x-y)}{|x-y|^{\frac{1}{4}}} \frac{1}{|y|^{\frac{1}{4}}|x-y|^{\frac{1}{4}}} \, dy, \\ |\phi(x)|^2 &\leqslant \int_{-\infty}^{\infty} \frac{|g(y)|^2}{|y|^{\frac{1}{4}}} \frac{|g(x-y)|^2}{|x-y|^{\frac{1}{4}}} \, dy \int_{-\infty}^{\infty} \frac{dy}{|y|^{\frac{1}{4}}|x-y|^{\frac{1}{4}}} \\ &= \frac{A}{|x|^{\frac{1}{4}}} \int_{-\infty}^{\infty} \frac{|g(y)|^2}{|y|^{\frac{1}{4}}} \frac{|g(x-y)|^2}{|x-y|^{\frac{1}{4}}} \, dy. \end{aligned}$$

Hence

$$\begin{split} \int\limits_{-\infty}^{\infty} |\phi(x)|^2 \, dx &\leqslant A \int\limits_{-\infty}^{\infty} \int\limits_{-\infty}^{\infty} \frac{|g(y)|^2 |g(x-y)|^2}{|x|^{\frac{1}{8}}|y|^{\frac{1}{8}}|x-y|^{\frac{1}{8}}} \, dx dy \\ &= A \int\limits_{-\infty}^{\infty} \int\limits_{-\infty}^{\infty} \frac{|g(y)|^2 |y|^{\frac{1}{16}}}{|x|^{\frac{1}{8}}|x-y|^{\frac{1}{16}}} \frac{|g(x-y)|^2 |x-y|^{\frac{1}{16}}}{|x|^{\frac{1}{8}}|y|^{\frac{1}{16}}} \, dx dy \\ &< A \int\limits_{-\infty}^{\infty} \int\limits_{-\infty}^{\infty} \left\{ \frac{|g(y)|^4 |y|^{\frac{1}{8}}}{|x|^{\frac{1}{8}}|x-y|^{\frac{1}{8}}} + \frac{|g(x-y)|^4 |x-y|^{\frac{1}{8}}}{|x|^{\frac{1}{8}}|y|^{\frac{1}{8}}} \right\} \, dx dy \\ &= A \int\limits_{-\infty}^{\infty} |g(y)|^4 \, dy \int\limits_{-\infty}^{\infty} \frac{|y|^{\frac{1}{8}}}{|x|^{\frac{1}{8}}|x-y|^{\frac{1}{8}}} \, dx = A \int\limits_{-\infty}^{\infty} |g(y)|^4 \, dy. \end{split}$$

Hence

$$\int\limits_{-\infty}^{\infty}|F(x)|^4\,dx < A\int\limits_{-\infty}^{\infty}|f(x)|^4\,x^2\,dx.$$

The proof now follows the usual lines. Let f(x) be any function such that $\{f(x)\}^4x^2$ is L. Approximating to f(x) over (a,b) by a sequence of functions of the special type, we prove as in § 4.3 that

$$\int\limits_{-\infty}^{\infty}|F(x,a)\!-\!F(x,b)|^4\,dx < A\Big(\int\limits_{-b}^{-a}+\int\limits_{a}^{b}\Big)|f(x)|^4x^2\,dx.$$

Hence F(x,a) converges in mean with exponent 4, and the theorem follows in the usual way.

(ii) It is possible to prove the theorem when q is any even integer by an extension of the above method, but, as in the Young-Hausdorff theorem, the other values remain to be filled in.

The simplest procedure is to begin by proving the corresponding result for series, and we shall quote this from Zygmund.[†] The case we require is that if f(x) has the Fourier coefficients c_m , then

$$\int_{-\pi}^{\pi} |f(x)|^q dx \leqslant K(q) \sum_{-\infty}^{\infty} |c_m|^q (|m|+1)^{q-2}.$$

Defining a_{ν} and $\Phi_{n}(x)$ as before, it follows that

$$\int_{-\pi\lambda}^{\pi\lambda} |\Phi_n(x)|^q dx = \lambda \int_{-\pi}^{\pi} \left| \sum_{\nu=-n}^n a_{\nu} e^{i\nu x} \right|^q dx$$

$$\leq \lambda K(q) \sum_{\nu=-n}^n |a_{\nu}|^q (|\nu|+1)^{q-2}.$$

$$\begin{array}{c} \text{If } \nu \geqslant 1, \\ |a_{\nu}|^q \leqslant \frac{1}{\lambda^{q-1}} \int\limits_{v(\lambda)}^{(\nu+1)/\lambda} |f(x)|^q \, dx \leqslant \frac{1}{\lambda \nu^{q-2}} \int\limits_{v(\lambda)}^{(\nu+1)/\lambda} |f(x)|^q x^{q-2} \, dx. \end{array}$$

There is a similar inequality for $\nu \leqslant -2$; and

$$|a_0| = \Big|\int\limits_0^{1/\lambda} f(x) x^{1-2/q} x^{2/q-1} dx\Big|$$

$$\leq \Big(\int\limits_0^{1/\lambda} |f(x)|^q x^{q-2} dx\Big)^{1/q} \Big(\int\limits_0^{1/\lambda} x^{(2/q-1) \, q/(q-1)} dx\Big)^{1-1/q} = o\,(\lambda^{-1/q}),$$

and similarly for a_{-1} . Hence, making $\lambda \to \infty$, we obtain

$$\int_{-\infty}^{\infty} |F(x,b) - F(x,a)|^q \, dx < K(q) \left(\int_{-b}^{-a} + \int_{a}^{b} \right) |f(x)|^q |x|^{q-2} \, dx.$$

The theorem now follows as in previous cases.

† Zygmund, Trigonometrical Series, § 9.4.

4.9. Theorem 80.† If f(x) belongs to L^p (1 < p < 2), then

$$\int\limits_{-\infty}^{\infty}|F(x)|^p|x|^{p-2}\,dx\leqslant K(p)\int\limits_{-\infty}^{\infty}|f(x)|^p\,dx.$$

Let g(x) be a function of $L^{p'}$, vanishing outside a finite interval (a, b), where a > 0. Then it also belongs to L^{p} . Hence, by Theorem 75,

$$\int_{-\infty}^{\infty} F(x)g(x) \ dx = \int_{-\infty}^{\infty} f(x)G(x) \ dx.$$

Also, by Theorem 79, with q = p',

$$\int\limits_{-\infty}^{\infty}|G(x)|^{p'}\,dx\leqslant K(p)\int\limits_{-\infty}^{\infty}|g(x)|^{p'}|x|^{p'-2}\,dx.$$

Hence

$$igg|\int\limits_{-\infty}^{\infty}F(x)g(x)\;dxigg|\leqslantiggl(\int\limits_{-\infty}^{\infty}|f(x)|^p\;dxiggr)^{1/p}iggl(\int\limits_{-\infty}^{\infty}|G(x)|^{p'}\;dxiggr)^{1/p'}\ \leqslant K(p)iggl(\int\limits_{-\infty}^{\infty}|f(x)|^p\;dxiggr)^{1/p}iggl(\int\limits_{-\infty}^{\infty}|g(x)|^{p'}|x|^{p'-2}\;dxiggr)^{1/p'}$$

Let $g(x) = |F(x)|^{p-1} \operatorname{sgn} \overline{F}(x) |x|^{p-2} \quad (a \leqslant x \leqslant b).$

Then

$$\int_a^b |F(x)|^p x^{p-2} dx \leqslant K(p) \left(\int_{-\infty}^\infty |f(x)|^p dx \right)^{1/p} \left(\int_a^b |F(x)|^p x^{p-2} dx \right)^{1/p'},$$

and hence $\int_a^b |F(x)|^p x^{p-2} dx \leqslant K(p) \int_{-\infty}^\infty |f(x)|^p dx.$

Making $a \to 0$, $b \to \infty$, we obtain the desired result for the integral over $(0,\infty)$; similarly for the integral over $(-\infty,0)$.

4.10. Another case of the Parseval formula.

THEOREM 81. Let f(x) be L^p , and let $|G(x)|^{p'}|x|^{p'-2}$ be L, where 1 . Then if <math>F and g are the transforms of f and G, (2.1.1) holds.

The proof is similar to that of Theorem 75, but now

$$\int_{-b}^{b} \{F(x,a) - F(x)\}G(x) \ dx = \int_{-b}^{b} \{F(x,a) - F(x)\}|x|^{(p-2)/p} \cdot G(x)|x|^{1-2/p'} \ dx$$

tends to 0 because, by Theorem 80, $F(x,a)|x|^{(p-2)/p}$ converges in mean

† Hardy and Littlewood (1).

to $F(x)|x|^{(p-2)/p}$, with exponent p. The proof concludes as before, but is justified by Theorem 79 instead of Theorem 74.

4.11. Failure of Theorems 75 and 79 for p > 2. That the Young-Hausdorff theorem fails for p > 2 follows incidentally from Theorem 80. For, if f(x) belongs to L^p , not only is $|F(x)|^{p'}$ integrable, but so is $|F(x)|^p|x|^{p-2}$; and hence so is

$$|F(x)|^r |x|^{r/p'-1} \quad (p \leqslant r \leqslant p').$$
 (4.11.1)

Call the class of functions with this property $L_1^{p'}$, so that $L_1^{p'}$ is a sub-set of $L^{p'}$

If f(x) belongs to L^q (q > 2), it does not necessarily belong to L_1^q , and is therefore not necessarily the transform of a function of L^q .

However, we can show by means of examples that even if f(x) belongs to L_1^q (q > 2), f(x) is not necessarily the transform of a function of the class $L^{q'}$. Presumably no condition which merely states the existence of an integral involving |f(x)| is a sufficient condition for f(x) to be the transform of a function of $L^{q'}$.

Consider the function \dagger (0 < a < 1, a < b)

$$f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{1} t^{-a-1} \cos t^{-b} \cos xt \, dt$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{1} \cos\left(xt + \frac{1}{t^{b}}\right) \frac{dt}{t^{a+1}} + \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{1} \cos\left(xt - \frac{1}{t^{b}}\right) \frac{dt}{t^{a+1}}$$

$$= \frac{1}{\sqrt{(2\pi)}} \{\phi(x) + \psi(x)\}. \qquad (4.11.2)$$
Let
$$\phi(x) = \int_{-\infty}^{(b-\xi)^{1/(b+1)}} \frac{\binom{b}{x} + \xi^{1/(b+1)}}{\binom{b}{x} - \xi^{1/(b+1)}} + \int_{\binom{b}{x} - \xi^{1/(b+1)}}^{1} = \phi_{1} + \phi_{2} + \phi_{3},$$

where $\xi = o(1/x)$ as $x \to \infty$. Then

$$\phi_1 = \int rac{d\sin(xt+t^{-b})}{xt^{a+1}-bt^{a-b}},$$

and here $(bt^{a-b}-xt^{a+1})^{-1}$ is positive, steadily increasing, and less than

$$\frac{t^{b-a}}{x\left(\frac{b}{x}-t^{b+1}\right)}\leqslant \frac{(b/x)^{\frac{b-a}{b+1}}}{x\xi}.$$

† Titchmarsh (2).

Hence, by the second mean-value theorem,

$$\phi_1 = O(\xi^{-1} x^{\frac{a-2b-1}{b+1}}).$$

In ϕ_3 , $(xt^{a+1}-bt^{a-b})^{-1}$ is positive and steadily decreasing, and we obtain the same result as for ϕ_1 . Finally

$$|\phi_2| \leqslant \frac{\left(\frac{b}{x} + \xi\right)^{1/(b+1)} - \left(\frac{b}{x} - \xi\right)^{1/(b+1)}}{\left(\frac{b}{x} - \xi\right)^{b+1}} = O\left(\frac{x^{-1/(b+1)}x\xi}{x^{-\frac{a+1}{b+1}}}\right) = O\left(\xi x^{\frac{a+b+1}{b+1}}\right).$$

Taking $\xi = x^{-\frac{3b+2}{2b+2}}$, it follows that, as $x \to \infty$,

$$\phi(x) = O\left(x^{\frac{2a-b}{2b+2}}\right).$$

Again,

$$\psi(x) = \int \frac{d\sin(xt-t^{-b})}{xt^{a+1}+bt^{a-b}}.$$

Now $(xt^{a+1}+bt^{a-b})^{-1}$ increases steadily from 0 to a maximum of the form Kx^{b+1} , where K depends on a and b only, and then decreases steadily. Hence the second mean-value theorem gives

$$\psi(x) = O\left(x^{\frac{a-b}{b+1}}\right).$$

$$f(x) = O\left(x^{\frac{2a-b}{2b+2}}\right),$$

Hence as $x \to \infty$

and plainly f(x) = O(1) as $x \to 0$. Hence, if q is a given number greater than 2, f(x) belongs to L_1^q if b is large enough.

If f(x) were the transform of a function F(x) of $L^{q'}$, we should have

$$F(x) = \sqrt{\left(\frac{2}{\pi}\right)} \frac{d}{dx} \int_{0}^{\infty} \frac{\sin xu}{u} f(u) du.$$

If we can insert (4.11.2) for f(u) and invert the order of integration, we obtain

$$F(x) = x^{-a-1}\cos x^{-b}$$
 (0 < x < 1), 0 (x > 1),

which does not belong to L^r for any $r \geqslant 1$. This gives the desired result.

The inversion is justified if we may invert

$$\int_{0}^{\infty} \frac{\sin xu}{u} du \int_{0}^{1} t^{-a-1} \cos t^{-b} (1-\cos ut) dt,$$

and we may if

$$\lim_{\lambda \to \infty} \int_0^1 t^{-a-1} \cos t^{-b} dt \int_{\lambda}^{+\infty} \frac{\sin x u (1 - \cos u t)}{u} du = 0.$$

Now

$$\left|\int_{\lambda}^{+\infty} \frac{\sin x u (1 - \cos u t)}{u} du\right| = \frac{1}{2} \left|\int_{\lambda x}^{\lambda (x+t)} \frac{\sin v}{v} dv - \int_{\lambda |x-t|}^{\lambda x} \frac{\sin v}{v} dv\right|$$

$$\leq \frac{1}{2} \log \left|\frac{x+t}{x-t}\right|,$$

and the result follows from dominated convergence.

4.12. Special conditions. In this section we give two sufficient conditions of special kinds for f(x) to be the transform of a function of L^p (1 < p < 2).

THEOREM 82.† Let f(x) be even, positive non-increasing for x > 0, $f(\infty) = 0$, and let $\{f(x)\}^p x^{p-2}$ $(1 belong to <math>L(0,\infty)$. Then F(x) belongs to L^p .

Since f(x) in non-increasing, and $f(\infty) = 0$, the integral

$$F(x) = F_c(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^{\infty} f(y) \cos xy \ dy$$

converges for every x > 0. Let

$$F(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{1/x} f(y) \cos xy \, dy + \sqrt{\left(\frac{2}{\pi}\right)} \int_{1/x}^{\infty} f(y) \cos xy \, dy$$

= $F_1(x) + F_2(x)$.

By the second mean-value theorem

$$F_2(x) = \sqrt{\left(\frac{2}{\pi}\right)} f\left(\frac{1}{x}\right) \int_{1/x}^{\xi} \cos xy \, dy = \sqrt{\left(\frac{2}{\pi}\right)} f\left(\frac{1}{x}\right) \frac{\sin x\xi - \sin 1}{x}.$$

Hence

$$|F_2(x)| \leqslant 2\sqrt{\left(\frac{2}{\pi}\right)}\frac{1}{x}f\left(\frac{1}{x}\right),$$

 $\int\limits_{-\infty}^{\infty}|F_2(x)|^p\ dx < A\int\limits_{-\infty}^{\infty}\left\{\frac{1}{x}f\left(\frac{1}{x}\right)\right\}^p\ dx = A\int\limits_{-\infty}^{\infty}\{f(t)\}^pt^{p-2}\ dt,$

† See Hardy and Littlewood (3).

which is finite by hypothesis. Also

$$|F_1(x)| \leqslant \sqrt{\left(\frac{2}{\pi}\right)} \int_0^{1/x} f(y) dy,$$

and we have to prove that this is L^p ; or, what is the same thing, that

$$x^{-2/p}\int\limits_0^x f(y)\ dy$$

is L^p . We are given that $f(x) = g(x)x^{2|p-1}$, where g(x) is L^p . Hence

$$x^{-2|p}\int_{0}^{x}f(y)\ dy=x^{-2|p}\int_{0}^{x}g(y)y^{2|p-1}\ dy\leqslant x^{-1}\int_{0}^{x}g(y)\ dy,$$

which is L^p , by a theorem of Hardy.† This proves the theorem.

Incidentally it must follow from our hypothesis that f(x) belongs to $L^{p'}$; and in fact

$$K > \int\limits_{rac{1}{2}}^{x} f^p(t) t^{p-2} \ dt \geqslant f^p(x) (rac{1}{2}x)^{p-2} rac{1}{2}x, \ f(x) < K x^{-(p-1)/p}, \ |f(x)|^{p'} < K |f(x)|^p x^{p-2}.$$

THEOREM 83. Let f(x) be the integral of order (2-p)/p of a function $\phi(x)$ of L^p . Then $F_c(x)$ exists and belongs to L^p .

Let
$$\Phi_a(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^a \phi(t) \sin(xt + \pi/p) dt.$$

Then, by Theorem 80, $x^{1-2/p}\Phi_a(x)$ converges in mean (p) to g(x) say. Let $G_c(x)$ be the cosine transform of g(x). Then $G_c(x)$ belongs to $L^{p'}$.

$$\int_{0}^{y} G_{c}(x) dx = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \frac{\sin xy}{x} g(x) dx$$

$$= \lim_{a \to \infty} \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \frac{\sin xy}{x} x^{1-2/p} \Phi_{a}(x) dx$$

$$= \lim_{a \to \infty} \frac{2}{\pi} \int_{0}^{\infty} \frac{\sin xy}{x^{2/p}} dx \int_{0}^{a} \phi(t) \sin(xt + \pi/p) dt$$

$$= \lim_{a \to \infty} \frac{2}{\pi} \int_{0}^{a} \phi(t) dt \int_{0}^{\infty} \frac{\sin xy \sin(xt + \pi/p)}{x^{2/p}} dx,$$

the inversion being justified by uniform convergence.

† See Titchmarsh, Theory of Functions, p. 396.

The inner integral is

$$\frac{\pi}{2\Gamma(2/p)} (y-t)^{2/p-1} \quad (t < y), \qquad 0 \quad (t > y).$$

Hence

$$\int_{0}^{y} G_{c}(x) dx = \frac{1}{\Gamma(2/p)} \int_{0}^{y} (y-t)^{2/p-1} \phi(t) dt = \int_{0}^{y} f(x) dx,$$

by hypothesis. Hence $G_c(x) = f(x)$ almost everywhere, and $g(x) = F_c(x)$ belongs to L^p .

4.13. Lipschitz conditions. In this section we shall give a condition of quite a different kind for a function to have a transform belonging to certain L-classes. The analysis originated with theorems of Bernstein and Szasz on Fourier series.†

THEOREM 84. Let f(x) belong to L^p (1 < $p \le 2$), and let

$$\int_{-\infty}^{\infty} |f(x+h) - f(x-h)|^p \, dx = O(h^{\alpha p}) \quad (0 < \alpha \leqslant 1) \quad (4.13.1)$$

as $h \to 0$. Then F(x) belongs to L^{β} for

$$\frac{p}{p+\alpha p-1}<\beta\leqslant\frac{p}{p-1}.$$

For a fixed h the transform of f(x+h) is $e^{-ixh}F(x)$. Hence the transform of f(x+h)-f(x-h), as a function of x, is $-2i\sin xhF(x)$. Hence

$$\int_{-\infty}^{\infty} |2\sin xhF(x)|^{p'} dx < K(p) \left\{ \int_{-\infty}^{\infty} |f(x+h)-f(x-h)|^{p} dx \right\}^{1/(p-1)}$$

$$< K(p)h^{\alpha p'}.$$

Since $|\sin xh| > Axh$ for $x \le 1/h$, the left-hand side is greater than

$$A\int_{0}^{1/h}x^{p'}h^{p'}|F(x)|^{p'}dx.$$

$$\int_{0}^{1/h}x^{p'}|F(x)|^{p'}dx = O(h^{(\alpha-1)p'}).$$

$$\phi(\xi) = \int_{0}^{\xi}|xF(x)|^{\beta}dx.$$

Hence

Let

† See Titchmarsh (12).

Then, if $\beta < p'$,

$$\phi(\xi) \leqslant \left(\int\limits_1^{\xi} |xF(x)|^{p'} dx\right)^{\beta/p'} \left(\int\limits_1^{\xi} dx\right)^{1-\beta/p'} = O(\xi^{1-\alpha\beta+\beta/p}).$$

Hence

$$\int_{1}^{\xi} |F(x)|^{\beta} dx = \int_{1}^{\xi} x^{-\beta} \phi'(x) dx = \xi^{-\beta} \phi(\xi) + \beta \int_{1}^{\xi} x^{-\beta-1} \phi(x) dx$$

$$= O(\xi^{1-\beta-\alpha\beta+\beta|p}) + O\left(\int_{1}^{\xi} x^{-\beta-\alpha\beta+\beta|p} dx\right) = O(\xi^{1-\beta-\alpha\beta+\beta|p}),$$

and this is bounded as $\xi \to \infty$ if $1-\beta-\alpha\beta+\beta/p \leqslant 0$, i.e. if

$$\beta > \frac{p}{p+\alpha p-1}$$
.

Similarly for the integral over $(-\xi, -1)$. This proves the theorem.

A particular case, corresponding to the original theorem of Bernstein, is that if the condition is satisfied with $\alpha > 1/p$, then F(x) belongs to $L(0,\infty)$, so that the Fourier integral

$$\int_{-\infty}^{\infty} F(t)e^{ixt} dt$$

is absolutely convergent for all values of x.

To show that the range for β in the above theorem cannot be extended, consider the even function

$$f(x) = \frac{1}{x^a + x} \quad (x > 0),$$

where 0 < a < 1/p. For x > 2h

$$|f(x+h)-f(x-h)| = 2h|f'(x+\theta h)| \quad (-1 < \theta < 1)$$

 $\leq 2h|f'(x-h)| \leq 2h|f'(\frac{1}{2}x)|,$

since |f'(x)| is positive and steadily decreasing. Hence

$$\int_{2h}^{\infty} |f(x+h)-f(x-h)|^p dx = O\left\{h^p \int_{2h}^{\infty} |f'(\frac{1}{2}x)|^p dx\right\}$$

$$= O\left\{h^p \left(\int_{2h}^{1} x^{-(a+1)p} dx + \int_{1}^{\infty} x^{-2} dx\right)\right\} = O(h^{1-ap}).$$

Also

$$\int_{0}^{2h} |f(x+h)-f(x-h)|^{p} dx = O\left(\int_{0}^{2h} \frac{dx}{|x-h|^{ap}}\right) = O(h^{1-ap}).$$

The conditions of the theorem are therefore satisfied with $\alpha = 1/p - a$. Hence F(x) belongs to L^{β} for $\beta > 1/(1-a)$. But $\uparrow F(x) \sim Kx^{a-1}$ as $x \to \infty$, so that F(x) does not belong to $L^{1/(1-a)}$.

In the case $\alpha < 1$, p = 2 it is possible to put the theorem into a form in which it is reversible.

THEOREM 85. If f(x) belongs to L^2 , the conditions

$$\int_{-\infty}^{\infty} |f(x+h) - f(x-h)|^2 dx = O(|h|^{2\alpha}) \quad (0 < \alpha < 1), \quad (4.13.2)$$

$$\left(\int_{-\infty}^{-X} + \int_{X}^{\infty} (F(x))^2 dx = O(X^{-2\alpha}) \quad (X \to \infty)$$
(4.13.3)

are equivalent.

Instead of an inequality we now obtain

$$\int_{-\infty}^{\infty} 4\sin^2xh |F(x)|^2 dx = \int_{-\infty}^{\infty} |f(x+h) - f(x-h)|^2 dx.$$
(4.13.4)

Suppose that (4.13.2) holds. Then (4.13.4) gives

$$\int_{1/(2h)}^{1/h} |F(x)|^2 dx < A \int_{0}^{\infty} \sin^2 x h |F(x)|^2 dx = O(h^{2\alpha}).$$

Hence

$$\int_{X}^{\infty} \{F(x)\}^{2} dx = \int_{X}^{2X} + \int_{2X}^{4X} + \dots = O\{X^{-2\alpha} + (2X)^{-2\alpha} + \dots\} = O(X^{-2\alpha}),$$
 and similarly for $(-\infty, -X)$.

On the other hand, if (4.13.3) holds, then writing

$$\phi(X) = \int\limits_X^\infty \{F(x)\}^2 \, dx,$$

$$\int\limits_0^X x^2 \{F(x)\}^2 \, dx = \int\limits_0^X -x^2 \phi'(x) \, dx = -X^2 \phi(X) + 2 \int\limits_0^X x \phi(x) \, dx$$

$$\leqslant 2 \int\limits_0^X O(x^{1-2\alpha}) \, dx = O(X^{2-2\alpha}).$$

Hence

$$\int_{-\infty}^{\infty} \sin^2 x h \{F(x)\}^2 dx = O\left(h^2 \int_{-1/h}^{1/h} \{F(x)\}^2 dx\right) + O\left(\int_{-\infty}^{-1/h} + \int_{1/h}^{\infty} \{F(x)\}^2 dx\right)$$
$$= O(h^{2\alpha}),$$

and (4.13.2) now follows from (4.13.4).

† See e.g. Theorems 126-7 below.

Since, if $\beta < 2$,

$$\int_{X}^{2X} |F(x)|^{\beta} dx \leqslant \left(\int_{X}^{2X} \{F(x)\}^{2} dx\right)^{\frac{1}{2}\beta} \left(\int_{X}^{2X} dx\right)^{1-\frac{1}{2}\beta}$$

$$= O(X^{-\alpha\beta})O(X^{1-\frac{1}{2}\beta}) = O(X^{1-\alpha\beta-\frac{1}{2}\beta}),$$

it follows again that F(x) belongs to L^{β} if $\beta > 1/(\alpha + \frac{1}{2})$, the case p = 2 of the above theorem. But this last step is, of course, not reversible.

4.14. Mellin transforms of the class L^p . Let us denote by \mathfrak{L}^p the class of functions f(x) such that

$$\int_{0}^{\infty} |f(x)|^{p} \, \frac{dx}{x} < \infty.$$

Then we have

THEOREM 86. If $\mathfrak{F}(k+it)$ belongs to L^p (1 , then its Mellin transform <math>f(x) exists, and $x^k f(x)$ belongs to \mathfrak{L}^p .

If $x^k f(x)$ belongs to \mathfrak{L}^p , then the Mellin transform $\mathfrak{F}(s)$ of f(x) exists, and $\mathfrak{F}(k+it)$ belongs to $L^{p'}$.

THEOREM 87. If $\mathfrak{F}(k+it)$ belongs to L^p , and $x^{1-k}g(x)$ to \mathfrak{L}^p , then

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s) \mathfrak{G}(1-s) \ ds = \int_{0}^{\infty} f(x) g(x) \ dx.$$

These are readily obtained by transformation from Theorems 74 and 75.

THEOREM 88. If $\mathfrak{F}(k+iv)$, $x^k f(x)$ are Mellin transforms of L^p , $\mathfrak{L}^{p'}$, and $x^{s-k}g(x)$, $\mathfrak{G}(s-k-iv)$ of $\mathfrak{L}^{p'}$, L^p , then (2.1.15) holds.

THEOREM 89. If $\mathfrak{F}(k+iv)$, $x^k f(x)$ are Mellin transforms of L^p , $\mathfrak{L}^{p'}$, and $\mathfrak{G}(s-k-iv)$, $x^{s-k}g(x)$ of L^q , $\mathfrak{L}^{q'}$, then (2.1.16) are Mellin transforms of $\mathfrak{L}^{p'}$, L^p .

Note. It has recently been proved by Zygmund (2) that if f(x) is L^p , 1 , then (3.11.2) holds almost everywhere, no logarithmic factor being required. If <math>f(x) satisfies the condition of Theorem 79, then $f(x)\log x$ is $L^2(1,\infty)$ (apply Hölder's inequality to the integral over $(2^n, 2^{n+1})$). Hence (3.11.2) holds almost everywhere by Theorem 62.

CONJUGATE INTEGRALS: HILBERT TRANSFORMS

5.1. Conjugate integrals. Fourier's integral formula may be written in the form

$$f(x) = \int_{0}^{\infty} \left\{ a(t)\cos xt + b(t)\sin xt \right\} dt, \qquad (5.1.1)$$

where

$$a(t) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(u)\cos ut \, du, \qquad b(t) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(u)\sin ut \, du.$$
(5.1.2)

The integral in (5.1.1) is, formally, the limit as $y \to 0$ of

$$\int_{0}^{\infty} \{a(t)\cos xt + b(t)\sin xt\}e^{-yt} dt = U(x, y)$$
 (5.1.3)

say; and this is the real part of

$$\int_{0}^{\infty} \{a(t) - ib(t)\} e^{izt} dt = \Phi(z)$$
 (5.1.4)

say, where z = x + iy.

The imaginary part of $\Phi(z)$ is

$$-\int_{0}^{\infty} \{b(t)\cos xt - a(t)\sin xt\}e^{-\nu t} dt = V(x,y)$$
 (5.1.5)

say. Writing -V(x,0) = g(x), we obtain

$$g(x) = \int_{0}^{\infty} \left\{ b(t)\cos xt - a(t)\sin xt \right\} dt \qquad (5.1.6)$$

$$=\frac{1}{\pi}\int_{0}^{\infty}dt\int_{-\infty}^{\infty}\sin(u-x)tf(u)\ du. \tag{5.1.7}$$

The integral (5.1.7) is called the *allied integral* of Fourier's integral. It is obtained formally from (5.1.1) by replacing a by b and b by -a.

Repeating the process, we return to minus the original integral. The relation between f(x) and g(x) is thus skew-reciprocal, i.e. reciprocal apart from a minus sign.

Again, we have formally

$$a(t) = \frac{1}{\sqrt{(2\pi)}} \{ F(t) + F(-t) \}, \qquad b(t) = \frac{1}{i\sqrt{(2\pi)}} \{ F(t) - F(-t) \}.$$

Hence
$$g(x) = \frac{1}{i\sqrt{(2\pi)}} \int_{0}^{\infty} \{F(t) - F(-t)\}\cos xt \, dt - \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \{F(t) + F(-t)\}\sin xt \, dt$$

$$= \frac{1}{i\sqrt{(2\pi)}} \left\{ \int_{0}^{\infty} F(t)e^{-ixt} \, dt - \int_{0}^{\infty} F(-t)e^{ixt} \, dt \right\}$$

$$= \frac{1}{i\sqrt{(2\pi)}} \int_{-\infty}^{\infty} F(t)\operatorname{sgn} t \, e^{-ixt} \, dt.$$
Thus
$$G(t) = -iF(t)\operatorname{sgn} t. \tag{5.1.8}$$

If f(x) is even, b(t) = 0, and g(x) is minus the sine transform of the cosine transform of f(x); similarly, if f(x) is odd, g(x) is the cosine transform of the sine transform of f(x).

Again, we have formally

$$\begin{split} g(x) &= \lim_{\lambda \to \infty} \frac{1}{\pi} \int_{0}^{\lambda} dt \int_{-\infty}^{\infty} \sin(u - x) t f(u) \ du \\ &= \lim_{\lambda \to \infty} \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{1 - \cos \lambda (u - x)}{u - x} f(u) \ du \\ &= \lim_{\lambda \to \infty} \frac{1}{\pi} \int_{0}^{\infty} \frac{1 - \cos \lambda t}{t} \{ f(x + t) - f(x - t) \} \ dt. \end{split}$$

If f(x) is a sufficiently regular function, the part involving $\cos \lambda t$ will tend to 0 as $\lambda \to \infty$, and we shall have

$$g(x) = \frac{1}{\pi} \int_{0}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt;$$
 (5.1.9)

and similarly
$$f(x) = -\frac{1}{\pi} \int_{0}^{\infty} \frac{g(x+t) - g(x-t)}{t} dt.$$
 (5.1.10)

The reciprocity expressed by (5.1.9), (5.1.10) was first noticed by Hilbert, and the two functions so connected are called *Hilbert transforms*.

Equivalent formulae are

$$g(x) = \frac{1}{\pi} P \int_{-\infty}^{\infty} \frac{f(t)}{t - x} dt, \qquad f(x) = -\frac{1}{\pi} P \int_{-\infty}^{\infty} \frac{g(t)}{t - x} dt,$$
 (5.1.11)

where P denotes a principal value at t = x.

Simple pairs of conjugate functions f(x), g(x), are

1 (0 < x < a), 0 elsewhere,
$$\frac{1}{\pi} \log \left| \frac{a+x}{a-x} \right|,$$

$$\frac{1}{1+x^2}, \qquad \frac{x}{1+x^2},$$

$$\cos x, \qquad -\sin x,$$

and any number of such examples can be written down by starting with a suitable analytic function $\Phi(z)$. Examples from Chapter VII are $|x|^{-\nu}J_{\nu}(|x|)$, $-\operatorname{sgn} x|x|^{-\nu}\mathbf{H}_{\nu}(|x|)$

from (7.1.11) and (7.2.8);

$$\operatorname{sgn} x |x|^{\nu} J_{\nu}(|x|), \qquad -|x|^{\nu} Y_{\nu}(|x|)$$

from (7.11.2) and (7.11.3); and

$$J_0(2\sqrt{|x|}), \qquad -\operatorname{sgn} x\{(2/\pi)K_0(2\sqrt{|x|}) + Y_0(2\sqrt{|x|})\}$$

from (7.11.2), with $\nu = 0$ and $x = \frac{1}{2}(u/a + a/u)$, and (7.12.8).

5.2. Conditions which would justify the above formalities directly would be extremely complicated. Actually the simplest rigorous argument gives the reciprocity in a slightly different form.

Theorem 90.† Let f(x) belong to $L^2(-\infty,\infty)$. Then the formula

$$g(x) = -\frac{1}{\pi} \frac{d}{dx} \int_{-\infty}^{\infty} f(t) \log \left| 1 - \frac{x}{t} \right| dt$$
 (5.2.1)

defines almost everywhere a function g(x), also belonging to $L^2(-\infty,\infty)$. The reciprocal formula

$$f(x) = \frac{1}{\pi} \frac{d}{dx} \int_{-\infty}^{\infty} g(t) \log \left| 1 - \frac{x}{t} \right| dt$$
 (5.2.2)

also holds almost everywhere; and

$$\int_{-\infty}^{\infty} \{f(x)\}^2 dx = \int_{-\infty}^{\infty} \{g(x)\}^2 dx.$$
 (5.2.3)

If we could perform the differentiations under the integral signs, we should obtain the reciprocity in the form already given. We shall see later that this is possible; but we begin with the form to which the theory of Fourier transforms leads directly.

Let F(x) be the Fourier transform of f(x), $G(x) = -iF(x)\operatorname{sgn} x$, and g(x) the transform of G(x). Then

$$\int_{-\infty}^{\infty} |g(x)|^2 dx = \int_{-\infty}^{\infty} |G(x)|^2 dx = \int_{-\infty}^{\infty} |F(x)|^2 dx = \int_{-\infty}^{\infty} |f(x)|^2 dx.$$
† Titchmarsh (5).

Also
$$g(x) = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} G(y) \frac{e^{-ixy}-1}{-iy} dy = \frac{1}{\sqrt{(2\pi)}} \frac{d}{dx} \int_{-\infty}^{\infty} F(y) \frac{e^{-ixy}-1}{|y|} dy.$$

The transform of $H(y) = (e^{-ixy}-1)/|y|$ is

$$\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \frac{e^{-ixy} - 1}{|y|} e^{-iuy} dy = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \frac{\cos(x+u)y - \cos uy}{y} dy$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \lim_{\delta \to 0} \int_{\delta}^{\infty} \frac{\cos(x+u)y - \cos uy}{y} dy$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \lim_{\delta \to 0} \left(\int_{\delta|x+u|}^{\infty} \frac{\cos v}{v} dv - \int_{\delta|u|}^{\infty} \frac{\cos v}{v} dv\right)$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \lim_{\delta \to 0} \int_{\delta|x+u|}^{\delta|u|} \frac{\cos v}{v} dv$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \lim_{\delta \to 0} \int_{\delta|x+u|}^{\delta|u|} \frac{dv}{v} = \sqrt{\left(\frac{2}{\pi}\right)} \log\left|\frac{u}{x+u}\right|.$$

Hence Parseval's formula gives

$$\int_{-\infty}^{\infty} F(y) \frac{e^{-ixy}-1}{|y|} dy = \frac{1}{\pi} \int_{-\infty}^{\infty} f(u) \log \left| \frac{u}{x-u} \right| du,$$

and (5.2.1) follows. The relation between F and G, and so between f and g, is skew-reciprocal, so that (5.2.2) also follows.

5.3. Theorem 91.† Let f(x) belong to $L^2(-\infty,\infty)$. Then the formula

$$g(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt$$
 (5.3.1)

defines almost everywhere a function g(x), also of $L^2(-\infty,\infty)$. The reciprocal formula

$$f(x) = -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{g(x+t) - g(x-t)}{t} dt$$
 (5.3.2)

† The analogue for series is due to Plessner (1). See also Hardy (14).

also holds almost everywhere; and

$$\int_{-\infty}^{\infty} \{f(x)\}^2 dx = \int_{-\infty}^{\infty} \{g(x)\}^2 dx.$$
 (5.3.3)

The functions g(x) of Theorems 90 and 91 are equivalent.

The integrals (5.1.2) defining a(t) and b(t) exist in the mean-square sense, and

 $a(t)-ib(t) = \sqrt{\left(\frac{2}{\pi}\right)}F(-t).$

Let $H(t) = e^{ist}$ (t > 0), 0 (t < 0). Then

$$h(u) = \frac{1}{\sqrt{(2\pi)}} \int_0^\infty e^{izt-izu} du = \frac{1}{i\sqrt{(2\pi)(u-z)}}.$$

Hence Parseval's formula, in the form

$$\int_{-\infty}^{\infty} F(-t)H(t) dt = \int_{-\infty}^{\infty} f(t)h(t) dt,$$

applied to (5.1.4) gives

$$\Phi(z) = \frac{1}{i\pi} \int_{-\infty}^{\infty} \frac{f(t)}{t-z} dt \quad (I(z) > 0).$$
 (5.3.4)

Taking real and imaginary parts separately, we obtain

$$U(x,y) = \frac{y}{\pi} \int_{-\infty}^{\infty} \frac{f(t)}{(t-x)^2 + y^2} dt,$$
 (5.3.5)

$$V(x,y) = -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{t-x}{(t-x)^2 + y^2} f(t) dt.$$
 (5.3.6)

Define g and G as in § 5.2, the integrals being now mean-square. Then we also have

$$\Phi(z) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} F(-t) e^{izt} dt = -i \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} G(-t) e^{izt} dt$$

$$= -i \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} G(-t) H(t) dt = -i \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} g(t) h(t) dt$$

$$= -\frac{1}{\pi} \int_{0}^{\infty} \frac{g(t)}{t-z} dt. \tag{5.3.7}$$

Hence

$$U(x,y) = -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{t-x}{(t-x)^2 + y^2} g(t) dt, \qquad (5.3.8)$$

$$V(x,y) = -\frac{y}{\pi} \int_{-\infty}^{\infty} \frac{g(t)}{(t-x)^2 + y^2} dt.$$
 (5.3.9)

By the theory of Cauchy's singular integral, § 1.17, $U(x,y) \to f(x)$ as $y \to 0$ for almost all values of x, and $V(x,y) \to -g(x)$ for almost all values of x. We now use the following theorem.

THEOREM 92. Let f(x) be any function such that f(x) belongs to L(0,1), and $x^{-1}f(x)$ to $L(1,\infty)$. Let V(x,y) be defined by (5.3.6). Then

$$\lim_{y \to 0} \left\{ V(x,y) + \frac{1}{\pi} \int_{t}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt \right\} = 0$$
 (5.3.10)

for almost all values of x.

We know that

$$\omega(y) = \int_{0}^{y} |f(x+t) - f(x-t)| dt = o(y)$$

for almost all values of x. Let x be a point where this holds. We have

$$\begin{split} V(x,y) + \frac{1}{\pi} \int_{v}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt \\ &= -\frac{1}{\pi} \int_{0}^{y} \frac{t}{t^2 + y^2} \{f(x+t) - f(x-t)\} dt + \\ &+ \frac{y^2}{\pi} \int_{v}^{1} \frac{f(x+t) - f(x-t)}{(t^2 + y^2)t} dt + \frac{y^2}{\pi} \int_{1}^{\infty} \frac{f(x+t) - f(x-t)}{(t^2 + y^2)t} dt \\ &= J_1 + J_2 + J_3 \end{split}$$

$$\begin{aligned} \text{say. As } y &\to 0, \\ |J_1| &\leqslant \frac{1}{2\pi y} \int\limits_0^y |f(x+t) - f(x-t)| \ dt = o \, (1), \\ |J_2| &\leqslant \frac{y^2}{\pi} \int\limits_y^1 \frac{|f(x+t) - f(x-t)|}{(t^2 + y^2)t} \ dt \\ &= \frac{y^2}{\pi} \left[\frac{\omega(t)}{(t^2 + y^2)t} \right]_{tt}^1 + \frac{y^2}{\pi} \int\limits_1^1 \frac{3t^2 + y^2}{(t^2 + y^2)^2 t^2} \, \omega(t) \ dt \end{aligned}$$

$$\leq \frac{y^2}{\pi} \frac{\omega(1)}{1+y^2} + o\left(y^2 \int_{y}^{1} \frac{3t^2 + y^2}{(t^2 + y^2)^2 t} dt\right)$$

$$= O(y^2) + o\left(\int_{1}^{1/y} \frac{3u^2 + 1}{(u^2 + 1)^2 u} du\right) = o(1),$$

and plainly $J_3 = o(1)$. Hence the result.

Since $V(x,y) \to -g(x)$ almost everywhere, (5.3.1) follows from (5.3.10). The relation between f and g being skew-reciprocal, (5.3.2) also follows; and (5.3.3) holds as before.

5.4. In this section we shall show that the same set of formulae may be obtained from a different source. We can take an analytic function $\Phi(z)$ satisfying certain conditions as the original function.

Theorem 93. Let $\Phi(z)$ be an analytic function, regular for y > 0, and let

$$\int_{-\infty}^{\infty} |\Phi(x+iy)|^2 dx$$

exist for every positive y, and be bounded. Then, as $y \to 0$, $\Phi(x+iy)$ converges in mean to a function $\Phi(x)$, and also $\Phi(x+iy) \to \Phi(x)$ for almost all x. For y > 0

$$\Phi(z) = \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u)}{u-z} du \quad (u \text{ real}).$$

If $\Phi(z) = U(x,y) + iV(x,y)$, $\Phi(x) = f(x) - ig(x)$, the functions U, V, f, and g are connected by the formulae of the previous section, and in particular f and g are conjugate.

We first prove the following

LEMMA. Let $\Phi(z)$ be analytic, and let

$$\int_{-\infty}^{\infty} |\Phi(x+iy)|^p dx \quad (p>1)$$

exist and be bounded for $y_1\leqslant y\leqslant y_2$. Then, as $x\to\pm\infty$, $\Phi(x+iy)\to 0$, uniformly for $y_1+\delta\leqslant y\leqslant y_2-\delta$.

Let $y_1 + \delta \leqslant y \leqslant u_2 - \delta$. Then, if $0 < \rho \leqslant \delta$,

$$\Phi(z) = rac{1}{2\pi i}\int\limits_{|w-z|=
ho} rac{\Phi(w)}{w-z}\,dw = rac{1}{2\pi}\int\limits_0^{2\pi} \Phi(z+
ho e^{i\phi})\,d\phi.$$

Hence

is bounded for $y_1 \leqslant v \leqslant y_2$, and tends to 0 as $x \to \infty$, for every v. Hence the right-hand side tends to 0, and the result follows.

To prove the theorem, let

$$\phi_a(t,y) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-a}^{a} \Phi(z)e^{-itx} dx.$$

For each y this converges in mean, to $\phi(t,y)$ say, as $a \to \infty$. Consider, however, the integral $\int \Phi(z)e^{-itz} dz$

taken round the rectangle with corners at $\pm a+iy_1$, $\pm a+iy_2$, where $0 < y_1 < y_2$. The integral along the right-hand side is

$$\int_{y_1}^{y_2} \Phi(a+iy)e^{-it(a+iy)}i \ dy = ie^{-ita} \int_{y_1}^{y_2} \Phi(a+iy)e^{ty} \ dy,$$

and, by the lemma, this tends to 0 as $a \to \infty$, for fixed y_1 and y_2 . Similarly, the integral along the left-hand side tends to 0. Hence, as $a \to \infty$,

$$\int_{-a}^{a} \Phi(x+iy_1)e^{-it(x+iy_1)} dx - \int_{-a}^{a} \Phi(x+iy_2)e^{-tt(x+iy_2)} dx \to 0,$$
i.e.
$$e^{ty_1}\phi_n(t,y_1) - e^{ty_2}\phi_n(t,y_2) \to 0.$$

Hence the mean-square limit of this sequence over any finite interval is also 0, i.e. $e^{ty_1}\phi(t,y_1) = e^{ty_2}\phi(t,y_2)$

for almost all t. We may therefore write

$$\phi(t,y)=e^{-ty}\phi(t),$$

 $\phi(t)$ being independent of y (e.g. by putting $\phi(t) = e'\phi(t, 1)$).

Also, by Parseval's theorem,

$$\int_{-\infty}^{\infty} |\phi(t)|^2 e^{-2ty} dt = \int_{-\infty}^{\infty} |\Phi(x+iy)|^2 dx.$$

Since this is bounded as $y \to \infty$, we must have $\phi(t) \equiv 0$ for t < 0; for

$$\int_{-\infty}^{-\delta} |\phi(t)|^2 dt \leqslant e^{-2\delta y} \int_{-\infty}^{-\delta} |\phi(t)|^2 e^{-2ty} dt < Ke^{-2\delta y} \to 0,$$

$$\int_{-\delta}^{-\delta} |\phi(t)|^2 dt = 0.$$

so that

Since it is also bounded as $y \to 0$, $\phi(t)$ belongs to $L^2(0,\infty)$.

Also, $\phi(t)(e^{-ty_1}-e^{-ty_2})$ is the transform of $\Phi(x+iy_1)-\Phi(x+iy_2)$. Hence

$$\int_{-\infty}^{\infty} |\Phi(x+iy_1) - \Phi(x+iy_2)|^2 dx = \int_{0}^{\infty} |\phi(t)|^2 (e^{-ty_1} - e^{-ty_2})^2 dt,$$

which tends to 0 as $y_1 \to 0$, $y_2 \to 0$. Hence $\Phi(x+iy)$ converges in mean as $y \to 0$, to $\Phi(x)$ say.

Next, if
$$y > 0$$
, $\Phi(z) = \frac{1}{2\pi i} \int \frac{\Phi(w)}{w-z} dw$,

the integral being taken round the rectangle $\pm a + iv_1$, $\pm a + iv_2$, where a > |x| and $v_1 < y < v_2$. As before, the integrals along the right- and left-hand sides tend to 0 as $a \to \infty$, and we obtain

$$\Phi(z) = \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u+iv_1)}{u+iv_1-z} du - \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u+iv_2)}{u+iv_2-z} du.$$

But

$$igg| \int_{-\infty}^{\infty} rac{\Phi(u+iv_2)}{u+iv_2-z} \, du igg|^2 \leqslant \int_{-\infty}^{\infty} |\Phi(u+iv_2)|^2 \, du \int_{-\infty}^{\infty} rac{du}{(u-x)^2+(v_2-y)^2} \ < rac{K}{v_2-y},$$

which tends to 0 as $v_2 \to \infty$. Hence

$$\Phi(z) = \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u + iv_1)}{u + iv_1 - z} du, \qquad (5.4.1)$$

and, making $v_1 \rightarrow 0$,

$$\Phi(z) = \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u)}{u-z} du. \qquad (5.4.2)$$

The integral with z replaced by \bar{z} is zero. Putting $\Phi(u) = f(u) - ig(u)$, $\Phi(z) = U(x,y) + iV(x,y)$, we obtain the formulae of the previous sections; and it follows from the theory of Cauchy's singular integral, and (5.3.5) and (5.3.9), that $\Phi(z) \to \Phi(x)$ for almost all x.

THEOREM 94. If $\psi(z)$ is regular and bounded for y > 0, then $\psi(z)$ tends to a limit as $y \to 0$ for almost all x.

For $\psi(z)/(z+i)$ satisfies the conditions of the above theorem, and so tends to a limit almost everywhere.

Notice also that, in the above theorem, $\phi(t)$ is the transform of $\Phi(x)$; for, if $\chi(t)$ is the transform of $\Phi(x)$, as $y \to 0$,

$$\int_{-\infty}^{\infty} |\chi(t) - \phi(t)e^{-ty}|^2 dt = \int_{-\infty}^{\infty} |\Phi(x) - \Phi(x + iy)|^2 dx \to 0,$$

$$\int_{-\infty}^{\infty} |\chi(t) - \phi(t)|^2 dt = 0.$$

Hence $\chi(t) \equiv \phi(t)$.

5.5. We also deduce

THEOREM 95. Alternative necessary and sufficient conditions that a complex $\Phi(x)$ of $L^2(-\infty,\infty)$ should be the limit as $z \to x$ of an analytic $\Phi(z)$ such that

$$\int_{-\infty}^{\infty} |\Phi(x+iy)|^2 dx < K$$

are

- (i) $\Phi(x) = f(x) ig(x)$, where f and g are conjugate functions of the class L^2 ;
 - (ii) $\phi(x)$, the transform of $\Phi(x)$, is null for x < 0.

The necessity and sufficiency of (i) follows at once from the above theorems.

The necessity of (ii) has been proved in the course of the previous proof. Conversely, let $\phi(x) \equiv 0$ for x < 0. Let Φ be its transform. Then

$$\Phi(u) = \lim_{a \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{0}^{a} e^{ixu} \phi(x) dx.$$

Let

$$\Phi(u+iv) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} e^{ix(u+iv)} \phi(x) dx \quad (v>0).$$

Then $\Phi(u+iv)$ is analytic for v>0, and

$$\int\limits_{-\infty}^{\infty}|\Phi(u+iv)|^2\,du=\int\limits_{0}^{\infty}e^{-2xv}|\phi(x)|^2\,dx\leqslant\int\limits_{0}^{\infty}|\phi(x)|^2\,dx.$$

Hence, by the above theorem, $\Phi(u+iv)$ converges in mean, and also almost everywhere, to $\Psi(u)$ say; and

$$\int_{0}^{U} \Phi(u) du = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \frac{e^{ixU} - 1}{ix} \phi(x) dx$$

$$= \lim_{v \to 0} \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \frac{e^{ix(U + iv)} - 1}{ix} \phi(x) dx$$

$$= \lim_{v \to 0} \int_{0}^{U} \Phi(u + iv) du = \int_{0}^{U} \Psi(u) du.$$

$$\Psi(u) \equiv \Phi(u).$$

Hence

i.e.

The result also follows from the transform formulae; for, if Φ satisfies the given conditions, Φ , f, and g are related as in § 5.1, and

$$\phi(x) = \text{l.i.m.} \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} \{f(u) - ig(u)\} e^{-ixu} du$$
$$= F(-x) - iG(-x) = 0 \quad (x < 0)$$

by (5.1.8). Conversely, let $\phi(x) \equiv 0$ for x < 0. Let $\Phi(u) = f(u) - ig(u)$, let a(x) and b(x) be defined as before in terms of f, and similarly $\alpha(x)$ and $\beta(x)$ in terms of g. Then

$$a(x)+ib(x)-i\{\alpha(x)+i\beta(x)\}=0 \quad (x<0),$$

 $a(x)=-\beta(x), \quad b(x)=\alpha(x) \quad (x<0).$

Hence g is the conjugate of f, and the sufficiency of the condition follows from condition (i).

5.6. THEOREM 96. A necessary and sufficient condition that $\Phi(x)$ should be the limit as $y \to 0$ of an analytic $\Phi(z)$ such that

$$\int_{-\infty}^{\infty} |\Phi(x+iy)|^2 dx = O(e^{2ky})$$

is that $\phi(x) \equiv 0$ for x < -k.

If k is the least number such that $\phi(x) \equiv 0$ for x < -k, then

$$\lim_{y\to\infty}\frac{1}{y}\log\int_{-\infty}^{\infty}|\Phi(x+iy)|^2\,dx=2k.$$

Suppose that $\Phi(z)$ satisfies the given conditions. Let

$$\Phi(z) = e^{-ikz}\Psi(z).$$

Then $\int_{-\infty}^{\infty} |\Psi(x+iy)|^2 dx = e^{-2ky} \int_{-\infty}^{\infty} |\Phi(x+iy)|^2 dx = O(1).$

Hence $\Psi(z) \to \Psi(x)$ almost everywhere, and, if $\psi(x)$ is the transform of $\Psi(x)$, $\psi(x) \equiv 0$ for x < 0. Now

$$\phi(x)= ext{l.i.m.} rac{1}{\sqrt{(2\pi)}}\int\limits_{-a}^{a}\Phi(u)e^{-ixu}\,du \ = ext{l.i.m.} rac{1}{\sqrt{(2\pi)}}\int\limits_{-a}^{a}\Psi(u)e^{-i(x+k)u}\,du = \psi(x+k).$$

Hence $\phi(x) \equiv 0$ for x < -k; and in view of the above theorem the argument is reversible. This proves the first part.

Again, since $\Phi(x+iy)$ is the transform of $e^{-uy}\phi(u)$,

$$\int\limits_{-\infty}^{\infty} |\Phi(x+iy)|^2 \, dx = \int\limits_{-\infty}^{\infty} e^{-2uy} |\phi(u)|^2 \, du = \int\limits_{-k}^{\infty} e^{-2uy} |\phi(u)|^2 \, du.$$
This is
$$\leqslant e^{2ky} \int\limits_{-k}^{\infty} |\phi(u)|^2 \, du;$$
and, if
$$\omega(u) = \int\limits_{-k}^{u} \phi |(u)|^2 \, du,$$
it equals
$$2y \int\limits_{-k}^{\infty} e^{-2uy} \omega(u) \, du \geqslant 2y \omega(-k+\delta) \int\limits_{-k+\delta}^{\infty} e^{-2uy} \, du$$

Hence the second part.

5.7. For a function having a mean value in a finite strip the corresponding theorem is as follows.

THEOREM 97. Let $\Phi(z)$ be an analytic function, regular for

$$y_1 < y < y_2,$$

$$\int\limits_0^\infty |\Phi(x+iy)|^2 dx$$

and such that

exists and is bounded for $y_1 < y < y_2$.

Then the boundary functions $\Phi(x+iy_1)$ and $\Phi(x+iy_2)$ exist as mean-square limits, and also almost everywhere as ordinary limits of $\Phi(x+iy)$. For $y_1 < y < y_2$,

$$\Phi(z) = \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u+iy_1)}{u+iy_1-z} du - \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{\Phi(u+iy_2)}{u+iy_2-z} du.$$

The transform of $\Phi(x+iy)$ is of the form $e^{-iy}\phi(t)$, where $e^{-iy}\phi(t)$ belongs to L^2 for $y_1 \leq y \leq y_2$.

This is an obvious consequence of the above analysis, except perhaps for the existence of the limit of $\Phi(x+iy)$ almost everywhere as $y \to y_1$ or y_2 . However, the previous analysis shows that

$$\int_{-\infty}^{\infty} \frac{\Phi(u+iy_1)}{u+iy_1-z} \, du$$

tends to a limit almost everywhere as $y \rightarrow y_1$ from above; and

$$\int_{-\infty}^{\infty} \Phi(u+iy_2) du$$

is regular for all $y < y_2$, and so tends to a limit everywhere as $y \to y_1$. Similarly for the case $y \to y_2$.

5.8. Theorem 98. Let f(x) belong to $L^2(-\infty,\infty)$. Then

$$f(x) = f_{+}(x) + f_{-}(x),$$

where $f_+(x)$ belongs to $L^2(-\infty,\infty)$, and is the mean-square limit of an analytic function $f_+(z)$, regular for $\mathbf{I}(z) > 0$; and similarly $f_-(x)$ is the mean-square limit of $f_-(z)$, regular for $\mathbf{I}(z) < 0$.

Let F(x) be the transform of f(x), and

$$f_{+}(z) = rac{1}{\sqrt{(2\pi)}} \int\limits_{0}^{\infty} F(u)e^{-izu} du, \qquad f_{-}(z) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{0} F(u)e^{-izu} du.$$

Plainly $f_{+}(z)$ and $f_{-}(z)$ are regular for y > 0, y < 0, respectively. The rest of the theorem follows as in § 5.4.

THEOREM 99. Let f(x) belong to $L^2(0,\infty)$. Then

$$f(x) = f_{(+)}(x) + f_{(-)}(x),$$

where $f_{(+)}(x)$ belongs to $L^2(0,\infty)$, and is the mean-square limit as $\arg z \to +0$ of an analytic function $f_{(+)}(z)$, regular for $\arg z >0$; and similarly $f_{(-)}(x)$ is the mean-square limit of $f_{(-)}(z)$, regular for $\arg z <0$.

This may be deduced from the previous theorem by putting z (of the present theorem) = e^{ζ} ; or deduced directly from Mellin transforms. In fact

$$f_{(+)}(z) = rac{1}{2\pi i} \int\limits_{1-i\infty}^{1} \mathfrak{F}(s) z^{-s} \, ds, \hspace{0.5cm} f_{(-)}(z) = rac{1}{2\pi i} \int\limits_{1}^{1+i\infty} \mathfrak{F}(s) z^{-s} \, ds.$$

5.9. Theorem 100†. If f(x) belongs to $L(-\infty,\infty)$, then

$$\int_{-\infty}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt$$

exists for almost all values of x.

We may suppose without loss of generality that $f(x) \ge 0$. Define U(x, y) and V(x, y) by (5.3.5), (5.3.6), and let

$$\Phi(z) = U(x,y) + iV(x,y) = \frac{1}{i\pi} \int_{-\infty}^{\infty} \frac{f(t)}{t-z} dt \quad (y > 0).$$

From its definition it is clear that $U(x,y) \geqslant 0$.

Let
$$\Psi(z) = e^{-\Phi(z)} = e^{-U(x,y)-iV(x,y)}.$$

Since $U(x,y) \ge 0$, $|\Psi(z)| \le 1$. Hence, as $y \to 0$, $\Psi(z)$ tends to a finite limit for almost all x (Theorem 94); and this limit can be 0 in a set of measure 0 only, since U(x,y) tends to the finite limit f(x) almost everywhere. Hence $\Psi(z)$ tends to a finite non-zero limit almost everywhere. Hence $\Phi(z)$ tends to a finite limit almost everywhere. Hence V(x,y) tends to a finite limit almost everywhere. The result then follows from Theorem 92.

5.10. Hilbert transforms of the class L^p .

Theorem 101. Let f(x) belong to $L^p(-\infty,\infty)$, where p>1. Then the formula

 $g(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt$ (5.10.1)

defines almost everywhere a function g(x), also belonging to $L^p(-\infty,\infty)$. The reciprocal formula

$$f(x) = -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{g(x+t) - g(x-t)}{t} dt$$
 (5.10.2)

[†] Plessner (1). I believe that this version of the argument is due to Littlewood.

also holds almost everywhere; and

$$\int_{-\infty}^{\infty} |g(x)|^p dx \leqslant M_p^p \int_{-\infty}^{\infty} |f(x)|^p dx, \qquad (5.10.3)$$

where M_p depends on p only.

This is M. Riesz's extension tof Theorem 91. There are three cases.

(i) Let p be an even integer. Let

$$\Phi_a(z) = \frac{1}{i\pi} \int_{-a}^{a} \frac{f(t)}{t-z} dt = U_a(x,y) + iV_a(x,y) \quad (y > 0).$$

Consider the integral

$$\int {\{\Phi_a(z)\}^p} \ dz$$

taken along the straight line from -R+iy to R+iy, and round the semicircle above it. For a fixed a, $\Phi_a(z) = O(1/|z|)$ as $|z| \to \infty$. Hence, making $R \to \infty$, we obtain

$$\int_{-\infty}^{\infty} {\{\Phi_a(x+iy)\}^p dx} = 0,$$

$$\int_{-\infty}^{\infty} (U_a+iV_a)^p dx = 0.$$

i.e.

Expanding the integrand by the binomial theorem, and taking the real part,

$$\int\limits_{-\infty}^{\infty} \left\{ V_a^p - {p \choose 2} V_a^{p-2} U_a^2 + {p \choose 4} V_a^{p-4} U_a^4 - \ldots \pm U_a^p \right\} dx = 0.$$
Hence
$$\int\limits_{-\infty}^{\infty} V_a^p dx \leqslant {p \choose 2} \int\limits_{-\infty}^{\infty} V_a^{p-2} U_a^2 dx + \ldots + \int\limits_{-\infty}^{\infty} U_a^p dx.$$
Now
$$\int\limits_{-\infty}^{\infty} V_a^{p-2k} U_a^{2k} dx \leqslant \left(\int\limits_{-\infty}^{\infty} V_a^p dx \right)^{(p-2k)/p} \left(\int\limits_{-\infty}^{\infty} U_a^p dx \right)^{2k/p}$$
Writing
$$X^p = \left(\int\limits_{-\infty}^{\infty} V_a^p dx \right) / \left(\int\limits_{-\infty}^{\infty} U_a^p dx \right).$$

Writing

$$X^{p} = \left(\int_{-\infty}^{\infty} V_{a}^{p} dx\right) / \left(\int_{-\infty}^{\infty} U_{a}^{p} dx\right),$$

it follows that

$$X^p\leqslant \binom{p}{2}X^{p-2}+\binom{p}{4}X^{p-4}+\ldots+1.$$

Hence X does not exceed the greatest positive root of the equation

$$X^{p} - \binom{p}{2} X^{p-2} - \dots - 1 = 0,$$

† M. Riesz (1), (2). For another method see Titchmarsh (7).

and so

$$X \leqslant M_p$$

where M_p depends on p only; i.e.

$$\int\limits_{-\infty}^{\infty}V_a^p\,dx\leqslant M_p^p\int\limits_{-\infty}^{\infty}U_a^p\,dx.$$

Now

$$|U_{a}(x,y)| = \frac{y}{\pi} \left| \int_{-a}^{a} \frac{f(t)}{(t-x)^{2}+y^{2}} dt \right| \leq \frac{y}{\pi} \int_{-a}^{a} \frac{|f(t)|}{(t-x)^{2}+y^{2}} dt,$$

$$|U_{a}(x,y)|^{p} \leq \frac{y^{p}}{\pi^{p}} \int_{-\infty}^{\infty} \frac{|f(t)|^{p}}{(t-x)^{2}+y^{2}} dt \left\{ \int_{-\infty}^{\infty} \frac{dt}{(t-x)^{2}+y^{2}} \right\}^{p-1}$$

$$= \frac{y}{\pi} \int_{-\infty}^{\infty} \frac{|f(t)|^{p}}{(t-x)^{2}+y^{2}} dt,$$

$$\int_{-\infty}^{\infty} |U_{a}|^{p} dx \leq \frac{y}{\pi} \int_{-\infty}^{\infty} |f(t)|^{p} dt \int_{-\infty}^{\infty} \frac{dx}{(t-x)^{2}+y^{2}}$$

$$= \int_{-\infty}^{\infty} |f(t)|^{p} dt. \qquad (5.10.4)$$

Hence

$$\int_{-\infty}^{\infty} \{V_a(x,y)\}^p \ dx \leqslant M_p^p \int_{-\infty}^{\infty} \{f(t)\}^p \ dt.$$

Making $a \to \infty$,

$$V_a(x,y) \to V(x,y) = -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{t-x}{(t-x)^2 + y^2} f(t) dt,$$
 (5.10.5)

and, making $y \to 0$, $V(x, y) \to -g(x)$ almost everywhere, by Theorems 92 and 100. It therefore follows from (5.10.5) and Fatou's theorem that (5.10.3) holds. (See Titchmarsh, *Theory of Functions*, § 10.81.)

(ii) Suppose next that p is not an integer. We may suppose without loss of generality that $f(t) \ge 0$. Then U(x,y) > 0, and $U_n(x,y) > 0$ for y > 0, $a > a_0$.

Some care is now needed in the definition of pth powers. Let

$$(U+iV)^p = e^{\frac{1}{2}p\log(U^2+V^2)+ip\arctan(V|U)},$$

where $-\frac{1}{2}\pi < \arctan(V/U) < \frac{1}{2}\pi$ for U > 0. Making $U \to 0$, we obtain

$$(iV)^p = |V|^p e^{\frac{1}{2}ip\pi} \quad (V > 0), \qquad |V|^p e^{-\frac{1}{2}ip\pi} \quad (V < 0).$$

With these definitions we have

$$egin{align} |(U+iV)^p-(iV)^p|&=pigg|\int\limits_{iV}^{U+iV}z^{p-1}\,dzigg| \ &\leqslant pU(U^2+V^2)^{rac{1}{2}p-rac{1}{2}}\leqslant p\,2^{rac{1}{2}(p-1)}(U^p+U|V|^{p-1}). \end{split}$$

Applying this to U_a , V_a , we obtain

$$\left|\int\limits_{-\infty}^{\infty}\left\{(U_a+iV_a)^p-(iV_a)^p\right\}dx\right|\leqslant K_p\bigg\{\int\limits_{-\infty}^{\infty}U_a^p\,dx+\int\limits_{-\infty}^{\infty}U_a|V_a|^{p-1}\,dx\bigg\}.$$
 Ut, as before

But, as before,

$$\int_{-\infty}^{\infty} \{U_a + iV_a\}^p \ dx = 0,$$

and
$$\left|\int_{-\infty}^{\infty} (iV_a)^p dx\right| \geqslant \left|\mathbf{R}\int_{-\infty}^{\infty} (iV_a)^p dx\right| = \left|\cos \frac{1}{2}p\pi\right|\int_{-\infty}^{\infty} |V_a|^p dx.$$

Hence

$$|\cos rac{1}{2}p\pi|\int\limits_{-\infty}^{\infty}|V_a|^p\ dx\leqslant K_p\int\limits_{-\infty}^{\infty}U_a^p\ dx+K_p\int\limits_{-\infty}^{\infty}U_a|V_a|^{p-1}\ dx,$$

and the proof of (5.10.1) and (5.10.3) can now be completed as in the previous case.

The above proof fails if p is an odd integer. Leaving this case aside for the moment, we next prove (5.10.2) in the above cases.

We have

$$\begin{split} |U(x,y)-f(x)|^p & \leqslant \frac{y^p}{\pi^p} \bigg| \int_{-\infty}^{\infty} \frac{f(t+x)-f(x)}{t^2+y^2} \, dt \bigg|^p \\ & < \frac{y}{\pi} \int_{-\infty}^{\infty} \frac{|f(t+x)-f(x)|^p}{t^2+y^2} \, dt, \\ & \int_{-\infty}^{\infty} |U(x,y)-f(x)|^p \, dx < \frac{y}{\pi} \int_{-\infty}^{\infty} \frac{dt}{t^2+y^2} \int_{-\infty}^{\infty} |f(t+x)-f(x)|^p \, dx. \end{split}$$

The inner integral (see Titchmarsh, Theory of Functions, p. 397, exs. 17-19) is bounded for all t, and tends to 0 with t; hence the right-hand side is less than

$$\begin{split} K_p y \int\limits_{\delta}^{\infty} \frac{dt}{t^2 + y^2} + \epsilon(\delta) y \int\limits_{0}^{\delta} \frac{dt}{t^2 + y^2} &< K_p y \int\limits_{\delta}^{\infty} \frac{dt}{t^2} + \epsilon(\delta) y \int\limits_{0}^{\infty} \frac{dt}{t^2 + y^2} \\ &< K_p y / \delta + \frac{1}{2} \pi \epsilon(\delta), \end{split}$$

which tends to 0 by choosing first δ and then y. Hence for any p

$$\lim_{y\to 0}\int_{-\infty}^{\infty}|U(x,y)-f(x)|^p\,dx=0.$$

Also, by (5.10.4),

$$\int\limits_{-\infty}^{\infty}|U(x,y)-U_{a}(x,y)|^{p}~dx\leqslant \Big(\int\limits_{-\infty}^{-a}+\int\limits_{a}^{\infty}\Big)|f(t)|^{p}~dt\rightarrow 0$$

as $a \to \infty$, uniformly with respect to y. Hence

$$\int_{-\infty}^{\infty} |U_a(x,y) - f(x)|^p dx \to 0$$
 (5.10.6)

as $a \to \infty$, $y \to 0$, in any manner.

Again, by the calculus of residues,

$$rac{1}{\pi i}P\int\limits_{y=\eta} rac{\Phi_a(z)}{z-\xi-i\eta}\,dz=\Phi_a(\xi+i\eta)\quad (\eta>0),$$

and, taking imaginary parts,

$$\frac{1}{\pi}P\int_{-\infty}^{\infty}\frac{U_a(x,\eta)}{x-\xi}\,dx=-V_a(\xi,\eta).$$

Hence the Hilbert transform of $U_a(x, y)$ is $-V_a(x, y)$, and it follows from (5.10.3) that, for the values of p already dealt with,

$$\int_{-\infty}^{\infty} |V_a(x,y) + g(x)|^p < K_p \int_{-\infty}^{\infty} |U_a(x,y) - f(x)|^p dx.$$
 (5.10.7)

Combining (5.10.6) and (5.10.7), it follows that

$$\int_{-\infty}^{\infty} |\Phi_a(z) - \{f(x) - ig(x)\}|^p dx \to 0$$

as $y \to 0$, $a \to \infty$, in any manner.

Now by the calculus of residues

$$rac{1}{2\pi i}\int\limits_{-\infty}^{\infty}rac{\Phi_a(z)}{z-\xi-i\eta}\,dz=\Phi_a(\xi+i\eta)\quad (y<\eta).$$

Making $a \to \infty$, $y \to 0$, it follows that

$$\frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{f(x) - ig(x)}{x - \xi - i\eta} dx = \Phi(\xi + i\eta),$$

and hence

$$\frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{-ig(x)}{x - \xi - i\eta} dx = \frac{1}{2} \Phi(\xi + i\eta).$$

Taking real parts,

$$\frac{1}{\pi}\int_{-\infty}^{\infty}\frac{x-\xi}{(x-\xi)^2+\eta^2}g(x)\,dx=-U(\xi,\eta).$$

Making $\eta \to 0$, the left-hand side tends almost everywhere to the Hilbert transform of g(x), by Theorem 92; and the right-hand side (Cauchy's singular integral) tends almost everywhere to -f(x). This proves (5.10.2).

(iii) To prove the case where p is an odd integer, we shall prove that if the theorem holds for any p it also holds for 2p. Since it holds when p is half an odd integer, it will follow that it holds when p is an odd integer.

Applying the calculus of residues as before, but now to $\{\Phi_a(z)\}^2$, we obtain

$$\frac{1}{\pi i} P \int_{-\infty}^{\infty} \frac{\{\Phi_a(x+iy)\}^2}{x-\xi} dx = \{\Phi_a(\xi+iy)\}^2 \quad (y>0),$$

i.e

$$\frac{1}{\pi i} \int_{-\infty}^{\infty} \frac{U_a^2 - V_a^2 + 2iU_aV_a}{x - \xi} dx = U_a^2(\xi, y) - V_a^2(\xi, y) + 2iU_a(\xi, y)V_a(\xi, y).$$

Taking imaginary parts, it follows that the Hilbert transform of $U_a^2 - V_a^2$ is $-2U_aV_a$. Let $\psi(x)$ be the transform of U_a^2 , and $\chi(x)$ that of V_a^2 . Then $\psi(x) - \chi(x) = -2U_aV_a$.

Hence

$$|\chi(x)|^p \leqslant 2^p |\psi(x)|^p + 2^{2p} |U_a V_a|^p,$$

$$\int_{-\infty}^{\infty} |\chi(x)|^p dx \leqslant 2^p \int_{-\infty}^{\infty} |\psi(x)|^p dx + 2^{2p} \int_{-\infty}^{\infty} |U_a V_a|^p dx.$$

$$\int_{-\infty}^{\infty} |U_a V_a|^p dx \leqslant \left(\int_{-\infty}^{\infty} |U_a|^{2p} dx \int_{-\infty}^{\infty} |V_a|^{2p} dx\right)^{\frac{1}{2}},$$

Now

and, by the fundamental inequality (5.10.3) (for p),

$$\int\limits_{-\infty}^{\infty} |\psi(x)|^p \ dx < K_p \int\limits_{-\infty}^{\infty} |U_a|^{2p} \ dx,$$

$$\int\limits_{-\infty}^{\infty} |V_a|^{2p} \ dx < K_p \int\limits_{-\infty}^{\infty} |\chi(x)|^p \ dx.$$

Altogether

$$\int\limits_{-\infty}^{\infty} |V_a|^{2p} \ dx < K_p \int\limits_{-\infty}^{\infty} |U_a|^{2p} \ dx \ + K_p \bigg(\int\limits_{-\infty}^{\infty} |U_a|^{2p} \ dx \int\limits_{-\infty}^{\infty} |V_a|^{2p} \ dx \bigg)^{\frac{1}{2}}$$

The result for 2p now follows as in the previous cases. This completes the proof.

5.11. THEOREM 102.† Let f(x) and g(x) be Hilbert transforms of the class L^p , and h(x) and k(x) Hilbert transforms of the class $L^{p'}$, where p' = p/(p-1). Then

$$\int_{-\infty}^{\infty} f(x)h(x) dx = \int_{-\infty}^{\infty} g(x)k(x) dx.$$
 (5.11.1)

If p = 2, p' = 2, and (5.3.3) gives

$$\int_{-\infty}^{\infty} \{f(x) + h(x)\}^2 dx = \int_{-\infty}^{\infty} \{g(x) + k(x)\}^2 dx, \qquad (5.11.2)$$

and the result follows in the usual way.

In the general case, define $U_a(x,y)$, $V_a(x,y)$ as before, and let $P_b(x,y)$, $Q_b(x,y)$ be the corresponding functions for h and k. We have seen that the Hilbert transform of U_a is $-V_a$, and similarly that of P_b is $-Q_b$. Since these functions belong to L^2 ,

$$\int_{-\infty}^{\infty} U_a(x,y) P_b(x,y') \ dx = \int_{-\infty}^{\infty} V_a(x,y) Q_b(x,y') \ dx. \tag{5.11.3}$$

Making $a \to \infty$, $y \to 0$, $b \to \infty$, $y' \to 0$, U_a and V_a converge in mean to f and -g, with exponent p; and P_b and Q_b converge in mean to h and -k, with exponent p'. Hence the result.

Example. Let h(x) = 1/(x-a) ($|x-a| > \delta$), 0 ($|x-a| \leqslant \delta$). Then

$$k(x) = \frac{1}{\pi} P \int_{-\infty}^{\infty} \frac{h(u)}{u - x} du = \frac{1}{\pi} P \left\{ \int_{-\infty}^{a - \delta} \frac{du}{(u - a)(u - x)} + \int_{a + \delta}^{\infty} \frac{du}{(u - a)(u - x)} \right\}$$
$$= \frac{1}{\pi (a - x)} \log \left| \frac{a + \delta - x}{a - \delta - x} \right|.$$

Hence

$$\int_{\delta}^{\infty} \frac{f(a+x) - f(a-x)}{x} dx = \frac{1}{\pi} \int_{-\infty}^{\infty} g(x) \log \left| \frac{a+\delta-x}{a-\delta-x} \right| \frac{dx}{a-x}$$

$$= \frac{1}{\pi} \int_{-\infty}^{\infty} g(t+a) \log \left| \frac{\delta+t}{\delta-t} \right| \frac{dt}{t}. \quad (5.11.4)$$

† M. Riesz (1), (2).

We can use (5.11.1) to give an alternative proof of the case of Theorem 101 where p is an odd integer.

Let h(x), k(x) be transforms of $L^{p'}$. Since the theorem has been proved for p', on making $b \to \infty$, $y' \to 0$ in (5.11.3) we obtain

$$\int_{-\infty}^{\infty} U_a(x,y)h(x) \ dx = \int_{-\infty}^{\infty} V_a(x,y)k(x) \ dx.$$

Hence

$$egin{aligned} \left|\int\limits_{-\infty}^{\infty}V_a(x,y)k(x)\;dx
ight|&\leqslant \left(\int\limits_{-\infty}^{\infty}|U_a|^p\;dx
ight)^{1/p}igg(\int\limits_{-\infty}^{\infty}|h(x)|^{p'}\;dxigg)^{1/p'}\ &\leqslant M_{p'}igg(\int\limits_{-\infty}^{\infty}|f(t)|^p\;dtigg)^{1/p}igg(\int\limits_{-\infty}^{\infty}|k(x)|^{p'}\;dxigg)^{1/p'} \end{aligned}$$

by (5.10.4) and (5.10.3) for p'. Here k(x) may be any function of $L^{p'}$. Take $k(x) = |V_n(x,y)|^{p-1} \operatorname{sgn} V_n(x,y).$

 $\begin{array}{ll} \text{Then} & \int\limits_{-\infty}^{\infty} |V_a|^p \ dx \leqslant M_{p'} \bigg(\int\limits_{-\infty}^{\infty} |f(t)|^p \ dt \bigg)^{1/p} \bigg(\int\limits_{-\infty}^{\infty} |V_a|^p \ dx \bigg)^{1/p'}, \\ \\ \text{or} & \int\limits_{-\infty}^{\infty} |V_a|^p \ dx \leqslant M_{p'}^p \int\limits_{-\infty}^{\infty} |f(t)|^p \ dt. \end{array}$

The theorem for p now follows as before.

It also follows that, if M_p is the least constant for which (5.10.3) holds, then $M_p \leqslant M_{p'}$. Hence, since p and p' are interchangeable, $M_p = M_{p'}$.

5.12. THEOREM 103. Let $\Phi(z)$ be an analytic function, regular for y > 0, and let $\int_{0}^{\infty} |\Phi(x+iy)|^p dx < K \quad (p > 1)$ (5.12.1)

for all values of y. Then $\Phi(x+iy)$ converges for almost all x, and also in the mean of order p, as $y \to 0$, to f(x)-ig(x), where f(x) and g(x) are Hilbert transforms of the class L^p .

It is convenient to use the following lemma.

LEMMA. Let $\lambda_n(x)$ be a sequence of functions such that

$$\int_{0}^{b} |\lambda_{n}(x)|^{p} dx < K,$$

while $\lambda_n(x) \to 0$ almost everywhere. Then if $\mu(x)$ belongs to $L^{p'}$,

$$\int_{a}^{b} \lambda_{n}(x)\mu(x) \ dx \to 0.$$

Suppose first that the interval (a, b) is finite. By Egoroff's theorem† $\lambda_n(x) \to 0$ uniformly in a set E of measure $b-a-\delta$, and hence

$$\int\limits_E \lambda_n(x)\mu(x)\ dx\to 0.$$

Also

$$\begin{split} \left| \int\limits_{CE} \lambda_n(x) \mu(x) \; dx \right| & \leqslant \left(\int\limits_{CE} |\lambda_n(x)|^p \; dx \right)^{1/p} \biggl(\int\limits_{CE} |\mu(x)|^{p'} \; dx \biggr)^{1/p'} \\ & < K \biggl(\int\limits_{CE} |\mu(x)|^{p'} \; dx \biggr)^{1/p'}, \end{split}$$

which tends to 0 with δ , and is independent of n. Hence the result. If $b = \infty$, we first take X so large that

$$\left|\int\limits_X^\infty \lambda_n(x)\mu(x)\;dx\right|\leqslant K\bigg(\int\limits_X^\infty |\mu(x)|^{p'}\;dx\bigg)^{1/p'}<\epsilon$$

and then argue as before with (a, X).

If $\Phi(z) \to f(x) - ig(x)$, it follows from Fatou's theorem that f and g belong to L^p . We prove (5.4.1) as before, and (5.4.2) now follows from (5.4.1) by the lemma, taking

$$\lambda_n(u) = \{\Phi(u+iv) - f(u) + ig(u)\} \frac{u-x+iy}{u-x+i(v-y)} \quad (n=v)$$

and

$$\mu(x) = \frac{1}{u - x + iy}.$$

Hence

$$\phi(z) = \frac{1}{2}(U+iV) - \frac{1}{2}i(P+iQ),$$

where U and V are (5.3.5), (5.3.6), and P, Q are defined similarly with g instead of f. Now make $g \to 0$. Denoting by f^* the conjugate of f, and by g^* that of g, we obtain

$$f-ig = \frac{1}{2}(f-if^*) - \frac{1}{2}i(g-ig^*).$$

Hence $f = -g^*$, $g = f^*$ almost everywhere.

That $\Phi(z)$ converges in mean to f(x)-ig(x) with index p follows from the analysis of § 5.10.

That $\Phi(z)$ tends to a limit almost everywhere was deduced by Hille and Tamarkin (5) from the corresponding theorem for series (Zygmund, *Trigonometrical Series*, § 7.53). It could be proved directly as follows. If Φ has no zeros, the result follows on applying Theorem 93

to $\{\Phi(z)\}^{\frac{1}{2}p}$. Otherwise, let z_p run through the zeros of Φ in y > 0, and let

 $B_n(z) = \prod_{\nu=1}^n \frac{z - z_{\nu}}{z - \bar{z}_{\nu}} \frac{\bar{z}_{\nu} + i}{z_{\nu} + i} \left| \frac{z_{\nu} + i}{\bar{z}_{\nu} + i} \right|$

(assuming that no z_{ν} is i). For a fixed n, $|B_n| \geqslant 1 - \epsilon$ for $y \leqslant \eta$, say, and all x. Let $\Phi(z) = G_n(z)B_n(z)$. Then

$$\int\limits_{-\infty}^{\infty}|G_n(x+iy)|^p\ dx<\frac{K}{(1-\epsilon)^p}$$

for $y \leq \eta$. Since $G_n(x+iy')$, $G_n(x+iy)$, (y < y') are related like the previous $\Phi(z)$, f(x)-ig(x), it follows from the analysis of § 5.10 (especially (5.10.4)) that

$$\int_{-\infty}^{\infty} |G_n(x+iy)|^p \, dx < K'$$

for all y, K' depending on K and p only.

If Φ has an infinity of zeros, a little consideration of Carleman's formula (Titchmarsh, *Theory of Functions*, § 3.7) shows that

$$\sum I(z_{\nu})/(1+|z_{\nu}|^2)$$

is convergent, and hence that $B(z) = \lim B_n(z)$ and $G(z) = \lim G_n(z)$ exist and are analytic. It follows that $\Phi(z) = G(z)B(z)$, where G(z) satisfies (5.12.1) with some K, and has no zeros, and $|B(z)| \leq 1$. Hence G(z) tends to a limit almost everywhere, as before, and so does B(z), by Theorem 94.

5.13. THEOREM 104. Let f(x) belong to $L^p(p > 1)$, and let g(x) be its conjugate. Let $\lambda(x)$ belong to L^q , where q > 1, $pq \leq p+q$, and let

$$h(x) = \int_{-\infty}^{\infty} \lambda(t) f(x-t) dt, \qquad k(x) = \int_{-\infty}^{\infty} \lambda(t) g(x-t) dt.$$

Then if pq < p+q, h(x) and k(x) are conjugates of the class L^p , where P = pq/(p+q-pq). If pq = p+q, h(x) and k(x) are conjugates, in the sense that

$$k(x) = \frac{1}{\pi} \int_{-\infty}^{+\infty} \frac{h(x+u) - h(x-u)}{u} du,$$

and reciprocally, for all values of x.

(i) Suppose first that pq < p+q. Then h(x) and k(x) belong to L^p , by Lemma β of § 4.2. We have to prove that they are conjugate.

Let $\Phi(z)$ be the analytic function of which f(x)-ig(x) is the boundary value, and let

$$\Psi(z) = \int_{-\infty}^{\infty} \lambda(t)\Phi(z-t) dt \quad (y > 0). \tag{5.13.1}$$

It follows from the lemma of § 5.4 that $\Phi(z)$ is bounded in any strip $0 < y_1 \leqslant y \leqslant y_2$. Hence

$$\begin{split} \left| \int\limits_{T}^{T'} \lambda(t) \Phi(z-t) \; dt \right| &\leqslant \int\limits_{T}^{T'} |\lambda(t)| \cdot |\Phi(z-t)|^{p|q'} |\Phi(z-t)|^{1-p|q'|} \, dt \\ &\leqslant K(y_1,y_2) \bigg(\int\limits_{T}^{T'} |\lambda(t)|^q \; dt \bigg)^{1/q} \bigg(\int\limits_{T}^{T'} |\Phi(z-t)|^p \; dt \bigg)^{1/q'}. \end{split}$$

Hence the integral (5.13.1) converges uniformly in $y_1 \leqslant y \leqslant y_2$. Hence $\Psi(z)$ is analytic for y > 0.

Also, by Lemma β of § 4.2,

$$\int_{-\infty}^{\infty} |\Psi(z)|^{P} dx \leqslant \left(\int_{-\infty}^{\infty} |\lambda(t)|^{q} dt\right)^{P/q} \left(\int_{-\infty}^{\infty} |\Phi(z-t)|^{p} dt\right)^{P/p},$$

which is bounded; and similarly

$$\int_{-\infty}^{\infty} |\Psi(z) - h(x) + ik(x)|^{P} dx$$

$$\leq \left(\int_{-\infty}^{\infty} |\lambda(t)|^{q} dt\right)^{P/q} \left(\int_{-\infty}^{\infty} |\Phi(z-t) - f(x-t) + ig(x-t)|^{p} dt\right)^{P/p},$$

which tends to 0 as $y \to 0$. Hence, by Theorem 103, h(x) and k(x) are conjugate.

(ii) If pq = p+q, it is known that h(x) and k(x) are continuous, and tend to 0 at infinity.†

In this case the integral defining h(x) converges uniformly over any finite range. Hence

$$\frac{1}{\pi} \int_{\delta}^{\Delta} \frac{h(x+u) - h(x-u)}{u} du = \frac{1}{\pi} \int_{\delta}^{\Delta} \frac{du}{u} \int_{-\infty}^{\infty} \lambda(t) \{f(x+u-t) - f(x-u-t)\} dt$$

$$= \frac{1}{\pi} \int_{-\infty}^{\infty} \lambda(t) dt \int_{\delta}^{\Delta} \frac{f(x+u-t) - f(x-u-t)}{u} du$$

$$= \int_{-\infty}^{\infty} \lambda(t) \{g_{\delta}(x-t) - g_{\Delta}(x-t)\} dt,$$

† See Titchmarsh, Theory of Functions, p. 398, exs. 20, 21.

where

$$g_{\delta}(x) = \frac{1}{\pi} \int_{\delta}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt.$$

Now

$$g_{\delta}(x-t) \rightarrow g(x-t), \qquad g_{\Delta}(x-t) \rightarrow 0$$

as $\delta \to 0$, $\Delta \to \infty$, for almost all t. Hence, by the lemma of § 5.12, it is sufficient to prove that

$$\int_{-\infty}^{\infty} |g_{\delta}(x)|^p \ dx < K$$

for all δ . By (5.11.4)

$$g_{\delta}(x) = \frac{1}{\pi^2} \int_{-\infty}^{\infty} g(t+x) \log \left| \frac{\delta+t}{\delta-t} \right| \frac{dt}{t}.$$

Hence

$$\begin{split} |g_{\delta}(x)|^p &\leqslant \frac{1}{\pi^{2p}} \int\limits_{-\infty}^{\infty} |g(t+x)|^p \bigg| \log \bigg| \frac{\delta + t}{\delta - t} \bigg| \bigg| \frac{dt}{|t|} \bigg(\int\limits_{-\infty}^{\infty} \bigg| \log \bigg| \frac{\delta + t}{\delta - t} \bigg| \bigg| \frac{dt}{|t|} \bigg)^{p-1} \\ &< K \int\limits_{-\infty}^{\infty} |g(t+x)|^p \bigg| \log \bigg| \frac{\delta + t}{\delta - t} \bigg| \bigg| \frac{dt}{|t|}, \end{split}$$

by putting $t = \delta u$ in the last factor. Integrating with respect to x and inverting, the result now follows.

5.14. The case p = 1. We have so far supposed that f(x) belongs to L^p , where p > 1. The general Theorem 101 fails in the case p = 1, in which f(x) belongs to L. We have seen (Theorem 100) that g(x) still exists almost everywhere in this case. But g(x) does not necessarily belong to L. Suppose for example that

$$f(t) = \frac{1}{t \log^2 t}$$
 $(t > 0), 0 \ (t \le 0).$

Then for x > 0

$$g(-x) = \frac{1}{\pi} P \int_{-\infty}^{\infty} \frac{f(t)}{t+x} dt > \frac{1}{\pi} \int_{0}^{x} \frac{dt}{2xt \log^2 t} = \frac{1}{2\pi x \log x}.$$

Hence g(x) does not belong to L. In fact it is possible to construct examples in which g(x) does not belong to L over any interval, however small.

We have, however, the following theorem.†

† Corresponding to a theorem of Kolmogoroff on Fourier series; see Littlewood (1), Titchmarsh (13), Zygmund, Trigonometrical Series, § 7.24.

THEOREM 105. Let f(x) belong to $L(-\infty,\infty)$. Then the formula

$$g(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt$$
 (5.14.1)

defines a finite g(x) for almost all values of x, and

$$\int_{-\infty}^{\infty} \frac{|g(x)|^p}{1+x^2} dx \tag{5.14.2}$$

is convergent if 0 .

We may suppose without loss of generality that $f(t) \ge 0$, and that f(t) is not null. Let

$$\Phi_a(z) = rac{1}{i\pi} \int \int dt = U_a + iV_a \quad (y > 0),$$

as before. Then

$$U_a(x,y) = rac{y}{\pi} \int_{-a}^{a} rac{f(t)}{(t-x)^2 + y^2} dt > 0$$

if a is large enough. Let 0 , and let

$$\{\Phi_a(z)\}^p = (U_a + iV_a)^p = e^{\frac{1}{2}p \log(U_a^* + V_a^*) + ip \arctan(V_a/U_a)},$$

where $-\frac{1}{2}\pi < \arctan(V_a/U_a) < \frac{1}{2}\pi$. For a fixed a, $\Phi_a(z) = O(1/|z|)$ as $|z| \to \infty$, and the calculus of residues gives

$$\int_{-\infty}^{\infty} \frac{\{\Phi_a(z)\}^p}{z^2 + 1} \, dz = \pi \{\Phi_a(i)\}^p \quad (0 < y < 1).$$

Now

$$|\Phi_{a}(i)| = \left| \frac{1}{i\pi} \int_{-a}^{a} \frac{f(t)}{t-i} dt \right| \leqslant \frac{1}{\pi} \int_{-\infty}^{\infty} |f(t)| dt.$$

$$\left| \int_{-a}^{\infty} \frac{(U_{a}+iV_{a})^{p}}{z^{2}+1} dz \right| < K_{p}.$$
 (5.14.3)

Hence

If U > 0, $|V| \geqslant 1$,

$$|(U+iV)^p-(iV)^p|=\left|p\int\limits_{iV}^{U+iV}z^{p-1}\,dz\right|\leqslant pU|V|^{p-1}\leqslant U,$$

while if U > 0, |V| < 1,

$$|(U+iV)^p-(iV)^p| \leq (U+1)^p+1 \leq U+2.$$

Hence

$$\left| \int_{-\infty}^{\infty} \frac{(U_a + iV_a)^p - (iV_a)^p}{z^2 + 1} dz \right| \le \int_{-\infty}^{\infty} \frac{U_a + 2}{x^2 + 1} dx$$

$$\le \int_{-\infty}^{\infty} U_a dx + 2\pi = \frac{y}{\pi} \int_{-\infty}^{\infty} dx \int_{-a}^{a} \frac{f(t)}{(t - x)^2 + y^2} dt + 2\pi$$

$$= \int_{-a}^{a} f(t) dt + 2\pi \le \int_{-\infty}^{\infty} f(t) dt + 2\pi. \quad (5.14.4)$$

From (5.14.3) and (5.14.4) it follows that

$$\left| \int_{-\infty}^{\infty} \frac{(iV_a)^p}{z^2+1} dz \right| < K_p.$$

Now

$$\mathbf{R} \left\{ \frac{(iV_a)^p}{z^2+1} \right\} = |V_a|^p \frac{(x^2-y^2+1)\cos\frac{1}{2}p\pi \pm 2xy\sin\frac{1}{2}p\pi}{(x^2-y^2+1)^2+4x^2y^2} > K_p \frac{|V_a|^p}{x^2+1}$$

for sufficiently small y and all x. Hence

$$\int_{-\infty}^{\infty} \frac{|V_a|^p}{x^2+1} dx < K_p,$$

and the result follows as in the proof of Theorem 101.

5.15. Lipschitz conditions. Theorem 106.† Let f(x) belong to L^p (p > 1), and let it satisfy the Lipschitz condition

$$|f(x+h)-f(x)| < K|h|^{\alpha} \quad (0 < \alpha < 1)$$
 (5.15.1)

uniformly in x, as $h \to 0$ (say for all x and $0 < h \le 1$). Then Hilbert's reciprocal formulae (5.1.9), (5.1.10) hold for all values of x; and g(x) also belongs to L^p and satisfies a Lipschitz condition with the same α as f(x).

In this case the integral (5.1.9) plainly exists for all values of x.

We next observe that if f(x) satisfies the given conditions then it is bounded—in fact it tends to 0 as $x \to \infty$. For since f(x) is continuous, the points where $|f(x)| \ge \delta > 0$ form a set of intervals. The length of such an interval (x_1, x_2) tends to 0 as $x_1 \to \infty$, since

$$(x_2-x_1)\delta^p\leqslant \int\limits_{x_1}^{x_1}|f(x)|^p\ dx\to 0.$$

† Titchmarsh (5). The result corresponds to Privaloff's theorem for Fourier series. See Zygmund, Trigonometrical Series, § 7.4.

4362

Now since $|f(x_1)| = \delta$,

$$|f(x)| \leqslant |f(x_1)| + |f(x) - f(x_1)| \leqslant \delta + K|x - x_1|^{\alpha} \leqslant \delta + K|x_2 - x_1|^{\alpha},$$

which tends to 0, by choosing first δ and then x_1 . Hence $f(x) \to 0$.

It follows that if (5.15.1) holds for small h it holds for all h, with a possible rechoice of the constant K.

Now as $y \to 0$,

$$V(x,y)+g(x) = \frac{1}{\pi} \int_{0}^{\infty} \left(\frac{1}{t} - \frac{t}{t^2 + y^2}\right) \{f(x+t) - f(x-t)\} dt$$

$$= \frac{y^2}{\pi} \int_{0}^{\infty} \frac{f(x+t) - f(x-t)}{t(t^2 + y^2)} dt = O\left(y^2 \int_{0}^{\infty} \frac{t^{\alpha - 1}}{t^2 + y^2} dt\right) = O(y^{\alpha}).$$

Also

$$\begin{split} \frac{\partial V}{\partial x} &= -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{(t-x)^2 - y^2}{\{(t-x)^2 + y^2\}^2} f(t) \ dt \\ &= -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{t^2 - y^2}{(t^2 + y^2)^2} f(t+x) \ dt = -\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{t^2 - y^2}{(t^2 + y^2)^2} \{f(t+x) - f(x)\} \ dt \\ &= O\bigg(\int_{-\infty}^{\infty} \frac{|t^2 - y^2|}{(t^2 + y^2)^2} |t|^{\alpha} \ dt\bigg) = O(y^{\alpha - 1}). \end{split}$$

Hence, taking h > 0,

$$|g(x+h)-g(x)| \le |g(x+h)+V(x+h,h)|+|V(x+h,h)-V(x,h)|+|g(x)+V(x,h)| = O(h^{\alpha})+O\left(\left|\int_{0}^{x+h} \frac{\partial}{\partial \xi} V(\xi,h) d\xi\right|\right)+O(h^{\alpha}) = O(h^{\alpha}),$$

so that g(x) satisfies the required Lipschitz condition.

The reciprocal formula (5.1.10), already known to hold for almost all values of x, now holds for all values of x, since each side is continuous.

If $\alpha = 1$, we obtain similarly

$$\begin{split} \frac{\partial V}{\partial x} &= O\left(\int_{-1}^{1} \frac{|t^2 - y^2|}{(t^2 + y^2)^2} |t| \ dt\right) + O\left(\int_{1}^{\infty} \frac{dt}{t^2}\right) \\ &= O\left(\frac{1}{y} \int_{-1/y}^{1/y} \frac{|u^2 - 1|}{u^2 + 1} |u| \ du\right) + O(1) = O\left(\frac{1}{y} \log \frac{1}{y}\right), \end{split}$$

and it follows that

$$g(x+h)-g(x) = O(|h| \log 1/|h|).$$

5.16. The allied integral. We next return to the allied integral (5.1.7), which is formally equal to g(x). We can now prove the following theorem.

THEOREM 107. Let f(x) belong to $L(-\infty,\infty)$. Then, for any positive α , the allied integral is summable (C,α) to g(x) wherever

$$g(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{f(x+t) - f(x-t)}{t} dt$$
 (5.16.1)

exists, and

$$\int_{0}^{h} |f(x+t) - f(x-t)| dt = o(h); \qquad (5.16.2)$$

and so almost everywhere.

It is plainly sufficient to suppose that $0 < \alpha < 1$.

We have to consider

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda} \right)^{\alpha} du \int_{-\infty}^{\infty} f(x+t) \sin ut \, dt$$

$$= \lim_{\lambda \to \infty} \int_{0}^{\infty} \left\{ f(x+t) - f(x-t) \right\} dt \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda} \right)^{\alpha} \sin ut \, du.$$

Now

$$\int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right)^{\alpha} \sin ut \, du = \lambda \int_{0}^{1} (1 - v)^{\alpha} \sin \lambda t v \, dv$$

$$= -\frac{1}{t} [(1 - v)^{\alpha} \cos \lambda t v]_{0}^{1} + \frac{\alpha}{t} \int_{0}^{1} (1 - v)^{\alpha - 1} \cos \lambda t v \, dv$$

$$= \frac{1}{t} + \frac{\alpha}{\lambda^{\alpha} t^{\alpha + 1}} \int_{0}^{\lambda t} \frac{\cos(\lambda t - w)}{w^{1 - \alpha}} \, dw.$$

Hence

$$\left|\int\limits_{1}^{\lambda}\left(1-\frac{u}{\lambda}\right)^{\alpha}\sin ut\ du-\frac{1}{t}\right|<\frac{K(\alpha)}{\lambda^{\alpha}t^{\alpha+1}}.$$

It follows at once that, if $\delta > 0$,

$$\lim_{\lambda\to\infty}\int_{\delta}^{\infty}\left\{f(x+t)-f(x-t)\right\}dt\int_{0}^{\lambda}\left(1-\frac{u}{\lambda}\right)\sin ut\ du=\int_{\delta}^{\infty}\frac{f(x+t)-f(x-t)}{t}\ dt.$$

Also

$$\int_{1/\lambda}^{\delta} \{f(x+t) - f(x-t)\} dt \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) \sin ut du$$

$$= \int_{1/\lambda}^{\delta} \frac{f(x+t) - f(x-t)}{t} dt + O\left\{\int_{1/\lambda}^{\delta} \frac{|f(x+t) - f(x-t)|}{t^{\alpha+1}\lambda^{\alpha}} dt\right\},$$

and, if (5.16.2) holds, the last term is

$$\begin{split} \left[\frac{1}{\lambda^{\alpha}t^{\alpha+1}}\int_{0}^{t}|f(x+u)-f(x-u)|\,du\right]_{1/\lambda}^{\delta} + \\ + \frac{(\alpha+1)}{\lambda^{\alpha}}\int_{1/\lambda}^{\delta}\frac{dt}{t^{\alpha+2}}\int_{0}^{t}|f(x+u)-f(x-u)|\,du \\ = o\left(1\right) + o\left(\frac{1}{\lambda^{\alpha}}\int_{1/\lambda}^{\delta}\frac{dt}{t^{\alpha+1}}\right) = o\left(1\right) \end{split}$$

by choosing first δ then λ . Finally,

$$\left| \int_{-\infty}^{\lambda} \left(1 - \frac{u}{\lambda} \right)^{\alpha} \sin ut \ du \right| < K\lambda,$$

and

$$\left| \int_{0}^{1/\lambda} \{f(x+t) - f(x-t)\} dt \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) \sin ut du \right|$$

$$< K\lambda \int_{0}^{1/\lambda} |f(x+t) - f(x-t)| dt = o(1).$$

This proves the theorem.

5.17. Application to Fourier Transforms. In this section we make an application of the theory of conjugate functions to the theory of Fourier transforms. There is one respect in which this theory is still incomplete. We have shown that if f(x) belongs to L^p (1) then <math>f(x) has a transform F(x) belonging to $L^{p'}$, and F(x) = 1.i.m. F(x, a).

We have not yet been able to show that the reciprocal relation f(x) = l.i.m. f(x, a), where

$$f(x,a) = \frac{1}{\sqrt{(2\pi)}} \int_{a}^{a} F(t)e^{-ixt} dt,$$

also holds. We can now supply this point.†

† Hille and Tamarkin (3).

THEOREM 108. If f(x) belongs to L^p (1 < p < 2), then f(x) = 1.i.m. f(x, a).

It follows from Parseval's formula (as in the proof of Theorem 58) that

$$f(x,a) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(u) \frac{\sin(x-u)a}{x-u} du$$

$$= \frac{\sin xa}{\pi} P \int_{-\infty}^{\infty} \frac{f(u)\cos ua}{x-u} du - \frac{\cos xa}{\pi} P \int_{-\infty}^{\infty} \frac{f(u)\sin ua}{x-u} du$$

$$= \sin xa \phi_a(x) - \cos xa \psi_a(x)$$

say, the integrals being principal values at u = x. By Theorem 101

$$\int\limits_{-\infty}^{\infty}|\phi_a(x)|^p\ dx< K_p\int\limits_{-\infty}^{\infty}|f(u)\cos ua|^p\ du< K_p\int\limits_{-\infty}^{\infty}|f(u)|^p\ du,$$

and similarly for $\psi_a(x)$. Hence

$$\int\limits_{-\infty}^{\infty}|f(x,a)|^p\ dx\leqslant \int\limits_{-\infty}^{\infty}(2^p|\phi_a(x)|^p+2^p|\psi_a(x)|^p)\ dx< K_p\int\limits_{-\infty}^{\infty}|f(u)|^p\ du.$$

This proves that

$$\int_{-\infty}^{\infty} |f(x)-f(x,a)|^p dx$$

is bounded as $a \to \infty$. We have to prove that in fact it tends to zero. We can construct a step-function $f^*(x)$, zero for |x| > X, and such that

$$\int_{-\infty}^{\infty} |f(x)-f^*(x)|^p dx < \epsilon.$$

Now

$$\left(\int_{-\infty}^{\infty} |f(x)-f(x,a)|^{p} dx\right)^{1/p} \leq \left(\int_{-\infty}^{\infty} |f(x)-f^{*}(x)|^{p} dx\right)^{1/p} + \left(\int_{-\infty}^{\infty} |f^{*}(x)-f^{*}(x,a)|^{p} dx\right)^{1/p} + \left(\int_{-\infty}^{\infty} |f^{*}(x,a)-f(x,a)|^{p} dx\right)^{1/p}$$

$$= J_{1}^{1/p} + J_{2}^{1/p} + J_{3}^{1/p},$$

say. By hypothesis, $|J_1| < \epsilon$; and by the above method

$$|J_3| < K_n |J_1| < K_n \epsilon.$$

Also $f^*(x,a) \to f^*(x)$ boundedly in any finite interval, say $(-\xi,\xi)$;

now if $\xi > 2X$,

$$\int_{\xi}^{\infty} |f^{*}(x) - f^{*}(x, a)|^{p} dx = \int_{\xi}^{\infty} \left| \sqrt{\left(\frac{2}{\pi}\right)} \int_{-X}^{X} f^{*}(t) \frac{\sin(x - t)a}{x - t} dt \right|^{p} dx$$

$$< K_{p} \int_{\xi}^{\infty} \left| \int_{-X}^{X} \frac{|f^{*}(t)|}{x - X} dt \right|^{p} dx < K_{p} \int_{\xi}^{\infty} \frac{dx}{(x - X)^{p}} \left(\int_{-X}^{X} |f^{*}(t)| dt \right)^{p}$$

We can choose ξ so large that this is less than ϵ , for all a; and, having fixed ξ ,

 $\int_{-\xi}^{\xi} |f^*(x) - f^*(x, a)|^p dx \to 0$

by bounded convergence. This completes the proof.

5.18. Further cases of Parseval's formula. We have already seen that (2.1.1) holds if f and G are L^p (1 . If <math>f and g are the given functions, and belong to L^p , $L^{p'}$ respectively, we cannot state the result, because the existence of G is not known. We require an additional condition.

THEOREM 109. If f is L^p (1 < p < 2), and g is L^2 and $L^{p'}$, then

$$\lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} F(x)G(x) \ dx = \int_{-\infty}^{\infty} f(x)g(-x) \ dx.$$

Let G be the transform of g. Then

$$G(x)$$
 $(|x| < \lambda), 0$ $(|x| > \lambda); g(x, \lambda),$

defined as in (3.1.2), are transforms of L^2 ; and the former is also L^p . They are therefore transforms of L^p , $L^{p'}$, and Theorem 75 gives

$$\int_{-\lambda}^{\lambda} F(x)G(x) dx = \int_{-\infty}^{\infty} f(x)g(-x,\lambda) dx.$$

As in the previous section

l.i.m.
$$(p')g(-x,\lambda) = g(-x),$$

and the result follows.

THEOREM 110. If f is L^p (1 $), g is <math>L^{p'}$, and the integral for G is uniformly convergent in any interval $0 < \delta \le x \le \lambda$, then

$$\lim_{\delta \to 0, \lambda \to \infty} \left(\int_{-\lambda}^{-\delta} + \int_{\frac{\lambda}{\delta}}^{\lambda} \right) F(x) G(x) \ dx = \int_{-\infty}^{\infty} f(x) g(-x) \ dx.$$

We can now prove by uniform convergence that

$$G(x)$$
 $(\delta \leqslant |x| \leqslant \lambda),$ 0 $(|x| < \delta, |x| > \lambda)$

is the transform of

$$\frac{1}{\pi} \int_{-\infty}^{\infty} g(u) \frac{\sin \lambda (u-x) - \sin \delta (u-x)}{u-x} du.$$

The proof now goes as before, but to complete it we want

$$\lim_{\delta\to 0}\int_{-\infty}^{\infty}f(x)\,dx\int_{-\infty}^{\infty}g(u)\frac{\sin\delta(u-x)}{u-x}\,du=0.$$

Now

$$\left| \int_{-\infty}^{\infty} g(u) \frac{\sin \delta(u-x)}{u-x} du \right|$$

$$\leq \delta \int_{x-1/\delta}^{x+1/\delta} |g(u)| du + \left(\int_{-\infty}^{x-1/\delta} + \int_{x+1/\delta}^{\infty} \right) \frac{|g(u)|}{|u-x|} du \to 0$$

as $\delta \to 0$, for any fixed x; and, as before, its mean p'th power is bounded. The result therefore follows from the lemma of § 5.12.

UNIQUENESS AND MISCELLANEOUS THEOREMS

6.1. Uniqueness of trigonometrical integrals. The classical uniqueness problem for trigonometrical series is to show that if

$$\frac{1}{2}a_0 + \sum_{n=1}^{\infty} (a_n \cos nx + b_n \sin nx) = 0$$

for all values of x in $(0, 2\pi)$, or all values with some exceptions, then $a_n = 0$, $b_n = 0$ for all values of n.

The corresponding problem for integrals is to show that if

$$\int_{0}^{\infty} \{a(y)\cos xy + b(y)\sin xy\} \, dy = 0 \tag{6.1.1}$$

in some sense or other for all values of x, possibly with some exceptions, then a(y) = 0, b(y) = 0 almost everywhere. A more general problem is to show that if a given function f(x) is represented by a trigonometrical integral,

$$\int_{0}^{\infty} \{a(y)\cos xy + b(y)\sin xy\} dy = f(x), \qquad (6.1.2)$$

in some sense, then the integral is necessarily of the Fourier form, i.e. in some sense

$$a(y) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(x) \cos xy \ dx, \qquad b(y) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(x) \sin xy \ dx.$$

Owing to the symmetry of the Fourier integral formula between a function and its transform, in the integral case this is not, formally, a new problem. It simply amounts to the question whether a(x) and b(x) are representable by Fourier integrals; and in some cases the answer follows from theorems which we have already proved.

Suppose, for example, that a(x) and b(x) belong to $L(0,\infty)$, and that

$$\int_{0}^{\infty} \{a(y)\cos xy + b(y)\sin xy\} dy = 0$$

for almost all values of x. Adding and subtracting the formulae with x and -x, it follows that both

$$\int_{0}^{\infty} a(y)\cos xy \ dy = 0, \quad \int_{0}^{\infty} b(y)\sin xy \ dy = 0$$

for almost all x. By Theorem 14 (for an even function)

$$\lim_{\lambda \to \infty} \frac{2}{\pi} \int_{0}^{\lambda} \left(1 - \frac{u}{\lambda}\right) \cos xu \ du \int_{0}^{\infty} a(y) \cos uy \ dy = a(x)$$

for almost all x, and in this case the left-hand side is 0 for all x. Hence a(x) = 0 almost everywhere. Similarly b(x) = 0 almost everywhere.

Theorem 22 can be used to give the same result.

Suppose, again, that a(y) belongs to $L^2(0,\infty)$, and that

$$\int_{0}^{+\infty} a(y)\cos xy \ dy = 0$$

for almost all values of x. Since the limit and mean limit of a sequence, if they both exist, are equal almost everywhere, the cosine transform of a(x), in the sense of Theorem 48, is null, and hence a(x) is the mean limit of a sequence of null functions, and is therefore null.

The uniqueness theory of Fourier series suggests a different type of theorem, in which the possible values of x for which (6.1.1) fails are much more restricted, but in which a(x) and b(x) do not necessarily belong to L-classes. The main difference between the theory for series and that for integrals is that the convergence of $\sum a_n \cos nx$, for example, in a set of positive measure, implies that $a_n \to 0$; but the convergence of

 $\int_{0}^{\infty} a(y)\cos xy \ dy$

does not imply that $a(x) \to 0$ as $x \to \infty$; for example, the integral is convergent if $a(x) = e^x \cos e^{2x}$.

6.2. The expression

$$\lim_{h \to 0} \frac{\Phi(x+h) + \Phi(x-h) - 2\Phi(x)}{h^2}$$
 (6.2.1)

is called the generalized second derivative of $\Phi(x)$. The uniqueness theory of Fourier series depends on the theorem of Schwarz (see Titchmarsh, Theory of Functions, § 13.84), that if $\Phi(x)$ is continuous, and has at all points of an interval the generalized second derivative 0, then $\Phi(x)$ is a linear function in the interval. Here we shall proceed at once to the general problem with f(x), and use

THEOREM 111.† Let $\Phi(x)$ be continuous in (a,b), and have at every point of this interval a finite generalized second derivative f(x), which

[†] De la Vallée-Poussin, Cours d'analyse infinitésimale, 1914 ed.

belongs to L(a,b). Then

$$\Phi(x) = \int_{a}^{x} du \int_{a}^{u} f(v) dv + a_{0} + a_{1}x \quad (a \leqslant x \leqslant b), \qquad (6.2.2)$$

where a_0 and a_1 are constants.

We first prove two lemmas.

LEMMA 1. Let $\Phi(x)$ be continuous in (a,b), and let

$$\frac{\overline{\lim}_{h \to 0} \Phi(x+h) + \Phi(x-h) - 2\Phi(x)}{h^2}$$
 (6.2.3)

be $\geqslant 0$ for every x in (a,b). Then no part of any arc of the curve $y = \Phi(x)$ can be above its chord.

Suppose that some points of an arc (x_1, x_2) lie above the chord, PQ say. Let

$$\Phi_{\epsilon}(x) = \Phi(x) + \frac{1}{2}\epsilon(x-x_1)(x-x_2) \quad (\epsilon > 0).$$

Then, if ϵ is small enough, some points of the corresponding arc of $y = \Phi_{\epsilon}(x)$ will lie above the chord. Let M be such a point of this curve, whose distance from PQ is not less than that of any other such point. Let x be the abscissa of M. Then, if λ is the tangent of the angle which PQ makes with the x-axis,

$$\frac{\Phi_{\epsilon}(x+h) - \Phi_{\epsilon}(x)}{h} \leqslant \lambda, \qquad \frac{\Phi_{\epsilon}(x) - \Phi_{\epsilon}(x-h)}{h} \geqslant \lambda.$$

Hence

$$\Phi_{\epsilon}(x+h) + \Phi_{\epsilon}(x-h) - 2\Phi_{\epsilon}(x) \leqslant 0,$$

i.e.

$$\Phi(x+h)+\Phi(x-h)-2\Phi(x) \leqslant -\epsilon$$

for all small h. This contradicts the hypothesis, and the result follows.

LEMMA 2. Let $\Phi(x)$ be continuous in (a,b), and let (6.2.3) be ≥ 0 for almost all x in (a,b) and be nowhere $-\infty$. Then no part of any arc of the curve $y = \Phi(x)$ can be above its chord.

If (6.2.3) is nowhere < 0, the result follows from the previous lemma. Otherwise, let E be the set, of measure 0, where (6.2.3) < 0. Let $\uparrow \chi(x)$ be a non-decreasing absolutely continuous function such that $\chi'(x) = + \infty$ in E, and $\chi(b) - \chi(a) < \epsilon$. Let

$$\chi_1(x) = \int_a^x \chi(u) \ du.$$

If x is a point of E, and M any positive number, however large,

$$\frac{\chi(x+u)-\chi(x-u)}{2u}\geqslant M \quad (|u|\leqslant \delta).$$

Hence, if $h \leq \delta$,

$$rac{\chi_1(x+h)+\chi_1(x-h)-2\chi_1(x)}{h^2} = rac{1}{h^2} \int_0^h \{\chi(x+u)-\chi(x-u)\} du$$

$$\geqslant rac{1}{h^2} \int_0^h 2uM \ du = M,$$

and so the left-hand side tends to infinity as $h \to 0$.

Let $\Omega(x) = \Phi(x) + \chi_1(x)$.

Then $\overline{\lim_{h\to 0}} \frac{\Omega(x+h) + \Omega(x-h) - 2\Omega(x)}{h^2} \geqslant 0$

for every x in (a, b). Hence no part of any arc of $y = \Omega(x)$ can be above its chord. Since this is true for arbitrarily small ϵ , the same result follows for $y = \Phi(x)$.

Proof of Theorem 111.

Let $p(x) = \min\{f(x), n\}, \quad q(x) = \max\{f(x), -n\}.$

Then $p(x) \leqslant f(x) \leqslant q(x)$, and, since f is integrable, so are p and q. Let

$$p_1(x) = \int_a^x p(u) du, \qquad p_2(x) = \int_a^x p_1(u) du,$$

and similarly for q.

Then $q_2(x) - \Phi(x)$ has almost everywhere the generalized second derivative $q(x) - f(x) \ge 0$, and

$$\frac{q_2(x+h)+q_2(x-h)-2q_2(x)}{h^2} = \frac{1}{h^2} \int_0^h du \int_{x-u}^{x+u} q(v) dv$$

$$\geqslant \frac{1}{h^2} \int_0^h du \int_{x-u}^{x+u} (-n) dv = -n.$$

Hence the generalized second derivative of $q_2(x) - \Phi(x)$ is nowhere $-\infty$. Hence no arc of $y = q_2(x) - \Phi(x)$ is above its chord.

The chord through the end-points a and b is

$$y = \frac{x-a}{b-a} \{q_2(b) - \Phi(b) + \Phi(a)\} - \Phi(a),$$

and hence

$$q_2(x) - \Phi(x) \leqslant \frac{x-a}{b-a} \{q_2(b) - \Phi(b) + \Phi(a)\} - \Phi(a) \quad (a \leqslant x \leqslant b).$$

Similarly, no arc of $y = p_2(x) - \Phi(x)$ is below its chord, and it follows that

$$p_2(x)-\Phi(x)\geqslant \frac{x-a}{b-a}\{p_2(b)-\Phi(b)+\Phi(a)\}-\Phi(a)\quad (a\leqslant x\leqslant b).$$

Making $n \to \infty$, $p_2(x)$ and $q_2(x)$ both tend to the limit

$$f_2(x) = \int_a^x du \int_a^u f(v) dv.$$

Hence

$$f_2(x)-\Phi(x)=\frac{x-a}{b-a}\{f_2(b)-\Phi(b)+\Phi(a)\}-\Phi(a)\quad (a\leqslant x\leqslant b),$$

the desired result.

6.3. THEOREM 112. Let a(y) and b(y) be integrable over any finite interval, and zero in an interval containing the origin. Let

$$\int_{0}^{\infty} \{a(y)\cos xy + b(y)\sin xy\} dy = f(x)$$

for all x in a certain interval. Then

$$\Phi(x) = -\int_{0}^{\infty} \{a(y)\cos xy + b(y)\sin xy\} \frac{dy}{y^{2}}$$

exists for every x of the interval, and has the generalized second derivative f(x).

The convergence of the integral for $\Phi(x)$ follows from the second mean-value theorem. Now

$$\frac{\cos(x+h)y+\frac{\cos(x-h)y-2\cos xy}{\sin xy}=-4\sin^{2}\frac{1}{2}hy\frac{\cos xy}{\sin xy}.$$

Hence

$$\frac{\Phi(x+h)+\Phi(x-h)-2\Phi(x)}{h^2} = \int_0^{+\infty} \{a(y)\cos xy + b(y)\sin xy\} \frac{4\sin^2\frac{1}{2}hy}{h^2y^2} dy,$$
(6.3.1)

and it is sufficient to prove that this integral converges uniformly with respect to h for $h \ge 0$.

Let
$$\int_{Y}^{+\infty} \{a(y)\cos xy + b(y)\sin xy\} dy = r(Y),$$

so that $|r(Y)| \leqslant \epsilon$ for $Y \geqslant Y_0(\epsilon)$. Then

$$\left| \int_{Y_{\bullet}}^{\infty} \left\{ a(y)\cos xy + b(y)\sin xy \right\} \frac{4\sin^2\frac{1}{2}hy}{h^2y^2} \, dy \right|$$

$$= \left| \left[-r(y) \frac{4\sin^2\frac{1}{2}hy}{h^2y^2} \right]_{Y_{\bullet}}^{\infty} + \int_{Y_{\bullet}}^{\infty} r(y) \, \frac{d}{dy} \left(\frac{4\sin^2\frac{1}{2}hy}{h^2y^2} \right) \, dy \right|$$

$$\leq \epsilon \frac{4\sin^2\frac{1}{2}hY_0}{h^2Y_0^2} + \epsilon \int_{Y_{\bullet}}^{\infty} \left| \frac{d}{dy} \left(\frac{4\sin^2\frac{1}{2}hy}{h^2y^2} \right) \right| \, dy$$

$$< \epsilon + \epsilon \int_{0}^{\infty} \left| \frac{d}{du} \left(\frac{4\sin^2\frac{1}{2}u}{u^2} \right) \right| \, du = A\epsilon$$

for all h > 0; hence the result.

6.4. We shall now prove the uniqueness theorem on the assumption that $a(y)/(1+y^2)$ and $b(y)/(1+y^2)$ belong to $L(0,\infty)$. Later it will be shown that this condition is superfluous.

Theorem 113.† Let $a(y)/(1+y^2)$ and $b(y)/(1+y^2)$ belong to $L(0,\infty)$, and let

$$\int_{0}^{+\infty} \{a(y)\cos xy + b(y)\sin xy\} \, dy = f(x) \tag{6.4.1}$$

for all values of x, where f(x) is everywhere finite and integrable over any finite interval. Then for almost all positive values of y

$$a(y) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda} \right) f(x) \cos xy \ dx, \tag{6.4.2}$$

$$b(y) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda} \right) f(x) \sin xy \ dx. \tag{6.4.3}$$

In particular, if f(x) = 0, then a(y) = 0, b(y) = 0 almost everywhere.

The condition (6.4.1) may be broken for a finite set of values of x, provided that $a(y) \to 0$, $b(y) \to 0$, as $y \to \infty$.

By replacing x by -x in (6.4.1) and adding or subtracting, we obtain similar formulae with the cosine or sine integral only. We may therefore consider them separately.

† Pollard (3), Jacob (2). The unrestricted result follows from Offord (7); the proof given here is by Offord and the author.

Suppose first then that b(y) = 0. Suppose also that a(y) = 0 for $0 < y < \delta$, and let

$$\Phi(x) = -\int_{0}^{\infty} \frac{a(y)}{y^2} \cos xy \, dy. \tag{6.4.4}$$

By Theorem 112, $\Phi(x)$ has the generalized second derivative f(x), and here f(x) is integrable. Hence, by Theorem 111,

$$\Phi(x) = \int_0^x du \int_0^u f(v) dv + p + qx,$$

where p and q are constants; and q = 0 in this case, since Φ and f are even. Writing

$$f_1(u) = \int_0^u f(v) dv,$$

we have

$$\Phi(x) = \int_0^x f_1(u) du + p,$$

and

$$\int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) f(x) \cos xy \, dx = \int_{0}^{\lambda} f_{1}(x) \left\{\frac{\cos xy}{\lambda} + \left(1 - \frac{x}{\lambda}\right) y \sin xy\right\} dx$$

$$= \left[\Phi(x) \left\{\frac{\cos xy}{\lambda} + \left(1 - \frac{x}{\lambda}\right) y \sin xy\right\}\right]_{0}^{\lambda} + \left(1 - \frac{x}{\lambda}\right) \Phi(x) \left\{\frac{2y \sin xy}{\lambda} - y^{2} \left(1 - \frac{x}{\lambda}\right) \cos xy\right\} dx$$

$$= \frac{\Phi(\lambda) \cos \lambda y}{\lambda} - \frac{p}{\lambda} + \frac{2y}{\lambda} \int_{0}^{\lambda} \Phi(x) \sin xy \, dx - y^{2} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) \Phi(x) \cos xy \, dx.$$
(6.4.5)

Since $a(y)/y^2$ is L, (6.4.4) and Theorem 14 give

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda} \right) \Phi(x) \cos xy \ dx = -\frac{1}{2} \pi \frac{a(y)}{y^2}$$
 (6.4.6)

almost everywhere. Also $\Phi(x) \to 0$ as $x \to \infty$, by Theorem 1, so that the remaining terms in (6.4.5) tend to 0 as $\lambda \to \infty$. Hence

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) f(x) \cos xy \ dx = \frac{1}{2} \pi a(y), \tag{6.4.7}$$

the required result with the conditions stated.

To remove the restriction that a(y) = 0 over $(0, \delta)$, let

$$a_{\delta}(x) = a(x) \quad (x \geqslant \delta), \qquad 0 \quad (x < \delta),$$

$$\int_{0}^{\delta} a(y) \cos xy \ dy = \chi(x).$$

and let

Then the result already obtained shows that

$$\lim_{\lambda\to\infty}\int\limits_0^\lambda \left(1-\frac{x}{\lambda}\right)\{f(x)-\chi(x)\}\cos xy\ dx=\tfrac{1}{2}\pi a_\delta(y)$$

almost everywhere. Also, by Theorem 14,

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda} \right) \chi(x) \cos xy \ dx = 0$$

for almost all y in (δ, ∞) . Hence

$$\lim_{\lambda \to \infty} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) f(x) \cos xy \ dx = \frac{1}{2} \pi a(y)$$

for almost all y in (δ, ∞) , and so, since δ is arbitrary, for almost all y in $(0, \infty)$. This is the required theorem for the cosine integral.

Next let a(y) = 0, and suppose that b(y) = 0 in $(0, \delta)$. Let

$$\Psi(x) = -\int_{0}^{\infty} \frac{b(y)}{y^2} \sin xy \, dy. \tag{6.4.8}$$

Arguing as before, we obtain

$$\Psi(x) = \int_{0}^{x} du \int_{0}^{x} f(v) dv + qx,$$
and
$$\int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) f(x) \sin xy \ dx = \int_{0}^{\lambda} \left\{f_{1}(x) + q\right\} \left\{\frac{\sin xy}{\lambda} - \left(1 - \frac{x}{\lambda}\right)y \cos xy\right\} dx$$

$$= \frac{\Psi(\lambda) \sin \lambda y}{\lambda} - \frac{2y}{\lambda} \int_{0}^{\lambda} \Psi(x) \cos xy \ dx - y^{2} \int_{0}^{\lambda} \left(1 - \frac{x}{\lambda}\right) \Psi(x) \sin xy \ dx.$$

The proof now concludes as in the cosine case.

If there are exceptional points where (6.4.1) does not hold, the argument merely shows that

$$y = \Phi(x) - \int_{-\infty}^{\infty} du \int_{-\infty}^{u} f(v) dv \qquad (6.4.10)$$

is a linear function in the intervals between these points. But now, if

$$|a(y)| \leqslant \epsilon$$
, $|b(y)| \leqslant \epsilon$ for $y \geqslant \Delta$,

$$\left| \int_0^\infty \left\{ a(y)\cos xy + b(y)\sin xy \right\} \frac{4\sin^2\frac{1}{2}hy}{h^2y^2} \, dy \right|$$

$$\leq \int_0^\Delta \left\{ |a(y)| + |b(y)| \right\} \, dy + 2\epsilon \int_\Delta^\infty \frac{4\sin^2\frac{1}{2}hy}{h^2y^2} \, dy$$

$$< K(\Delta) + A\epsilon/h.$$

Multiplying (6.3.1) by h, and choosing first ϵ and then h, it follows that

 $\lim_{h\to 0}\left\{\frac{\Phi(x+h)-\Phi(x)}{h}-\frac{\Phi(x)-\Phi(x-h)}{h}\right\}=0.$

Taking x to be one of the exceptional points, it follows that the slopes of the straight lines which make up the graph of (6.4.10) are the same on each side of the point. Hence (6.4.10) is a single linear function, and the result then follows as before.

6.5. To remove the restriction on a(y) and b(y) we require some more preliminary theorems.

Theorem 114. If
$$\int_{0}^{+\infty} f(t)\cos yt \ dt$$

converges uniformly in any finite interval, to $\sqrt{(\frac{1}{2}\pi)}F_c(y)$ say, then

$$\lim_{\lambda \to \infty} \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\lambda} \left(1 - \frac{y}{\lambda}\right) F_{c}(y) \cos xy \ dy = f(x)$$

for almost all x.

This is merely a variant of Theorem 20; the proof is substantially the same, depending on the particular case of the data, that

$$\int_{0}^{\infty} f(t) dt$$

exists.

We require a similar theorem for the sine integral, but in this case the argument is more complicated.

Theorem 115.† Let x_n be a sequence of numbers tending to infinity, such that $x_n \geqslant k^{n-1}x_{n-1} \quad (n=2,3,...),$

where k > 1. Then, given any interval (α, β) , there is a number Ω in (α, β) , and a sequence of integers $y_1, y_2, ...$, such that

$$x_n \Omega - (2y_n + 1) \to 0.$$
 (6.5.1)

Suppose that $0 < \alpha < \beta$, and divide (α, β) into three equal parts (α, γ) , (γ, δ) , (δ, β) . Let x_{ν} be the first of the x_n which is greater than both

 $\frac{3}{(k-1)(\beta-\alpha)}$ and $\frac{6}{\beta-\alpha}$.

Choose y_{ν} so that $(2y_{\nu}+1)/x_{\nu}$ falls in (γ, δ) . This is possible, since $x_{\nu} > 6/(\beta-\alpha)$. Then determine $y_{\nu+1}, y_{\nu+2}, ...$ so that

$$\left| (2y_{n+1}+1)-(2y_n+1)\frac{x_{n+1}}{x_n} \right| \leqslant 1 \quad (n=\nu,\nu+1,\ldots).$$

If in any case there is more than one such y_{n+1} , take the least; the process is then unique; $y_1, ..., y_{\nu-1}$ can have any values.

The numbers x_n and y_n now determine a sequence of fractions $(2y_n+1)/x_n$, which tend to a limit Ω ; for, by the above inequality,

$$\left| \frac{2y_{n+m}+1}{x_{n+m}} - \frac{2y_n+1}{x_n} \right| \leqslant \frac{1}{x_{n+1}} + \dots + \frac{1}{x_{n+m}} \to 0.$$

Also

$$\left|\Omega - \frac{2y_n + 1}{x_n}\right| \leqslant \frac{1}{x_{n+1}} + \frac{1}{x_{n+2}} + \dots \leqslant \frac{1}{x_n} \left(\frac{1}{k^n} + \frac{1}{k^{2n}} + \dots\right) = \frac{1}{x_n(k^n - 1)},$$

so that (6.5.1) follows. Finally, Ω is in (α, β) , since

$$\left|\Omega - \frac{2y_{\nu}+1}{x_{\nu}}\right| \leqslant \frac{1}{x_{\nu}(k^{\nu}-1)} \leqslant \frac{1}{x_{\nu}(k-1)} < \frac{1}{3}(\beta - \alpha).$$

THEOREM 116. If $\int_{0}^{+\infty} \phi(u) \sin xu \ du$

is convergent for all values of x, then as $n \to \infty$

$$r_n = \max_{0 \leqslant \xi \leqslant 1} \int_{u}^{n+\xi} \phi(u) \ du \to 0.$$

If the theorem is false, there is a positive ϵ and a sequence n_{ν} such that $|r_{n_{\nu}}| \ge \epsilon$; and from this sequence we can select a sub-sequence n_{μ} satisfying $n_{\mu} \ge 2^{\mu-1}n_{\mu-1}$. Hence there is an x in $(0, \frac{1}{8}\pi)$, and integers $y_1, y_2, ...$, such that

$$2xn_{\mu}/\pi - (2y_{\mu} + 1) \rightarrow 0.$$

Hence, if μ is large enough,

$$|xn_{\mu}-(y_{\mu}+\tfrac{1}{2})\pi|\leqslant \tfrac{1}{8}\pi.$$

4362

Hence, for $n_{\mu} \leq u \leq n_{\mu} + \xi$, where $\xi \leq 1$, $xu - y_{\mu}\pi$ lies between $\frac{1}{4}\pi$ and $\frac{3}{4}\pi$; hence $\sin xu$ is monotonic in at most two intervals, and $|\sin xu| \geq 1/\sqrt{2}$. If e.g. $\sin xu$ is steadily increasing,

$$\int_{n_{\mu}}^{n_{\mu}+\xi} \phi(u) \ du = \int_{n_{\mu}}^{n_{\mu}+\xi} \frac{\phi(u)\sin xu}{\sin xu} \ du = \frac{1}{\sin xn_{\mu}} \int_{n_{\mu}}^{n_{\mu}+\xi'} \phi(u)\sin xu \ du$$

by the second mean-value theorem, and the right-hand side tends to 0. Similarly in other cases.

THEOREM 117. If
$$\int_{0}^{\infty} f(t)\sin yt \ dt$$
 (6.5.2)

converges uniformly in every finite interval, to $\sqrt{(\frac{1}{2}\pi)}F_s(y)$ say, then

$$\lim_{\lambda \to \infty} \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\lambda} \left(1 - \frac{y}{\lambda}\right) F_s(y) \sin xy \ dy = f(x)$$
 (6.5.3)

for almost all x.

We can insert (6.5.2) in (6.5.3) and invert, by uniform convergence. We obtain

$$\frac{1}{\pi} \int_{0}^{\infty} f(t) \left\{ \frac{1 - \cos \lambda(x - t)}{\lambda(x - t)^2} - \frac{1 - \cos \lambda(x + t)}{\lambda(x + t)^2} \right\} dt. \tag{6.5.4}$$

Let T > x, and consider

$$I = \int_{T}^{+\infty} \frac{f(t)\cos \lambda t}{\lambda (x-t)^2} dt = \left(\int_{T}^{\frac{1}{2}N\pi/\lambda} + \sum_{n=N}^{\infty} \int_{\frac{1}{2}n\pi/\lambda}^{\frac{1}{2}(n+1)\pi/\lambda} \right) \frac{f(t)\cos \lambda t}{\lambda (x-t)^2} dt,$$

where N is the integer next above $2\lambda T/\pi$. By the second mean-value theorem

$$\int_{\frac{1}{2}n\pi/\lambda}^{\frac{1}{2}(n+1)\pi/\lambda} \frac{f(t)\cos\lambda t}{\lambda(x-t)^2} dt = O\left\{\frac{1}{\lambda(n/\lambda)^2} \int_{\xi}^{\eta} f(t) dt\right\},\,$$

where $\frac{1}{2}n\pi/\lambda \leqslant \xi < \eta \leqslant \frac{1}{2}(n+1)\pi/\lambda$. By Theorem 116 the last integral tends to 0 as $n \to \infty$, uniformly for $\lambda > 2/\pi$. Hence

$$I = o\left(\frac{\lambda}{N^2} + \sum_{n=N}^{\infty} \frac{\lambda}{n^2}\right) = o\left(\frac{\lambda}{N}\right) = o\left(\frac{1}{T}\right)$$

uniformly with respect to λ . Similar arguments apply to the rest of (6.5.4) with $t \ge T$. Also, for a fixed T, the part with t < T tends to f(x) almost everywhere. The result therefore follows.

6.6. THEOREM 118. The results of Theorem 113 hold if a(y) and b(y) are integrable over any finite interval, and (6.4.1) holds.

We again define $\Phi(x)$ by (6.4.4), but now the integral is not necessarily absolutely convergent, and $\Phi(x)$ does not necessarily tend to 0 at infinity. However, since

$$a_1(y) = \int\limits_0^y a(u) \ du \to \text{limit}$$

as $y \to \infty$ (by putting x = 0 in (6.4.1)), and

$$\int_{0}^{Y} \frac{a(y)}{y^{2}} \cos xy \ dy = \frac{a_{1}(Y) \cos xY}{Y^{2}} + \int_{0}^{Y} a_{1}(y) \left(\frac{x \sin xy}{y^{2}} + \frac{2 \cos xy}{y^{3}} \right) dy,$$

the integral (6.4.4) converges uniformly over any finite interval. We therefore deduce (6.4.6) from Theorem 114. Also

$$\Phi(x) = -\int_{0}^{\infty} a_1(y) \left\{ \frac{x \sin xy}{y^2} + \frac{2 \cos xy}{y^3} \right\} dy = o(x)$$

as $x \to \infty$, by Theorem 1; and, for a fixed y,

$$\int_{0}^{\lambda} \Phi(x)\sin xy \, dx = \int_{0}^{\lambda} \left\{ o\left(1\right) - \int_{y+1}^{\infty} a_{1}(u) \frac{x \sin xu}{u^{2}} \, du \right\} \sin xy \, dx$$

$$= o\left(\lambda\right) - \frac{1}{2} \int_{y+1}^{\infty} \frac{a_{1}(u)}{u^{2}} \left\{ \frac{\lambda \sin \lambda(u-y)}{u-y} - \frac{1 - \cos \lambda(u-y)}{u-y} - \frac{\lambda \sin \lambda(u+y)}{u+y} + \frac{1 - \cos \lambda(u+y)}{u+y} \right\} du = o\left(\lambda\right)$$

as $\lambda \to \infty$. Hence (6.4.7) again follows from (6.4.5).

In the sine case we obtain (6.4.8) as before, but now we get no result by putting x = 0 in (6.4.1), and we have to use Theorems 115-17.

We have

$$\int_{y_1}^{y_2} \frac{b(y)}{y^2} \sin xy \ dy = \left(\int_{y_1}^{\frac{1}{2}m\pi/x} + \sum_{\nu=m}^{n} \int_{\frac{1}{2}\nu\pi/x}^{\frac{1}{2}(\nu+1)\pi/x} + \int_{\frac{1}{2}(n+1)\pi/x}^{y_2} \right) \frac{b(y)}{y^2} \sin xy \ dy.$$

The second mean-value theorem gives

$$\int_{\frac{1}{2}\nu\pi/x}^{\frac{1}{2}(\nu+1)\pi/x}\frac{b(y)}{y^2}\sin xy\ dy = O\left(\frac{x^2}{\nu^2}\int_{\xi}^{\eta}b(y)\ dy\right),$$

where $\frac{1}{2}\nu\pi/x \leqslant \xi < \eta \leqslant \frac{1}{2}(\nu+1)\pi/x$; and the last integral is o(1) as $\nu \to \infty$, $x \to \infty$, and is o(1/x) as $\nu \to \infty$, $x \to 0$ (by considering O(1/x) terms of type r_n). Hence as x tends to 0 or ∞

$$\int_{y_1}^{y_2} \frac{b(y)}{y^2} \sin xy \ dy = o\left\{x^2\left(1+\frac{1}{x}\right)\frac{1}{m}\right\} = o\left(\frac{x+1}{y_1}\right).$$

Hence (6.4.8) converges uniformly over any finite interval, and

$$\Psi(x)=o\left(x\right)$$

as $x \to \infty$. Also

$$\int_{0}^{\lambda} \Psi(x) \cos xy \ dx$$

$$= o(\lambda) - \frac{1}{2} \int_{u}^{\infty} \frac{b(u)}{u^2} \left\{ \frac{1 - \cos \lambda(u+y)}{u+y} + \frac{1 - \cos \lambda(u-y)}{u-y} \right\} du = o(\lambda)$$

as $\lambda \to \infty$, by an argument similar to that just used for $\Phi(x)$. The result now follows as in the previous case.

We can also state the above result as a direct theorem on Fourier's integral formula.

THEOREM 119. Let f(x) be integrable over any finite interval, and let

$$\lim_{\mu \to \infty} \int_{-\mu}^{\mu} f(t) \cos xt \ dt, \qquad \lim_{\mu \to \infty} \int_{-\mu}^{\mu} f(t) \sin xt \ dt$$

have finite values for every x, and let these values be integrable over any finite interval. Then

$$f(x) = \lim_{\lambda \to \infty} \frac{1}{\pi} \int_{-\lambda}^{\lambda} \left(1 - \frac{|u|}{\lambda} \right) \left(\lim_{\mu \to \infty} \int_{-\mu}^{\mu} f(t) \cos u(x-t) dt \right) du.$$

6.7. Integrals in the complex form. The result in this form is Theorem 120. Let F(y) be integrable over any finite interval, and let

$$\lim_{\lambda \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-\lambda}^{\lambda} F(y)e^{-ixy} dy = f(x)$$
 (6.7.1)

for all values of x, where f(x) is everywhere finite and integrable over any finite interval. Then for almost all y

$$F(y) = \lim_{\lambda \to \infty} \frac{1}{\sqrt{(2\pi)}} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda}\right) f(x) e^{ixy} dx. \tag{6.7.2}$$

In particular, if f(x) = 0, then F(y) = 0 almost everywhere.

Since

$$\int_{-\lambda}^{\lambda} F(y)e^{-ixy} dy$$

$$= 2\int_{0}^{\lambda} \left[\{F(y) + F(-y)\}\cos xy - i\{F(y) - F(-y)\}\sin xy \right] dy,$$

the theorem is equivalent to Theorem 118.

There is a specially simple argument for the case f(x) = 0. Let

$$F_1(x) = \int_0^x F(u) \, du.$$

Then

$$\int_{-\lambda}^{\lambda} F(y)e^{-ixy} dy = F_1(\lambda)e^{-ix\lambda} - F_1(-\lambda)e^{ix\lambda} + ix\int_{-\lambda}^{\lambda} F_1(y)e^{-ixy} dy.$$

Replacing x by -x, and adding,

$$\int_{-\lambda}^{\lambda} F(y) \cos xy \ dy = \{F_1(\lambda) - F_1(-\lambda)\} \cos x\lambda + x \int_{-\lambda}^{\lambda} F_1(y) \sin xy \ dy.$$
Now
$$F_1(\lambda) - F_1(-\lambda) = \int_{\lambda}^{\lambda} F(y) \ dy \to 0$$

Now

as $\lambda \to \infty$, as a particular case of the data. Hence

$$\lim_{\lambda \to \infty} x \int_{-\lambda}^{\lambda} F_1(y) \sin xy \ dy = 0,$$

and so

$$\lim_{\lambda\to\infty}\int\limits_0^\lambda \{F_1(y)-F_1(-y)\}\sin xy\ dy=0$$

for $x \neq 0$, by the above argument; and for x = 0 because the integrand is 0.

We deduce that
$$F_1(y) - F_1(-y) = 0$$

from Theorem 118; but now b(y) is bounded, and all that we want about $\Psi(x)$ is obvious from its definition. Hence

$$F(y)+F(-y)\equiv 0.$$

The argument applies equally well with $F(y)e^{i\xi y}$, with any ξ , instead of F(y). Hence

$$F(y)e^{i\xi y}+F(-y)e^{-i\xi y}\equiv 0.$$

for every y and ξ . Hence $F(y) \equiv 0$.

Offord (7) has recently proved the remarkable theorem that if F(y) is integrable over every finite interval, and (6.7.1) is summable (C, 1) to 0 for all x, i.e.

 $\lim_{\lambda \to 0} \int_{-\lambda}^{\lambda} \left(1 - \frac{|y|}{\lambda} \right) F(y) e^{-ixy} dy = 0$

for all x, then F(y) = 0 almost everywhere.

This theorem is a 'best possible' both in the sense that one exceptional point is sufficient to render the conclusion false, and in the sense that it is not possible to replace (C, 1) by $(C, 1+\delta)$ for any positive δ . Thus

$$\int_{-\infty}^{\infty} e^{ixu} du = 0 \quad (C, 1) \quad (x \neq 0),$$

$$\int_{-\infty}^{\infty} u e^{ixu} du = 0 \quad (C, 1+\delta)$$

and

for all x.

6.8. Parseval's formula. The above results enable us to prove still another theorem on Parseval's formula.

Suppose that f and g are given functions, f belongs to $L(-\infty,\infty)$, G exists as the transform of g in some sense or other, and G is $L(-\infty,\infty)$. We are unable to use Theorem 35, since we do not know that g is the transform of G. We have, however, the following theorem.

THEOREM 121. Let f(x) be $L(-\infty,\infty)$, g(x) integrable over any finite interval, and let

$$G(x) = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} g(t)e^{ixt} dt$$
 (6.8.1)

for all x, and let G(x) be everywhere finite, and $L(-\infty,\infty)$. Then (2.1.1) holds.

The convergence of (6.8.1) may fail for a finite set of values of x, provided that $g(x) \to 0$ as $x \to \pm \infty$.

For g(x) is the transform of G(x), by Theorem 120, and the result therefore follows from Theorem 35.

6.9. Another uniqueness theorem. We shall next prove a uniqueness theorem of a different type, in which (6.4.1) is not necessarily convergent, but in which there is an additional condition.

Theorem 122.† Let $e^{-yt}a(t)$, $e^{-yt}b(t)$ belong to $L(0,\infty)$ for every positive y, and let

$$U(x,y) = \int_{0}^{\infty} \{a(t)\cos xt + b(t)\sin xt\}e^{-yt} dt$$

be bounded for y > 0 and all x. Then

$$\lim_{y\to 0}U(x,y)=f(x)$$

exists and is finite for almost all values of x, and

$$a(t) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda} \right) f(x) \cos xt \ dx,$$

$$b(t) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda} \right) f(x) \sin xt \ dx,$$

for almost all values of t.

Let
$$\Phi(z) = \int_{0}^{\infty} \{a(t) - ib(t)\} e^{izt} dt$$

be the analytic function of which U(x,y) is the real part, and let $\Psi(z) = \exp\{-\Phi(z)\}$. Then $|\Psi(z)|$ is bounded above and below. Hence, by Theorem 94, $\Psi(z)$ tends to a finite non-zero limit for almost all x, as $y \to 0$. Hence $\Phi(z)$ tends to a finite limit for almost all x, and hence so does U(x,y). Let the limit of U(x,y) be f(x). Now for y > 0, $\eta > 0$

$$\begin{split} \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{\eta}{(\xi - x)^2 + \eta^2} U(x, y) \, dx &= \frac{1}{\pi} \int_{0}^{\infty} e^{-yt} a(t) \, dt \int_{-\infty}^{\infty} \frac{\eta \cos xt}{(\xi - x)^2 + \eta^2} \, dx \, + \\ &\quad + \frac{1}{\pi} \int_{0}^{\infty} e^{-yt} b(t) \, dt \int_{-\infty}^{\infty} \frac{\eta \sin xt}{(\xi - x)^2 + \eta^2} \, dx \\ &= \int_{0}^{\infty} e^{-yt} \{a(t) e^{-\eta t} \cos \xi t + b(t) e^{-\eta t} \sin \xi t\} \, dt = U(\xi, y + \eta), \end{split}$$

the inversion being justified by absolute convergence. Making $y \rightarrow 0$,

$$U(\xi,\eta) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{\eta}{(\xi-x)^2 + \eta^2} f(x) dx$$

by dominated convergence.

By Theorem 14,

em 14,

$$e^{-yt}a(t) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \left(1 - \frac{|x|}{\lambda}\right) U(x, y) \cos xt \, dx$$

$$= \frac{1}{\pi} \lim_{\lambda \to \infty} \frac{1}{\lambda} \int_{0}^{\lambda} d\xi \int_{-\xi}^{\xi} U(x, y) \cos xt \, dx \qquad (6.9.1)$$

for any positive y, for almost all values of t. Now

$$\int_{-\xi}^{\xi} U(x,y)\cos xt \, dx = \frac{1}{\pi} \int_{-\xi}^{\xi} \cos xt \, dx \int_{-\infty}^{\infty} \frac{y}{(x-u)^2 + y^2} f(u) \, du$$

$$= \frac{1}{\pi} \int_{-\infty}^{\infty} f(u) \, du \int_{-\xi}^{\xi} \frac{y \cos xt}{(x-u)^2 + y^2} \, dx$$

$$= \frac{1}{\pi} \int_{0}^{\infty} \{f(u) + f(-u)\} \, du \int_{0}^{\xi} \left\{ \frac{y}{(x-u)^2 + y^2} + \frac{y}{(x+u)^2 + y^2} \right\} \cos xt \, dx$$

$$= \frac{1}{\pi} \left(\int_{0}^{\xi} \int_{0}^{\infty} - \int_{0}^{\xi} \int_{\xi}^{\infty} + \int_{\xi}^{\infty} \int_{0}^{\xi} \right) = \frac{1}{\pi} (J_1 + J_2 + J_3).$$

Also

$$\int_{0}^{\infty} \left\{ \frac{y}{(x-u)^{2} + y^{2}} + \frac{y}{(x+u)^{2} + y^{2}} \right\} \cos xt \, dx$$

$$= \int_{-\infty}^{\infty} \frac{y}{(x-u)^{2} + y^{2}} \cos xt \, dx = \pi e^{-yt} \cos ut.$$

$$\text{See} \qquad J_{1} = \pi e^{-yt} \int_{0}^{\xi} \{f(u) + f(-u)\} \cos ut \, du. \tag{6.9.2}$$

Hence

Again, by the second mean-value theorem,

$$\left| \int_{\xi}^{\infty} \frac{y \cos xt}{(x+u)^2 + y^2} dx \right| \leqslant \frac{2}{t} \frac{y}{(u+\xi)^2 + y^2},$$

$$\left| \int_{\xi}^{\infty} \frac{y \cos xt}{(x-u)^2 + y^2} dx \right| \leqslant \frac{2}{t} \frac{y}{(u-\xi)^2 + y^2},$$

$$\leqslant \int_{\infty}^{\infty} \frac{y dx}{(x-u)^2 + y^2} = \pi.$$

and also

Hence, if $|f(u)+f(-u)| \leq M$,

$$egin{aligned} |J_2| &\leqslant rac{2M}{t} \int\limits_0^{\xi} rac{y}{(x+\xi)^2+y^2} \, dx + rac{2M}{t} \int\limits_0^{\xi-\sqrt{y}} rac{y}{(\xi-x)^2+y^2} \, dx + M\pi \int\limits_{\xi-\sqrt{y}}^{\xi} \, dy \ &= rac{2M}{t} igg(rgama rac{2\xi}{y} - rctan rac{\xi}{y} igg) + rac{2M}{t} igg(rgama ext{catan} rac{1}{\sqrt{y}} - rctan rac{\xi}{y} igg) + M\pi \sqrt{y}. \end{aligned}$$

Also, if $u > \xi$,

$$\left| \int_{0}^{\xi} \frac{y \cos xt}{(x+u)^2 + t^2} dx \right| \leqslant \frac{2}{t} \frac{y}{u^2 + y^2},$$

$$\left| \int_{0}^{\xi} \frac{y \cos xt}{(x-u)^2 + t^2} dx \right| \leqslant \frac{2}{t} \frac{y}{(u-\xi)^2 + y^2}$$

$$\leqslant \pi$$

and also

as before. Hence

$$|J_3|\leqslant rac{2M}{t}\int\limits_{\xi}^{\infty}rac{y}{u^2+y^2}du+rac{2M}{t}\int\limits_{\xi+\sqrt{y}}^{\infty}rac{y}{(u-\xi)^2+y^2}du+M\pi\int\limits_{\xi}^{\xi+\sqrt{y}}du \ =rac{2M}{t}igg(rac{\pi}{2}-rctanrac{\xi}{y}igg)+rac{2M}{t}igg(rac{\pi}{2}-rctanrac{1}{\sqrt{y}}igg)+M\pi\sqrt{y}.$$

Let $y_1, y_2, ...$ be a sequence of values of y tending to 0, and let E be the set of values of t for which (6.9.1) fails for any of these values of y. Then E is of measure 0. Let t be a point not in E. Then we can choose $y = y_n$ so small that the contribution of J_2 and J_3 to (6.9.1) is $< \epsilon$ for all $\lambda > 1$. Having fixed y, it follows from (6.9.2) that

$$\left|e^{-yt}a(t)-e^{-yt}rac{1}{\pi\lambda}\int\limits_0^\lambda d\xi\int\limits_0^\xi \{f(u)+f(-u)\}\cos ut\ du
ight|<2\epsilon$$

for λ sufficiently large. Hence

$$a(t) = \lim_{\lambda \to \infty} \frac{1}{\pi \lambda} \int_{0}^{\lambda} d\xi \int_{0}^{\xi} \{f(u) + f(-u)\} \cos ut \ du.$$

Similarly we can prove the corresponding result for b(t).

6.10. Special properties of transforms. In this section we consider some special properties of sine and cosine transforms.

THEOREM 123. Let f(x) be non-increasing over $(0,\infty)$, integrable over (0,1), and let $f(x) \to 0$ as $x \to \infty$. Then $F_s(x) \ge 0$.

For
$$F_s(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^{\infty} f(y) \sin xy \ dy = \sqrt{\left(\frac{2}{\pi}\right)} \sum_{n=0}^{\infty} \int_{n\pi/x}^{(n+1)\pi/x} f(y) \sin xy \ dy.$$

This is a series of alternately positive and negative terms, non-increasing in absolute value. Its sum is therefore ≥ 0 .*

THEOREM 124. Let f(x) be a bounded function, which decreases steadily to 0 as $x \to \infty$ and is convex downwards. Then $F_c(x)$ is positive and belongs to $L(0,\infty)$.

The conditions imply that f(x) is the integral of f'(x), which is negative and non-decreasing, and tends to a limit at infinity; and the limit is 0 since f(x) is bounded. We can now integrate Fourier's cosine integral by parts, and obtain

$$F_c(x) = -\sqrt{\left(\frac{2}{\pi}\right)} \frac{1}{x} \int_0^\infty f'(y) \sin xy \ dy,$$

and this is positive, by the previous theorem. Also we may now take x = 0 in the analysis of Theorem 6, and obtain

$$\sqrt{\left(\frac{2}{\pi}\right)}\int_{-\infty}^{\infty}F_c(u)\ du=\tfrac{1}{2}f(+0).$$

Hence $F_c(x)$ belongs to $L(0, \infty)$.

Neither of these theorems is true for transforms of the opposite kind. If f(x) = 1 ($0 \le x \le 1$), 0 (x > 1), then

$$F_c(x) = \sqrt{\left(\frac{2}{\pi}\right) \frac{\sin x}{x}},$$

which takes both signs. If $f(x) = e^{-x}$, then

$$F_s(x) = \sqrt{\left(\frac{2}{\pi}\right)\frac{x}{1+x^2}},$$

which is positive but does not belong to $L(0,\infty)$. In fact if $F_s(x)$ belongs to $L(0,\infty)$, $f(x) \to 0$ both as $x \to 0$ and as $x \to \infty$, and so cannot be monotonic.

6.11. It is not quite easy to construct a monotonic f(x) for which $F_c(x)$ is not integrable ver $(0,\infty)$. To do so, we first prove that there is a function

 $\phi(x) = \sum_{n=1}^{\infty} b_n \sin \lambda_n x$

[†] For a similar result for series see Szidon (1).

with $b_n \geqslant 0$, $\sum\limits_{1}^{\infty} b_n$ convergent, and

$$\int_{1}^{1} \frac{|\phi(x)|}{x} \, dx$$

divergent.

We have

$$\begin{split} \int\limits_{1/\lambda_n}^1 \frac{|\phi(x)|}{x} \, dx \geqslant \int\limits_{1/\lambda_n}^1 b_n |\sin \lambda_n \, x| \, \frac{dx}{x} - \int\limits_{1/\lambda_n}^1 \left| \sum_{\nu=1}^{n-1} b_\nu \sin \lambda_\nu \, x \right| \frac{dx}{x} - \\ - \int\limits_{1/\lambda_n}^1 \left| \sum_{\nu=n+1}^\infty b_\nu \sin \lambda_\nu \, x \right| \frac{dx}{x} = J_1 - J_2 - J_3. \end{split}$$

Now

$$egin{aligned} J_1 &= b_n \int\limits_1^{\lambda_n} \left| rac{\sin u}{u}
ight| du > A b_n \log \lambda_n, \ J_2 &< \int\limits_0^1 \sum\limits_{
u=1}^{n-1} b_
u \lambda_
u \, dx = \sum\limits_{
u=1}^{n-1} b_
u \lambda_
u, \ J_3 &< \int\limits_1^1 \sum\limits_{
u=n+1}^{\infty} b_
u rac{dx}{x} = \log \lambda_n \sum\limits_{
u=n+1}^{\infty} b_
u. \end{aligned}$$

Hence $J_3 < \frac{1}{2}J_1$ if $b_{\nu} = k^{-\nu}$ with a sufficiently large k; and $J_1 \to \infty$, $J_2 = o(J_1)$ if e.g. $\lambda_1 = 1$, and

$$\lambda_n = 2^{(\lambda_1 + \dots + \lambda_{n-1})/b_n},$$

$$\int_1^1 \frac{|\phi(x)|}{x} dx \to \infty,$$

Then

the required result.

THEOREM 125. There is a function f(x), continuous and steadily decreasing to 0 as $x \to \infty$, such that $F_c(x)$ does not belong to $L(0,\infty)$.

We first obtain the result for a non-increasing function. Let $f(x)=c_n$ in $(a_{n-1}+\delta, a_n-\delta)$, where $c_n\to 0$ steadily and $a_n\to \infty$ steadily, and $0<\delta<1$; and let f(x) be continuous and linear in the remaining intervals. Then, taking $a_0+\delta=0$,

$$\sqrt{(\frac{1}{2}\pi)}F_{c}(x) = \sum_{n=1}^{\infty} c_{n} \int_{a_{n-1}+\delta}^{a_{n}-\delta} \cos xt \, dt + + \sum_{n=1}^{\infty} \int_{a_{n}-\delta}^{a_{n}+\delta} \left\{ \frac{1}{2}(c_{n}+c_{n+1}) + \frac{(c_{n}-c_{n+1})(a_{n}-t)}{2\delta} \right\} \cos xt \, dt$$

$$\begin{split} &=\sum_{n=1}^{\infty}c_n\frac{\sin(a_n-\delta)x-\sin(a_{n-1}+\delta)x}{x}+\\ &+\sum_{n=1}^{\infty}\biggl[c_{n+1}\frac{\sin(a_n+\delta)x}{x}-c_n\frac{\sin(a_n-\delta)x}{x}+\\ &+\frac{c_n-c_{n+1}}{2\delta x^2}\{\cos(a_n-\delta)x-\cos(a_n+\delta)x\}\biggr]\\ &=\frac{\sin\delta x}{\delta x^2}\sum_{n=1}^{\infty}\left(c_n-c_{n+1})\sin a_n\,x. \end{split}$$

By the lemma, we can choose c_n and a_n so that

$$\int_{0}^{1} \left| \sum_{n=1}^{\infty} (c_n - c_{n+1}) \sin a_n x \right| \frac{dx}{x}$$

is divergent, and then so is $\int_0^1 |F_c(x)| dx$, since $\sin \delta x/(\delta x) \to 1$ as $x \to 0$.

We can plainly construct a steadily decreasing function g(x), having derivatives of as many orders as we please, such that f(x)-g(x) belongs to $L(0,\infty)$. Then the cosine transform of f(x)-g(x) is bounded. Hence the cosine transform of g(x) does not belong to $L(0,\infty)$.

6.12. Under special conditions $F_c(x)$ and $F_s(x)$ behave asymptotically like a power of x, either as $x \to 0$ or as $x \to \infty$, or both. The two simplest theorems of this character are as follows.†

THEOREM 126. Let $f(x) = x^{-\alpha}\phi(x)$, where $0 < \alpha < 1$, and $\phi(x)$ is of bounded variation in $(0, \infty)$. Then

$$F_c(x) \sim \phi(+0) \sqrt{\left(\frac{2}{\pi}\right)} \Gamma(1-\alpha) \sin \frac{1}{2} \pi \alpha x^{\alpha-1} \quad (x \to \infty),$$

$$F_c(x) \sim \phi(\infty) \sqrt{\left(\frac{2}{\pi}\right)} \Gamma(1-\alpha) \sin \frac{1}{2} \pi \alpha x^{\alpha-1} \quad (x \to 0).$$

 $F_s(x)$ satisfies similar conditions with $\sin \frac{1}{2}\pi\alpha$ replaced by $\cos \frac{1}{2}\pi\alpha$.

We may suppose that $\phi(x)$ is positive non-increasing in $(0,\infty)$. Take the case $x \to \infty$. We have

$$\sqrt{(\frac{1}{2}\pi)}F_c(x) = \int\limits_0^\infty t^{-\alpha}\phi(t)\cos xt\ dt$$
† Titchmarsh (9).

$$=x^{lpha-1}\int\limits_0^\infty u^{-lpha}\phi(u/x)\cos u\ du$$
 $=x^{lpha-1}\Big(\int\limits_0^\Delta+\int\limits_{\Delta}^\infty\Big)=x^{lpha-1}(I_1+I_2).$

By the second mean-value theorem

$$I_2 = \phi\left(\frac{\Delta}{x}\right) \int_{\Lambda}^{\Delta'} u^{-\alpha} \cos u \ du = O(\Delta^{-\alpha})$$

uniformly with respect to x. Now, for a fixed Δ ,

$$\int_{0}^{\Delta} \{\phi(+0) - \phi(u/x)\} u^{-\alpha} \cos u \, du = \{\phi(+0) - \phi(\Delta/x)\} \int_{0}^{\Delta} u^{-\alpha} \cos u \, du$$

$$= O\{\phi(+0) - \phi(\Delta/x)\} = o(1),$$

and

Then

$$\phi(+0)\int_0^\Delta u^{-\alpha}\cos u\ du \to \phi(+0)\int_0^\infty u^{-\alpha}\cos u\ du = \phi(+0)\Gamma(1-\alpha)\sin\tfrac12\pi\alpha.$$

Hence the result. Similarly in the case $x \to 0$.

THEOREM 127. Let f(x) and f'(x) be integrable over any finite interval not ending at x = 0; let $x^{\alpha+1}f'(x)$ be bounded for all x, and let $f(x) \sim x^{-\alpha}$ as $x \to \infty$ $(x \to 0)$. Then

$$F_c(x) \sim \sqrt{\left(\frac{2}{\pi}\right)} \Gamma(1-\alpha) \sin \frac{1}{2} \pi \alpha x^{\alpha-1}$$

as $x \to 0$ $(x \to \infty)$. $F_s(x)$ satisfies similar conditions with $\sin \frac{1}{2}\pi\alpha$ replaced by $\cos \frac{1}{2}\pi\alpha$.

Consider the case $x \to \infty$. We have

$$\sqrt{(\frac{1}{2}\pi)}F_c(x) = \int\limits_0^\infty f(t)\cos xt \ dt = \int\limits_0^{\Delta/x} + \int\limits_{\Delta/x}^\infty = I_1 + I_2.$$
 $I_2 = -f\left(\frac{\Delta}{x}\right) \frac{\sin \Delta}{x} - \frac{1}{x} \int\limits_{\Delta/x}^\infty f'(t)\sin xt \ dt$

$$= O(\Delta^{-\alpha}x^{\alpha-1}) + \frac{1}{x} \int_{\Delta/x}^{\infty} O(t^{-\alpha-1}) dt$$
$$= O(\Delta^{-\alpha}x^{\alpha-1}).$$

Let $m(\xi) = \max_{x \in \xi} |x^{\alpha} f(x) - 1|$, so that $m(\xi) \to 0$ as $\xi \to 0$. Then

$$\begin{split} I_1 &= \int_0^{\Delta/x} t^{-\alpha} \cos xt \, dt + \int_0^{\Delta/x} \{t^{\alpha} f(t) - 1\} t^{-\alpha} \cos xt \, dt \\ &= x^{\alpha - 1} \int_0^{\Delta} u^{-\alpha} \cos u \, du + O\left(m\left(\frac{\Delta}{x}\right)\left(\frac{\Delta}{x}\right)^{1 - \alpha}\right) \\ &= x^{\alpha - 1} \int_0^{\infty} u^{-\alpha} \cos u \, du + O(x^{\alpha - 1}\Delta^{-\alpha}) + O\left(m\left(\frac{\Delta}{x}\right)\left(\frac{\Delta}{x}\right)^{1 - \alpha}\right), \end{split}$$

and the result follows on choosing Δ large enough, and then x large enough.

Similarly if $x \to 0$.

6.13. Order of magnitude of transforms. There are various more or less trivial results; if $(1+|x|^n)f(x)$ belongs to $L(-\infty,\infty)$, the equation

 $F(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(t)e^{ixt} dt$

can be differentiated n times. It follows that F(x), F'(x), ..., $F^{(n)}(x)$ are all continuous and tend to 0 at infinity.

If $f(x), ..., f^{(n-1)}(x)$ are continuous and tend to 0 at infinity, and $f^{(n)}(x)$ is L, then by repeated integration by parts

$$F(x) = \left(\frac{i}{x}\right)^n \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f^{(n)}(t)e^{ixt} dt.$$

Hence $x^n F(x) \to 0$ as $x \to \pm \infty$.

Similarly, if $(1+|x|^n)f(x)$ belongs to $L^2(-\infty,\infty)$, then

$$F(x), \ldots, F^{(n-1)}(x)$$

are continuous and tend to 0 at infinity, and $F^{(n)}(x)$ is $L^2(-\infty,\infty)$; and conversely.

Other results have been given in Theorem 26.

The idea underlying the following theorem is that both a function and its transform cannot be too small at infinity.† The result is

THEOREM 128. Let f(x) and $F_c(x)$ be Fourier cosine transforms, and let each be $O(e^{-\frac{1}{2}x^2})$ as $x \to \infty$. Then

$$f(x) = F_c(x) = Ce^{-\frac{1}{2}x^2}.$$

† Hardy (19), Ingham (1), Morgan (1), Paley and Wiener, Fourier Transforms, § 19.

We use the following lemma.

LEMMA. Let $\phi(z)$ be an integral function, $\phi(z) = O(e^{a|z|})$ for all z, and $\phi(x) = O(e^{-ax})$ for real positive $x \to \infty$, a being a positive constant. Then $\phi(z) = Ce^{-az}$.

There is a constant C such that

$$|\phi(x)| \leqslant Ce^{-ax}, \qquad |\phi(re^{i\alpha})| \leqslant Ce^{ar},$$

where $0 < \alpha < \pi$. Hence, by a theorem of Phragmén and Lindelöf,†

$$|\phi(re^{i\theta})| \leqslant Ce^{rH(\theta)} \quad (0 \leqslant \theta \leqslant \alpha),$$

where

$$H(\theta) = \frac{-a\sin(\alpha - \theta) + a\sin\theta}{\sin\alpha} = a\frac{\sin(\theta - \frac{1}{2}\alpha)}{\sin\frac{1}{2}\alpha}.$$

Here we can keep θ fixed and make $\alpha \to \pi$. Then $H(\theta) \to -a \cos \theta$, and it follows that

$$|\phi(z)| \leqslant Ce^{-ar\cos\theta} \quad (0 \leqslant \theta < \pi).$$

Similarly, we obtain the same inequality for $-\pi < \theta \le 0$; and also, by continuity, for $\theta = \pi$. Hence $e^{az}\phi(z)$ is a bounded integral function, and so is a constant.

To prove the theorem we have

$$F_c(z) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty f(t) \cos zt \ dt.$$

By the condition on f(t) this is an even integral function of z; and if |z| = r,

 $|F_c(z)| \leqslant K \int_0^\infty e^{-it^2} \cosh rt \ dt = Ke^{ir^2}.$

Hence $F_c(\sqrt{z})$ is an integral function which satisfies the conditions of the lemma, with $a = \frac{1}{2}$. Hence

$$F_c(\sqrt{z}) = Ce^{-\frac{1}{2}z},$$
 $F_c(z) = Ce^{-\frac{1}{2}z^2}.$

and also the result for f(z) follows, by a familiar formula.

More general results can be obtained in a similar way. Suppose for example that

$$f(x) = O(x^{2k}e^{-\frac{1}{2}x^2}), \qquad F_c(x) = O(x^{2k}e^{-\frac{1}{2}x^2}),$$

[†] The argument is that of Titchmarsh, Theory of Functions, § 5.71, with $\delta = 0$.

where k is an integer. Then $F_c(z)$ is an even integral function; and

$$|F_c(z)| < K \int_0^\infty e^{-\frac{1}{4}t^2t} \cosh rt \ dt$$

$$= K \left(\frac{d}{dt}\right)^{2k} \int_0^\infty e^{-\frac{1}{4}t^2} \cosh rt \ dt$$

$$= K \left(\frac{d}{dt}\right)^{2k} e^{\frac{1}{4}r^2} = O(r^{2k}e^{\frac{1}{4}r^2}).$$

Let $\phi(z) = F_c(\sqrt{z})$. Then $\phi(z)$ is an integral function; and, if $a_0, ..., a_{k-1}$ are properly chosen, so is

$$\psi(z) = z^{-2k} \{ \phi(z) - (a_0 + a_1 z + \ldots + a_{k-1} z^{k-1}) e^{-\frac{1}{2}z} \}.$$

Hence $\psi(z)$ satisfies the conditions of the lemma, and so is $Ce^{-\frac{1}{2}z}$. Hence $F_c(z) = (a_0 + a_1 z^2 + ... + a_k z^{2k})e^{-\frac{1}{2}z^k}$,

and f(z) is another expression of the same form.

EXAMPLES AND APPLICATIONS†

7.1. Cosine transforms. Simple pairs of cosine transforms are

1
$$(0,a)$$
, θ (a,∞) , $\sqrt{\left(\frac{2}{\pi}\right)} \frac{\sin ax}{x}$, (7.1.1)

$$\cos x \ (0,a), \quad 0 \ (a,\infty), \qquad \frac{1}{\sqrt{(2\pi)}} \left\{ \frac{\sin a(1-x)}{1-x} + \frac{\sin a(1+x)}{1+x} \right\}, \quad (7.1.2)$$

$$\sin x \ (0,a), \quad 0 \ (a,\infty), \qquad \frac{1}{\sqrt{(2\pi)}} \left\{ \frac{1-\cos a(1-x)}{1-x} + \frac{1-\cos a(1+x)}{1+x} \right\},$$
(7.1.3)

$$e^{-x}$$
, $\sqrt{\left(\frac{2}{\pi}\right)\frac{1}{1+x^2}}$. (7.1.4)

Generally, the cosine transform of any even rational function, regular on the real axis and $O(1/x^2)$ at infinity, can be evaluated by contour integration; for example

$$\frac{1}{1+x^4}$$
, $\sqrt{\left(\frac{\pi}{2}\right)}e^{-x/\sqrt{2}}\sin\left(\frac{x}{\sqrt{2}}+\frac{\pi}{4}\right)$. (7.1.5)

Another familiar process of contour integration gives the pair

$$\frac{1}{\cosh \pi x}, \qquad \frac{1}{\sqrt{(2\pi)\cosh \frac{1}{2}x}}.$$
 (7.1.6)

Next

$$\sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} e^{-\frac{1}{2}x^{2}} \cos xu \, dx$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} e^{-\frac{1}{2}x^{2} + ixu} \, dx = \frac{1}{\sqrt{(2\pi)}} e^{-\frac{1}{2}u^{2}} \int_{-\infty}^{\infty} e^{-\frac{1}{2}(x - iu)^{2}} \, dx$$

$$= \frac{1}{\sqrt{(2\pi)}} e^{-\frac{1}{2}u^{2}} \int_{-\infty}^{\infty} e^{-\frac{1}{2}\xi^{2}} \, d\xi = Ce^{-\frac{1}{2}u^{2}}$$

(by Cauchy's theorem). The cosine formula then gives $C^2 = 1$, whence C = 1 since C > 0. Hence we have the pair

$$e^{-\frac{1}{2}x^2}$$
, $e^{-\frac{1}{2}x^2}$. (7.1.7)

All the above examples belong to obvious L-classes.

† An extensive list of Fourier transforms is given by Campbell and Foster, Fourier Integrals for Practical Applications.

4362

Next we have

$$\sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} e^{-\frac{1}{4}\lambda y^{2}} \cos xy \ dy = \sqrt{\left(\frac{2}{\pi\lambda}\right)} \int_{0}^{\infty} e^{-\frac{1}{4}y^{2}} \cos \frac{xy}{\sqrt{\lambda}} dy$$
$$= \frac{1}{\sqrt{\lambda}} e^{-\frac{1}{4}x^{2}/\lambda}$$

by the formula just proved.

This is true primarily for real positive λ , ut it can be extended by the theory of analytic continuation to all values of λ with $\mathbf{R}(\lambda) \geqslant 0$. Taking $\lambda = i$,

$$\sqrt{\binom{2}{\pi}}\int_{a}^{\infty}e^{-\frac{1}{2}iy^{2}}\cos xy\ dy=e^{\frac{1}{2}ix^{2}-\frac{1}{2}i\pi}.$$

Taking real and imaginary parts, we obtain the transforms

$$\cos \frac{1}{2}x^2$$
, $\frac{1}{\sqrt{2}}(\cos \frac{1}{2}x^2 + \sin \frac{1}{2}x^2)$, (7.1.8)

$$\sin \frac{1}{2}x^2$$
, $\frac{1}{\sqrt{2}}(\cos \frac{1}{2}x^2 - \sin \frac{1}{2}x^2)$. (7.1.9)

The Fourier formulae arising from these give examples of Theorem 11, case (i).

We define the Bessel function of order ν by

$$J_{\nu}(x) = \sum_{n=0}^{\infty} \frac{(-1)^n (\frac{1}{2}x)^{\nu+2n}}{n! \Gamma(\nu+n+1)} \quad (\nu > -1).$$
 (7.1.10)

Then †

$$\int_{0}^{1} (1-y^{2})^{\nu-\frac{1}{2}} \cos xy \, dy = \sum_{n=0}^{\infty} \frac{(-1)^{n} x^{2n}}{(2n)!} \int_{0}^{1} (1-y^{2})^{\nu-\frac{1}{2}} y^{2n} \, dy$$

$$= \frac{1}{2} \sum_{n=0}^{\infty} \frac{(-1)^{n} \Gamma(\nu+\frac{1}{2}) \Gamma(n+\frac{1}{2})}{(2n)! \Gamma(\nu+n+1)} x^{2n}$$

$$= \frac{1}{2} \sqrt{\pi} \Gamma(\nu+\frac{1}{2}) \sum_{n=0}^{\infty} \frac{(-1)^{n} x^{2n}}{2^{2n} n! \Gamma(\nu+n+1)}.$$

Hence we have the cosine transforms

$$(1-x^2)^{\nu-\frac{1}{2}}$$
 $(0 < x < 1), \quad 0 \quad (x > 1), \quad 2^{\nu-\frac{1}{2}}\Gamma(\nu+\frac{1}{2})x^{-\nu}J_{\nu}(x).$ (7.1.11)

[†] Watson, Theory of Bessel Functions, § 3.3 (2). This work is referred to later as 'Watson'.

These belong to L^2 if $\nu > 0$, and to L^p , $L^{p'}$ respectively if $1/p > \frac{1}{2} - \nu$, $\nu > -\frac{1}{2}$.

In each case we can introduce a parameter, since the transform of $f(\lambda x)$ is $\frac{1}{\lambda}F_c\left(\frac{x}{\lambda}\right)$,

and similarly for sine transforms.

7.2. Sine transforms. A simple pair is

$$e^{-x}$$
, $\sqrt{\frac{2}{\pi}} \frac{x}{1+x^2}$, (7.2.1)

and, generally, the sine transform of any odd rational function, regular on the real axis and O(1/x) at infinity, can be evaluated by contour integration.

Other familiar methods of contour integration give the pairs

$$\frac{1}{e^{x\sqrt{(2\pi)}}-1} - \frac{1}{x\sqrt{(2\pi)}}, \qquad \frac{1}{e^{x\sqrt{(2\pi)}}-1} - \frac{1}{x\sqrt{(2\pi)}}, \tag{7.2.2}$$

$$\frac{1}{\sinh x\sqrt{(\frac{1}{2}\pi)}} - \frac{1}{x\sqrt{(\frac{1}{2}\pi)}}, \quad \tanh\{x\sqrt{(\frac{1}{2}\pi)}\} - 1.$$
 (7.2.3)

The pair
$$xe^{-\frac{1}{2}x^2}$$
, $xe^{-\frac{1}{2}x^2}$ (7.2.4)

may be obtained by differentiation from (7.1.7). Next†

$$\int_{0}^{\infty} \frac{\sin y}{y} \sin xy \ dy = \frac{1}{2} \int_{0}^{\infty} \frac{\cos(1-x)y - \cos(1+x)y}{y} \ dy = \frac{1}{2} \log \left| \frac{1+x}{1-x} \right|.$$

Hence we obtain the pair

$$\frac{\sin x}{x}, \qquad \frac{1}{\sqrt{(2\pi)}} \log \left| \frac{1+x}{1-x} \right|. \tag{7.2.5}$$

If $\nu > \frac{1}{2}$ we obtain by partial integration from (7.1.11) the pair $x(1-x^2)^{\nu-\frac{1}{2}}$ (0 < x < 1), 0 (x > 1), $2^{\nu-\frac{1}{2}}\Gamma(\nu-\frac{1}{2})x^{1-\nu}J_{\nu}(x)$. (7.2.6)

We define the Struve's function \ddagger of order ν by

$$\mathbf{H}_{\nu}(x) = \sum_{n=0}^{\infty} \frac{(-1)^n (\frac{1}{2}x)^{\nu+2n+1}}{\Gamma(n+\frac{3}{2})\Gamma(\nu+n+\frac{3}{2})} \quad (\nu > -\frac{3}{2}). \tag{7.2.7}$$

† c.g. as in § 5.2,

Watson, § 10.4.

Then

$$\begin{split} \int_0^1 (1-y^2)^{\nu-\frac{1}{2}} \sin xy \ dy &= \sum_{n=0}^\infty \frac{(-1)^n x^{2n+1}}{(2n+1)!} \int_0^1 (1-y^2)^{\nu-\frac{1}{2}} y^{2n+1} \ dy \\ &= \frac{1}{2} \sum_{n=0}^\infty \frac{(-1)^n \Gamma(\nu+\frac{1}{2}) n!}{(2n+1)! \Gamma(\nu+n+\frac{3}{2})} x^{2n+1} \\ &= \frac{1}{2} \sqrt{\pi} \Gamma(\nu+\frac{1}{2}) \sum_{n=0}^\infty \frac{(-1)^n x^{2n+1}}{2^{2n} \Gamma(n+\frac{3}{2}) \Gamma(\nu+n+\frac{3}{2})}. \end{split}$$

Hence we have the sine transforms

$$(1-x^2)^{\nu-\frac{1}{2}}$$
 $(0 < x < 1), \quad 0 \ (x > 1), \quad 2^{\nu-\frac{1}{2}}\Gamma(\nu+\frac{1}{2})x^{-\nu}\mathbf{H}_{\nu}(x).$ (7.2.8)

7.3. The Parseval formulae. We obtain simple examples of (2.1.4) or (2.1.6) by taking f and g rational functions; for example, let $f(x) = 1/(x^2+a^2)$, $F_c(x) = \sqrt{(\frac{1}{2}\pi)e^{-ax}/a}$, and similarly g, G_c , with b for a. We obtain

$$\int_{0}^{\infty} \frac{dx}{(x^{2} + a^{2})(x^{2} + b^{2})} = \frac{\pi}{2ab} \int_{0}^{\infty} e^{-ax - bx} dx = \frac{\pi}{2ab(a + b)}.$$
 (7.3.1)

As another type, let f(x) = 1 (0 < x < a), 0 $(x \ge a)$, $F_c(x) = \sin ax/x$, and similarly g, G_c , with b for a. We obtain

$$\int_{0}^{\infty} \frac{\sin ax \sin bx}{x^{2}} dx = \frac{1}{2}\pi \int_{0}^{\min(a,b)} dx = \frac{1}{2}\pi \min(a,b). \quad (7.3.2)$$

Similarly, from (2.1.6) and (7.2.5) we obtain

$$\int_{0}^{\infty} \log \left| \frac{a+x}{a-x} \right| \log \left| \frac{b+x}{b-x} \right| dx = 2\pi \int_{0}^{\infty} \frac{\sin ax \sin bx}{x^2} dx = \pi^2 \min(a,b),$$
(7.3.3)

by (7.3.2); or we can obtain this directly from Theorem 91. All the above formulae come under Theorem 52.

We can deduce some of the familiar Γ -function formulae from Parseval's formula. \dagger Define $\Gamma(a)$ by

$$\Gamma(a) = \int_{0}^{\infty} e^{-x} x^{a-1} dx \quad (a > 0), \tag{7.3.4}$$

and let

$$c(a) = \int_{0}^{\infty} x^{a-1} \cos x \, dx \quad (0 < a < 1), \tag{7.3.5}$$

$$s(a) = \int_{0}^{\infty} x^{a-1} \sin x \, dx \quad (-1 < a < 1). \tag{7.3.6}$$

Then the cosine transform of x^{a-1} is

$$\sqrt{\left(\frac{2}{\pi}\right)}\int_{0}^{\infty}y^{a-1}\cos xy\ dy = x^{-a}\sqrt{\left(\frac{2}{\pi}\right)}\int_{0}^{\infty}u^{a-1}\cos u\ du = x^{-a}\sqrt{\left(\frac{2}{\pi}\right)}c(a).$$

Similarly, the sine transform of x^{a-1} is $x^{-a} \sqrt{\left(\frac{2}{\pi}\right)} s(a)$.

We can prove, by contour integration or term-by-term integration of series, that

$$\int_{0}^{\infty} \frac{x^{-a}}{1+x^{2}} dx = \frac{1}{2}\pi \sec \frac{1}{2}a\pi \quad (-1 < a < 1). \tag{7.3.7}$$

In (2.1.4) let $f(x) = e^{-x}$, $g(x) = x^{a-1}$. We obtain

$$\Gamma(a) = \frac{2}{\pi} \int_{0}^{\pi} \frac{x^{-a}c(a)}{1+x^{2}} dx = c(a)\sec \frac{1}{2}a\pi \quad (0 < a < 1), \quad (7.3.8)$$

e.g. by Theorem 35 or Theorem 37. Similarly, by (2.1.6),

$$\Gamma(a) = \frac{2}{\pi} \int_{0}^{\infty} \frac{x}{1+x^2} x^{-a} s(a) \, dx = s(a) \operatorname{cosec} \frac{1}{2} a \pi \quad (0 < a < 1).$$
(7.3.9)

Also, by the above rule, the cosine transform of $x^{-a} / \left(\frac{2}{\pi}\right) c(a)$ is

$$x^{1-a}\frac{2}{\pi}c(a)c(1-a);$$

since it is also x^{1-a} , by Theorem 6, it follows that

$$c(a)c(1-a) = \frac{1}{2}\pi \quad (0 < a < 1).$$
 (7.3.10)

Similarly,
$$s(a)s(1-a) = \frac{1}{2}\pi \quad (0 < a < 1).$$
 (7.3.11)

Also, (7.3.8) and (7.3.10) give

$$\Gamma(a)\Gamma(1-a) = \frac{c(a)c(1-a)}{\frac{1}{3}\sin a\pi} = \frac{\pi}{\sin a\pi}.$$
 (7.3.12)

In particular

$$c(\frac{1}{2}) = \sqrt{(\frac{1}{2}\pi)}, \quad s(\frac{1}{2}) = \sqrt{(\frac{1}{2}\pi)}, \quad \Gamma(\frac{1}{2}) = \sqrt{\pi}.$$
 (7.3.13)

We have also obtained the cosine transforms

$$x^{a-1}$$
, $\sqrt{\left(\frac{2}{\pi}\right)}\Gamma(a)\cos\frac{1}{2}a\pi x^{-a}$ $(0 < a < 1)$ $(7.3.14)$

and the sine transforms

$$x^{a-1}$$
, $\sqrt{\left(\frac{2}{\pi}\right)}\Gamma(a)\sin\frac{1}{2}a\pi x^{-a}$ $(0 < a < 1)$. $(7.3.15)$

7.4. Some Bessel-function examples. From (2.1.4), (7.1.11), and (7.3.14) we obtain

$$\int_{0}^{\infty} J_{\nu}(x)x^{a-\nu-1} dx = \sqrt{\left(\frac{2}{\pi}\right)} \frac{\Gamma(a)\cos\frac{1}{2}a\pi}{2^{\nu-1}\Gamma(\nu+\frac{1}{2})} \int_{0}^{1} (1-x^{2})^{\nu-\frac{1}{2}}x^{-a} dx$$

$$= \frac{\Gamma(a)\cos\frac{1}{2}a\pi}{2^{\nu-1}\sqrt{\pi}\Gamma(\nu+\frac{1}{2})} \frac{\Gamma(\nu+\frac{1}{2})\Gamma(\frac{1}{2}-\frac{1}{2}a)}{2\Gamma(\nu-\frac{1}{2}a+1)}$$

$$= \frac{2^{a-\nu-1}\Gamma(\frac{1}{2}a)}{\Gamma(\nu-\frac{1}{2}a+1)}.$$
(7.4.1)†

This is a case of Theorem 37 if $\nu > -\frac{1}{2}$, 0 < a < 1 (taking $f(x) = (1-x^2)^{\nu-\frac{1}{2}}$ (0 < x < 1), 0 (x > 1), and $g(x) = x^{-a}$). Actually the integral converges if $0 < a < \nu + \frac{3}{2}$, so that the result holds by analytic continuation in this wider range.

Similarly, from (2.1.6), (7.2.8), and (7.3.15) we obtain

$$\int_{0}^{\infty} \mathbf{H}_{\nu}(x) x^{a-\nu-1} dx = \sqrt{\left(\frac{2}{\pi}\right)} \frac{\Gamma(a) \sin \frac{1}{2} a \pi}{2^{\nu-\frac{1}{2}} \Gamma(\nu+\frac{1}{2})} \int_{0}^{1} (1-x^{2})^{\nu-\frac{1}{2}} x^{-a} dx$$

$$= \frac{2^{a-\nu-1} \Gamma(\frac{1}{2}a) \tan \frac{1}{2} a \pi}{\Gamma(\nu-\frac{1}{2}a+1)} \quad (-1 < a < \nu+\frac{3}{2}).$$
(7.4.2)

As an example of (2.1.8) let $-\frac{1}{2} < \nu < 0$,

 $f(x) = \sqrt{(2/\pi)}\Gamma(2\nu+1)\cos\nu\pi|x|^{-2\nu-1}\operatorname{sgn} x, \quad F(x) = -|x|^{2\nu}\operatorname{sgn} x$ by (7.3.15), and

$$g(x) = 2^{\frac{1}{4}-\nu}(1-x^2)^{\nu-\frac{1}{4}}/\Gamma(\nu+\frac{1}{2}) \quad (|x| < 1), \qquad 0 \quad (|x| > 1),$$

$$G(x) = |x|^{-\nu}J_{\nu}(|x|),$$

by (7.1.11). Then if $x \neq 1$,

$$2 \int_{0}^{+\infty} t^{\nu} J_{\nu}(t) \sin xt \, dt$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \frac{\Gamma(2\nu+1)\cos\nu\pi}{2^{\nu-\frac{1}{2}}\Gamma(\nu+\frac{1}{2})} \int_{-1}^{1} (1-u^{2})^{\nu-\frac{1}{2}}|x-u|^{-2\nu-1} \operatorname{sgn}(x-u) \, du.$$

$$+ \operatorname{Watson}, \S 13.24 (1). \qquad \sharp \operatorname{Ibid.}, \S 13.24 (2). \qquad (7.4.3)$$

This may be justified by Theorem 37, with f and g interchanged and g(u) replaced by g(u+x).

If x > 1, the integral on the right-hand side is \dagger

$$\int_{-1}^{1} (1-u^2)^{\nu-\frac{1}{2}} (x-u)^{-2\nu-1} du = \frac{\{\Gamma(\nu+\frac{1}{2})\}^2}{\Gamma(2\nu+1)} \frac{2^{2\nu}}{(x^2-1)^{\nu+\frac{1}{2}}}.$$

If 0 < x < 1, the integral is 0; for consider

$$\int (1-w^2)^{\nu-\frac{1}{2}}(x-w)^{-2\nu-1} dw.$$

This integral, taken round a circle of centre the origin and radius R, tends to 0 as $R \to \infty$. On reducing the contour to the real axis from -1 to 1 described twice, and allowing for the change of value of the integrand at -1, x, and 1, we obtain a multiple of the above integral.

It follows that we have the sine transforms

$$x^{\nu}J_{\nu}(x), \qquad \frac{2^{\nu+\frac{1}{4}}}{\Gamma(\frac{1}{2}-\nu)(x^2-1)^{\nu+\frac{1}{4}}} \quad (x>1), \quad 0 \ (x<1). \ (7.4.4)^{\frac{1}{4}}$$

Actually (7.4.3) converges for $\nu < \frac{1}{2}$, and the result holds by analytic continuation in this wider range. The functions belong to L^p , $L^{p'}$ if

$$p > \frac{1}{\frac{1}{2} - \nu} \quad (\nu \geqslant 0); \qquad \frac{1}{\frac{1}{2} - \nu} (7.4.5)$$

From (2.1.6), generalized as in (2.1.20), and (7.2.6) and (7.4.4), we obtain, if 0 < a < b,

$$\begin{split} \int\limits_0^\infty {(ax)^\mu J_\mu (ax)(bx)^{1 - \nu} J_\nu (bx)} \; dx \\ &= \frac{1}{ab} \int\limits_a^b {\frac{{2^{\mu + \frac{1}{b}}}}{{\Gamma (\frac{1}{2} - \mu)}} {\left({\frac{{{x^2}}}{{{a^2}}} - 1} \right)^{ - \mu - \frac{1}{b}}} \frac{{2^{\frac{1}{b} - \nu }}}{{\Gamma (\nu - \frac{1}{2})} \frac{x}{b}} {\left({1 - \frac{{{x^2}}}{{{b^2}}}} \right)^{\nu - \frac{1}{b}}} \; dx, \end{split}$$

while the left-hand side is 0 if $0 < b \leqslant a$. Hence

$$\int_{0}^{\infty} x^{\mu-\nu+1} J_{\mu}(ax) J_{\nu}(bx) dx$$

$$= \frac{2^{\mu-\nu+1} a^{\mu} (b^{2}-a^{2})^{\nu-\mu-1}}{\Gamma(\nu-\mu)b^{\nu}} \quad (0 < a < b), \qquad 0 \quad (a \ge b). \quad (7.4.6)$$

The process is justified by L^2 theory if $-\frac{1}{4} < \mu < 0$, $\nu > 1$. As usual, the result holds in a wider range.

† See Titchmarsh, Theory of Functions, p. 63, ex. 19. ‡ Watson, § 6.13 (3).

Next let

$$f(x) = \sin x/x, \qquad F(x) = \sqrt{(\frac{1}{2}\pi)} \ (|x| < 1), \quad 0 \ (|x| > 1),$$
 and $g(x) = J_0(x) \quad (x > 0), \qquad 0 \quad (x < 0),$ $G(x) = rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty J_0(y) \cos xy \ dy + rac{i}{\sqrt{(2\pi)}} \int\limits_0^\infty J_0(y) \sin xy \ dy$

$$egin{align} egin{align} egin{align} egin{align} egin{align} eta_0(y)\cos xy & dy & + rac{1}{\sqrt{(2\pi)}}\int\limits_0^x eta_0(y)\sin xy & dy \ & = rac{1}{\sqrt{\{2\pi(1-x^2)\}}} & (|x| < 1), & rac{i \, {
m sgn} \, x}{\sqrt{\{2\pi(x^2-1)\}}} & (|x| > 1), \ \end{pmatrix} \end{aligned}$$

by (7.1.11) and (7.4.4). Then (2.1.8) gives

$$\int_{0}^{\infty} J_{0}(t) \frac{\sin(x-t)}{x-t} dt = \frac{1}{2} \int_{-1}^{1} \frac{e^{-ixt}}{\sqrt{(1-t^{2})}} dt = \int_{0}^{1} \frac{\cos xt}{\sqrt{(1-t^{2})}} dt = \frac{1}{2} \pi J_{0}(x).$$
(7.4.7)

Here f(x) and G(x) belong to L^p if 1 .

From (7.2.1), (7.4.4), and (2.1.6) we deduce

$$\int_{0}^{\infty} e^{-ax} x^{\nu} J_{\nu}(x) dx = \sqrt{\left(\frac{2}{\pi}\right)} \int_{1}^{\infty} \frac{x}{a^{2} + x^{2}} \frac{2^{\nu + \frac{1}{2}}}{\Gamma(\frac{1}{2} - \nu)(x^{2} - 1)^{\nu + \frac{1}{2}}} dx$$

$$= \frac{2^{\nu}}{\sqrt{\pi} \Gamma(\frac{1}{2} - \nu)} \int_{0}^{1} \frac{u^{\nu - \frac{1}{2}}}{(a^{2}u + 1)(1 - u)^{\nu + \frac{1}{2}}} du = \frac{2^{\nu} \Gamma(\nu + \frac{1}{2})}{\sqrt{\pi}(a^{2} + 1)^{\nu + \frac{1}{2}}}.$$
(7.4.8)†

Here the L^p theory applies if p satisfies (7.4.5). The result holds by analytic continuation if $\nu > -\frac{1}{2}$.

7.5. Some integrals of Ramanujan. || Let

$$\phi(x) = \int_{-\infty}^{\infty} \frac{e^{-i\pi u^{i}}}{\cosh \pi u} e^{-ixu} du.$$
 (7.5.1)

Then

$$\phi(x+i\pi) + \phi(x-i\pi) = 2 \int_{-\infty}^{\infty} e^{-i\pi u^{2}-ixu} du$$

$$= 2e^{\frac{ix^{2}}{4\pi} - \frac{i\pi}{4}}.$$
(7.5.2)

Again, by (7.1.8) and (7.1.9) the transform of $e^{-i\pi x^2}$ is $(2\pi)^{-\frac{i}{4}}e^{\frac{ix^3}{4}-\frac{i\pi}{4}}$,

and that of sech πx is $(2\pi)^{-\frac{1}{2}}$ sech $\frac{1}{2}x$. Hence, by (2.1.8),

$$\phi(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{e^{\frac{i(x-u)^2}{4\pi} - \frac{i\pi}{4}}}{\cosh \frac{1}{2}u} du = e^{\frac{ix^2}{4\pi} - \frac{i\pi}{4}} \int_{-\infty}^{\infty} \frac{e^{i\pi v^2 - ixv}}{\cosh \pi v} dv.$$

This integral is of the same form as the original one, and we can repeat the process which gave (7.5.2). We obtain

$$e^{-\frac{i(x+i\pi)^{3}}{4\pi}}\phi(x+i\pi)+e^{-\frac{i(x-i\pi)^{2}}{4\pi}}\phi(x-i\pi)=2e^{-\frac{i\pi}{4}}\int_{-\infty}^{\infty}e^{i\pi v^{3}-ixv}\,dv$$

$$=2e^{-\frac{ix^{2}}{4\pi}},$$

i.e.

$$e^{\frac{i}{2}x}\phi(x+i\pi)+e^{-\frac{i}{2}x}\phi(x-i\pi)=2e^{-\frac{i\pi}{4}}.$$
 (7.5.3)

Eliminating $\phi(x-i\pi)$,

$$(e^{\frac{i}{4}x} - e^{-\frac{i}{4}x})\phi(x + i\pi) = 2e^{-\frac{i\pi}{4}} (1 - e^{\frac{i\pi^2}{4} - \frac{i}{4}x}), \tag{7.5.4}$$

and, replacing x by $x-i\pi$, we obtain

$$\phi(x) = \frac{e^{i\pi} - ie^{ix}}{\cosh \frac{1}{2}x}.$$
 (7.5.5)

Taking real and imaginary parts, we obtain the formulae

$$\int_{0}^{\infty} \frac{\cos \pi u^{2} \cos x v}{\cosh \pi u} du = \frac{\sin \frac{x^{2}}{4\pi} + \frac{1}{\sqrt{2}}}{2 \cosh \frac{1}{2}x},$$
 (7.5.6)

$$\int_{0}^{\infty} \frac{\sin \pi u^{2} \cos x u}{\cosh \pi u} du = \frac{\cos \frac{x^{2}}{4\pi} - \frac{1}{\sqrt{2}}}{2 \cosh \frac{1}{2}x}.$$
 (7.5.7)

Similar integrals with denominator $\sinh \pi u$ may be evaluated in a similar way, or deduced from the previous ones, as follows. We have

$$\phi(x+i\pi) = \int_{-\infty}^{\infty} e^{-i\pi u^{2}-ixu} \frac{e^{\pi u}}{\cosh \pi u} du$$

$$= \int_{-\infty}^{\infty} e^{-i\pi u^{2}-ixu} (1+\tanh \pi u) du$$

$$= e^{\frac{ix^{2}}{4}-i\pi} - 2i \int_{0}^{\infty} e^{-i\pi u^{2}} \sin xu \tanh \pi u du. \quad (7.5.8)$$

Now by (7.5.4)

$$\phi(x+i\pi) - e^{\frac{ix^3}{4\pi} - \frac{i\pi}{4}} = e^{-\frac{i\pi}{4}} \left\{ 1 - e^{\frac{ix^3}{4\pi} - \frac{i}{4}x} - e^{\frac{ix^3}{4\pi}} \right\}$$

$$= \frac{e^{-\frac{i\pi}{4}}}{\sinh \frac{1}{2}x} \left(1 - e^{\frac{ix^3}{4\pi}} \cosh \frac{1}{2}x \right).$$

$$\tanh \pi u = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{\sin uv}{\sinh \frac{1}{2}v} \, dv.$$

Also

Hence the second term on the right of (7.5.8) is

$$-\frac{2i}{\pi} \int_{0}^{\infty} e^{-i\pi u^{2}} \sin xu \ du \int_{0}^{\infty} \frac{\sin uv}{\sinh \frac{1}{2}v} \ dv$$

$$= -\frac{2i}{\pi} \int_{0}^{\infty} \frac{dv}{\sinh \frac{1}{2}v} \int_{0}^{\infty} e^{-i\pi u^{4}} \sin xu \sin vu \ du$$

$$= -\frac{1}{\pi} e^{\frac{ix^{4}}{4\pi} - \frac{i\pi}{4}} \int_{0}^{\infty} \frac{e^{i\pi^{2}} \sin(xv/2\pi)}{\sinh \frac{1}{2}v} \ dv$$

$$= -2e^{\frac{ix^{4}}{4\pi} - \frac{i\pi}{4}} \int_{0}^{\infty} \frac{e^{i\pi v^{4}} \sin xy}{\sinh \pi y} \ dy.$$
Hence
$$\int_{0}^{\infty} \frac{e^{i\pi v^{4}} \sin xy}{\sinh \pi y} \ dy = \frac{\cosh \frac{1}{2}x - e^{-\frac{ix^{4}}{4\pi}}}{2 \sinh \frac{1}{2}x},$$
i.e.
$$\int_{0}^{\infty} \frac{\cos \pi y^{2} \sin xy}{\sinh \pi y} \ dy = \frac{\cosh \frac{1}{2}x - \cos(x^{2}/4\pi)}{2 \sinh \frac{1}{2}x},$$
(7.5.9)
and
$$\int_{0}^{\infty} \frac{\sin \pi y^{2} \sin xy}{\sinh \pi y} \ dy = \frac{\sin(x^{2}/4\pi)}{2 \sinh \frac{1}{2}x}.$$
(7.5.10)

7.6. Some Γ-function formulae.† The formula

$$\int_{-\frac{1}{2}\pi}^{\frac{1}{2}\pi} (\cos t)^{a-2} e^{ixt} dt = \frac{\pi \Gamma(a-1)}{2^{a-2} \Gamma(\frac{1}{2}a + \frac{1}{2}x) \Gamma(\frac{1}{2}a - \frac{1}{2}x)} \quad (a > 1)$$
† Ramanujan (4), (6).

may be obtained by calculating

$$\int \left(w + \frac{1}{w}\right)^{a-2} w^{x-1} dw$$

taken round the contour formed by joining the points -i, i by the imaginary axis, and by the right-hand half of the unit circle.

The reciprocal formula is

$$\int_{-\infty}^{\infty} \frac{e^{-ixt}}{\Gamma(\frac{1}{2}a + \frac{1}{2}x)\Gamma(\frac{1}{2}a - \frac{1}{2}x)} dx = \frac{2^{a-1}(\cos t)^{a-2}}{\Gamma(a-1)} \quad (|t| < \frac{1}{2}\pi)$$

$$= 0 \quad (|t| \geqslant \frac{1}{2}\pi), \quad (7.6.2)$$

or, putting $a = \alpha + \beta$, $x = 2u + \alpha - \beta$, $t = \frac{1}{2}y$,

$$\int_{-\infty}^{\infty} \frac{e^{-iuy}}{\Gamma(\alpha+u)\Gamma(\beta-u)} du = \frac{(2\cos\frac{1}{2}y)^{\alpha+\beta-2}e^{\frac{1}{2}iy(\alpha-\beta)}}{\Gamma(\alpha+\beta-1)} \quad (|y| < \pi)$$

$$= 0 \quad (|y| \ge \pi).$$
(7.6.3)

Here

$$F(x) = \{\Gamma(\alpha+u)\Gamma(\beta-u)\}^{-1} = O(|u|^{1-\alpha-\beta})$$

as $u \to \pm \infty$. The functions F(x), f(x), related by (7.6.3) both belong to L^p ($p \ge 1$) if $\alpha + \beta > 2$; if $1 < \alpha + \beta \le 2$ they belong to L^p , $L^{p'}$ respectively if $p(\alpha + \beta - 1) > 1$. In the latter case (7.6.3) is non-absolutely convergent; this may be verified from the asymptotic expressions for the Γ -functions, or by Theorem 59 and its extension to L^p .

The particular case y = 0 is

$$\int_{-\infty}^{\infty} \frac{du}{\Gamma(\alpha+u)\Gamma(\beta-u)} = \frac{2^{\alpha+\beta-2}}{\Gamma(\alpha+\beta-1)} \quad (\alpha+\beta>1).$$
 (7.6.4)

Since

$$\sin m\pi u/\sin \pi u = e^{i(m-1)\pi u} + e^{i(m-3)\pi u} + ... + e^{-i(m-1)\pi u},$$

(7.6.3), with $\alpha = \beta$, gives

$$\int_{0}^{\infty} \frac{\sin m\pi u}{\sin \pi u} \frac{du}{\Gamma(\alpha+u)\Gamma(\alpha-u)} = \frac{2^{2\alpha-3}}{\Gamma(2\alpha-1)} \quad (m \text{ odd}), \qquad 0 \quad (m \text{ even}).$$
(7.6.5)

The particular case $\alpha = n+1$ (n an integer) is

$$\int_{0}^{\infty} \frac{\sin m\pi u \, du}{u \left(1 - \frac{u^2}{1^2}\right) \left(1 - \frac{u^2}{2^2}\right) \dots \left(1 - \frac{u^2}{n^2}\right)} = \frac{\pi}{2} \frac{2^{2n} (n!)^2}{2n!} \quad (m \text{ odd}), \qquad 0 \quad (m \text{ even}).$$
(7.6.6)

Again, apply (2.1.1) with

$$F(x) = \frac{1}{\Gamma(\alpha+x)\Gamma(\beta-x)}, \qquad G(x) = \frac{1}{\Gamma(\gamma+x)\Gamma(\delta-x)}.$$

Then, by (7.6.3), if $\alpha+\beta+\gamma+\delta > 3$,

$$\int_{-\infty}^{\infty} \frac{dx}{\Gamma(\alpha+x)\Gamma(\beta-x)\Gamma(\gamma+x)\Gamma(\delta-x)}$$

$$= \frac{1}{2\pi} \frac{1}{\Gamma(\alpha+\beta-1)\Gamma(\gamma+\delta-1)} \int_{-\pi}^{\pi} (2\cos\frac{1}{2}y)^{\alpha+\beta+\gamma+\delta-4} e^{\frac{1}{2}iy(\alpha-\beta-\gamma+\delta)} dy$$

$$= \frac{\Gamma(\alpha+\beta+\gamma+\delta-3)}{\Gamma(\alpha+\beta-1)\Gamma(\gamma+\delta-1)\Gamma(\alpha+\delta-1)\Gamma(\beta+\gamma-1)}, (7.6.7)$$

using (7.6.1) again. Here F and g are L^p if $2-\gamma-\delta<1/p<\alpha+\beta-1$. The formula (2.1.8) with the same functions and $x=\pi, \alpha+\delta=\beta+\gamma$, gives

$$\int_{-\infty}^{\infty} \frac{e^{-i\pi x}}{\Gamma(\alpha+x)\Gamma(\beta-x)\Gamma(\gamma+x)\Gamma(\delta-x)} dx$$

$$= \frac{1}{2\pi} \frac{e^{\frac{1}{2}i\pi(\alpha-\beta)}}{\Gamma(\alpha+\beta-1)\Gamma(\gamma+\delta-1)} \int_{0}^{\pi} (2\cos\frac{1}{2}y)^{\alpha+\beta-2} (2\sin\frac{1}{2}y)^{\gamma+\delta-2} dy$$

$$= \frac{e^{\frac{1}{2}i\pi(\alpha-\beta)}}{2\Gamma(\frac{1}{2}(\alpha+\beta))\Gamma(\frac{1}{2}(\gamma+\delta))\Gamma(\alpha+\delta-1)}.$$
(7.6.8)

In particular,

$$\int_{0}^{\infty} \frac{\cos \pi x}{\{\Gamma(\alpha+x)\Gamma(\alpha-x)\}^{2}} dx = \frac{1}{4\Gamma(2\alpha-1)\{\Gamma(\alpha)\}^{2}}.$$
 (7.6.9)

Other integrals which may be evaluated in the same way are

$$\int_{-\infty}^{\infty} \frac{e^{i\pi x}}{\Gamma(\alpha+x)\Gamma(\beta-x)\Gamma(\gamma+2x)\Gamma(\delta-2x)} dx$$

$$= \frac{2^{\alpha+\beta+\gamma+\delta-5}e^{\frac{1}{2}i\pi(\beta-\alpha)}\Gamma(\frac{1}{2}(\alpha+\beta+\gamma+\delta-3))}{\sqrt{\pi}\Gamma(\frac{1}{2}(\alpha+\beta))\Gamma(\gamma+\delta-1)\Gamma(2\alpha+\delta-2)} (7.6.10)$$

provided that $2(\alpha - \beta) = \gamma - \delta$; if $\alpha + \beta + \gamma + \delta = 4$, then

$$\int_{-\infty}^{\infty} \frac{\cos \pi (x+\beta+\gamma)}{\Gamma(\alpha+x)\Gamma(\beta-x)\Gamma(\gamma+2x)\Gamma(\delta-2x)} dx$$

$$= \frac{1}{2\Gamma(\gamma+\delta-1)\Gamma(2\alpha+\delta-2)\Gamma(2\beta+\gamma-2)}. \quad (7.6.11)$$

If $2(\alpha-\beta) = \gamma - \delta + k$, where k is ± 1 or ± 2 , then

$$\int_{-\infty}^{\infty} \frac{\sin \pi (2x + \alpha - \beta)}{\Gamma(\alpha + x)\Gamma(\beta - x)\Gamma(\gamma + 2x)\Gamma(\delta - 2x)} dx$$

$$= \frac{2^{2\alpha - \gamma - 3}}{\sqrt{\pi}\Gamma(\beta + \gamma - \alpha + \frac{1}{3})\Gamma(2\alpha + \delta - 2)}. \quad (7.6.12)$$

If $3(\alpha - \beta) = \gamma - \delta + k$, where k is ± 1 or ± 2 ,

$$\int_{-\infty}^{\infty} \frac{\sin \pi (2x + \alpha - \beta)}{\Gamma(\alpha + x)\Gamma(\beta - x)\Gamma(\gamma + 3x)\Gamma(\delta - 3x)} dx$$

$$= \pm \frac{3^{3\alpha + \frac{\alpha}{4}} \Gamma(2\alpha - \beta + \delta - 2)}{4\pi\Gamma(\gamma + \delta - 1)\Gamma(3\alpha + \delta - 3)}. \quad (7.6.13)$$

The sign on the right-hand side in each case is that of k.

We next take some integrals of a similar kind, but with Γ -functions in the numerator. Consider

$$I = \int_{-\infty}^{\infty} \frac{\Gamma(\alpha + x)}{\Gamma(\beta + x)} e^{ixt} dx \quad (\mathbf{I}(\alpha) < 0)$$

$$= \pi \int_{-\infty}^{\infty} \frac{1}{\Gamma(\beta + x)\Gamma(1 - \alpha - x)} \frac{e^{ixt}}{\sin \pi(\alpha + x)} dx. \quad (7.6.14)$$

Now

$$\frac{1}{\sin\pi(\alpha+x)} = \frac{2i}{e^{i\pi(\alpha+x)} - e^{-i\pi(\alpha+x)}} = 2i\sum_{m=0}^{\infty} e^{-i\pi(2m+1)(\alpha+x)}.$$

Hence

$$I = 2i\pi \sum_{m=0}^{\infty} \int_{-\infty}^{\infty} \frac{e^{ixt - i\pi(2m+1)(\alpha+x)}}{\Gamma(\beta+x)\Gamma(1-\alpha-x)} dx,$$

and these integrals are of the form (7.6.3), with $y = (2m+1)\pi - t$. Hence I = 0 if $t \le 0$. If t > 0, the only non-zero term is that in which $m = \lfloor t/2\pi \rfloor$; the value of I may thus be obtained from (7.6.3).

We can now pass to

$$\int_{-\infty}^{\infty} \Gamma(\alpha+x)\Gamma(\beta-x)e^{ixt} dx \qquad (7.6.15)$$

in a similar way. This is

$$\pi \int_{-\infty}^{\infty} \frac{\Gamma(\alpha+x)}{\Gamma(1-\beta+x)} \frac{e^{ixt}}{\sin \pi(\beta-x)} dx$$

$$= 2i\pi \sum_{m=0}^{\infty} \int_{-\infty}^{\infty} \frac{\Gamma(\alpha+x)}{\Gamma(1-\beta+x)} e^{ixt-i\pi(2m+1)(\beta-x)} dx$$

if $I(\beta) < 0$. Hence (7.6.15) can be evaluated in terms of (7.6.14).

The above results may be used to evaluate some integrals involving Bessel functions, in which the order is the variable of integration.† Using (7.6.4), we have

$$\int_{-\infty}^{\infty} \frac{J_{\mu+x}(a)}{a^{\mu+x}} \frac{J_{\nu-x}(b)}{b^{\nu-x}} dx$$

$$= \sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{(-1)^{m+n}}{m! \, n!} \left(\frac{1}{2}\right)^{\mu+\nu+2m+2n} \int_{-\infty}^{\infty} \frac{a^{2m}b^{2n}}{\Gamma(\mu+x+m+1)\Gamma(\nu-x+n+1)} dx$$

$$= \sum_{m=0}^{\infty} \sum_{n=0}^{\infty} \frac{(-1)^{m+n}}{m! \, n!} \frac{a^{2m}b^{2n}}{2^{\mu+\nu+2m+2n}} \frac{2^{\mu+\nu+m+n}}{\Gamma(\mu+\nu+m+n+1)}$$

$$= \sum_{r=0}^{\infty} \frac{(-1)^r}{2^r\Gamma(\mu+\nu+r+1)} \sum_{m=0}^r \frac{a^{2m}b^{2r-2m}}{m! \, (r-m)!}$$

$$= \sum_{r=0}^{\infty} \frac{(-1)^r}{2^r\Gamma(\mu+\nu+r+1)} \frac{(a^2+b^2)^r}{r!},$$
i.e.
$$\int_{-\infty}^{\infty} \frac{J_{\mu+x}(a)}{a^{\mu+x}} \frac{J_{\nu-x}(b)}{b^{\nu-x}} dx = \frac{J_{\mu+\nu}[\sqrt{\{2(a^2+b^2)\}}]}{\{\frac{1}{2}(a^2+b^2)\}^{\frac{1}{2}(\mu+\nu)}}.$$
(7.6.16)

In particular $\int_{-\infty}^{\infty} J_{\mu+x}(a)J_{\nu-x}(a) \ dx = J_{\mu+\nu}(2a). \tag{7.6.17}$

The values of corresponding integrals containing a factor e^{inx} may be deduced in the same way from (7.6.3).

7.7. Mellin transforms. The simplest example of Mellin transforms is $f(x) = e^{-x}$, $\mathfrak{F}(s) = \Gamma(s)$ $(\sigma > 0)$. (7.7.1)

† Watson, § 13.8.

Here $f(x)x^{k-1}$ belongs to $L(0,\infty)$ if k>0; and $\mathfrak{F}(s)$ belongs to $L(k-i\infty,k+i\infty)$ for k>0.

Other straightforward examples are

1
$$(x < a)$$
, $0 (x \ge a)$, a^s/s $(\sigma > 0)$, $(7.7.2)$

$$\log(a/x)$$
 $(x < a)$, 0 $(x \geqslant a)$, a^{s}/s^{2} $(\sigma > 0)$, $(7.7.3)$

$$\frac{1}{e^x-1}$$
, $\Gamma(s)\zeta(s)$ $(\sigma>1)$, $(7.7.4)$

$$1$$
 $(x < a)$, 0 $(x \geqslant a)$, a^s/s $(\sigma > 0)$, $(7.7.2)$ $\log(a/x)$ $(x < a)$, 0 $(x \geqslant a)$, a^s/s^2 $(\sigma > 0)$, $(7.7.3)$ $\frac{1}{e^x-1}$, $\Gamma(s)\zeta(s)$ $(\sigma > 1)$, $(7.7.4)$ $\Gamma(s)L(s)$ $(\sigma > 0)$, $(7.7.5)$

where L(s) is (9.12.1). Here $f(x)x^{k-1}$ belongs to $L(0,\infty)$ for k>0 in each case except (7.7.4), when it is k > 1.

We also observe that if f(x) and $\mathfrak{F}(s)$ are Mellin transforms so are $x^{\lambda}f(x)$ and $\mathfrak{F}(s+\lambda)$, and also $f(x^{\alpha})$ and $\frac{1}{\alpha}\mathfrak{F}\left(\frac{s}{\alpha}\right)$. This enables us to introduce parameters in each case.

Consider next the integral

$$f(z) = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(s)\Gamma(a-s)\Gamma(b-s)}{\Gamma(c-s)} \frac{\Gamma(c)}{\Gamma(a)\Gamma(b)} z^{-s} ds, \quad (7.7.6)$$

where R(a) > 0, R(b) > 0, c is not 0, -1, ..., and

$$0 < k < \min\{\mathbf{R}(a), \mathbf{R}(b)\}.$$

Since

$$\Gamma(s)\Gamma(a-s)\Gamma(b-s)/\Gamma(c-s) = O(e^{-\pi|t|}|t|^{\mathbf{R}(a+b-c)-1}), \quad |z^{-s}| = r^{-\sigma}e^{\theta t},$$

the integral represents an analytic function of z, regular for r > 0, $-\pi < \theta < \pi$. If z = x, where 0 < x < 1, it may be evaluated by moving the line of integration away to infinity on the left, and evaluating the residues at $s = 0, -1, \dots$ We obtain

$$f(x) = 1 - \frac{ab}{c \cdot 1!} x + \frac{a(a+1)b(b+1)}{c(c+1) \cdot 2!} x^2 - \dots$$

= $F(a, b; c; -x)$

with the usual hypergeometric notation. For x > 1, f(x) is therefore the analytic continuation of this function (and may, of course, be expressed as a sum of hypergeometric series by moving the contour the other way). We therefore obtain as Mellin transforms

$$f(x) = F(a,b;c;-x), \qquad \mathfrak{F}(s) = \frac{\Gamma(s)\Gamma(a-s)\Gamma(b-s)}{\Gamma(c-s)} \frac{\Gamma(c)}{\Gamma(a)\Gamma(b)}$$
$$(0 < \sigma < \min\{\mathbf{R}(a),\mathbf{R}(b)\}). \quad (7.7.7)$$

Particular cases are

where $P_n(x)$ is the Legendre polynomial of degree n.

In each case $\mathfrak{F}(\sigma+it)$ belongs to $L(-\infty,\infty)$ for the range of values of σ stated. In cases (7.7.8), (7.7.9), and (7.7.10) the integral

$$\int_{0}^{\infty} f(x)x^{s-1} \, dx \tag{7.7.12}$$

can easily be proved to be equal to $\mathfrak{F}(s)$.

Another Mellin pair of the same type is

$$f(x) = \frac{\{\sqrt{(x^2+1)} - x\}^a}{\sqrt{(x^2+1)}}, \qquad \mathfrak{F}(s) = 2^{-s} \frac{\Gamma(s)\Gamma(\frac{1}{2} + \frac{1}{2}a - \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}a + \frac{1}{2}s)}$$
$$(0 < \sigma < \mathbf{R}(a) + 1). \quad (7.7.13)$$

Here the integral (7.7.12) may be evaluated by putting

$$x = \frac{1}{2}y/\sqrt{y+1}.$$

Another class of Mellin transforms is

$$f(x) = \begin{cases} (1-x)^{a-1} & (0 < x < 1), \\ 0 & (x \ge 1), \end{cases} \quad \mathfrak{F}(s) = \frac{\Gamma(s)\Gamma(a)}{\Gamma(s+a)} \quad \begin{pmatrix} \sigma > 0, \\ R(a) > 0 \end{pmatrix},$$

$$\begin{cases} 0 & (0 < x \le 1), & \Gamma(a-s)\Gamma(1-a) \\ (x-1)^{-a} & (x > 1), & \Gamma(1-s) \end{cases} \quad (\sigma < R(a) < 1),$$

$$\begin{cases} 0 & (0 < x \le 1), & (7.7.15) \end{cases}$$

$$\begin{cases} 0 & (0 < x \le 1), & (7.7.15) \end{cases}$$

$$\begin{cases} (x-\sqrt{(x^2-1)})^a + \{x-\sqrt{(x^2-1)}\}^{-a} & 2^{-s}\Gamma(\frac{1}{2}a + \frac{1}{2} - \frac{1}{2}s)\Gamma(\frac{1}{2} - \frac{1}{2}s - \frac{1}{2}a) \\ \Gamma(1-s) & \Gamma(1-s) \end{cases}$$

$$(x > 1), & (\sigma < |R(a)| + 1), \quad (7.7.16) \end{cases}$$

$$\log \left| \frac{1+x}{1-x} \right|, & \frac{\pi}{s} \tan \frac{1}{2}s\pi \quad (-1 < \sigma < 1).$$

$$(7.7.17) \end{cases}$$

In each case f(x) belongs to a different analytic function for 0 < x < 1 and for x > 1, while $x^{k-1}f(x)$ belongs to $L(0,\infty)$ for some k. The evaluation of (7.7.12) is immediate in cases (7.7.14) and (7.7.15); for (7.7.16), put $x = \frac{1}{2}(y+1/y)$. For (7.7.17),

$$\int_{0}^{1} \log \binom{1+x}{1-x} x^{s-1} dx = 2 \int_{0}^{1} \left(x + \frac{x^{3}}{3} + \ldots \right) x^{s-1} dx = 2 \left(\frac{1}{1+s} + \frac{1}{3(3+s)} + \ldots \right),$$

$$\int_{1}^{\infty} \log \left(\frac{x+1}{x-1} \right) x^{s-1} dx = \int_{0}^{1} \log \left(\frac{1+u}{1-u} \right) u^{1-s} \frac{du}{u^{2}} = 2 \left(\frac{1}{1-s} + \frac{1}{3(3-s)} + \ldots \right),$$
and
$$\mathfrak{F}(s) = 4 \left(\frac{1}{1^{2}-s^{2}} + \frac{1}{3^{2}-s^{2}} + \ldots \right) = \frac{\pi}{s} \tan \frac{1}{2} s\pi.$$

7.8. Further gamma-function formulae. In (2.1.12) let $f(x)=x^ae^{-x}$, $\mathfrak{F}(s)=\Gamma(s+a)$, $g(x)=x^{b-1}e^{-x}$, $\mathfrak{G}(s)=\Gamma(s+b-1)$. Then

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(a+s)\Gamma(b-s) ds = \int_{0}^{\infty} x^{a+b-1}e^{-2x} dx$$

$$= 2^{-a-b}\Gamma(a+b) \quad (-a < k < b). \quad (7.8.1)$$

This process and the following ones are justified by Theorem 42. The result is a particular case of the reciprocity (7.7.9).

Taking b = a and the line of integration the imaginary axis, we obtain

$$\int_{0}^{\infty} |\Gamma(a+it)|^{2} dt = 2^{-2a} \pi \Gamma(2a) \quad (a > 0), \tag{7.8.2}$$

and there are similar particular cases of the following formulae.

Next let

$$f(x) = \frac{x^b}{(1+x)^a}, \quad \Re(s) = \frac{\Gamma(b+s)\Gamma(a-b-s)}{\Gamma(a)}.$$

and so g, \mathfrak{G} , with c, d for a, b. Then

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(b+s)\Gamma(a-b-s)\Gamma(d+1-s)\Gamma(c-d-1+s)}{\Gamma(a)\Gamma(c)} ds$$

$$= \int_{0}^{\infty} \frac{x^{b+d}}{(1+x)^{a+c}} dx = \frac{\Gamma(b+d+1)\Gamma(a+c-b-d-1)}{\Gamma(a+c)},$$

4362

or, writing $c-d-1=\alpha$, $b=\beta$, $1+d=\gamma$, $a-b=\delta$,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(\alpha+s)\Gamma(\beta+s)\Gamma(\gamma-s)\Gamma(\delta-s) ds$$

$$= \frac{\Gamma(\alpha+\gamma)\Gamma(\alpha+\delta)\Gamma(\beta+\gamma)\Gamma(\beta+\delta)}{\Gamma(\alpha+\beta+\gamma+\delta)} \quad (-\alpha < k, -\beta < k, \gamma > k, \delta > k).$$
(7.8.3)†

 $f(x)=x^b(1-x)^{a-1}$ $(0< x<1), \quad 0$ $(x>1), \quad \mathfrak{F}(s)=rac{\Gamma(b+s)\Gamma(a)}{\Gamma(a+b+s)},$ and so g, \mathfrak{G} , with c, d for a, b. Then

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(b+s)\Gamma(a)\Gamma(1+d-s)\Gamma(c)}{\Gamma(a+b+s)\Gamma(c+d+1-s)} ds = \int_{0}^{1} x^{b+d} (1-x)^{a+c-2} dx$$

$$= \frac{\Gamma(b+d+1)\Gamma(a+c-1)}{\Gamma(a+b+c+d)},$$

or, writing $a = \beta - \alpha$, $b = \alpha$, $c = \delta - \gamma$, $d = \gamma - 1$,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\alpha+s)\Gamma(\gamma-s)}{\Gamma(\beta+s)\Gamma(\delta-s)} ds = \frac{\Gamma(\alpha+\gamma)\Gamma(\beta+\delta-\alpha-\gamma-1)}{\Gamma(\beta-\alpha)\Gamma(\delta-\gamma)\Gamma(\beta+\delta-1)} (-\alpha < k, -\beta < k, \gamma > k, \delta > k). \quad (7.8.4)$$

Defining f(x) as in the last example and g(x) as in (7.8.3), we obtain

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(b+s)\Gamma(a)\Gamma(d+1-s)\Gamma(c-d-1+s)}{\Gamma(a+b+s)\Gamma(c)} ds$$

$$= \int_{0}^{1} \frac{x^{b+d}(1-x)^{a-1}}{(1+x)^{c}} dx.$$

The integral can be evaluated in finite terms if c = 1-a. It is then

$$\int\limits_0^1 x^{b+d} (1-x^2)^{a-1} \, dx = \tfrac{1}{2} \int\limits_0^1 y^{\frac{1}{2}(b+d-1)} (1-y)^{a-1} \, dy = \frac{1}{2} \, \frac{\Gamma\{\frac{1}{2}(b+d+1)\}\Gamma(a)}{\Gamma\{a+\frac{1}{2}(b+d+1)\}}.$$

Putting $b = \alpha$, $a = 1 - \beta - \gamma$, $d = \gamma - 1$, we obtain

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\alpha+s)\Gamma(\beta+s)\Gamma(\gamma-s)}{\Gamma(1+\alpha-\beta-\gamma+s)} ds = \frac{\Gamma(\frac{1}{2}\alpha+\frac{1}{2}\gamma)\Gamma(\beta+\gamma)}{2\Gamma(1-\beta+\frac{1}{2}\alpha-\frac{1}{2}\gamma)} (-\alpha < k, -\beta < k, \gamma > k, \beta+\gamma-\alpha-1 < k). \quad (7.8.5)$$

† Barnes (1); see Whittaker and Watson, Modern Analysis, § 14.52.

Another formula of Barnes† is

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\alpha+s)\Gamma(\beta+s)\Gamma(\gamma+s)\Gamma(\delta-s)\Gamma(-s)}{\Gamma(\epsilon+s)} ds$$

$$= \frac{\Gamma(\alpha)\Gamma(\beta)\Gamma(\gamma)\Gamma(\alpha+\delta)\Gamma(\beta+\delta)\Gamma(\gamma+\delta)}{\Gamma(\epsilon-\alpha)\Gamma(\epsilon-\beta)\Gamma(\epsilon-\gamma)}, \quad (7.8.6)$$

where

To prove this, we use the formula

$$\frac{1}{2\pi i}\int_{k-i\infty}^{k+i\infty}\mathfrak{F}_{1}(s)\mathfrak{F}_{2}(s)\mathfrak{F}_{3}(s)\,ds=\int_{0}^{\infty}\int_{0}^{\infty}f_{1}(u)f_{2}(v)f_{3}\left(\frac{1}{uv}\right)\frac{dudv}{uv},$$

derived from (2.1.18) with n=2. Take

$$f_1(x) = rac{x^{lpha}}{(1+x)^{lpha+\delta}}, \qquad \qquad \mathfrak{F}_1(s) = rac{\Gamma(lpha+s)\Gamma(\delta-s)}{\Gamma(lpha+\delta)}, \ f_2(x) = rac{x^{eta}}{(1+x)^{eta}}, \qquad \qquad \mathfrak{F}_2(s) = rac{\Gamma(eta+s)\Gamma(-s)}{\Gamma(eta)}, \ f_3(x) = x^{\gamma}(1-x)^{\epsilon-\gamma-1} \ (0 < x < 1) \ 0 \ (x > 1), \qquad \mathfrak{F}_3(s) = rac{\Gamma(\gamma+s)\Gamma(\epsilon-\gamma)}{\Gamma(\epsilon+s)}.$$

Denoting the left-hand side of (7.8.6) by I, we obtain

$$\frac{\Gamma(\epsilon-\gamma)}{\Gamma(\alpha+\delta)\Gamma(\beta)}I=\iint\limits_{\Omega}\frac{u^{\alpha}}{(1+u)^{\alpha+\delta}}\frac{v^{\beta}}{(1+v)^{\beta}}\frac{1}{(uv)^{\gamma}}\Big(1-\frac{1}{uv}\Big)^{\epsilon-\gamma-1}\frac{dudv}{uv}.$$

Putting $u=\frac{1}{x}-1$, $v=\frac{1}{y}-1$, the right-hand side becomes $\iint\limits_{x+y<1} x^{\gamma+\delta-1}y^{\gamma-1}(1-x)^{\alpha-\epsilon}(1-y)^{\beta-\epsilon}(1-x-y)^{\epsilon-\gamma-1}\,dxdy.$

Putting y = z(1-x), we obtain

$$\int_{0}^{1} z^{\gamma-1} (1-z)^{\epsilon-\gamma-1} dz \int_{0}^{1} x^{\gamma+\delta-1} (1-x)^{\alpha-1} (1-z+zx)^{\beta-\epsilon} dx.$$

The inner integral can be evaluated in terms of Γ -functions if $\alpha + \beta + \gamma + \delta = \epsilon$. It is then equal to \ddagger

$$\frac{\Gamma(\alpha)\Gamma(\gamma+\delta)}{\Gamma(\alpha+\gamma+\delta)}\frac{1}{(1-z)^{\alpha}}.$$

Hence we obtain

$$\frac{\Gamma(\alpha)\Gamma(\gamma+\delta)}{\Gamma(\alpha+\gamma+\delta)}\int_{0}^{1}z^{\gamma-1}(1-z)^{\epsilon-\gamma-\alpha-1}\,dz=\frac{\Gamma(\alpha)\Gamma(\gamma+\delta)}{\Gamma(\alpha+\gamma+\delta)}\frac{\Gamma(\gamma)\Gamma(\epsilon-\gamma-\alpha)}{\Gamma(\epsilon-\alpha)},$$

and the result follows. The necessary inversions are all justified by absolute convergence.

7.9. Bessel functions. In (7.4.1) we may take a = s to be complex, provided that $0 < \sigma < \nu + \frac{3}{2}$. We thus obtain the Mellin transforms

$$x^{-\nu}J_{\nu}(x), \qquad \frac{2^{s-\nu-1}\Gamma(\frac{1}{2}s)}{\Gamma(\nu-\frac{1}{2}s+1)} \qquad (0 < \sigma < \nu+\frac{3}{2}).$$
 (7.9.1)†

Equivalent pairs are

$$J_{\nu}(x), \qquad \frac{2^{s-1}\Gamma(\frac{1}{2}s+\frac{1}{2}\nu)}{\Gamma(\frac{1}{2}\nu-\frac{1}{2}s+1)} \qquad (-\nu < \sigma < \frac{3}{2}), \tag{7.9.2}$$

$$x^{\nu}J_{\nu}(x), \qquad \frac{2^{s+\nu-1}\Gamma(\frac{1}{2}s+\nu)}{\Gamma(1-\frac{1}{2}s)} \qquad (-2\nu < \sigma < \frac{3}{2}-\nu), \qquad (7.9.3)$$

and
$$x^{\frac{1}{2}}J_{\nu}(x)$$
, $\frac{2^{s-\frac{1}{2}}\Gamma(\frac{1}{2}s+\frac{1}{2}\nu+\frac{1}{4})}{\Gamma(\frac{1}{2}\nu-\frac{1}{2}s+\frac{3}{4})}$ $(-\nu-\frac{1}{2}<\sigma<1)$. $(7.9.4)$

Taking $\nu = -\frac{1}{2}$, $\nu = \frac{1}{2}$ in the last pair, we obtain

$$\cos x, \qquad 2^{s-1} \sqrt{\pi} \frac{\Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{2} - \frac{1}{2}s)} = \Gamma(s) \cos \frac{1}{2} s \pi \quad (0 < \sigma < 1),$$
(7.9.5)

$$\sin x, \qquad 2^{s-1} \sqrt{\pi} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}s)}{\Gamma(1 - \frac{1}{2}s)} = \Gamma(s) \sin \frac{1}{2}s\pi \quad (-1 < \sigma < 1).$$
(7.9.6)

We define

$$Y_{\nu}(x) = \frac{J_{\nu}(x)\cos\nu\pi - J_{-\nu}(x)}{\sin\nu\pi}.$$

By (7.9.2) the Mellin transform of $Y_{\nu}(x)$ is

$$\begin{split} \frac{1}{\sin\nu\pi} & \left\{ \frac{2^{s-1}\Gamma(\frac{1}{2}s + \frac{1}{2}\nu)}{\Gamma(\frac{1}{2}\nu - \frac{1}{2}s + 1)} \cos\nu\pi - \frac{2^{s-1}\Gamma(\frac{1}{2}s - \frac{1}{2}\nu)}{\Gamma(1 - \frac{1}{2}\nu - \frac{1}{2}s)} \right\} \\ &= \frac{2^{s-1}\Gamma(\frac{1}{2}s + \frac{1}{2}\nu)\Gamma(\frac{1}{2}s - \frac{1}{2}\nu)}{\pi\sin\nu\pi} \{ \sin(\frac{1}{2}s - \frac{1}{2}\nu)\pi\cos\nu\pi - \sin(\frac{1}{2}s + \frac{1}{2}\nu)\pi \} \\ &= -2^{s-1}\pi^{-1}\Gamma(\frac{1}{2}s + \frac{1}{2}\nu)\Gamma(\frac{1}{2}s - \frac{1}{2}\nu)\cos(\frac{1}{2}s - \frac{1}{2}\nu)\pi. \end{split}$$

Hence we have the Mellin pairs

$$Y_{\nu}(x), \qquad -2^{s-1}\pi^{-1}\Gamma(\frac{1}{2}s+\frac{1}{2}\nu)\Gamma(\frac{1}{2}s-\frac{1}{2}\nu)\cos(\frac{1}{2}s-\frac{1}{2}\nu)\pi \quad (|\nu|<\sigma<\frac{3}{2}), \tag{7.9.7}$$

$$x^{-\nu}Y_{\nu}(x), \qquad -2^{s-\nu-1}\pi^{-1}\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2}s-\nu)\cos(\frac{1}{2}s-\nu)\pi$$

$$(\nu+|\nu|<\sigma<\nu+\frac{3}{2}). \quad (7.9.8)$$

† Watson, § 6.5 (7).

1 Ibid., § 13.24 (5).

From (7.9.2) we also deduce

$$\begin{split} J_{\nu}(x) + J_{-\nu}(x), & 2^s \pi^{-1} \Gamma(\tfrac{1}{2} s + \tfrac{1}{2} \nu) \Gamma(\tfrac{1}{2} s - \tfrac{1}{2} \nu) \sin \tfrac{1}{2} s \pi \cos \tfrac{1}{2} \nu \pi \\ & (|\nu| < \sigma < \tfrac{3}{2}), \quad (7.9.9) \\ J_{\nu}(x) - J_{-\nu}(x), & -2^s \pi^{-1} \Gamma(\tfrac{1}{2} s + \tfrac{1}{2} \nu) \Gamma(\tfrac{1}{2} s - \tfrac{1}{2} \nu) \cos \tfrac{1}{2} s \pi \sin \tfrac{1}{2} \nu \pi \end{split}$$

$$J_{\nu}(x) - J_{-\nu}(x), \qquad -2^{s}\pi^{-1}\Gamma(\frac{1}{2}s + \frac{1}{2}\nu)\Gamma(\frac{1}{2}s - \frac{1}{2}\nu)\cos\frac{1}{2}s\pi\sin\frac{1}{2}\nu\pi$$

$$(|\nu| < \sigma < \frac{3}{2}). \quad (7.9.10)$$

Again, by (7.9.1) and (7.9.8),

$$x^{-\nu}\{J_{\nu}(x)+iY_{\nu}(x)\}$$

$$= \frac{1}{2\pi^{2}i} \int_{k-i\infty}^{k+i\infty} 2^{s-\nu-1} \Gamma(\frac{1}{2}s) \Gamma(\frac{1}{2}s-\nu) \{ \sin(\frac{1}{2}s-\nu)\pi - i\cos(\frac{1}{2}s-\nu)\pi \} x^{-s} ds$$

$$= -\frac{1}{2\pi^{2}} \int_{k-i\infty}^{k+i\infty} 2^{s-\nu-1} \Gamma(\frac{1}{2}s) \Gamma(\frac{1}{2}s-\nu) e^{i(\frac{1}{2}s-\nu)\pi} x^{-s} ds,$$

and here we can (by analytic continuation) replace x by ix. We obtain \dagger

$$\begin{aligned} x^{-\nu} e^{-\frac{1}{2}i\pi\nu} \frac{2}{\pi i} e^{-\frac{1}{2}\nu\pi i} K_{\nu}(x) \\ &= -\frac{1}{2\pi^2} \int\limits_{k-i\infty}^{k+i\infty} 2^{s-\nu-1} \Gamma(\frac{1}{2}s) \Gamma(\frac{1}{2}s-\nu) e^{i\pi(\frac{1}{2}s-\nu)} x^{-s} e^{-\frac{1}{2}i\pi s} \, ds, \end{aligned}$$

so that we obtain as Mellin transforms

$$x^{-\nu}K_{\nu}(x), \qquad 2^{s-\nu-2}\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2}s-\nu) \quad (\sigma > \max(0,2\nu)). \quad (7.9.11)^{+}$$

An equivalent pair is

$$x^{\nu}K_{\nu}(x)$$
, $2^{s+\nu-2}\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2}s+\nu)$ $(\sigma > \max(0,-2\nu))$. (7.9.12)

Hence we verify that $K_{\nu}(x)$ is an even function of ν . For $\nu = \frac{1}{2}$ (7.9.12) reduces to (7.7.1).

From (7.4.2) we obtain the pair

$$x^{-\nu}\mathbf{H}_{\nu}(x), \qquad \frac{2^{s-\nu-1}\Gamma(\frac{1}{2}s)\tan\frac{1}{2}s\pi}{\Gamma(\nu-\frac{1}{2}s+1)} \quad (-1 < \sigma < \nu+\frac{3}{2}), \quad (7.9.13)$$

and variants of this can easily be obtained in the above way.

To justify the inversion formulae in the above cases, consider e.g. (7.9.1). We obtained (7.4.1) directly; the inverse formula is

$$\frac{1}{2\pi i} \lim_{\lambda \to \infty} \int_{k-i\lambda}^{k+i\lambda} \frac{2^{s-\nu-1}\Gamma(\frac{1}{2}s)}{\Gamma(\nu-\frac{1}{2}s+1)} x^{-s} \, ds = \frac{J_{\nu}(x)}{x^{\nu}}.$$
 (7.9.14)

† Watson, § 3.7 (8).

‡ Ibid., § 6.5 (3-6).

This follows from Theorem 28 if $0 < k < \nu + \frac{1}{2}$, and from Theorem 30 if $0 < k \le \nu + 1$. The leading terms in the asymptotic expansion of $x^{-\nu}J_{\nu}(x)$ are of the form $x^{-\nu-\frac{1}{2}}(a\cos x + b\sin x)$. Hence, in Theorem 30,

$$\phi(x) = e^{(k-\nu-\frac{1}{2})x}, \qquad \psi(x) = e^x,$$

and the crucial condition (1.12.1) is satisfied if $k \leq \nu + 1$.

We might begin by proving (7.9.14) by the calculus of residues. We have then to deduce (7.4.1). We have as $t \to \infty$

$$\frac{2^{s-\nu-1}\Gamma(\frac{1}{2}s)}{\Gamma(\nu-\frac{1}{2}s+1)} = e^{\frac{1}{2}(\nu-1)i\pi+i(t\log t-t)}t^{\sigma-\nu-1}\left\{1+\frac{a}{t}+O\left(\frac{1}{t^2}\right)\right\}.$$

The result follows from Theorem 29 if $0 < k < \nu$. We can also apply Theorem 11; here

$$\phi(t) = t^{k-\nu-1}, \qquad \psi(t) = t \log t - t,$$

and (1.12.1) is satisfied if $k < \nu + \frac{3}{2}$.

7.10. Products of Bessel functions. By (2.1.16) and (7.9.1) the Mellin transform of $x^{-\mu-\nu}J_{\mu}(x)J_{\nu}(x)$ is

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} 2^{w-\mu-1} \frac{\Gamma(\frac{1}{2}w)}{\Gamma(1+\mu-\frac{1}{2}w)} 2^{s-w-\nu-1} \frac{\Gamma(\frac{1}{2}s-\frac{1}{2}w)}{\Gamma(1+\nu-\frac{1}{2}s+\frac{1}{2}w)} dw,$$

and putting w = 2w' and using (7.8.4), we obtain the Mellin pairt

$$\frac{J_{\mu}(x)J_{\nu}(x)}{x^{\mu+\nu}}, \frac{2^{s-\mu-\nu-1}\Gamma(\frac{1}{2}s)\Gamma(1+\mu+\nu-s)}{\Gamma(1+\nu-\frac{1}{2}s)\Gamma(1+\mu-\frac{1}{2}s)\Gamma(1+\mu+\nu-\frac{1}{2}s)}.$$
(7.10.1)

Similarly, by (7.9.11), the Mellin transform of $x^{-\mu-\nu}K_{\mu}(x)K_{\nu}(x)$ is

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} 2^{w-\mu-2} \Gamma(\frac{1}{2}w) \Gamma(\frac{1}{2}w-\mu) 2^{s-w-\nu-2} \Gamma(\frac{1}{2}s-\frac{1}{2}w) \Gamma(\frac{1}{2}s-\frac{1}{2}w-\nu) \ dw,$$

and, using (7.8.3), we obtain the Mellin pair

$$\frac{K_{\mu}(x)K_{\nu}(x)}{x^{\mu+\nu}}, \qquad \frac{2^{s-\mu-\nu-1}\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2}s-\mu)\Gamma(\frac{1}{2}s-\nu)\Gamma(\frac{1}{2}s-\mu-\nu)}{\Gamma(s-\mu-\nu)}. \tag{7.10.2}$$

From (7.9.1) and (7.9.11), the Mellin transform of $x^{-2\nu}J_{\nu}(x)K_{\nu}(x)$ is

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} 2^{w-\nu-2} \Gamma(\frac{1}{2}w) \Gamma(\frac{1}{2}w-\nu) 2^{s-ic-\nu-1} \frac{\Gamma(\frac{1}{2}s-\frac{1}{2}w)}{\Gamma(1+\nu-\frac{1}{2}s+\frac{1}{2}w)} dw,$$

† Watson 13.41 (1), (2), 13.33 (1).

and, using (7.8.5), we obtain the Mellin pair

$$x^{-2\nu}J_{\nu}(x)K_{\nu}(x), \qquad \frac{2^{s-2\nu-2}\Gamma(\frac{1}{4}s)\Gamma(\frac{1}{2}s-\nu)}{\Gamma(1+\nu-\frac{1}{4}s)}. \qquad (7.10.3)$$

By combining particular cases of (7.10.1), we obtain as Mellin transforms

$$J_{\nu}(x)Y_{\nu}(x), \qquad -\frac{1}{2\sqrt{\pi}} \frac{\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2}s+\nu)}{\Gamma(\frac{1}{3}s+\frac{1}{3})\Gamma(1+\nu-\frac{1}{3}s)}. \tag{7.10.4}$$

Other particular cases of (7.10.1) give the Mellin transforms

$$\cos x J_{\nu}(x), \qquad \frac{2^{s-1}\pi^{\frac{1}{2}}\Gamma(\frac{1}{2}s+\frac{1}{2}\nu)\Gamma(\frac{1}{2}-s)}{\Gamma(1+\frac{1}{2}\nu-\frac{1}{2}s)\Gamma(\frac{1}{2}-\frac{1}{2}\nu-\frac{1}{2}s)\Gamma(\frac{1}{2}+\frac{1}{2}\nu-\frac{1}{2}s)}, \qquad (7.10.5)$$

$$\sin x J_{\nu}(x), \qquad \frac{2^{s-1}\pi^{\frac{1}{2}}\Gamma(\frac{1}{2}s+\frac{1}{2}\nu+\frac{1}{2})\Gamma(\frac{1}{2}-s)}{\Gamma(\frac{1}{2}+\frac{1}{2}\nu-\frac{1}{2}s)\Gamma(1-\frac{1}{2}\nu-\frac{1}{2}s)\Gamma(1+\frac{1}{2}\nu-\frac{1}{2}s)}.$$

$$(7.10.6)$$

Combining these, we have

$$e^{ix}J_{\nu}(x) = \frac{1}{2\pi i} \int_{-\infty}^{k+i\infty} \frac{2^{-s}}{\sqrt{\pi}} \frac{\Gamma(s+\nu)\Gamma(\frac{1}{2}-s)}{\Gamma(1+\nu-s)} e^{\frac{i}{2}i\pi(s+\nu)}x^{-s} ds.$$

As in (7.9.11), we may now replace x by ix, and obtain the Mellin transforms

$$e^{-x}I_{\nu}(x), \qquad \frac{\Gamma(s+\nu)\Gamma(\frac{1}{2}-s)}{2^{s}\pi^{\frac{1}{2}}\Gamma(1+\nu-s)}.$$
 (7.10.7)

Again, from (7.10.1) and (7.10.4), the Mellin transform of

$$J_{
u}(x)\{J_{
u}(x)+iY_{
u}(x)\} \ -ie^{rac{1}{4}is\pi} \Gamma(rac{1}{2}s+
u)\Gamma(rac{1}{2}s)\Gamma(rac{1}{2}-rac{1}{2}s)}{\Gamma(1+
u-rac{1}{8}s)},$$

is

and hence, replacing x by ix, we obtain as Mellin transforms

$$I_{\nu}(x)K_{\nu}(x), \qquad \frac{\Gamma(\frac{1}{2}s+\nu)\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2}-\frac{1}{2}s)}{4\sqrt{\pi}\Gamma(1+\nu-\frac{1}{2}s)}. \qquad (7.10.8)$$

Similarly, the Mellin transform of $e^{-ix}\{J_{\nu}(x)+iY_{\nu}(x)\}$ is

$$\frac{-2^{1-s}i}{\pi^{\frac{3}{2}}}e^{\frac{1}{2}i\pi(s-\nu)}\cos\nu\pi\,\Gamma(\frac{1}{2}-s)\Gamma(s+\nu)\Gamma(s-\nu),$$

and replacing x by ix we obtain

$$e^{x}K_{\nu}(x),$$
 $2^{-s}\pi^{-\frac{1}{2}}\cos\nu\pi\Gamma(\frac{1}{2}-s)\Gamma(s+\nu)\Gamma(s-\nu).$ (7.10.9)†

The processes of this section can be justified by Theorem 73; for † 1bid., § 6.51.

example, $x^k x^{-\mu} J_{\mu}(x)$ belongs to \mathfrak{L}^2 if $0 < k < \mu + \frac{1}{2}$, and $x^{\sigma - k} x^{-\nu} J_{\nu}(x)$ belongs to \mathfrak{L}^2 if $0 < \sigma - k < \nu + \frac{1}{2}$. We can choose σ and k to satisfy these conditions if $\mu > -\frac{1}{2}$, $\nu > -\frac{1}{2}$. This would not include (7.10.5) directly; for this we could consider first $(1 - \cos x) J_{\nu}(x)$, $x^k (1 - \cos x)$ belonging to \mathfrak{L}^2 if -2 < k < 0. We can of course also extend the ranges of validity of the formulae by analytic continuation.

7.11. Integrals involving Bessel functions. We are now in a position to evaluate a large class of integrals involving Bessel functions.

By transformation from (7.7.15), we have the Mellin pair

$$(x^2-1)^{-\nu-\frac{1}{2}}$$
 $(x>1)$, 0 $(x<1)$, $\frac{\Gamma(\nu+\frac{1}{2}-\frac{1}{2}s)\Gamma(\frac{1}{2}-\nu)}{2\Gamma(1-\frac{1}{2}s)}$. (7.11.1)

Using this and (7.9.6), (2.1.23) gives

$$\int_{1}^{\infty} \frac{\sin ax}{(x^{2}-1)^{\nu+\frac{1}{2}}} dx = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} 2^{s-1} \pi^{\frac{1}{2}} \frac{\Gamma(\frac{1}{2}+\frac{1}{2}s)}{\Gamma(1-\frac{1}{2}s)} \frac{\Gamma(\nu+\frac{1}{2}s)\Gamma(\frac{1}{2}-\nu)}{2\Gamma(\frac{1}{2}+\frac{1}{2}s)} a^{-s} ds$$

$$= \frac{\pi^{\frac{1}{2}} \Gamma(\frac{1}{2}-\nu)}{4\pi i} \int_{k-i\infty}^{k+i\infty} 2^{s-1} \frac{\Gamma(\nu+\frac{1}{2}s)}{\Gamma(1-\frac{1}{2}s)} a^{-s} ds$$

$$= 2^{-\nu-1} \pi^{\frac{1}{2}} \Gamma(\frac{1}{2}-\nu) a^{\nu} J_{\nu}(a) \quad (-\frac{1}{2} < \nu < \frac{1}{2}) \quad (7.11.2)^{\frac{1}{2}}$$

by (7.9.3). The formula is the sine-reciprocal of (7.4.3). Similarly, using (7.9.5) and (7.9.7),

$$\int_{1}^{\infty} \frac{\cos ax}{(x^{2}-1)^{\nu+\frac{1}{2}}} dx = -2^{-\nu-1}\pi^{\frac{1}{2}}\Gamma(\frac{1}{2}-\nu)a^{\nu}Y_{\nu}(a) \quad (-\frac{1}{2} < \nu < \frac{1}{2}).$$
(7.11.3)‡

These processes come under Theorem 42 if $\frac{1}{4} < \nu < \frac{1}{2}$; for

$$\mathfrak{F}(s) = 2^{s-1}\pi^{\frac{1}{2}} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}s)}{\Gamma(1 - \frac{1}{2}s)} a^{-s} = O(|t|^{k-\frac{1}{2}}),$$

which is L if $k < -\frac{1}{2}$; and $x^{-k}(x^2-1)^{-\nu-\frac{1}{2}}$ is L if $-\frac{1}{2}k < \nu < \frac{1}{2}$. The result can be extended to the full range $-\frac{1}{2} < \nu < \frac{1}{2}$ by analytic continuation, or we can use Theorem 43. For $0 < \sigma < 1$

$$\int_{a}^{1} \sin x \, x^{s-1} \, dx = O(1)$$
and
$$\int_{1}^{b} \sin x \, x^{s-1} \, dx = \left[-\cos x \, x^{s-1} \right]_{1}^{b} + (s-1) \int_{1}^{b} \cos x \, x^{s-2} \, dx$$

$$= O(1) + O(|t|) = O(|t|).$$
† Watson, 6.13 (3).
‡ Ibid., 6.13 (4).

Also, for $\sigma = 1 - k$,

$$\int_{1}^{b} \frac{x^{s-1}}{(x^{2}-1)^{\nu+\frac{1}{2}}} = O(1)$$

if $-\frac{1}{2}k < \nu < \frac{1}{2}$. The other conditions are plainly fulfilled if $-\frac{1}{2} < \nu < \frac{1}{2}$. By taking k arbitrarily near to 1 the result follows.

Again, from the Mellin pairs

$$x^{\nu+1}J_{\nu}(ax), \qquad 2^{s+\nu}a^{-s-\nu-1}\frac{\Gamma(\frac{1}{2}s+\nu+\frac{1}{2})}{\Gamma(\frac{1}{2}-\frac{1}{2}s)},$$
 (7.11.4)

$$x^{\nu+1}J_{\nu}(ax), \qquad 2^{s+\nu}a^{-s-\nu-1}\frac{\Gamma(\frac{1}{2}s+\nu+\frac{1}{2})}{\Gamma(\frac{1}{2}-\frac{1}{2}s)}, \qquad (7.11.4)$$

$$(x^{2}+1)^{-\mu-1}, \qquad \frac{\Gamma(\frac{1}{2}s)\Gamma(\mu+1-\frac{1}{2}s)}{2\Gamma(\mu+1)} \qquad (7.11.5)$$

(transformations of (7.9.3) and (7.7.9)), we obtain

$$\begin{split} &\int\limits_{0}^{\infty} \frac{x^{\nu+1} J_{\nu}(ax)}{(x^{2}+1)^{\mu+1}} \, dx \\ &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} 2^{s+\nu} a^{-s-\nu-1} \frac{\Gamma(\frac{1}{2}s+\nu+\frac{1}{2})}{\Gamma(\frac{1}{2}-\frac{1}{2}s)} \frac{\Gamma(\frac{1}{2}-\frac{1}{2}s)\Gamma(\mu+\frac{1}{2}+\frac{1}{2}s)}{2\Gamma(\mu+1)} \, ds \\ &= \frac{1}{4\pi i \Gamma(\mu+1)} \int\limits_{k+2\nu+1-i\infty}^{k+2\nu+1+i\infty} 2^{s'-\nu-1} a^{-s'+\nu} \Gamma(\frac{1}{2}s')\Gamma(\frac{1}{2}s'+\mu-\nu) \, ds' \\ &= \frac{a^{\mu} K_{\mu-\nu}(a)}{2^{\mu} \Gamma(\mu+1)} \end{split} \tag{7.11.6}$$

by (7.9.12). Here the $\Re(s)$ of (7.11.4) is $L(k-i\infty, k+i\infty)$ if $-2\nu-1 < k < -\nu-1$; and $x^{-k}(x^2+1)^{-\mu-1}$ is L if $-2\mu-1 < k < 1$. These conditions are consistent if $0 < \nu < 2\mu$, and the result then follows from Theorem 42. The formula is actually valid if $-1 < \nu < 2\mu + \frac{3}{2}$. It can be extended to the full range either by analytic continuation or by Theorem 43.

The following examples can be obtained in a similar way, and present no particular difficulty: ‡

$$\int_{1}^{\infty} e^{-ax} (x^{2}-1)^{\nu-1} dx = \frac{2^{\nu} \Gamma(\nu+\frac{1}{2}) K_{\nu}(a)}{\Gamma(\frac{1}{2}) a^{\nu}}, \qquad (7.11.7)$$

[†] Ibid., 13.6 (2).

[‡] The corresponding formulae in Watson are 6.15 (4), 13.2 (5), 13.2 (8), 13.3 (4), 13.3 (5), 13.45 (2), 13.6 (3), 13.6 (5).

$$\int_{0}^{\infty} e^{-ax} J_{\nu}(x) x^{\nu} dx = \frac{2^{\nu} \Gamma(\nu + \frac{1}{2})}{\sqrt{\pi (a^{2} + 1)^{\nu + \frac{1}{2}}}},$$
(7.11.8)

$$(=7.4.8)$$

$$\int_{0}^{\infty} e^{-ax} J_{\nu}(x) \, dx = \frac{\{\sqrt{(1+a^2)-a}\}^{\nu}}{\sqrt{(1+a^2)}},\tag{7.11.9}$$

$$\int_{0}^{\infty} J_{\nu}(ax)e^{-bx^{2}}x^{\nu+1} dx = \frac{a^{\nu}}{(2b)^{\nu+1}}e^{-\frac{1}{4}a^{2}/b}, \tag{7.11.10}$$

$$\int_{0}^{\infty} J_{2\nu}(ax)e^{-bx^{2}} dx = \frac{\pi^{\frac{1}{2}}}{2b^{\frac{1}{2}}}e^{-a^{2}/8b}I_{\nu}\left(\frac{a^{2}}{8b}\right), \tag{7.11.11}$$

$$\int_{0}^{\infty} K_{\mu}(at) J_{\nu}(bt) t^{\mu+\nu+1} dt = \frac{(2a)^{\mu} (2b)^{\nu} \Gamma(\mu+\nu+1)}{(a^{2}+b^{2})^{\mu+\nu+1}}, \quad (7.11.12)$$

$$\int_{0}^{\infty} \frac{J_{\nu}(ax)}{\sqrt{(x^{2}+1)}} dx = I_{\frac{1}{2}\nu}(\frac{1}{2}a)K_{\frac{1}{2}\nu}(\frac{1}{2}a), \tag{7.11.13}$$

$$\int_{0}^{\infty} \frac{x^{\nu+1} J_{\nu}(ax)}{(x^{4}+4)^{\nu+\frac{1}{4}}} dx = \frac{(\frac{1}{2}a)^{\nu} \pi^{\frac{1}{4}}}{2^{2\nu} \Gamma(\nu+\frac{1}{2})} J_{\nu}(a) K_{\nu}(a). \quad (7.11.14)$$

A more delicate example is

$$\begin{split} \int\limits_{0}^{\infty} J_{\nu}(ax) J_{\nu+1}(x) \ dx &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} \frac{2^{s-1} \Gamma(\frac{1}{2}s+\frac{1}{2}\nu)}{\Gamma(\frac{1}{2}\nu-\frac{1}{2}s+1)} \frac{2^{-s} \Gamma(1+\frac{1}{2}\nu-\frac{1}{2}s)}{\Gamma(\frac{1}{2}\nu+\frac{1}{2}s+1)} a^{-s} \ ds \\ &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} \frac{a^{-s}}{s+\nu} \ ds \quad (k>-\nu) \\ &= a^{\nu} \ (0 < a < 1), \quad \frac{1}{2} \ (a = 1), \quad 0 \ (a > 1). \end{split}$$

If $a \neq 1$ this may be justified by Theorem 43, with

$$\chi(\xi) = \int_0^\infty J_{\nu}(ax) J_{\nu+1}(\xi x) \ dx.$$

The conditions may be verified as in the proof of (7.11.1). If a = 1, $\chi(\xi)$ is discontinuous at $\xi = 1$, and the method fails to evaluate $\chi(1)$. We can fill in this case by proving directly that $\chi(1) = \frac{1}{2} \{\chi(1+0) + \chi(1-0)\}$; or we can apply the method in the opposite direction, which gives the whole result, but with more

tedious details; or we can write

$$\int_{0}^{\infty} J_{\nu}(x)J_{\nu+1}(x) dx = \int_{0}^{\infty} \frac{d}{dx} \{x^{\nu+1}J_{\nu+1}(x)\} \cdot x^{\nu+1}J_{\nu+1}(x) \frac{dx}{x^{2\nu+2}}$$

$$= -\int_{0}^{\infty} x^{\nu+1}J_{\nu+1}(x) \left\{ \frac{x^{\nu+1}J_{\nu}(x)}{x^{2\nu+2}} - (2\nu+2) \frac{x^{\nu+1}J_{\nu+1}(x)}{x^{2\nu+3}} \right\} dx,$$
so that
$$\int_{0}^{\infty} J_{\nu}(x)J_{\nu+1}(x) dx = (\nu+1) \int_{0}^{\infty} \frac{J_{\nu+1}^{2}(x)}{x} dx.$$

The last integral is $1/(2\nu+2)$, by taking $\nu=\mu$, $s=2\mu$, in (7.10.1).

As an example on (7.8.6), we have, by (7.9.12) and (7.10.8),

$$\int_{0}^{\infty} x^{-\frac{1}{2}} K_{\mu}(2x) I_{\nu}(x) K_{\nu}(x) dx
= \frac{1}{8\sqrt{\pi}} \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(\frac{1}{2}s - \frac{1}{2}\mu - \frac{1}{4}) \Gamma(\frac{1}{2}s + \frac{1}{2}\mu - \frac{1}{4}) \times
\times \frac{\Gamma(\frac{1}{2} + \nu - \frac{1}{2}s) \Gamma(\frac{1}{2} - \frac{1}{2}s) \Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{2} + \nu + \frac{1}{2}s)} ds
= \frac{1}{4\sqrt{\pi}} \frac{1}{2\pi i} \int_{\frac{1}{4}(k-1) - i\infty}^{\frac{1}{4}(k-1) + i\infty} \Gamma(s' - \frac{1}{2}\mu + \frac{1}{4}) \Gamma(s' + \frac{1}{2}\mu + \frac{1}{4}) \times
\times \frac{\Gamma(\nu - s') \Gamma(\frac{1}{2} + s') \Gamma(-s')}{\Gamma(1 + \nu + s')} ds'
= \frac{\Gamma(\frac{1}{4} - \frac{1}{2}\mu) \Gamma(\frac{1}{4} + \frac{1}{2}\mu) \Gamma(\nu + \frac{1}{4} - \frac{1}{2}\mu) \Gamma(\nu + \frac{1}{4} + \frac{1}{2}\mu)}{4\Gamma(\frac{3}{4} + \nu + \frac{1}{2}\mu) \Gamma(\frac{3}{4} + \nu - \frac{1}{2}\mu)}.$$
(7.11.16)

From (7.7.11), and (7.11.4) with $\nu = 0$, we obtain

$$\begin{split} \int\limits_{0}^{\infty} \frac{1}{(1+x^2)^n} P_{n-1} \Big(\frac{1-x^2}{1+x^2} \Big) x J_0(ax) \; dx \\ &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} 2^s \frac{\Gamma(\frac{1}{2}s+\frac{1}{2}) \; \Gamma(\frac{1}{2}-\frac{1}{2}s) \{\Gamma(n-\frac{1}{2}+\frac{1}{2}s)\}^2}{2\Gamma(\frac{1}{2}+\frac{1}{2}s) \{\Gamma(n)\}^2} a^{-s-1} \; ds \\ &= \frac{1}{2\pi i \{\Gamma(n)\}^2} \int\limits_{k-i\infty}^{k+i\infty} 2^{s-1} \{\Gamma(n-\frac{1}{2}+\frac{1}{2}s)\}^2 a^{-s-1} \; ds \end{split}$$

$$= \frac{1}{2\pi i \{\Gamma(n)\}^2} \int_{k+2n-1-i\infty}^{k+2n-1+i\infty} 2^{s'-2n} \{\Gamma(\frac{1}{2}s')\}^2 a^{-s'-2+2n} ds'$$

$$= \frac{(\frac{1}{2}a)^{2n-2}}{\{\Gamma(n)\}^2} K_0(a). \tag{7.11.17}$$

7.12. Some non-absolutely convergent integrals. We have, by (7.9.5) and (7.9.6), the Mellin transforms

$$\cos x^{\alpha}, \qquad \frac{1}{\alpha} \Gamma\left(\frac{s}{\alpha}\right) \cos \frac{s\pi}{2\alpha}$$
 (7.12.1)

$$\sin x^{\alpha}, \qquad \frac{1}{\alpha} \Gamma\left(\frac{s}{\alpha}\right) \sin\frac{s\pi}{2\alpha}.$$
 (7.12.2)

From (7.12.1), with $\alpha = 1$ and $\alpha = 2$, we obtain

$$\int_{0}^{\infty} \cos x^{2} \cos ax \, dx = \frac{1}{4\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(s) \cos \frac{1}{2} s \pi \, \Gamma(\frac{1}{2} - \frac{1}{2} s) \cos \frac{1}{4} \pi (1-s) a^{-s} \, ds$$

$$= \frac{1}{4\pi i} \int_{k-i\infty}^{k+i\infty} 2^{s-1} \pi^{\frac{1}{4}} \Gamma(\frac{1}{2} s) \cos \frac{1}{4} \pi (1-s) a^{-s} \, ds$$

$$= \frac{1}{2} \pi^{\frac{1}{4}} \cos \frac{1}{4} (\pi - a^{2}), \qquad (7.12.3)$$

by (7.12.1) and (7.12.2) with $\alpha = 2$. Similarly,

$$\int_{0}^{\infty} \sin x^{2} \cos ax \, dx = \frac{1}{2} \pi^{\frac{1}{4}} \sin \frac{1}{4} (\pi - a^{2}). \tag{7.12.4}$$

The results are equivalent to (7.1.8) and (7.1.9). The process is justified by Theorem 39. As in § 7.11 we have

$$\int_{\lambda}^{\mu} x^{s-1} \cos x \, dx = O(|t|)$$

for all λ and μ . Also

$$\int\limits_{\lambda}^{\mu}\cos(ax-x^2)\;dx=\int\limits_{\lambda}^{\frac{1}{4}a-1}+\int\limits_{\frac{1}{4}a-1}^{\frac{1}{4}a+1}+\int\limits_{\frac{1}{4}a+1}^{\mu}=J_1+J_2+J_3$$

(with obvious modifications if $\lambda > \frac{1}{2}a-1$ or $\mu < \frac{1}{2}a+1$). Plainly $J_2 = O(1)$; and

$$J_3 = \int_{x=\frac{1}{2}a+1}^{\mu} \frac{d\sin(ax-x^2)}{a-2x} = -\frac{1}{2} \int_{x=\frac{1}{2}a+1}^{\xi} d\sin(ax-x^2) = O(1)$$

by the second mean-value theorem. Similarly, $J_1 = O(1)$. The conditions of Theorem 39 are thus satisfied.

Next, combine the cases $\alpha = 1$ and $\alpha = 3$ of (7.12.1). We obtain

$$\int_{0}^{+\infty} \cos x^{3} \cos ax \ dx = \frac{1}{6\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(s) \cos \frac{1}{2} s \pi \Gamma(\frac{1}{3} - \frac{1}{3} s) \cos \frac{1}{6} \pi (1-s) a^{-s} \ ds.$$

Now'

$$\Gamma(s)\Gamma(\frac{1}{3} - \frac{1}{3}s) = \frac{3^{s-\frac{1}{2}}}{2\pi} \Gamma(\frac{1}{3}s)\Gamma(\frac{1}{3}s + \frac{1}{3})\Gamma(\frac{1}{3}s + \frac{2}{3})\Gamma(\frac{1}{3} - \frac{1}{3}s)$$

$$= \frac{3^{s-\frac{1}{2}}}{2} \Gamma(\frac{1}{3}s)\Gamma(\frac{1}{3}s + \frac{1}{3})/\sin\frac{1}{3}\pi(1-s).$$

We thus obtain

$$\frac{1}{24\pi i} \int_{s-i\infty}^{s+i\infty} 3^{s-\frac{1}{2}} \{1 + 2\cos\frac{1}{3}\pi(1-s)\} \Gamma(\frac{1}{3}s) \Gamma(\frac{1}{3}s + \frac{1}{3})a^{-s} ds.$$

In the first part put $s = \frac{3}{2}s'$. We obtain

$$\begin{split} &\frac{1}{16\pi i}\int\limits_{\frac{1}{4}k-i\infty}^{\frac{1}{4}k+i\infty} 3^{\frac{1}{8}s'-\frac{1}{4}}\Gamma(\frac{1}{2}s')\Gamma(\frac{1}{2}s'+\frac{1}{3})a^{-\frac{1}{8}s'}\,ds'\\ &=\frac{1}{4\pi i 2^{\frac{1}{4}}3^{\frac{1}{4}}}\int\limits_{\frac{1}{4}k-i\infty}^{\frac{1}{4}k+i\infty} 2^{s'+\frac{1}{2}-2}\Gamma(\frac{1}{2}s')\Gamma(\frac{1}{2}s'+\frac{1}{3})\binom{2a^{\frac{3}{4}}}{3\sqrt{3}}^{-s'}\,ds'=\frac{a^{\frac{1}{4}}}{6}K_{\frac{1}{4}}\{2(\frac{1}{3}a)^{\frac{3}{4}}\}, \end{split}$$

by (7.9.12). In the second part put $s = \frac{3}{2}s' - \frac{1}{2}$. We obtain

$$\begin{split} \frac{1}{8\pi i} \int\limits_{\frac{1}{2}k+\frac{1}{2}-i\infty}^{\frac{1}{2}k+\frac{1}{2}+i\infty} & 3^{\frac{1}{2}s'-1}\sin\tfrac{1}{2}\pi s' \Gamma(\tfrac{1}{2}s'-\tfrac{1}{6})\Gamma(\tfrac{1}{2}s'+\tfrac{1}{6})a^{-\frac{2}{2}s'+\frac{1}{2}}\,ds' \\ &= \frac{\sqrt{a}}{24\pi i} \int \, 2^{s'}\sin\tfrac{1}{2}\pi s' \Gamma(\tfrac{1}{2}s'-\tfrac{1}{6})\Gamma(\tfrac{1}{2}s'+\tfrac{1}{6})\{2(\tfrac{1}{3}a)^{\frac{1}{2}}\}^{-s'}\,ds' \\ &= \frac{\pi\sqrt{a}}{6\sqrt{3}} \big[J_{\frac{1}{2}}\{2(\tfrac{1}{3}a)^{\frac{3}{2}}\} + J_{-\frac{1}{2}}\{2(\tfrac{1}{3}a)^{\frac{3}{2}}\}\big], \end{split}$$

by (7.9.9). Hence†

$$\int_{0}^{\infty} \cos x^{3} \cos ax \, dx = \frac{\pi \sqrt{a}}{6\sqrt{3}} \left[J_{i} \left\{ 2 \left(\frac{1}{3} a \right)^{\frac{3}{2}} \right\} + J_{-i} \left\{ 2 \left(\frac{1}{3} a \right)^{\frac{3}{2}} \right\} \right] + \frac{\sqrt{a}}{6} K_{i} \left\{ 2 \left(\frac{1}{3} a \right)^{\frac{3}{2}} \right\}.$$

$$(7.12.5)$$

Similarly,

$$\int_{0}^{+\infty} \sin x^{3} \sin ax \ dx = \frac{\pi \sqrt{a}}{6\sqrt{3}} \left[J_{\frac{1}{2}} \left\{ 2(\frac{1}{3}a)^{\frac{3}{2}} \right\} + J_{-\frac{1}{2}} \left\{ 2(\frac{1}{3}a)^{\frac{3}{2}} \right\} \right] - \frac{\sqrt{a}}{6} K_{\frac{1}{2}} \left\{ 2(\frac{1}{3}a)^{\frac{3}{2}} \right\}.$$
(7.12.6)

† Watson, § 6.4.

The process is justified by Theorem 39, as in the previous case. We have

$$\int_{\lambda}^{\mu} x^{s-1} \cos x^{\alpha} dx = \frac{1}{\alpha} \int_{\lambda'}^{\mu'} \xi^{s/\alpha - 1} \cos \xi d\xi = O(|t|)$$

for $0 < \sigma/\alpha < 1$, as before. Also the integrals

$$\int_{0}^{+\infty} \cos(ax \pm x^3) \ dx$$

converge uniformly in any finite a-interval.

Again,

$$x^{\nu-1}\cos\frac{a^2}{x} = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(s)\cos\frac{1}{2}s\pi x^{\nu-1} \left(\frac{a^2}{x}\right)^{-s} ds$$

$$= \frac{1}{2\pi i} \int_{1-\nu-k-i\infty}^{1-\nu-k+i\infty} \Gamma(1-\nu-s)\cos\frac{1}{2}\pi (1-\nu-s) a^{-2+2\nu+2s}x^{-s} ds.$$

Using this Mellin transformation and that of $\cos x$, we obtain

$$\int_{0}^{\infty} \cos x \cos \frac{a^{2}}{x} x^{\nu-1} dx = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(s) \cos \frac{1}{2} s \pi \Gamma(s-\nu) \cos \frac{1}{2} \pi(s-\nu) a^{2\nu-2s} ds$$

$$= \frac{1}{8\pi i} \int_{2k-2\nu-i\infty}^{2k-2\nu+i\infty} \Gamma(\nu+\frac{1}{2}s') \Gamma(\frac{1}{2}s') \{\cos \frac{1}{2} \pi \nu + \cos \frac{1}{2} \pi(s'+\nu)\} a^{-s'} ds'$$

$$= \cos \frac{1}{2} \pi \nu a^{\nu} K_{\nu}(2a) + \frac{\pi a^{\nu}}{4 \sin \frac{1}{4} \pi \nu} \{J_{-\nu}(2a) - J_{\nu}(2a)\}, \qquad (7.12.7)$$

by (7.9.12) and (7.9.10). Similarly,

$$\int_{0}^{\infty} \sin x \sin \frac{a^{2}}{x} x^{\nu-1} dx = \cos \frac{1}{2} \pi \nu a^{\nu} K_{\nu}(2a) - \frac{\pi a^{\nu}}{4 \sin \frac{1}{2} \pi \nu} \{ J_{-\nu}(2a) - J_{\nu}(2a) \}.$$
(7.12.8)

The process may be justified as in the previous cases.

7.13. Laplace transforms. Simple functions f(x), $\phi(s)$ connected by (1.4.1), (1.4.2) are

$$e^{-x}$$
, $\frac{1}{s+1}$, $(7.13.1)$

$$\cos x, \frac{s}{s^2+1},$$
 (7.13.2)

$$\sin x$$
, $\frac{1}{s^2+1}$, (7.13.3)

$$x^{\alpha-1}$$
, $\Gamma(\alpha)s^{-\alpha}$, $(7.13.4)$

$$\frac{1}{\sqrt{x}}e^{-1/x}, \qquad \sqrt{\left(\frac{\pi}{s}\right)}e^{-2\sqrt{s}}. \tag{7.13.5}$$

The pair

$$x^{\nu}J_{\nu}(x), \qquad \frac{2^{\nu}\Gamma(\nu+\frac{1}{2})}{\sqrt{\pi}(1+s^2)^{\nu+\frac{1}{2}}}$$
 (7.13.6)

results from (7.4.8); the pair

$$J_{\nu}(x), \qquad \frac{\{\sqrt{(1+s^2)-s}\}^{\nu}}{\sqrt{(1+s^2)}}$$
 (7.13.7)

from (7.11.9); the pair

$$J_{\nu}(x)/x$$
, $\{\sqrt{(1+s^2)-s}\}^{\nu}/\nu$ (7.13.8)

by integrating (7.11.9) with respect to a; the pair

$$x^{\frac{1}{4}\nu}J_{\nu}(\sqrt{x}), \qquad 2^{-\nu}s^{-\nu-1}e^{-1/(4s)}$$
 (7.13.9)

comes from (7.11.10).

Writing

$$C(x) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{x} \frac{\cos u}{\sqrt{u}} du, \qquad S(x) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{x} \frac{\sin u}{\sqrt{u}} du,$$
(7.13.10)

we have

$$\int_{0}^{\infty} e^{-xs} C(x) dx = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \frac{\cos u}{\sqrt{u}} du \int_{u}^{\infty} e^{-xs} dx$$

$$= \frac{1}{\sqrt{(2\pi)}s} \int_{0}^{\infty} \frac{\cos u}{\sqrt{u}} e^{-us} du = \frac{1}{2s} \left(\frac{1}{\sqrt{(1+s^2)}} + \frac{s}{1+s^2} \right)^{\frac{1}{2}}, \quad (7.13.11)$$

e.g. by (7.13.7). Similarly, we have the pair

$$S(x), \qquad \frac{1}{2s} \left(\frac{1}{\sqrt{(1+s^2)}} - \frac{s}{1+s^2} \right)^{\frac{1}{2}}.$$
 (7.13.12)

Defining

$$\vartheta(x) = 1 + 2\sum_{n=1}^{\infty} e^{-n^{i}\pi x}, \qquad (7.13.13)$$

we have

$$\int_{0}^{\infty} e^{-xs} \vartheta(x) \, dx = \int_{0}^{\infty} e^{-sx} \, dx + 2 \sum_{n=1}^{\infty} \int_{0}^{\infty} e^{-(s+n^{2}\pi)x} \, dx$$

$$= \frac{1}{s} + 2 \sum_{n=1}^{\infty} \frac{1}{s+n^{2}\pi} = \frac{1}{\sqrt{s \tanh \sqrt{s}}}.$$
 (7.13.14)

In each case the real part of s has to be greater than some lower limit—in fact 0 in all the above examples.

7.14. The formula (2.1.20) gives a number of interesting examples. The simplest is obtained by taking f(x), $\phi(s)$, as in (7.13.4), and g, ψ , with β for α . We obtain the familiar result

$$\int_{0}^{x} y^{\alpha-1} (x-y)^{\beta-1} dy = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(\alpha) \Gamma(\beta) \frac{e^{xs}}{s^{\alpha+\beta}} ds$$

$$= \frac{\Gamma(\alpha) \Gamma(\beta)}{\Gamma(\alpha+\beta)} x^{\alpha+\beta-1}. \tag{7.14.1}$$

The formula (2.1.20) is equivalent to Parseval's formula for the Fourier transforms

$$f(x)e^{-kx}$$
 $(x>0)$, 0 $(x<0)$, $(2\pi)^{-\frac{1}{2}}\phi(k+it)$,

and similarly with g and ψ . Here the e^{-kx} makes problems of f and g at infinity trivial. In the above case the L^2 theory applies if $\alpha > \frac{1}{2}$, $\beta > \frac{1}{2}$. The result holds for $\alpha > 0$, $\beta > 0$. The extension may be made e.g. by Theorem 38.

The following examples are easily justified in a similar way.

Take f(x) and $\phi(s)$ as in (7.13.6), but with parameter μ , and g, ψ , with ν . Then

$$\int_{0}^{x} y^{\mu} J_{\mu}(y) (x-y)^{\nu} J_{\nu}(x-y) dy$$

$$= \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{2^{\mu+\nu} \Gamma(\mu+\frac{1}{2}) \Gamma(\nu+\frac{1}{2})}{\pi} \frac{e^{sx}}{(1+s^{2})^{\mu+\nu+1}} ds$$

$$= \frac{\Gamma(\mu+\frac{1}{2}) \Gamma(\nu+\frac{1}{2})}{\sqrt{(2\pi)} \Gamma(\mu+\nu+1)} x^{\mu+\nu+\frac{1}{2}} J_{\mu+\nu+\frac{1}{2}}(x). \tag{7.14.2}$$

The particular case $\mu = \frac{1}{2}$, $\nu = 0$ is

$$\int_{0}^{x} \sin y \, J_0(x-y) \, dy = x J_1(x). \tag{7.14.3}$$

A number of similar formulae derivable from this are given by Watson, § 12.21.

The particular case $\mu = 0$, $\nu = 0$ is

$$\int_{0}^{x} J_{0}(y)J_{0}(x-y) dy = \sin x. \qquad (7.14.4)$$

† Hardy (10), Watson, §§ 12.2-12.22.

Similarly, (7.13.8) gives

$$\int_{0}^{x} \frac{J_{\mu}(y)J_{\nu}(x-y)}{y(x-y)} dy = \frac{1}{2\pi i \mu \nu} \int_{k-i\infty}^{k+i\infty} \{\sqrt{(1+s^{2})} - s\}^{\mu+\nu} e^{xs} ds$$

$$= \frac{\mu+\nu}{\mu\nu} \frac{J_{\mu+\nu}(x)}{x}.$$
(7.14.5)

Taking f, ϕ , as in (7.13.8), but with μ for ν , and g, ψ , as in (7.13.7), we obtain

$$\int_{0}^{x} J_{\mu}(y) J_{\nu}(x-y) \frac{dy}{y} = \frac{1}{2\pi i \mu} \int_{k-i\infty}^{k+i\infty} \frac{\{\sqrt{(1+s^{2})}-s\}^{\mu+\nu}}{\sqrt{(1+s^{2})}} e^{xs} ds$$

$$= \frac{J_{\mu+\nu}(x)}{\mu}. \tag{7.14.6}$$

Taking g, ψ , as in (7.13.7), and f, ϕ with μ for ν , we obtain

$$\int_{0}^{x} J_{\mu}(y)J_{\nu}(x-y) dy = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\{\sqrt{(1+s^{2})-s}\}^{\mu+\nu}}{1+s^{2}} e^{xs} ds$$

$$= \frac{1}{\pi i} \int \left[\{\sqrt{(1+s^{2})-s}\}^{\mu+\nu+1} - \{\sqrt{(1+s^{2})-s}\}^{\mu+\nu+3} + \dots \right] \frac{e^{xs}}{\sqrt{(1+s^{2})}} ds$$

$$= 2\{J_{\mu+\nu+1}(x) - J_{\mu+\nu+3}(x) + \dots \}. \tag{7.14.7}$$

The integral is expressible in finite terms if $\mu+\nu$ is an integer; for example

$$\int_{0}^{x} J_{-\nu}(y) J_{\nu}(x-y) \, dy = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{e^{xs}}{1+s^2} \, ds = \sin x. \quad (7.14.8)$$

A slightly different type of formula is obtained by taking

$$f(x) = x^{\frac{1}{4}\mu}J_{\mu}(a\sqrt{x}), \qquad \phi(s) = \frac{a^{\mu}}{2^{\mu}s^{\mu}+1}\exp\left(-\frac{a^2}{4s}\right),$$

and g, ψ , with b, ν for a, μ . Then

$$\begin{split} \int\limits_{0}^{s} y^{\frac{1}{4}\mu} J_{\mu}(a\sqrt{y})(x-y)^{\frac{1}{4}\nu} J_{\nu} \{b\sqrt{(x-y)}\} \, dy \\ &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} \frac{a^{\mu}b^{\nu}}{2^{\mu+\nu} s^{\mu+\nu+2}} \exp\left(-\frac{a^{2}+b^{2}}{4s} + xs\right) ds \\ &= \frac{2a^{\mu}b^{\nu}}{(a^{2}+b^{2})^{\frac{1}{4}(\mu+\nu+1)}} J_{\mu+\nu+1} [\sqrt{(a^{2}+b^{2})x}]. \quad (7.14.9) \end{split}$$

† The result is equivalent to Watson 12.13 (1).

4362

Taking the same g, ψ , and the f, ϕ of (7.13.4), we obtain

$$\int_{0}^{x} y^{\alpha-1}(x-y)^{\frac{1}{2}\nu} J_{\nu}\{b\sqrt{(x-y)}\} dy = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\alpha)b^{\nu}}{2^{\nu}s^{\nu+\alpha+1}} \exp\left(-\frac{b^{2}}{4s} + xs\right) ds$$

$$= \frac{2^{\alpha}\Gamma(\alpha)}{b^{\alpha}} x^{\frac{1}{2}\nu} J_{\nu+\alpha}(b\sqrt{x}). \tag{7.14.10}$$

From (7.13.11) and (7.13.12) we obtain ‡

$$\int_{0}^{x} C(y)S(x-y) dy = \frac{1}{8\pi i} \int_{k-i\infty}^{k+i\infty} \frac{e^{xs}}{s^{2}(1+s^{2})} ds$$

$$= \frac{1}{4}(x-\sin x), \qquad (7.14.11)$$

$$\int_{0}^{x} \{C(y)C(x-y)-S(y)S(x-y)\} dy = \frac{1}{4\pi i} \int_{k-i\infty}^{k+i\infty} \frac{e^{xs}}{s(1+s^{2})} ds$$

$$= \frac{1}{4\pi i} \int_{k-i\infty}^{k+i\infty} \left(\frac{1}{s} - \frac{s}{1+s^{2}}\right) e^{xs} ds = \frac{1}{2}(1-\cos x), \quad (7.14.12)$$

and there are some similar formulae involving J_0 and J_1 .

The method also leads to an integral equation || satisfied by the function $\vartheta(x)$. From (7.13.14) we deduce

$$\int_{0}^{x} \vartheta(y)\vartheta(x-y) \, dy = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{e^{xs}}{s \tanh^{2} \sqrt{s}} \, ds.$$
$$\frac{d}{ds} \left(\frac{1}{\tanh \sqrt{s}}\right) = \frac{1}{2\sqrt{s}} - \frac{1}{2\sqrt{s} \tanh^{2} \sqrt{s}}.$$

Now

Hence the right-hand side is

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{e^{xs}}{s} ds - \frac{1}{\pi i} \int_{k-i\infty}^{k+i\infty} \left(\frac{d}{ds} \frac{1}{\tanh \sqrt{s}} \right) \frac{e^{xs}}{\sqrt{s}} ds$$

$$= 1 + \frac{1}{\pi i} \int_{k-i\infty}^{k+i\infty} \frac{1}{\tanh \sqrt{s}} \left(\frac{x}{\sqrt{s}} - \frac{1}{2s!} \right) e^{xs} ds$$

$$= 1 + 2x\vartheta(x) - \int_{0}^{x} \vartheta(u) du,$$

since
$$\int_{0}^{x} \vartheta(u) du = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{e^{xs}}{s^{\frac{1}{2}} \tanh \sqrt{s}} ds - \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{ds}{s^{\frac{1}{2}} \tanh \sqrt{s}},$$

and the last term is 0, as is seen by moving the line of integration away to the right. We have thus proved that

$$\int_{0}^{x} \vartheta(y)\vartheta(x-y) \ dy = 1 + 2x\vartheta(x) - \int_{0}^{x} \vartheta(u) \ du. \tag{7.14.13}$$

VIII

GENERAL TRANSFORMATIONS

8.1. Generalization of Fourier's formulae. In the foregoing chapters we have studied two formulae of the form

$$f(x) = \int_0^\infty k(xu) \, du \int_0^\infty k(uy) f(y) \, dy \qquad (8.1.1)$$

for an arbitrary function f(x); $k(x) = \sqrt{\binom{2}{\pi}} \cos x$ gives Fourier's cosine formula, and $k(x) = \sqrt{\binom{2}{\pi}} \sin x$ gives Fourier's sine formula. There

are, however, other formulae of the same form, the best known being Hankel's formula, in which

$$k(x) = x^{\frac{1}{2}} J_{\nu}(x). \tag{8.1.2}$$

There are also formulae of the form

$$f(x) = \int_0^\infty k(xu) \ du \int_0^\infty h(uy) f(y) \ dy \tag{8.1.3}$$

in which the two cosines of Fourier's formula are replaced by different functions. The simplest formula of this type is that in which

$$k(x) = x^{\frac{1}{2}}Y_{\nu}(x), \qquad h(x) = x^{\frac{1}{2}}\mathbf{H}_{\nu}(x).$$
 (8.1.4)

As usual, (8.1.1) may be written as a pair of reciprocal formulae

$$g(x) = \int_0^\infty f(y)k(xy) \, dy, \tag{8.1.5}$$

$$f(x) = \int_{0}^{\infty} g(y)k(xy) dy. \qquad (8.1.6)$$

A function k(x) giving rise to a formula of the form (8.1.1) will be called a *Fourier kernel*. The main object of this chapter is to give an account of such functions.

Suppose that we multiply (8.1.5) by x^{s-1} and integrate over $(0, \infty)$. We obtain formally

$$\int_{0}^{\infty} g(x)x^{s-1} dx = \int_{0}^{\infty} x^{s-1} dx \int_{0}^{\infty} f(y)k(xy) dy$$

$$= \int_{0}^{\infty} f(y) dy \int_{0}^{\infty} k(xy)x^{s-1} dx = \int_{0}^{\infty} f(y)y^{-s} dy \int_{0}^{\infty} k(u)u^{s-1} du,$$
† Hardy and Titchmarsh (8), Watson (2).

i.e., with the usual notation for Mellin transforms,

$$\mathfrak{G}(s) = \mathfrak{F}(1-s)\mathfrak{R}(s). \tag{8.1.7}$$

Similarly (8.1.6) gives

$$\mathfrak{F}(s) = \mathfrak{G}(1-s)\mathfrak{R}(s). \tag{8.1.8}$$

Changing s into 1-s in one of these equations, and multiplying, we deduce that $\Re(s)\Re(1-s) = 1.$ (8.1.9)

We should therefore expect that a Fourier kernel k(x) would be in some sense of the form

$$k(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re(s) x^{-s} ds,$$
 (8.1.10)

where $\Re(s)$ satisfies the functional equation (8.1.9).

8.2. The condition (8.1.9) may also be expected to be in some sense sufficient.

A characteristic property of a Fourier kernel k(x) is that, if

$$k_1(x) = \int_0^x k(u) \, du, \qquad (8.2.1)$$

then

$$\int_{0}^{\infty} k(xu) \frac{k_{1}(\xi u)}{u} du = \begin{cases} 1 & (0 < x < \xi), \\ 0 & (x > \xi). \end{cases}$$
 (8.2.2)

For if we put f(x) = 1 $(0 < x \le \xi)$, 0 $(x > \xi)$, in (8.1.1), we obtain (8.2.2); and conversely (8.2.2) leads to

$$\int_{0}^{x} f(y) dy = \int_{0}^{\infty} f(y) dy \int_{0}^{\infty} k(yu) \frac{k_{1}(xu)}{u} du$$
$$= \int_{0}^{\infty} \frac{k_{1}(xu)}{u} du \int_{0}^{\infty} k(uy) f(y) dy,$$

from which (8.1.1) follows by formal differentiation.

Now (8.1.10) gives formally

$$k_1(x) = \frac{1}{2\pi i} \int_{-\infty}^{c+i\infty} \Re(s) \frac{x^{1-s}}{1-s} ds.$$
 (8.2.3)

Hence, by a formal application of Parseval's formula for Mellin

transforms, in the form (2.1.20),

$$\int_{0}^{\infty} k(xu) \frac{k_{1}(\xi u)}{u} du = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re(s) \frac{\Re(1-s)}{s} x^{-s} \xi^{s} ds \quad (c > 0)$$

$$= \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \left(\frac{\xi}{x}\right)^{s} \frac{ds}{s} = 1 \quad (0 < x < \xi), \quad 0 \quad (x > \xi).$$

Hence (8.2.2), and so the Fourier formula, is formally a consequence of (8.1.9).

8.3. Similar analysis holds for the unsymmetrical formulae arising from (8.1.3). If we now write

$$f(x) = \int_0^\infty k(xu)g(u) \ du, \tag{8.3.1}$$

$$g(u) = \int_0^\infty h(uy)f(y) \, dy, \qquad (8.3.2)$$

the relations between Mellin transforms are

$$\mathfrak{F}(s) = \mathfrak{G}(1-s)\mathfrak{R}(s), \tag{8.3.3}$$

$$\mathfrak{G}(s) = \mathfrak{F}(1-s)\mathfrak{H}(s), \tag{8.3.4}$$

and, eliminating & and &,

$$\Re(s)\Re(1-s) = 1.$$
 (8.3.5)

We may regard (8.3.2) as the solution of the integral equation (8.3.1) for the unknown function g(u), the 'solving kernel' h(x) being given by c+ix c+ix

$$h(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \mathfrak{H}(s) x^{-s} ds = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \frac{x^{-s}}{\mathfrak{R}(1-s)} ds.$$

8.4. Examples. Before proceeding farther we shall give a number of examples.

(1) If
$$\Re(s) = 2^{s-\frac{1}{2}} \frac{\Gamma(\frac{1}{2}\nu + \frac{1}{2}s + \frac{1}{4})}{\Gamma(\frac{1}{2}\nu - \frac{1}{2}s + \frac{3}{4})} \quad (\nu > -1),$$
 then
$$k(x) = x^{\frac{1}{2}} J_{\nu}(x),$$

and the formula is that of Hankel; the cases $\nu = -\frac{1}{2}$ and $\nu = \frac{1}{2}$ are Fourier's cosine and sine formulae.

If
$$-2<
u<-1$$
, then $k(x)=x^{rac{1}{2}}\Big\{J_{
u}(x)-rac{(rac{1}{2}x)^{
u}}{\Gamma(
u+1)}\Big\},$

and generally, if $-m-1 < \nu < -m$, then

$$k(x) = x^{\frac{1}{2}} \left\{ J_{\nu}(x) - \sum_{n=0}^{m} \frac{(-1)^n (\frac{1}{2}x)^{\nu+2n}}{n! \Gamma(\nu+n+1)} \right\},\,$$

the sum being the first m+1 terms in the power series for $J_{\nu}(x)$.

(2) If
$$\dagger$$
 $k(x) = x^{\frac{1}{2}}Y_{\nu}(x)$,

then, by (7.9.7),

$$\Re(s) = -2^{s-\frac{1}{2}}\pi^{-1}\Gamma(\frac{1}{2}s+\frac{1}{2}\nu+\frac{1}{2})\Gamma(\frac{1}{2}s-\frac{1}{2}\nu+\frac{1}{2})\cos(\frac{1}{2}s-\frac{1}{2}\nu+\frac{1}{2})\pi.$$

This does not satisfy (8.1.9), so that k(x) is not a Fourier kernel. But (8.3.5) gives

$$\mathfrak{H}(s) = 2^{s-\frac{1}{2}} \frac{\Gamma(\frac{1}{2}s + \frac{1}{2}\nu + \frac{1}{4})}{\Gamma(\frac{1}{2}\nu - \frac{1}{2}s + \frac{3}{4})} \tan(\frac{1}{2}s + \frac{1}{2}\nu + \frac{1}{4})\pi.$$

It then follows from (7.9.13) that

$$h(x) = x^{\frac{1}{2}}\mathbf{H}_{\nu}(x).$$

(3) There are a number of very general transformations, due to Fox, \ddagger in which k(x) and h(x) are linear combinations of generalized hypergeometric functions. From our present point of view these originate as follows.

Suppose that $\alpha_1 > 0$, that ρ_1 and ρ_2 are any real numbers other than negative integers, and that

and let

$$egin{aligned} \phi &= lpha_1 -
ho_1 -
ho_2 + rac{1}{2}, \ \Re(s) &= 2^{s-1} rac{\Gamma(lpha_1 + rac{1}{2}\phi - rac{1}{2}s)\Gamma(rac{1}{2}s - rac{1}{2}\phi)}{\Gamma(
ho_1 + rac{1}{2}\phi - rac{1}{2}s)\Gamma(
ho_2 + rac{1}{2}\phi - rac{1}{2}s)}. \end{aligned}$$

We deduce from the calculus of residues that

$$k(x) = (\frac{1}{2}x)^{-\phi} \sum_{n=0}^{\infty} \frac{\Gamma(\alpha_1 + n)}{\Gamma(\rho_1 + n)\Gamma(\rho_2 + n)} \frac{(-\frac{1}{4}x^2)^n}{n!}$$
$$= (\frac{1}{2}x)^{-\phi} {}_{1}F_{2}(\alpha_1, \rho_1, \rho_2; -\frac{1}{4}x^2)$$

in the usual hypergeometric series notation. If we now calculate $\mathfrak{H}(s)$ from (8.3.5), and then h(x) by summation of residues, we find

$$h(x) = h_1(x) + h_2(x),$$

where

$$h_1(x) = \frac{\sin(\alpha_1 - \rho_1)\pi}{\sin(\rho_2 - \rho_1)\pi} (\frac{1}{2}x)^{2\rho_1 + \phi - 1} {}_1F_2(1 - \alpha_1 + \rho_1, 1 - \rho_2 + \rho_1, \rho_1; -\frac{1}{4}x^2),$$

and $h_2(x)$ is derived from $h_1(x)$ by interchange of ρ_1 and ρ_2 . The formulae thus obtained are those of Fox's Theorem 1, in the special

case p = 1. In the general case $\Re(s)$ is a more complicated product of gamma-functions of the same type.

The case $\alpha_1 = 1$, $\rho_1 = \frac{3}{2}$, $\rho_2 = \nu + \frac{3}{2}$ gives example (2) above. The case $\alpha_1 = 1$, $\rho_1 = a+1$, $\rho_2 = \nu + a+1$ gives a more general transformation found by Hardy† and discussed by Cooke.‡ The case $\alpha_1 = \nu + \frac{3}{2}$, $\rho_1 = \nu + 1$, $\rho_2 = 2\nu + 1$ gives

$$k(x) = \frac{1}{2} \dot{\sqrt{\pi}} \frac{d}{dx} \{ x J_{\nu}^{2}(\frac{1}{2}x) \}, \qquad h(x) = -\sqrt{\pi} J_{\nu}(\frac{1}{2}x) Y_{\nu}(\frac{1}{2}x),$$

a transformation due to Bateman. || The case $\alpha_1 = \nu + a + \frac{1}{2}$, $\rho_1 = \nu + a + 1$, $\rho_2 = 2\nu + a + 1$ gives a more general transformation due to Titchmarsh.†† Fuller details concerning these transformations will be found in § 5.2 of Fox's paper.

If we take

$$\Re(s) = 2^{s-rac{1}{2}} rac{\Gamma(a_1 + rac{1}{2}s)\Gamma(a_2 + rac{1}{2}s)\Gamma(a_3 - rac{1}{2}s)}{\Gamma(b_1 + rac{1}{2}s)\Gamma(b_2 - rac{1}{2}s)\Gamma(b_3 - rac{1}{2}s)},
onumber \ a_1 + a_2 + a_3 + rac{1}{2} = b_1 + b_2 + b_3,$$

where

we obtain examples of Fox's Theorem 2. For example, if

$$a_1 = \frac{1}{2}\mu + \frac{1}{2}\nu + \frac{1}{2}, \qquad a_2 = \frac{1}{2} - \frac{1}{2}\mu - \frac{1}{2}\nu, \qquad a_3 = \frac{1}{2},$$

$$b_1 = 1, \qquad b_2 = \frac{1}{2}\nu - \frac{1}{2}\mu + \frac{1}{2}, \qquad b_3 = \frac{1}{2}\mu - \frac{1}{2}\nu + \frac{1}{2},$$

k(x) and h(x) are each combinations of two hypergeometric functions, and can be reduced to the forms

$$k(x) = \frac{\sqrt{\pi}}{2\sqrt{2}\sin\frac{1}{2}(\mu+\nu)\pi}x\{J_{-\mu}(\frac{1}{2}x)J_{-\nu}(\frac{1}{2}x) - J_{\mu}(\frac{1}{2}x)J_{\nu}(\frac{1}{2}x)\},$$

$$h(x) = \frac{\sqrt{\pi}}{2\sqrt{2}\cos\frac{1}{2}(\mu-\nu)\pi}\frac{d}{dx}\{x[J_{\mu}(\frac{1}{2}x)J_{-\nu}(\frac{1}{2}x) + J_{-\mu}(\frac{1}{2}x)J_{\nu}(\frac{1}{2}x)]\}.$$

$$(4) \text{ If } k(x) = x^{\frac{1}{2}}\{Y_{\nu}(x) + \frac{2}{\pi}\cos a\pi K_{\nu}(x)\},$$

then, by (7.9.7) and (7.9.11), $\Re(s)$ is

$$\begin{array}{c} 2^{2s-1}\Gamma(\frac{1}{4}s+\frac{1}{4}\nu+\frac{1}{8})\Gamma(\frac{1}{4}s+\frac{1}{4}\nu+\frac{5}{8})\Gamma(\frac{1}{4}s-\frac{1}{4}\nu+\frac{1}{8})\Gamma(\frac{1}{4}s-\frac{1}{4}\nu+\frac{5}{8})}{\Gamma(\frac{1}{4}s-\frac{1}{4}\nu+\frac{1}{8}+\frac{1}{2}a)\Gamma(\frac{7}{8}+\frac{1}{4}\nu-\frac{1}{4}s-\frac{1}{2}a)\times\\ \times\Gamma(\frac{1}{4}s-\frac{1}{4}\nu+\frac{1}{8}-\frac{1}{2}a)\Gamma(\frac{7}{8}+\frac{1}{4}\nu-\frac{1}{4}s+\frac{1}{2}a)\end{array}$$

In this case again h(x) is the sum of two hypergeometric series. There are two interesting particular cases. If $\nu = 0$, $\alpha = 1$, then

$$\Re(s) = -2^{2s-1} \left(\frac{\Gamma(\frac{1}{4}s + \frac{1}{8})}{\Gamma(\frac{3}{8} - \frac{1}{4}s)} \right)^2$$

which satisfies (8.1.9), so that, by (7.9.7), (7.9.11),

$$k(x) = x^{\frac{1}{2}} \left\{ Y_0(x) - \frac{2}{\pi} K_0(x) \right\}$$

is a Fourier kernel. If $\nu = 2$, a = 0, then

$$\Re(s) = 2^{2s-1} \left(\frac{\Gamma(\frac{1}{4}s + \frac{9}{8})}{\Gamma(\frac{11}{8} - \frac{1}{4}s)} \right)^2 \frac{\frac{1}{4}s - \frac{3}{8}}{\frac{1}{4}s + \frac{1}{8}},$$

and $\Re(s)\Re(1-s) = -1$, so that

$$k(x) = -h(x) = x^{\frac{1}{2}} \Big\{ Y_2(x) + \frac{2}{\pi} K_2(x) \Big\}.$$

The formulae in this case are due to A. L. Dixon and Hardy.† Much more general formulae of a similar character have been obtained by Steen‡ and Kuttner.||

(5) If
$$\Re(s) = e^{-ai(s-\frac{1}{2})^s}$$
 $(a > 0)$, then $\Re(s) = e^{ai(s-\frac{1}{2})^s}$.

Taking $c = \frac{1}{2}$ in (8.1.10), we find that

$$k(x) = \frac{1}{2\pi\sqrt{x}} \int_{-\infty}^{\infty} e^{ait^*} \cos(t \log x) dt = \frac{e^{\frac{1}{4}\pi i}}{2\sqrt{(\pi ax)}} e^{-i(\log x)^{\frac{n}{4}a}},$$

and h(x) is the conjugate function. The Fourier formula thus obtained may be reduced by a change of variable to the exponential form of the ordinary Fourier formula. It is

$$f(x) = \frac{1}{4\pi a} \int_0^\infty e^{-i(\log xu)^{\bullet/4a}} \frac{du}{\sqrt{(xu)}} \int_0^\infty e^{i(\log uy)^{\bullet/4a}} \frac{f(y)}{\sqrt{(uy)}} dy;$$

and, if we put $a = \frac{1}{2}$, and

$$x = e^{\xi}, \quad u = e^{\zeta}, \quad y = e^{\eta}, \quad g(\xi) = e^{ii\xi^{\eta} + i\xi}f(e^{\xi}),$$

we obtain

$$g(\xi) = rac{1}{2\pi} \int\limits_{-\infty}^{\infty} \, e^{-i\xi\zeta} \, d\zeta \int\limits_{-\infty}^{\infty} \, e^{i\zeta\eta} g(\eta) \, d\eta.$$

This formula is not included in our standard form, since the limits are $-\infty$ and ∞ instead of 0 and ∞ .

- † Hardy (17); see also Hardy and Titchmarsh (8).
- ‡ Steen (1). || Kuttner (1).

(6) If
$$\Re(s) = e^{a(s-\frac{1}{2})^s}$$
 $(a > 0),$

then k(x) is a Fourier kernel. Taking $c = \frac{1}{2}$ in (8.1.10), we find

$$\begin{split} k(x) &= \frac{1}{\pi \sqrt{x}} \int_{0}^{+\infty} \cos(at^3 + t \log x) \, dt \\ &= \frac{1}{3\sqrt{(3ax)}} \left(\log \frac{1}{x} \right)^{\frac{1}{2}} \left[J_{\frac{1}{2}} \left(\frac{2}{3\sqrt{(3a)}} \left(\log \frac{1}{x} \right)^{\frac{3}{2}} \right) + J_{-\frac{1}{2}} \left(\frac{2}{3\sqrt{(3a)}} \left(\log \frac{1}{x} \right)^{\frac{3}{2}} \right) \right] \\ &= \frac{1}{3\pi \sqrt{(ax)}} (\log x)^{\frac{1}{2}} K_{\frac{1}{2}} \left(\frac{2}{3\sqrt{(3a)}} (\log x)^{\frac{3}{2}} \right) \quad (x > 1), \end{split}$$

by (7.12.5) and (7.12.6).

(7) If
$$\Re(s) = \exp\{ie^{-i(s-t)}\} = \exp(ie^t)$$
 $(s = \frac{1}{2} + it)$, then $\Re(s) = \exp(-ie^{-t})$.

We obtain

$$\begin{split} k(x) &= \frac{1}{2\pi} \int\limits_{-\infty}^{\infty} \exp(ie^t) x^{-\frac{1}{2} - it} \, dt = \frac{x^{-\frac{1}{2}}}{2\pi} \int\limits_{0}^{\infty} e^{iu} u^{-i\log x} \frac{du}{u} \\ &= \frac{x^{-\frac{1}{2}}}{2\pi} \int\limits_{0}^{\infty} e^{-v} (ve^{\frac{1}{2}i\pi})^{-i\log x} \frac{dv}{v} = \frac{x^{\frac{1}{2}\pi - \frac{1}{2}}}{2\pi} \Gamma(-i\log x), \end{split}$$

and h(x) is the conjugate.

(8) If
$$\Re(s) = 1,$$

then (8.1.9) is satisfied, but the integral (8.1.10) is not convergent. If, however, we regard (8.2.3), with 0 < c < 1, as the definition of $k_1(x)$, we have

$$k_1(x) = \frac{1}{2\pi i} \int_{-\infty}^{c+i\infty} \frac{x^{1-s}}{1-s} ds = \frac{1}{0} \quad (x > 1),$$
 $0 \quad (0 < x < 1).$

If we replace (8.1.5), (8.1.6) by

$$g(x) = rac{1}{x} \int_{0}^{\infty} f(y) \ dk_{1}(xy), \qquad f(x) = rac{1}{x} \int_{0}^{\infty} g(y) \ dk_{1}(xy),$$

then our formulae become

$$g(x) = \frac{1}{x} f\left(\frac{1}{x}\right), \quad f(x) = \frac{1}{x} g\left(\frac{1}{x}\right),$$

which are plainly consequences of one another.

† Paley and Wiener, Fourier Transforms, § 16.

(9) In all these examples $\Re(s)$ is analytic. Suppose, however, that $c = \frac{1}{2}$, and $\Re(\frac{1}{2} + it) = i \operatorname{sgn} t$.

Then \mathfrak{H} , defined by (8.3.5), is $-\mathfrak{R}$. The integral (8.1.10) is not convergent, but it is formally

$$\frac{i}{2\pi\sqrt{x}}\left(\int\limits_{0}^{\infty}x^{-it}\,dt-\int\limits_{-\infty}^{0}x^{-it}\,dt\right)=\frac{1}{\pi\sqrt{x}}\int\limits_{0}^{\infty}\sin(t\log x)\,dt=\frac{1}{\pi\sqrt{x}\log x},$$

the integral being summable (C, 1). Our formulae become

$$f(x) = \frac{1}{\pi} \int_{0}^{\infty} \frac{g(y) \, dy}{\sqrt{(xy) \log(xy)}}, \qquad g(x) = -\frac{1}{\pi} \int_{0}^{\infty} \frac{f(y) \, dy}{\sqrt{(xy) \log(xy)}}.$$

If we replace x and y by e^{ξ} and e^{η} , and interpret the integrals as principal values, we obtain formulae equivalent to those of the theory of Hilbert transforms.

(10) If
$$\Re(s) = \cot \frac{1}{2} s\pi,$$

then (8.1.9) is satisfied. The integral (8.1.10) is of the same type as in (9). A formal application of the theorem of residues gives

$$k(x) = \frac{2}{\pi} \frac{1}{1-x^2},$$

and we again obtain formulae of the Hilbert transform type.

(11) We obtain formulae of a somewhat different type by taking $\Re(\frac{1}{3}+it)=e^{it}.$

Then (8.1.9) is satisfied, and

$$k(x) = \frac{1}{2\pi\sqrt{x}} \int_{-\infty}^{\infty} e^{it} x^{-it} dt = \frac{1}{\pi\sqrt{x}} \int_{0}^{\infty} \cos\left(\frac{1}{t} - t\log x\right) dt.$$

The integral is summable (C, 1) if $x \neq 1$, and has the value

$$-\left(x\log\frac{1}{x}\right)^{-\frac{1}{2}}J_1\left\{2\left(\log\frac{1}{x}\right)^{\frac{1}{2}}\right\} \quad (0 < x < 1), \qquad 0 \quad (x > 1).$$

If x = 1, the integral for k(x) diverges to infinity, and $k_1(x)$ has a discontinuity, as in example (8). The formula which results is therefore

$$g(x) = \frac{1}{x} f\left(\frac{1}{x}\right) - \int_{0}^{1/x} \frac{J_{1}\left\{2(-\log \frac{xy}{xy})^{1}\right\}}{(-xy\log \frac{xy}{xy})^{1}} f(y) \ dy.$$

If we put

$$x = e^{-\xi}, \qquad y = e^{-\eta}, \qquad e^{-i\xi}f(e^{-\xi}) = \phi(\xi), \qquad e^{-i\xi}g(e^{-\xi}) = \psi(\xi),$$

we obtain

$$\psi(\xi) = \phi(-\xi) - \int_{-\xi}^{\infty} \frac{J_1\{2\sqrt{(\xi+\eta)}\}}{\sqrt{(\xi+\eta)}}\phi(\eta) d\eta.$$

The reciprocal formula is obtained by interchanging ϕ and ψ . The Fourier formula which results may be verified by using the integral

$$\int_{\lambda}^{\infty} \frac{J_1\{2\sqrt{(x+\lambda)}\}J_1\{2\sqrt{(x+\mu)}\}}{\sqrt{(x+\lambda)}\sqrt{(x+\mu)}} dx = \frac{J_1\{2\sqrt{(\mu-\lambda)}\}}{\sqrt{(\mu-\lambda)}} \quad (\lambda < \mu).$$

(12) The kernels which arise in the summation formulae obtained formally in § 2.9 are Fourier kernels. For example, in the argument of § 2.9 we obtain $2\cos 2\pi x$ and $4K_0(4\pi\sqrt{x}) - 2\pi Y_0(4\pi\sqrt{x})$ as the Mellin transforms of

 $\frac{\zeta(1-s)}{\zeta(s)}, \qquad \frac{\zeta^2(1-s)}{\zeta^2(s)}$

respectively. These functions of course satisfy (8.1.9).

Note also that, if k(x) is a Fourier kernel, so are $\sqrt{a} k(ax)$ and $\lambda x^{i(\lambda-1)} k(x^{\lambda})$.

8.5. L^2 -theory. In the theory of Fourier integrals we have proved theorems of two kinds, theorems on convergence in the ordinary sense, and theorems on mean-convergence. There are also theorems of both kinds for general transforms; but here the mean-convergence theory is both easier and more general than the other, and we begin with this.

In the first place, we need only assume the existence of the function $\Re(s)$ on the line $\sigma = \frac{1}{2}$. The equation (8.1.9) then takes the form

$$\Re(\frac{1}{2}+it)\Re(\frac{1}{2}-it) = 1. \tag{8.5.1}$$

We might simply write $\Re(\frac{1}{2}+it) = \phi(t)$, and $\phi(t)\phi(-t) = 1$; but we shall retain the previous notation to preserve the appearance of the formulae.

We should now have formally

$$k(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \Re(\frac{1}{2} + it) x^{-\frac{1}{2} - it} dt.$$
 (8.5.2)

There is no reason in general to suppose that this integral will exist † Watson, § 13.47 (10).

in any sense. However, the formula for $k_1(x)$ obtained by formal integration will exist in the sense that

$$k_1(x) = \frac{x}{2\pi} \lim_{T \to \infty} \int_{-T}^{T} \frac{\Re(\frac{1}{2} + it)}{\frac{1}{2} - it} x^{-1-it} dt.$$
 (8.5.3)

If, as in most of our formulae, $\Re(s)$ takes conjugate values for conjugate values of s, then (8.5.1) gives

$$|\Re(\frac{1}{2} + it)| = 1. \tag{8.5.4}$$

Hence $\Re(\frac{1}{2}+it)/(\frac{1}{2}-it)$ belongs to $L^2(-\infty,\infty)$, the integral in (8.5.3) exists in the mean-square sense, and $k_1(x)/x$ belongs to $L^2(0,\infty)$.

It follows that our theorems have to be stated in terms, not of k(x), but of $k_1(x)$. For example, (8.2.2) is no longer significant. However, the formula obtained by formal integration with respect to x is

$$\int_{0}^{\infty} k_{1}(xu)k_{1}(\xi u)\frac{du}{u^{2}} = \min(x,\xi). \tag{8.5.5}$$

This integral is absolutely convergent in the general case, and (8.5.5) by itself may be taken as the basis of a Fourier theory.

The theory takes different forms according to whether (8.5.5) appears explicitly or not. The results may be summed up in the following theorems.

THEOREM 129. Let $\Re(\frac{1}{2}+it)$ be any function of t satisfying (8.5.1) and (8.5.4), so that $\frac{\Re(\frac{1}{2}+it)}{\frac{1}{2}-it}$

belongs to $L^2(-\infty,\infty)$. Let $k_1(x)$ be defined by (8.5.3). Let f(x) be any function of $L^2(0,\infty)$. Then the formula

$$g(x) = \frac{d}{dx} \int_{0}^{\infty} k_1(xu) f(u) \frac{du}{u}$$
 (8.5.6)

defines almost everywhere a function g(x), also belonging to $L^2(0,\infty)$; the reciprocal formula

$$f(x) = \frac{d}{dx} \int_{0}^{\infty} k_1(xu)g(u) \frac{du}{u}$$
 (8.5.7)

also holds almost everywhere; and

$$\int_{0}^{\infty} \{f(x)\}^{2} dx = \int_{0}^{\infty} \{g(x)\}^{2} dx.$$
 (8.5.8)

THEOREM 130. If $\Re(\frac{1}{2}+it)$ satisfies the conditions of Theorem 129, then $k_1(x)/x$ belongs to $L^2(0,\infty)$, and (8.5.5) holds.

THEOREM 131. Let $k_1(x)$ be such that $k_1(x)/x$ belongs to $L^2(0,\infty)$, and let (8.5.5) hold for all values of x and ξ . Then the reciprocal formulae of Theorem 129 hold.

Theorem 129 is thus a consequence of Theorems 130 and 131. But it is possible to prove it directly.

The above theory is due to Watson. \dagger We shall call functions f(x) and g(x) connected by (8.5.6), (8.5.7) k-transforms; and (8.5.8) the Parseval formula for k-transforms.

8.6. Proof \ddagger of Theorems 129, 130. Let f(x) be any function of $L^2(0,\infty)$, and $\mathfrak{F}(s)$ its Mellin transform, so that $\mathfrak{F}(\frac{1}{2}+it)$ belongs to $L^2(-\infty,\infty)$. Since $|\mathfrak{R}(\frac{1}{2}+it)|=1$, $\mathfrak{R}(\frac{1}{2}+it)\mathfrak{F}(\frac{1}{2}-it)$ also belongs to L^2 . Let g(x) be its Mellin transform. Then

$$\int_{0}^{x} g(u) du = \frac{1}{2\pi i} \int_{1-i\infty}^{1+i\infty} \Re(s) \Re(1-s) \frac{x^{1-s}}{1-s} ds.$$

Now $k_1(x)/x$ is the Mellin transform of $\Re(s)/(1-s)$. Hence, by the Parseval formula for Mellin transforms,

$$\frac{1}{2\pi i} \int_{1-i\infty}^{1+i\infty} \frac{\Re(s)}{1-s} \, \Re(1-s)x^{-s} \, ds = \int_{0}^{\infty} \frac{k_1(xy)}{xy} f(y) \, dy.$$

$$\int_{1}^{x} g(u) \, du = \int_{1}^{\infty} \frac{k_1(xy)}{y} f(y) \, dy,$$

Hence

and (8.5.6) follows almost everywhere. The k-transform g(x) of f(x) is thus the Mellin transform of $\Re(s)\Im(1-s)$ (on $\sigma=\frac{1}{2}$). By the same rule, the k-transform of g(x) is the Mellin transform of

$$\Re(s)\Re(1-s)\Im(s) = \Im(s).$$

Thus the k-transform of g(x) is f(x). All these transformations are of the class L^2 , so that the necessary uniqueness theorems hold.

This formula and the next come under Theorem 72, extended as in (2.1.23).

Finally,

$$\begin{split} \int\limits_{0}^{\infty} \{g(x)\}^2 \, dx &= \frac{1}{2\pi} \int\limits_{-\infty}^{\infty} \, |\Re(\frac{1}{2} + it) \Im(\frac{1}{2} - it)|^2 \, dt \\ &= \frac{1}{2\pi} \int\limits_{-\infty}^{\infty} \, |\Im(\frac{1}{2} + it)|^2 \, dt = \int\limits_{0}^{\infty} \{f(x)\}^2 \, dx, \end{split}$$

the Parseval theorem for k-transforms.

Theorem 130 also follows at once from the Parseval formula for Mellin transforms. Since $k_1(x)/x$ is the Mellin transform of $\Re(s)/(1-s)$,

$$\int_{0}^{\infty} \frac{k_{1}(ax)}{ax} \frac{k_{1}(bx)}{bx} dx = \frac{1}{2\pi i} \int_{1-i\infty}^{\frac{1}{4}+i\infty} \frac{\Re(s)a^{-s}}{1-s} \frac{\Re(1-s)b^{s-1}}{s} ds$$

$$= \frac{1}{2\pi i} \int_{1-i\infty}^{\frac{1}{4}+i\infty} \frac{a^{-s}b^{s-1}}{(1-s)s} ds,$$

by the functional equation for $\Re(s)$. If a > b, the integral may be evaluated by moving the line of integration away to the right, and the value is a^{-1} . If a > b, its value, obtained by moving the line of integration to the left, is b^{-1} . Also the integral on the left is continuous at a = b. Hence the result.

8.7. Proof of Theorem 131.† Suppose first that f(x) has a continuous derivative, and that it vanishes for all sufficiently small and sufficiently large values of x. Let

$$g_1(y) = \int_0^\infty \frac{k_1(xy)}{x} f(x) dx = \int_0^\infty \frac{k_1(u)}{u} f\left(\frac{u}{y}\right) du.$$

Then $g_1(y)$ is clearly differentiable, and

$$g(y) = g'_1(y) = -\frac{1}{y^2} \int_0^\infty k_1(u) f'\left(\frac{u}{y}\right) du = -\frac{1}{y} \int_0^\infty k_1(xy) f'(x) dx.$$

Hence

$$\int_{0}^{\infty} \{g(y)\}^{2} dy = \int_{0}^{\infty} \frac{dy}{y^{2}} \int_{0}^{\infty} k_{1}(xy)f'(x) dx \int_{0}^{\infty} k_{1}(\xi y)f'(\xi) d\xi$$

$$= \int_{0}^{\infty} f'(x) dx \int_{0}^{\infty} f'(\xi) d\xi \int_{0}^{\infty} \frac{k_{1}(xy)k_{1}(\xi y)}{y^{2}} dy$$

† Titchmarsh (15); see also Plancherei (6).

$$= \int_{0}^{\infty} f'(x) \, dx \int_{0}^{\infty} f'(\xi) \min(x, \xi) \, d\xi$$

$$= \int_{0}^{\infty} f'(x) x \, dx \int_{x}^{\infty} f'(\xi) \, d\xi + \int_{0}^{\infty} f'(\xi) \xi \, d\xi \int_{\xi}^{\infty} f'(x) \, dx$$

$$= -2 \int_{0}^{\infty} f(x) f'(x) x \, dx$$

$$= \left[-x \{ f(x) \}^{2} \right]_{0}^{\infty} + \int_{0}^{\infty} \{ f(x) \}^{2} \, dx$$

$$= \int_{0}^{\infty} \{ f(x) \}^{2} \, dx.$$

All the transformations are easily justified if f(x) satisfies the given conditions.

Next let f(x) be any function of $L^2(0,\infty)$. Then it is known that there is a sequence of functions $f_n(x)$, each satisfying the conditions previously imposed on f(x), and such that

$$\lim_{n\to\infty}\int_{0}^{\infty}\{f(x)-f_n(x)\}^2\,dx=0.$$

Let $g_n(x)$ correspond to $f_n(x)$ in the same way as the above g(x) does to f(x). Then

$$\int_{0}^{\infty} \{g_{m}(x) - g_{n}(x)\}^{2} dx = \int_{0}^{\infty} \{f_{m}(x) - f_{n}(x)\}^{2} dx,$$

which tends to 0 as m and n tend to infinity. Hence the sequence $g_n(x)$ converges in mean, to a function g(x) say. Then

$$\int_{0}^{\infty} \{g(x)\}^{2} dx = \lim_{n \to \infty} \int_{0}^{\infty} \{g_{n}(x)\}^{2} dx = \lim_{n \to \infty} \int_{0}^{\infty} \{f_{n}(x)\}^{2} dx = \int_{0}^{\infty} \{f(x)\}^{2} dx,$$

the Parseval formula.

Also

$$\int_{0}^{y} g(u) du = \lim_{n \to \infty} \int_{0}^{y} g_{n}(u) du$$

$$= \lim_{n \to \infty} \int_{0}^{\infty} \frac{k_{1}(xy)}{x} f_{n}(x) dx$$

$$= \int_{0}^{\infty} \frac{k_{1}(xy)}{x} f(x) dx,$$

so that

$$g(y) = \frac{d}{dy} \int_{x}^{\infty} \frac{k_1(xy)}{x} f(x) dx,$$

i.e. g(y) is the k-transform of f(x).

Let $\phi(x)$ be another function of $L^2(0,\infty)$, and $\psi(x)$ its k-transform. Then the Parseval formula gives

$$\int_0^\infty g(x)\psi(x)\ dx = \int_0^\infty f(x)\phi(x)\ dx.$$
 Let
$$\phi(x) = 1 \ (x \leqslant u), \quad 0 \ (x > u).$$
 Then
$$\int_0^\infty \frac{k_1(xy)}{x}\phi(x)\ dx = \int_0^u \frac{k_1(xy)}{x}\ dx = \int_0^{uy} \frac{k_1(x)}{x}\ dx,$$
 and hence
$$\psi(y) = \frac{d}{dy}\int_0^{uy} \frac{k_1(x)}{x}\ dx = \frac{k_1(uy)}{y}.$$
 Hence
$$\int_0^\infty g(x)\frac{k_1(ux)}{x}\ dx = \int_0^u f(x)\ dx,$$

and the reciprocal formula (8.5.7) follows.

8.8. Necessity of the conditions.† It is also easily seen that the conditions imposed on $k_1(x)$ and $\Re(s)$ in the above theorem are necessary. For suppose that the reciprocal formulae

$$\int_{0}^{x} f(y) dy = \int_{0}^{\infty} \frac{k_{1}(xu)}{u} g(u) du,$$

$$\int_{0}^{x} g(y) dy = \int_{0}^{\infty} \frac{k_{1}(xu)}{u} f(u) du$$
(8.8.2)

 $\int_{0}^{\infty} g(y) dy = \int_{0}^{\infty} \frac{\ln \chi(u, y)}{u} f(u) du \qquad (8.8.2)$

hold for any function f(x) of $L^2(0, \infty)$. Let f(x) = 1 $(x \leqslant \xi)$, 0 $(x > \xi)$. Then (8.8.2) gives

$$\int_{0}^{x} g(y) dy = \int_{0}^{\xi} \frac{k_{1}(xu)}{u} du = \int_{0}^{\xi x} \frac{k_{1}(v)}{v} dv,$$
$$g(x) = \frac{k_{1}(\xi x)}{x}.$$

so that

† Busbridge (1).

Substituting in (8.8.1), we obtain

$$\int_{0}^{\infty} \frac{k_1(xu)k_1(\xi u)}{u^2} du = \min(x,\xi).$$

In particular $(x = \xi = 1) k_1(u)/u$ belongs to $L^2(0, \infty)$.

If $\Re(s)/(1-s)$ is the Mellin transform of $k_1(x)/x$, Parseval's formula for Mellin transforms gives

$$\frac{1}{2\pi i} \int_{1-i\infty}^{1+i\infty} \frac{\Re(s)a^{-s}}{1-s} \frac{\Re(1-s)b^{s-1}}{s} ds = \int_{0}^{\infty} \frac{k_1(ax)}{ax} \frac{k_1(bx)}{bx} dx = \frac{\min(a,b)}{ab}.$$
But also
$$\frac{1}{2\pi i} \int_{1-s}^{1+i\infty} \frac{a^{-s}}{1-s} \frac{b^{s-1}}{s} ds = \frac{\min(a,b)}{ab}.$$

Hence (taking b = 1)

$$\frac{1}{2\pi i} \int_{\frac{1}{s}-i\infty}^{\frac{1}{s}+i\infty} \frac{1-\Re(s)\Re(1-s)}{(1-s)s} a^{-s} ds = 0$$

for all values of a. Since the integrand (as the product of functions of L^2) belongs to L, it follows from Theorem 32, p. 47, that it must be null, i.e. that $\Re(s)\Re(1-s)=1.$

8.9. The unsymmetrical formulae. For the transformation arising from the equation (8.3.5) a similar set of theorems holds. We now assume that $\mathfrak{H}(\frac{1}{2}+it)$ and $\mathfrak{R}(\frac{1}{2}+it)$ are both bounded. Let $h_1(x)/x$ and $k_1(x)/x$ be the Mellin transforms of $\mathfrak{H}(\frac{1}{2}+it)/(\frac{1}{2}-it)$ and $\Re(\frac{1}{2}+it)/(\frac{1}{2}-it)$. Then a given function f(x) of $L^2(0,\infty)$ has two transforms

$$g_h(x) = rac{d}{dx} \int\limits_0^\infty rac{h_1(xy)}{y} f(y) \ dy, \qquad g_k(x) = rac{d}{dx} \int\limits_0^\infty rac{k_1(xy)}{y} f(y) \ dy.$$

The k-transform of $g_h(x)$ is f(x), and so is the h-transform of $g_k(x)$. The usual Parseval formula is replaced by the relation

$$\int_0^\infty g_h(x)g_k(x)\ dx = \int_0^\infty \{f(x)\}^2\ dx,$$

together with the inequalities

$$\int\limits_{0}^{\infty}|g_{h}(x)|^{2}~dx < c\int\limits_{0}^{\infty}\{f(x)\}^{2}~dx,$$

$$\int\limits_{0}^{\infty}|g_{k}(x)|^{2}~dx < c\int\limits_{0}^{\infty}\{f(x)\}^{2}~dx.$$

The proof of Theorem 129 extends without substantial change to this case. That of 131 holds only if $\Re(\frac{1}{2}+it)$ and $\Re(\frac{1}{2}+it)$ are conjugate, so that $g_h(x)g_k(x)=|g_h(x)|^2$. In the general case the result still holds, but now we have to prove (8.3.5) as in § 8.8, and thence proceed as in the proof of Theorem 129.

8.10. A convergence theorem. In the foregoing theory the transformation is expressed in terms of $k_1(x)$, which is not necessarily differentiable. To obtain the forms (8.1.5), (8.1.6) we require further restrictions, both on the kernel and on the function represented.

THEOREM 132. Suppose (i) that $\hat{\mathbf{R}}(\frac{1}{2}+it)$ satisfies (8.5.1) and (8.5.4), so that $x^{-1}k_1(x)$, defined by (8.2.3) with $c=\frac{1}{2}$, belongs to $L^2(0,\infty)$; (ii) that $k_1(x)$ is the integral of k(x); (iii) that $x^{-1}k_1(x)$ is bounded.

Let
$$f(x) = \frac{1}{x} \int_{0}^{x} \phi(y) dy,$$
 (8.10.1)

where $\phi(y)$ belongs to $L^2(0,\infty)$. Then

$$f(x) = \int_{-\infty}^{\infty} k(xu) du \int_{-\infty}^{\infty} k(uy) f(y) dy$$
 (8.10.2)

for every positive x.

We have

$$|f(x)| \leqslant \frac{1}{x} \left\{ \int_{0}^{x} |\phi(y)|^{2} dy \int_{0}^{x} dy \right\}^{\frac{1}{2}} = o(x^{-\frac{1}{2}})$$

as $x \to 0$; and

$$|f(x)| \leq \frac{1}{x} \int_{0}^{x} |\phi(y)| dy + \frac{1}{x} \left(\int_{X}^{\infty} |\phi(y)|^{2} dy \int_{0}^{x} dy \right)^{\frac{1}{2}} = o(x^{-\frac{1}{2}})$$

as $x \to \infty$, by choosing first X and then x.

Let $\psi(x)$ be the k-transform of $\phi(x)$. Then $\psi(x)$ belongs to L^2 , and

$$\int_{0}^{x} \phi(y) dy = \int_{0}^{\infty} \frac{k_{1}(xu)}{u} \psi(u) du.$$

$$g(u) = \int_{x}^{\infty} \frac{\psi(v)}{v} dv.$$
(8.10.3)

Let

Then

$$\int_{0}^{x} \phi(y) dy = -\int_{0}^{\infty} k_{1}(xu)g'(u) du$$

$$= \lim_{\delta \to 0, \Delta \to \infty} \left\{ -\left[k_{1}(xu)g(u)\right]_{\delta}^{\Delta} + x \int_{\delta}^{\Delta} k(xu)g(u) du \right\}. \quad (8.10.4)$$

$$|g(u)| \leqslant \left\{ \int_{0}^{\infty} |\psi(v)|^{2} dv \int_{0}^{\infty} \frac{dv}{v^{2}} \right\}^{\frac{1}{2}} = o(u^{-\frac{1}{2}})$$

Now

as $u \to \infty$; and

$$|g(u)| \leqslant \int_{\delta}^{\infty} \left|\frac{\psi(v)}{v}\right| dv + \left\{\int_{0}^{\delta} |\psi(v)|^{2} dv \int_{u}^{\infty} \frac{dv}{v^{2}}\right\}^{\frac{1}{2}} = o(u^{-\frac{1}{2}})$$

as $u \to 0$, by choosing first δ and then u. Hence the integrated terms in (8.10.4) tend to 0, and we obtain

$$f(x) = \frac{1}{x} \int_{0}^{x} \phi(y) \, dy = \int_{-\infty}^{\infty} k(xu)g(u) \, du. \tag{8.10.5}$$

Again, (8.10.3) may be written

$$g(u) = \int_{0}^{\infty} \psi(v)\mu(v) dv,$$

where $\mu(v) = 0$ (v < u), 1/v (v > u). Hence by the Parseval formula

$$g(u) = \int_{v}^{\infty} \phi(v) \lambda(v) \ dv,$$

where

$$\begin{split} \lambda(v) &= \frac{d}{dv} \int\limits_0^\infty \mu(t) \frac{k_1(vt)}{t} \, dt = \frac{d}{dv} \int\limits_u^\infty \frac{k_1(vt)}{t^2} \, dt \\ &= \frac{d}{dv} \bigg\{ v \int\limits_{t}^\infty \frac{k_1(\xi)}{\xi^2} \, d\xi \bigg\} = \int\limits_{t}^\infty \frac{k_1(\xi)}{\xi^2} \, d\xi - \frac{k_1(uv)}{uv}. \end{split}$$

Hence

$$g(u) = \int_{0}^{\infty} \phi(v) \, dv \int_{uv}^{\infty} \frac{k_{1}(\xi)}{\xi^{2}} \, d\xi - \int_{0}^{\infty} \phi(v) \frac{k_{1}(uv)}{uv} \, dv$$

$$= g_{1}(u) - g_{2}(u) \tag{8.10.6}$$

say. Integrating by parts,

$$g_1(u) = \left[vf(v) \int_{v_1}^{\infty} \frac{k_1(\xi)}{\xi^2} d\xi \right]_0^{\infty} + \int_{-\infty}^{\infty} vf(v) u \frac{k_1(uv)}{u^2 v^2} dv, \quad (8.10.7)$$

and the integrated terms vanish since $vf(v) = o(v^{\frac{1}{2}})$, and, since $k_1(\xi)/\xi$ is L^2 ,

 $\int_{0}^{\infty} \frac{k_1(\xi)}{\xi^2} d\xi = o(v^{-\frac{1}{2}})$

as in the case of g(u). Also

$$g_2(u) = \left[v f(v) \frac{k_1(uv)}{uv} \right]_0^{\infty} - \int_{-\infty}^{\infty} v f(v) \frac{k(uv)}{v} dv + \int_{-\infty}^{\infty} v f(v) \frac{k_1(uv)}{uv^2} dv, \tag{8.10.8}$$

and the integrated terms vanish since $v^{-\frac{1}{2}}k_1(uv) = O(1)$.

From (8.10.6), (8.10.7), and (8.10.8) it follows that

$$g(u) = \int_{-\infty}^{\infty} k(uv)f(v) dv, \qquad (8.10.9)$$

and (8.10.5) and (8.10.9) give the theorem.

8.11. The resultant of two Fourier kernels.† Let

$$m(x) = \int_{0}^{\infty} k(xy)l(y) dy$$

be the resultant of k(x) and l(x). Then a formal rule is that, if k(x) and l(x) are Fourier kernels, so is m(x). We may, for example, put

$$\iint_{0}^{\infty} m(xu)m(ut)f(t) \ dudt = \iiint_{0}^{\infty} \iint_{0}^{\infty} k(xuy)k(utz)l(y)l(z)f(t) \ dudtdydz,$$

and the substitution t = v/z, y = zw gives

$$\int_{0}^{\infty} \int_{0}^{\infty} l(z)l(zw) \ dzdw \int_{0}^{\infty} \int_{0}^{\infty} k(xzwu)k(uv)f\left(\frac{v}{z}\right) dudv$$

$$= \iint_{0}^{\infty} l(z)l(zw)f(xw) \ dzdw = f(x)$$

if k and l are Fourier kernels.

We can also argue in terms of Mellin transforms. If \Re and $\mathfrak L$ are the Mellin transforms of k and l, that of m is

$$\mathfrak{M}(s) = \int_{0}^{\infty} m(x)x^{s-1} dx = \int_{0}^{\infty} x^{s-1} dx \int_{0}^{\infty} k(xy)l(y) dy$$

$$= \int_{0}^{\infty} l(y) dy \int_{0}^{\infty} k(xy)x^{s-1} dx = \int_{0}^{\infty} l(y)y^{-s} dy \int_{0}^{\infty} k(u)u^{s-1} du$$

$$= \mathfrak{L}(1-s)\mathfrak{R}(s).$$
† Hardy (20).

Hence

$$\mathfrak{M}(s)\mathfrak{M}(1-s) = \mathfrak{R}(s)\mathfrak{R}(1-s)\mathfrak{L}(1-s)\mathfrak{L}(s) = 1,$$

and the result again follows. The argument is still of course purely formal.

The L^2 theory gives

THEOREM 133. Let $k_1(x)$ and $l_1(x)$ satisfy the conditions of Theorem 131, and let $m_1(1/x)$ be the l-transform of $k_1(x)/x$. Then $m_1(x)$ also satisfies the conditions of Theorem 131.

Here $m_1(1/x)$ is defined by

$$\int_{0}^{x} m_{1}\left(\frac{1}{u}\right) du = \int_{0}^{\infty} \frac{l_{1}(xu)}{u} \frac{k_{1}(u)}{u} du.$$

Now $m_1(a/x)$ is the *l*-transform of $k_1(ax)/x$. Hence by Parseval's formula for *l*-transforms

$$\int_0^\infty \frac{m_1(ax)m_1(bx)}{x^2} dx = \int_0^\infty m_1\left(\frac{a}{x}\right)m_1\left(\frac{b}{x}\right) dx$$

$$= \int_0^\infty \frac{k_1(ax)k_1(bx)}{x^2} dx = \min(a,b),$$

the required result.

As a particular case, let $s_1(x) = 0$ (x < 1), 1 $(x \ge 1)$, so that

$$g(x) = \frac{1}{x} f\left(\frac{1}{x}\right), \qquad f(x) = \frac{1}{x} g\left(\frac{1}{x}\right).$$

We call this the transformation S. If $k_1 = l_1$, then

$$\int_{0}^{x} m_{1} \left(\frac{1}{y}\right) dy = \int_{0}^{\infty} \frac{k_{1}(t)k_{1}(xt)}{t^{2}} dt = \min(1, x),$$

and $m_1 \equiv s_1$. If $l_1 = s_1$, then

$$\int_0^x m_1\left(\frac{1}{y}\right) dy = \int_{1/x}^\infty \frac{k_1(t)}{t^2} dt = \int_0^x k_1\left(\frac{1}{t}\right) dt,$$

and $m_1 \equiv k_1$. Thus the resultant of k and k is s, the resultant of k and s is k.

Examples. (1) If k and l are the cosine and sine transformations,

$$m_1(x) = \frac{1}{\pi} \log \left| \frac{x-1}{x+1} \right|, \quad m(x) = \frac{2}{\pi} \frac{1}{1-x^2},$$

and the m-transformation is

$$g(x) = \frac{2}{\pi} \int\limits_0^\infty \frac{f(t)}{1 - x^2 t^2} dt.$$

If f is even, this gives

$$\frac{1}{x}g\left(\frac{1}{x}\right) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{f(t)}{x-t} dt,$$

the Hilbert transform of f(t).

The resultant of this transformation and k is defined by

$$m(x) = \frac{2}{\pi} \int_{0}^{\infty} \frac{k(xt)}{1-t^2} dt,$$

or, regarding k(x) as even, by

$$m(x) = \frac{1}{\pi} \int_{-\pi}^{\infty} \frac{k(t)}{x-t} dt.$$

Thus the conjugate of a Fourier kernel is a Fourier kernel.

(2) The function $l_1(x) = x$ (x < 1), 0 ($x \ge 1$), satisfies (8.5.5). We conclude that, if k(x) is a Fourier kernel, then so is

$$m(x) = \int_{0}^{\infty} k(xt) \, dl_{1}(t) = \int_{0}^{1} k(xt) \, dt - k(x) = \frac{1}{x} \int_{0}^{x} k(u) \, du - k(x).$$

Similarly, taking $l_1(x) = 0$ (x < 1), $\log x - 1$ $(x \ge 1)$, we find that

$$\int_{a}^{\infty} \frac{k(u)}{u} du - k(x)$$

is a Fourier kernel.

- (3) The resultant of $t^{\frac{1}{2}}J_{\nu}(t)$ and $t^{-\frac{1}{2}}J_{\nu-1}(1/t)$ is $J_{2\nu-1}(2t^{\frac{1}{2}})$.
- (4) The resultant of $\sqrt{(2/\pi)\cos x}$, $\sqrt{(2/\pi)x^{-1}\cos x^{-1}}$ is (§ 7.12)

$$\frac{2}{\pi}\{K_0(2\sqrt{x})-Y_0(2\sqrt{x})\};$$

that of $\sqrt{(2/\pi)\sin x}$, $\sqrt{(2/\pi)x^{-1}\sin x^{-1}}$ is $(2/\pi)\{K_0(2\sqrt{x})+Y_0(2\sqrt{x})\}$. The last kernel is also the conjugate of $J_0(2\sqrt{x})$.

(5) The resultant of $J_0(2\sqrt{x})$ and $\cos x$ is $-\sin x$, and that of $J_0(2\sqrt{x})$ and $\sin x$ is $\cos x$. This may be proved as in § 7.12.

† Watson, § 13.61 (1), or as in § 7.12.

(6) We have

$$x^{\frac{1}{2}}\int_{0}^{\infty}tJ_{\mu}(xt)J_{-\mu}(t)\ dt = -\frac{2\sin\mu\pi}{\pi}\frac{x^{\mu+\frac{1}{2}}}{1-x^{2}}$$
 (C, 1),

if $x \neq 1$, while when x = 1 the integral diverges like

$$\frac{\cos \mu \pi}{\pi} \int dt.$$

The divergence indicates that when we form the resultant of $x^{\frac{1}{2}}J_{\mu}(x)$ and $x^{\frac{1}{2}}J_{-\mu}(x)$, there will be a discontinuity in $m_1(x)$ at x=1. In fact

$$m_1(x) = -rac{2\sin\mu\pi}{\pi} \int_0^x rac{t^{\mu+1}}{1-t^2} dt \quad (x < 1),$$
 $-rac{2\sin\mu\pi}{\pi} \int_0^x rac{t^{\mu+1}}{1-t^2} dt + \cos\mu\pi \quad (x \geqslant 1).$

The inversion formulae are

$$g(x) = -\frac{2\sin \mu\pi}{\pi} \int_{0}^{\infty} \frac{(xt)^{\mu+\frac{1}{2}}}{1-x^{2}t^{2}} f(t) dt + \cos \mu\pi \frac{1}{x} f\left(\frac{1}{x}\right)$$

and the reciprocal formula.

(7) If we form the resultant m(x) of $\sqrt{(2/\pi)\cos x}$ and $J_{\frac{1}{2}}(2\sqrt{x})$, and then replace m(x) by $2^{-\frac{1}{2}}x^{-1}m(1/2x)$, we obtain the Fourier kernel

$$(2x)^{\frac{1}{2}} \{\cos(x-\frac{1}{8}\pi)J_{\frac{1}{2}}(x)+\sin(x-\frac{1}{8}\pi)J_{-\frac{1}{2}}(x)\}.$$

8.12. Convergence of k-integrals. We now leave the transform theory, and prove quite independently a theorem on convergence in the ordinary sense. To do this we have to make very special assumptions, and the theory is practically restricted to those examples in § 8.4 in which $\Re(s)$ is a product of Γ -functions. For such functions, however, we obtain a direct generalization of Theorem 3.

THEOREM 134.† Let $\Re(s)$ be regular in a strip $\sigma_1 < \sigma < \sigma_2$, where $\sigma_1 < 0$, $\sigma_2 > 1$, except perhaps for a finite number of simple poles on the imaginary axis; and let $\Re(s)$ be of the forms

$$\Re_0(s)\left\{\alpha + \frac{\beta}{s} + O\left(\frac{1}{|s|^2}\right)\right\}, \qquad \Re_0(s)\left\{\gamma + \frac{\delta}{s} + O\left(\frac{1}{|s|^2}\right)\right\}$$

† Hardy and Titchmarsh (8).

for large positive and negative t respectively, where

$$\Re_0(s) = \Gamma(s) \cos \tfrac{1}{2} s \pi$$

is the Mellin transform of $\cos x$. Let $\Re(s)$ satisfy (8.1.9), and let k(x) be the Mellin transform of $\Re(s)$.

Let x > 0, and let f(y) be $L(0, \infty)$, and be of bounded variation near y = x. Then

$$\int_{0}^{\infty} k(xu) \, du \int_{0}^{\infty} k(uy) f(y) \, dy = \frac{1}{2} \{ f(x+0) + f(x-0) \}. \quad (8.12.1)$$

The function $\Re_0(s)$ is regular in any strip $\sigma_1 < \sigma < \sigma_2$, except for a finite number of simple poles at points where $\sigma \leqslant 0$. If t is large and positive, then

$$\Re_0(\sigma+it) = Ct^{\sigma-\frac{1}{2}}e^{i(t\log t-t)}\left\{1+\frac{a}{t}+O\left(\frac{1}{t^2}\right)\right\},\,$$

where C and a are complex, and a depends on σ ; and $\Re_0(\sigma - it)$ satisfies the conjugate formula.

The functions

$$\Gamma(s)\sin\frac{1}{2}s\pi$$
, $\Gamma(s)\frac{\cos\frac{1}{2}s\pi}{1-s}$, $\Gamma(s)\frac{\sin\frac{1}{2}s\pi}{2-s}$

are the Mellin transforms of

$$\sin x$$
, $\frac{\sin x}{x}$, $\frac{\sin x - x \cos x}{x^2}$,

and are of the form

$$\begin{split} \Re_0(s) \Big\{ i \operatorname{sgn} t + O\Big(\frac{1}{|s|^2}\Big) \Big\}, & \Re_0(s) \Big\{ -\frac{1}{s} + O\Big(\frac{1}{|s|^2}\Big) \Big\}, \\ \Re_0(s) \Big\{ \frac{-i \operatorname{sgn} t}{s} + O\Big(\frac{1}{|s|^2}\Big) \Big\}, \end{split}$$

for large t. If $\Re(s)$ satisfies the conditions of the theorem, we can find constants a_1 , a_2 , a_3 , a_4 , such that

$$\Re(s) = \Re^{(1)}(s) + \Re^{(2)}(s) + \Re^{(3)}(s),$$

where

$$\Re^{(1)}(s) = a_1 \Gamma(s) \cos \frac{1}{2} s \pi + a_2 \Gamma(s) \sin \frac{1}{2} s \pi$$

$$\Re^{(2)}(s) = a_3 \Gamma(s) \frac{\cos \frac{1}{2} s \pi}{1-s} + a_4 \Gamma(s) \frac{\sin \frac{1}{2} s \pi}{2-s},$$

and

$$\Re^{(3)}(s) = O\{|\Re_0(s)s^{-2}|\} = O(|t|^{\sigma - \frac{1}{2}})$$

for large s of the strip. Let $k^{(1)}(x),...$ be the Mellin transforms of $\Re^{(1)}(s),...$.

8.13. Lemma α . k(x) is bounded for all positive x.

This is true for $k^{(1)}(x)$ and $k^{(2)}(x)$, so that it is enough to prove that

$$k^{(3)}(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re^{(3)}(s) x^{-s} ds$$

is bounded. If $x \ge 1$, we take $c = 1 + \delta$, where $0 < \delta < \frac{1}{2}$, $1 + \delta < \sigma_2$. Since $\Re^{(3)}(s)$ is then $O(|s|^{\delta - \frac{1}{2}})$, $k^{(3)}(x)$ is bounded, and indeed is $O(x^{-1-\delta})$. If $0 < x \le 1$, we take $c = -\delta$, where $\sigma_1 < -\delta < 0$. Then

$$k^{(3)}(x) = rac{1}{2\pi i} \int_{-\delta - i\infty}^{-\delta + i\infty} \Re^{(3)}(s) x^{-s} ds +
ho,$$

the latter term being the sum of the residues at any poles on the imaginary axis. It is plain that ρ is bounded, and the integral is bounded because $\Re^{(3)}(s) = O(|s|^{-\delta-1})$. Hence k(x) is bounded for all positive x.

8.14. LEMMA β . Let

$$\phi(\lambda, x, y) = \int_{1/\lambda}^{\lambda} k(xu)k(yu) du, \qquad (8.14.1)$$

where $\lambda > 1$ and x is positive and fixed. Then

$$|\phi| < B(x,\zeta) \tag{8.14.2}$$

for all positive y for which $|y-x| \geqslant \zeta$.

In view of Lemma α we may replace ϕ by

$$\chi(\lambda, x, y) = \int_{1}^{\lambda} k(xu)k(yu) du$$

$$= \int_{1}^{\lambda} \{k^{(1)}(xu) + k^{(2)}(xu)\}\{k^{(1)}(yu) + k^{(2)}(yu) + k^{(3)}(yu)\} du + \int_{1}^{\lambda} k^{(3)}(xu)k(yu) du.$$

The last term is bounded because $k^{(3)}(xu) = O(u^{-1-\delta})$ and k(yu) is bounded.

Denote the integral involving $k^{(p)}(xu)k^{(q)}(yu)$ by $\chi_{p,q}$. Then $\chi_{1,1}$ is clearly bounded. Next, $\chi_{1,2}$ splits up into four terms, a typical term being

$$\int_{1}^{\lambda} \sin xu \frac{\sin yu - yu \cos yu}{y^{2}u^{2}} du = \int_{1}^{1/y} \sin xu \frac{\sin yu - yu \cos yu}{y^{2}u^{2}} du +$$

$$+ \int_{1/y}^{\lambda} \frac{\sin xu \sin yu}{y^{2}u^{2}} du - \int_{1/y}^{\lambda} \frac{\sin xu \cos yu}{yu} du.$$

Since $(\sin x - x \cos x)/x^2$ is positive increasing in 0 < x < 1, the first term on the right is

$$(\sin 1 - \cos 1) \int_{u_1}^{1/y} \sin xu \ du,$$

where $0 < u_1 < 1/y$. The second and third are

$$\int_{1/y}^{u_*} \sin xu \sin yu \ du, \qquad -\int_{1/y}^{u_*} \sin xu \cos yu \ du,$$

where $u_2 > 1/y$, $u_3 > 1/y$. All these are bounded, and the other terms of $\chi_{1,2}$ may be shown to be bounded in the same way. Hence $\chi_{1,2}$ is bounded.

A similar argument applies to $\chi_{2,1}$ and $\chi_{2,2}$. Thus a typical term of $\chi_{2,2}$ is

$$\int_{1}^{\lambda} \frac{\sin xu}{xu} \frac{\sin yu}{yu} du$$

$$= \int_{1}^{1/y} + \int_{1/y}^{\lambda} = \frac{\sin y}{y} \int_{1}^{u_{1}} \frac{\sin xu}{xu} du + \int_{1/y}^{u_{1}} \frac{\sin xu}{xu} \sin yu du,$$

and each of these is bounded.

A typical term in $\chi_{1,3}$ is

$$\frac{1}{2\pi i}\int\limits_{1}^{\lambda}\cos xu\;du\int\limits_{c-i\infty}^{c+i\infty}(yu)^{-s}\Re^{(3)}(s)\;ds.$$

If $\Re^{(3)}$ has no pole on the imaginary axis, we may take c=0, and invert, and obtain

$$\frac{1}{2\pi}\int_{-\infty}^{\infty}y^{-u}\mathcal{R}^{(3)}(it)\ dt\int_{1}^{\lambda}u^{-u}\cos xu\ du.$$

The inner integral is $O(\lambda) = O(t)$ for $t > \lambda$, while for $0 < t < \lambda$ it

differs by O(t) from

$$\int_{t}^{\lambda} \frac{\cos xu}{u^{it}} du = \left[\frac{\sin xu}{xu^{it}}\right]_{t}^{\lambda} - it \left[\frac{\cos xu}{x^{2}u^{it+1}}\right]_{t}^{\lambda} - \frac{it(it+1)}{x^{2}} \int_{t}^{\lambda} \frac{\cos xu}{u^{it+2}} du,$$

each term of which is O(t). Since $\Re^{(3)}(it) = O(|t|^{-\frac{1}{2}})$ for large t, the term in question is bounded.

If there are poles on the imaginary axis, it is sufficient to consider one of them, say at $s = i\tau$ with residue C. Let

$$\Re^{(3)}(s) = C\Gamma(s-i\tau) + \Re^{(4)}(s), \qquad k^{(3)}(x) = Cx^{-i\tau}e^{-x} + k^{(4)}(x).$$

Then $\mathfrak{R}^{(4)}$ satisfies the conditions imposed above on $\mathfrak{R}^{(3)}$, and the additional term is

$$Cy^{-i\tau}\int_{1}^{\lambda}u^{-i\tau}e^{-yu}\cos xu\ du=Cy^{-i\tau}\int_{1}^{\lambda'}u^{-i\tau}\cos xu\ du=O(1)$$

by the argument used for the above inner integral. Hence $\chi_{1,3}$ is bounded. Practically the same argument proves that $\chi_{2,3}$ is bounded, and the lemma follows.

8.15. LEMMA γ . Let

$$\psi(\lambda, x, y) = \int_{1/\lambda}^{\lambda} k(xu) \frac{k_1(yu)}{u} du,$$

where

$$k_1(x) = \int_0^x k(u) \ du = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re(s) \frac{x^{1-s}}{1-s} \ ds.$$

Then $|\psi| < B(x,\zeta)$ for $\lambda > 1$, x > 0, and $0 < x-\zeta < y < x+\zeta$; and $\psi(\lambda, x, y)$ converges (boundedly) as $\lambda \to \infty$ to the limit

$$0 \quad (y < x), \qquad \frac{1}{2} \quad (y = x), \qquad 1 \quad (y > x).$$

Since k(u) = O(1), $k_1(u) = O(u)$ for small u, the integral over $(1/\lambda, 1)$ is bounded. We now write

$$k(x) = k^{(1)}(x) + k^{(5)}(x),$$

where $k^{(1)}$ is the same as before; and

$$\begin{split} & \int\limits_{1}^{\lambda} k(xu) \frac{k_{1}(yu)}{u} \, du \\ & = \int\limits_{1}^{\lambda} k^{(1)}(xu) \frac{k_{1}^{(1)}(yu)}{u} \, du + \int\limits_{1}^{\lambda} k(xu) \frac{k_{1}^{(5)}(yu)}{u} \, du + \int\limits_{1}^{\lambda} k^{(5)}(xu) \frac{k_{1}^{(1)}(yu)}{u} \, du. \end{split}$$

The first term is a multiple of

$$\begin{split} \int\limits_{1}^{\lambda} \left(a_{1} \cos x u + a_{2} \sin x u\right) \frac{a_{1} \sin y u + a_{2} (1 - \cos y u)}{u} \, du \\ &= a_{1}^{2} \int\limits_{1}^{\lambda} \frac{\cos x u \sin y u}{u} \, du + a_{1} a_{2} \int\limits_{1}^{\lambda} \frac{\cos x u - \cos(x + y) u}{u} \, du + \\ &+ a_{2}^{2} \int\limits_{1}^{\lambda} \frac{\sin x u (1 - \cos y u)}{u} \, du, \end{split}$$

each term of which converges boundedly. Also k(u) and $k_1^{(1)}(u)$ are bounded,

$$k^{(5)}(u) = O\left(\int_{-\infty}^{\infty} (1+|t|)^{\sigma-\frac{3}{2}}u^{-\sigma} dt\right) = O(u^{-\frac{1}{2}+\delta}),$$

taking $\sigma = \frac{1}{2} - \delta$; and $k_1^{(5)}(u)$, like $uk^{(3)}(u)$, is $O(u^{-\delta})$. The remaining terms are therefore bounded.

This proves the lemma except as regards the value of the limit. To calculate this directly requires some further examination of the argument, but the result can be obtained from the transform theory. We have in fact proved that

$$\int\limits_0^\infty \frac{k_1(u)k(xu)}{u}\,du$$

converges boundedly for $x \ge \delta$, where $0 < \delta < 1$, and uniformly except near x = 1; hence, if its value is $\phi(x)$,

$$\int_{\delta}^{x} \phi(u) \ du = \int_{0}^{\infty} \frac{k_{1}(x)\{k_{1}(xu) - k_{1}(\delta u)\}}{u^{2}} \ du = \min(x, 1) - \delta$$

and hence If x = 1,

$$\phi(x) = 1$$
 $(x < 1)$, 0 $(x > 1)$.

$$\int_{0}^{X} \frac{k_{1}(u)k(u)}{u} du = \frac{k_{1}^{2}(X)}{X} - \int_{0}^{X} \frac{k(u)k_{1}(u)}{u} du + \int_{0}^{X} \frac{k_{1}^{2}(u)}{u^{2}} du,$$

and since each integral tends to a limit as $X \to \infty$, so does $k_1^2(X)/X$, and this limit must be zero since $k_1^2(X)/X^2$ belongs to $L(0,\infty)$. Hence

$$2\int_{-\infty}^{\infty} \frac{k_1(u)k(u)}{u} du = \int_{-\infty}^{\infty} \frac{k_1^2(u)}{u^2} du = 1$$

8.16. The Riemann-Lebesgue theorem is here replaced by the following theorem, due to Hobson (1).

LEMMA 8. Let f(t) belong to L(a,b); let $\phi(\lambda,t)$ belong to L(a,b) for all values of λ , let it be bounded uniformly with respect to λ in (a,b); and let

 $\int_{\alpha}^{\beta} \phi(\lambda, t) dt \to 0$

as $\lambda \to \infty$, uniformly in α and β for $a \leqslant \alpha < \beta \leqslant b$. Then

$$\lim_{\lambda\to\infty}\int\limits_a^b f(t)\phi(\lambda,t)\ dt=0.$$

Suppose first that f(t) is absolutely continuous in (a, b). Let

$$\int_{a}^{t} \phi(\lambda, u) du = \phi_{1}(\lambda, t).$$

Then

$$\int_a^b f(t)\phi(\lambda,t)\ dt = f(b)\phi_1(\lambda,b) - \int_a^b f'(t)\phi_1(\lambda,t)\ dt.$$

Given ϵ , we have

$$|\phi_1(\lambda,t)| < \epsilon$$
 $(\lambda > \lambda_0(\epsilon), a \leqslant t \leqslant b),$

and hence

$$\left|\int_a^b f(t)\phi(\lambda,t)\,dt\right| \leqslant \epsilon \left\{|f(b)| + \int_a^b |f'(t)|\,dt\right\} \quad (\lambda > \lambda_0).$$

The result therefore follows in this case.

In the general case we can, given ϵ , define an absolutely continuous function $\chi(t)$ such that

$$\int_{a}^{b} |f(t)-\chi(t)| dt < \epsilon.$$

Having fixed t and $\chi(t)$, we can, by the first part, choose λ_0 so large that

 $\left|\int_a^b \chi(t)\phi(\lambda,t)\ dt\right|<\epsilon.$

If $|\phi(\lambda,t)| \leq M$, it follows that

$$\left| \int_{a}^{b} f(t)\phi(\lambda,t) dt \right| \leq \left| \int_{a}^{b} \chi(t)\phi(\lambda,t) dt \right| + \left| \int_{a}^{b} \{f(t) - \chi(t)\}\phi(\lambda,t) dt \right|$$
$$\leq \epsilon + M\epsilon \quad (\lambda > \lambda_{0}).$$

This proves the lemma.

8.17. Proof of Theorem 134. By Lemma α the integral

$$\int\limits_0^\infty k(uy)f(y)\;dy$$

is uniformly convergent for $1/\lambda \leqslant u \leqslant \lambda$, so that we may multiply by k(xu) and integrate under the integral sign. Hence

$$\int_{1/\lambda}^{\lambda} k(xu) du \int_{0}^{\infty} k(uy)f(y) dy = \int_{0}^{\infty} f(y)\phi(\lambda, x, y) dy$$

$$= \int_{0}^{\delta} + \int_{\delta}^{x-\zeta} + \int_{x-\zeta}^{x} + \int_{x}^{x+\zeta} + \int_{x+\zeta}^{\Delta} + \int_{\Delta}^{\infty}$$

$$= I_{1} + I_{2} + I_{3} + I_{4} + I_{5} + I_{6}.$$

It follows from Lemma β that

$$|I_1| \leqslant B\int\limits_0^\delta |f(y)| \ dy < \epsilon, \qquad |I_6| < B\int\limits_\Delta^\infty |f(y)| \ dy < \epsilon$$

for $\delta = \delta(\epsilon)$, $\Delta = \Delta(\epsilon)$, $\lambda > 2$.

Next

$$\int_{\alpha}^{\beta} \phi \ dy = \int_{1/\lambda}^{\lambda} k(xu) \frac{k_1(\beta u) - k_1(\alpha u)}{u} \ du$$
$$= \psi(\lambda, x, \beta) - \psi(\lambda, x, \alpha).$$

If $\alpha < \beta < x$, or $x < \alpha < \beta$, this tends to 0, when $\lambda \to \infty$, by Lemma γ . Hence, by Lemma δ ,

$$\lim_{\lambda \to \infty} I_2 = 0, \qquad \lim_{\lambda \to \infty} I_5 = 0$$

when ζ , δ , and Δ are fixed.

We may suppose ζ small enough to ensure that f(y) is of bounded variation in $(x-\zeta,x+\zeta)$, and then

$$f(y)-f(x-0) = f_1(y)-f_2(y),$$

where f_1 and f_2 are positive and decreasing and tend to 0 when $y \to x$ from below. Then

$$\begin{split} I_3 &= f(x-0)\{\psi(\lambda,x,x) - \psi(\lambda,x,x-\zeta)\} + \\ &+ \int\limits_{x-\zeta}^x f_1(y)\phi(\lambda,x,y) \ dy - \int\limits_{x-\zeta}^x f_2(y)\phi(\lambda,x,y) \ dy. \end{split}$$

The first term tends to $\frac{1}{2}f(x-0)$. The second is

$$f_1(x-\zeta)\int_{x-\zeta}^{\eta}\phi\ dy=f_1(x-\zeta)\{\psi(\lambda,x,\eta)-\psi(\lambda,x,x-\zeta)\},$$

where $x-\zeta < \eta < x$, and, since ψ is bounded, this is less than ϵ (for all λ in question) if ζ is sufficiently small. A similar argument applies to the third term. Hence

$$\left|\overline{\lim_{\lambda\to\infty}}I_3-\frac{1}{2}f(x-0)\right|\leqslant 2\epsilon$$

if ζ is sufficiently small. There is a corresponding result for I_4 , and it follows that

$$\lim_{\lambda\to\infty}\int_{1/\lambda}^{\lambda}k(xu)\int_{0}^{\infty}k(uy)f(y)\,dy=\tfrac{1}{2}\{f(x+0)+f(x-0)\}.$$

The *u*-integrand is, however, bounded as $u \to 0$, so that this may be replaced by (8.12.1). This proves the theorem.

It is easily verified that the $\Re(s)$ which gives rise to Hankel's theorem satisfies the above conditions if $\nu \geqslant -\frac{1}{2}$; and so do all the other \Re 's which are products of Γ -functions if the parameters involved are subject to suitable restrictions.

8.18. Hankel's theorem.† The most important particular case of the foregoing theorem is that in which $k(x) = \sqrt{x} J_{\nu}(x)$. This case can be obtained much more simply.

THEOREM 135. If f(x) belongs to $L(0,\infty)$, and is of bounded variation near the point x, then for $\nu \geqslant -\frac{1}{2}$

$$\frac{1}{2}\{f(x+0)+f(x-0)\} = \int_{0}^{\infty} J_{\nu}(xu)\sqrt{(xu)} \, du \int_{0}^{\infty} J_{\nu}(uy)\sqrt{(uy)}f(y) \, dy.$$
(8.18.1)

Let δ be a small positive number. Then

$$\int_{0}^{\lambda} J_{\nu}(xu)\sqrt{(xu)} du \int_{0}^{x-\delta} J_{\nu}(uy)\sqrt{(uy)}f(y) dy$$

$$= \sqrt{x} \int_{0}^{x-\delta} \sqrt{y}f(y) dy \int_{0}^{\lambda} J_{\nu}(xu)J_{\nu}(uy)u du$$

$$= \sqrt{x} \lambda \int_{0}^{x-\delta} \frac{xJ_{\nu+1}(\lambda x)J_{\nu}(\lambda y)-yJ_{\nu+1}(\lambda y)J_{\nu}(\lambda x)}{x^{2}-y^{2}} \sqrt{y}f(y) dy \qquad (8.18.2)$$

$$= O(\sqrt{\lambda}) \int_{0}^{x-\delta} J_{\nu}(\lambda y) \frac{\sqrt{y}f(y)}{x^{2}-y^{2}} dy + O(\sqrt{\lambda}) \int_{0}^{x-\delta} J_{\nu+1}(\lambda y) \frac{y^{4}f(y)}{x^{2}-y^{2}} dy, \qquad (8.18.3)$$

† Watson, Chap. 14.

for any fixed x and δ . Now

$$\int_{0}^{1/\lambda} J_{\nu}(\lambda y) \frac{\sqrt{y} f(y)}{x^{2} - y^{2}} dy = O\left(\int_{0}^{1/\lambda} (\lambda y)^{\nu} y^{\frac{1}{2}} |f(y)| dy\right)$$

$$= O\left(\lambda^{\nu} \int_{0}^{1/\lambda} y^{\nu + \frac{1}{2}} |f(y)| dy\right) = O\left(\lambda^{-\frac{1}{2}} \int_{0}^{1/\lambda} |f(y)| dy\right) = O\left(\lambda^{-\frac{1}{2}}\right).$$

For $\lambda y \geqslant 1$ we have

$$J_{\nu}(\lambda y) = \frac{A\cos\lambda y + B\sin\lambda y}{(\lambda y)^{\frac{1}{2}}} + O\left(\frac{1}{(\lambda y)^{\frac{1}{2}}}\right).$$

The O-term contributes

$$O\left(\lambda^{-\frac{1}{2}}\int_{1/\lambda}^{x-\delta}|f(y)|\frac{dy}{y}\right)$$

$$=O\left(\lambda^{-\frac{1}{2}}\int_{0}^{1/\sqrt{\lambda}}|f(y)|dy\right)+O\left(\lambda^{-1}\int_{1/\sqrt{\lambda}}^{x-\delta}|f(y)|dy\right)=o(\lambda^{-\frac{1}{2}}),$$

and the main term contributes

$$\lambda^{-\frac{1}{4}} \int_{1/\lambda}^{x-\delta} (A\cos\lambda y + B\sin\lambda y) \frac{f(y)}{x^2 - y^2} dy = o(\lambda^{-\frac{1}{4}})$$

by the Riemann-Lebesgue theorem. The second term in (8.18.3) may be dealt with in a similar way. Hence (8.18.2) tends to 0 as $\lambda \to \infty$.

Next, we may invert

$$\int_{0}^{\lambda} J_{\nu}(xu)\sqrt{(xu)} \ du \int_{x+\lambda}^{\infty} J_{\nu}(uy)\sqrt{(uy)}f(y) \ dy$$

by the uniform convergence of the y-integral. The proof that this part tends to 0 is then similar, but simpler, since here y is not small.

We can suppose δ so small that f(y) is of bounded variation over $(x-\delta,x+\delta)$. Then so is $y^{-\nu-i}f(y)$. Hence in $(x,x+\delta)$ we can write

$$y^{-\nu-\frac{1}{2}}f(y) = x^{-\nu-\frac{1}{2}}f(x+0) + \chi_1(y) - \chi_2(y),$$

where χ_1 and χ_2 are positive, increasing, and less than ϵ . Then

$$\int_{0}^{\lambda} J_{\nu}(xu) \sqrt{(xu)} \, du \int_{x}^{x+\delta} J_{\nu}(uy) \sqrt{(uy)} f(y) \, dy$$

$$= \sqrt{x} \int_{0}^{\lambda} J_{\nu}(xu) u \, du \int_{x}^{x+\delta} J_{\nu}(uy) y^{\nu+1} \{x^{-\nu-\frac{1}{2}} f(x+0) + \chi_{1}(y) - \chi_{2}(y)\} \, dy.$$

The first term in the bracket contributes

$$x^{-\nu}f(x+0)\int_{0}^{\Lambda}J_{\nu}(xu)[J_{\nu+1}\{(x+\delta)u\}(x+\delta)^{\nu+1}-J_{\nu+1}(xu)x^{\nu+1}]du$$

$$\to x^{-\nu}f(x+0)(x^{\nu}-\frac{1}{2}x^{\nu})=\frac{1}{2}f(x+0),$$

by (7.11.15). The second term contributes

$$\sqrt{x} \int_{0}^{\lambda} J_{\nu}(xu)u \ du \int_{x}^{x+\delta} J_{\nu}(uy)y^{\nu+1}\chi_{1}(y) \ dy$$

$$= \sqrt{x} \int_{x}^{x+\delta} \chi_{1}(y)y^{\nu+1} \ dy \int_{0}^{\lambda} J_{\nu}(xu)J_{\nu}(yu)u \ du$$

$$= \sqrt{x} \chi_{1}(x+\delta) \int_{\xi}^{x+\delta} y^{\nu+1} \ dy \int_{0}^{\lambda} J_{\nu}(xu)J_{\nu}(yu)u \ du$$

$$= \sqrt{x} \chi_{1}(x+\delta) \int_{0}^{\lambda} J_{\nu}(xu)[(x+\delta)^{\nu+1}J_{\nu+1}\{(x+\delta)u\} - \xi^{\nu+1}J_{\nu+1}(\xi u)] \ du$$

where $x < \xi < x + \delta$. Now for $x \ge x_0 > 0$, $y \ge x_0$,

$$\int_{0}^{\lambda} J_{\nu}(xu) J_{\nu+1}(yu) du
= O(1) + \frac{2}{\pi} \int_{1}^{\lambda} \cos(x - \frac{1}{2}\nu\pi - \frac{1}{4}\pi) \sin(x - \frac{1}{2}\nu\pi - \frac{1}{4}\pi) \frac{du}{\sqrt{(xy)}u} + \int_{1}^{\lambda} O\left(\frac{1}{u^{i}}\right) du
= O(1)$$

for all λ . The contribution of the χ_1 term is therefore $O(\epsilon)$. Similarly so is that of the χ_2 term.

The theorem therefore follows on choosing first δ sufficiently small, and then, having fixed δ , λ sufficiently large.

8.19. Formulae derived from Hankel's theorem. Simple pairs of Hankel transforms may be derived from (7.4.6), (7.11.6), (7.11.8)-(7.11.15), and (7.11.17). Another elegant pair is

$$\begin{split} 2^{3\nu-1}\Gamma(\nu+\tfrac{1}{2})\sqrt{\pi} & \left(\frac{pq}{x}\right)^{\nu} \sqrt{x} \, J_{\nu}(px) J_{\nu}(qx), \\ \{x^2-(p-q)^2\}^{\nu-\frac{1}{2}} \{(p+q)^2-x^2\}^{\nu-\frac{1}{2}} x^{\frac{1}{2}-\nu} \quad (|p-q| < x < p+q), \\ 0 \quad \text{elsewhere.} \qquad (8.19.1)^{-\frac{1}{2}} (p+q)^{-\frac{1}{2}} x^{\frac{1}{2}-\nu} \quad (|p-q| < x < p+q), \end{split}$$

To prove this,† put $\nu = -\frac{1}{2}$, $\mu = \lambda - \frac{1}{2}$, x = 1, $y = \sin^2\theta$ in † This is Sonine's proof referred to by Watson, § 11.41.

(7.14.9). We obtain

$$a^{\lambda-\frac{1}{4}}\frac{J_{\lambda}\left\{\sqrt{(a^2+b^2)}\right\}}{(a^2+b^2)^{\frac{1}{4}\lambda}} = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\pi} \sin^{\lambda+\frac{1}{4}\theta} J_{\lambda-\frac{1}{4}}(a\sin\theta) e^{ib\cos\theta} d\theta. \tag{8.19.2}$$

Putting $a = q \sin \phi$, $b = p - q \cos \phi$, and multiplying by $\sin^{\lambda+1} \phi$ and integrating, we obtain

$$\int_{0}^{\pi} \frac{J_{\lambda}\{\sqrt{(p^{2}+q^{2}-2pq\cos\phi)}\}\sin^{2}\lambda\phi\,d\phi}{(p^{2}+q^{2}-2pq\cos\phi)^{\frac{1}{2}\lambda}}\sin^{2}\lambda\phi\,d\phi$$

$$= \frac{q^{\frac{1}{2}-\lambda}}{\sqrt{(2\pi)}} \int_{0}^{\pi} \sin^{\lambda+\frac{1}{2}}\phi\,d\phi \int_{0}^{\pi} \sin^{\lambda+\frac{1}{2}}\theta\,J_{\lambda-\frac{1}{2}}(q\sin\theta\sin\phi)\,e^{i\cos\theta(p-q\cos\phi)}\,d\theta$$

$$= \frac{q^{\frac{1}{2}-\lambda}}{\sqrt{(2\pi)}} \int_{0}^{\pi} \sin^{\lambda+\frac{1}{2}}\theta\,e^{ip\cos\theta}\,d\theta \int_{0}^{\pi} \sin^{\lambda+\frac{1}{2}}\phi\,J_{\lambda-\frac{1}{2}}(q\sin\theta\sin\phi)\,e^{-iq\cos\theta\cos\phi}\,d\phi$$

$$= q^{-\lambda} \int_{0}^{\pi} \sin^{\lambda+\frac{1}{2}}\theta\,e^{ip\cos\theta}\sin^{\lambda-\frac{1}{2}}\theta\,J_{\lambda}(q)\,d\theta$$

$$= 2^{\lambda}\Gamma(\lambda+\frac{1}{2})\sqrt{\pi}(pq)^{-\lambda}J_{\lambda}(p)J_{\lambda}(q). \tag{8.19.3}$$

The result stated follows on taking $p^2+q^2-2pq\cos\phi=\xi$ as a new variable.

The reciprocal formulat is

$$\int_{0}^{\infty} x^{1-\nu} J_{\nu}(px) J_{\nu}(qx) J_{\nu}(ux) dx$$

$$= \frac{\{u^{2} - (p-q)^{2}\}^{\nu-\frac{1}{2}} \{(p+q)^{2} - u^{2}\}^{\nu-\frac{1}{2}}}{2^{3\nu-1}\sqrt{\pi} \Gamma(\nu+\frac{1}{2})(pqu)^{\nu}}$$
(8.19.4)

if |p-q| < u < p+q, and 0 otherwise.

Still other results can now be deduced from the Parseval formula.‡ For example, (7.11.12) gives the Hankel transforms

$$x^{\lambda+\nu+\frac{1}{2}}K_{\lambda}(ax), \qquad 2^{\lambda+\nu}a^{\lambda}x^{\nu+\frac{1}{2}}\Gamma(\lambda+\nu+1)(a^2+x^2)^{-\lambda-\nu-1}, \quad (8.19.5)$$

and we deduce

$$\begin{split} &\int\limits_0^\infty x^{\lambda+\mu+2\nu+1}K_{\lambda}(ax)K_{\mu}(bx)\;dx\\ &=\;2^{\lambda+\mu+2\nu}a^{\lambda}b^{\mu}\Gamma(\lambda+\nu+1)\Gamma(\mu+\nu+1)\int\limits_0^\infty \frac{x^{2\nu+1}\;dx}{(a^2+x^2)^{\lambda+\nu+1}(b^2+x^2)^{\mu+\nu+1}}. \end{split}$$

† See Watson, § 13.46; Nicholson (1).

1 See Titchmarsh (11).

We can put $x = b \tan \theta$ and expand the integral in powers of $(b^2-a^2)/b^2$; if a = b, the result (with $\nu = \frac{1}{2}\rho - \frac{1}{2}\lambda - \frac{1}{2}\mu - 1$) is

$$\int_{0}^{\infty} K_{\lambda}(ax)K_{\mu}(ax)x^{\rho-1} dx$$

$$= \frac{2^{\rho-3}}{\Gamma(\rho)a^{\rho}} \Gamma\left(\frac{\rho+\lambda-\mu}{2}\right) \Gamma\left(\frac{\rho-\lambda+\mu}{2}\right) \Gamma\left(\frac{\rho-\lambda-\mu}{2}\right) \Gamma\left(\frac{\rho+\lambda+\mu}{2}\right) \quad (8.19.6)$$

Similarly, from the Hankel transforms (see (7.11.14))

$$\frac{x^{\nu+\frac{1}{2}}\sqrt{\pi}}{2^{3\nu}a^{2\nu}\Gamma(\nu+\frac{1}{2})}J_{\nu}(ax)K_{\nu}(ax), \qquad \frac{x^{\nu+\frac{1}{2}}}{(x^4+4a^4)^{\nu+\frac{1}{2}}}, \qquad (8.19.7)$$

we deduce

$$\int_{0}^{\infty} J_{\nu}^{2}(ax) K_{\nu}^{2}(ax) x^{2\nu+1} dx = \frac{2^{\nu-3}}{a^{2\nu+2}} \frac{\Gamma(\frac{1}{2}\nu + \frac{1}{2})\Gamma(\nu + \frac{1}{2})\Gamma(\frac{3}{2}\nu + \frac{1}{2})}{\sqrt{\pi} \Gamma(\nu+1)}, \quad (8.19.8)$$

and from (8.19.1), with p = q, we deduce

$$\int_{0}^{\infty} J_{\nu}^{4}(px)x^{1-2\nu} dx = \frac{p^{2\nu-2}}{2} \frac{\Gamma(\nu)\Gamma(2\nu)}{\pi\{\Gamma(\nu+\frac{1}{2})\}^{2}\Gamma(3\nu)}.$$
 (8.19.9)

SELF-RECIPROCAL FUNCTIONS

9.1. Formalities. In previous chapters we have noticed a number of functions which are their own Fourier cosine or sine transforms, i.e. functions f(x) such that

$$f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) \cos xy \ dy \tag{9.1.1}$$

or

$$f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) \sin xy \ dy. \tag{9.1.2}$$

The simplest solutions of (9.1.1) are

$$x^{-\frac{1}{2}}, e^{-\frac{1}{2}x^2}, \operatorname{sech}\{x\sqrt{(\frac{1}{2}\pi)}\}.$$

Similar solutions of (9.1.2) are

$$x^{-\frac{1}{2}}, \qquad xe^{-\frac{1}{4}x^{2}}, \qquad \frac{1}{e^{x\sqrt{(2\pi)}}-1} - \frac{1}{x\sqrt{(2\pi)}}.$$

There are also functions which are their own Hankel transforms of order ν , i.e. solutions of

$$f(x) = \int_{0}^{\infty} f(y) \sqrt{(xy)} J_{\nu}(xy) dy. \qquad (9.1.3)$$

Solutions of (9.1.1), (9.1.2), (9.1.3) will be called R_c , R_s , R_{ν} respectively.

Other functions are 'skew-reciprocal', i.e. satisfy (9.1.1), (9.1.2), or (9.1.3) with the sign of the right-hand side changed. Such functions will be called $-R_c$, $-R_s$, $-R_{\nu}$ respectively.

The first problem of this chapter is to determine all self-reciprocal functions; or (since complete generality is hardly attainable) all such functions of certain classes, such as the class L^2 . We shall take (9.1.1) as the typical case.

In a sense, there is an immediate solution. If g(x) belongs to L^2 , then $g(x) + G_c(x)$ is also a function of L^2 , and is plainly self-reciprocal. Also any self-reciprocal f(x) may be expressed as

$$\frac{1}{2}f(x) + \frac{1}{2}f(x) = \frac{1}{2}f(x) + \frac{1}{2}F_c(x)$$
.

The formula $g(x)+G_c(x)$ therefore gives the complete solution of the problem.

On the other hand, it is obvious that none of the examples quoted above have been obtained in this way, and the solution does not enable us to decide (unless by actual verification) whether a given f(x) is self-reciprocal. To determine whether f(x) is of the form $g+G_c$ is to solve another integral equation, viz.

$$f(x) = g(x) + \sqrt{\binom{2}{\pi}} \int_{0}^{\infty} g(y) \cos xy \ dy. \tag{9.1.4}$$

We shall consider such equations in §11.15; but it is easier to attack (9.1.1) directly.

Let $\mathfrak{F}(s)$ be the Mellin transform of f(x). Then (9.1.1) gives formally

$$\mathfrak{F}(s) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} x^{s-1} dx \int_{0}^{\infty} f(y) \cos xy dy$$
$$= \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) dy \int_{0}^{\infty} x^{s-1} \cos xy dx$$
$$= \sqrt{\left(\frac{2}{\pi}\right)} \Gamma(s) \cos \frac{1}{2} s \pi \int_{0}^{\infty} f(y) y^{-s} dy,$$

i.e. $\mathfrak{F}(s)$ satisfies the functional equation

$$\mathfrak{F}(s) = \sqrt{\left(\frac{2}{\pi}\right)} \Gamma(s) \cos \frac{1}{2} s \pi \, \mathfrak{F}(1-s). \tag{9.1.5}$$

If now we write $\mathfrak{F}(s) = 2^{\frac{1}{2}s}\Gamma(\frac{1}{2}s)\psi(s)$, then

$$\psi(s) = \psi(1-s), \tag{9.1.6}$$

and, by Mellin's formula,

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} 2^{\frac{1}{2}s} \Gamma(\frac{1}{2}s) \psi(s) x^{-s} ds.$$
 (9.1.7)

We may therefore expect (9.1.7), where $\psi(s)$ satisfies (9.1.6), i.e. is an even function of $s-\frac{1}{2}$, to be a general formula for functions of R_c .

The simplest example is

$$\psi(s) = 1, \quad f(x) = 2e^{-\frac{1}{2}x^2}.$$

We can deal with (9.1.2), or generally (9.1.3), in a similar way. If f(x) satisfies (9.1.3), then

$$\mathfrak{F}(s) = \int_{-\infty}^{\infty} y^{\frac{1}{2}} f(y) \ dy \int_{-\infty}^{\infty} x^{s-\frac{1}{2}} J_{\nu}(xy) \ dx = \frac{2^{s-\frac{1}{2}} \Gamma(\frac{1}{2}\nu + \frac{1}{2}s + \frac{1}{4})}{\Gamma(\frac{1}{2}\nu - \frac{1}{2}s + \frac{3}{4})} \int_{-\infty}^{\infty} f(y) y^{-s} \ dy,$$

$$\mathfrak{F}(s) = 2^{s-\frac{1}{4}} \frac{\Gamma(\frac{1}{2}\nu + \frac{1}{2}s + \frac{1}{4})}{\Gamma(\frac{1}{2}\nu - \frac{1}{2}s + \frac{3}{4})} \mathfrak{F}(1-s). \tag{9.1.8}$$

Putting $\mathfrak{F}(s) = 2^{\frac{1}{2}s}\Gamma(\frac{1}{2}\nu + \frac{1}{2}s + \frac{1}{4})\psi(s)$, we obtain as a general solution of (9.1.3)

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} 2^{\frac{1}{6}s} \Gamma(\frac{1}{2}\nu + \frac{1}{2}s + \frac{1}{4}) \psi(s) x^{-s} ds, \qquad (9.1.9)$$

where $\psi(s)$ again satisfies (9.1.6).

9.2. Another formal solution of the problem is obtained by considering $_{\infty}$

 $\chi(s) = \int_{0}^{\infty} f(x)e^{-\frac{1}{2}s^{2}x^{2}} dx. \qquad (9.2.1)$

Then (9.1.1) gives

$$\chi(s) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} e^{-\frac{1}{8}s^{2}x^{2}} dx \int_{0}^{\infty} f(y)\cos xy dy$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) dy \int_{0}^{\infty} e^{-\frac{1}{8}s^{2}x^{2}}\cos xy dx$$

$$= \frac{1}{8} \int_{0}^{\infty} f(y)e^{-\frac{1}{8}\nu^{2}/8^{2}} dy,$$

i.e.

$$\chi(s) = \frac{1}{s} \chi \left(\frac{1}{s}\right). \tag{9.2.2}$$

If
$$\mu(s) = s^{\dagger} \chi(s^{\dagger})$$
, then $\mu(s) = s^{\dagger} \chi(s^{\dagger})$

 $\mu(s) = \mu\left(\frac{1}{s}\right). \tag{9.2.3}$

We may write (9.2.1) as

$$\chi(s^{\frac{1}{2}}) = \int_{0}^{\infty} (2u)^{-\frac{1}{2}} f\{(2u)^{\frac{1}{2}}\} e^{-su} du,$$

and the reciprocal formula is

$$(2u)^{-\frac{1}{2}} f\{(2u)^{\frac{1}{2}}\} = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \chi(s^{\frac{1}{2}}) e^{su} ds,$$

$$f(x) = \frac{x}{2\pi i} \int_{c+i\infty}^{c+i\infty} \mu(s) e^{\frac{1}{2}x^{\frac{1}{2}}s} s^{-\frac{1}{2}} ds. \qquad (9.2.4)$$

or

Hence (9.2.4), where $\mu(s)$ satisfies (9.2.3), may be expected to be R_c .

The simplest example is

$$\mu(s) = 1, f(x) = \frac{2^{t}}{\Gamma(\frac{1}{4})} x^{-\frac{1}{4}}.$$
st
$$\chi(s) = \int_{0}^{\infty} f(x) x^{\nu + \frac{1}{4}} e^{-\frac{1}{4} s^{2} x^{4}} dx. (9.2.5)$$

For general ν , let

Then (9.1.3) gives

$$\chi(s) = \int_{0}^{\infty} e^{-\frac{1}{8}s^{2}x^{2}} x^{\nu+\frac{1}{8}} dx \int_{0}^{\infty} f(y) \sqrt{(xy)} J_{\nu}(xy) dy$$

$$= \int_{0}^{\infty} f(y) \sqrt{y} dy \int_{0}^{\infty} e^{-\frac{1}{8}s^{2}x^{2}} x^{\nu+1} J_{\nu}(xy) dx$$

$$= \int_{0}^{\infty} f(y) \sqrt{y} \frac{y^{\nu}}{s^{2\nu+2}} e^{-\nu^{2}/2s^{2}} dy$$

$$= \frac{1}{s^{2\nu+2}} \chi \left(\frac{1}{s}\right).$$

If $\mu(s) = s^{\frac{1}{4}\nu + \frac{1}{4}}\chi(s^{\frac{1}{4}})$, then $\mu(s)$ again satisfies (9.2.3), and

$$f(x) = \frac{x^{\frac{1}{4}-\nu}}{2\pi i} \int_{c-i\infty}^{c+i\infty} \mu(s)e^{\frac{1}{4}x^{2}s}s^{-\frac{1}{4}\nu-\frac{1}{4}} ds.$$
 (9.2.6)

9.3. Still other formulae of the same kind can be obtained by replacing the e^{-ix^2} of the above example by other functions which are self-reciprocal, and which also are the kernels of a general transformation. We may take, for example, the function

$$x^{\frac{1}{2}}J_{-\frac{1}{4}}(\frac{1}{2}x^2).$$

Proceeding as before, we obtain

$$\chi(s) = \int_{0}^{\infty} (sx)^{\frac{1}{2}} J_{-\frac{1}{2}}(\frac{1}{2}s^{2}x^{2}) f(x) \ dx = \frac{1}{s} \chi(\frac{1}{s}),$$

and f(x) can be expressed in terms of $\chi(s)$ by Hankel's theorem. This transformation has been studied in detail by Mehrotra (8).

9.4. Functions of L^2 . We shall now justify the above arguments under a variety of conditions. The simplest conditions are provided by the L^2 -theory of Mellin transforms.

THEOREM 136.† A necessary and sufficient condition that a function f(x) of $L^2(0,\infty)$ should be its own cosine transform is that it should

† Hardy and Titchmarsh (4). Proof suggested by Miss Busbridge.

be of the form (9.1.7), where $c = \frac{1}{2}$, the integral is a mean-square integral, $\widetilde{\chi}(\frac{1}{2} + it) = 2^{\frac{1}{2} + \frac{1}{2}it} \Gamma(\frac{1}{2} + \frac{1}{2}it) \psi(\frac{1}{2} + it) \qquad (9.4.1)$

belongs to $L^2(-\infty,\infty)$, and (9.1.6) holds, i.e. $\psi(\frac{1}{2}+it)$ is an even function of t.

In view of the L^2 theory of Mellin transforms, all that we have to prove is the equivalence of the self-reciprocal property of f(x) to (9.1.6).

Now f(y) and $\sin xy/y$ belong to $L^2(0,\infty)$, and the Mellin transform of the latter is

$$\Gamma(\frac{1}{2}+it)\cos{\frac{1}{2}\pi(\frac{1}{2}+it)}x^{\frac{1}{2}-it}/(\frac{1}{2}-it).$$

Hence

$$\sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) \frac{\sin xy}{y} dy$$

$$= \frac{1}{2\pi} \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{\infty} \mathfrak{F}(\frac{1}{2} - it) \Gamma(\frac{1}{2} + it) \cos \frac{1}{2\pi} \pi(\frac{1}{2} + it) \frac{x^{\frac{1}{2} - it}}{\frac{1}{2} - it} dt.$$
Also
$$\int_{0}^{x} f(y) dy = \frac{1}{2\pi} \int_{-\infty}^{\infty} \mathfrak{F}(\frac{1}{2} + it) \frac{x^{\frac{1}{2} - it}}{\frac{1}{2} - it} dt.$$

If f is self-reciprocal, the right-hand sides must be equal; since each integrand belongs to $L(-\infty,\infty)$, they must be equal almost everywhere (Theorem 32, p. 47). Hence $\psi(\frac{1}{2}+it)$ is even. Conversely, if $\psi(\frac{1}{2}+it)$ is even, the right-hand sides are equal; hence so are the left-hand sides, and so f is self-reciprocal.

9.5. Functions of L^p .

THEOREM 137. If a function f(x) of $L^p(0,\infty)$, where 1 , is its own cosine transform, then it is of the form (9.1.7), where

$$\mathfrak{F}(s) = 2^{\frac{1}{2}s}\Gamma(\frac{1}{2}s)\psi(s)$$

is an analytic function which (i) is regular in the strip

$$\frac{1}{p'} < \sigma < \frac{1}{p} \quad \left(p' = \frac{p}{p-1} \right),$$
 (9.5.1)

(ii) tends to 0 uniformly as $s \to \infty$ inside any interior strip, and (iii) satisfies (9.1.5); the integral in (9.1.7) is a mean-square integral along any line of the strip (9.5.1).

This is a one-sided theorem, with conditions which are necessary only and not sufficient.

If f(x) belongs to L^p , its cosine transform belongs to $L^{p'}$, so that f(x) here belongs to both L^p and $L^{p'}$, and therefore to all intermediate classes L^q . In particular it belongs to L^2 , and so satisfies the conditions of Theorem 136.

The function $\mathfrak{F}(s)$ reduces to the $\mathfrak{F}(\frac{1}{2}+it)$ of Theorem 136 when $s=\frac{1}{2}+it$, but $\mathfrak{F}(s)$ is now an analytic function, regular in the strip (9.5.1). In fact

$$\int_{1}^{\infty} |f(x)| x^{\sigma-1} dx \\ \leq \left(\int_{0}^{1} |f|^{p'} dx \right)^{1/p'} \left(\int_{0}^{1} x^{p(\sigma-1)} dx \right)^{1/p} + \left(\int_{1}^{\infty} |f|^{p} dx \right)^{1/p} \left(\int_{1}^{\infty} x^{p'(\sigma-1)} dx \right)^{1/p'},$$

and these integrals converge for the values of σ stated. It follows in the usual manner that $\mathfrak{F}(s)$ is regular in the strip, and bounded in any interior strip.

Again, we can write $\mathfrak{F}(s)$ in the form

$$\mathfrak{F}(s) = \left(\int_0^{\delta} + \int_{\delta}^{\Delta} + \int_{\Delta}^{\infty}\right) f(x) x^{\sigma-1} x^{it} dx = \mathfrak{F}_1(s) + \mathfrak{F}_2(s) + \mathfrak{F}_3(s).$$

Let $\eta > 0$, and

$$1/p' + \eta \leqslant \sigma \leqslant 1/p - \eta. \tag{9.5.2}$$

Then

$$|\mathfrak{F}_1(s)| \leqslant \left(\int\limits_0^\delta |f|^{p'}\,dx\right)^{1/p'} \left(\int\limits_0^\delta x^{p(\sigma-1)}\,dx\right)^{1/p} \,=\, O(\delta^\eta)$$

as $\delta \to 0$, and we can therefore choose δ so that $|\mathfrak{F}_1| < \epsilon$ for all s in (9.5.2). Similarly, we can choose Δ so that $|\mathfrak{F}_3| < \epsilon$. When δ and Δ are fixed, $\mathfrak{F}_2 \to 0$ uniformly as $s \to \infty$ in (9.5.2). Hence $\mathfrak{F} \to 0$ uniformly in (9.5.2).

It follows from Theorem 136 that $\mathfrak{F}(s)$ satisfies (9.1.5) on $s = \frac{1}{2} + it$, and so throughout (9.5.1).

Thus $\mathfrak{F}(s)$ possesses the properties stated in the theorem, and it remains only to prove (9.1.7). This is true for $c=\frac{1}{2}$, by Theorem 136, so that it is sufficient to prove that the value of the integral is independent of c; and this follows by the argument of § 5.4.

9.6. The previous theorem is a one-sided theorem, and we cannot, in view of the asymmetry of the theory of transforms about the number 2, expect in this case a theorem as satisfactory as Theorem 136. There is, however, a very similar class of functions for which we can obtain a complete solution.

We shall say that f(x) belongs to $L_p^*(0,\infty)$, where $1 , if <math>x^{\alpha}f(x)$ belongs to $L^2(0,\infty)$ for

$$-\alpha_0 = -rac{1}{p} + rac{1}{2} < lpha < rac{1}{p} - rac{1}{2} = lpha_0.$$

It is plain that f(x) then belongs to $L^q(0,1)$ for $q\leqslant 2$. Suppose now that p< q< 2. Then we can choose $\alpha<\alpha_0$ so that $2q\alpha>2-q$; and then

$$\int\limits_{1}^{\infty}|f|^{q}~dx\leqslant\left(\int\limits_{1}^{\infty}x^{2lpha}|f|^{2}~dx
ight)^{q/2}\!\!\left(\int\limits_{1}^{\infty}x^{-2qlpha|(2-q)}~dx
ight)^{(2-q)/2}<\infty$$
 ,

so that f(x) belongs to L^q . If also f(x) is its own cosine transform, it belongs to $L^{q'}$, so that a self-reciprocal f(x) of L_p^* belongs to all L-classes between L^p and $L^{p'}$, though not usually to either of these.

The class of self-reciprocal functions of L_p^* is thus in this respect a little wider than the class of those of L^p . In other respects it is narrower. Suppose, for example, that h(x) is defined by

$$h(x) = 2^{-n}$$
 $(n!-1 \le x \le n!+1, n = 2, 3,...),$

and h(x) = 0 elsewhere. Then h(x) belongs to L^r for every positive r, but to no L_p^* , since $\sum 2^{-rn}$ is convergent, but $\sum (n!)^{2\alpha}2^{-2n}$ divergent for every positive α . The cosine transform of h(x) is

$$H_c(x) = \sqrt{\left(\frac{2}{\pi}\right)} \sum_{x = 1}^{n+1} \cos xy \ dy = 2\sqrt{\left(\frac{2}{\pi}\right)} \frac{\sin x}{x} \sum_{x = 1}^{n+1} \frac{\cos n! \ x}{2^n},$$

which is continuous and $O(x^{-1})$ at infinity, so that $H_c(x)$ belongs to L^r for r > 1, and to L_p^* for $1 . Thus <math>h(x) + H_c(x)$ is a self-reciprocal function which belongs to L^r for all r > 1, but to no L_p^* .

THEOREM 138. A necessary and sufficient condition that a function f(x) of $L_p^*(0,\infty)$ should be its own cosine transform is that it should be of the form (9.1.7), where $\mathfrak{F}(s)$ satisfies the conditions (i), (ii), (iii) of Theorem 137, and (iv) belongs to $L^2(-\infty,\infty)$, qua function of t, for all σ of (9.5.1).

(i) The condition is necessary. Since f(x) belongs to L^r for p < r < p', we have only to show that $\mathfrak{F}(s)$ satisfies condition (iv). This results immediately from the theory of transforms, since

$$\mathfrak{F}(s) = \int_0^\infty f(x) x^{\sigma - \frac{1}{2}} x^{-\frac{1}{2} + it} dx,$$

and $x^{\sigma-\frac{1}{2}}f(x)$ belongs to L^2 if $|\sigma-\frac{1}{2}| < \alpha_0$, i.e. if $1/p' < \sigma < 1/p$.

(ii) The condition is sufficient. Since $\mathfrak{F}(s)$ belongs to L^2 on the

line s = c + it, the integral (9.1.7) exists as a mean-square integral for all c in question, and as before its value is independent of c. It therefore defines a function f(x) independent of c. Since

$$x^{c-\frac{1}{2}}f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \mathfrak{F}(c+it)x^{-\frac{1}{2}-it} dt$$

for $|c-\frac{1}{2}| < \alpha_0$, and the right-hand side belongs to L^2 for every such c, f(x) belongs to L_p^* . Finally, by Theorem 136, f(x) is self-reciprocal.

9.7. Analytic functions. We shall say that f(x) belongs to $A(\alpha, a)$, where $0 < \alpha \le \pi$, $a < \frac{1}{2}$, if (i) it is an analytic function of $x = re^{i\theta}$ regular in the angle defined by r > 0, $|\theta| < \alpha$, and (ii) it is $O(|x|^{-a-\epsilon})$ for small x, and $O(|x|^{a-1+\epsilon})$ for large x, for every positive ϵ and uniformly in any angle $|\theta| \le \alpha - \eta < \alpha$.

THEOREM 139. A necessary and sufficient condition that a function f(x) of $A(\alpha, a)$ should be its own cosine transform is that it should be of the form (9.1.7), where $\psi(s)$ is regular, and satisfies (9.1.6), in the strip

$$a < \sigma < 1 - a; \tag{9.7.1}$$

$$\psi(s) = O(e^{(\frac{1}{4}\pi - \alpha + \eta)|t|}) \tag{9.7.2}$$

for every positive η and uniformly in any strip interior to (9.7.1); and c is any value of σ in (9.7.1).

(i) The condition is necessary. The integral

$$\int_{0}^{\infty} f(x)x^{s-1} dx \tag{9.7.3}$$

is absolutely convergent for $a < \sigma < 1-a$, so that $\mathfrak{F}(s)$ is regular in (9.7.1). Also, f(x) belongs to L^2 , and it follows from Theorem 136 that $\mathfrak{F}(s)$ satisfies (9.1.5) on $\sigma = \frac{1}{2}$, and therefore throughout (9.7.1), or, what is the same thing, $\psi(s)$ satisfies (9.1.6).

Also, f(x) satisfies the conditions of Theorem 31, with $\beta = \alpha$ and b = 1-a. Since

$$\psi(s) = \Re(s) 2^{-\frac{1}{2}s} / \Gamma(\frac{1}{2}s), \qquad |\Gamma(\frac{1}{2}s)| \sim Ce^{-\frac{1}{2}\pi|l|} |\frac{1}{2}t|^{\frac{1}{2}\sigma-\frac{1}{2}},$$

it follows that $\psi(s)$ satisfies the conditions stated.

- (ii) The condition is also sufficient because Theorems 31 and 136, on which we have based the argument, are reversible.
- 9.8. More general conditions. The next theorem is of a more general kind; here f(x) does not necessarily belong to any L- or $A(\alpha, a)$ -class.

THEOREM 140. Let f(x) be integrable over any finite interval; let

$$F_c(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^{-\infty} f(y) \cos xy \ dy \tag{9.8.1}$$

exist for every x; and let

$$\mathfrak{F}(s) = \int\limits_{-\infty}^{-\infty} f(x)x^{s-1} dx \qquad (9.8.2)$$

exist for $|\sigma - \frac{1}{2}| < \alpha$, where $\alpha > 0$. Then a necessary and sufficient condition that $F_c(x) = f(x)$ almost everywhere is that $\mathfrak{F}(s)$ should satisfy (9.1.5) for $|\sigma - \frac{1}{2}| < \alpha$.

Let
$$\frac{1}{2} < \beta < \frac{1}{2} + \alpha$$
, and

$$g(x) = \int_{1}^{x} f(\xi) \xi^{\beta-1} d\xi.$$

Then g(x) is bounded. Hence if $\sigma < \beta$

$$\int_{1}^{X} f(x)x^{s-1} dx = \int_{1}^{X} g'(x)x^{s-\beta} dx$$

$$= g(X)X^{s-\beta} - (s-\beta) \int_{1}^{X} g(x)x^{s-\beta-1} dx$$

$$= O(1) + O\left(|s| \int_{1}^{X} x^{\sigma-\beta-1} dx\right) = O(|t|)$$

for all X. Similarly,

$$\int_{1/X}^{1} f(x)x^{s-1} dx = O(|t|)$$

for $\sigma > \frac{1}{2} - \alpha$. Thus

$$\int_{1/X}^X f(x)x^{s-1} dx = O(|t|)$$

in any strip interior to $|\sigma - \frac{1}{2}| < \alpha$.

It follows by dominated convergence that

$$\begin{split} \frac{1}{2\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} \mathfrak{F}(s) \frac{x^{3-s}}{(s-1)(s-2)(s-3)} \, ds \\ &= \frac{1}{2\pi i} \int_{-\infty}^{\infty} f(\xi) \frac{x^3}{\xi} \, d\xi \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} \frac{(x/\xi)^{-s}}{(s-1)(s-2)(s-3)} \, ds \\ &= -\frac{1}{2} \int_{0}^{x} f(\xi)(x-\xi)^2 \, d\xi, \end{split}$$

and that

$$\begin{split} \frac{1}{2\pi i} \int\limits_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} \frac{\Gamma(s)\cos\frac{1}{2}s\pi}{(s-1)(s-2)(s-3)} \, & \mathfrak{F}(1-s)x^{3-s} \, ds \\ &= \frac{1}{2\pi i} \int\limits_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} \Gamma(s-3)\cos\frac{1}{2}s\pi \, \, \mathfrak{F}(1-s)x^{3-s} \, ds \\ &= \frac{x^3}{2\pi i} \int\limits_{-0}^{+\infty} f(\xi) \, d\xi \int\limits_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} \Gamma(s-3)\cos\frac{1}{2}s\pi \, (x\xi)^{-s} \, ds \\ &= \frac{1}{2\pi i} \int\limits_{-0}^{+\infty} \frac{f(\xi)}{\xi^3} \, d\xi \int\limits_{-2\frac{1}{2}-i\infty}^{-2\frac{1}{2}+i\infty} \Gamma(s)\sin\frac{1}{2}s\pi \, (x\xi)^{-s} \, ds \\ &= \int\limits_{-\infty}^{+\infty} f(\xi) \frac{\sin x\xi - x\xi}{\xi^3} \, d\xi. \end{split}$$

If $\mathfrak{F}(s)$ satisfies (9.1.5), it follows that

$$\frac{1}{2}\int_{0}^{x}f(\xi)(x-\xi)^{2}\,d\xi=\sqrt{\left(\frac{2}{\pi}\right)}\int_{0}^{\infty}f(\xi)\frac{x\xi-\sin x\xi}{\xi^{3}}\,d\xi.$$

But, as in the proof of Theorem 118, (9.8.1) gives

$$\int_{0}^{x} du \int_{0}^{u} F_{c}(v) dv = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(\xi) \frac{1-\cos x\xi}{\xi^{2}} d\xi.$$

We may integrate over (0, x) by uniform convergence; hence

$$\frac{1}{2}\int_{0}^{x}f(\xi)(x-\xi)^{2}\,d\xi=\int_{0}^{x}d\xi\int_{0}^{\xi}du\int_{0}^{u}F_{c}(v)\,dv,$$

and, differentiating three times, it follows that $f(x) = F_c(x)$ almost everywhere.

Conversely, if $f(x) = F_c(x)$ almost everywhere, the argument shows that

$$\frac{1}{2\pi i} \int_{-i\infty}^{\frac{1}{2}+i\infty} \frac{\Re(s) - \sqrt{(2/\pi)\Gamma(s)\cos\frac{1}{2}s\pi} \Re(1-s)}{(s-1)(s-2)(s-3)} x^{-s} ds = 0$$

for all values of x. Since the integrand belongs to L, it must be null (Theorem 32, second part). Hence $\mathfrak{F}(s)$ satisfies (9.1.5) on $\sigma = \frac{1}{2}$, and so throughout its region of regularity.

If the conditions are satisfied, f(x) is represented by (9.1.7) in the (C, 1) sense, by Theorem 32.

9.9. A general theorem. Even if (as in the case $f(x) = x^{-1}$) the integral (9.8.2) does not exist for any s, it is possible to obtain a result corresponding to (9.1.5), but involving the functions $\mathfrak{F}_+(s)$, $\mathfrak{F}_-(s)$. We shall deduce our result from the following theorem, which is frequently useful.

Theorem 141. Let $\phi(w)$ be regular in the strip $a_1 \leq v \leq a_2$, and let $\phi(u+iv)$ be $L(-\infty,\infty)$ (or $L^2(-\infty,\infty)$), and tend to 0 as $u \to \pm \infty$, for v in the above interval. Let $\psi(w)$ have similar properties in $b_1 \leq v \leq b_2$, where $b_2 < a_1$. Let

$$\int_{ia-\infty}^{ia+\infty} \phi(w)e^{-ixw} dw + \int_{ib-\infty}^{ib+\infty} \psi(w)e^{-ixw} dw = 0$$
 (9.9.1)

for all x, where $a_1 < a < a_2$, $b_1 < b < b_2$. Then ϕ and ψ are regular for $b_1 < v < a_2$, their sum is 0 in this strip, and they tend to 0, as $u \to \pm \infty$, uniformly in any interior strip.

Consider first the L case. Multiply (9.9.1) by $e^{ix\zeta}$, where $\zeta = \xi + i\eta$, $a < \eta < a_2$, and integrate with respect to x over (0, ∞). We can invert the order of integration by absolute convergence, and we obtain

$$\int_{ia-\infty}^{ia+\infty} \frac{\phi(w)}{w-\zeta} dw + \int_{ib-\infty}^{ib+\infty} \frac{\psi(w)}{w-\zeta} dw = 0 \quad (a < \eta < a_2). \quad (9.9.2)$$

Now move the line of integration of the ϕ -integral to $v = a_2$. We obtain

$$\int_{ia_{2}-\infty}^{ia_{2}+\infty} \frac{\phi(w)}{w-\zeta} dw + \int_{ib-\infty}^{ib+\infty} \frac{\psi(w)}{w-\zeta} dw = -2\pi i \phi(\zeta) \quad (a < \eta < a_{2}).$$

$$(9.9.3)$$

The left-hand side is now regular for $b < \eta < a_2$. It therefore provides the analytic continuation of $-2\pi i\phi(\zeta)$ throughout this strip.

Similarly, multiplying (9.9.1) by $e^{ix\xi}$, where $b_1 < \eta < b$, and integrating over $(-\infty, 0)$, we obtain (9.9.2) with $b_1 < \eta < b$. Moving the line of integration of the ψ -integral to $\eta = b_1$, we obtain

$$\int_{ia-\infty}^{ia+\infty} \frac{\phi(w)}{w-\zeta} dw + \int_{ib_1-\infty}^{ib_1+\infty} \frac{\psi(w)}{w-\zeta} dw = 2\pi i \psi(\zeta) \quad (b_1 < \eta < b). \tag{9.9.4}$$

This provides the analytic continuation of $2\pi i \psi(\zeta)$ over $b_1 < \eta < a$.

If $b < \eta < a$, the left-hand sides of (9.9.3) and (9.9.4) are equal, by an obvious application of Cauchy's theorem. Hence

$$\phi(\zeta) = -\psi(\zeta)$$

in the strip.

Also

$$\begin{split} &\left|\int_{ia_1-\infty}^{ia_1+\infty} \frac{\phi(w)}{w-\zeta} dw\right| \\ &\leqslant \frac{1}{|a_2-\eta|} \left(\int_{-\infty}^{-U} + \int_{U}^{\infty} \right) |\phi(u+ia_2)| \ du + \frac{1}{|U-\xi|} \int_{-U}^{U} |\phi(u+ia_2)| \ du, \end{split}$$

which tends to 0 as $\xi \to \pm \infty$, by choosing first U and then ξ . Similarly, for the other term on the left of (9.9.3). Hence $\phi(\zeta) \to 0$ uniformly in the strip.

In the L^2 case (9.9.2) follows from (9.9.1) by the L^2 case of Parseval's formula, and the argument then proceeds as before; in the last part we put

$$\begin{split} \left| \int_{ia_{1}-\infty}^{ia_{1}+\infty} \frac{\phi(w)}{w-\zeta} \, dw \right| & \leqslant \left\{ \int_{-\infty}^{-U} \frac{du}{|w-\zeta|^{2}} \int_{-\infty}^{-U} |\phi(u+ia_{2})|^{2} \, du \right\}^{\frac{1}{2}} + \\ & + \left\{ \int_{U}^{\infty} \frac{du}{|w-\zeta|^{2}} \int_{U}^{\infty} |\phi(u+ia_{2})|^{2} \, du \right\}^{\frac{1}{2}} + \\ & + \frac{1}{|U-\xi|} \int_{-U}^{U} |\phi(u+ia_{2})| \, du, \end{split}$$

and again choose first U and then ξ .

9.10. Application. Theorem 142. Let f(x) be integrable over every finite interval, and tend to 0 at infinity, and let

$$f(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{+\infty} f(y)\cos xy \ dy \qquad (9.10.1)$$

for all but a finite set of values of x. Then almost everywhere

$$f(x) = \frac{1}{2\pi i} \int_{\alpha - i\infty}^{\alpha + i\infty} \mathfrak{F}_{+}(s) x^{-s} ds + \frac{1}{2\pi i} \int_{-\beta - i\infty}^{\beta + i\infty} \mathfrak{F}_{-}(s) x^{-s} ds$$

$$(\beta < 0, \alpha > 1), \quad (9.10.2)$$

where the integrals are (C, 1) integrals; $\mathfrak{F}_{-}(s)$ is regular for $\sigma < 0$, and $\mathfrak{F}_{+}(s)$ for $\sigma > 1$; and the functions

$$\mathfrak{F}_{+}(s) - \sqrt{\frac{2}{\pi}} \mathfrak{F}_{-}(1-s)\Gamma(s)\cos\frac{1}{2}s\pi,$$

$$\mathfrak{F}_{-}(s) - \sqrt{\frac{2}{\pi}} \mathfrak{F}_{+}(1-s)\Gamma(s)\cos\frac{1}{2}s\pi$$
(9.10.3)

are regular for $0 \le \sigma \le 1$, except possibly for simple poles at s = 0; and their sum is 0 in the strip.

Let
$$f_1(x) = \int_0^x f(u) \ du, \quad f_2(x) = \int_0^x f_1(u) \ du,$$

etc. Then, as in the proof of Theorem 113, (9.10.1) gives

$$f_{2}(x) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) \frac{1 - \cos xy}{y^{2}} dy. \tag{9.10.4}$$
Let $\mathcal{F}_{-}(s) = \int_{0}^{\infty} f(x) x^{s-1} dx$, $\mathcal{F}_{+}(s) = \int_{0}^{1} f(x) x^{s-1} dx$.

These are clearly regular for $\sigma < 0$ and $\sigma > 1$ respectively; and (9.10.2) holds by the (C, 1) analogue of Theorem 24, for Mellin integrals. Let

$$\begin{split} \Phi(x) &= \frac{1}{2\pi i} \int_{-\frac{1}{2} - i\infty}^{-\frac{1}{2} + i\infty} \mathfrak{F}_{-}(s) \frac{x^{2-s}}{(s-1)(s-2)} \, ds \, - \\ &- \frac{1}{2\pi i} \int_{-\frac{1}{2} - i\infty}^{-\frac{1}{2} + i\infty} \sqrt{\left(\frac{2}{\pi}\right)} \mathfrak{F}_{+}(1-s) \Gamma(s) \cos \frac{1}{2} s \pi \frac{x^{2-s}}{(s-1)(s-2)} \, ds \, + \\ &+ \frac{1}{2\pi i} \int_{\frac{1}{2} - i\infty}^{-\frac{1}{2} + i\infty} \mathfrak{F}_{+}(s) \frac{x^{2-s}}{(s-1)(s-2)} \, ds \, - \\ &- \frac{1}{2\pi i} \int_{\frac{1}{2} - i\infty}^{-\frac{1}{2} + i\infty} \sqrt{\left(\frac{2}{\pi}\right)} \mathfrak{F}_{-}(1-s) \Gamma(s) \cos \frac{1}{2} s \pi \frac{x^{2-s}}{(s-1)(s-2)} \, ds \\ &= \Phi_{1}(x) + \Phi_{2}(x) + \Phi_{3}(x) + \Phi_{4}(x). \end{split}$$

We may insert the above integrals for $\mathfrak{F}_{-}(s)$, etc., and invert, by absolute convergence. We obtain

$$\Phi_{1}(x) = \frac{x}{2\pi i} \int_{1}^{\infty} f(y) \, dy \int_{-\frac{1}{2} - i\infty}^{-\frac{1}{2} + i\infty} \frac{(x/y)^{1-s}}{(s-1)(s-2)} \, ds$$

$$= \int_{1}^{x} f(y)(x-y) \, dy \quad (x > 1), \qquad 0 \quad (x < 1),$$

4362

$$\Phi_{3}(x) = \frac{x}{2\pi i} \int_{0}^{1} f(y) \, dy \int_{1-i\infty}^{1+i\infty} \frac{(x/y)^{1-s}}{(s-1)(s-2)} \, ds$$

$$= \begin{cases} -\int_{0}^{1} y f(y) \, dy & (x > 1), \\ -\int_{0}^{x} y f(y) \, dy - x \int_{x}^{1} f(y) \, dy & (x < 1). \end{cases}$$

Hence $\Phi_1(x) + \Phi_3(x) = \int_{0}^{x} f(y)(x-y) \, dy - x \int_{0}^{1} f(y) \, dy$

for all x > 0. Also

$$\begin{split} \Phi_2(x) &= -\sqrt{\left(\frac{2}{\pi}\right)} \frac{x^2}{2\pi i} \int_0^1 f(y) \, dy \int_{-\frac{1}{2} - i\infty}^{-\frac{1}{2} + i\infty} \Gamma(s-2) \cos \frac{1}{2} s \pi(xy)^{-s} \, ds \\ &= \sqrt{\left(\frac{2}{\pi}\right)} \int_0^1 f(y) \frac{\cos xy - 1 - \frac{1}{2} x^2 y^2}{y^2} \, dy, \\ \Phi_4(x) &= -\sqrt{\left(\frac{2}{\pi}\right)} \frac{x^2}{2\pi i} \int_1^{\infty} f(y) \, dy \int_{\frac{1}{2} - i\infty}^{\frac{1}{2} + i\infty} \Gamma(s-2) \cos \frac{1}{2} s \pi(xy)^{-s} \, ds \\ &= \sqrt{\left(\frac{2}{\pi}\right)} \int_0^{\infty} f(y) \frac{\cos xy - 1}{y^2} \, dy. \end{split}$$

Altogether

$$\Phi(x) = f_2(x) - \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty f(y) \frac{1 - \cos xy}{y^2} dy + ax + bx^2,$$

where a and b are constants. Hence, by (9.10.4),

$$\int_{-1-i\infty}^{-\frac{1}{4}+i\infty} \left\{ \mathfrak{F}_{-}(s) - \sqrt{\left(\frac{2}{\pi}\right)} \mathfrak{F}_{+}(1-s) \Gamma(s) \cos \frac{1}{2} s \pi + a + 2b \frac{s-1}{s} \right\} \frac{x^{2-s} ds}{(s-1)(s-2)} + \int_{\frac{1}{4}-i\infty}^{1+i\infty} \left\{ \mathfrak{F}_{+}(s) - \sqrt{\left(\frac{2}{\pi}\right)} \mathfrak{F}_{-}(1-s) \Gamma(s) \cos \frac{1}{2} s \pi - a - 2b \frac{s-1}{s} \right\} \frac{x^{2-s} ds}{(s-1)(s-2)} = 0$$

for every positive x. The result therefore follows from Theorem 141, with an obvious change of variable.

A similar result holds if f(x) does not tend to 0, but (9.10.1) holds

everywhere. The proof is similar, but with an extra factor s-3 in the denominator.

9.11. The Second Solution. A similar set of theorems might be constructed for the second solution obtained in § 9.2. It will perhaps be sufficient to prove one of them, and we take the case of analytic functions. We shall say that f(x) belongs to $A^*(\omega, a)$, where $0 < \omega \leq \frac{1}{4}\pi$, $0 < \alpha < \frac{1}{2}$, if (i) it is an analytic function of $x = re^{i\theta}$ regular in the angle A^* defined by r > 0, $|\theta| < \omega$, and (ii) it is $O(|x|^{-a-\frac{1}{2}-\delta})$ for small x, and $O(|x|^{a-\frac{1}{2}+\delta})$ for large x, for every positive δ and uniformly in any angle $|\theta| \leq \omega - \eta < \omega$.

THEOREM 143. A necessary and sufficient condition that a function f(x) of $A^*(\omega, a)$ should be its own cosine transform is that it should be of the form (9.2.4), where c is any positive number, the integral is the limit of an integral over (c-iT, c+iT), and $\mu(s)$ has the properties

- (i) $\mu(s) = \mu(\rho e^{i\phi})$ is an analytic function of s, regular in the angle $B(\omega, a)$ defined by $\rho > 0$, $|\phi| < \frac{1}{2}\pi + 2\omega$;
- (ii) $\mu(s)$ is $O(|s|^{-\frac{1}{4}a-\delta})$ for small s, and $O(|s|^{\frac{1}{4}a+\delta})$ for large s, for every positive δ and uniformly in any angle $|\phi| \leq \frac{1}{2}\pi + 2\omega \zeta < \frac{1}{2}\pi + 2\omega$;
 - (iii) $\mu(s)$ satisfies (9.2.3) in $B(\omega, a)$.

The conditions of regularity and order follow from Theorem 31.

It is then only a question of proving that (9.2.3) is necessary and sufficient for f(x) to be self-reciprocal.

Integrating (9.2.4), we get

$$\begin{split} f_1(x) &= \frac{1}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} \mu(s) s^{-\frac{1}{4}} \frac{e^{\frac{1}{4}x^4s} - 1}{s} \, ds \\ &= \frac{1}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} \mu(s) s^{-\frac{1}{4}} e^{\frac{1}{4}x^4s} \, ds, \end{split}$$

the other term being zero, by an obvious application of Cauchy's theorem. Again,

$$\frac{\sin xy}{y} = \frac{\sqrt{\pi}}{4\pi i} \int_{\cdot}^{c+i\infty} e^{x^*s-\nu^*/(4s)}s^{-\frac{1}{2}} ds,$$

and hence

$$\int_{0}^{\infty} f(y) \frac{\sin xy}{y} \, dy = \frac{\sqrt{\pi}}{4\pi i} \int_{c-i\infty}^{c+i\infty} e^{x^{1}s} s^{-1} \, ds \int_{0}^{\infty} f(y) e^{-y^{2}/(4s)} \, dy$$

$$\begin{split} &=\frac{\sqrt{\pi}}{4\pi i}\int\limits_{c-i\infty}^{c+i\infty}e^{x^{\mathbf{1}}s}s^{-\frac{1}{2}}\chi\{(2s)^{-\frac{1}{2}}\}\;ds\\ &=\frac{\sqrt{(2\pi)}}{4\pi i}\int\limits_{c-i\infty}^{c+i\infty}e^{\frac{i}{2}x^{\mathbf{1}}s}s^{-\frac{1}{2}}\mu\left(\frac{1}{s}\right)\;ds, \end{split}$$

the inversion being justified by absolute convergence. It follows that

$$f_1(x) - \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty f(y) \frac{\sin xy}{y} dy = \frac{1}{2\pi i} \int_{s-i\infty}^{c+i\infty} e^{\frac{i}{4}x^2s} e^{-\frac{i}{4}} \left\{ \mu(s) - \mu\left(\frac{1}{s}\right) \right\} ds,$$

and hence that the condition (9.2.3) is both necessary and sufficient.

9.12. Examples.

(1) If $\psi(s) = 1$ in (9.1.7), then

$$f(x) = 2e^{-\frac{1}{4}x^2}, \qquad f(x) = 2^{\frac{1}{4}-\frac{1}{4}\nu}x^{\nu+\frac{1}{4}}e^{-\frac{1}{4}x^2}$$

in the cosine and general cases respectively. The conditions of Theorem 139 (and a fortiori those of the less special theorems) are satisfied.

If $\psi(s) = P(\frac{1}{2} - s)$, where P(u) is an even polynomial, or an even integral function of order less than 1, we find that

$$f(x) = 2\sum_{n=0}^{\infty} \frac{(-\frac{1}{2}x^2)^n}{n!} P(2n + \frac{1}{2})$$

is its own cosine transform. If P(u) is a polynomial, $f(x) = e^{-\frac{1}{4}x^2}Q(x^2)$, where Q(u) is a polynomial.

(2) Sonine's polynomials $T_{\nu}^{n}(x)$ are defined by

$$T_{\nu}^{n}(x) = \sum_{r=0}^{n} \frac{(-1)^{r} x^{n-r}}{r! (n-r)! \Gamma(n+\nu-r+1)}.$$

$$f(x) = x^{\nu} T^{n}(x) e^{-\frac{1}{2}x}$$

If

$$f(x) = x^{\nu} T^n_{\nu}(\dot{x}) e^{-\frac{1}{2}x},$$

then

$$\phi(s) = \int_{0}^{\infty} f(x)e^{-sx} dx = \sum_{r=0}^{n} \frac{(-1)^{r}}{r! (n-r)! \Gamma(n+\nu-r+1)} \int_{0}^{\infty} x^{n-r+\nu}e^{-(s+\frac{1}{2})x} dx$$

$$= \sum_{r=0}^{n} \frac{(-1)^{r}}{r! (n-r)!} (s+\frac{1}{2})^{r-n-\nu-1} = \frac{1}{n!} \frac{(\frac{1}{2}-s)^{n}}{(\frac{1}{2}+s)^{n+\nu+1}}.$$
If
$$g(x) = x^{\nu+\frac{1}{2}}e^{-\frac{1}{2}x^{2}}T_{\nu}^{n}(x^{2}),$$

then

$$\begin{split} \mu(s) &= s^{\frac{1}{2}\nu + \frac{1}{6}} \int_{0}^{\infty} g(x) x^{\nu + \frac{1}{6}} e^{-\frac{1}{6}x^{8}s} \, dx \\ &= \frac{1}{2} s^{\frac{1}{6}\nu + \frac{1}{6}} \int_{0}^{\infty} f(\xi) e^{-\frac{1}{6}\xi s} \, d\xi = \frac{2^{\nu}}{n!} (s^{\frac{1}{6}} + s^{-\frac{1}{6}})^{-\nu - 1} \left(\frac{1 - s}{1 + s}\right)^{n}. \\ \mu(s) &= (-1)^{n} \mu\left(\frac{1}{s}\right), \end{split}$$

Hence

and g(x) is $\pm R_{\nu}$ according as n is even or odd.

The parabolic cylinder functions $D_n(x)$ may be defined for integral n by

 $T_{-\frac{1}{2}}^{n}(x^{2}) = \frac{2^{n}}{(2n)!\sqrt{\pi}}e^{\frac{1}{2}x^{2}}D_{2n}(x\sqrt{2})$

and

$$xT_{\frac{1}{2}}^{n}(x^{2}) = \frac{2^{n+\frac{1}{2}}}{(2n+1)!\sqrt{\pi}}e^{\frac{1}{2}x^{2}}D_{2n+1}(x\sqrt{2}).$$

Thus $D_{2n}(x\sqrt{2})$ is $\pm R_c$ according as n is even or odd, and $D_{2n+1}(x\sqrt{2})$ is $\pm R_s$ according as n is even or odd. This is equivalent to the self-reciprocal property of Hermite polynomials (§ 3.8). In fact it is easily verified that

$$H_{2n}(x) = (2n)! \sqrt{\pi} T_{-1}^n(x^2), \qquad H_{2n+1}(x) = (2n+1)! \sqrt{\pi} x T_{-1}^n(x^2).$$

In the case $\nu = \frac{1}{2}$ Parseval's formula gives

$$\begin{split} \int\limits_0^\infty \{f(x)\}^2 e^{-sx} \, dx &= \frac{1}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} \phi(w) \phi(s-w) \, dw \\ &= \frac{1}{2\pi i (n!)^2} \int\limits_{-i\infty}^{c+i\infty} \frac{(\frac{1}{2}-w)^n (\frac{1}{2}-s+w)^n}{(\frac{1}{2}+w)^{n+\frac{1}{2}} (\frac{1}{2}+s-w)^{n+\frac{1}{2}}} \, dw. \end{split}$$

Denoting this by $\omega(s)$, and putting $w = w' + \frac{1}{2}s$, we obtain

$$\omega(s) = \frac{1}{2\pi i (n!)^2} \int_{c'-im}^{c'+i\infty} \frac{\{(\frac{1}{2} - \frac{1}{2}s)^2 - w'^2\}^n}{\{(\frac{1}{2} + \frac{1}{2}s)^2 - w'^2\}^{n+\frac{1}{2}}} dw'$$

Changing s into 1/s, and then putting w' = w''/s, it follows that $\omega(1/s) = s^2\omega(s)$.

If now $\mu_1(s) = s \int\limits_0^\infty x^{-\frac{1}{2}} D_{2n+1}^2(x) . x^{\frac{1}{2}} e^{-\frac{1}{2}x^{\frac{1}{2}s}} dx,$ we have $\mu_1(s) = \frac{\{(2n+1)!\}^2 \pi}{2^{\frac{1}{2}n+1}} s \omega(s),$

† A. Milne (1), B. M. Wilson (1).

and hence $\mu_1(s)$ satisfies (9.2.3). Hence †

$$x^{-\frac{1}{2}}D_{2n+1}^2(x)$$

is R_1 .

For negative (not necessarily integral) n we have

$$D_n(x) = \frac{e^{-\frac{1}{4}x^{\bullet}}}{\Gamma(-n)} \int_0^{\infty} e^{-tx-\frac{1}{2}t^{\bullet}} t^{-n-1} dt.$$

It follows that

$$\int\limits_{0}^{\infty} x^{s-1} e^{\frac{i}{4}x^{s}} D_{n}(x) \ dx = \frac{\Gamma(s) \Gamma(-\frac{1}{2}n - \frac{1}{2}s)}{2^{\frac{i}{4}n + \frac{1}{4}s + 1} \Gamma(-n)}.$$

It is then easily verified from the formulae of § 9.1 that;

$$x^{\nu+\frac{1}{2}}e^{\frac{1}{4}x^2}D_{-2\nu-3}(x), \qquad x^{\nu-\frac{1}{2}}e^{\frac{1}{4}x^2}D_{-2\nu}(x)$$

are R_{ν} .

(3) If $f(x) = \operatorname{sech} x \sqrt{(\frac{1}{2}\pi)}$, we find that

$$\mathfrak{F}(s) = 2\left(\frac{2}{\pi}\right)^{4s}\Gamma(s)L(s),$$

where

$$L(s) = \frac{1}{1^s} - \frac{1}{3^s} + \frac{1}{5^s} - \dots, (9.12.1)$$

and $\mathfrak{F}(s)$ satisfies (9.1.5) by the functional equation for L(s) (§ 2.11). This is another example of Theorem 139.

If
$$f(x) = \frac{1}{e^{x\sqrt{(2\pi)}} - 1} - \frac{1}{x\sqrt{(2\pi)}}$$

we find that $\mathfrak{F}(s) = (2\pi)^{-\frac{1}{2}s}\Gamma(s)\zeta(s)$. Taking $\nu = \frac{1}{2}$ in the formulae at the end of § 9.1, we obtain

$$\psi(s) = \frac{1}{2}\pi^{-\frac{1}{2}s}\Gamma(\frac{1}{2}s)\zeta(s) = \frac{\xi(s)}{s(s-1)},$$

where $\xi(s)$ is Riemann's ξ -function. This is an example of the analogue of Theorem 139 for sine transforms.

Other self-reciprocal functions are associated in a similar way with the functional equations of other Dirichlet's L-functions. For example the functions

$$\frac{\cosh(\frac{1}{2}x\sqrt{\pi})}{\cosh(x\sqrt{\pi})}, \frac{1}{1+2\cosh\{x\sqrt{(\frac{2}{3}\pi)}\}}$$

† Mitra (1), Watson (4).

1 Varma (1).

are R_c ; they are associated with

$$egin{align} L_1(s) &= \sum_{r=0}^{\infty} (-1)^r \Big\{ rac{1}{(4r+1)^s} + rac{1}{(4r+3)^s} \Big\}, \ L_2(s) &= \sum_{r=0}^{\infty} \Big\{ rac{1}{(3r+1)^s} - rac{1}{(3r+2)^s} \Big\} \end{split}$$

respectively. And

$$\frac{\sinh(\frac{1}{2}x\sqrt{\pi})}{\cosh(x\sqrt{\pi})}, \frac{\sinh\{x\sqrt{(\frac{1}{6}\pi)}\}}{2\cosh\{x\sqrt{(\frac{2}{3}\pi)}\}-1}$$

are R_s ; they are associated with

$$\begin{split} L_3(s) &= \sum_{r=0}^{\infty} (-1)^r \Big\{ \frac{1}{(4r+1)^s} - \frac{1}{(4r+3)^s} \Big\}, \\ L_4(s) &= \sum_{r=0}^{\infty} (-1)^r \Big\{ \frac{1}{(6r+1)^s} - \frac{1}{(6r+5)^s} \Big\}. \end{split}$$

respectively.

(4) It is easily verified from (7.1.8), (7.1.9) that

$$f(x) = \cos(\frac{1}{2}x^2 - \frac{1}{8}\pi)$$

is its own cosine transform. This does not belong to any *L*-class, but is an example of Theorem 140. The integral (9.8.2) exists for $0 < \sigma < 2$, and $\mathfrak{F}(s) = 2^{\frac{1}{8}s-1}\Gamma(\frac{1}{8}s)\cos\frac{1}{4}\pi(s-\frac{1}{8})$ satisfies (9.1.5).

(5) The function $f(x) = x^{-\frac{1}{2}}$ is its own cosine transform, and is an example of Theorem 142. Here

$$\mathfrak{F}_+(s) = \frac{1}{s-\frac{1}{s}}, \qquad \mathfrak{F}_-(s) = -\frac{1}{s-\frac{1}{s}},$$

and

$$\mathfrak{F}_+(s) - \sqrt{\left(\frac{2}{\pi}\right)} \mathfrak{F}_-(1-s) \Gamma(s) \cos \frac{1}{2} s \pi = \frac{1-\sqrt{(2/\pi)} \Gamma(s) \cos \frac{1}{2} s \pi}{s-\frac{1}{2}},$$

which has a simple pole at s = 0, and is regular for $\sigma > 0$.

A more general example of the same kind is

$$f(x) = 2^{\frac{1}{2}a} \Gamma(\frac{1}{2}a) x^{-a} + 2^{\frac{1}{2} - \frac{1}{2}a} \Gamma(\frac{1}{2} - \frac{1}{2}a) x^{a-1} \quad (0 < a < 1).$$

(6) It follows from (7.5.6) and (7.5.7) that

$$\frac{\cos\frac{1}{2}x^2 + \sin\frac{1}{2}x^2}{\cosh\{x\sqrt{(\frac{1}{2}\pi)}\}}$$

is R_c , and from (7.5.10) that

$$\frac{\sin\frac{1}{2}x^2}{\sinh\{x\sqrt{(\frac{1}{2}\pi)}\}}$$

is R_s . These are examples of Theorems 139 and 143, but $\psi(s)$ and $\mu(s)$ do not seem to be particularly simple.

(7) Taking
$$f(x) = x^{\frac{1}{2}} J_{\frac{1}{2}\nu}(\frac{1}{2}x^{2}),$$
 we find
$$\psi(s) = \frac{\pi^{\frac{1}{2}} 2^{-\frac{1}{2}\nu - \frac{3}{2}}}{\Gamma(\frac{1}{2}\nu + \frac{5}{8} + \frac{1}{4}s)\Gamma(\frac{1}{4}\nu + \frac{7}{8} - \frac{1}{4}s)},$$

so that $\psi(s) = \psi(1-s)$. By Theorem 140 f(x) is its own Hankel transform of order ν . In this case, however, f(x) behaves like $x^{\nu+\frac{1}{2}}$ for small x, and $x^{-\frac{1}{2}}$ for large x, and does not belong to L^2 , or to any L^p for which $p \leq 2$.

In the case $\nu = -\frac{1}{2}$ the resulting formula is

$$\sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \sqrt{y} J_{-\frac{1}{2}(\frac{1}{2}y^{2})} \cos xy \ dy = \sqrt{x} J_{-\frac{1}{2}(\frac{1}{2}x^{2})}.$$

Differentiating twice with respect to x, we obtain formally

$$\sqrt{\left(\frac{2}{\pi}\right)}\int_{0}^{\infty}y^{\frac{1}{2}}J_{-\frac{1}{2}}(\frac{1}{2}y^{2})\cos xy\ dy = x^{\frac{1}{2}}J_{-\frac{1}{2}}(\frac{1}{2}x^{2}).$$

This is true if the integral is taken in the (C, 1) sense, but it does not come under any of our general theorems. A discussion of functions self-reciprocal in this sense is given by Mehrotra (8).

(8) Let
$$f(x) = x^{\frac{1}{4}-\nu}(x^2-b^2)^{\frac{1}{4}(\nu-1)}J_{\frac{1}{4}\nu-\frac{1}{4}}\{b\sqrt{(x^2-b^2)}\}$$
 $(x>b>0),$ Then
$$0 \qquad (0< x< b).$$

$$\begin{split} \mathfrak{F}(s) &= \int\limits_{b}^{\infty} x^{s-\frac{1}{2}-\nu} (x^2-b^2)^{\frac{1}{2}(\nu-1)} J_{\frac{1}{2}\nu-\frac{1}{2}} \{b\sqrt{(x^2-b^2)}\} \\ &= b^{s-\frac{1}{2}\nu} \int\limits_{0}^{\infty} (1+u^2)^{\frac{1}{2}(s-\nu-\frac{1}{2})} u^{\frac{1}{2}\nu+\frac{1}{2}} J_{\frac{1}{2}\nu-\frac{1}{2}} (b^2u) \ du \\ &= \frac{b^{\frac{1}{2}\nu-\frac{1}{2}} K_{\frac{1}{2}(\frac{1}{2}-s)} (b^2)}{2^{\frac{1}{2}\nu-\frac{1}{2}s-\frac{1}{2}} \Gamma(\frac{3}{2}+\frac{1}{2}\nu-\frac{1}{2}s)} \end{split}$$

by (7.11.6); and, $K_{\mu}(x)$ being an even function of μ , (9.1.8) follows. Here f(x) is $O\{(x-b)^{\frac{1}{2}\nu-\frac{1}{2}}\}$ near x=b, and $O(x^{-\frac{1}{2}\nu-\frac{1}{2}})$ at infinity; it belongs to L^2 if $\nu>0$, and to L^p and L^{*}_p if $\nu>|1-2/p|$. If $-1<\nu\leqslant 0$ it is a case of Theorem 140.

(9) The function

$$f(x) = x^{\nu+\frac{1}{2}}(x^2+a^2)^{-\frac{1}{2}\nu-\frac{1}{2}}K_{\frac{1}{2}\nu+\frac{1}{2}}\{a\sqrt{(x^2+a^2)}\} \quad (a>0)$$

is R_{ν} (see Watson § 13.47 (2)). By Watson § 13.47 (6) we find

$$\psi(s) = 2^{-\frac{1}{2}} \left(\frac{2}{a}\right)^{\frac{1}{4}\nu + \frac{1}{2}} K_{\frac{1}{2}(s-\frac{1}{2})}(a^2).$$

9.13. Lattice-point formulae. There are some interesting examples of self-reciprocal functions in the analytic theory of numbers. \dagger Let r(n) denote the number of representations of n as a sum of two squares, and let

$$\bar{P}(x) = \sum_{0 \le n \le x} r(n) - \pi x,$$
 (9.13.1)

the dash implying the insertion of a factor $\frac{1}{2}$ in the last term of the sum when x is an integer. Then

$$f(x) = x^{-\frac{1}{4}} \left\{ \overline{P} \left(\frac{x^2}{2\pi} \right) - 1 \right\}$$
 (9.13.2)

belongs to R_2 .

It is clear from the definition that $f(x) = O(x^{\frac{1}{2}})$ as $x \to 0$. That $\overline{P}(x) = O(x^{\frac{1}{2}})$ as $x \to \infty$ is comparatively trivial, and in fact it is known; that $\overline{P}(x) = O(x^{\frac{1}{2}})$. Hence $f(x) = O(x^{-\frac{1}{2}})$ as $x \to \infty$. Hence f(x) is L^2 , and is L^p and L^*_p if $p > \frac{6}{5}$.

We have

$$\mathfrak{F}(s) = \int_{0}^{\infty} \left\{ \bar{P}\left(\frac{x^{2}}{2\pi}\right) - 1 \right\} x^{s-\frac{1}{2}} dx = \frac{1}{2} (2\pi)^{\frac{1}{2}s-\frac{3}{2}} \int_{0}^{\infty} \{\bar{P}(x) - 1\} x^{\frac{1}{2}s-\frac{3}{2}} dx,$$

the integral being convergent, and $\mathfrak{F}(s)$ analytic, for $-\frac{1}{2} < \sigma < \frac{5}{6}$. The last integral is

$$\int_{0}^{\infty} \left\{ \sum_{1 \leq n \leq x} r(n) - \pi x \right\} x^{\frac{1}{8}s - \frac{7}{4}} dx = -\frac{\pi}{\frac{1}{2}s + \frac{1}{4}} + \int_{1}^{\infty} \left\{ \sum_{1 \leq n \leq x} r(n) - \pi x \right\} x^{\frac{1}{8}s - \frac{7}{4}} dx,$$

and this provides the analytic continuation of $\mathfrak{F}(s)$ to $\sigma < -\frac{1}{2}$, there being a simple pole at $s = -\frac{1}{2}$. If $\sigma < -\frac{1}{2}$

and
$$\int_{1}^{\infty} (-\pi x) x^{\frac{1}{8}s-\frac{7}{4}} dx = \frac{\pi}{\frac{1}{2}s+\frac{7}{4}},$$

$$\int_{1}^{\infty} \sum_{1 \leq n \leq x} r(n) x^{\frac{1}{8}s-\frac{7}{4}} dx = \sum_{\nu=1}^{\infty} \int_{\nu}^{\nu+1} \{r(1) + ... + r(\nu)\} x^{\frac{1}{4}s-\frac{7}{4}} dx$$

$$= \sum_{\nu=1}^{\infty} \{r(1) + ... + r(\nu)\} \frac{(\nu+1)^{\frac{1}{4}s-\frac{7}{4}} - \nu^{\frac{1}{4}s-\frac{7}{4}}}{\frac{1}{2}s-\frac{3}{4}}.$$

$$= -\frac{1}{\frac{1}{2}s-\frac{3}{4}} \sum_{\nu=1}^{\infty} r(\nu) \nu^{\frac{1}{4}s-\frac{7}{4}} = \frac{Z(\frac{3}{4}-\frac{1}{2}s)}{\frac{3}{4}-\frac{1}{2}s},$$
where
$$Z(s) = \sum_{n=1}^{\infty} \frac{r(n)}{n^s} = 4\zeta(s)L(s),$$

† See Hardy (17), Hardy and Titchmarsh (4), 212-3.

1 Landau, Vorlesungen über Zahlentheorie, 2, 204-8.

L(s) being (9.12.1). Hence

$$\mathfrak{F}(s) = \frac{1}{2} (2\pi)^{\frac{1}{8}s - \frac{3}{4}} \frac{Z(\frac{3}{4} - \frac{1}{2}s)}{\frac{3}{4} - \frac{1}{2}s}.$$

Now

$$Z(1-s) = \pi^{1-2s} \frac{\Gamma(s)}{\Gamma(1-s)} Z(s)$$

by the functional equations for $\zeta(s)$ and L(s) (pp. 65-6), or independently.[†] Hence $\mathfrak{F}(s)$ satisfies (9.1.8) with $\nu=2$, and so f(x) is R_2 .

It follows from the analogue for J_2 -transforms of Theorem 136 (p. 248) that (9.1.3), with $\nu = 2$, holds in the mean-square sense. In fact it holds in the ordinary sense, i.e.

$$\frac{1}{\xi^{\frac{1}{2}}}\left\{\overline{P}\left(\frac{\xi^{2}}{2\pi}\right)-1\right\} = \int_{0}^{\infty} \frac{1}{y^{\frac{1}{2}}}\left\{\overline{P}\left(\frac{y^{2}}{2\pi}\right)-1\right\}(\xi y)^{\frac{1}{2}}J_{2}(\xi y) dy \qquad (9.13.3)$$

for every positive ξ . To prove this we require the analogue for J_2 -transforms of Theorem 58 (p. 83). It is easy to obtain this analogue by adapting the argument of § 8.18. There we justified the inversion of

$$\int\limits_0^\lambda J_\nu(xu)\sqrt{(xu)}\;du\int\limits_0^\infty J_\nu(uy)\sqrt{(uy)}f(y)\;dy$$

by the uniform convergence of the inner integral. If f(x) is L^2 , the inversion is justified by the mean convergence of the inner integral, and the result is a case of Parseval's formula for Hankel transforms. Having obtained the inversion, the rest of the proof is the same as that given in § 8.18.

Putting $y = t/\xi$, $\xi = \sqrt{(2\pi x)}$, (9.13.3) gives

$$\frac{\bar{P}(x)-1}{2\pi x} = \int_{0}^{\infty} \left\{ \bar{P}\left(\frac{t^{2}}{4\pi^{2}x}\right)-1\right\} \frac{J_{2}(t)}{t} dt = \int_{0}^{\infty} \bar{P}\left(\frac{t^{2}}{4\pi^{2}x}\right) \frac{J_{2}(t)}{t} dt - \frac{1}{2}.$$

Now

$$\int_{0}^{2\pi\sqrt{((N+1)x)}} \overline{P}\left(\frac{t^{2}}{4\pi^{2}x}\right) \frac{J_{2}(t)}{t} dt = \sum_{n=0}^{N} \int_{2\pi\sqrt{(nx)}}^{2\pi\sqrt{((n+1)x)}} \left\{r(0) + ... + r(n) - \frac{t^{2}}{4\pi x}\right\} \frac{J_{2}(t)}{t} dt$$

$$= \sum_{n=0}^{N} \left\{r(0) + ... + r(n)\right\} \left\{\frac{J_{1}\left\{2\pi\sqrt{(nx)}\right\}}{2\pi\sqrt{(nx)}} - \frac{J_{1}\left[2\pi\sqrt{((n+1)x)}\right]}{2\pi\sqrt{((n+1)x)}}\right\} - \frac{1}{4\pi x} \int_{0}^{2\pi\sqrt{(N+1)x}} t J_{2}(t) dt$$

† See e.g. Mordell (2), Potter (1).

$$= \sum_{n=0}^{N} r(n) \frac{J_1\{2\pi\sqrt{(nx)}\}}{2\pi\sqrt{(nx)}} - \{r(0) + \dots + r(N)\} \frac{J_1[2\pi\sqrt{(N+1)x}\}]}{2\pi\sqrt{(N+1)x}} - \frac{1}{4\pi x} \left\{ -2\pi\sqrt{(N+1)x} J_1[2\pi\sqrt{(N+1)x}] + 2\int_{0}^{+\infty} J_1(t) dt + o(1) \right\}$$

for $N \to \infty$, x fixed. The terms involving $J_1[2\pi\sqrt{(N+1)x}]$ are

$$-\{r(0)+\ldots+r(N)-(N+1)\pi\}\frac{J_1[2\pi\sqrt{\{(N+1)x\}}]}{2\pi\sqrt{\{(N+1)x\}}}=O(N^{-1}),$$

and, since $\int\limits_0^\infty J_1(t)\ dt=1$, we obtain finally \dagger

$$\bar{P}(x) = \sqrt{x} \sum_{n=1}^{\infty} \frac{r(n)}{\sqrt{n}} J_1\{2\pi\sqrt{(nx)}\}.$$
 (9.13.4)

It has been proved by Walfisz[‡] and Oppenheim|| that, if $r_p(n)$ is the number of representations of n as a sum of p squares, and

$$egin{aligned} & \overline{P}_p(x) = \sum_{0\leqslant n\leqslant x}' r_p(n) - rac{\pi^{rac{1}{p}}}{\Gamma(1+rac{1}{2}p)} x^{rac{1}{p}}, \ & \overline{P}_p(x) = x^{rac{1}{p}} \sum_{n \stackrel{1}{1}p}' rac{r_p(n)}{n^{rac{1}{p}}} J_{rac{1}{2}p} \{2\pi \sqrt{(nx)}\}, \end{aligned}$$

then

the series being summable by Cesàro's means of sufficiently high

order. It follows that $x^{-1-ip}\left\{\overline{P}_p\left(\frac{x^2}{2\pi}\right)-1\right\}$

belongs to $R_{1+\frac{1}{2}p}$. If we take p=3, and use Walfisz's result $\overline{P}_3(x) = O(x^{\frac{1}{2}+\epsilon})$, we find that f(x) falls under the obvious extension of Theorem 136. This is not true for any larger p.

If we take p = 1, we find that

$$f(x) = \frac{1}{x} \left(\frac{x}{\sqrt{(2\pi)}} - \left[\frac{x}{\sqrt{(2\pi)}} \right] \right),$$

where [u] is the integral part of u, belongs to R_{i} , as may be verified directly.

9.14. Formulae connecting different classes of self-reciprocal functions.†† The simplest such formula is given by

Rule 1. If f(x) is its own cosine (sine) transform, then

$$g(x) = \int_{0}^{\infty} f(t)e^{-xt} dt$$
 (9.14.1)

is its own sine (cosine) transform.

† See Hardy and Landau (1), Hardy (15). ‡ Walfisz (1), (2). || Oppenheim (1) †† Phillips (1), Hardy and Titchmarsh (6), Mehrotra (1), (6).

Supposing, for example, that f(x) is its own cosine transform, we have

$$\sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} g(t) \sin xt \ dt = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \sin xt \ dt \int_{0}^{\infty} f(y) e^{-ty} \ dy$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) \ dy \int_{0}^{\infty} e^{-ty} \sin xt \ dt$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(y) \frac{x}{x^{2} + y^{2}} \ dy.$$

Now $\sqrt{\left(\frac{2}{\pi}\right)} \frac{x}{x^2 + y^2}$ is the cosine transform of e^{-xy} , and f(y) is its own cosine transform. Hence Parseval's theorem for cosine transforms gives

 $\sqrt{\left(\frac{2}{\pi}\right)}\int\limits_0^\infty f(y)\frac{x}{x^2+y^2}\,dy=\int\limits_0^\infty f(y)e^{-xy}\,dy=g(x),$

and the rule follows. The example with $f(t) = \operatorname{sech}\{t\sqrt{(\frac{1}{2}\pi)}\}$, g(x) R_s , has been observed by various authors. Rule 1 is a particular case of

Rule 2. If f(x) belongs to R_{μ} , and

$$k(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} 2^s \Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s) \Gamma(\frac{1}{4} + \frac{1}{2}\nu + \frac{1}{2}s) \chi(s) x^{-s} ds,$$
(9.14.2)

where

$$\chi(s) = \chi(1-s),$$
 (9.14.3)

then

$$g(x) = \int_{0}^{\infty} f(y)k(xy) dy \qquad (9.14.4)$$

belongs to R_{ν} .

A general formula for f(x) of R_{μ} is

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} 2^{\frac{1}{8}\Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s)\psi(s)x^{-s} ds, \qquad (9.14.5)$$

where $\psi(s) = \psi(1-s)$. By (2.1.22),

$$g(x) = rac{1}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} 2^{rac{1}{2}-rac{1}{2}s} \Gamma(rac{3}{4} + rac{1}{2}\mu - rac{1}{2}s) \psi(1-s) 2^{s} \Gamma(rac{1}{4} + rac{1}{2}\mu + rac{1}{2}s) imes
onumber \ imes \Gamma(rac{1}{4} + rac{1}{2}
u + rac{1}{2}s) \chi(s) x^{-s} \, ds$$

$$=\frac{1}{2\pi i}\int\limits_{c-i\infty}^{c+i\infty}2^{\frac{1}{2}s}\Gamma(\frac{1}{4}+\frac{1}{2}\nu+\frac{1}{2}s)\psi_{1}(s)\dot{x}^{-s}\,ds,$$

$$\psi_{1}(s)=2^{\frac{1}{2}}\Gamma(\frac{3}{4}+\frac{1}{2}\mu-\frac{1}{2}s)\Gamma(\frac{1}{4}+\frac{1}{2}\mu+\frac{1}{2}s)\psi(1-s)\chi(s).$$

where Since

Since $\psi_1(s) = \psi_1(1-s)$, g(x) is of the same form as (9.14.5), with μ replaced by ν . This proves the rule.

Since the rule is symmetrical in μ and ν , a kernel which transforms R_{μ} into R_{ν} also effects the converse transformation.

Taking $\mu = \frac{1}{2}$, $\nu = -\frac{1}{2}$, or vice versa, we obtain

$$k(x) = rac{2\sqrt{\pi}}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} \Gamma(s) \chi(s) x^{-s} \, ds,$$

where $\chi(s)$ satisfies (9.14.3), as the general kernel which transforms R_c into R_s (or vice versa). Rule 1 is the case $\chi(s)=1/2\sqrt{\pi}$. Taking

$$\chi(s) = \frac{1}{2\Gamma(\frac{1}{2} + \frac{1}{2}s)\Gamma(1 - \frac{1}{2}s)}, \qquad \frac{1}{\Gamma(\frac{1}{2}s)\Gamma(\frac{1}{2} - \frac{1}{2}s)}, \qquad \frac{\sqrt{3}}{4\pi}\Gamma(s - \frac{1}{3})\Gamma(\frac{2}{3} - s),$$

we obtain

$$k(x) = J_0(x), \qquad xJ_0(x), \qquad x^{-\frac{1}{4}}e^{\frac{1}{4}x}K_{\frac{1}{4}}(\frac{1}{2}x)$$

as other kernels with the same property.

Taking $\chi(s) = 2^{\frac{1}{4}\mu + \frac{1}{4}\nu - \frac{3}{4}}$ in the general rule, we obtain

$$k(x) = x^{\frac{1}{4}\mu + \frac{1}{4}\nu + \frac{1}{4}}K_{\frac{1}{4}\nu - \frac{1}{4}\mu}(x),$$

and, in particular, $K_0(x)$ transforms R_c into itself.

Taking
$$\chi(s) = \frac{2^{\frac{1}{4}\nu - \frac{1}{4}\mu - \frac{1}{2}}}{\Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s)\Gamma(\frac{3}{4} + \frac{1}{2}\mu - \frac{1}{2}s)},$$
 we obtain $k(x) = x^{\frac{1}{4}\nu - \frac{1}{4}\mu + \frac{1}{4}}J_{\frac{1}{4}\mu + \frac{1}{4}\nu}(x);$ and taking $\chi(s) = \frac{2^{\frac{1}{4}\mu - \frac{1}{4}\nu - \frac{1}{2}s}}{\Gamma(\frac{1}{4} + \frac{1}{2}\nu + \frac{1}{2}s)\Gamma(\frac{3}{4} + \frac{1}{2}\nu - \frac{1}{2}s)},$ we obtain $k(x) = x^{\frac{1}{4}\mu - \frac{1}{4}\nu + \frac{1}{4}}J_{\frac{1}{4}\mu + \frac{1}{4}\nu}(x).$

Naturally any of these rules, once they have been obtained, can be verified in the same way as Rule 1.

9.15. Other rules for such transformations may be obtained by combining those already known. For example, if we iterate Rule 1, we obtain

$$g(x) = \int_{0}^{\infty} e^{-xy} \, dy \int_{0}^{\infty} f(t)e^{-yt} \, dt = \int_{0}^{\infty} \frac{f(t)}{t+x} \, dt \qquad (9.15.1)$$

as a function which is $R_c(R_s)$ if f(x) is $R_c(R_s)$. This transformation is not of the above form; but it is of the form

$$g(x) = \frac{1}{x} \int_{0}^{\infty} f(y)k\left(\frac{y}{x}\right) dy, \qquad (9.15.2)$$

with k(u) = 1/(1+u). This suggests a general rule for transformations of this type.

Rule 3. If f(x) belongs to R_{μ} , and

$$k(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s) \Gamma(\frac{3}{4} + \frac{1}{2}\nu - \frac{1}{2}s) \chi(s) x^{-s} ds,$$
(9.15.3)

where $\chi(s) = \chi(1-s)$, then (9.15.2) belongs to R_{ν} .

If f(x) is given by (9.14.5), (2.1.17) gives

$$g(x) = rac{1}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} 2^{rac{1}{4}s} \Gamma(rac{1}{4} + rac{1}{2}
u + rac{1}{2}s)\psi_1(s)x^{-s} ds,$$

where $\psi_1(s) = \Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s)\Gamma(\frac{3}{4} + \frac{1}{2}\mu - \frac{1}{2}s)\psi(s)\chi(1-s)$.

This verifies the rule as before.

In the particular case $\mu = \nu$, (9.15.3) reduces simply to

$$k(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \chi_1(s) x^{-s} ds,$$

where $\chi_1(s) = \chi_1(1-s)$; and this is equivalent to

$$k\left(\frac{1}{x}\right) = xk(x). \tag{9.15.4}$$

Hence

RULE 4. If f(x) belongs to R_{ν} , and k(x) satisfies (9.15.4), then g(x), defined by (9.15.2), belongs to R_{ν} .

It is easy to verify this directly in the usual way.

Particular cases are

$$k(x) = \frac{1}{(1+x^{\alpha})^{1/\alpha}},$$

or, more generally,

$$k(x) = \frac{x^{\frac{1}{2}(\alpha\beta-1)}}{(1+x^{\alpha})^{\beta}}.$$

Taking $f(x) = x^{\nu + \frac{1}{2}}e^{-\frac{1}{2}x^2}$, which belongs to R_{ν} , and $\alpha = 2$, $\beta = 1 - \nu$,

we obtain

$$\begin{split} g(x) &= \frac{1}{x} \int_{0}^{\infty} y^{\nu + \frac{1}{2}} e^{-\frac{i}{2}y^{2}} \left(\frac{y}{x} \right)^{\frac{1}{2} - \nu} \left(1 + \frac{y^{2}}{x^{2}} \right)^{\nu - 1} dy \\ &= x^{\frac{1}{2} - \nu} e^{\frac{i}{2}x^{2}} \int_{x}^{\infty} t^{2\nu - 1} e^{-\frac{i}{2}t^{2}} dt \end{split}$$

(putting $x^2+y^2=t^2$) as a function of R_{ν} . In particular

$$e^{\frac{1}{2}x^2}\int\limits_r^\infty e^{-\frac{1}{2}t^2}\,dt$$

is R_s .

The formula $g(x) = x^{\frac{1}{2} + \nu} \int_{0}^{\infty} y^{\nu + \frac{1}{2}} h(y) e^{-\frac{1}{2}x^{2}y^{2}} dy$

for functions of R_{ν} , where h(y) = h(1/y), is derivable from the above rule by taking $f(x) = x^{\nu + \frac{1}{2}}e^{-\frac{1}{2}x^{\nu}}$, and making obvious transformations.

Taking $\mu = -\frac{1}{2}$, $\nu = \frac{1}{2}$, we obtain

Rule 5. If f(x) belongs to R_c , and

$$k(x) = \frac{1}{2i} \int_{-\infty}^{c+i\infty} \frac{\chi(s)}{\sin \frac{1}{2}s\pi} x^{-s} ds,$$

where $\chi(s) = \chi(1-s)$, then

$$g(x) = \frac{1}{x} \int_{-\infty}^{\infty} f(y) k \left(\frac{y}{x} \right) dy$$

belongs to R_s .

For example, if $\chi(s) = 1$, then $k(x) = 1/(1+x^2)$, and

$$g(x) = x \int_0^\infty \frac{f(y)}{x^2 + y^2} dy.$$

If
$$\chi(s) = \sqrt{\pi/\Gamma(\frac{1}{2} + \frac{1}{2}s)}\Gamma(1 - \frac{1}{2}s)$$
, then
$$k(x) = (1 - x^2)^{-\frac{1}{2}} \quad (0 < x < 1), \qquad 0 \quad (x > 1),$$

and

$$g(x) = \int_0^x \frac{f(y) dy}{\sqrt{(x^2-y^2)}}.$$

There are, of course, similar rules for transformations from R_s to R_c .

9.16. The main interest of the above rules is in their formal appearance. A considerable variety of theorems about them could be constructed.† We shall give here only one, applying to Rule 2. The reader should have little difficulty in dealing with the other rules in the same way.

THEOREM 144. Let f(x) and k(x) be $L^2(0,\infty)$, and let f(x) be R_{μ} . Let

$$g(x) = \int_{0}^{\infty} f(y)k(xy) dy \qquad (9.16.1)$$

be also $L^2(0,\infty)$. Then, in order that g(x) should be R_{ν} , it is necessary and sufficient that $\Re(s)$, the Mellin transform of k(x), should be of the form $\Re(s) = 2^s \Gamma(\frac{1}{2} + \frac{1}{2}\mu + \frac{1}{2}s)\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)\chi(s), \qquad (9.16.2)$

where $\chi(s) = \chi(1-s)$, and the right-hand side is $L^2(\frac{1}{2}-i\infty, \frac{1}{2}+i\infty)$.

By Theorem 72, with g(x) replaced by k(xy).

$$\int\limits_0^\infty f(y)k(xy)\;dy=\frac{1}{2\pi i}\int\limits_{\frac{1}{s}-i\infty}^{\frac{1}{s}+i\infty}\mathfrak{F}(s)\mathfrak{R}(1-s)y^{s-1}\;ds.$$

The Mellin transform of g(x) is therefore

$$\mathfrak{G}(s) = \mathfrak{F}(1-s)\mathfrak{R}(s).$$

By the analogue for R_{μ} of Theorem 136

$$\mathfrak{F}(s) = 2^{\frac{1}{8}}\Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s)\psi(s),$$

where $\psi(s) = \psi(1-s)$. If g(x) is R_{ν} , we have also

$$\mathfrak{G}(s) = 2^{\frac{1}{4}s} \Gamma(\frac{1}{4} + \frac{1}{2}\nu + \frac{1}{2}s)\omega(s),$$

where $\omega(s) = \omega(1-s)$. Hence

$$\Re(s) = \frac{2^{\frac{1}{4}s}\Gamma(\frac{1}{4} + \frac{1}{2}\nu + \frac{1}{2}s)\omega(s)}{2^{\frac{1}{4} - \frac{1}{4}s}\Gamma(\frac{3}{4} + \frac{1}{2}\mu - \frac{1}{2}s)\psi(1-s)},$$

which is of the form (9.16.2), with

$$\chi(s) = \sqrt{2} \, \Gamma(\frac{1}{4} + \frac{1}{2}\mu + \frac{1}{2}s) \Gamma(\frac{3}{4} + \frac{1}{2}\mu - \frac{1}{2}s) \omega(s) / \psi(1-s)$$

satisfying $\chi(s) = \chi(1-s)$. Hence the form (9.16.2) is necessary; and the reversed argument shows that it is sufficient.

9.17. A series formula for self-reciprocal functions may be derived from the function

$$f(x) = \frac{1}{x} \left(\frac{x}{\sqrt{(2\pi)}} - \left[\frac{x}{\sqrt{(2\pi)}} \right] \right),$$

† See e.g. Mehrotra (1).

which belongs to R_i . Let k(x) be a kernel which transforms R_c to R_r . Then, by Rule 3,

$$egin{align} k(x) &= rac{1}{2\pi i} \int\limits_{c-i\infty}^{c+i\infty} 2^s \Gamma(rac{1}{2}s) \Gamma(rac{1}{4} + rac{1}{2}
u + rac{1}{2}s) \chi(s) x^{-s} \, ds; \ x k'(x) &= -rac{1}{\pi i} \int\limits_{c}^{c+i\infty} 2^s \Gamma(1 + rac{1}{2}s) \Gamma(rac{1}{4} + rac{1}{2}
u + rac{1}{2}s) \chi(s) x^{-s} \, ds \end{array}$$

is a kernel which transforms R_i to R_{ν} . Hence

$$\begin{split} g(x) &= \int\limits_{0}^{\infty} f(y) x y k'(xy) \; dy \\ &= \sum\limits_{n=1}^{\infty} x \int\limits_{(n-1)\sqrt{(2\pi)}}^{n\sqrt{(2\pi)}} \left\{ \frac{y}{\sqrt{(2\pi)}} - n + 1 \right\} k'(xy) \; dy \\ &= \sum\limits_{n=1}^{\infty} \left\{ \left[\left(\frac{y}{\sqrt{(2\pi)}} - n + 1 \right) k(xy) \right]_{(n-1)\sqrt{(2\pi)}}^{n\sqrt{(2\pi)}} - \frac{1}{\sqrt{(2\pi)}} \int\limits_{(n-1)\sqrt{(2\pi)}}^{n\sqrt{(2\pi)}} k(xy) \; dy \right\} \\ &= \sum\limits_{n=1}^{\infty} k \{ nx \sqrt{(2\pi)} \} - \frac{1}{\sqrt{(2\pi)x}} \int\limits_{0}^{\infty} k(u) \; du \end{split}$$

should be a function of R_{n} .

The rule may be verified as follows. Let

$$k(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re(s) x^{-s} ds \quad (c > 0).$$

Then

$$\begin{split} \sum_{n=1}^{\infty} k \{ nx \sqrt{(2\pi)} \} &= \frac{1}{2\pi i} \int_{c'-i\infty}^{c'+i\infty} \Re(s) \sum_{n=1}^{\infty} \{ nx \sqrt{(2\pi)} \}^{-s} \, ds \quad (c' > 1) \\ &= \frac{1}{2\pi i} \int_{c'-i\infty}^{c'+i\infty} \Re(s) (2\pi)^{-\frac{1}{4}s} \zeta(s) x^{-s} \, ds \\ &= \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re(s) (2\pi)^{-\frac{1}{4}s} \zeta(s) x^{-s} \, ds + \frac{\Re(1)}{\sqrt{(2\pi)}x}. \end{split}$$
 Hence
$$g(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Re(s) (2\pi)^{-\frac{1}{4}s} \zeta(s) x^{-s} \, ds.$$

4362

If k(x) transforms R_c to R_v , then by Rule 3

$$\Re(s) = 2^s \Gamma(\frac{1}{2}s) \Gamma(\frac{1}{4} + \frac{1}{2}\nu + \frac{1}{2}s) \chi(s),$$

where $\chi(s) = \chi(1-s)$. Hence

$$\mathfrak{G}(s) = \Re(s)(2\pi)^{-\frac{1}{6}s}\zeta(s) = 2^{\frac{1}{6}s}\Gamma(\frac{1}{4} + \frac{1}{2}\nu + \frac{1}{2}s)\chi_1(s),$$

where

$$\chi_1(s) = \pi^{-\frac{1}{2}s} \Gamma(\frac{1}{2}s) \zeta(s) \chi(s).$$

Hence

$$\chi_1(s) = \chi_1(1-s)$$

by the functional equation for $\zeta(s)$.

Hence g(x) belongs to R_{ν} , by (9.1.9).

As examples, let $k(x) = e^{-x}$, $\nu = \frac{1}{2}$. Then

$$g(x) = \frac{1}{e^{x\sqrt{(2\pi)}}-1} - \frac{1}{x\sqrt{(2\pi)}}$$

is R_s , as in § 9.12 (3).

Taking $k(x) = J_0(x)$, another R_s function is \dagger

$$\sum_{n=1}^{\infty} J_0\{nx\sqrt{(2\pi)}\} - \frac{1}{x\sqrt{(2\pi)}} = \sqrt{\left(\frac{2}{\pi}\right)} \sum_{n < x/\sqrt{(2\pi)}} \frac{1}{\sqrt{(x^2 - 2n^2\pi)}} - \frac{1}{2}.$$

Taking $k(x) = x^{\frac{1}{4}\nu + \frac{1}{4}}K_{\frac{1}{4}\nu + \frac{1}{4}}(x)$, we obtain ‡

$$\sum_{n=1}^{\infty} \{nx\sqrt{(2\pi)}\}^{\frac{1}{4}\nu + \frac{1}{4}} K_{\frac{1}{4}\nu + \frac{1}{4}} \{nx\sqrt{(2\pi)}\} - \frac{2^{\frac{1}{4}\nu - \frac{1}{4}} \Gamma(\frac{3}{4} + \frac{1}{2}\nu)}{x}$$

as a function of R_{ν} .

Watson (1).

DIFFERENTIAL AND DIFFERENCE EQUATIONS

10.1. Introduction. In this chapter we use Fourier's integral formula and its related formulae to obtain the solutions of certain differential equations. The general method is to transform a differential (or other functional) equation, involving an unknown function, into a relation involving the Fourier transform, or some similar transform, of the original function. The new relation may be simpler, and so lead to the solution.

That certain differential equations can be solved by means of definite integrals was shown by Laplace and Cauchy. The main object of this chapter is to present some cases of this familiar method as exercises in the use of Fourier's integral formula.

The chapter is merely a collection of examples illustrating the possibilities of the method. Most of them are familiar, and the solutions are to be found in standard works. The methods usually employed, however, are more or less tentative, and often make no explicit use of Fourier's theorem. Here we aim at solving the equations subject to simple conditions which justify a priori the process used.

10.2. Ordinary differential equations. We shall first give a method of solving ordinary linear differential equations, due to Bromwich.† The method, in its rigorous form, depends on Theorems 33 and 34.

One of Bromwich's examples is

$$\left(\frac{d^2}{dt^2} - 4\frac{d}{dt}\right)x - \left(\frac{d}{dt} - 1\right)y = 0,
\left(\frac{d}{dt} + 6\right)x + \left(\frac{d^2}{dt^2} - \frac{d}{dt}\right)y = 0,$$
(10.2.1)

where $x(0) = x_0$, $x'(0) = x_1$, $y(0) = y_0$, $y'(0) = y_1$ are given constants.

It can be seen a priori that x(t) and y(t) are integral functions of exponential type. For example, by further differentiation and elimination we obtain

$$x'''(t)+c_1x''(t)+c_2x'(t)+c_3x(t)=0.$$
† Bromwich (1).

Hence

$$x^{(n+3)}(t)+c_1x^{(n+2)}(t)+c_2x^{(n+1)}(t)+c_3x^{(n)}(t)=0.$$

If $|x^{(m)}(t)| \leq \{K(t)\}^m$ for m = 1, 2, ..., n+2, it follows that

$$|x^{(n+3)}(t)| \leq (|c_1| + ... + |c_3|) \{K(t)\}^{n+2} \leq \{K(t)\}^{n+3}$$

provided that $K(t) \ge 1$, $K(t) \ge |c_1| + ... + |c_3|$. Hence x(t) is expansible by Taylor's theorem, and

$$|x(t)| = \left|\sum_{n=0}^{\infty} \frac{x^{(n)}(0)t^n}{n!}\right| \leqslant \sum_{n=0}^{\infty} \frac{\{K(0)\}^n |t|^n}{n!} = e^{K(0)|t|},$$

so that x(t) is an integral function of exponential type. Similarly for y(t).

Hence, by Theorem 33,

$$x(t) = \frac{1}{2\pi i} \int_{C} \xi(w) e^{wt} dw, \qquad y(t) = \frac{1}{2\pi i} \int_{C} \eta(w) e^{wt} dw,$$
 (10.2.2)

where $\xi(w)$ and $\eta(w)$ are regular for |w| > R, say, and zero at infinity; and C is a simple closed curve surrounding |w| = R.

The differential equations then give

$$\int_{C} \{\xi(w)(w^{2}-4w)-\eta(w)(w-1)\}e^{wt} dw = 0,
\int_{C} \{\xi(w)(w+6)+\eta(w)(w^{2}-w)\}e^{wt} dw = 0.$$
(10.2.3)

Let

$$\begin{cases} \xi(w)(w^2-4w)-\eta(w)(w-1) = p(w), \\ \xi(w)(w+6)+\eta(w)(w^2-w) = q(w). \end{cases}$$
 (10.2.4)

Then p(w) and q(w) are regular for |w| > R, except for poles of the first order at infinity. Hence, by Theorem 34, they are linear functions of w, say

$$p(w) = a + bw, q(w) = \alpha + \beta w.$$
 (10.2.5)

Also, from (10.2.2),

$$x_0 = \frac{1}{2\pi i} \int_C \xi(w) dw, \qquad x_1 = \frac{1}{2\pi i} \int_C \xi(w) w dw,$$

and hence, by Laurent's theorem,

$$\xi(w) = \frac{x_0}{w} + \frac{x_1}{w^2} + O\left(\frac{1}{|w|^3}\right)$$
 (10.2.6)

for large |w|. Similarly,

$$\eta(w) = \frac{y_0}{w} + \frac{y_1}{w^2} + O\left(\frac{1}{|w|^3}\right). \tag{10.2.7}$$

Substituting (10.2.5), (10.2.6), (10.2.7) in (10.2.4), and equating coefficients, we find

$$p(w) = (w-4)x_0+x_1-y_0, \qquad q(w) = x_0+(w-1)y_0+y_1.$$

Solving (10.2.4) for ξ and η , we obtain

$$\xi(w) = \frac{wp(w) + q(w)}{(w+1)(w-2)(w-3)}, \quad \eta(w) = \frac{-(w+6)p(w) + (w^2 - 4w)q(w)}{(w^2 - 1)(w-2)(w-3)}.$$

The values of x(t) and y(t) now follow from (10.2.2) by the calculus of residues. For example, the term in x(t) corresponding to the pole at w = -1 is

$$\frac{-p(-1)+q(-1)}{12}e^{-t} = \frac{6x_0-x_1-y_0+y_1}{12}e^{-t}.$$

Naturally the method is quite general. Another simple example is

$$\frac{d^2x}{dt^2} + 2n\frac{dx}{dt} + n^2x = 0,$$

$$\frac{d^2y}{dt^2} + 2n\frac{dy}{dt} + n^2y = \mu \frac{dx}{dt},$$

where x(0) = 0, $x'(0) = \lambda$, y(0) = 0, y'(0) = 0. This is given by Jeffreys, Operational Methods, § 3.31, as an example of the operational equivalent of the above method.

10.3. If we are given a linear equation whose coefficients are polynomials of degree m, and treat it by the above method, we have to integrate by parts m times, and the transform of the original function satisfies a differential equation of the mth order. Consider, for example, Bessel's equation,

$$\frac{d^2z}{dx^2} + \frac{1}{x}\frac{dz}{dx} + \left(1 - \frac{v^2}{x^2}\right)z = 0,$$

where $\nu > 0$. Putting $z = x^{\nu}y$, we obtain

$$\frac{d^2y}{dx^2} + \frac{2\nu + 1}{x} \frac{dy}{dx} + y = 0.$$

Let us seek a solution which is an integral function of exponential type. Let it be

$$y = \frac{1}{2\pi i} \int_{\Omega} \eta(w) e^{xw} \, dw.$$

Then

$$egin{align} xy &= -rac{1}{2\pi i}\int\limits_C \, \eta'(w)e^{xw}\,dw, \ rac{dy}{dx} &= rac{1}{2\pi i}\int\limits_C \, \eta(w)we^{xw}\,dw, \ xrac{d^2y}{dx^2} &= -rac{1}{2\pi i}\int\limits_C \, \{2w\eta(w)\!+\!w^2\eta'(w)\}e^{xw}\,dw. \ \int\limits_C \, \{(1\!+\!w^2)\eta'(w)\!-\!(2\nu\!-\!1)w\eta(w)\}e^{xw}\,dw &= 0. \end{align}$$

Hence

The factor in brackets has at most a pole of the first order at infinity; hence, by Theorem 34,

$$(1+w^2)\eta'(w)-(2\nu-1)w\eta(w)=a,$$
 $\eta(w)=a(1+w^2)^{\nu-\frac{1}{2}}\int \frac{dw}{(1+w^2)^{\nu+\frac{1}{2}}}+b(1+w^2)^{\nu-\frac{1}{2}}.$

Since $\eta(w)$ is regular at infinity, b = 0, and

$$\eta(w) = a(1+w^2)^{\nu-\frac{1}{2}} \int_{0}^{\infty} \frac{d\zeta}{(\zeta^2+1)^{\nu+\frac{1}{2}}},$$

where we take the branches of $(w^2+1)^{\nu-\frac{1}{2}}$ and $(\zeta^2+1)^{\nu+\frac{1}{2}}$ which are real on the real axis, and suppose the plane cut along the imaginary axis from -i to i. Then

$$y = \frac{a}{2\pi i} \int\limits_C (1+w^2)^{\nu-\frac{1}{2}} e^{xw} dw \int\limits_w^{\infty} \frac{d\zeta}{(\zeta^2+1)^{\nu+\frac{1}{2}}}.$$

This can be reduced to a more familiar form. Let w be a point on the imaginary axis between 0 and i, w' the same point after a circuit has been made round i. Then

$$\begin{split} \eta(w) - \eta(w') \\ &= a(1+w^2)^{\nu-\frac{1}{2}} \int\limits_{w}^{\infty} \frac{d\zeta}{(\zeta^2+1)^{\nu+\frac{1}{2}}} - ae^{2\pi i(\nu-\frac{1}{2})} (1+w^2)^{\nu-\frac{1}{2}} \int\limits_{w}^{-\infty} \frac{d\zeta}{(\zeta^2+1)^{\nu+\frac{1}{2}}}. \\ \text{Now} \qquad \int\limits_{w}^{-\infty} \frac{d\zeta}{(\zeta^2+1)^{\nu+\frac{1}{2}}} = -e^{-2\pi i(\nu+\frac{1}{2})} \int\limits_{-\infty}^{w} \frac{d\zeta}{(\zeta^2+1)_1^{\nu+\frac{1}{2}}}, \end{split}$$

where the suffix denotes the branch obtained by cuts from $-i\infty$ to -i and i to $i\infty$. Hence

$$\eta(w) - \eta(w') = a(1+w^2)^{\nu-\frac{1}{2}} \int_{-\infty}^{\infty} \frac{d\zeta}{(\zeta^2+1)_1^{\nu+\frac{1}{2}}} = aK(1+w^2)^{\nu-\frac{1}{2}},$$

where K depends on ν only. Hence

$$y = K_1 \int_{-1}^{1} (1-v^2)^{\nu-\frac{1}{2}} e^{ixv} dv = K_2 \int_{0}^{1} (1-v^2)^{\nu-\frac{1}{2}} \cos xv dv.$$

10.4. If we put $z = x^{-\nu}y$, $x^2 = 4t$ in Bessel's equation, we obtain

$$t\frac{d^2y}{dt^2} + (1-\nu)\frac{dy}{dt} + y = 0. (10.4.1)$$

The solution corresponding to $z = J_{\nu}(x)$ is not an integral function for general ν . Let us look instead for solutions y(t) such that $y(t) = O(e^{ct})$ as $t \to \infty$ for some positive c, and y(0) = 0.

Such a solution is representable by a Fourier integral, by Theorem 24. Let f(t) = y(t) for t > 0, and f(t) = 0 for t < 0. Then $F_{-}(w) = 0$, and $F_{+}(w)$ is

$$Y(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} y(t)e^{iwt} dt$$
 (10.4.2)

for v > c. Since y(t) is continuous and of bounded variation in any finite interval, Theorem 24 gives

$$y(t) = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ia-\lambda}^{ia+\lambda} Y(w)e^{-iwt} dw, \qquad (10.4.3)$$

for t > 0, a > c.

Integrating (10.4.2) by parts, we have

$$\sqrt{(2\pi)}Y(w) = -\frac{1}{iw} \int_{0}^{\infty} y'(t)e^{iwt} dt, \qquad (10.4.4)$$

the integrated term vanishing. Similarly,

$$\sqrt{(2\pi)Y'(w)} = \int_0^\infty ity(t)e^{iwt} dt$$

$$= -\frac{1}{w} \int_0^{+\infty} \{y(t) + ty'(t)\}e^{iwt} dt. \qquad (10.4.5)$$

Integrating by parts again,

$$\begin{split} \sqrt{(2\pi)}Y'(w) &= \frac{1}{iw^2} \int_0^{\infty} \{2y'(t) + ty''(t)\}e^{iwt} dt \\ &= \frac{1}{iw^2} \int_0^{\infty} \{(\nu+1)y'(t) - y(t)\}e^{iwt} dt \end{split} \tag{10.4.6}$$

by (10.4.1). The last integral is convergent since (10.4.4) is; the integrated term

 $-\frac{1}{iw^2}\{y(t)+ty'(t)\}e^{iwt}$

must therefore tend to a limit as $t \to \infty$, and the limit can only be 0. From (10.4.2), (10.4.4), and (10.4.6) it follows that

$$Y'(w) = \left(-\frac{\nu+1}{w} - \frac{1}{iw^2}\right)Y(w).$$

Hence

i.e.

$$Y(w) = Kw^{-\nu-1}e^{1/iw},$$

$$y(t) = \frac{K}{\sqrt{(2\pi)}} \int_{-\infty}^{ia+\infty} \frac{e^{-iwt+1/iw}}{w^{\nu+1}} dw.$$

This is in fact a multiple of $t^{\frac{1}{4}\nu}J_{\nu}(2\sqrt{t})$, by (7.13.9).

10.5. A similar method may be used to solve differential equations with a given function on the right-hand side. To take a simple case, consider

 $\frac{d^2y}{dt^2}+k^2y=\phi(t) \quad (t\geqslant 0),$

where all the functions concerned are of the form $O(e^{ct})$ as $t \to \infty$. If v > c,

$$egin{align} \sqrt{(2\pi)}Y(w) &= \int\limits_0^\infty y(t)e^{iwt}\,dt \ &= -rac{y(0)}{iw} - rac{y'(0)}{w^2} - rac{1}{w^2}\int\limits_0^\infty y''(t)e^{iwt}\,dt, \end{gathered}$$

integrating by parts twice. Hence

$$egin{align} \Phi(w) &= rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty \phi(t) e^{iwt} \, dt = rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty \{y''(t) + k^2 y(t)\} e^{iwt} \, dt \ &= -w^2 Y(w) + rac{iwy(0)}{\sqrt{(2\pi)}} - rac{y'(0)}{\sqrt{(2\pi)}} + k^2 Y(w), \ &Y(w) &= rac{\Phi(w)}{k^2 - w^2} - rac{1}{\sqrt{(2\pi)}} rac{iwy(0) - y'(0)}{k^2 - w^2}. \end{align}$$

Hence for a sufficiently large, in particular a > c,

$$y(t) = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{i\sigma-\lambda}^{i\sigma+\lambda} \left\{ \frac{\Phi(w)}{k^2 - w^2} - \frac{1}{\sqrt{(2\pi)}} \frac{iwy(0) - y'(0)}{k^2 - w^2} \right\} e^{-iwt} dw.$$

In the first part we can insert the Fourier integral for $\Phi(w)$, and invert, by absolute convergence. We obtain

$$\frac{1}{2\pi}\int_{0}^{\infty}\phi(x)\,dx\int_{ia-\infty}^{ia+\infty}\frac{e^{iw(x-t)}}{k^2-w^2}\,dw=\frac{1}{k}\int_{0}^{t}\phi(x)\sin k(t-x)\,dx,$$

evaluating the inner integral by the calculus of residues. The remaining terms are

 $y(0)\cos kt + y'(0)\frac{\sin kt}{k}.$

Hence

$$y(t) = y(0)\cos kt + y'(0)\frac{\sin kt}{k} + \frac{1}{k}\int_{0}^{t}\phi(x)\sin k(t-x) dx,$$

the usual solution.

10.6. Partial differential equations. Obtain the solution v(x,t) of

$$\frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial x^2} \quad (-\infty < x < \infty, \ t > 0) \tag{10.6.1}$$

such that

$$v(x,0) = f(x) \quad (-\infty < x < \infty).$$

This is the classical problem of the flow of heat in an infinite rod with a given initial temperature distribution, v being the temperature, t the time, and x the distance along the rod. †

Formally we proceed as follows. Let

$$V(\xi,t) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} v(x,t)e^{ix\xi} dx.$$

Then

$$\begin{split} \frac{\partial V}{\partial t} &= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \frac{\partial v}{\partial t} e^{ix\xi} dx = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \frac{\partial^2 v}{\partial x^2} e^{ix\xi} dx \\ &= -\frac{\xi^2}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} v e^{ix\xi} dx = -\xi^2 V, \end{split}$$

integrating by parts twice, and assuming that the terms at the limits vanish. Hence

$$V(\xi,t)=A(\xi)e^{-\xi^2t},$$

where $A(\xi)$ depends on ξ only. Putting t=0,

$$A(\xi) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(x)e^{ix\xi} dx = F(\xi),$$

† Riemann-Weber, 2, § 36; Carslaw, Heat, § 16.

F being the transform of f. Hence

$$V(\xi,t) = F(\xi)e^{-\xi^2t},$$

and the solution is

$$v(x,t) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} F(\xi)e^{-\xi^{\epsilon}t - ix\xi} d\xi,$$

or, in terms of f(x), $v(x,t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{-\xi^{2}t - ix\xi} d\xi \int_{-\infty}^{\infty} f(u)e^{i\xi u} du$ $= \frac{1}{2\pi} \int_{-\infty}^{\infty} f(u) du \int_{-\infty}^{\infty} e^{-\xi^{2}t - i\xi(x-u)} d\xi$ $= \frac{1}{2\sqrt{(\pi t)}} \int_{-\infty}^{\infty} f(u)e^{-\frac{1}{2}(x-u)^{2}/t} du. \qquad (10.6.2)$

That in fact $v(x,t) \to f(x)$ as $t \to 0$ follows from the theory of Weierstrass's singular integral. The method would be justified e.g. if all the functions concerned belong to $L(-\infty,\infty)$. But the following procedure is much more general.

Suppose that $|v(x,t)| < Ke^{c|x|}$ for some c and all t, with similar conditions on any of the partial derivatives which occur. We shall say that such a function is of exponential type.

Let $v(x,t) \to f(x)$, as $t \to 0$, for almost all values of x. Then $|f(x)| < Ke^{c|x|}$ almost everywhere.

Let
$$V_+(\zeta,t) = \frac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty v(x,t) e^{ix\zeta} dx, \qquad V_-(\zeta,t) = \frac{1}{\sqrt{(2\pi)}} \int\limits_0^0 v(x,t) e^{ix\zeta} dx,$$

where $\zeta = \xi + i\eta$. Then V_+ exists and is regular for $\eta > c$, V_- for $\eta < -c$.

Now if $\eta > c$,

$$\sqrt{(2\pi)} \frac{\partial V_{+}}{\partial t} = \int_{0}^{\infty} \frac{\partial v}{\partial t} e^{ix\zeta} dx = \int_{0}^{\infty} \frac{\partial^{2}v}{\partial x^{2}} e^{ix\zeta} dx$$

$$= \left[\frac{\partial v}{\partial x} e^{ix\zeta} \right]_{0}^{\infty} - i\zeta \int_{0}^{\infty} \frac{\partial v}{\partial x} e^{ix\zeta} dx$$

$$= \left[\frac{\partial v}{\partial x} e^{ix\zeta} \right]_{0}^{\infty} - i\zeta [ve^{ix\zeta}]_{0}^{\infty} - \zeta^{2} \int_{0}^{\infty} ve^{ix\zeta} dx$$

$$= -v_{x}(0, t) + i\zeta v(0, t) - \zeta^{2} \sqrt{(2\pi)} V_{+}(\zeta, t).$$

This is an ordinary differential equation, of which (e.g. as in § 10.5) the solution is

$$V_+(\zeta,t) = A(\zeta)e^{-\zeta^{2}\!t} - rac{e^{-\zeta^{2}\!t}}{\sqrt{(2\pi)}} \int\limits_0^t \{v_x(0, au) - i\zeta v(0, au)\}e^{\zeta^{2}\! au}\,d au.$$

Making $t \to 0$,

$$A(\zeta) = \lim_{t \to 0} V_{+}(\zeta, t) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} f(x)e^{ix\zeta} dx = F_{+}(\zeta)$$

by dominated convergence, since $v(x,t) \rightarrow f(x)$ for almost all x, and

$$|v(x,t)e^{ix\zeta}| < Ke^{(c-\eta)x}$$

Hence

$$V_{+}(\zeta,t) = F_{+}(\zeta)e^{-\zeta t} - \chi(\zeta,t),$$

where $\chi(\zeta, t)$ is an integral function of ζ which $\to 0$ as $\xi \to \pm \infty$.

In the corresponding argument with V_{-} the integrated terms appear with the opposite sign, and we obtain

$$V_{-}(\zeta,t)=F_{-}(\zeta)e^{-\zeta^{2}t}+\chi(\zeta,t).$$

Hence, by Theorem 24,

$$\begin{split} v(x,t) &= \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int\limits_{ia-\lambda}^{\infty+1} \{F_+(\zeta)e^{-\zeta \cdot \mathbf{q}} - \chi(\zeta,t)\}e^{-ix\zeta} \, d\zeta \, + \\ &\quad + \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int\limits_{ib-\lambda}^{ib+\lambda} \{F_-(\zeta)e^{-\zeta \cdot \mathbf{q}} + \chi(\zeta,t)\}e^{-ix\zeta} \, d\zeta. \end{split}$$

The contribution of χ is plainly 0. The contribution of F_+ is

$$\frac{1}{2\pi}\int_{ia-\infty}^{ia+\infty}e^{-\zeta^{2}t-ix\zeta}\,d\zeta\int_{0}^{\infty}f(u)e^{iu\zeta}\,du=\frac{1}{2\pi}\int_{0}^{\infty}f(u)\,du\int_{ia-\infty}^{ia+\infty}e^{-\zeta^{2}t-i\zeta(x-u)}\,d\zeta$$

(inverting by absolute convergence)

$$=\frac{1}{2\sqrt{(\pi t)}}\int\limits_{0}^{\infty}f(u)e^{-(x-u)^{2}/(4t)}\,du.$$

Similarly for F_{-} , and we obtain (10.6.2) as before.

We do not know whether, for a unique solution, it is necessary to assume that v(x,t) is of exponential type. But some condition bearing on v(x,t) and not merely on f(x) is necessary. It is an easily verified rule that, if v(x,t) is a solution of (10.6.1), then so is

$$t^{-\frac{1}{4}}e^{-\frac{1}{4}x^{2}/t}v\bigg(\!\frac{x}{t},-\frac{1}{t}\!\bigg).$$

Hence, v(x,t) = x being a solution, so is

$$xt^{-\frac{1}{2}}e^{-\frac{1}{4}x^{2}/t}$$

as is easily verified; and this function tends to 0, as $t \to 0$, for every x, without vanishing identically. It is of course unbounded near x = 0 as $t \to 0$ (e.g. for $x = \sqrt{t}$), so that it does not satisfy the conditions of the above analysis.

10.7. Obtain the solution v(x,t) of

$$\frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial x^2} \quad (x > 0, \ t > 0)$$

such that v(x,0) = 0 (x > 0), v(0,t) = f(t) (t > 0).

This is the problem \dagger of the conduction of heat in a semi-infinite rod, initially at zero temperature, the end being suddenly raised to, and maintained at, a given temperature f(t).

For a formal solution let

$$V_s(\xi,t) = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty v(x,t) \sin \xi x \ dx.$$

Then

$$\frac{\partial V_s}{\partial t} = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty \frac{\partial v}{\partial t} \sin \xi x \, dx$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty \frac{\partial^2 v}{\partial x^2} \sin \xi x \, dx$$

$$= -\sqrt{\left(\frac{2}{\pi}\right)} \xi \int_0^\infty \frac{\partial v}{\partial x} \cos \xi x \, dx$$

$$= -\sqrt{\left(\frac{2}{\pi}\right)} \xi \left[\left[v \cos \xi x \right]_0^\infty + \xi \int_0^\infty v \sin \xi x \, dx \right]$$

$$= \sqrt{\left(\frac{2}{\pi}\right)} \xi f(t) - \xi^2 V_s.$$

$$V_s = A(\xi) e^{-\xi^2 t} + \sqrt{\left(\frac{2}{\pi}\right)} \xi e^{-\xi^2 t} \int_0^t e^{\xi^2 u} f(u) \, du.$$

Hence

Since v(x,0) = 0, $V_s(\xi,0) = 0$, and hence $A(\xi) = 0$. Hence

$$v(x,t) = \frac{2}{\pi} \int_{0}^{\infty} \xi e^{-\xi^{2}t} \sin \xi x \, d\xi \int_{0}^{t} e^{\xi^{2}u} f(u) \, du$$

† Riemann-Weber, 2, § 40; Carslaw, Heat, § 23.

$$= \frac{2}{\pi} \int_{0}^{t} f(u) du \int_{0}^{\infty} \xi e^{\xi^{2}(u-t)} \sin \xi x d\xi$$

$$= \frac{x}{2\sqrt{\pi}} \int_{0}^{t} f(u)(t-u)^{-\frac{1}{2}} e^{-\frac{1}{2}x^{2}/(t-u)} du.$$

That this tends to f(x) in general follows from Theorem 13.

For a rigorous proof suppose again that v(x,t) is of exponential type. Let

$$V(\zeta,t) = rac{1}{\sqrt{(2\pi)}} \int_{s}^{\infty} v(x,t) e^{i\zeta x} dx \quad (\eta > c).$$

Then

$$egin{aligned} \sqrt{(2\pi)} rac{\partial V}{\partial t} &= \int\limits_0^\infty rac{\partial v}{\partial t} e^{i\zeta x} \, dx = \int\limits_0^\infty rac{\partial^2 v}{\partial x^2} e^{i\zeta x} \, dx \ &= -v_x(0,t) + i\zeta f(t) - \sqrt{(2\pi)}\zeta^2 V, \end{aligned}$$

integrating by parts twice. Hence

$$V(\zeta,t) = A(\zeta)e^{-\zeta^{2}t} + \frac{e^{-\zeta^{2}t}}{\sqrt{(2\pi)}} \int_{0}^{t} \{i\zeta f(u) - v_{x}(0,u)\}e^{\zeta^{2}u} du.$$

As before, $A(\zeta) = 0$; if f(u) and $v_x(0, u)$ are bounded in a finite interval, the other term is

$$O\left(e^{-\xi^{*}l}\int_{0}^{t}|\xi|e^{\xi^{*}u}\,du\right)=O(|\xi|^{-1})$$

as $\xi \to \pm \infty$. Hence, by L^2 theory,

$$v(x,t) = rac{1}{2\pi} rac{d}{dx} \int\limits_{ia-\infty}^{ia+\infty} rac{e^{-i\zeta x}-1}{-i\zeta} e^{-\zeta^{*}t} d\zeta \int\limits_{0}^{t} \{i\zeta f(u)-v_{x}(0,u)\} e^{\zeta^{*}u} du$$

for x > 0, while the right-hand side is 0 for x < 0. The repeated integral is absolutely convergent, and we may invert, and then replace a by 0. The term in f(u) contributes

$$\frac{1}{2\pi} \frac{d}{dx} \int_{0}^{t} f(u) du \int_{-\infty}^{\infty} (1 - e^{-i\xi x}) e^{\xi^{s}(u-t)} d\xi
= \frac{1}{2\sqrt{\pi}} \frac{d}{dx} \int_{0}^{t} f(u) \{1 - e^{-\frac{1}{4}x^{s}/(t-u)}\} \frac{du}{\sqrt{(t-u)}}
= \frac{x}{4\sqrt{\pi}} \int_{0}^{t} f(u) e^{-\frac{1}{4}x^{s}/(t-u)} \frac{du}{(t-u)^{\frac{1}{4}}}.$$

The contribution of the other term is an even function of x. Changing the sign of x and subtracting, we obtain the same result as before.

10.8. Obtain the solution of

$$\frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial x^2} - \alpha^2 v,$$

where v(x,0) = 0 (x > 0), v(0,t) = f(t) (t > 0).

Proceeding as before, we obtain

$$\frac{\partial V_s}{\partial t} = \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty \left(\frac{\partial^2 v}{\partial x^2} - \alpha^2 v\right) \sin \xi x \, dx$$
$$= \sqrt{\left(\frac{2}{\pi}\right)} \xi f(t) - \xi^2 V_s - \alpha^2 V_s.$$

Hence

$$V_s = A(\xi)e^{-(\xi^*+lpha^*)t} + \sqrt{\left(rac{2}{\pi}
ight)} \xi e^{-(\xi^*+lpha^*)t} \int\limits_0^t e^{(\xi^*+lpha^*)u} f(u) \ du,$$

 $A(\xi) = 0$ as before, and

$$\begin{split} v(x,t) &= \frac{2}{\pi} \int_{0}^{t} f(u) \ du \int_{0}^{\infty} \xi e^{(\xi^{1} + \alpha^{1})(u - t)} \sin \xi x \ d\xi \\ &= \frac{x}{2\sqrt{\pi}} \int_{0}^{t} f(u)(t - u)^{-\frac{1}{4}} e^{-\alpha^{1}(t - u) - \frac{1}{4}x^{1}/(t - u)} \ du. \end{split}$$

The rigorous solution may be obtained as before.

10.9. Solvet

$$\frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial x^2} - \alpha^2 v \quad (0 < x < l),$$

where $v = v_0$ (x = 0, t > 0), v = 0 (t = 0, 0 < x < l), and

$$\frac{\partial v}{\partial x} = 0 \quad (x = l).$$

Here we take t as the variable of the Fourier integral, and suppose that $|v(x,t)| < Ke^{c|t|}$ for all x. Let

$$V(x,\zeta) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} v(x,t)e^{i\zeta t} dt \quad (\eta > c).$$

† See Jeffreys, Operational Methods, p. 70.

Then

$$\begin{split} \frac{\partial^2 V}{\partial x^2} &= \frac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty \frac{\partial^2 v}{\partial x^2} e^{i\zeta t} \, dt \\ &= \frac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty \left(\frac{\partial v}{\partial t} + \alpha^2 v \right) e^{i\zeta t} \, dt \\ &= \frac{1}{\sqrt{(2\pi)}} \left[v e^{i\zeta t} \right]_0^\infty - \frac{i\zeta}{\sqrt{(2\pi)}} \int\limits_0^\infty v e^{i\zeta t} \, dt + \frac{\alpha^2}{\sqrt{(2\pi)}} \int\limits_0^\infty v e^{i\zeta t} \, dt \\ &= (\alpha^2 - i\zeta) V. \end{split}$$

Hence $V = A(\zeta) \cosh{\{\sqrt{(\alpha^2 - i\zeta)x\}} + B(\zeta) \sinh{\{\sqrt{(\alpha^2 - i\zeta)x\}}\}}$

When x=0,

$$V(0,\zeta) = rac{v_0}{\sqrt{(2\pi)}} \int\limits_0^\infty e^{i\zeta t} dt = -rac{v_0}{i\zeta\sqrt{(2\pi)}}.$$

Hence

$$A(\zeta) = -\frac{v_0}{i\zeta\sqrt{(2\pi)}}.$$

When x = l, $\frac{\partial V}{\partial x} = 0$. Hence

$$A(\zeta)\sinh\{\sqrt{(\alpha^2-i\zeta)l}\} + B(\zeta)\cosh\{\sqrt{(\alpha^2-i\zeta)l}\} = 0.$$

Hence

$$\begin{split} V &= -\frac{v_0}{i\zeta\sqrt{(2\pi)}}\bigg[\cosh\{\sqrt{(\alpha^2-i\zeta)x}\} - \frac{\sinh\{\sqrt{(\alpha^2-i\zeta)l}\}\sinh\{\sqrt{(\alpha^2-i\zeta)x}\}}{\cosh\{\sqrt{(\alpha^2-i\zeta)l}\}}\bigg] \\ &= -\frac{v_0}{i\zeta\sqrt{(2\pi)}}\frac{\cosh\{\sqrt{(\alpha^2-i\zeta)(x-l)}\}}{\{\cosh\sqrt{(\alpha^2-i\zeta)l}\}}. \end{split}$$

Hence for t>0 $v(x,t)=-\frac{v_0}{2\pi i}\int\limits_{-\infty}^{ia+\infty}\frac{\cosh\{\sqrt{(\alpha^2-i\zeta)(x-l)}\}}{\cosh\{\sqrt{(\alpha^2-i\zeta)l}\}}e^{-i\zeta t}\frac{d\zeta}{\zeta}.$

Here $\arg \sqrt{(\alpha^2 - i\zeta)}$ varies from $\frac{1}{4}\pi$ to $-\frac{1}{4}\pi$, and the integral is absolutely convergent if 0 < x < l.

10.10. Obtain the solution of †

$$\frac{\partial v}{\partial t} = \frac{\partial^2 v}{\partial r^2} + \frac{1}{r} \frac{dv}{\partial r}$$
 $(t > 0, r > a)$

such that v(r, 0) = 0 (r > a), v(a, t) = f(t).

† See Nicholson (2), Goldstein (2).

Suppose that $v(r,t) = O(e^{cr+ct})$. If

$$V(r,\zeta) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} v(r,t)e^{i\zeta t} dt \quad (\eta > c),$$

then

$$\begin{split} \sqrt{(2\pi)} & \left(\frac{\partial^2 V}{\partial r^2} + \frac{1}{r} \frac{\partial V}{\partial r} \right) = \int\limits_0^\infty \frac{\partial v}{\partial t} e^{i\zeta t} \, dt \ &= \left[v e^{i\zeta t} \right]_0^\infty - i\zeta \int\limits_0^\infty v e^{i\zeta t} \, dt \ &= -i\zeta \sqrt{(2\pi)} V. \end{split}$$

The solution of this may be written

$$V(r,\zeta) = A(\zeta)H_0^{(1)}\{r\sqrt{(i\zeta)}\} + B(\zeta)H_0^{(2)}\{r\sqrt{(i\zeta)}\},$$

where

$$H_0^{(1)}(z) = J_0(z) + iY_0(z), \qquad H_0^{(2)}(z) = J_0(z) - iY_0(z).$$

Let
$$\zeta = \xi + ik$$
, $\sqrt{(i\zeta)} = \zeta' = \xi' + i\eta'$. Then $\xi'^2 - \eta'^2 = -k$,

i.e. ζ' varies along a branch of this rectangular hyperbola, say the upper branch. On it

$$|H_0^{(1)}(r\zeta')| \sim \frac{Ae^{-r\eta'}}{\sqrt{(|r\zeta'|)}}, \qquad |H_0^{(2)}(r\zeta')| \sim \frac{Ae^{r\eta'}}{\sqrt{(|r\zeta'|)}}.$$

Since $V(r,\zeta) = O(e^{cr})$ in the upper half-plane, we must have $B(\zeta) = 0$.

Also, as $r \to a$,

$$egin{align} V(r,\zeta) &
ightarrow rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty f(t) e^{i\zeta t} \, dt = F(\zeta). \ A(\zeta) &= rac{F(\zeta)}{H^{(1)}\{q_*/(i\zeta)\}}, \end{aligned}$$

Hence

and the solution is

$$v(r,t) = rac{1}{\sqrt{(2\pi)}}\int\limits_{ik-\infty}^{ik+\infty} F(\zeta) rac{H_0^{(1)}\{r\sqrt{(i\zeta)}\}}{H_0^{(1)}\{a\sqrt{(i\zeta)}\}} e^{-i\zeta t} d\zeta,$$

where $\frac{1}{4}\pi \leqslant \arg \sqrt{(i\zeta)} \leqslant \frac{3}{4}\pi$.

For suitable functions f(t) we can make $k \to 0$, and obtain the solution in the form of integrals along the real axis.

such that

10.11. Obtain the solution of

$$\frac{\partial^2 v}{\partial x^2} + \frac{\partial^2 v}{\partial y^2} = 0$$
 $(x > 0, 0 < y < b)$
 $v = f(x)$ $(y = 0, 0 < x < \infty),$
 $v = 0$ $(y = b, 0 < x < \infty),$
 $v = 0$ $(x = 0, 0 < y < b).$

This is the problem of the steady distribution of heat in a semi-infinite strip with the edges kept at given temperatures.†

Formally, let

$$V_{e}(\xi,y) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} v(x,y) \sin \xi x \ dx.$$

Then

$$\begin{split} \frac{\partial^2 V_s}{\partial y^2} &= \sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty \frac{\partial^2 v}{\partial y^2} \sin \xi x \, dx \\ &= -\sqrt{\left(\frac{2}{\pi}\right)} \int_0^\infty \frac{\partial^2 v}{\partial x^2} \sin \xi x \, dx \\ &= -\sqrt{\left(\frac{2}{\pi}\right)} \left[\frac{\partial v}{\partial x} \sin \xi x\right]_0^\infty + \sqrt{\left(\frac{2}{\pi}\right)} \xi \int_0^\infty \frac{\partial v}{\partial x} \cos \xi x \, dx \\ &= \left[\sqrt{\left(\frac{2}{\pi}\right)} \xi v \cos \xi x\right]_0^\infty + \sqrt{\left(\frac{2}{\pi}\right)} \xi^2 \int_0^\infty v \sin \xi x \, dx \\ &= \xi^2 V_s. \end{split}$$

Hence

$$V_s = A(\xi) \cosh \xi y + B(\xi) \sinh \xi y.$$

Making
$$y \to 0$$
, $A(\xi) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} f(x) \sin \xi x \, dx$,

so that $A(\xi)$ is the sine transform of f(x), $A(\xi) = F_{\varepsilon}(\xi)$.

Putting y = b,

$$\begin{split} A(\xi) &\cosh \xi b + B(\xi) \sinh \xi b = 0, \\ B(\xi) &= - \coth \xi b \, F_s(\xi). \\ V_s &= F_s(\xi) (\cosh \xi y - \sinh \xi y \coth \xi b) \\ &= F_s(\xi) \frac{\sinh \xi (b-y)}{\sinh \xi b}, \end{split}$$

Hence

† Carslaw, Heat, § 45.

4362

and

$$v(x,y) = \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} F_s(\xi) \frac{\sinh \xi (b-y)}{\sinh \xi b} \sin \xi x \, d\xi.$$

In terms of f this gives

$$\begin{split} v(x,y) &= \frac{2}{\pi} \int_0^\infty f(u) \, du \int_0^\infty \frac{\sinh \xi (b-y)}{\sinh \xi b} \sin \xi x \sin \xi u \, d\xi \\ &= \frac{1}{2b} \sin \frac{\pi y}{b} \int_0^\infty f(u) \left(\frac{1}{\cos(b-y)\pi/b + \cosh(x-u)\pi/b} - \frac{1}{\cos(b-y)\pi/b + \cosh(x+u)\pi/b} \right) du. \end{split}$$

That this tends to $\frac{1}{2}\{f(x+0)+f(x-0)\}\$ wherever it exists, follows from Theorem 18; for

$$\frac{\sinh \xi(b-y)}{\sinh \xi b} = e^{-\xi y} - e^{-\xi b} \frac{\sinh \xi y}{\sinh \xi b},$$

and the contribution of the last term is clearly 0.

Suppose now that $v(x,y) = O(e^{cx})$ as $x \to \infty$, uniformly with respect to y, where $\pi/b < c < 2\pi/b$.

 $V(\zeta,y) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} v(x,y)e^{i\zeta x} dx,$ Let

where $c < \eta < 2\pi/b$. Then

$$\begin{split} \sqrt{(2\pi)} & \frac{\partial^2 V}{\partial y^2} = \int\limits_0^\infty \frac{\partial^2 v}{\partial y^2} e^{i\zeta x} \, dx \ &= -\int\limits_0^\infty \frac{\partial^2 v}{\partial x^2} e^{i\zeta x} \, dx \ &= -\left[\frac{\partial v}{\partial x} e^{i\zeta x}\right]_0^\infty + i\zeta \left[v e^{i\zeta x}\right]_0^\infty + \zeta^2 \int\limits_0^\infty v e^{i\zeta x} \, dx \ &= g(y) + \sqrt{(2\pi)} \zeta^2 V, \end{split}$$

where $y(y) = v_x(0, y)$. Hence

 $V(\zeta,y) = A(\zeta) \cosh \zeta y + B(\zeta) \sinh \zeta y + \frac{1}{\sqrt{(2\pi)}\zeta} \int_{-\infty}^{\infty} \sinh \zeta (y-u) g(u) du.$

Making $y \to 0$, we obtain, by dominated convergence,

$$A(\zeta) = rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty f(x) e^{i\zeta x} dx = F(\zeta),$$

and making $y \to b$,

$$0 = A(\zeta) \cosh \zeta b + B(\zeta) \sinh \zeta b + \frac{1}{\sqrt{(2\pi)\zeta}} \int_{0}^{b} \sinh \zeta (b-u) g(u) du.$$

Hence

$$V(\zeta,y) = F(\zeta) rac{\sinh \zeta (b-y)}{\sinh \zeta b} + rac{\sinh \zeta (y-b)}{\sqrt{(2\pi)\zeta \sinh \zeta b}} \int\limits_0^y \sinh \zeta u \, g(u) \, du - rac{\sinh \zeta y}{\sqrt{(2\pi)\zeta \sinh \zeta b}} \int\limits_u^b \sinh \zeta (b-u) \, g(u) \, du.$$

Hence if $c < a < 2\pi/b$,

$$\frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F(\zeta) \frac{\sinh \zeta(b-y)}{\sinh \zeta b} e^{-i\zeta x} d\zeta + \\ + \frac{1}{2\pi} \int_{0}^{y} g(u) du \int_{ia-\infty}^{ia+\infty} \frac{\sinh \zeta(y-b)\sinh \zeta u}{\zeta \sinh \zeta b} e^{-i\zeta x} d\zeta - \\ - \frac{1}{2\pi} \int_{y}^{b} g(u) du \int_{ia-\infty}^{ia+\infty} \frac{\sinh \zeta y \sinh \zeta(b-u)}{\zeta \sinh \zeta b} e^{-i\zeta x} d\zeta \\ = v(x,y) \quad (x>0), \qquad 0 \quad (x<0).$$

Replacing x by -x and subtracting, we find that, for x > 0, v(x, y)is equal to the above expression with $e^{-i\zeta x}$ replaced by $-2i\sin\zeta x$. In this form, if we replace a by 0 in the last two terms, we obtain 0; but if $a > \pi/b$, the pole at $\zeta = i\pi/b$ gives a residue term of the form

$$K\sin\frac{\pi y}{b}\sinh\frac{\pi x}{b}. (10.11.1)$$

In the first term we may insert the Fourier integral for $F(\zeta)$ and invert, by absolute convergence. The result (again allowing for the pole at $\zeta = i\pi/b$) is

where
$$\chi(u) = rac{1}{2b} \sin rac{\pi y}{b} igg(rac{1}{\cos rac{b-y}{b} \pi + \cosh rac{x-u}{b} \pi} - rac{1}{\cos rac{b-y}{b} \pi + \cosh rac{x+u}{b} \pi} igg) - rac{2}{b} \sin rac{\pi y}{b} \sinh rac{\pi x}{b} e^{-\pi u/b}.$$

We obtain the same solution as before, plus a term of the form (10.11.1), which is a solution of the corresponding problem with f = 0. The solution in this case is therefore not unique, unless we make some hypothesis which excludes such a term.

10.12. Obtain the solution of

$$\frac{\partial^2 y}{\partial t^2} = \frac{\partial^2 y}{\partial x^2} \quad (-\infty < x < \infty, \ t > 0),$$

such that $y(x, 0) = f(x), y_t(x, 0) = g(x)$.

This is the problem of the motion of an infinite string with a given initial displacement and velocity, y being the displacement at distance x along the string at time t.

For a formal solution, let

$$Y(\xi,t) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} y(x,t) e^{i\xi x} dx.$$

Then

$$egin{aligned} rac{\partial^2 Y}{\partial t^2} &= rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} rac{\partial^2 y}{\partial t^2} e^{i\xi x} \, dx \ &= rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} rac{\partial^2 y}{\partial x^2} e^{i\xi x} \, dx \ &= -\xi^2 Y, \end{aligned}$$

integrating by parts twice. Hence

$$Y = A(\xi)\cos \xi t + B(\xi)\sin \xi t$$

and clearly $A(\xi) = F(\xi), \, \xi B(\xi) = G(\xi)$. Hence

$$y(x,t) = rac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^{\infty} F(\xi) \cos \xi t \, e^{-i\xi x} \, d\xi + rac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^{\infty} rac{G(\xi)}{\xi} \sin \xi t \, e^{-i\xi x} \, d\xi.$$

The first term is

$$\frac{1}{4\pi} \int_{-\infty}^{\infty} \left\{ e^{-i\xi(x-t)} + e^{-i\xi(x+t)} \right\} d\xi \int_{-\infty}^{\infty} f(u)e^{i\xi u} du = \frac{1}{2} \{ f(x-t) + f(x+t) \},$$

and the second is

$$\begin{split} \frac{1}{2\pi} \int\limits_{-\infty}^{\infty} \frac{\sin \xi t}{\xi} e^{-i\xi x} \, d\xi \int\limits_{-\infty}^{\infty} g(u) e^{i\xi u} \, du \\ &= \frac{1}{\pi} \int\limits_{-\infty}^{\infty} g(u) \, du \int\limits_{0}^{\infty} \frac{\sin \xi t \cos \xi (x-u)}{\xi} \, d\xi = \frac{1}{2} \int\limits_{x-t}^{x+t} g(u) \, du. \end{split}$$

$$y(x,t) = \frac{1}{2} \{ f(x+t) + f(x-t) \} + \frac{1}{2} \int_{x-t}^{x+t} g(u) \ du,$$

the classical solution.

For a rigorous solution let $y(x,t) = O(e^{c|x|})$, and similarly for the partial derivatives. Let

$$Y_+(\zeta,t)=rac{1}{\sqrt{(2\pi)}}\int\limits_0^\infty y(x,t)e^{i\zeta x}\,dx, \qquad Y_-(\zeta,t)=rac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^0\ y(x,t)e^{i\zeta x}\,dx,$$

where Y_+ exists for $\eta > c$, Y_- for $\eta < -c$. Now

$$\begin{split} \sqrt{(2\pi)} \frac{\partial^2 Y_+}{\partial t^2} &= \int_0^\infty \frac{\partial^2 y}{\partial t^2} e^{i\zeta x} \, dx = \int_0^\infty \frac{\partial^2 y}{\partial x^2} e^{i\zeta x} \, dx \\ &= \left[\frac{\partial y}{\partial x} e^{i\zeta x} \right]_0^\infty - i\zeta [y e^{i\zeta x}]_0^\infty - \zeta^2 \int_0^\infty y e^{i\zeta x} \, dx \\ &= -\phi(t) + i\zeta \psi(t) - \zeta^2 \sqrt{(2\pi)} Y_+, \end{split}$$

where $\phi(t) = y_x(0, t)$, $\psi(t) = y(0, t)$. Hence

$$Y_{+} = A(\zeta)\cos\zeta t + B(\zeta)\sin\zeta t - \frac{1}{\sqrt{(2\pi)\zeta}}\int_{0}^{t}\sin\zeta (t-u)\{\phi(u)-i\zeta\psi(u)\}\,du,$$

and the initial conditions give $A(\zeta) = F_{+}(\zeta)$, $\zeta B(\zeta) = G_{+}(\zeta)$. Then

$$Y_{+}(\zeta,t) = F_{+}(\zeta)\cos\zeta t + \zeta^{-1}G_{+}(\zeta)\sin\zeta t + \chi(\zeta,t),$$

where χ is an integral function which tends to 0 for $\zeta = \xi + ik$, $\xi \to \pm \infty$. Similarly,

$$Y_{-}(\zeta,t) = F_{-}(\zeta)\cos\zeta t + \zeta^{-1}G_{-}(\zeta)\sin\zeta t - \chi(\zeta,t).$$

$$Now y(x,t) = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ia-\lambda}^{ia+\lambda} Y_{+}(\zeta,t) e^{-i\zeta x} d\zeta + \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{ib-\lambda}^{ib+\lambda} Y_{-}(\zeta,t) e^{-i\zeta x} d\zeta,$$

where a > c, b < -c. The contribution of $\chi(\zeta, t)$ to this is 0. The contribution of F is

$$\begin{split} \frac{1}{\sqrt{(2\pi)}} & \lim_{\lambda \to \infty} \int\limits_{ia-\lambda}^{ia+\lambda} F_{+}(\zeta) \frac{1}{2} \{e^{-i\zeta(x+t)} + e^{-i\zeta(x-t)}\} d\zeta + \\ & + \frac{1}{\sqrt{(2\pi)}} \lim\limits_{\lambda \to \infty} \int\limits_{ib-\lambda}^{ib+\lambda} F_{-}(\zeta) \frac{1}{2} \{e^{-i\zeta(x+t)} + e^{-i\zeta(x-t)}\} d\zeta = \frac{1}{2} \{f(x+t) + f(x-t)\}. \end{split}$$

The contribution of G_{+} is

$$\frac{1}{2\pi}\int_{0}^{\infty}g(u)\ du\int_{ia-\infty}^{ia+\infty}\frac{\sin\zeta t}{\zeta}e^{i\zeta(u-x)}\ d\zeta$$

(inverting by the bounded convergence of the ζ -integral). The ζ -integral is π if x-t < u < x+t, and otherwise 0. Similarly for G_{-} , and we obtain the same result as before.

10.13 Obtain the solution of †

$$\frac{\partial^2 y}{\partial t^2} - \frac{\partial^2 y}{\partial x^2} = 0 \quad (0 < x < l, \ t > 0)$$

such that y(0,t) = 0, y(l,t) = 0, y(x,0) = f(x), and $y_l(x,0) = 0$.

This is the problem of the vibration of an elastic string with fixed ends, y being the displacement of the string at distance x along the string at time t.

Suppose that y(x,t) and its derivatives are $O(e^{ct})$ for some c. Let

$$Y(x,\zeta) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} y(x,t)e^{i\zeta t} dt$$

for $\eta > c$. Then

$$\begin{split} \sqrt{(2\pi)} \frac{\partial^2 Y}{\partial x^2} &= \int_0^\infty \frac{\partial^2 y}{\partial x^2} e^{i\zeta t} \, dt = \int_0^\infty \frac{\partial^2 y}{\partial t^2} e^{i\zeta t} \, dt \\ &= \left[\frac{\partial y}{\partial t} e^{i\zeta t} \right]_0^\infty - i\zeta \int_0^\infty \frac{\partial y}{\partial t} e^{i\zeta t} \, dt \\ &= -i\zeta [y e^{i\zeta t}]_0^\infty - \zeta^2 \int_0^\infty y e^{i\zeta t} \, dt = i\zeta f(x) - \zeta^2 \sqrt{(2\pi)} Y. \end{split}$$

Hence

$$Y = A(\zeta)\cos\zeta x + B(\zeta)\sin\zeta x + \frac{i}{\sqrt{(2\pi)}}\int_{0}^{x}f(u)\sin\zeta(x-u)\ du.$$

The initial conditions give $Y(0,\zeta) = 0$, $Y(l,\zeta) = 0$. Hence $A(\zeta) = 0$, and

$$B(\zeta)\sin\zeta l + \frac{i}{\sqrt{(2\pi)}}\int_{0}^{t}\sin\zeta(l-u)f(u)\,du = 0.$$

Hence
$$Y = -\frac{i}{\sqrt{(2\pi)}} \frac{\sin \zeta x}{\sin \zeta l} \int_{0}^{l} f(u) \sin \zeta (l-u) \ du + \frac{i}{\sqrt{(2\pi)}} \int_{0}^{x} f(u) \sin \zeta (x-u) \ du$$

$$= -\frac{i}{\sqrt{(2\pi)}} \frac{\sin \zeta (l-x)}{\sin \zeta l} \int_{0}^{x} f(u) \sin \zeta u \ du - \frac{i}{\sqrt{(2\pi)}} \frac{\sin \zeta x}{\sin \zeta l} \int_{x}^{l} f(u) \sin \zeta (l-u) \ du.$$

Hence
$$y(x,t) = \frac{1}{2\pi i} \int_{ia-\infty}^{ia+\infty} \frac{\sin \zeta(l-x)}{\sin \zeta l} e^{-i\zeta t} d\zeta \int_{0}^{x} f(u)\sin \zeta u du + \frac{1}{2\pi i} \int_{ia-\infty}^{ia+\infty} \frac{\sin \zeta x}{\sin \zeta l} e^{-i\zeta t} d\zeta \int_{x}^{l} f(u)\sin \zeta(l-u) du. \quad (10.13.1)$$

If we replace t by t+2l and subtract, we introduce a factor $2i\sin\zeta l\,e^{-i\zeta l}$, and the resulting integrals tend to 0 when $a\to -\infty$, if t>0. Hence the solution has the period 2l, and we may suppose 0< t< 2l. We then write

$$\frac{1}{\sin \zeta l} = -2ie^{i\zeta l} + \frac{e^{2i\zeta l}}{\sin \zeta l},$$

and the contribution of the last term is seen to be 0 on making $a \to \infty$. The contribution of the first term may then be deduced from Fourier's theorem. For example, the first term in (10.13.1) gives

$$\frac{1}{4\pi}\int_{id-\infty}^{ia+\infty}\left\{e^{i\zeta(2l-x-t)}-e^{i\zeta(x-t)}\right\}d\zeta\int_{0}^{x}f(u)(e^{i\zeta u}-e^{-i\zeta u})du.$$

The first terms in each bracket give

$$-\frac{1}{2}f(2l-x-t)$$

if 2l-2x < t < 2l-x, and otherwise 0. The complete solution may be written $\frac{1}{2}\{f(x+t)+f(x-t)\},$

where f(x) is defined outside (0, l) by saying that it is odd and has the period 2l.

If f''(x) exists everywhere and is continuous, the whole process is plainly valid. In other cases the differential equation is not satisfied everywhere, and the given conditions are not strictly consistent.

Suppose, for example, that f(x) = x $(0 \le x \le \frac{1}{2}l)$, f(x) = l - x $(\frac{1}{2}l \le x \le l)$. Then f'(x) = 1 or -1, and f''(x) = 0 where it exists.

To cover a case of this kind we could restate the problem by assuming that, instead of the differential equation, y satisfies

$$\left(\frac{\partial y}{\partial x}\right)_{x=x_1} - \left(\frac{\partial y}{\partial x}\right)_{x=x_1} = \frac{\partial^2}{\partial t^2} \int_{x_1}^{x_1} y \ dx$$

for every t, for all but a finite number of values of x, and that $\partial y/\partial x$ is bounded in x for each t. Then

$$\begin{split} \left(\frac{\partial Y}{\partial x}\right)_{x=x_1} - \left(\frac{\partial Y}{\partial x}\right)_{x=x_1} &= \frac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty \left\{ \left(\frac{\partial y}{\partial x}\right)_{x=x_1} - \left(\frac{\partial y}{\partial x}\right)_{x=x_1} \right\} e^{i\zeta t} \, dt \\ &= \frac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty \left(\frac{\partial^2}{\partial t^2} \int\limits_{x_1}^{x_2} y \, dx \right) e^{i\zeta t} \, dt \\ &= -\frac{i\zeta}{\sqrt{(2\pi)}} \left[\int\limits_{x_1}^{x_2} y \, dx \, e^{i\zeta t} \right]_0^\infty - \frac{\zeta^2}{\sqrt{(2\pi)}} \int\limits_0^\infty e^{i\zeta t} \, dt \int\limits_{x_1}^{x_2} y \, dx \\ \left(\text{assuming that } \frac{\partial}{\partial t} \int\limits_{x_1}^{x_2} y \, dx \to 0 \text{ as } t \to 0 \right) \\ &= \frac{i\zeta}{\sqrt{(2\pi)}} \int\limits_{x_1}^{x_2} f(x) \, dx - \zeta^2 \int\limits_{x_1}^{x_2} Y \, dx. \end{split}$$

Hence $\frac{\partial^2 Y}{\partial x^2}$ exists, and is equal to $\frac{i\zeta f(x)}{\sqrt{(2\pi)}} - \zeta^2 Y$. The analysis then proceeds as before.

Another equation which may be solved in a similar way is

$$\frac{\partial^2 y}{\partial t^2} - \frac{\partial^2 y}{\partial x^2} = g \quad (0 < x < l),$$

where g is a constant, $y(x, 0) = g(lx - \frac{1}{2}x^2)$, y(0, t) = 0, $y_l(x, 0) = 0$.

10.14. The problem \ddagger of the waves on a plane sheet of water, caused by a disturbance of strength f(t) at a fixed point (the origin), depends on the solution of

$$\frac{\partial^2 \phi}{\partial t^2} = c^2 \left(\frac{\partial^2 \phi}{\partial r^2} + \frac{1}{r} \frac{\partial \phi}{\partial r} \right) \quad (r > 0, t > 0),$$

† Jeffreys, Operational Methods, p. 59.

1 Lamb, Hydrodynamics, p. 297.

where
$$\lim_{r\to 0} \left(-2\pi r \frac{\partial \phi}{\partial r}\right) = f(t) \quad (t>0).$$
 Let
$$\Phi(r,\zeta) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \phi(r,t) e^{i\zeta t} dt.$$
 Then
$$c^{2} \left(\frac{\partial^{2} \Phi}{\partial r^{2}} + \frac{1}{r} \frac{\partial \Phi}{\partial r}\right) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \frac{\partial^{2} \phi}{\partial t^{2}} e^{i\zeta t} dt.$$

If the surface of the water is initially flat and at rest, $\phi(r,0) = 0$ and $\phi_t(r,0) = 0$, and the usual partial integration shows that the right-hand side is $-\zeta^2\Phi$. Hence

$$\Phi(r,\zeta) = A(\zeta)H_0^{(1)}\left(\frac{r\zeta}{c}\right) + B(\zeta)H_0^{(2)}\left(\frac{r\zeta}{c}\right).$$

Since Φ must be bounded for $\mathbf{I}(\zeta) > 0$, $B(\zeta) = 0$. Also

$$F(\zeta) = rac{1}{\sqrt{(2\pi)}} \int\limits_0^{\infty} f(t) e^{i\zeta t} \, dt = \lim_{r o 0} \left\{ -\sqrt{(2\pi)}r \int\limits_0^{\infty} rac{\partial \phi}{\partial r} e^{i\zeta t} \, dt
ight\}$$
 $= \lim_{r o 0} \left(-2\pi r rac{\partial \Phi}{\partial r}
ight) = \lim_{r o 0} rac{2\pi r \zeta}{c} A(\zeta) H_1^{(1)} \left(rac{r \zeta}{c}
ight) = 2\pi A(\zeta).$ Hence $\Phi(r,\zeta) = rac{1}{2\pi} F(\zeta) H_0^{(1)} \left(rac{r \zeta}{c}
ight),$ $\phi(r,t) = rac{1}{(2\pi)!} \int\limits_{ia - \infty}^{ia + \infty} F(\zeta) H_0^{(1)} \left(rac{r \zeta}{c}
ight) e^{-i\zeta t} \, d\zeta$ $= rac{1}{4\pi^2} \int\limits_0^{\infty} f(u) \, du \int\limits_{ia - \infty}^{ia + \infty} H_0^{(1)} \left(rac{r \zeta}{c}
ight) e^{i\zeta(u - t)} \, d\zeta.$

The inner integral is 0 if t < u+r/c, and otherwise it is

$$\sqrt{\{(u-t)^2-r^2/c^2\}}$$
. Hence $\phi(r,t)=rac{1}{2\pi}\int\limits_{0}^{t-r/c}rac{f(u)\ du}{\sqrt{\{(t-u)^2-r^2/c^2\}}}=rac{1}{2\pi}\int\limits_{0}^{\cosh^{-1}ct/r}f\Big(t-rac{r}{c}\cosh\lambda\Big)\ d\lambda.$

10.15. Obtain the solution of ‡

$$\frac{\partial^2 u}{\partial x \partial y} = u \quad (x > 0, \ y > 0)$$

such that u(x,0) = a, u(0,y) = a.

$$U(\zeta,y) = rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty u(x,y) e^{i\zeta x} \, dx$$

for $\eta > c$. Then

$$egin{aligned} rac{\partial U}{\partial y} &= rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty rac{\partial u}{\partial y} e^{i\zeta x} \, dx \ &= rac{1}{\sqrt{(2\pi)}} iggl[rac{\partial u}{\partial y} rac{e^{i\zeta x}}{i\zeta} iggr]_0^\infty - rac{1}{i\zeta\sqrt{(2\pi)}} \int\limits_0^\infty rac{\partial^2 u}{\partial x \partial y} e^{i\zeta x} \, dx \ &= -rac{U}{i\zeta}, \end{aligned}$$

since (with sufficient continuity) $u_y(0,y) = 0$. Hence

$$U(\zeta, y) = A(\zeta)e^{iy/\zeta}.$$

Making $y \to 0$,

$$A(\zeta) = rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty a e^{i\zeta x} \, dx = -rac{a}{i\zeta\sqrt{(2\pi)}}.$$

Hence

$$u(x,y) = -\frac{a}{2\pi i} \int_{ik-\infty}^{ik+\infty} \frac{e^{iy|\zeta-ix\zeta}}{\zeta} d\zeta$$

$$= aI_0\{2\sqrt{(xy)}\}$$

by (7.13.9).

10.16. Differential-difference equations.† We shall illustrate the general method of solution by considering the simple special case

$$f'(x) = \frac{1}{2h} \{ f(x+h) - f(x-h) \}. \tag{10.16.1}$$

We shall first assume that $f(x) = O(e^{c|x|})$ for some positive c. It follows by repeated appeal to the equation that f(x) has derivatives of all orders, each of which is $O(e^{c|x|})$; and if f(x) satisfies the equation, so does $f^*(x) = f(x) - f(0) - xf'(0)$, and $f^*(0) = f^*(0) = 0$. Hence we may suppose without loss of generality that f(0) = f'(0) = 0.

Define $F_{+}(w)$, $F_{-}(w)$ as usual, for v > c, v < -c respectively. Then

$$F_+(w) = -rac{1}{\sqrt{(2\pi)w^2}}\int\limits_0^\infty f''(x)e^{ixw}\,dx = O\Bigl(rac{1}{|w|^3}\Bigr)$$

† Hilb (2), Titchmarsh (16).

as $|w| \to \infty$; and similarly for $F_{-}(w)$. Hence (1.3.4) and the formula obtained from it by differentiating under the integral sign are both valid; and (10.16.1) gives

$$\int_{-iw-\frac{e^{-ihw}-e^{ihw}}{2h}}^{ia+\infty}\left(-iw-\frac{e^{-ihw}-e^{ihw}}{2h}\right)F_{+}(w)e^{-ixw}\,dw+\ldots=0,$$

where +... indicates the corresponding term with b and F_{-} instead of a and F_{+} . Hence, by Theorem 141,

$$\left(\frac{\sin hw}{h}-w\right)F_{+}(w)=\chi(w), \qquad \left(\frac{\sin hw}{h}-w\right)F_{-}(w)=-\chi(w),$$

where $\chi(w)$ is regular for $b \leqslant v \leqslant a$, and $\chi(w) \to 0$ as $u \to \pm \infty$. Hence

$$f(x) = \frac{h}{\sqrt{(2\pi)}} \int_{ia\cdot x}^{ia+\infty} \frac{\chi(w)e^{-ixw}}{\sin hw - hw} dw - \frac{h}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} \frac{\chi(w)e^{-ixw}}{\sin hw - hw} dw.$$

This is the sum of the residues at poles in the strip b < v < a. There is a triple pole at the origin, giving a quadratic in x. The other zeros of the denominator give exponential terms. Hence

$$f(x) = A + Bx + Cx^2 + \sum_{\nu} C_{\nu} e^{-ixw_{\nu}},$$
 (10.16.2)

where A, B, C, C_{ν} are constants, and w_{ν} runs through zeros of $\sin hw - hw$ other than 0 such that $|\mathbf{I}(w_{\nu})| \leq c$.

If we do not assume that $f(x) = O(e^{c|x|})$, we can proceed as follows.

Let

$$F_{\alpha,\beta}(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\beta} f(x)e^{iwx} dx.$$

Then

$$\int_{\alpha}^{\beta} f(x+h)e^{iwx} dx = \int_{\alpha+h}^{\beta+h} f(x)e^{iw(x-h)} dx$$

$$= e^{-iwh} \sqrt{(2\pi)} \{F_{\alpha,\beta}(w) + F_{\beta,\beta+h}(w) - F_{\alpha,\alpha+h}(w)\},$$

and similarly with -h. Also

$$\int\limits_{lpha}^{eta}f'(x)e^{iwx}\,dx=f(eta)e^{iweta}-f(lpha)e^{iwlpha}-iw\sqrt{(2\pi)}F_{lpha,eta}(w).$$

On multiplying (10.16.1) by e^{iwx} and integrating over (α, β) , we thus obtain

$$\sqrt{(2\pi)ih^{-1}(\sin hw-hw)}F_{\alpha,\beta}(w)=\Phi_{\alpha}(w)-\Phi_{\beta}(w),$$

where

$$\Phi_{\alpha}(w) = f(\alpha)e^{iw\alpha} - \frac{\sqrt{(2\pi)}}{2h}\left\{e^{-iwh}F_{\alpha,\alpha+h}(w) - e^{iwh}F_{\alpha-h,\alpha}(w)\right\},\,$$

and similarly for $\Phi_{\beta}(w)$. Now

$$f(x) = rac{1}{\sqrt{(2\pi)}} \int\limits_{i\alpha = -\infty}^{i\alpha + \infty} F_{lpha,eta}(w) e^{-iwx} dw \quad (lpha < x < eta)$$

for any real a. Choose a so that no zero of $\sin hw - hw$ has the imaginary part a, e.g. take a positive and sufficiently small. Then

$$f(x) = \frac{h}{2\pi i} \int_{ia-\infty}^{ia+\infty} \frac{\Phi_{\alpha}(w)e^{-iwx}}{\sin hw - hw} dw - \frac{h}{2\pi i} \int_{ia-\infty}^{ia+\infty} \frac{\Phi_{\beta}(w)e^{-iwx}}{\sin hw - hw} dw.$$

It is easily verified that for a fixed β , and I(w) = v > 0,

$$\Phi_{\beta}(w) = O(e^{v(h-\beta)}).$$

Also we can choose a sequence of contours, e.g. the squares C_n with vertices at $2n\pi h^{-1}(\pm 1 \pm i)$, on which $|\sin hw - hw| > C|w|$. The usual process of contour integration then gives

$$\frac{h}{2\pi i}\int_{ia-r}^{ia+\infty}\frac{\Phi_{\beta}(w)e^{-iwx}}{\sin hw-hw}dw=\sum\frac{\Phi_{\beta}(w_{\nu})e^{-iw_{\nu}x}}{\cos hw_{\nu}-1}\quad(x<\beta-h),$$

where w_{ν} runs through the zeros of $\sin hw - hw$ in the upper half-plane. The coefficients in this series are independent of β , since

$$\begin{split} \frac{\partial \Phi_{\beta}(w_{\nu})}{\partial \beta} &= e^{iw_{\nu}\beta} \Big\{ f'(\beta) - \frac{1}{2h} f(\beta + h) + \frac{1}{2h} f(\beta - h) \Big\} + \\ &+ e^{iw_{\nu}\beta} \Big(iw_{\nu} + \frac{1}{2h} e^{-ihw_{\nu}} - \frac{1}{2h} e^{ihw_{\nu}} \Big) = 0. \end{split}$$

Similarly the term involving Φ_{α} gives a series depending on the zeros of $\sin hw - hw$ in the lower half-plane, convergent for $x > \alpha + h$, together with a quadratic in x arising from the triple zero at w = 0. The result is that (10.16.2) again holds, w_{ν} now running through all zeros of $\sin hw - hw$ except w = 0, and the series converging uniformly in any finite interval.

10.17. The equation †

$$f^{(n)}(x) + \sum_{\nu=0}^{n-1} a_{\nu} f^{(\nu)}(x+b_{\nu}) = g(x)$$

† For another method see Schmidt (1).

can be treated in a similar way. Consider the case in which each function is $O(e^{c|x|})$. By putting

$$f^*(x) = f(x) - f(0) - \dots - \frac{x^{n-1}}{(n-1)!} f^{(n-1)}(0)$$

we can reduce it to a similar problem in which

$$f(0) = f'(0) = \dots = f^{(n-1)}(0) = 0;$$

and repeated partial integration then shows that $w^n F_+(w)$ belongs to $L^2(ia-\infty,ia+\infty)$ if a>c, and similarly for $F_-(w)$. It follows as in the previous section that

$$\int_{ia-\infty}^{ia+\infty} \{F_{+}(w)K(w)-G_{+}(w)\}e^{-ixw} dw + \dots = 0,$$

$$K(w) = (-iw)^{n} + \sum_{i=0}^{n-1} a_{\nu}(-iw)^{\nu}e^{-ib_{\nu}w},$$

where

and the integrals are mean-square integrals. Hence, by Theorem 141,

$$F_{+}(w)K(w)-G_{+}(w)=\chi(w), \qquad F_{-}(w)K(w)-G_{-}(w)=-\chi(w)$$

where $\chi(w)$ is regular for $b \leqslant v \leqslant a$. Hence

$$f(x) = rac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} rac{G_{+}(w) + \chi(u_{i})}{K(w)} e^{-ixw} dw + rac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} rac{G_{-}(w) - \chi(w)}{K(w)} e^{-ixw} dw,$$

where a and b can be chosen so that all the zeros of K(w) in

$$-c \leqslant v \leqslant c$$
,

but no others, lie in $b \le v \le a$. The terms involving $\chi(w)$ can be calculated by the theorem of residues. The result is

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} \frac{G_{+}(w)}{K(w)} e^{-ixw} dw + \frac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} \frac{G_{-}(w)}{K(w)} e^{-ixw} dw + \sum_{ib-\infty} C_{ii} e^{-ixw} dw$$

where w_r runs through the zeros of K(w) in the strip $-c \le v \le c$, and C_r is a constant for simple zeros, a linear function of x for double zeros, and so on.

We have used L^2 theory in the proof, but the is no difficulty in avoiding it, e.g. by first integrating twice, so that all the integrals dealt with are absolutely convergent.

The problem can also be solved by the method of the last section in the case in which the functions are not $O(e^{c|x|})$.

10.18. Difference equations. A pure difference equation can be solved in the same way. Take, for example,

$$f(x+1) - f(x) = g(x)$$

with the usual assumptions about f and g. This is equivalent to

$$\int_{ia-\infty}^{ia+\infty} \{F_{+}(w)e^{-i(x+1)w} - F_{+}(w)e^{-ixw} - G_{+}(w)e^{-ixw}\} dw + \int_{ib-\infty}^{ib+\infty} \{F_{-}(w)e^{-i(x+1)w} - F_{-}(w)e^{-ixw} - G_{-}(w)e^{-ixw}\} dw = 0.$$
 Hence
$$F_{+}(w)(e^{-iw} - 1) - G_{+}(w) = \chi(w),$$

$$F_{-}(w)(e^{-iw} - 1) - G_{-}(w) = -\chi(w),$$

where $\chi(w)$ is regular for b < v < a. Hence

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} \frac{\chi(w) + G_{+}(w)}{e^{-iw} - 1} e^{-ixw} dw - \frac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} \frac{\chi(w) - G_{-}(w)}{e^{-iw} - 1} e^{-ixw} dw.$$

The terms involving $\chi(w)$ merely represent a function of period 1, which is obviously part of the solution. Hence the solution is

$$f(x) = f^*(x) + \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} \frac{G_+(w)}{e^{-iw}-1} e^{-ixw} dw + \frac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} \frac{G_-(w)}{e^{-iw}-1} e^{-ixw} dw,$$

where $f^*(x)$ is any function of period 1.

The formulae are valid in the L^2 sense if $g(x)e^{-c|x|}$ belongs to L^2 for some c. Under more special circumstances we can reduce it to other forms. If we expand $1/(e^{-iw}-1)$ in powers of e^{-iw} , we obtain formally $f(x) = f^*(x) - g(x) - g(x+1) - \dots$

which is obviously a solution if the series converges.

INTEGRAL EQUATIONS

11.1. Introduction. The most familiar form of integral equation is

$$f(x) = g(x) + \lambda \int_a^b k(x, y) f(y) dy,$$

where g(x) and k(x, y) are given functions, and f(x) is to be determined.

The equation can be solved by means of Fourier integrals in certain special cases; these are, roughly, the cases in which k(x, y) is of such a form that the integral is a 'resultant' of one kind or another.

We shall usually suppress the factor λ , which is of no importance in most of our results.

First take k(x,y) = k(x-y), and the limits $-\infty$, ∞ , so that the equation is

$$f(x) = g(x) + \int_{-\infty}^{\infty} k(x-y)f(y) dy \quad (-\infty < x < \infty).$$
 (11.1.1)

A formal solution may be obtained as follows. With our standard notation for transforms, we have

$$F(u) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \left\{ g(x) + \int_{-\infty}^{\infty} k(x-y)f(y) \, dy \right\} e^{ixu} \, dx$$

$$= G(u) + \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) \, dy \int_{-\infty}^{\infty} k(x-y)e^{ixu} \, dx$$

$$= G(u) + \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) \, dy \int_{-\infty}^{\infty} k(t)e^{i(y+t)u} \, dt$$

$$= G(u) + \sqrt{(2\pi)}F(u)K(u). \tag{11.1.2}$$

Hence

$$F(u) = \frac{G(u)}{1 - \sqrt{(2\pi)K(u)}},$$
 (11.1.3)

and the solution may be written

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \frac{G(u)}{1 - \sqrt{(2\pi)}K(u)} e^{-ixu} du.$$
 (11.1.4)

Also, (11.1.4) gives

$$f(x)-g(x) = rac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \left\{ rac{G(u)}{1-\sqrt{(2\pi)}K(u)} - G(u) \right\} e^{-ixu} du$$

$$= \int_{-\infty}^{\infty} G(u) rac{K(u)}{1-\sqrt{(2\pi)}K(u)} e^{-ixu} du.$$

$$M(u) = rac{K(u)}{1-\sqrt{(2\pi)}K(u)},$$

If

and m(x) is the transform of M(u), this gives

$$f(x) = g(x) + \int_{-\infty}^{\infty} g(t)m(x-t) dt$$
 (11.1.5)

as another formal solution.

The equation

$$f(x) = g(x) + \int_{0}^{\infty} f(y)k\left(\frac{x}{y}\right)\frac{dy}{y}$$
 (11.1.6)

may be reduced to the form (11.1.1), or solved similarly by Mellin integrals. The formal process is

$$\mathfrak{F}(s) = \mathfrak{G}(s) + \int_0^\infty x^{s-1} dx \int_0^\infty f(y) k \left(\frac{x}{y}\right) \frac{dy}{y}$$

$$= \mathfrak{G}(s) + \int_0^\infty f(y) \frac{dy}{y} \int_0^\infty k \left(\frac{x}{y}\right) x^{s-1} dx$$

$$= \mathfrak{G}(s) + \int_0^\infty f(y) y^{s-1} dy \int_0^\infty k(u) u^{s-1} du$$

$$= \mathfrak{G}(s) + \mathfrak{F}(s) \mathfrak{R}(s),$$

and the solution is

$$f(x) = \frac{1}{2\pi i} \int_{-\pi}^{c+i\infty} \frac{\mathfrak{G}(s)}{1 - \mathfrak{R}(s)} x^{-s} ds.$$
 (11.1.7)

This can also be reduced to a form corresponding to (11.1.5).

The simplest conditions under which the process is valid are given by

THEOREM 145. Let g(x) belong to $L^2(-\infty,\infty)$, and k(x) to $L(-\infty,\infty)$, and let the upper bound of K(u) be less than $1/\sqrt{(2\pi)}$. Then (11.1.4) gives a solution of the equation of the class L^2 , and any other solution of L^2 is equal to it almost everywhere.

Clearly $G(u)/\{1-\sqrt{(2\pi)}K(u)\}$ belongs to L^2 , so that (11.1.4) exists in the L^2 sense, and defines a function f(x) of L^2 ; as in § 3.13

$$h(x) = \int_{-\infty}^{\infty} k(x-y)f(y) dy$$

exists for almost all x, and belongs to L^2 ; and, by Theorem 65, if F, H, K are the transforms of f, h, k,

$$H(u) = \sqrt{(2\pi)}F(u)K(u) = \frac{\sqrt{(2\pi)}G(u)K(u)}{1-\sqrt{(2\pi)}K(u)}.$$

Hence the transform of g(x)+h(x) is

$$G(u) + \frac{\sqrt{(2\pi)}G(u)K(u)}{1 - \sqrt{(2\pi)}K(u)} = \frac{G(u)}{1 - \sqrt{(2\pi)}K(u)} = F(u).$$

$$g(x) + h(x) = f(x)$$

Hence

almost everywhere, i.e. the equation is satisfied.

Conversely, if f and g are L^2 , k is L, and (11.1.1) holds, then by Theorem 65 (11.1.2) holds, and hence (11.1.4). This proves the theorem.

If also k is L^2 , so are K and M, and (11.1.5) is equivalent to (11.1.4).

11.2. The homogeneous equation. We have shown that, so far as the class L^2 goes, the solution is unique. But under special circumstances there may be other solutions not of L^2 . If there were two solutions of (11.1.1), their difference would satisfy the homogeneous equation

$$f(x) = \int_{-\infty}^{\infty} k(x-y)f(y) dy \quad (-\infty < x < \infty).$$
 (11.2.1)

This equation is satisfied formally by putting $f(x) = e^{ax}$, if a is such that

$$\int_{-\infty}^{\infty} k(t)e^{-at} dt = 1. \qquad (11.2.2)$$

We shall next show that, under fairly simple conditions, the only solutions of the homogeneous equation are of this type.

THEOREM 146. Let 0 < c < c', and let $e^{c'|x|}k(x)$ belong to L and $e^{-c|x|}f(x)$ to $L^2(-\infty,\infty)$. Then, if f(x) satisfies (11.2.1), it is of the form

$$f(x) = \sum_{\nu} \sum_{p=1}^{q} C_{\nu,p} x^{p-1} e^{-iw_{\nu}x}, \qquad (11.2.3)$$

where w_r runs through all the zeros of $1-\sqrt{(2\pi)}K(w)$ such that $|I(w_r)| \leq c$, the $C_{\nu,p}$ are constants, and q is the order of multiplicity of the zero w_r .

It is at once verified that (11.2.3) is a solution of (11.2.1).

To prove the theorem we observe that, with the usual notation, K(w) is analytic for -c' < v < c', $F_{+}(w)$ is analytic for v > c, and $F_{-}(w)$ analytic for v < -c. For c < a < c'

$$\frac{1}{\sqrt{(2\pi)}} \int_{ia^{-\infty}}^{ia^{+\infty}} F_{+}(w) e^{-ixw} dw = f(x) \quad (x > 0), \qquad 0 \quad (x < 0)$$

in the mean-square sense; and by Theorem 65

$$\int_{0}^{\infty} k(x-y)f(y) \ dy = \int_{ia-\infty}^{ia+\infty} F_{+}(w)K(w)e^{-ixw} \ dw,$$

also in the mean-square sense. Similarly for $F_{-}(w)$, with a replaced by b, where -c' < b < -c. Hence (11.2.1) gives

$$\int\limits_{ia-\infty}^{ia+\infty}F_{+}(w)\{1-\sqrt{(2\pi)}K(w)\}e^{-ixw}\,dw + \int\limits_{ib-\infty}^{ib+\infty}F_{-}(w)\{1-\sqrt{(2\pi)}K(w)\}e^{-ixw}\,dw = 0$$

in the mean-square sense.

It therefore follows from Theorem 141 that $F_+(w)\{1-\sqrt{(2\pi)}K(w)\}$ and $F_-(w)\{1-\sqrt{(2\pi)}K(w)\}$ can both be continued throughout the strip b < v < a, and $F_+(w) = -F_-(w)$ in this strip. Hence $F_+(w)$ and $F_-(w)$ are regular in the strip except possibly for poles at the zeros of $1-\sqrt{(2\pi)}K(w)$.

We can now write

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w)e^{-ixw} dw - \frac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} F_{+}(w)e^{-ixw} dw,$$

and, since $F_+(w) \to 0$ as $u \to \pm \infty$, we can evaluate the right-hand side by the calculus of residues in the usual way. This proves the theorem.

In particular, the result is true if $k(x) = O(e^{-c'|x|})$ and $f(x) = O(e^{c|x|})$, where 0 < c < c'. In this case it can be obtained without recourse to L^2 theory. For, if $c < a < \eta$,

$$\lim_{T\to\infty}\int_{ia-T}^{ia+T}\frac{F_{+}(w)}{w-\zeta}\,dw=\lim_{T\to\infty}\frac{1}{\sqrt{(2\pi)}}\int_{ia-T}^{ia+T}\frac{dw}{w-\zeta}\int_{0}^{\infty}f(x)e^{ixw}\,dx$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} f(x) dx \lim_{T \to \infty} \int_{ia-T}^{ia+T} \frac{e^{ixw}}{w-\zeta} dw$$
$$= \sqrt{(2\pi)i} \int_{0}^{\infty} f(x)e^{ix\zeta} dx$$

by dominated convergence; and similarly

$$\lim_{T \to \infty} \int_{ia-T}^{ia+T} \frac{F_{+}(w)K(w)}{w-\zeta} dw = \frac{1}{2\pi} \int_{0}^{\infty} f(x) dx \int_{-\infty}^{\infty} k(y) dy \lim_{T \to \infty} \int_{ia-T}^{ia+T} \frac{e^{iw(x+y)}}{w-\zeta} dw$$

$$= i \int_{0}^{\infty} f(x) dx \int_{-x}^{\infty} k(y)e^{i\zeta(x+y)} dy$$

$$= i \int_{0}^{\infty} e^{i\zeta t} dt \int_{0}^{\infty} f(x)k(t-x) dx.$$

Hence

$$\int_{ia-\infty}^{ia+\infty} \frac{F_{+}(w)\{1-\sqrt{(2\pi)K(w)}\}}{w-\zeta} dw$$

$$= \sqrt{(2\pi)i} \int_{0}^{\infty} e^{i\zeta t} dt \Big\{ f(t) - \int_{0}^{\infty} f(x)k(t-x) dx \Big\}.$$

Similarly

$$\int_{ib-\infty}^{ib+\infty} \frac{F_{-}(w)\{1-\sqrt{(2\pi)K(w)}\}}{w-\zeta} dw = -\sqrt{(2\pi)i} \int_{0}^{\infty} e^{i\zeta t} dt \int_{-\infty}^{0} f(x)k(t-x) dx.$$

Hence the sum of the terms on the left is zero, the result of Theorem 141 again holds, and the theorem follows as before.†

11.3. Examples. (i) Let

$$g(x) = e^{-|x|}, \qquad k(x) = \lambda e^x \ (x < 0), \quad 0 \ (x > 0).$$

Then

$$G(u) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{-|x|+ixu} dx = \sqrt{\left(rac{2}{\pi}\right)} rac{1}{1+u^2},$$
 $K(u) = rac{\lambda}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{0} e^{x+ixu} dx = rac{\lambda}{\sqrt{(2\pi)}} rac{1}{1+iu}.$

† A solution under different conditions is given by Bochner (2).

The L^2 solution is therefore

$$f(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{e^{-ixu}}{(1-iu)(1-\lambda+iu)} du.$$

Suppose, for example, that $0 < \lambda < 1$. Then

$$f(x) = \frac{2}{2-\lambda}e^{-x} \quad (x \geqslant 0), \qquad = \frac{2}{2-\lambda}e^{(1-\lambda)x} \quad (x < 0).$$

This is plainly a solution, and so the only L^2 solution. There are similar solutions for other values of λ .

The equation is

$$f(x) = e^{-|x|} + \lambda e^x \int_{x}^{\infty} e^{-y} f(y) \, dy, \qquad (11.3.1)$$

and is reducible to differential equations. Let

$$\phi(x)=\int\limits_x^{\infty}e^{-\nu}f(y)\,dy, \qquad \phi'(x)=-e^{-x}f(x).$$
 Then, for $x>0, \qquad -\phi'(x)=e^{-2x}+\lambda\phi(x),$ so that $\qquad \phi(x)=rac{e^{-2x}}{2-\lambda}+Ce^{-\lambda x}.$ For $x<0, \qquad -\phi'(x)=1+\lambda\phi(x),$ so that $\qquad \phi(x)=-rac{1}{\lambda}+C'e^{-\lambda x}.$

Since $\phi(x)$ is continuous at x=0, $C'=C+\frac{1}{2-\lambda}+\frac{1}{\lambda}$. Hence

$$f(x) = \frac{2}{2-\lambda}e^{-x} + C\lambda e^{(1-\lambda)x} \quad (x > 0)$$
$$= \left(\frac{2}{2-\lambda} + C\lambda\right)e^{(1-\lambda)x} \quad (x < 0).$$

The complete solution therefore contains a term with an arbitrary constant; and in fact $f(x) = e^{(1-\lambda)x}$

is a solution of the homogeneous equation

$$f(x) = \lambda \int_{x}^{\infty} e^{x-y} f(y) dy, \qquad (11.3.2)$$

corresponding to the zero $w = i(1-\lambda)$ of the function

$$1 - \sqrt{(2\pi)}K(w) = 1 - \frac{\lambda}{1 + iw}.$$

(ii) Let
$$t(x) = \lambda e^{-|x|}$$
 ($\lambda < \frac{1}{2}$). Then

$$K(u) = \sqrt{\left(\frac{2}{\pi}\right)} \frac{\lambda}{1+x^2}, \qquad M(u) = \sqrt{\left(\frac{2}{\pi}\right)} \frac{\lambda}{1+x^2-2\lambda},$$
 $m(x) = \frac{\lambda}{\sqrt{(1-2\lambda)}} e^{-|x|\sqrt{(1-2\lambda)}},$

and, if g(x) is L^2 , the L^2 solution is

$$f(x) = g(x) + \frac{\lambda}{\sqrt{(1-2\lambda)}} \int_{-\infty}^{\infty} g(t)e^{-|t-x|\sqrt{(1-2\lambda)}} dt.$$

$$1 - \sqrt{(2\pi)}K(w) = \frac{1+w^2-2\lambda}{1+w^2},$$

Also

so that

$$Ae^{x\sqrt{(1-2\lambda)}}+Be^{-x\sqrt{(1-2\lambda)}}$$

is a solution of the homogeneous equation if $\lambda > 0$.

(iii) Consider the homogeneous equation in which $k(x) = e^{-\frac{1}{2}x^2}$.

(iv) Let
$$\lambda = \frac{1}{\pi}$$
, $k(x) = \frac{1}{\pi} \frac{1}{1+x^2}$, $g(x) = \frac{x}{1+x^2}$. Then
$$K(u) = \sqrt{\left(\frac{1}{2\pi}\right)} e^{-|u|}, \qquad G(u) = i \operatorname{sgn} u \, \pi e^{-|u|},$$

and the solution is

$$f(x) = rac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} rac{i\pi \operatorname{sgn} u \, e^{-|u|}}{1 - e^{-|u|}} e^{-ixu} \, du$$

$$= \sqrt{(2\pi)} \int\limits_{0}^{\infty} rac{\sin xu}{e^{u} - 1} \, du = \sqrt{(2\pi)} \left(rac{\pi}{2} \coth \pi x - rac{1}{2x} \right).$$

This just fails to come under the above conditions, and in fact f(x) is not L^2 .

(v) Let $\ddagger k(u) = \lambda/(1+u)$ in (11.1.6). Then $\Re(s) = \lambda \pi \csc s \pi$, and the solution is

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \frac{\mathfrak{G}(s)}{1-\lambda\pi \csc s\pi} x^{-s} ds,$$

or $f(x) = g(x) + \frac{1}{2\pi i} \int_{s-i\infty}^{c+i\infty} \frac{\lambda \pi}{\sin s\pi - \lambda \pi} \mathfrak{G}(s) x^{-s} ds.$

† Picard (1).

‡ A. C. Dixon (1).

If $\lambda \pi = \sin \alpha \pi$, where $0 < \alpha < \frac{1}{2}$, we have

$$\frac{1}{2\pi i}\int_{c-i\infty}^{c+i\infty}\frac{\sin\alpha\pi}{\sin s\pi-\sin\alpha\pi}u^{-s}\,du=\frac{\tan\alpha\pi}{\pi}\frac{u^{-\alpha}-u^{1+\alpha}}{1-u^2},$$

and, by Parseval's formula, the solution may be written

$$f(x) = g(x) + x \frac{\tan \alpha \pi}{\pi} \int_{0}^{\infty} g(y) \frac{(x/y)^{-1-\alpha} - (x/y)^{\alpha}}{y^{2} - x^{2}} dy$$

(vi) The homogeneous equation

$$f(x) = \lambda \int_{0}^{\infty} \frac{f(y)}{x+y} dy$$
 (11.3.3)

is reduced by the substitutions

$$x = e^{\xi}, \quad y = e^{\eta}, \quad e^{\frac{1}{4}\xi}f(e^{\xi}) = \phi(\xi)$$

$$\phi(\xi) = \lambda \int_{-2}^{\infty} \frac{\phi(\eta)}{2\cosh\frac{1}{2}(\xi - \eta)} d\eta.$$

to

The only solutions of this of the form $\phi(\xi) = O(e^{c|\xi|})$, $0 < c < \frac{1}{2}$, are exponentials. We have

$$k(\xi) = \frac{\lambda}{2\cosh\frac{1}{2}\xi}, \qquad K(w) = \frac{\lambda\sqrt{\pi}}{\sqrt{2\cosh\pi w}},$$
$$1 - \sqrt{(2\pi)}K(w) - 1 - \frac{\pi\lambda}{\cosh\pi w}.$$

and

This has an infinity of zeros, some of which may lie in $-\frac{1}{2} < v < \frac{1}{2}$, and give solutions. For example, if $\lambda = 1/\pi$, there is a double zero at w = 0, and

$$\phi(\xi) = A + B\xi$$

is the solution, i.e. $f(x) = \frac{A + B \log x}{\sqrt{x}}$

is the solution of (11.3.3).

That there are in fact no other solutions of any kind is proved by Hardy and Titchmarsh (3).

(vii) The homogeneous equation

$$x^{\alpha}f(x) = \frac{\lambda}{\Gamma(\alpha)} \int_{0}^{x} (x-y)^{\alpha-1}f(y) dy \quad (0 < \alpha < 1) \quad (11.3.4)$$

is reduced by the substitution

$$x = e^{\xi}, \quad y = e^{\eta}, \quad e^{\alpha \xi} f(e^{\xi}) = \phi(\xi)$$

to the form

$$\phi(\xi) = \frac{\lambda}{\Gamma(\alpha)} \int_{-\infty}^{\xi} (e^{\xi-\eta} - 1)^{\alpha-1} \phi(\eta) d\eta.$$

Here

$$k(\xi) = \frac{\lambda}{\Gamma(\alpha)} (e^{\xi} - 1)^{\alpha - 1} \quad (\xi \geqslant 0), \qquad 0 \quad (\xi < 0),$$

and

$$K(w) = \frac{\lambda}{\sqrt{(2\pi)}} \frac{\Gamma(1-iw-\alpha)}{\Gamma(1-iw)} \quad (v > -(1-\alpha)).$$

11.4. Various forms. Various other forms of equation are reducible to that just considered.

For example, considert

$$f(x) = g(x) + \int_{a}^{x} \frac{1}{y} k\left(\frac{y}{x}\right) f(y) dy.$$
 (11.4.1)

Putting $x = e^{-\xi}$, $y = e^{-\eta}$, and writing

$$f(e^{-\xi}) = \phi(\xi), \qquad g(e^{-\xi}) = \psi(\xi), \qquad k(e^{\xi}) = \kappa(\xi),$$

we obtain

$$\phi(\xi) = \psi(\xi) + \int_{\xi}^{\infty} \kappa(\xi - \eta)\phi(\eta) d\eta. \qquad (11.4.2)$$

This is of the standard form if $\kappa(\xi) = 0$ for $\xi > 0$.

Another related form is

$$g(x) = \int_{0}^{x} k \left(\frac{y}{x}\right) f(y) dy.$$
 (11.4.3)

If $f_1(x) = \int_0^x f(t) dt$, and k is an integral,

$$\int_{x}^{x} k \left(\frac{y}{x}\right) f(y) \ dy = k(1) f_1(x) - \frac{1}{x} \int_{x}^{x} k' \left(\frac{y}{x}\right) f_1(y) \ dy,$$

and if $k(1) \neq 0$ the equation is

$$f_1(x) = \frac{g(x)}{k(1)} + \int_0^x \frac{1}{y} \frac{1}{k(1)} \frac{y}{x} k' \left(\frac{y}{x}\right) f_1(y) dy.$$

This is of the same form as (11.4.1).

11.5. The equation with finite limits. Another equation of some interest is obtained by putting f(x) = 0, g(x) = 0, k(x) = 0, for x < 0, in (11.1.1). We obtain the equation

$$f(x) = g(x) + \int_{0}^{x} k(x-y)f(y) dy \quad (x > 0),$$
 (11.5.1)

considered by Doetsch† and Fock.‡ Theorem 145 of course still applies; but now there is a more general solution of the same type.

THEOREM 147. Let $g(x)e^{-cx}$ belong to $L^2(0,\infty)$, and $k(x)e^{-cx}$ to $L(0,\infty)$, for some positive c. Then there is just one solution f(x) of (11.5.1) such that $f(x)e^{-c'x}$ belongs to $L^2(0,\infty)$ for some positive c'; it is given by

 $f(x) = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{a-\lambda}^{a+\lambda} \frac{G(w)}{1 - \sqrt{(2\pi)}K(w)} e^{-ixw} dw \qquad (11.5.2)$

if a is sufficiently large.

The equation (11.5.1) is unchanged if we replace f(x), g(x), and k(x) by $f(x)e^{-ax}$, $g(x)e^{-ax}$, and $k(x)e^{-ax}$ respectively, and we may argue in terms of these functions; or, what comes to the same thing, we may apply the argument of § 11.1 to K(u+ia), etc., instead of to K(u). We have

$$|K(u+ia)| \leqslant \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} |k(x)|e^{-ax} dx < \frac{1}{\sqrt{(2\pi)}}$$

if a is sufficiently large. The solution then proceeds as before.

The solution (11.5.2) may also be written

$$f(x) = g(x) + 1.i.m. \int_{a}^{ia+\lambda} G(w) \frac{K(w)}{1 - \sqrt{(2\pi)K(w)}} e^{-ixw} dw. \quad (11.5.3)$$

Suppose that $k(x)e^{-cx}$ is also L^2 . Then K(w) is L^2 , and hence so is

$$M(w) = \frac{K(w)}{1 - \sqrt{(2\pi)K(w)}},$$
 (11.5.4)

and

$$m(x) = \frac{1}{\sqrt{(2\pi)}} \text{l.i.m.} \int_{i\sigma-1}^{i\sigma+\lambda} M(w)e^{-ixw} dw.$$

Making $a \to \infty$, it is seen that m(x) = 0 for x < 0. The solution can therefore be put in the form

$$f(x) = g(x) + \int_{0}^{x} g(y)m(x-y) dy. \qquad (11.5.5)$$

The relation (11.5.4) is equivalent to

$$m(x) = k(x) + \int_{0}^{x} k(t)m(x-t) dt;$$
 (11.5.6)

† Doetsch (1), (2).

1 Fock (1).

in fact it is at once verified that (11.5.5) is a solution of (11.5.1) if (11.5.6) holds and the inversions are justified.

(11.5.4) also gives

$$M(w) = \sum_{n=1}^{\infty} (2\pi)^{\frac{1}{4}n-\frac{1}{4}} \{K(w)\}^n,$$

and this is equivalent to

$$m(x) = \sum_{n=1}^{\infty} k^{(n)}(x),$$

where

$$k^{(1)}(x) = k(x), \qquad k^{(n)}(x) = \int_{0}^{x} k(t)k^{(n-1)}(x-t) dt.$$

This is the well-known Volterra form of the solution.†

It has been proved by Wiener that, if k(x) is $L(0,\infty)$, a necessary and sufficient condition that (11.5.6) should have a solution h(x) of $L(0,\infty)$ is that $1-\sqrt{(2\pi)}K(u)\neq 0$ for u real. This is bound up with Wiener's Tauberian theory, which we do not discuss here.

Examples. (i) Let $k(x) = \lambda e^x$ (x > 0), 0 (x < 0). Then

$$\sqrt{(2\pi)}K(w) = -rac{\lambda}{1+iw}, \qquad M(w) = -rac{1}{\sqrt{(2\pi)}}rac{\lambda}{1+\lambda+iw}, \ m(x) = \lambda e^{x(1+\lambda)}.$$

Hence the solution of

$$f(x) = g(x) + \lambda \int_{0}^{x} e^{x-y} f(y) dy$$

 $f(x) = g(x) + \lambda \int_{0}^{x} e^{(1+\lambda)(x-y)} g(y) dy.$

is

If $\phi(x) = \int_0^x e^{-y} f(y) \, dy$, the equation reduces to the differential equation $\phi'(x) - \lambda \phi(x) = e^{-x} g(x).$

This gives for f(x) the above solution, together with $Ae^{(\lambda+1)x}$; but A=0, since all the other terms vanish for x<0.

(ii) Let ||k(x)| be a finite sum of exponentials,

$$k(x) = Pe^{px} + Qe^{qx} + \dots \quad (x > 0).$$

Then

$$K(w) = -\frac{1}{\sqrt{(2\pi)}} \left\{ \frac{P}{p+iw} + \frac{Q}{q+iw} + \ldots \right\}.$$

|| E. T. Whittaker (1).

[†] See Goursat's Cours d'analyse, t. 3, § 548-9.

[‡] Paley and Wiener, Fourier Transforms, § 18.

Hence M(w) is a rational function, and may be written

$$M(w) = \frac{(p+iw)(q+iw)...}{(\alpha+iw)(\beta+iw)...}K(w),$$

where $i\alpha$, $i\beta$,... are the zeros of $1-\sqrt{(2\pi)}K(w)$. The calculus of residues then gives

 $m(x) = -\sum_{\beta} \frac{(p-\alpha)(q-\alpha)\dots}{(\beta-\alpha)(\gamma-\alpha)\dots} e^{\alpha x},$

since $\sqrt{(2\pi)}K(i\alpha) = 1$.

A similar expression may be obtained for the solution if k(x) is a polynomial.

(iii) Let
$$f(x) = k(x) = \lambda J_0(x)$$
. Then

$$G(w) = K(w) = \frac{\lambda}{\sqrt{(2\pi)}} \int_0^\infty J_0(x) e^{ixw} dx = \frac{\lambda}{\sqrt{(2\pi)}} \frac{1}{\sqrt{(1-w^2)}},$$

and the solution is

$$f(x) = \frac{\lambda}{2\pi} \int_{ia-\infty}^{ia+\infty} \frac{e^{-ixw}}{\sqrt{(1-w^2)-\lambda}} dw = \frac{\lambda}{2\pi i} \int_{a-i\infty}^{a+i\infty} \frac{e^{xs}}{\sqrt{(1+s^2)-\lambda}} ds \quad (a > 0)$$

$$= \frac{\lambda}{2\pi i} \int_{a-i\infty}^{a+i\infty} \frac{\sqrt{(1+s^2)-s}}{1-\lambda^2+s^2} e^{xs} ds + \frac{\lambda}{2\pi i} \int_{a-i\infty}^{a+i\infty} \frac{e^{xs}}{1-\lambda^2+s^2} ds + \frac{\lambda}{2\pi i} \int_{a-i\infty}^{a+i\infty} \frac{e^{xs}}{1-\lambda^2+s^$$

by (7.13.2), (7.13.3), and (7.13.8).

11.6. Another type. Another integral equation which can be solved formally by means of Fourier integrals is

$$g(x) = \int_{-\infty}^{\infty} k(x-y)f(y) \, dy.$$
 (11.6.1)

This gives formally

$$G(u) = rac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} e^{ixu} dx \int_{-\infty}^{\infty} k(x-y) f(y) dy$$

† Fock (1).

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) \, dy \int_{-\infty}^{\infty} k(x-y)e^{ixu} \, dx$$

$$= \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) \, dy \int_{-\infty}^{\infty} k(t)e^{i(t+y)u} \, dt$$

$$= \sqrt{(2\pi)} F(u)K(u). \tag{11.6.2}$$

Hence the solution is

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{G(u)}{K(u)} e^{-ixu} du.$$
 (11.6.3)

For this to be an actual solution K(u) has to satisfy special conditions.

THEOREM 148. Let g(x) belong to $L^2(-\infty,\infty)$, and k(x) to $L(-\infty,\infty)$. Then, in order that there should be a solution f(x) of $L^2(-\infty,\infty)$, it is necessary and sufficient that G(u)/K(u) should belong to $L^2(-\infty,\infty)$.

Suppose that g, k, and f belong to the given L-classes, and (11.6.1) holds. Then (11.6.2) holds, by Theorem 65, p. 90, and F is L^2 . Hence G/K is L^2 .

Conversely, if G/K is L^2 , then f, defined by (11.6.3), is L^2 , and, by Theorem 65, the transform of the right-hand side of (11.6.1) is

$$\sqrt{(2\pi)}K(u)\frac{1}{\sqrt{(2\pi)}}\frac{G(u)}{K(u)}=G(u).$$

Hence (11.6.1) holds.

A similar equation soluble in terms of Mellin transforms is

$$g(x) = \int_0^\infty k(xy)f(y) dy. \qquad (11.6.4)$$

This gives formally

$$\mathfrak{G}(s) = \int_{0}^{\infty} x^{s-1} dx \int_{0}^{\infty} k(xy)f(y) dy$$

$$= \int_{0}^{\infty} f(y) dy \int_{0}^{\infty} k(xy)x^{s-1} dx$$

$$= \int_{0}^{\infty} f(y)y^{-s} dy \int_{0}^{\infty} k(u)u^{s-1} du$$

$$= \mathfrak{F}(1-s)\mathfrak{R}(s).$$

Hence

$$\mathfrak{F}(s) = \frac{\mathfrak{G}(1-s)}{\mathfrak{R}(1-s)},$$

and the solution is

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \frac{\mathfrak{G}(1-s)}{\mathfrak{R}(1-s)} x^{-s} ds.$$
 (11.6.5)

11.7. Laplace's integral equation. This is

$$g(x) = \int_{0}^{\infty} f(y)e^{-xy} dy.$$
 (11.7.1)

A formal solution

$$f(x) = \frac{1}{2\pi i} \int_{\frac{1}{2}-i\infty}^{\frac{1}{2}+i\infty} \frac{6(1-s)}{\Gamma(1-s)} x^{-s} ds$$
 (11.7.2)

is given by (11.6.5). The equation can, however, be solved directly by Fourier's integral formula. This gives

$$g(ix) = \int_{0}^{\infty} f(y)e^{-ixy} dy,$$

and hence

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} g(iy)e^{ixy} dy \quad (x > 0),$$
 (11.7.3)

the right-hand side being zero for x < 0.

If g(x) is given originally for real x, the solution (11.7.3) involves an appeal to analytic continuation. The solution (11.7.2), with the usual definition of \mathfrak{G} , only involves explicitly g(x) for real x; but it contains the factor $1/\Gamma(1-s)$, which is exponentially large at infinity, and it seems difficult to justify it except by an argument involving analytic continuation. In fact the equation (11.7.1) can only be satisfied if g(x) has the values assumed on the real axis by an analytic function g(z) regular for x > 0, so that some reference to the analytic character of g(x) is almost inevitable.

We shall prove that a necessary and sufficient condition that (11.7.2) should exist in the mean-square sense, and define a solution of (11.7.1) belonging to $L^2(0,\infty)$, is that g(x) should have the values assumed on the real axis by an analytic function g(z), regular for $|\arg z| < \frac{1}{2}\pi$, and such that

$$\int_{0}^{\infty} |g(re^{i\theta})|^2 dr < K \tag{11.7.4}$$

for $-\frac{1}{2}\pi < \theta < \frac{1}{2}\pi$.

Suppose first that f(x) satisfies the equation and belongs to $L^2(0,\infty)$.

Plainly g(z) is regular for R(z) > 0, i.e. $|\arg z| < \frac{1}{2}\pi$. Now by Theorem 99, p. 131, we can write

$$f(u) = f_{(+)}(u) + f_{(-)}(u),$$

where $f_{(+)}(w)$ is regular for $\arg w > 0$, $f_{(-)}(w)$ for $\arg w < 0$. Then if $-\frac{1}{2}\pi < \theta < \frac{1}{2}\pi$,

$$g(re^{i heta}) = \int\limits_0^\infty e^{-re^{i heta}u} f_{(+)}(u) \ du + \int\limits_0^\infty e^{-re^{i heta}u} f_{(-)}(u) \ du.$$

In the first integral we can turn the line of integration through an angle $\frac{1}{2}\pi - \theta$, and in the second through $-\frac{1}{2}\pi - \theta$. We obtain

$$g(re^{i\theta}) = e^{i(\frac{\epsilon}{4}\pi - \theta)} \int\limits_0^\infty e^{-ir\rho} f_{(+)}(\rho e^{i(\frac{\epsilon}{4}\pi - \theta)}) \ d\rho \ + e^{-i(\frac{\epsilon}{4}\pi + \theta)} \int\limits_0^\infty e^{ir\rho} f_{(-)}(\rho e^{-i(\frac{\epsilon}{4}\pi + \theta)}) \ d\rho,$$

and since $f_{(+)}$ and $f_{(-)}$ belong to L^2 along every line $\arg w = \text{const}$, (11.7.4) follows.

Conversely, suppose that g(z) satisfies the above condition. We have

$$\mathfrak{G}(1-s)=\int\limits_{0}^{\infty}g(x)x^{-s}\,dx.$$

If t > 0, we rotate the line of integration through $-\frac{1}{2}\pi$, giving,

$$\mathfrak{G}(1-s) = -i \int_{0}^{\infty} g(-iy)(ye^{-\frac{1}{2}\pi i})^{-s} dy$$
$$= -ie^{\frac{1}{2}\pi i\sigma}e^{-\frac{1}{2}\pi i} \int_{0}^{\infty} g(-iy)y^{-\sigma-it} dy.$$

For $\sigma = \frac{1}{2}$ this is $e^{-\frac{1}{4}\pi t}$ multiplied by a function of $L^2(0,\infty)$; a similar argument with t < 0 and a rotation through $\frac{1}{2}\pi$ shows that $\mathfrak{G}(1-s)$ is $e^{-\frac{1}{4}\pi |t|}$ multiplied by a function of $L^2(-\infty,\infty)$. Also

$$|1/\Gamma(1-s)| = O(e^{\frac{1}{4}\pi|t|}).$$

Hence the integral in (11.7.2) exists in the mean-square sense. That the f(x) so defined satisfies the equation follows from Theorem 72.

Alternative forms of solution have been given by Widder (1), Palcy and Wiener, § 13.

11.8. Stieltjes's integral equation. If we iterate the previous equation, i.e. put

$$g(x) = \int_0^\infty f(y)e^{-xy} dy, \qquad h(x) = \int_0^\infty g(y)e^{-xy} dy,$$

we obtain formally

$$h(x) = \int_{0}^{\infty} e^{-xy} dy \int_{0}^{\infty} f(u)e^{-yu} du$$

$$= \int_{0}^{\infty} f(u) du \int_{0}^{\infty} e^{-(x+u)y} dy$$

$$= \int_{0}^{\infty} \frac{f(u)}{x+u} du, \qquad (11.8.1)$$

another integral equation of a similar type. This equation has been considered in connexion with Stieltjes's moment problem.†

Putting $x = e^{\xi}$, $y = e^{\eta}$, $e^{i\xi}h(e^{\xi}) = \psi(\xi)$, $e^{i\xi}f(e^{\xi}) = \phi(\xi)$, the equation becomes

$$\psi(\xi) = \int_{-\infty}^{\infty} \frac{\phi(\eta)}{2\cosh\frac{1}{2}(\xi - \eta)} d\eta. \tag{11.8.2}$$

This is of the form (11.6.1), with

$$k(\xi) = \frac{1}{2} \operatorname{sech} \frac{1}{2} \xi, \qquad K(u) = \sqrt{(\frac{1}{2}\pi)} \operatorname{sech} \pi u,$$

and the formal solution is

$$\phi(\xi) = \frac{1}{\pi \sqrt{(2\pi)}} \int_{-\infty}^{\infty} \Psi(u) \cosh \pi u \, e^{-i\xi u} \, du$$

$$= \frac{1}{(2\pi)^{\frac{1}{2}}} \int_{-\infty}^{\infty} \Psi(u) (e^{-i(\xi + i\pi)u} + e^{-i(\xi - i\pi)u}) \, du$$

$$= \frac{1}{2\pi} \{ \psi(\xi + i\pi) + \psi(\xi - i\pi) \}, \qquad (11.8.3)$$

or

$$f(x) = \frac{i}{2\pi} \{ h(xe^{i\pi}) - h(xe^{-i\pi}) \}.$$
 (11.8.4)

An appeal to analytic continuation is again obviously involved.

We shall show that a necessary and sufficient condition that (11.8.3) should define a solution of (11.8.2) belonging to $L^2(-\infty,\infty)$ is that $\psi(z)$ should be an analytic function, regular for $-\pi < y < \pi$, and that

$$\int_{-\infty}^{\infty} |\psi(x+iy)|^2 \, dy < K$$

for $-\pi < y < \pi$.

As in § 5.4, the condition implies that there are limit-functions $\psi(x+i\pi)$ and $\psi(x-i\pi)$ belonging to $L^2(-\infty,\infty)$.

If ϕ is L^2 and ψ is defined by (11.8.2), then $\psi(z)$ is plainly analytic for $-\pi < y < \pi$; and by Theorem 64,

$$\psi(z) = \sqrt{(\frac{1}{2}\pi)} \int_{-\infty}^{\infty} \frac{\Phi(u)}{\cosh \pi u} e^{-izu} du,$$

where Φ is the transform of ϕ . Hence

$$\int_{-\infty}^{\infty} |\psi(x+iy)|^2 dy = \pi^2 \int_{-\infty}^{\infty} \frac{\{\Phi(u)\}^2}{\cosh^2 \pi u} e^{2yu} du$$

$$\leqslant \pi^2 \int_{-\infty}^{\infty} \frac{\{\Phi(u)\}^2}{\cosh^2 \pi u} e^{2\pi |u|} du.$$

Hence the condition is necessary.

Conversely, if ψ is of the given form, then

$$e^{\pm\pi u}\Psi(u)=rac{1}{\sqrt{(2\pi)}}\int\limits_{-\infty}^{\infty}\psi(\xi\pm i\pi)e^{i\xi u}\,du$$

belong to $L^2(0,\infty)$ and $L^2(-\infty,0)$ respectively. Hence $\Psi(u)\cosh \pi u$ belongs to $L^2(-\infty,\infty)$, and (11.8.3) defines a function ϕ of L^2 . That it is a solution of (11.8.2) follows from Theorem 64.

In terms of the original functions, a necessary and sufficient condition for (11.8.1) to have a solution of L^2 is that $g(z) = g(re^{i\theta})$ should be analytic for $-\pi < \theta < \pi$, and that

$$\int\limits_0^\infty |g(re^{i\theta})|^2\,dr$$

should be bounded for $-\pi < \theta < \pi$.

That (11.8.4) is a solution of the original problem is easily verified, for the right-hand side of (11.8.1) is then

$$\frac{i}{2\pi}\int\limits_0^\infty \frac{h(ue^{i\pi})}{x+u}\,du - \frac{i}{2\pi}\int\limits_0^\infty \frac{h(ue^{-i\pi})}{x+u}\,du.$$

Rotating the line of integration of the second integral through 2π , and allowing for the residue at $u = xe^{i\pi}$, we see that this is equal to h(x).

11.9. Stieltjes's moment problem.† A note on Stieltjes's moment problem itself may be inserted here. The problem is to determine f(x) such that

$$\int_{0}^{\infty} x^{n} f(x) dx = c_{n} \quad (n = 0, 1, ...),$$

where $c_0, c_1,...$ are given.

Suppose that $f(x)e^{k\sqrt{x}}$ is $L(0,\infty)$ for some positive k. Let

$$\phi(s) = \sum_{n=0}^{\infty} \frac{(-1)^n c_n s^{2n}}{(2n)!} = \int_0^{\infty} f(x) \sum_{n=0}^{\infty} \frac{(-1)^n x^n s^{2n}}{(2n)!} dx$$
$$= \int_0^{\infty} f(x) \cos s \sqrt{x} dx = 2 \int_0^{\infty} \xi f(\xi^2) \cos s \xi d\xi.$$

The inversion is justified by the convergence of

$$\int_{0}^{\infty} |f(x)| \sum_{n=0}^{\infty} \frac{x^{n} |s|^{2n}}{(2n)!} dx = \int_{0}^{\infty} f(x) \cosh|s| \sqrt{x} dx,$$

provided that |s| < k. The final integral, however, converges if $s = \sigma + it$, -k < t < k. Hence $\phi(s)$ is an analytic function, regular in this strip, and $\phi(s) \to 0$ as $\sigma \to \pm \infty$ in the strip. Also

$$\xi f(\xi^2) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_0^{\lambda} \left(1 - \frac{s}{\lambda} \right) \phi(s) \cos s \xi \ ds$$

for almost all ξ . Hence f(x) is unique apart from sets of zero measure. To show that this is actually a solution, we have, if a > 0,

$$\frac{1}{2\pi i} \int_{ia-\infty}^{ia+\infty} \frac{e^{-is\xi}}{s^{2n+1}} ds = \frac{(-1)^{n+1}\xi^{2n}}{(2n)!} \quad (\xi > 0), \qquad 0 \quad (\xi < 0).$$

Hence

$$\int_{0}^{\infty} x^{n} f(x) dx = 2 \int_{0}^{\infty} \xi^{2n+1} f(\xi^{2}) d\xi$$

$$= \frac{(-1)^{n+1} 2^{n}!}{\pi i} \int_{-\infty}^{\infty} |\xi| f(\xi^{2}) d\xi \int_{ia-\infty}^{ia+\infty} \frac{e^{-is\xi}}{s^{2n+1}} ds$$

† See Hardy (7).

$$= \frac{(-1)^{n+1}2n!}{\pi i} \int_{ia-\infty}^{ia+\infty} \frac{ds}{s^{2n+1}} \int_{-\infty}^{\infty} |\xi| f(\xi^2) e^{-is\xi} d\xi$$

$$= \frac{(-1)^{n+1}2n!}{\pi i} \int_{ia-\infty}^{ia+\infty} \frac{\phi(s)}{s^{2n+1}} ds.$$

The inversion is justified by absolute convergence if n > 0, and by the bounded convergence of the s-integral if n = 0; in the latter case the final integral is $\lim_{\lambda \to \infty} \int_{ia-\lambda}^{ia+\lambda}$. The result is, of course, a case of Parseval's formula.

Now

$$\int_{-\infty}^{\infty} \frac{\phi(\sigma+ia)}{(\sigma+ia)^{2n+1}} d\sigma = \int_{-\infty}^{\infty} \frac{\phi(-\sigma'+ia)}{(-\sigma'+ia)^{2n+1}} d\sigma' = -\int_{-\infty}^{\infty} \frac{\phi(\sigma'-ia)}{(\sigma'-ia)^{2n+1}} d\sigma',$$

since ϕ is even. Hence

$$\frac{1}{\pi i} \int_{ia-\infty}^{ia+\infty} \frac{\phi(s)}{s^{2n+1}} \, ds = \frac{1}{2\pi i} \left(\int_{ia-\infty}^{ia+\infty} - \int_{-ia-\infty}^{-ia+\infty} \right) \frac{\phi(s)}{s^{2n+1}} \, ds = \frac{(-1)^{n+1} c_n}{2n!}$$

by the theorem of residues. The desired result therefore follows.

The method, of course, does not show whether a particular set of c_n correspond to a function f(x) of the class considered. For example, if $c_n = 1$ for every n, then $\phi(s) = \cos s$, which is not the transform of a function integrable in the ordinary sense. It is here that Stieltjes integrals become relevant.

If
$$c_n = 1/(n+1)$$
, then

$$\phi(s) = 2s^{-2}(s\sin s + \cos s - 1),$$

and

$$\begin{aligned} \xi f(\xi^2) &= \frac{1}{\pi} \int\limits_0^\infty \left(\frac{2 \sin s}{s} - \frac{1 - \cos s}{s^2} \right) \cos s \xi \, ds \\ &= \xi \quad (0 < \xi < 1), \qquad 0 \quad (\xi > 1). \end{aligned}$$

Since

$$\int_{0}^{\infty} x^{n} e^{-x^{\mu}\cos\alpha} \sin(x^{\mu}\sin\alpha) \ dx = \frac{1}{\mu} \Gamma\left(\frac{n+1}{\mu}\right) \sin\frac{(n+1)\alpha}{\mu}$$

if $\mu > 0$ and $0 < \alpha < \frac{1}{2}\pi$, the function

$$f(x) = e^{-x^{\mu}\cos\mu\pi}\sin(x^{\mu}\sin\mu\pi)$$

4362

satisfies

$$\int_{0}^{\infty} x^{n} f(x) dx = 0 \quad (n = 0, 1, ...)$$

for every value of μ less than $\frac{1}{2}$. The solution is therefore not unique if we merely assume that $f(x) = O(e^{-kx^{\mu}})$, where $\mu < \frac{1}{2}$.

11.10. Finite limits. The equation

$$g(x) = \int_{0}^{x} k(x-y)f(y) \, dy \quad (x > 0)$$
 (11.10.1)

is, formally, the particular case of (11.6.1) in which f(x) and k(x), and so also g(x), vanish for x < 0. The formal process of § 11.6 gives as before $G(w) = \sqrt{(2\pi)}F(w)K(w), \qquad (11.10.2)$

and the formal solution is

$$f(x) = \frac{1}{2\pi} \int_{ia-\infty}^{ia+\infty} \frac{G(w)}{K(w)} e^{-ixw} dw.$$
 (11.10.3)

As before, for this to be a solution, special conditions on K(w) or special relations between G(w) and K(w) are required.

THEOREM 149. Let $g(x)e^{-cx}$ belong to $L^2(0,\infty)$, and $k(x)e^{-cx}$ to $L(0,\infty)$. Then, in order that there should be a solution f(x) such that $f(x)e^{-cx}$ belongs to $L^2(0,\infty)$, it is necessary and sufficient that

$$\int_{-\infty}^{\infty} \left| \frac{G(u+iv)}{K(u+iv)} \right|^2 du \leqslant M,$$

where M is a constant independent of v, for all $v \ge c$.

We can replace f(x), g(x), and k(x) in (11.10.1) by $e^{-ax}f(x)$, $e^{-ax}g(x)$ and $e^{-ax}k(x)$ respectively, and the result follows from Theorems 148 and 95.

That the solution of (11.10.1), if it exists, is unique, can be proved more generally.

THEOREM 150. Let $f(x)e^{-cx}$ and $k(x)e^{-cx}$ belong to $L(0, \infty)$ for some positive c, and let

$$\int_0^x k(x-y)f(y)\ dy = 0 \quad (x>0).$$

Then at least one of k and f is null.

For (11.10.2) holds, the inversion in the proof being justified by absolute convergence; and now G(w) = 0. Hence either F(w) or K(w) is 0 for all w, and the corresponding f or k is null, by Theorem 14.

We shall next show that the same result holds without any restriction on the behaviour of the functions at infinity.† We use the following lemmas.

LEMMA α . Let $\phi(w)$ be regular in the upper half-plane,

$$\phi(w) = O(e^{k|w|}), \qquad |\phi(u)| \leqslant 1 \quad (w = u + iv),$$

and let $\phi(iv)$ be real. If $\epsilon > 0$, the connected region in which

$$|\phi(w)| \geqslant 1 + \epsilon$$
,

if it exists, contains arbitrarily large purely imaginary values of w.

Let w_1 be a point (if there is one) at which $|\phi(w_1)| \geqslant 1+\epsilon$, let D be the connected region containing w_1 in which $|\phi(w)| \geqslant 1+\epsilon$, and let D_1 and D_2 be the parts of D in the first and second quadrants. If the lemma is false, D_1 and D_2 meet at most along a finite stretch of the imaginary axis. Let $|\phi(w)| \leqslant m$ on this stretch. Since $\phi(w) = 1+\epsilon$ on the boundary of D, $|\phi(w)| \leqslant M = \max(1+\epsilon, m)$ on the boundary of D_1 , and so, by the Phragmén-Lindelöf theorem, throughout D_1 ; and similarly throughout D_2 .

But actually $m \leqslant 1+\epsilon$, so that $M=1+\epsilon$. For the function $\psi(w)=(w+i)^{-\eta}\phi(w)$, where $\eta>0$, satisfies $|\psi(w)|\leqslant 1+\epsilon$ on the boundary of D, and $\psi(w)\to 0$ as $|w|\to\infty$ in D. Hence $|\psi(w)|\leqslant 1+\epsilon$ throughout D. Hence

$$|\phi(w)| \leqslant (1+\epsilon)|w+i|^{\eta}$$

throughout D, and, making $\eta \to 0$, $|\phi(w)| \le 1 + \epsilon$. Since the reversed inequality also holds, $\phi(w) = C$, where $|C| = 1 + \epsilon$. This is inconsistent with $|\phi(u)| \le 1$, so that D must contain arbitrarily large purely imaginary values. Also since $\phi(iv)$ is real, the region

$$|\phi(1+w)| \geqslant 1+\epsilon$$

is symmetrical about the imaginary axis, and it is easily seen that two regions with the properties of the above D would have to overlap. Hence there is only one such connected region.

LEMMA β . Let F(w) and K(w) both have the properties of the above $\phi(w)$, and let $|F(w)K(w)| \leq e^{-\gamma v}$, where $\gamma > 0$, for all v > 0. Then there exist α and β such that $\alpha + \beta = \gamma$, and $|F(w)| \leq e^{-\alpha v}$, $|K(w)| \leq e^{-\beta v}$ for all v > 0.

Consider the regions D and D' in which

$$|F(w)e^{\alpha'w}| \geqslant 1+\epsilon, \qquad |K(w)e^{\beta'w}| \geqslant 1+\epsilon,$$

where α' and β' are any fixed real numbers whose sum is γ , and $\epsilon > 0$. We shall show that either D or D' is empty.

By applying Lemma α to $F(w)e^{\alpha'w}$ and $K(w)e^{\beta'w}$, we see that D and D', if they exist, both contain arbitrarily large purely imaginary values of w. Let iv_1 be a point of D, iv_2 a point of D' with $v_2 > v_1$, and iv_3 a point of D with $v_3 > v_2$. Since D is a connected region, and is symmetrical about the imaginary axis, there must be a closed curve joining iv_1 and iv_3 , lying entirely in D, surrounding iv_2 . On this $|F(w)e^{\alpha'w}| \geqslant 1+\epsilon$, and so

$$|K(w)e^{(\gamma-\alpha')iw}| \leq 1/(1+\epsilon).$$

This inequality therefore holds throughout the area enclosed by the curve, and in particular at iv_2 . This involves a contradiction, so that either D or D' is empty.

Suppose that, for some w_1 and w_2 ,

$$|F(w_1)| > e^{-\alpha' v_1}$$
 and $|K(w_2)| > e^{-\beta' v_2}$.

Then, for some positive ϵ ,

$$|F(w_1)|\geqslant (1+\epsilon)e^{-\alpha' v_1}, \qquad |K(w_2)|\geqslant (1+\epsilon)e^{-\beta' v_2}.$$

Since we have shown that this cannot be so, it follows that either $|F(w)| \leq e^{-\alpha' v}$ for all v > 0, or $|K(w)| \leq e^{-\beta' v}$ for all v > 0. Let α be the upper bound of values of α' for which the first inequality holds. If it held for all α' , F(w) would be identically zero; if it held for no α' , the second inequality would hold for all β' , and K(w) would be identically zero. Otherwise $0 < \alpha < \infty$, $|F(w)| \leq e^{-(\alpha - \epsilon)v}$ for all w and arbitrarily small ϵ , and so $|F(w)| \leq e^{-\alpha v}$. If $\alpha' = \alpha + \epsilon$, the second of the above alternatives holds, so that

$$|K(w)| \leqslant e^{-(\gamma-\alpha-\epsilon)v} = e^{-(\beta-\epsilon)v}$$

for all w, and so $|K(w)| \leqslant e^{-\beta v}$ for all w. This proves the lemma.

THEOREM 151. Let f and k belong to $L(0,\gamma)$, and let

$$g(x) = \int_0^x f(y)k(x-y) dy = 0$$

for almost all x in $(0, \gamma)$. Then f(x) = 0 for almost all x in $(0, \alpha)$, and k(x) = 0 for almost all x in $(0, \beta)$, where $\alpha + \beta = \gamma$.

We may suppose that

$$\frac{1}{\sqrt{(2\pi)}}\int\limits_0^\gamma |f(x)|\;dx\leqslant 1,\qquad \frac{1}{\sqrt{(2\pi)}}\int\limits_0^\gamma |k(x)|\;dx\leqslant 1,$$

and that f(x) and k(x) are null for x < 0 and $x > \gamma$. Then g(x) is null for $x < \gamma$ and $x > 2\gamma$, and

$$\int_{\gamma}^{2\gamma} |g(x)| dx \leqslant \int_{0}^{2\gamma} dx \int_{0}^{x} |f(y)k(x-y)| dy$$

$$= \int_{0}^{2\gamma} |f(y)| dy \int_{\gamma}^{2\gamma} |k(x-y)| dx$$

$$\leqslant \int_{0}^{2\gamma} |f(y)| dy \int_{0}^{2\gamma} |k(t)| dt \leqslant 2\pi.$$

As before, the transforms F(w) of f(x), etc., are related by

$$G(w) = \sqrt{(2\pi)}F(w)K(w);$$

hence

$$|F(w)K(w)| = rac{1}{2\pi} \left| \int\limits_{\gamma}^{2\gamma} g(x) e^{ixw} \ dx
ight| \leqslant rac{e^{-\gamma v}}{2\pi} \int\limits_{\gamma}^{2\gamma} |g(x)| \ dx \leqslant e^{-\gamma v}.$$

Hence, by Lemma β , either $F(w) \equiv 0$ or $K(w) \equiv 0$, or there exist α and β such that $\alpha + \beta = \gamma$, and $|F(w)| \leq e^{-\alpha v}$, $|K(w)| \leq e^{-\beta v}$ for v > 0. Now

$$\int_{0}^{\xi} f(x) dx = \frac{1}{\sqrt{(2\pi)}} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} F(w) \frac{e^{-i\xi w} - 1}{-iw} dw$$

by Theorem 22, and the ordinary method of integrating round a contour in the upper half-plane shows that this is 0 if $\xi < \alpha$. Hence f(x) is null in $(0, \alpha)$, and similarly k(x) is null in $(0, \beta)$.

THEOREM 152. If f and k are integrable over any finite interval, k is not null, and

$$\int_{0}^{x} f(y)k(x-y) dy = 0 \quad (0 < x < \infty),$$

then f is null in $(0,\infty)$.

By the previous theorem f is null in $(0, \alpha)$, where $\alpha + \beta = \gamma$, γ is arbitrarily large, β bounded.

11.11. Another example of an equation with finite limits ist

$$f(x) = \int_{0}^{1} k(t)f(x-t) dt, \qquad (11.11.1)$$

where k(t) belongs to $L^2(0, 1)$, and f(t) to L^2 over any finite interval. Here the integral represents a continuous function, so that f(x) is, in fact, continuous.

Let $f(x) = e^{cx}g(x)$. Then

$$g(x) = \int_0^1 k(t)e^{-ct}g(x-t) dt.$$

Taking c so large that $\int\limits_0^1 |k(t)| e^{-ct} \, dt < 1$, it follows that

$$|g(x)| < \max_{x-1 \le \xi \le x} |g(\xi)|,$$

and hence that g(x) is bounded as $x \to \infty$. Hence $f(x) = O(e^{cx})$. If we assume also that $f(x) = O(e^{c|x|})$ as $x \to -\infty$, the theory of § 11.2 applies, (11.11.1) being the particular case of (11.2.1) in which k(t) = 0 for t < 0 and for t > 1.

We can, however, prove without this assumption

THEOREM 153. The solution of (11.11.1) is

$$f(x) = \sum C_{\nu} e^{-iw_{\nu}x}, \qquad (11.11.2)$$

where w, runs through the zeros of

$$G(w) = 1 - \int_0^1 k(t)e^{iwt} dt$$

with $I(w_{\nu}) \leqslant c$, and C_{ν} is a constant at simple zeros, linear at double zeros, etc.

Let $F_{\alpha}(w) = \frac{1}{\sqrt{(2\pi)}} \int\limits_{\alpha}^{\infty} f(x)e^{iwx} dx \qquad (11.11.3)$

(cf. § 10.16). Then $F_{\alpha}(w)$ is regular for v > c, with the above c. The formal argument is then as follows. If a > c,

$$\frac{1}{\sqrt{(2\pi)}}\int_{ia=-\infty}^{ia+\infty}F_{\alpha}(w)e^{-iwx}\,dw=\begin{cases} f(x) & (x>\alpha),\\ 0 & (x<\alpha). \end{cases}$$
 (11.11.4)

† Schurer (1), Titchmarsh (16).

Hence, if $x > \alpha + 1$,

$$\int_{0}^{1} k(t)f(x-t) dt = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{\alpha}(w)e^{-iwx} dw \int_{0}^{1} k(t)e^{iwt} dt,$$

and (11.11.1) gives

$$\int_{ia-\infty}^{ia+\infty} F_{\alpha}(w)G(w)e^{-iwx} dw = 0 \quad (x > \alpha+1).$$

Multiplying by $e^{ix\zeta}$, where $\zeta = \xi + i\eta$, $\eta > a$, and integrating over $(\alpha + 1, \infty)$, we obtain

$$\int_{ia-\infty}^{ia+\infty} F_{\alpha}(w)G(w)\frac{e^{i(\zeta-w)(\alpha+1)}}{\zeta-w}\,dw=0.$$

The result may be justified by mean-square theory, as in Theorem 141.

Moving the integral to the parallel line through ia', where $a' > \eta$, we obtain

$$\int\limits_{ia'-\infty}^{ia'+\infty}F_{lpha}(w)G(w)rac{e^{i(\zeta-w)(lpha+1)}}{\zeta-w}\,dw=-2\pi iF_{lpha}(\zeta)G(\zeta).$$

The left-hand side is an analytic function of ζ , regular for $\eta < a'$. It therefore provides the analytic continuation of the right-hand side throughout $\eta < a'$. It follows that $F_{\alpha}(\zeta)$ is regular for $\eta < a'$, except possibly for poles at the zeros of $G(\zeta)$. Also

$$F_{\alpha}(\zeta)G(\zeta) = o(e^{-\eta(\alpha+1)})$$

as $\zeta \to \infty$ uniformly for $\eta \leqslant a < a'$. If the zeros of $G(\zeta)$ are separated by suitable contours on which $|G(\zeta)| > \text{const.}$, (11.11.2) follows on applying the usual contour integration to (11.11.4). The result certainly holds if k(t) is absolutely continuous near t = 1, and $k(1) \neq 0$; for then we can integrate by parts and obtain

$$G(w) = 1 - \frac{k(1)e^{iw}}{iw} + o\left(\frac{e^{-v}}{|w|}\right)$$

from which the result easily follows.

Finally, the C_{ν} are independent of α ; for example, at a simple zero of G(w),

$$\begin{split} C_{\nu} &= \frac{1}{\sqrt{(2\pi)G'(w_{\nu})}} \int\limits_{ia'-\infty}^{ia'+\infty} F_{\alpha}(w)G(w) \frac{e^{i(w_{\nu}-w)(\alpha+1)}}{w_{\nu}-w} \, dw, \\ \frac{\partial C_{\nu}}{\partial \alpha} &= -\frac{f(\alpha)e^{iw_{\nu}(\alpha+1)}}{2\pi G'(w_{\nu})} \int\limits_{ia'-\infty}^{ia'+\infty} \frac{G(w)e^{-iw}}{w_{\nu}-w} \, dw + \\ &+ \frac{ie^{iw_{\nu}(\alpha+1)}}{\sqrt{(2\pi)G'(w_{\nu})}} \int\limits_{ia'-\infty}^{ia'+\infty} F_{\alpha}(w)G(w)e^{-iw(\alpha+1)} \, dw. \end{split}$$

Each of these integrals is zero by Cauchy's theorem, the integrands being regular in the half-plane below the path of integration.

Similar methods can be applied to the solution of many other problems.†

11.12. Examples. The following example of (11.10.1) is considered by Bateman (6). A tradesman buys and sells various articles. It is assumed (i) that buying and selling are continuous processes, and that goods bought begin to be sold at once; (ii) that when the tradesman buys a new supply of any article, he buys just as much as he can sell in time T, the same for all such purchases; (iii) that the new supply sells uniformly during the time T.

The tradesman starts with a new supply of unit value, and it is required to find the law according to which purchases must be made if the value of the stock is to remain constant.

The amount of the original stock remaining after time t is k(t), where k(t) = 1 - t/T $(t \le T)$, 0 (t > T).

Suppose that articles of value $f(\tau)$ $\delta \tau$ are purchased in the interval of time between τ and $\tau + \delta \tau$. This stock is reduced by sales in such a way that the value of the remainder at time $t > \tau$ is

$$k(t-\tau)f(\tau)\delta\tau$$
.

The value at time t of the unsold stock due to purchases is therefore

$$\int_0^t k(t-\tau)f(\tau)\ d\tau.$$

† Busbridge (6), Cooper (2).

Hence f satisfies the integral equation

$$1-k(t)=\int_{0}^{t}k(t-\tau)f(\tau)\ d\tau.$$

Here

$$K(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{T} \left(1 - \frac{t}{T}\right) e^{itw} dt = \frac{1}{\sqrt{(2\pi)}} \left(-\frac{1}{iw} + \frac{1 - e^{iwT}}{Tw^2}\right),$$

and

$$egin{align} G(w) &= rac{1}{\sqrt{(2\pi)}} \int\limits_0^\infty e^{itw} \, dt - K(w) \ &= rac{1}{\sqrt{(2\pi)iw}} - K(w) = -rac{1}{\sqrt{(2\pi)}} rac{1 - e^{iwT}}{Tw^2}. \end{split}$$

We can take I(w) > 0, and the solution is

$$f(t) = -\frac{1}{2\pi} \int_{ia-\infty}^{ia+\infty} \frac{1 - e^{iwT}}{iwT + 1 - e^{iwT}} e^{-iwt} dw \quad (a > 0).$$

This can be expanded in various forms. If we move the line of integration to a parallel line through w = -ib, where b > 0, we obtain $-ib + \infty$

$$f(t) = \frac{2}{T} - \frac{1}{2\pi} \int_{-ih-\infty}^{-ih+\infty} \frac{1 - e^{iwt}}{iwT + 1 - e^{iwT}} e^{-iwt} dw,$$

and the last integral is exponentially small as $t \to \infty$. Further terms in the approximation arise from the zeros of the denominator.†

11.13. As another example we shall sum the series‡

$$f(x) = \sum_{n=1}^{\infty} n J_n(x) J_n(\xi). \tag{11.13.1}$$

We have $|J_n(x)| \leqslant 1$ for all n and x, and hence

$$|J_n(x)| = \left|\frac{x}{2n} \{J_{n-1}(x) + J_{n+1}(x)\}\right| \leqslant x$$

for $n \ge 1$ and $x \ge 0$. Also, for a fixed ξ , as $n \to \infty$

$$J_n(\xi) = O\{(\frac{1}{2}\xi)^n/n!\}.$$

Hence we may multiply (11.13.1) by $J_0(t-x)/x$ and integrate term-

† See also Goldstein (1).

1 See Watson, § 16.32.

by-term over (0, t). We obtain

$$\int_{0}^{t} \frac{f(x)}{x} J_{0}(t-x) dx = \sum_{n=1}^{\infty} n J_{n}(\xi) \int_{0}^{t} \frac{J_{n}(x)}{x} J_{0}(t-x) dx$$

$$= \sum_{n=1}^{\infty} J_{n}(\xi) J_{n}(t) = \frac{1}{2} J_{0}(t-\xi) - \frac{1}{2} J_{0}(\xi) J_{0}(t),$$

by (7.14.6) and the 'addition formula' for Bessel coefficients.†

This is an integral equation for f(x)/x of the form (11.10.1); by the above inequalities, f(x)/x is bounded, so that it is given by (11.10.3). Here

$$K(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} J_0(x) e^{ixw} dx = \frac{1}{\sqrt{(2\pi(1-w^2))}}$$

where u > 0, and the branch which is real and positive on the real axis is taken. Similarly,

$$egin{align} G(w) &= rac{1}{2\sqrt{(2\pi)}} \int\limits_0^\infty \{J_0(x-\xi) - J_0(x)J_0(\xi)\} e^{ixw} \ dx \ &= rac{1}{2\sqrt{(2\pi)}} rac{1}{iw} \int\limits_0^\infty \{J_1(x-\xi) - J_1(x)J_0(\xi)\} e^{ixw} \ dx \ \end{aligned}$$

on integrating by parts. Hence

$$\frac{f(x)}{x} = \frac{1}{4\pi} \lim_{\lambda \to \infty} \int_{ia-\lambda}^{ia+\lambda} \frac{\sqrt{(1-w^2)}}{iw} e^{-ixw} dw \int_{0}^{\infty} \{J_1(t-\xi) - J_1(t)J_0(\xi)\} e^{itw} dt,$$

where a > 0. Now

$$\begin{split} -\frac{1}{4\pi} \lim_{\lambda \to \infty} \int\limits_{ia-\lambda}^{ia+\lambda} e^{-ixw} \, dw \int\limits_{0}^{\infty} \{J_{1}(t-\xi) - J_{1}(t)J_{0}(\xi)\} e^{itw} \, dt \\ &= -\frac{1}{2} \{J_{1}(x-\xi) - J_{1}(x)J_{0}(\xi)\}, \end{split}$$

and the remainder is

$$\begin{split} \frac{1}{4\pi} \int\limits_{ia-\infty}^{u+\infty} & \frac{\sqrt{(1-w^2)+iw}}{iw} e^{-ixw} \, dw \int\limits_{0}^{\infty} \{J_1(t-\xi)-J_1(t)J_0(\xi)\} e^{itw} \, dt \\ & = \frac{1}{4\pi} \int\limits_{0}^{\infty} \{J_1(t-\xi)-J_1(t)J_0(\xi)\} \, dt \int\limits_{ia-\infty}^{ia+\infty} \frac{\sqrt{(1-w^2)+iw}}{iw} \, e^{i(t-x)w} \, dw. \end{split}$$

The inner integral is 0 if $t \ge x$ (by making $a \to \infty$). For t < x its \dagger See Watson, § 2.4.

derivative with respect to t is $2\pi J_1(t-x)/(t-x)$, by (7.13.8). Hence, on integrating the repeated integral by parts, we obtain

$$\begin{split} \frac{1}{2} \int_{0}^{x} \left\{ J_{0}(t-\xi) - J_{0}(t)J_{0}(\xi) \right\} \frac{J_{1}(t-x)}{t-x} dt \\ &= \frac{1}{2} \int_{0}^{x} J_{0}(t-\xi) \frac{J_{1}(t-x)}{t-x} dt - \frac{1}{2} J_{1}(x)J_{0}(\xi), \end{split}$$

by (7.14.6). Hence

$$\frac{f(x)}{x} = \frac{1}{2} \int_{0}^{x} J_0(t-\xi) \frac{J_1(t-x)}{t-x} dt - \frac{1}{2} J_1(x-\xi) = \frac{1}{2} \int_{0}^{\xi} J_0(t-\xi) \frac{J_1(t-x)}{t-x} dt,$$
 again by (7.14.6).

11.14. Abel's integral equation. This is

$$g(x) = \int_{0}^{x} (x-y)^{-\alpha} f(y) dy$$
 (0 < \alpha < 1),

and is of the form (11.10.1), with $k(x) = x^{-\alpha}$. Here

$$K(w) = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} x^{-\alpha} e^{ixw} dx = \frac{1}{\sqrt{(2\pi)}} (-iw)^{\alpha-1} \Gamma(1-\alpha),$$

where $(-iw)^{\alpha-1}$ is real on the positive imaginary axis. The formal solution is therefore

$$f(x) = \frac{1}{\sqrt{(2\pi)\Gamma(1-\alpha)}} \int_{i\alpha=-\infty}^{i\alpha+\infty} \frac{G(w)}{(-iw)^{\alpha-1}} e^{-ixw} dw.$$

If this is an L^2 solution, its integral is

$$f_1(x) = \frac{1}{\sqrt{(2\pi)}\Gamma(1-\alpha)} \int_{ia-\infty}^{ia+\infty} \frac{G(w)}{(-iw)^{\alpha-1}} \frac{1-e^{-ixw}}{iw} dw$$

$$= \frac{1}{2\pi\Gamma(1-\alpha)} \int_{ia-\infty}^{ia+\infty} \frac{e^{-ixw}-1}{(-iw)^{\alpha}} dw \int_{0}^{\infty} g(t)e^{iwt} dt$$

$$= \frac{1}{2\pi\Gamma(1-\alpha)} \int_{0}^{\infty} g(t) dt \int_{ia-\infty}^{ia+\infty} \frac{e^{iw(t-x)}-e^{iw'}}{(-iw)^{\alpha}} dw.$$

The inner integral is 0 if t > x (by making $a \to +\infty$). For 0 < t < x the contribution of e^{iut} is still 0, while the other part is (by deforming

(11.15.2)

the line of integration into the negative imaginary axis)

$$\int_{0}^{\infty} v^{-\alpha} e^{v(t-x)} (e^{i\pi\alpha} - e^{-i\pi\alpha}) (-i) \ dv = 2\sin\pi\alpha \ \Gamma(1-\alpha)(x-t)^{\alpha-1}.$$
Hence
$$f_{1}(x) = \frac{\sin\pi\alpha}{\pi} \int_{0}^{x} (x-t)^{\alpha-1} g(t) \ dt,$$

and

$$f(x) = \frac{\sin \pi \alpha}{\pi} \frac{d}{dx} \int_{0}^{x} (x-t)^{\alpha-1} g(t) dt,$$

the usual form of solution.†

11.15. An equation of Fox. Another equation of 'resultant' type is $_{\infty}$

 $f(x) = g(x) + \int_{-\infty}^{\infty} k(x+y)f(y) \, dy.$ (11.15.1)

This is equivalent to the equation considered by Fox (2). The solution is a little more elaborate. We have as before

$$F(u) = G(u) + \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} e^{ixu} dx \int_{-\infty}^{\infty} k(x+y)f(y) dy$$

$$= G(u) + \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) dy \int_{-\infty}^{\infty} k(x+y)e^{ixu} dx$$

$$= G(u) + \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} f(y) dy \int_{-\infty}^{\infty} k(t)e^{iu(t-y)} dt$$

$$= G(u) + \sqrt{(2\pi)}F(-u)K(u).$$

Changing the sign of u,

$$F(-u) = G(-u) + \sqrt{(2\pi)}F(u)K(-u),$$

and, eliminating F(-u),

$$F(u) = \frac{G(u) + \sqrt{(2\pi)G(-u)K(u)}}{1 - 2\pi K(u)K(-u)}.$$

Hence

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{\infty} \frac{G(u) + \sqrt{(2\pi)}G(-u)K(u)}{1 - 2\pi K(u)K(-u)} e^{-ixu} du.$$

The form actually considered by Fox is

$$f(x) = g(x) + \int_{0}^{\infty} k(ux)f(u) du, \qquad (11.15.3)$$

† See Bosanquet (1) for a direct study of the solution.

which is connected with (11.15.1) by obvious transformations. The corresponding analysis for this equation goes in terms of Mellin transforms, and the solution is

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \frac{\mathfrak{G}(s) + \mathfrak{R}(s)\mathfrak{G}(1-s)}{1 - \mathfrak{R}(s)\mathfrak{R}(1-s)} x^{-s} ds.$$
 (11.15.4)

THEOREM 154. Let g(x) belong to L^2 and k(x) to L, and let the upper bound of K(u)K(-u) be less than $1/2\pi$. Then the equation (11.15.1) has just one solution of L^2 , given by (11.15.2).

As in §11.1, f(x) belongs to L^2 and satisfies the equation. Also the difference between two solutions of L^2 satisfies

$$f(x) = \int_{-\pi}^{\infty} k(x+y)f(y) \, dy, \qquad (11.15.5)$$

and so its transform satisfies

$$F(u) = \sqrt{(2\pi)}F(-u)K(u). \tag{11.15.6}$$

Hence

$$F(-u) = \sqrt{(2\pi)}F(u)K(-u),$$

and so

$$F(u)F(-u)\{1-2\pi K(u)K(-u)\}=0.$$

Hence F(u) or F(-u) is 0 for almost all u. But, by (11.15.6), if F(-u) = 0, then F(u) = 0. Hence F(u) = 0 for almost all u, and hence f(x) = 0 for almost all x.

There are obvious extensions, e.g. we could simply say that $|1-2\pi K(u)K(-u)| \ge A > 0$.

EXAMPLES. (i) In (11.15.3) let

$$k(x) = \lambda \sqrt{\left(\frac{2}{\pi}\right)} \cos x.$$

Then

$$\Re(s) = \lambda \Gamma(s) \cos \frac{1}{2} s \pi,$$

and

$$\Re(s)\Re(1-s) = \lambda^2.$$

Hence, if $\lambda^2 \neq 1$, the solution is

$$f(x) = rac{1}{2\pi i(1-\lambda^2)} \int\limits_{c-i\,\infty}^{c+i\,\infty} \{\mathfrak{G}(s) + \lambda\mathfrak{G}(1-s)\Gamma(s)\cos{rac{1}{2}s\pi}\}x^{-s}\,ds$$

$$= rac{1}{1-\lambda^2}g(x) + rac{\lambda}{1-\lambda^2}\sqrt{rac{2}{\pi}} \int\limits_{0}^{\infty}g(u)\cos{xu}\,du.$$

This may be verified by Fourier's cosine formula.

(ii) In (11.15.3) let $k(x) = \pi^{-\frac{1}{2}}e^{-x}$ and

$$g(x) = \frac{\log(1+x)}{x}$$
 $(0 < x < 1), \frac{\log(1+x)}{x} - \frac{\pi}{x}$ $(x > 1).$

Then

$$\Re(s) = \pi^{-\frac{1}{2}}\Gamma(s), \qquad \Im(s) = \frac{\pi}{1-s}\left(\frac{1}{\sin s\pi}-1\right),$$

and the solution is

$$f(x) = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \left\{ \frac{\pi}{s-1} - \sqrt{\pi} \, \Gamma(s-1) \right\} x^{-s} \, ds$$

$$= \sqrt{\pi} \frac{1 - e^{-x}}{x} \quad (0 < x < 1), \qquad \sqrt{\pi} \frac{1 - e^{-x}}{x} - \frac{\pi}{x} \quad (x > 1).$$

11.16. 'Dual' integral equations. In some problems the unknown function satisfies one integral equation over part of the range $(0,\infty)$, and a different equation over the rest of the range.

For example, \dagger let $v(\rho, z)$ be the potential of a flat circular electrified disk of conducting material, its centre being at the origin, and its axis along the axis of z. The potential satisfies the differential equation

$$\frac{\partial^2 v}{\partial \rho^2} + \frac{1}{\rho} \frac{\partial v}{\partial \rho} + \frac{\partial^2 v}{\partial z^2} = 0.$$
 (11.16.1)

$$V(u,z) = \int_{0}^{\infty} \rho v(\rho,z) J_0(\rho u) \, d\rho \quad (z > 0).$$
 (11.16.2)

Then

$$rac{\partial^2 V}{\partial z^2} = \int\limits_0^\infty
ho rac{\partial^2 v}{\partial z^2} J_0(
ho u) \ du = - \int\limits_0^\infty \left(
ho rac{\partial^2 v}{\partial
ho^2} + rac{\partial v}{\partial
ho}\right) J_0(
ho u) \ d
ho,$$

and

$$\int_{0}^{\infty} \rho \frac{\partial^{2} v}{\partial \rho^{2}} J_{0}(\rho u) d\rho = - \int_{0}^{\infty} \frac{\partial v}{\partial \rho} \{ J_{0}(\rho u) + \rho u J_{0}'(\rho u) \} d\rho.$$

Hence

$$egin{aligned} rac{\partial^2 V}{\partial z^2} &= \int\limits_0^\infty rac{\partial v}{\partial
ho}
ho u J_0'(
ho u) \ d
ho = -u \int\limits_0^\infty v \{J_0'(
ho u) +
ho u J_0''(
ho u)\} \ d
ho \ &= u^2 \int\limits_0^\infty v
ho J_0(
ho u) \ d
ho = u^2 V. \ &V &= A(u) e^{-uz} + B(u) e^{uz}. \end{aligned}$$

Hence

† Riemann-Weber, 1, § 134.

and plainly B(u) = 0. Hence, by Hankel's theorem,

$$v(
ho,z)=\int\limits_0^\infty uA(u)e^{-uz}J_0(
ho u)\ du.$$

Taking the radius of the disk to be unity, the boundary conditions are

$$v = \text{const.}$$
 $(z = 0, \ 0 < \rho < 1);$ $\frac{\partial v}{\partial z} = 0 \ (z = 0, \ \rho > 1).$

Hence, writing uA(u) = f(u), f(u) must satisfy

$$\int_{0}^{\infty} f(u)J_{0}(\rho u) du = g(\rho) \quad (0 < \rho < 1), \tag{11.16.3}$$

$$\int_{0}^{\infty} f(u)uJ_{0}(\rho u) du = 0 \qquad (\rho > 1), \qquad (11.16.4)$$

where, in the above case, $g(\rho)$ is a constant.

To solve these equations formally†, apply Parseval's formula for Mellin transforms to the left-hand sides. We obtain

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s) \frac{2^{-s} \Gamma(\frac{1}{2} - \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}s)} \rho^{s-1} ds = g(\rho) \quad (0 < \rho < 1),$$
(11.16.5)

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \mathfrak{F}(s) \frac{2^{1-s} \Gamma(1-\frac{1}{2}s)}{\Gamma(\frac{1}{2}s)} \rho^{s-2} ds = 0 \qquad (\rho > 1), \quad (11.16.6)$$

where 0 < k < 1. Putting

$$\mathfrak{F}(s) = \frac{2^s \Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{8} - \frac{1}{8}s)} \chi(s), \tag{11.16.7}$$

the equations become

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{2}+\frac{1}{2}s)} \chi(s) \rho^{s-1} ds = g(\rho) \quad (0 < \rho < 1),$$
(11.16.8)

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(1-\frac{1}{2}s)}{\Gamma(\frac{1}{2}-\frac{1}{2}s)} \chi(s) \rho^{s-1} ds = 0 \qquad (\rho > 1).$$
 (11.16.9)

In this form the Γ -function factor in (11.16.8) has no poles or zeros for $\sigma > 0$, and that in (11.16.9) has no poles or zeros for $\sigma < 1$.

† See also Busbridge (2), Copson (2).

Multiplying (11.16.8) by ρ^{-u} , where $\sigma - u > 0$, and integrating over (0, 1), we obtain

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{2}+\frac{1}{2}s)} \frac{\chi(s)}{s-w} ds = \int_{0}^{1} g(\rho)\rho^{-w} d\rho = \mathfrak{G}(1-w) \quad (u < k).$$

Moving the line of integration to $\sigma = k' < u$,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k'+i\infty} \frac{\Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{2}+\frac{1}{2}s)} \frac{\chi(s)}{s-w} ds = \mathfrak{G}(1-w) - \frac{\Gamma(\frac{1}{2}w)}{\Gamma(\frac{1}{2}+\frac{1}{2}w)} \chi(w).$$

The left-hand side is regular for u > k', and so for u > 0. Hence so is the right-hand side. Hence so also is

$$\chi(w) = \frac{\Gamma(\frac{1}{2} + \frac{1}{2}w)}{\Gamma(\frac{1}{2}w)} \mathfrak{G}(1-w).$$

Hence (assuming suitable conditions at infinity)

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \left\{ \chi(s) - \frac{\Gamma(\frac{1}{2} + \frac{1}{2}s)}{\Gamma(\frac{1}{2}s)} \mathfrak{G}(1-s) \right\} \frac{ds}{s-w} = 0 \quad (u < k).$$
(11.16.10)

Similarly, multiplying (11.16.9) by ρ^{-u} , where $\sigma - u < 0$, and integrating over $(1, \infty)$, we obtain

$$\frac{1}{2\pi i} \int_{k'=1/2}^{k'+i\infty} \frac{\Gamma(1-\frac{1}{2}s)}{\Gamma(\frac{1}{2}-\frac{1}{2}s)} \frac{\chi(s)}{s-w} ds = 0 \quad (u > k').$$

We conclude as before that $\{\Gamma(1-\frac{1}{2}s)/\Gamma(\frac{1}{2}-\frac{1}{2}s)\}\chi(s)$, and so $\chi(s)$, is regular for $\sigma < 1$. Hence

$$\frac{1}{2\pi i}\int_{0}^{k'+i\infty}\frac{\chi(s)}{s-w}\,ds=0\quad (u>k').$$

Moving the line of integration from k' to k > u,

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\chi(s)}{s-w} \, ds = \chi(w) \quad (u < k). \tag{11.16.11}$$

From (11.16.10), (11.16.11)

$$\chi(w) = \frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}s)}{\Gamma(\frac{1}{2}s)} \frac{\mathfrak{G}(1-s)}{s-w} ds \quad (u < k),$$
(11.16.12)

and this, with (11.16.7) and Mellin's inversion formula for f(x), gives the solution.

If $g(\rho) = v_0 = \text{const.}$, then

$$\mathfrak{G}(1-s) = v_0 \int_0^1 \rho^{-s} \, ds = \frac{v_0}{1-s},$$

$$\chi(w) = \frac{v_0}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}s)}{(1-s)\Gamma(\frac{1}{2}s)} \frac{ds}{s-w} = \frac{v_0}{\sqrt{\pi}(1-w)}$$

(from the pole at s=1). Hence

$$\mathfrak{F}(s) = \frac{v_0}{\sqrt{\pi(1-s)}} \frac{2^s \Gamma(\frac{1}{2}s)}{\Gamma(\frac{1}{2} - \frac{1}{2}s)} = \frac{v_0 2^{s-1} \Gamma(\frac{1}{2}s)}{\sqrt{\pi} \Gamma(\frac{3}{2} - \frac{1}{2}s)},$$
and, by (7.9.6),
$$f(u) = \frac{2v_0 \sin u}{\pi}.$$
 (11.16.13)

Hence

$$egin{aligned} v &= rac{2v_0}{\pi} \int\limits_0^\infty e^{-zu} J_0(
ho u) rac{\sin u}{u} \, du \ &= rac{2v_0}{\pi} rcsin iggl\{ \sqrt{\{(
ho-1)^2 + z^2\} + \sqrt{\{(
ho+1)^2 + z^2\}}} iggr\}, \end{aligned}$$

the solution obtained by Weber.

The pair of equations†

$$\int_{0}^{\infty} y^{\alpha} f(y) J_{\nu}(xy) \, dy = g(x) \quad (0 < x < 1), \qquad (11.16.14)$$

$$\int_{0}^{\infty} f(y) J_{\nu}(xy) \, dy = 0 \qquad (x > 1) \qquad (11.16.15)$$

can be solved in a similar way. They are equivalent to

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)} \chi(s)x^{s-1-\alpha} ds = g(x) \quad (0 < x < 1),$$

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}\alpha - \frac{1}{2}s)} \chi(s)x^{s-1} ds = 0 \quad (x > 1),$$
where
$$\mathfrak{F}(s) = 2^{s-\alpha} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}a - \frac{1}{2}s)} \chi(s).$$

where

† See King (1).

Multiplying (11.16.16) by $x^{\alpha-w}$, where $\sigma-u>0$, and integrating over (0,1), we obtain

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)} \frac{\chi(s)}{s-w} ds = \int_{0}^{1} g(x)x^{\alpha-w} dx = \mathfrak{G}(\alpha-w+1).$$

Moving the line of integration to $\sigma = k' < u$, we obtain

$$\frac{1}{2\pi i} \int_{k'-i\infty}^{k'+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)} \frac{\chi(s)}{s - w} ds$$

$$= -\frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}w)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}w)} \chi(w) + \mathfrak{G}(\alpha - w + 1).$$

Hence the right-hand side is regular for u > 0, and we deduce as before that

$$\frac{1}{2\pi i} \int_{k-i\infty}^{k+i\infty} \left\{ \chi(s) - \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)} \mathfrak{G}(\alpha - s + 1) \right\} \frac{ds}{s - w} = 0 \quad (u < k).$$

From (11.16.17) we deduce (11.16.11) as before. Hence

$$\begin{split} \chi(w) &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)} \frac{\mathfrak{G}(\alpha + 1 - s)}{s - w} \, ds \\ &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)} \, ds \int\limits_{0}^{1} g(\lambda) \lambda^{\alpha - s} \, d\lambda \int\limits_{0}^{1} \mu^{s - w - 1} \, d\mu \\ &= \frac{1}{2\pi i} \int\limits_{0}^{1} g(\lambda) \lambda^{\alpha} \, d\lambda \int\limits_{0}^{1} \mu^{-w - 1} \, d\mu \int\limits_{k-i\infty}^{k+i\infty} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu - \frac{1}{2}\alpha + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)} \left(\frac{\lambda}{\mu}\right)^{-s} \, ds \\ &= \frac{2}{\Gamma(\frac{1}{2}\alpha)} \int\limits_{0}^{1} g(\lambda) \lambda^{\alpha} \, d\lambda \int\limits_{\lambda}^{1} \mu^{-w - 1} \left(\frac{\lambda}{\mu}\right)^{\nu - \alpha + 1} \left(1 - \frac{\lambda^{2}}{\mu^{2}}\right)^{\frac{1}{4}\alpha - 1} \, d\mu \\ &= \frac{2}{\Gamma(\frac{1}{2}\alpha)} \int\limits_{0}^{1} \mu^{-w - 1} \, d\mu \int\limits_{0}^{\mu} g(\lambda) \lambda^{\alpha} \left(\frac{\lambda}{\mu}\right)^{\nu - \alpha + 1} \left(1 - \frac{\lambda^{2}}{\mu^{2}}\right)^{\frac{1}{4}\alpha - 1} \, d\lambda \\ &= \frac{2}{\Gamma(\frac{1}{2}\alpha)} \int\limits_{0}^{1} \mu^{\alpha - w} \, d\mu \int\limits_{0}^{1} g(\rho\mu) \rho^{\nu + 1} (1 - \rho^{2})^{\frac{1}{4}\alpha - 1} \, d\rho. \end{split}$$

Hence

$$\begin{split} f(x) &= \frac{1}{2\pi i} \int\limits_{k-i\infty}^{k+i\infty} \frac{2^{s-\alpha} \Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}\alpha - \frac{1}{2}s)} \chi(s) x^{-s} \, ds \\ &= \frac{1}{\pi i \Gamma(\frac{1}{2}\alpha)} \int\limits_{0}^{1} \mu^{\alpha} \, d\mu \int\limits_{0}^{1} g(\rho\mu) \rho^{\nu+1} (1-\rho^{2})^{\frac{1}{2}\alpha-1} \, d\rho \times \\ &\qquad \qquad \times \int\limits_{k-i\infty}^{k+i\infty} 2^{s-\alpha} \frac{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}s)}{\Gamma(\frac{1}{2} + \frac{1}{2}\nu + \frac{1}{2}\alpha - \frac{1}{2}s)} (\mu x)^{-s} \, ds \\ &= \frac{(2x)^{1-\frac{1}{2}\alpha}}{\Gamma(\frac{1}{2}\alpha)} \int\limits_{0}^{1} \mu^{1+\frac{1}{2}\alpha} J_{\nu+\frac{1}{2}\alpha}(\mu x) \, d\mu \int\limits_{0}^{1} g(\rho\mu) \rho^{\nu+1} (1-\rho^{2})^{\frac{1}{2}\alpha-1} \, d\rho. \end{split}$$

For this form of the solution to hold we must suppose that $\alpha > 0$; the previous equations correspond to $\nu = 0$, $\alpha = -1$.

As an example, let $\alpha = 1$, $\nu = 0$, g(x) = 1; the solution is

$$f(x) = \frac{2}{\pi} \left(\frac{\sin x}{x^2} - \frac{\cos x}{x} \right).$$

11.17. The method of Hopf and Wiener.† A method of Hopf and Wiener for solving the homogeneous equation

$$f(x) = \int_{0}^{\infty} k(x-y)f(y) \, dy \quad (0 < x < \infty)$$
 (11.17.1)

will now be given. It depends on the following lemma.

LEMMA. Let $\phi(w)$ be an analytic function, regular in the strip -1 < v < 1, and let

$$\int_{-\infty}^{\infty} |\phi(u+iv)|^2 du < K = K(\alpha)$$

in any interior strip $-1 < -\alpha \le v \le \alpha < 1$ (so that, in particular, by the lemma of §5.4, $\phi(u+iv) \to 0$ as $u \to \pm \infty$ uniformly in any interior strip).

In any interior strip $-1 < -\beta \le v \le \beta < 1$, $1-\phi(w)$ has only a finite number of zeros. If they are $w_1,...,w_n$, we can write

$$1 - \phi(w) = \frac{\phi_1(w)}{\phi_2(w)}(w - w_1)...(w - w_n), \qquad (11.17.2)$$

† Wiener and Hopf (1); Hopf, Radiative Equilibrium, Chap. IV; Paley and Wiener, Fourier Transforms, Chap. IV.

where $\phi_1(w)$ is regular and free from zeros in $v \leq \beta$, $\phi_2(w)$ is regular and free from zeros in $v \geq -\beta$, and, in their respective half-planes of regularity,

$$|\phi_1(w)| > K|w|^{-\frac{1}{4}n-k}, \qquad |\phi_2(w)| > K|w|^{\frac{1}{4}n-k}, \qquad (11.17.3)$$

where k is a positive integer depending on ϕ .

Let
$$\psi(w) = \{1 - \phi(w)\} \frac{(w^2 + 1)^{\frac{1}{2}n}}{(w - w_1)...(w - w_n)} \left(\frac{w - i}{w + i}\right)^k$$

where $(w^2+1)^{\frac{1}{2}n}$ is that single-valued branch in the strip $-\beta \leqslant v \leqslant \beta$ which behaves like w^n for large w, and where k is an integer still to be determined. Then $\psi(w) \to 1$ as $u \to \pm \infty$. Hence we can choose k so that the variation of $\log \psi(w)$ along the whole strip is 0. Having fixed k, let $\log \psi(w)$ denote the branch which tends to 0 as $u \to \pm \infty$. Since

$$\psi(w) = \{1 - \phi(w)\} \Big\{ 1 + O\Big(\frac{1}{|w|}\Big) \Big\},$$

 $|\log \psi(w)|$ belongs to L^2 uniformly in the strip. Hence

$$\log \psi(w) = \frac{1}{2\pi i} \int_{-i\gamma - \infty}^{-i\gamma + \infty} \frac{\log \psi(z)}{z - w} dz - \frac{1}{2\pi i} \int_{i\gamma - \infty}^{i\gamma + \infty} \frac{\log \psi(z)}{z - w} dz$$
$$= \chi_1(w) - \chi_2(w) \qquad (-\gamma < v < \gamma),$$

where $0 < \beta < \gamma < 1$, but $\gamma - \beta$ is so small that no zeros of $\psi(w)$ lie in $\beta < v \le \gamma$. Now $\chi_1(w)$ is regular for $v > -\gamma$, and regular and bounded for $v \ge -\beta$; and similarly χ_2 for $v \le \beta$. Since

$$1-\phi(w) = \frac{e^{\chi_1(w)}}{e^{\chi_2(w)}} \frac{(w-i)^{-\frac{1}{2}n-k}}{(w+i)^{\frac{1}{2}n-k}} (w-w_1)...(w-w_n),$$

the result now follows.

Suppose now that f(x) is a function which satisfies (11.17.1) for all real x, and let $k(x) = O(e^{-|x|})$, or more generally let $e^{\lambda |x|}k(x)$ belong to $L^2(-\infty,\infty)$ for all $\lambda < 1$; and

$$f(x) = O(e^{c|x|}) \quad (0 < c < 1).$$

Then $F_{+}(w)$ is regular for v > c, $F_{-}(w)$ is regular for v < -c, and K(w) is regular for -1 < v < 1. Now

$$\int_{0}^{\infty} k(x-y)f(y) \ dy = \int_{-\infty}^{\infty} k(x-y)f_{1}(y) \ dy$$
(where $f_{1}(y) = f(y) \ (y > 0)$, $0 \ (y < 0)$)
$$= \int_{ia-\infty}^{ia+\infty} F_{+}(w)K(w)e^{-ixw} \ dw \quad (c < a < 1),$$

by Theorem 64. The equation (11.17.1) therefore gives

Hence each integrand is regular throughout the strip $-a \leqslant v \leqslant a$, and in this strip

$$F_{+}(w)\{1-\sqrt{(2\pi)}K(w)\}+F_{-}(w)=0.$$

If $w_1,..., w_n$ are the zeros of $1-\sqrt{(2\pi)}K(w)$ in $-a \le v \le a$, by the lemma

 $F_{+}(w)\frac{\phi_{1}(w)}{\phi_{2}(w)}(w-w_{1})...(w-w_{n})+F_{-}(w)=0,$

where ϕ_1 and ϕ_2 have the properties stated in the lemma. We can write this $F_{-}(w) = F_{-}(w)$

 $\frac{F_{+}(w)}{\phi_{2}(w)}(w-w_{1})...(w-w_{n}) = -\frac{F_{-}(w)}{\phi_{1}(w)},$

and here the left-hand side is regular for $v \ge -a$, the right-hand side for $v \le a$. Hence each side is an integral function, and by (11.17.3) this must be a polynomial of degree not exceeding $\frac{1}{2}n-k$. Hence

$$F_{+}(w) = \frac{\phi_{2}(w)P(w)}{(w-w_{1})...(w-w_{n})},$$

where P(w) is a polynomial. Hence

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{i\pi}^{i\pi+\infty} \frac{\phi_2(w)P(w)}{(w-w_1)...(w-w_n)} e^{-ixw} dw$$

satisfies the original equation (and vanishes for x < 0).

As a simple example, let

$$k(x) = \lambda e^{-|x|} \quad (0 < \lambda < \frac{1}{2}),$$
 $K(w) = \frac{\lambda}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{-|x|+ixw} \, dx = \frac{2\lambda}{\sqrt{(2\pi)}} \frac{1}{1+w^2},$ $1-\sqrt{(2\pi)}K(w) = 1 - \frac{2\lambda}{1+w^2} = \frac{w^2-(2\lambda-1)}{w^2+1}.$

The roots are $w = \pm \sqrt{(2\lambda - 1)} = w_1, w_2$, and

$$1 - \sqrt{(2\pi)}K(w) = \frac{(w - w_1)(w - w_2)}{(w - i)(w + i)},$$

$$\phi_1(w) = \frac{1}{w-i}, \qquad \phi_2(w) = w+i,$$

$$F_{+}(w) = rac{(w+i)P(w)}{w^2 - 2\lambda + 1}, \qquad P(w) = {
m const.},$$

$$f(x) = rac{C}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} rac{w+i}{w^2 - 2\lambda + 1} e^{-ixw} dw$$

$$= C' \left\{ rac{1+\sqrt{(1-2\lambda)}}{2\sqrt{(1-2\lambda)}} e^{x\sqrt{(1-2\lambda)}} - rac{1-\sqrt{(1-2\lambda)}}{2\sqrt{(1-2\lambda)}} e^{-x\sqrt{(1-2\lambda)}}
ight\}.$$

11.18. An equation of A. C. Dixon. A similar problem is presented by the equation †

$$f(x) = g(x) + \lambda \int_{0}^{1} \frac{f(t)}{x+t} dt.$$
 (11.18.1)

This is satisfied formally by

$$f(x) = g(x) + \lambda \int_{0}^{1} g(t)\chi(x, t) dt$$
 (11.18.2)

if $\chi(x,t)$ satisfies the integral equation

$$\chi(x,t) = \frac{1}{x+t} + \lambda \int_{0}^{1} \frac{\chi(y,t)}{x+y} \, dy.$$
 (11.18.3)

Putting $x = e^{-\xi}$, $y = e^{-\eta}$, $t = e^{-\beta}$, this is

$$e^{-i\xi}\chi(e^{-\xi},e^{-eta})=rac{e^{-i\xi}}{e^{-\xi}+e^{-eta}}+\lambda\int\limits_0^\inftyrac{e^{-i\eta}\chi(e^{-\eta},e^{-eta})}{2\coshrac{1}{2}(\xi-\eta)}\,d\eta,$$

or, writing $e^{-\frac{1}{2}\xi}\chi(e^{-\xi},e^{-\beta})=\phi(\xi)$,

$$\phi(\xi) = \frac{e^{-\frac{1}{2}\xi}}{e^{-\xi} + e^{-\beta}} + \lambda \int_{0}^{\infty} \frac{\phi(\eta)}{2 \cosh \frac{1}{2}(\xi - \eta)} d\eta.$$
 (11.18.4)

Suppose that $\phi(\xi) = O(e^{c|\xi|})$, where $0 < c < \frac{1}{2}$. Let $c < a < \frac{1}{2}$. Then (11.18.4) is equivalent to

$$\int_{ia-\infty}^{ia+\infty} \Phi_{+}(w)e^{ixw} dw + \int_{-ia-\infty}^{-ia+\infty} \Phi_{-}(w)e^{ixw} dw$$

$$= \sqrt{\left(\frac{\pi}{2}\right)} \int_{ib-\infty}^{ib+\infty} \frac{e^{i\beta+iw\beta}}{\cosh \pi w} e^{ixw} dw + \pi \lambda \int_{ia-\infty}^{ia+\infty} \frac{\Phi_{+}(w)}{\cosh \pi w} e^{ixw} dw,$$

where $-\frac{1}{2} < b < \frac{1}{2}$. We can take b = a. It follows that

$$\Phi_{+}(w)\left(1 - \frac{\pi\lambda}{\cosh\pi w}\right) - \sqrt{\left(\frac{\pi}{2}\right)} \frac{e^{\frac{1}{2}\beta + i\beta w}}{\cosh\pi w}$$

$$+ \text{ A. C. Dixon (2).}$$
(11.18.5)

and $-\Phi_{-}(w)$ are regular and equal in the strip -a < v < a. Hence (11.18.5) is regular for $v < \frac{1}{2}$. Hence $\Phi_{+}(w)$ is regular for $v < \frac{1}{2}$, except possibly for simple poles at the zeros of $\cosh \pi w - \pi \lambda$. Suppose for example that $\pi \lambda = \sin \alpha \pi$, where $0 < \alpha < \frac{1}{2}$. Then the zeros are at $(\frac{1}{2} - \alpha)i, (-\frac{3}{2} - \alpha)i, ..., (-\frac{1}{2} + \alpha)i, (-\frac{5}{2} + \alpha)i, ...$

Hence

$$\Psi(w) = \frac{\Phi_{+}(w)}{\Gamma(-\frac{1}{4} + \frac{1}{2}\alpha - \frac{1}{2}iw)\Gamma(\frac{1}{4} - \frac{1}{2}\alpha - \frac{1}{2}iw)}$$

is regular for $v < \frac{1}{2}$.

To cancel the poles of sech πw in (11.18.5) we must also have

$$\Phi_{+}(w) = -\frac{1}{\lambda \sqrt{(2\pi)}} e^{i\beta + i\beta w}$$

at $w = -\frac{1}{2}i,..., -(n+\frac{1}{2})i,...$ Hence

$$\Psi\{-(n+\frac{1}{2})i\} = -\frac{1}{\lambda\sqrt{(2\pi)}} \frac{e^{(n+1)\beta}}{\Gamma(\frac{1}{2}\alpha - \frac{1}{2}n)\Gamma(\frac{1}{2} - \frac{1}{2}\alpha - \frac{1}{2}n)} = a_n,$$

say. The most obvious function with these properties is

$$\Psi(w) = \frac{i}{\Gamma(\frac{1}{2} - iw)} \sum_{n=0}^{\infty} \frac{(-1)^n}{n!} \frac{a_n}{w + (n + \frac{1}{2})i},$$

and it is easily verified that this does in fact give a solution.

The difference between two solutions of (11.18.4) satisfies

$$\phi(\xi) = \lambda \int_{0}^{\infty} \frac{\phi(\eta)}{2\cosh\frac{1}{2}(\xi - \eta)} d\eta, \qquad (11.18.6)$$

which is of the form (11.17.1). Here

$$\begin{split} 1 - \sqrt{(2\pi)}K(w) &= 1 - \frac{\pi\lambda}{\cosh\pi w} \\ &= \frac{2\pi\Gamma(\frac{1}{2} - iw)\Gamma(\frac{1}{2} + iw)}{\Gamma(\frac{1}{4} - \frac{1}{2}\alpha + \frac{1}{2}iw)\Gamma(\frac{1}{4} - \frac{1}{2}\alpha - \frac{1}{2}iw)\Gamma(\frac{3}{4} + \frac{1}{2}\alpha + \frac{1}{2}iw)\Gamma(\frac{3}{4} + \frac{1}{2}\alpha - \frac{1}{2}iw)} \\ &= \frac{\phi_1(w)}{\phi_2(w)}(\frac{1}{4} - \frac{1}{2}\alpha + \frac{1}{2}iw)(\frac{1}{4} - \frac{1}{2}\alpha - \frac{1}{2}iw), \end{split}$$

where

$$\begin{split} \phi_1(w) &= \frac{2\pi\Gamma(\frac{1}{2}+iw)}{\Gamma(-\frac{3}{4}-\frac{1}{2}\alpha+\frac{1}{2}iw)\Gamma(\frac{3}{4}+\frac{1}{2}\alpha+\frac{1}{2}iw)},\\ \phi_2(w) &= \frac{\Gamma(-\frac{3}{4}-\frac{1}{2}\alpha-\frac{1}{2}iw)\Gamma(\frac{3}{4}+\frac{1}{2}\alpha-\frac{1}{2}iw)}{\Gamma(\frac{1}{2}-iw)} \end{split}$$

have the properties stated in § 11.17.

11.19. A problem of radiative equilibrium. \uparrow Consider a medium stratified in planes perpendicular to the axis of x, extending indefinitely on the positive side of its boundary x = 0.

Let I (a function of x and θ) be the intensity of radiation of all wavelengths, at any point, in a direction making an angle θ with the negative direction of the axis of x. Let ρ be the density at any point, and k the coefficient of mass-absorption, supposed independent of the wave-length. Let B (a function of x) be the intensity of black-body radiation corresponding to the temperature of the matter at x.

The rate of absorption of energy per unit volume from the radiation in a solid angle ω is

 $k
ho \int \int I \ d\omega,$ while the rate of emission is

$$k\rho \int \int B d\omega = k\rho B\omega.$$

Consider a narrow circular cylinder, area of cross-section a, the centres of whose ends are at x and x', and whose axis makes an angle θ with the negative x-axis. The energy radiated from the x'-end through a distant area in the line of the axis of the cylinder, at which all points of the cylinder subtend approximately the same small solid angle ω , is $I(x',\theta)a\omega$; this is made up of $I(x,\theta)a\omega$ from the x-end, together with $\begin{cases} k\rho(B-I)\omega \ dv \end{cases}$

from the interior of the cylinder, v being its element of volume.

In the limit as $a \to 0$, $\omega \to 0$ we obtain

$$I(x', \theta) = I(x, \theta) - \int\limits_{x}^{x'} k \rho(B-I) \sec \theta \ d\xi,$$

and, making $x' \to x$,

$$\frac{\partial I}{\partial x} = k\rho \sec \theta (I - B). \tag{11.19.1}$$

For radiative equilibrium, the rate of absorption of energy per unit volume from all directions is equal to the rate of emission in all directions; this gives

$$4\pi k
ho B = k
ho\int\limits_0^{2\pi}d\phi\int\limits_0^\pi I\sin\theta\ d\theta$$

$$= 2\pi k
ho\int\limits_0^\pi I\sin\theta\ d\theta,$$

† E. A. Milne (1), Hopf (3); Hopf, Radiative Equilibrium.

$$2B = \int_{0}^{\pi} I \sin \theta \, d\theta. \tag{11 19.2}$$

Putting

$$\tau = \int_0^x k\rho \ dx,$$

(11.19.1) becomes

$$\frac{\partial I}{\partial \tau} = \sec \theta (I - B). \tag{11.19.3}$$

Hence

$$I = e^{\tau \sec \theta} \left\{ K - \int_{-\infty}^{\infty} B(t) \sec \theta \, e^{-t \sec \theta} \, dt \right\}.$$

The boundary condition is that the incident radiation is zero, i.e. that I=0 for $x=0, \frac{1}{2}\pi < \theta \leqslant \pi$. Hence

$$I = -e^{-\tau \sec \theta} \int_0^{\tau} B(t) \sec \theta \, e^{-t \sec \theta} \, dt \quad (\frac{1}{2}\pi < \theta \leqslant \pi). \tag{11.19.4}$$

For $0 < \theta \leqslant \frac{1}{2}\pi$ we choose K so that I is not exponentially large as $\tau \to \infty$, i.e. we obtain

$$I = e^{ au\sec{ heta}}\int\limits_{ au}^{\infty}B(t){
m sec}\, heta\,e^{-t\sec{ heta}}\,dt \quad (0< heta\leqslant rac{1}{2}\pi).$$

Inserting these results in (11.19.2), we obtain

$$B(\tau) = \frac{1}{2} \int_{0}^{\frac{1}{4}\pi} e^{\tau \sec \theta} \sin \theta \ d\theta \int_{\tau}^{\infty} B(t) \sec \theta \ e^{-t \sec \theta} \ dt - \frac{1}{2} \int_{\frac{1}{4}\pi}^{\pi} e^{\tau \sec \theta} \sin \theta \ d\theta \int_{0}^{\tau} B(t) \sec \theta \ e^{-t \sec \theta} \ dt$$

$$= \frac{1}{2} \int_{\tau}^{\infty} B(t) \ dt \int_{0}^{\frac{1}{4}\pi} e^{(\tau - t) \sec \theta} \tan \theta \ d\theta - \frac{1}{2} \int_{0}^{\tau} B(t) k(\tau - t) \ dt, \qquad (11.19.5)$$
here
$$k(t) = \frac{1}{2} \int_{0}^{\pi} e^{-t |\theta| \sec \theta} \tan \theta \ d\theta - \frac{1}{2} \int_{0}^{\infty} e^{-\lambda} \ d\lambda$$

where

$$k(\xi) = \frac{1}{2} \int_{0}^{4\pi} e^{-|\xi| \sec \theta} \tan \theta \ d\theta = \frac{1}{2} \int_{|\xi|}^{\infty} \frac{e^{-\lambda}}{\lambda} \ d\lambda.$$

We can now appeal to the theory of § 11.17. We have, if v < 1,

$$K(w) = \frac{1}{2\sqrt{(2\pi)}} \int_{-\infty}^{\infty} e^{ixw} dx \int_{|x|}^{\infty} \frac{e^{-\lambda}}{\lambda} d\lambda = \frac{1}{\sqrt{(2\pi)}} \int_{0}^{\infty} \cos xw \, dw \int_{x}^{\infty} \frac{e^{-\lambda}}{\lambda} d\lambda$$

$$= \frac{1}{\sqrt{(2\pi)w}} \int_{0}^{\infty} \sin xw \frac{e^{-x}}{x} dx = \frac{1}{\sqrt{(2\pi)}} \frac{\arctan w}{w},$$
so that
$$1 - \sqrt{(2\pi)}K(w) = 1 - \frac{\arctan w}{w}.$$

This has a double zero at the origin, and no other zeros in the strip -1 < v < 1. Hence, with the notation of §11.17, we put

$$\psi(w) = \left(1 - \frac{\arctan w}{w}\right) \frac{w^2 + 1}{w^2},$$

no additional factor being needed. Hence

$$\chi_2(w) = \frac{1}{2\pi i} \int_{i\gamma - \infty}^{i\gamma + \infty} \log \left\{ \left(1 - \frac{\arctan z}{z} \right) \frac{z^2 + 1}{z^2} \right\} \frac{dz}{z - w} \quad (v < \gamma).$$

Also $P(w) = \alpha + \beta w$, where α and β are constants, and the solution is

$$B(\tau) = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} \frac{\alpha+\beta w}{w^2} e^{-i\tau w + \chi_2(w)} dw.$$

11.20. The limiting form of Milne's equation. \dagger Writing $\int_{0}^{x} B(\tau) d\tau = f(x)$, (11.19.5) may also be written

$$f'(x) = \frac{1}{2} \int_{0}^{\infty} f'(t) dt \int_{|x-t|}^{\infty} \frac{e^{-y}}{y} dy$$

$$= \frac{1}{2} \int_{0}^{\infty} \frac{e^{-y}}{y} dy \int_{\max(0,x-y)}^{x+y} f'(t) dt$$

$$= \frac{1}{2} \int_{0}^{x} \frac{e^{-y}}{y} \{f(x+y) - f(x-y)\} dy + \frac{1}{2} \int_{x}^{\infty} \frac{e^{-y}}{y} f(x+y) dy.$$
(11.20.1)

† E. A. Milne (1), Hardy and Titchmarsh (1), (2). See also Hopf (2).

For large values of x this approximates to the form

$$f'(x) = \frac{1}{2} \int_{0}^{\infty} \frac{e^{-y}}{y} \{ f(x+y) - f(x-y) \} dy.$$
 (11.20.2)

THEOREM 155. If $f(x) = O(e^{c|x|})$, where 0 < c < 1, and both sides of (11.20.2) are finite and equal for every x, then f(x) is a quadratic.

The formal argument is similar to that of § 11.2. We have

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w)e^{-ixw} dw + \frac{1}{\sqrt{(2\pi)}} \int_{ib-\infty}^{ib+\infty} F_{-}(w)e^{-ixw} dw,$$
(11.20.3)

where 1 > a > c, -1 < b < -c. Hence

$$f(x+y)-f(x-y) = -\frac{2i}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w)\sin yw \, e^{-ixw} \, dw - ...,$$

$$\int_{0}^{\infty} \frac{e^{-y}}{y} \{f(x+y)-f(x-y)\} \, dy$$

$$= -\frac{2i}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w)e^{-ixw} \, dw \int_{0}^{\infty} \frac{e^{-y}}{y} \sin yw \, dy - ...$$

$$= -\frac{2i}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w) \arctan w \, e^{-ixw} \, dw - ...,$$
(11.20.4)

the dots indicating in each case the corresponding term involving $F_{-}(w)$. Also

$$f'(x) = -\frac{i}{\sqrt{(2\pi)}} \int_{ia-\infty}^{ia+\infty} F_{+}(w)w e^{-ixw} dw - \dots$$
 (11.20.5)

Hence (11.20.2) gives

$$\int_{ia-\infty}^{ia+\infty} F_{+}(w)(w-\arctan w)e^{-ixw} dw + \dots = 0.$$

Hence, by Theorem 141, p. 255, $F_{+}(w)$ and $F_{-}(w)$ are regular for $b \leq v \leq a$, except possibly for a triple pole at the origin corresponding to the triple zero of w-arctan w; and $F_{+}(w) = -F_{-}(w)$. Evaluating (11.20.3) by the calculus of residues, it follows that f(x) is a quadratic.

To justify the process we shall first prove that $e^{-c'|x|}f'(x)$ belongs to $L^2(-\infty,\infty)$ if c'>c. For (11.20.2) gives

$$f'(x) = \frac{1}{2} \int_{0}^{1} \frac{f(x+y) - f(x-y)}{y} dy + \frac{1}{2} \left[\int_{0}^{1} \frac{e^{-y} - 1}{y} \{f(x+y) - f(x-y)\} dy + \int_{1}^{\infty} \frac{e^{-y}}{y} \{f(x+y) - f(x-y)\} dy \right]$$

$$= \frac{1}{2} \phi(x) + \frac{1}{2} \psi(x),$$

say. If $|f(x)| \leqslant Ke^{c|x|}$,

$$|\psi(x)| \leqslant K \int_{0}^{1} (e^{c|x+y|} + e^{c|x-y|}) dy + \int_{1}^{\infty} e^{-y} (e^{c|x+y|} + e^{c|x-y|}) dy < Ke^{c|x|}.$$

We may write

$$\phi(x) = \int_{t-1}^{x+1} \frac{f(t)}{t-x} dt,$$

where the integral is a principal value at t = x. We now appeal to the theory of conjugate functions. Let

$$\phi_1(x) = \int\limits_{\xi-2}^{\xi+2} \frac{f(t)}{t-x} \, dt.$$
 Then
$$\int\limits_{\xi-1}^{\xi+1} |\phi_1(x)|^2 \, dx \leqslant \int\limits_{-\infty}^{\infty} |\phi_1(x)|^2 \, dx = \pi^2 \int\limits_{\xi-2}^{\xi+2} |f(t)|^2 \, dt$$

by (5.3.3). Also, for $\xi - 1 < x < \xi + 1$,

$$\begin{aligned} |\phi_1(x) - \phi(x)| &\leq \left| \int_{x+1}^{\xi+2} \frac{f(t)}{t-x} dt \right| + \left| \int_{\xi-2}^{x-1} \frac{f(t)}{t-x} dt \right| \\ &\leq \left(\int_{x+1}^{\xi+2} \frac{dt}{(t-x)^2} \int_{x+1}^{\xi+2} |f(t)|^2 dt \right)^{\frac{1}{2}} + \left(\int_{\xi-2}^{x-1} \frac{dt}{(t-x)^2} \int_{\xi-2}^{x-1} |f(t)|^2 dt \right)^{\frac{1}{2}} \\ &< A \left(\int_{\xi-2}^{\xi+2} |f(t)|^2 dt \right)^{\frac{1}{2}} \end{aligned}$$

Altogether it follows that

$$\int\limits_{\xi-1}^{\xi+1} |\phi(x)|^2 \, dx < A \int\limits_{\xi-2}^{\xi+2} |f(x)|^2 \, dx < K \, e^{2c|\xi|}.$$

Hence

$$\int\limits_{\xi-1}^{\xi+1} |f'(x)|^2\,dx < K\,e^{2c|\xi|},$$

$$\int\limits_{\xi-1}^{\xi+1} e^{-2c'|x|} |f'(x)|^2\,dx < K\,e^{2(c-c')|\xi|},$$

and the result stated follows.

It now follows that the integrals (11.20.5) exist in the mean-square sense, $wF_{+}(w)$ being $L^{2}(ia-\infty,ia+\infty)$ if a>c. Also the inversion of (11.20.4) is justified by absolute convergence; for $\sin yw$ is $O(e^{ay})$ for all y, and O(|yw|) for small |yw|, and so is

$$O(|yw|^{\frac{1}{4}}e^{ay})$$

for all y and w; and

$$\int_{|a|=\infty}^{|a|+\infty} |F_{+}(w)w^{\frac{1}{2}} dw| \int_{0}^{\infty} y^{-\frac{3}{2}} e^{(a-1)y} dy$$

is convergent. This completes the proof.

It has been proved by a more complicated method† that the result holds under less restrictive assumptions.

11.21. Bateman's equation.‡ Suppose that a function f(x) is represented by Fourier's single-integral formula (1.1.7), not merely in the limit, but for some value of λ , $\lambda = a$ say, exactly. Then

$$f(x) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(y) \frac{\sin a(x-y)}{x-y} dy$$
 (11.21.1)

for a given a and all x.

This is an integral equation of the form (11.2.1), but the conditions if § 11.2 are not satisfied, and the solution takes quite a different form.

Suppose that f(x) belongs to $L^2(-\infty,\infty)$. Let

$$g(x) = \sin ax/x$$
, $G(x) = \sqrt{(\frac{1}{2}\pi)} (|x| < a)$, $0 (|x| > a)$.

Then (2.1.8) gives

$$\int_{-\infty}^{\infty} f(y) \frac{\sin a(x-y)}{x-y} \, dy = \sqrt{\binom{\pi}{2}} \int_{-a}^{a} F(t)e^{-ixt} \, dt. \quad (11.21.2)$$

- † Hardy and Titchmarsh (9).
- ‡ Bateman (1), Hardy (2), Hardy and Titchmarsh (1), (2).

Hence

$$f(x) = \frac{1}{\sqrt{(2\pi)}} \int_{-a}^{a} F(t)e^{-ixt} dt,$$
 (11.21.3)

i.e. f(x) is a finite trigonometrical integral. Conversely, if f(x) is of the form (11.21.3), where F is L^2 , (11.21.1) follows from (11.21.2). Hence

THEOREM 156. A necessary and sufficient condition that a function f(x) of L^2 should be a solution of (11.21.1) is that it should be of the form (11.21.3), where F is $L^2(-a,a)$.

There are, however, simple solutions of (11.21.1) not belonging to L^2 ; for example, $\cos bx$ and $\sin bx$ are solutions if -a < b < a, though not if |b| > |a|. The next theorem includes these solutions.

THEOREM 157. Let f(x)/(1+|x|) belong to $L^2(-\infty,\infty)$, and let

$$\int_{-\infty}^{\infty} \frac{\cos ax \frac{f(x)}{x} dx}{\sin ax} dx, \qquad \int_{-\infty}^{\infty} \frac{\cos ax \frac{f(x)}{x} dx}{\sin ax} dx$$

exist. Then, if f(x) satisfies (11.21.1), it is of the form

$$f(x) = f(0) + x \int_{-a}^{a} \chi(u)e^{-ixu} du,$$

where $\chi(u)$ belongs to $L^2(-a,a)$.

It is easily verified that

$$\int_{-a}^{a} (e^{iyu} - e^{iya \operatorname{sgn} u}) e^{-ixu} du = \frac{2y}{x} \left\{ \frac{\sin a(x-y)}{x-y} - \frac{\sin ay}{y} \right\}.$$

Hence

$$f(x)-f(0) = \frac{1}{\pi} \int_{-\infty}^{\infty} \left\{ \frac{\sin a(x-y)}{x-y} - \frac{\sin ay}{y} \right\} f(y) \, dy$$

$$= \frac{x}{2\pi} \int_{-\infty}^{\infty} \frac{f(y)}{y} \, dy \int_{-a}^{a} (e^{iyu} - e^{iya \operatorname{sgn} u}) e^{-ixu} \, du$$

$$= \frac{x}{2\pi} \int_{-a}^{a} e^{-ixu} \, du \int_{-\infty}^{\infty} \frac{f(y)}{y} (e^{iyu} - e^{iya \operatorname{sgn} u}) \, dy$$

if we can invert the order of integration. This is obviously per-

missible for the part with |y| < 1, and for the part involving $e^{iya \operatorname{sgn} u}$ and |y| > 1. Also f(y)/y is $L^2(-\infty, -1)$ and $L^2(1, \infty)$, and the integrals

 $\int_{-\infty}^{-1} \frac{f(y)}{y} e^{iyu} dy, \qquad \int_{1}^{\infty} \frac{f(y)}{y} e^{iyu} dy$

exist in the mean-square sense. The inversion for these is a case of Parseval's formula in L^2 theory. The y-integral represents a function of $L^2(-a,a)$, and this is the result stated.

THEOREM 158. Let

$$f(x) = f(0) + x \int_{-a}^{a} \chi(u)e^{-ixu} du,$$

where $\chi(u)/(a^2-u^2)$ is L(-a,a). Then

$$f(x) = \frac{1}{\pi} \lim_{\lambda \to \infty} \int_{-\lambda}^{\lambda} \frac{\sin ay}{y} f(x-y) \ dy.$$

We may suppose f(0) = 0. Then

$$\int_{-\lambda}^{\lambda} \frac{\sin ay}{y} f(x-y) dy = \int_{-\lambda}^{\lambda} \frac{\sin ay}{y} (x-y) dy \int_{-a}^{a} \chi(u) e^{-i(x-y)u} du$$

$$= x \int_{-a}^{a} \chi(u) e^{-ixu} du \int_{-\lambda}^{\lambda} \frac{\sin ay}{y} e^{iyu} dy - \int_{-a}^{a} \chi(u) e^{-ixu} du \int_{-\lambda}^{\lambda} \sin ay e^{iyu} dy.$$

The first term tends to $\pi x \int_{-a}^{a} e^{-ixu} \chi(u) du = \pi f(x)$, by the bounded convergence of the inner integral. The second term is

$$-2i\int_{-a}^{a}e^{-ixu}\chi(u) du \int_{0}^{\lambda}\sin ay \sin yu dy$$

$$= -i\int_{-a}^{a}e^{-ixu}\chi(u)\left\{\frac{\sin(a-u)\lambda}{a-u} - \frac{\sin(a+u)\lambda}{a+u}\right\} du,$$

which tends to 0 as $\lambda \to \infty$ with the given conditions. Hence the result.

The function $f(x) = \sin bx$ (|b| < a) is a case of both these theorems; $f(x) = \sin ax$ is not a solution of the equation.

Also, if $0 \le m \le n$, $x^{-m}J_n(x)$ is a solution of the equation with a = 1. For

$$\begin{split} \frac{J_n(x)}{x^m} &= \frac{x^{n-m}}{2^n \sqrt{\pi} \, \Gamma(n+\frac{1}{2})} \int_{-1}^1 \, (1-y^2)^{n-\frac{1}{2}} e^{ixy} \, dy \\ &= \frac{i^{n-m}}{2^n \sqrt{\pi} \, \Gamma(n+\frac{1}{2})} \int_{-1}^1 \, \left(\frac{d}{dy}\right)^{n-m} (1-y^2)^{n-\frac{1}{2}} \cdot e^{ixy} \, dy. \end{split}$$

11.22. Kapteyn's equation.† A Neumann series (for an odd function) is an expansion of the form

$$f(x) = \sum_{n=0}^{\infty} a_{2n+1} J_{2n+1}(x).$$
 (11.22.1)

If f(x) is given, the coefficients a_{2n+1} may be obtained formally as follows. We have (e.g. from (7.10.1))

$$\int_{0}^{\infty} J_{2m+1}(t) J_{2n+1}(t) \frac{dt}{t} = 0 \qquad (m \neq n), \qquad (11.22.2)$$

Hence, multiplying by $J_{2m+1}(t)/t$ and integrating over $(0,\infty)$, we obtain

$$a_{2m+1} = (4m+2) \int_{0}^{\infty} f(t) \frac{J_{2m+1}(t)}{t} dt.$$
 (11.22.3)

The series formed with these coefficients is

$$\sum_{n=0}^{\infty} (4n+2)J_{2n+1}(x) \int_{0}^{\infty} f(t) \frac{J_{2n+1}(t)}{t} dt$$

$$= \int_{0}^{\infty} \frac{f(t)}{t} \left\{ \sum_{n=0}^{\infty} (4n+2)J_{2n+1}(x)J_{2n+1}(t) \right\} dt$$

(provided that we may integrate term-by-term)

$$= \frac{1}{2} \int_{0}^{\infty} f(t) dt \int_{0}^{x} \left\{ \frac{J_{1}(t-v)}{t-v} + \frac{J_{1}(t+v)}{t+v} \right\} J_{0}(x-v) dv$$

$$= \frac{1}{2} \int_{0}^{x} J_{0}(x-v) dv \int_{0}^{\infty} f(t) \left\{ \frac{J_{1}(t-v)}{t-v} + \frac{J_{1}(t+v)}{t+v} \right\} dt$$

[†] See Watson § 16.4, Hardy and Titchmarsh (1).

by §11.12. The inner integral is

$$\begin{split} \int_{-v}^{\infty} f(u+v) \frac{J_1(u)}{u} \, du \, + \, \int_{v}^{\infty} f(u-v) \frac{J_1(u)}{u} \, du \\ = \, \int_{0}^{\infty} \{f(u+v) + f(u-v)\} \frac{J_1(u)}{u} \, du \, + 2 \int_{0}^{v} f(v-u) \frac{J_1(u)}{u} \, du, \end{split}$$

and the last term gives

$$\int_{0}^{x} J_{0}(x-v) dv \int_{0}^{v} f(u) \frac{J_{1}(v-u)}{v-u} du = \int_{0}^{x} f(u) du \int_{u}^{x} J_{0}(x-v) \frac{J_{1}(v-u)}{v-u} dv$$

$$= \int_{0}^{x} f(u)J_{1}(x-u) du = f(x) - \int_{0}^{x} f'(u)J_{0}(x-u) du$$

on integrating by parts. The sum of the series is therefore f(x) if

$$\int_{0}^{x} f'(u)J_{0}(x-u) du = \frac{1}{2} \int_{0}^{x} J_{0}(x-u) du \int_{0}^{\infty} \{f(\xi+u) + f(\xi-u)\} \frac{J_{1}(\xi)}{\xi} d\xi,$$

and, by Theorem 150, this implies that

$$f'(u) = \frac{1}{2} \int_{0}^{\infty} \{f(\xi+u) + f(\xi-u)\} \frac{J_1(\xi)}{\xi} d\xi.$$
 (11.22.4)

This is Kapteyn's integral equation.

11.23. Before proceeding to rigorous analysis, we shall prove the following lemma.

LEMMA. For x > 0, t > 0,

$$\sum_{n=0}^{\infty} (4n+2)|J_{2n+1}(x)J_{2n+1}(t)| = O\{\min(x^3, x^{\frac{1}{2}})\min(t^3, t^{-\frac{1}{2}})\}.$$

We have

$$J_n(x) = O(x^{-\frac{1}{2}}) \qquad (n \leqslant \frac{1}{2}x),$$
 (11.23.1)

$$= O(1)$$
 (all n and x), (11.23.2)

and

$$J_n(x) = O\left\{\frac{(\frac{1}{2}x)^n}{n!}\right\} = O\left\{\frac{1}{\sqrt{n}}\left(\frac{xe}{2n}\right)^n\right\}$$
 (all n and x), (11.23.3)

so that in particular

$$J_n(x) = O(2^{-n}) \quad (n \geqslant ex).$$
 (11.23.4)

For x > 1, $2ex \le t$ the above sum is therefore

$$\sum_{2n+1=3}^{ex} O\left(\frac{n}{\sqrt{t}}\right) + \sum_{ex}^{\frac{1}{2}t} O\left(\frac{n}{2^{n}\sqrt{t}}\right) + \sum_{\frac{1}{2}t}^{\infty} O\left(\frac{n}{2^{n}}\right)$$

$$= O(x^{2}t^{-\frac{1}{2}}) + O(t^{-\frac{1}{2}}) + O(e^{-At}) = O(x^{2}t^{-\frac{1}{2}}).$$

For 1 < x < t < 2ex it is

$$\sum_{2n+1=3}^{ex} O(n) + \sum_{ex}^{\infty} O\left(\frac{n}{2^n}\right) = O(x^2) = O(x^{\frac{6}{2}t^{-\frac{1}{2}}}).$$

For $x \leq 1 < t$ it is

$$\begin{split} \sum_{2n+1=3}^{\frac{1}{2}t} O\left\{\frac{x^3}{\sqrt{t}} \sqrt{n} \left(\frac{e}{2^n}\right)^n\right\} + \sum_{\frac{1}{2}t}^{\infty} O\left\{x^3 \sqrt{n} \left(\frac{e}{2^n}\right)^n\right\} \\ &= O\left(\frac{x^3}{\sqrt{t}}\right) + O(x^3 e^{-At}) = O\left(\frac{x^3}{\sqrt{t}}\right). \end{split}$$

For $x \leq 1$, $t \leq 1$ it is

$$\sum^{\infty} O\left\{x^3t^3\left(\frac{e}{2^n}\right)^{2n}\right\} = O(x^3t^3).$$

THEOREM 159. Let f(x) be an odd function of x, and let $f(x)/(1+|x|)^{\frac{3}{2}}$ belong to $L(-\infty,\infty)$. Then a necessary and sufficient condition that f(x) should be expressible by the Neumann series (11.22.1), with the coefficients (11.22.3), is that f(x) should satisfy (11.22.4).

Suppose first that f(x) is expressible by the above series.

It follows from the lemma that, for a fixed x,

$$\sum_{0}^{\infty} (4n+2)|J_{2n+1}(x)J_{2n+1}(t)| = O\{\min(1,t^{-1})\},\,$$

and the inversion of the above summation and integration is justified by absolute convergence.

It is also clear from (11.23.3) that, if f(x) is expressible by (11.22.1), it is (like the sum of a power series) differentiable any number of times within the range of convergence of the series (here 0 to ∞). The final integration by parts is therefore justified. Hence Kapteyn's equation holds.

Conversely, if (11.22.4) holds, then f'(u) is continuous, by the uniform convergence of the integral. The argument can therefore be reversed.

11.24. Solution of Kapteyn's equation. Since, in § 11.22, f(x) is odd, (11.22.4) may be written

$$f'(u) = \frac{1}{2} \int_{0}^{\infty} \{f(u+\xi) - f(u-\xi)\} \frac{J_{1}(\xi)}{\xi} d\xi, \qquad (11.24.1)$$

or

$$f'(x) = -\frac{1}{2} \int_{-\infty}^{\infty} f(x-y) \frac{J_1(y)}{y} \operatorname{sgn} y \, dy, \qquad (11.24.2)$$

and in this form f(x) is not necessarily odd.

THEOREM 160. Let f(x) belong to $L^2(-\infty,\infty)$. Then a necessary and sufficient condition that (11.24.2) should hold for all values of x is that

$$f(x) = \int_{-1}^{1} \phi(u)e^{-ixu} du,$$

where $\phi(u)$ belongs to $L^2(-1,1)$.

The Fourier transform of $g(x) = x^{-1}J_1(x)\operatorname{sgn} x$ is

$$G(x) = i \sqrt{\left(\frac{2}{\pi}\right)} \int_{0}^{\infty} \frac{J_1(y)}{y} \sin xy \, dy$$

= $i \sqrt{\left(\frac{2}{\pi}\right)} x$ ($|x| \leqslant 1$), $i \sqrt{\left(\frac{2}{\pi}\right)} \frac{\operatorname{sgn} x}{|x| + \sqrt{(x^2 - 1)}}$ ($|x| > 1$).

Hence, if F is the transform of f,

$$\int_{-\infty}^{\infty} f(x-y) \frac{J_1(y)}{y} \operatorname{sgn} y \, dy = -i \sqrt{\left(\frac{2}{\pi}\right)} \int_{-\infty}^{1} \frac{F(t)e^{-ixt}}{|t| + \sqrt{(t^2 - 1)}} \, dt + i \sqrt{\left(\frac{2}{\pi}\right)} \int_{-1}^{1} tF(t)e^{-ixt} \, dt + i \sqrt{\left(\frac{2}{\pi}\right)} \int_{1}^{\infty} \frac{F(t)e^{-ixt}}{t + \sqrt{(t^2 - 1)}} \, dt;$$

 $-\frac{1}{2}$ of the integral of this with respect to x is

$$-\frac{1}{\sqrt{(2\pi)}} \int_{-\infty}^{-1} \frac{F(t)e^{-ixt}}{|t| + \sqrt{(t^2 - 1)}} \frac{dt}{t} + \frac{1}{\sqrt{(2\pi)}} \int_{-1}^{1} F(t)e^{-ixt} dt + \frac{1}{\sqrt{(2\pi)}} \int_{1}^{\infty} \frac{F(t)e^{-ixt}}{t + \sqrt{(t^2 - 1)}} \frac{dt}{t},$$

and, by the theory of § 3, the necessary and sufficient condition that this should equal f(x), or differ from it by a constant, is that $F(t) \equiv 0$ for |t| > 1. This proves the theorem.

THEOREM 161. If

$$f(x) = f(0) + x \int_{-1}^{1} \chi(u)e^{-ixu} du,$$

where $\chi(u)/\sqrt{(1-u^2)}$ belongs to L(-1,1), then f(x) is a solution of (11.24.1).

The term f(0) is a solution, so may be omitted. We then have

$$\frac{1}{2} \int_{0}^{\infty} \{f(x+\xi) - f(x-\xi)\} \frac{J_{1}(\xi)}{\xi} d\xi$$

$$= \frac{1}{2} \int_{0}^{\infty} \frac{J_{1}(\xi)}{\xi} (x+\xi) d\xi \int_{-1}^{1} \chi(u) e^{-i(x+\xi)u} du - \frac{1}{2} \int_{0}^{\infty} \frac{J_{1}(\xi)}{\xi} (x-\xi) d\xi \int_{-1}^{1} \chi(u) e^{-i(x-\xi)u} du$$

$$= -ix \int_{-1}^{1} \chi(u) e^{-ixu} du \int_{0}^{\infty} \frac{J_{1}(\xi)}{\xi} \sin \xi u d\xi + \frac{1}{2} \int_{-1}^{1} \chi(u) e^{-ixu} du \int_{0}^{\infty} J_{1}(\xi) \cos \xi u d\xi$$

$$= -ix \int_{-1}^{1} \chi(u) u e^{-ixu} du + \int_{0}^{1} \chi(u) e^{-ixu} du = f'(x)$$

(Watson, § 13.42) if the inversions are justified.

The repeated integral with the factor x outside is absolutely convergent; the inversion of the other is justified by dominated convergence provided that

$$\left|\int\limits_0^T J_1(\xi) {\cos \xi u} \; d\xi
ight| < rac{K}{\sqrt{(1-u^2)}}$$

for all T.

Here the leading term in the asymptotic expansion gives terms like

$$\int_{1}^{T} \frac{\cos \xi \cos \xi u}{\sqrt{\xi}} d\xi = \int_{1}^{T} \frac{\cos \xi (1-u)}{\sqrt{\xi}} d\xi + \dots = O\left(\frac{1}{\sqrt{(1-u)}}\right) + \dots,$$

and the result follows.

 $f(x) = \sin x$ is an example of this theorem.

THEOREM 162. Let f(x)/(1+|x|) belong to $L^2(-\infty,\infty)$, and $f(x)/(1+|x|)^{\frac{3}{2}}$ to $L(-\infty,\infty)$. Then, if f(x) satisfies Kapteyn's equation,

$$f(x) = x \int_{-1}^{1} \chi(u) e^{-ixu} du,$$

where $\chi(u)$ belongs to $L^2(-1,1)$.

The formal argument here is that, if f(x) satisfies Kapteyn's equation, it is expansible in the form (11.22.1); and then

$$\frac{1}{\pi} \int_{-\infty}^{\infty} f(x) \frac{\sin(\xi - x)}{\xi - x} dx = \sum_{n=0}^{\infty} \frac{a_{2n+1}}{\pi} \int_{-\infty}^{\infty} \frac{\sin(\xi - x)}{\xi - x} J_{2n+1}(x) dx$$
$$= \sum_{n=0}^{\infty} a_{2n+1} J_{2n+1}(\xi) = f(\xi).$$

Thus f(x) satisfies Bateman's equation (with a = 1), and so is equal to a Fourier integral with limits (-1,1). Owing to convergence difficulties we have to apply the argument indirectly. We have instead

$$\frac{1}{\pi} \int_{-\infty}^{\infty} \frac{f(x) - a_1 J_1(x) \sin(\xi - x)}{x^3} dx = \sum_{n=1}^{\infty} \frac{a_{2n+1}}{\pi} \int_{-\infty}^{\infty} \frac{\sin(\xi - x) J_{2n+1}(x)}{\xi - x} dx$$
$$= \sum_{n=1}^{\infty} a_{2n+1} \frac{J_{2n+1}(\xi)}{\xi^3} = \frac{f(\xi) - a_1 J_1(\xi)}{\xi^3}.$$

This inversion is justified by absolute convergence, since the lemma of §11.23 shows that

$$\sum_{n=1}^{\infty} (4n+2) \int_{0}^{\infty} \left| f(t) \atop t J_{2n+1}(t) \right| dt \int_{-\infty}^{\infty} \left| \frac{\sin(\xi-x)}{\xi-x} \right| \left| \frac{J_{2n+1}(x)}{x^3} \right| dx$$

is convergent.

It now follows from Theorem 156 that

$$\frac{f(x) - a_1 J_1(x)}{x^3} = \int_{-1}^{1} \phi(u) e^{-ixu} du,$$

where $\phi(u)$ belongs to $L^2(-1,1)$. Hence

$$\phi(u) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{f(x) - a_1 J_1(x)}{x^3} e^{ixu} dx \quad (-1 < u < 1),$$

and, since $\{f(x)-a_1J_1(x)\}/x$ belongs to $L^2(-\infty,\infty)$, $\phi(u)$ is the integral of the integral of a function of L^2 . Integrating by parts twice, we obtain

$$\frac{f(x)-a_1J_1(x)}{x} = x(ae^{ix}+be^{-ix}) + (ce^{ix}+de^{-ix}) + \int_{-1}^{1} \chi(u)e^{-ixu} du,$$

where χ is $L^2(-1,1)$; since the left-hand side is $L^2(-\infty,\infty)$, a,b,c, and d must vanish. This proves the theorem.

11.25. A differential equation of fractional order. The integral equation \dagger

 $f(x) = \frac{\lambda}{\Gamma(\alpha)} \int_{x}^{\infty} (y-x)^{\alpha-1} f(y) dy \qquad (11.25.1)$

may be regarded as a differential equation of order α . Suppose, for example, that α is a positive integer p, that f(x) tends to 0, as $x \to \infty$, with sufficient rapidity, and that

$$f_1(x) = \int\limits_x^\infty f(y) \ dy, \qquad f_2(x) = \int\limits_x^\infty f_1(y) \ dy, \dots.$$

Then, if we integrate repeatedly by parts, and write z for $f_p(x)$, (11.25.1) becomes $\frac{d^p z}{dx^p} = (-1)^p \lambda z.$

The only solutions of this are finite combinations of exponentials. The general equation (11.25.1) is of the form (11.2.1), with

$$k(x) = \frac{\lambda x^{\alpha-1}}{\Gamma(\alpha)} \quad (x > 0), \qquad 0 \quad (x < 0).$$

The theory of §11.2 is not applicable, since k(x) does not satisfy (11.2.3). But the equation still has exponential solutions. The conditions that $f(x) = e^{-ax}$ should be a solution are that $\mathbf{R}(a) > 0$ and $\lambda = a^{\alpha}$, where a^{α} means $e^{\alpha \log a}$, and $\log a$ has its principal value. If $\lambda > 0$, a may have any of the values

$$\lambda^{1/\alpha}e^{2r\pi i/\alpha}$$
 $(r=0,\pm 1,...)$

for which $|2r\pi/\alpha| < \frac{1}{2}\pi$. If $\alpha \leq 4$, and in particular if $\alpha < 1$, the only admissible value of a is $\lambda^{1/\alpha}$. We shall prove that in this case, at any rate, the solution is unique.

[†] Hardy and Titchmarsh (7).

i.e.

THEOREM 163. Let f(x) be integrable over any finite interval $0 < x_1 \le x \le x_2$, let $\lambda > 0$, $0 < \alpha < 1$, and let

$$f(x) = \frac{\lambda}{\Gamma(\alpha)} \int_{-r}^{\infty} (y - x)^{\alpha - 1} f(y) \, dy \qquad (11.25.2)$$

for every positive x. Then

$$f(x) = Ce^{-ax},$$

where $a = \lambda^{1/\alpha}$, and C is a constant.

If $\mathfrak{F}(s)$ is the Mellin transform of f(x), we have formally

$$\mathfrak{F}(s) = rac{\lambda}{\Gamma(lpha)} \int\limits_0^\infty x^{s-1} \, dx \int\limits_x^\infty (y-x)^{lpha-1} f(y) \, dy$$

$$= rac{\lambda}{\Gamma(lpha)} \int\limits_0^\infty f(y) \, dy \int\limits_0^y x^{s-1} (y-x)^{lpha-1} \, dx$$

$$= rac{\lambda \Gamma(s)}{\Gamma(s+lpha)} \mathfrak{F}(s+lpha).$$

We shall prove that this is in fact true, and base our solution on it; but we cannot justify the inversions as they stand, and we have to proceed indirectly. We require the following lemma.

Let
$$\uparrow$$
 $f_{\alpha}^{*}(x) = \frac{1}{\Gamma(\alpha)} \int_{-\infty}^{c} f(y)(y-x)^{\alpha-1} dy$

for every positive x. Then, if $\beta > 0$,

$$\frac{1}{\Gamma(\beta)} \int_{-\xi}^{c} f_{\alpha}^{*}(x)(x-\xi)^{\beta-1} dx = \frac{1}{\Gamma(\alpha+\beta)} \int_{-\xi}^{c} f(y)(y-\xi)^{\alpha+\beta-1} dy,$$

$$(f_{\alpha}^{*})_{\beta}^{*} = f_{\alpha+\beta}^{*}. \tag{11.25.3}$$

To prove this, we have to justify the inversion

$$\int_{-\xi}^{c} (x-\xi)^{\beta-1} dx \int_{-x}^{c} f(y)(y-x)^{\alpha-1} dy = \int_{-\xi}^{c} f(y) dy \int_{\xi}^{y} (x-\xi)^{\beta-1} (y-x)^{\alpha-1} dx.$$
Clearly
$$\int_{-\xi}^{c} \dots \int_{-x}^{c} \dots = \int_{-\xi}^{c} \dots \int_{-x}^{y} \dots,$$

and it is sufficient to prove that

$$I = \int_{-\xi}^{\xi + \delta} f(y) \, dy \int_{\xi}^{y} (x - \xi)^{\beta - 1} (y - x)^{\alpha - 1} \, dx \to 0$$

† See Bosanquet (1) for a proof under much more general conditions.

and

$$J = \int_{\xi+\delta}^{c} f(y) \, dy \int_{\xi}^{\xi+\delta} (x-\xi)^{\beta-1} (y-x)^{\alpha-1} \, dx \to 0$$

as $\delta \to 0$. Now

$$I = \frac{\Gamma(\alpha)\Gamma(\beta)}{\Gamma(\alpha+\beta)} \int_{-\xi}^{\xi+\delta} f(y)(y-\xi)^{\alpha+\beta-1} dy,$$

and the result for this follows from the second mean-value theorem and the existence of $\int_{-\xi}^{\xi} f(y)(y-\xi)^{\alpha-1} dy$.

Also

$$J=\int\limits_{\xi+\delta}^cf(y)(y-\xi)^{\alpha-1}\,dy\int\limits_{\xi}^{\xi+\delta}(x-\xi)^{\beta-1}\!\!\left(\!\!\!\begin{array}{c}y-x\\y-\xi\end{array}\!\!\!\right)^{\alpha-1}dx,$$

and the inner integral is steadily increasing with y, and its value when y = c is $O(\delta^{\beta})$. Hence the result for this part also follows from the second mean-value theorem.

Proof of Theorem 163. Let c > 1, $0 < x \le \frac{1}{2}c$, and write

$$f(x) = \frac{\lambda}{\Gamma(\alpha)} \int_{-x}^{c} f(y)(y-x)^{\alpha-1} dy + \frac{\lambda}{\Gamma(\alpha)} \int_{c}^{\infty} f(y)(y-x)^{\alpha-1} dy$$
$$= \lambda f_{\alpha}^{*}(x) + \lambda g(x),$$

say. Then

$$f_{\alpha}^{*}(x) = \frac{\lambda}{\Gamma(\alpha)} \int_{-x}^{c} \{f_{\alpha}^{*}(y) + g(y)\}(y-x)^{\alpha-1} dy$$

= $\lambda f_{2\alpha}^{*}(x) + \lambda g_{\alpha}^{*}(x)$

by (11.25.3). Hence

$$f(x) = \lambda g(x) + \lambda^2 g_{\alpha}^{*}(x) + \lambda^2 f_{2\alpha}^{*}(x).$$

Repeating the argument, we obtain

$$f(x) = \lambda g(x) + \lambda^2 g_{\alpha}^*(x) + \dots + \lambda^n g_{(n-1)\alpha}^*(x) + \lambda^n f_{n\alpha}^*(x), \quad (11.25.4)$$

where

$$f_{n\alpha}^*(x) = \frac{1}{\Gamma(n\alpha)} \int_{x}^{c} f(y)(y-x)^{n\alpha-1} dy.$$

By taking n large enough, in particular $n\alpha > 1$, we obtain

$$|f(x)| \leq \phi(x) + \frac{1}{2c} \int_{x}^{c} |f(y)| dy,$$
 (11.25.5)

where $\phi(x)$ is bounded as $x \to 0$. Hence f(x) is bounded as $x \to 0$; otherwise there would be a sequence of values of x such that

 $|f(x)| \ge f(y)$ $(x \le y \le c)$, $|f(x)| \to \infty$, which is inconsistent with (11.25.5). It then follows that $f_{n\alpha}^*(x)$ is continuous for $0 \le x \le c$, and hence so is f(x). Denote its limit as $x \to 0$ by f(0).

We can also differentiate (11.25.4), and it follows that f'(x) is bounded near the origin. The argument could be carried on indefinitely, but all we require is that

$$f(x)-f(0) = O(x)$$
as $x \to 0$.
Now $\mathfrak{F}(s) = \int_{s}^{1} \{f(x)-f(0)\}x^{s-1} dx + \int_{s}^{\infty} f(x)x^{s-1} dx + \frac{f(0)}{s}$,

primarily for $0 < \sigma < \alpha$, and then, as an analytic continuation of $\mathfrak{F}(s)$, for $-1 < \sigma < \alpha$. Since

$$\int_{1}^{\infty} f(0)x^{s-1} dx = -\frac{f(0)}{s} \quad (\sigma < 0),$$

$$\mathfrak{F}(s) = \int_{0}^{\infty} \{f(x) - f(0)\}x^{s-1} dx \quad (-1 < \sigma < 0).$$

Inserting values of f(x) and f(0) given by (11.25.2), we obtain formally

$$\mathfrak{F}(s) = \frac{\lambda}{\Gamma(\alpha)} \int_{0}^{\infty} x^{s-1} dx \int_{x}^{\infty} \{(y-x)^{\alpha-1} - y^{\alpha-1}\} f(y) dy + \\ + \frac{\lambda}{\Gamma(\alpha)} \int_{0}^{\infty} x^{s-1} dx \int_{0}^{x} y^{\alpha-1} f(y) dy \\ = \frac{\lambda}{\Gamma(\alpha)} \int_{0}^{\infty} f(y) dy \int_{0}^{y} x^{s-1} \{(y-x)^{\alpha-1} - y^{\alpha-1}\} dx + \\ + \frac{\lambda}{\Gamma(\alpha)} \int_{0}^{\infty} y^{\alpha-1} f(y) dy \int_{y}^{\infty} x^{s-1} dx \\ = \frac{\lambda}{\Gamma(\alpha)} \int_{0}^{\infty} f(y) \left\{ \frac{\Gamma(s)\Gamma(\alpha)}{\Gamma(s+\alpha)} y^{s+\alpha-1} - \frac{y^{s+\alpha-1}}{s} \right\} dy + \\ + \frac{\lambda}{s\Gamma(\alpha)} \int_{0}^{\infty} f(y) y^{s+\alpha-1} dy = \frac{\lambda\Gamma(s)}{\Gamma(s+\alpha)} \mathfrak{F}(s+\alpha).$$

We shall show that this process is valid if $-\alpha < \sigma < 0$.

For the first term, the integral over $y \leqslant N$ can plainly be inverted, and it is sufficient to prove that

$$\int_{0}^{N} x^{s-1} dx \int_{N}^{-\infty} \{(y-x)^{\alpha-1} - y^{\alpha-1}\} f(y) dy$$

$$\int_{0}^{\infty} x^{s-1} dx \int_{0}^{\infty} \{(y-x)^{\alpha-1} - y^{\alpha-1}\} f(y) dy$$
(11.25.6)
$$(11.25.7)$$

and

tend to 0 as $N \to \infty$. Now by the second mean-value theorem

$$\int_{N}^{\infty} \{(y-x)^{\alpha-1} - y^{\alpha-1}\} f(y) \ dy = \left\{ \left(1 - \frac{x}{N}\right)^{\alpha-1} - 1 \right\} \int_{N}^{\xi} y^{\alpha-1} f(y) \ dy;$$

the last integral is bounded, and

$$\int_{0}^{N} x^{\sigma-1} \left\{ \left(1 - \frac{x}{N}\right)^{\alpha-1} - 1 \right\} dx = N^{\sigma} \int_{0}^{1} \left\{ (1 - u)^{\alpha-1} - 1 \right\} u^{\sigma-1} du = O(N^{\sigma}),$$

giving the result for (11.25.6). Also (11.25.7) is

$$\frac{\Gamma(\alpha)}{\lambda} \int_{V}^{\infty} x^{s-1} f(x) \ dx - \int_{N}^{\infty} x^{s-1} \ dx \int_{T}^{\infty} y^{\alpha-1} f(y) \ dy,$$

which plainly tends to 0.

The inversion of the second term is equivalent to integration by parts:

$$\int_0^\infty x^{s-1} dx \int_0^x y^{\alpha-1} f(y) dy = \left[\frac{x^s}{s} \int_0^x y^{\alpha-1} f(y) dy \right]_0^\infty - \frac{1}{s} \int_0^\infty x^{s+\alpha-1} f(x) dx,$$

and the integrated term tends to 0 at each limit.

Let
$$\chi(s) = \lambda^{s/\alpha} \frac{\mathfrak{F}(s)}{\Gamma(s)}.$$

Then the above result is equivalent to

$$\chi(s+\alpha)=\chi(s).$$

Thus $\chi(s)$ has the period α , and is regular for $-1 < \sigma < 0$, and so everywhere. Also, if $h(x) = \int_{-\infty}^{\infty} f(y) y^{\alpha-1} dy = o(1)$,

$$\mathfrak{F}(s) = O(1) + \int_{1}^{\infty} f(x) x^{s-1} dx$$

$$= O(1) + h(1) + (s-\alpha) \int_{1}^{\infty} h(x) x^{s-\alpha-1} dx$$

$$= O(|t|) \quad (-1 < \sigma < 0).$$

Hence

$$\chi(s) = O(|t|^A e^{\frac{1}{4}\pi|t|})$$

for $-1 < \sigma < 0$, and so on any line parallel to the imaginary axis. Hence

 $\chi \binom{\alpha \log z}{2\pi i}$

is one-valued, and $O(\log^4 r r^{\frac{1}{4}\alpha})$ as $|z| = r \to \infty$, and $O(\log^4 (1/r) r^{-\frac{1}{4}\alpha})$ as $r \to 0$. Hence $\chi(s)$ is a constant,

$$\mathfrak{F}(s) = C\Gamma(s)\lambda^{-s/\alpha},$$

and, by Theorem 32,

$$f(x) = rac{C}{2\pi i} \int_{k-i\infty}^{k+i\infty} \Gamma(s) \lambda^{-s/\alpha} x^{-s} ds \quad (0 < k < lpha)$$

$$= Ce^{-x\lambda^{1/\alpha}}.$$

11.26. A probability problem.† A function f(x), such that $f(x) \ge 0$ and

 $\int_{-\infty}^{\infty} f(x) \ dx = 1,$

defines a law of errors, which asserts that the probability that the error in making a certain measurement lies within the range (x_1, x_2) is $\int_{x_1}^{x_2} f(x) dx$; or that, for small δx , the probability that the error lies in $(x, x + \delta x)$ is to a first approximation f(x) δx .

Suppose that we observe two quantities P and Q, and that the errors in observing them, p and q, are distributed according to laws f(x) and g(x). It is required to find the corresponding law for P+Q.

If p and q are capable of taking integer values only, and the proportion of times that p is x is f(x), and that q is y is g(y), then the proportion of times that p+q is ξ is

$$\sum_{x+y=\xi} \sum_{x=\xi} f(x)g(y) = \sum_{x} f(x)g(\xi-x),$$

i.e. the 'resultant' of f and g.

In the continuous case, a similar argument with f(x) δx and g(y) δy leads to

$$\int_{-\infty}^{\infty} f(x)g(\xi-x) \ dx$$

as the law for p+q. We can prove this rigorously as follows. Strictly, p and q run through sets of points E_1 and E_2 such that

$$mE_1(p\leqslant x)=\int\limits_{-\infty}^x f(u)\ du, \qquad mE_2(q\leqslant x)=\int\limits_{-\infty}^x g(u)\ du.$$
 Let $f_1(x)=\int\limits_{-\infty}^x f(u)\ du, \qquad g_1(x)=\int\limits_{-\infty}^x g(u)\ du.$

Consider the sum

$$S = \sum_{n=-\infty}^{\infty} g_1(\xi - n\delta) [f_1\{(n+1)\delta\} - f_1(n\delta)].$$

The term in n represents the probability that p is in $(n\delta, (n+1)\delta)$ and q is $\leqslant \xi - n\delta$. For such p and q, $p+q \leqslant \xi + \delta$. On the other hand, if $p+q \leqslant \xi$, then $n\delta \leqslant p \leqslant (n+1)\delta$ and $q \leqslant \xi - n\delta$ for some n. Hence $mE(p+q \leqslant \xi) \leqslant S \leqslant mE(p+q \leqslant \xi + \delta).$

Since f is L, and g_1 is continuous and tends to 0 as $x \to -\infty$ and to 1 as $x \to \infty$, it is easily seen that

$$\lim_{\delta \to 0} \sum_{n=-\infty}^{\infty} \int_{n\delta}^{(n+1)\delta} f(t) \{g_1(\xi-t) - g_1(\xi-n\delta)\} dt = 0,$$
i.e. that
$$\lim_{\delta \to 0} S = \int_{-\infty}^{\infty} f(t) g_1(\xi-t) dt.$$
Hence
$$mE(p+q \leqslant \xi) = \int_{-\infty}^{\infty} f(t) g_1(\xi-t) dt$$

$$= \int_{-\infty}^{\infty} f(t) dt \int_{-\infty}^{\xi-t} g(x) dx = \int_{-\infty}^{\infty} f(t) dt \int_{-\infty}^{\xi} g(u-t) du$$

$$= \int_{-\infty}^{\xi} du \int_{-\infty}^{\infty} f(t) g(u-t) dt,$$

which is equivalent to the result stated.

If f(x) gives a law of errors, so does $\frac{1}{a}f\left(\frac{x}{a}\right)$. We now ask for what law of errors the resultant of two laws of this form is also of the same form.

THEOREM 164. Let $f(x) \ge 0$, let f(x) and $x^2 f(x)$ belong to $L(-\infty, \infty)$, and let

$$\frac{1}{c}f\left(\frac{x}{c}\right) = \frac{1}{ab}\int_{-\infty}^{\infty} f\left(\frac{y}{a}\right)f\left(\frac{x-y}{b}\right) dy$$
 (11.26.1)

for every x, where a, b, c, are given positive numbers.

Then the conditions are consistent only if $c^2 = a^2 + b^2$; and in that case

 $f(x) = \frac{1}{\sqrt{(2\pi k)}} e^{-\frac{1}{8}x^2/k}$

almost everywhere, where k is a constant.

The integral

$$\int_{-\infty}^{\infty} x^m f(x) \ dx = K_m$$

exists for m = 0, 1, 2. Now, for m = 0, 1, 2,

$$\frac{1}{c} \int_{-\infty}^{\infty} x^m f\left(\frac{x}{c}\right) dx = \frac{1}{ab} \int_{-\infty}^{\infty} dx \int_{-\infty}^{\infty} x^m f\left(\frac{y}{a}\right) f\left(\frac{x-y}{b}\right) dy$$
$$= \frac{1}{ab} \int_{-\infty}^{\infty} f\left(\frac{y}{a}\right) dy \int_{-\infty}^{\infty} (y+u)^m f\left(\frac{u}{b}\right) du,$$

and, in the three cases m = 0, 1, 2 this gives

$$K_0 = K_0^2, (11.26.2)$$

$$cK_1 = aK_1K_0 + bK_0K_1, (11.26.3)$$

$$c^2K_2 = a^2K_2K_0 + 2abK_1^2 + b^2K_0K_2. (11.26.4)$$

Assuming that f(x) is not null, (11.26.2) gives $K_0 = 1$. Hence (11.26.3) gives $(a+b-c)K_1 = 0$, (11.26.5)

and (11.26.4) gives
$$(c^2-a^2-b^2)K_2 = 2abK_1^2$$
. (11.26.6)

But by Schwarz's inequality

$$K_1^2 < K_0 K_2 = K_2,$$

so that (11.26.6) gives

$$c^2-a^2-b^2 < 2ab,$$

 $c < a+b.$

Hence (11.26.5) gives $K_1 = 0$, and (11.26.6) gives

$$c^2 = a^2 + b^2$$
.

Let

$$\frac{a}{c} = \alpha, \qquad \frac{b}{c} = \beta.$$

Then, putting $x = c\xi$, $y = c\eta$, (11.26.1) becomes

$$f(\xi) = \frac{1}{\alpha\beta} \int_{-\infty}^{\infty} f\left(\frac{\eta}{\alpha}\right) f\left(\frac{\xi - \eta}{\beta}\right) d\eta, \qquad (11.26.7)$$

where $\alpha^2 + \beta^2 = 1$. (11.26.8)

Let

$$\Phi(x) = \sqrt{(2\pi)}F(x) = \int_{-\infty}^{\infty} f(t)e^{ixt} dt.$$

Then, by Theorem 41, (11.26.7) gives

$$\Phi(x) = \Phi(\alpha x)\Phi(\beta x). \tag{11.26.9}$$

Using (11.26.9) for each term on the right, we obtain

$$\Phi(x) = \Phi(\alpha^2 x) \Phi(\beta \alpha x) \Phi(\alpha \beta x) \Phi(\beta^2 x),$$

and so generally

$$\Phi(x) = \Phi(\gamma_{m,1}x)\Phi(\gamma_{m,2}x)...\Phi(\gamma_{m,m}x),$$

where $m = 2^n$, and the *m* numbers $\gamma_{m,1},..., \gamma_{m,m}$ are the 2^n terms obtained by expanding $(\alpha + \beta)^n$.

Hence

$$\gamma_{m,1}^2 + \gamma_{m,2}^2 + ... + \gamma_{m,m}^2 = (\alpha^2 + \beta^2)^n = 1.$$

Also $\gamma_{m,\mu}$ is of the form $\alpha^p \beta^q$, where p+q=n; and hence, supposing $\alpha \geqslant \beta$, we have $\gamma_{m,\mu} \leqslant \alpha^n \quad (\mu=1,2,...,m)$.

Hence $\max \gamma_{m,\mu} \to 0$ as $m \to \infty$.

Now since f(x), xf(x), and $x^2f(x)$ belong to $L, \Phi(x), \Phi'(x)$, and $\Phi''(x)$ are continuous; and

$$\Phi(0) = \int_{-\infty}^{\infty} f(t) dt = K_0 = 1,$$

$$\Phi'(0) = \int_{-\infty}^{\infty} f(t) it dt = iK_1 = 0,$$

$$\Phi''(0) = -\int_{-\infty}^{\infty} t^2 f(t) dt = -k,$$

and

say. Hence, in the neighbourhood of x = 0,

$$\log \Phi(x) = u(x) + iv(x),$$

where u(x) and v(x) are continuous, and

$$u(0) = v(0) = u'(0) = v'(0) = 0,$$

and

$$u''(0) = -k, \quad v''(0) = 0.$$

Hence

$$\begin{split} \log \Phi(x) &= \sum_{\mu=1}^{m} \{ u(\gamma_{m,\mu} x) + i v(\gamma_{m,\mu} x) \} \\ &= \frac{1}{2} x^2 \sum_{\mu=1}^{m} \gamma_{m,\mu}^2 \{ u''(\theta_{m,\mu} \gamma_{m,\mu} x) + i v''(\theta'_{m,\mu} \gamma_{m,\mu} x) \}, \end{split}$$

where $0 < \theta_{m,\mu} < 1$, $0 < \theta'_{m,\mu} < 1$. As $m \to \infty$, each $\gamma_{m,\mu} \to 0$, and hence so does each $\theta_{m,\mu} \gamma_{m,\mu}$, and uniformly with respect to μ .

Hence

$$\begin{split} \log \Phi(x) &= \frac{1}{2} x^2 \sum_{\mu=1}^{m} \gamma_{m,\mu}^2 \{ -k + o(1) \} \\ &= -\frac{1}{2} k x^2 + o(1), \\ \log \Phi(x) &= -\frac{1}{2} k x^2, \quad \Phi(x) = e^{-\frac{1}{2} k x^2}, \end{split}$$

i.e.

and so (by Theorem 27) almost everywhere

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{-\frac{1}{4}kx^4 - ixt} dt = \frac{1}{\sqrt{(2\pi k)}} e^{-\frac{1}{4}x^4/k},$$

i.e. the law is Gauss's law.

11.27. A problem in statistical dynamics.† Consider an assemblage of atoms moving in one dimension in such a way that the fraction of them with velocities between v and $v+\delta v$ is $\int_{v}^{v+\delta v} f(x) dx$.

At a subsequent instant let a fraction $\int_{w}^{w+\delta w} \phi(v,x) \, dx$ of those with velocity v have acquired increments of velocity between w and $w+\delta w$. Then, by an argument similar to that used in the previous section, the fraction of the whole which finish with velocities between v' and $v'+\delta v'$

$$v'+\delta v'$$
 is $\int\limits_{v'}^{v'+\delta v'}g(x)\;dx$, where
$$g(v')=\int\limits_{-\infty}^{\infty}f(v)\phi(v,v'-v)\;dv. \tag{11.27.1}$$

For a steady state $g \equiv f$, so that f satisfies the integral equation

$$f(v') = \int_{-\infty}^{\infty} f(v)\phi(v, v'-v) \, dv. \tag{11.27.2}$$

Suppose now that the motion is defined as follows. The centre of mass of atoms moving with velocity v moves according to the equation

 $\frac{dV}{dt} = -\lambda V \quad (\lambda > 0),$

so that after time t its velocity is $ve^{-\lambda t}$. Superimposed on this motion, the atoms are given increments of velocity u in time t, the proportion of those with increment between u and $u+\delta u$ being $\psi(u)$ δu ; and this increment is uncorrelated with v, so that the proportion of those

† E. A. Milne (2); Fowler, Statistical Mechanics, § 19.5. Milne's original method requires heavier restrictions than those assumed here.

with velocity v having the additional increment between u and $u+\delta u$ is also $\psi(u)\delta u$. It follows that

$$\phi(v, ve^{-\lambda t} - v + u) = \psi(u),$$

where $\psi(u)$ is independent of v, but of course depends on t. Putting $v' = ve^{-\lambda t} + u$, this is

$$\phi(v,v'-v)=\psi(v'-ve^{-\lambda t}).$$

The condition for a steady state is therefore

$$f(v') = \int_{-\infty}^{\infty} f(v)\psi(v'-ve^{-\lambda t}) dv,$$

where, as in the last section,

$$\int_{-\infty}^{\infty} f(x) \ dx = 1, \qquad \int_{-\infty}^{\infty} \psi(x) \ dx = 1.$$

Let F and Ψ be the transforms of f and ψ . Then

$$\begin{split} F(\xi) &= \frac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} e^{i\xi v'} \, dv' \int\limits_{-\infty}^{\infty} f(v) \psi(v' - v e^{-\lambda l}) \, dv \\ &= \frac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} f(v) \, dv \int\limits_{-\infty}^{\infty} \psi(v' - v e^{-\lambda l}) e^{i\xi v'} \, dv' \\ &= \frac{1}{\sqrt{(2\pi)}} \int\limits_{-\infty}^{\infty} f(v) \, dv \int\limits_{-\infty}^{\infty} \psi(x) e^{i\xi v e^{-\lambda l} + i\xi x} \, dx \\ &= \sqrt{(2\pi)} F(\xi e^{-\lambda l}) \Psi(\xi). \end{split}$$

We now show that a certain assumption about the limiting behaviour of $\psi(x)$ as $t \to 0$ actually determines all the functions completely. We assume that positive and negative increments u are equally likely, so that $\psi(u) = \psi(-u)$; and also that, as $t \to 0$, for any fixed positive δ ,

$$\int_{0}^{\delta} x^{2} \psi(x) dx \sim at, \qquad \int_{\delta}^{\infty} \psi(x) dx = o(t),$$

where a is a constant. It follows that, as $t \to 0$, for a fixed ξ ,

$$\sqrt{(2\pi)}\Psi(\xi) - 1 = \int_{-\infty}^{\infty} \psi(x)(e^{i\xi x} - 1) dx = 2 \int_{0}^{\infty} \psi(x)(\cos \xi x - 1) dx$$

$$= -\xi^{2} \int_{0}^{\delta} x^{2} \psi(x) dx + \int_{0}^{\delta} \psi(x) O(x^{4}) dx + O\left(\int_{\delta}^{\infty} \psi(x) dx\right)$$

$$= -\xi^{2} \{at + o(t)\} + O(\delta^{2}t) + o(t),$$

and by choosing first δ and then t sufficiently small it follows that

$$\sqrt{(2\pi)}\Psi(\xi)-1 \sim -\xi^2 at.$$

Hence

$$\frac{F(\xi) - F(\xi e^{-\lambda t})}{\xi - \xi e^{-\lambda t}} = \frac{F(\xi e^{-\lambda t}) \{ \sqrt{(2\pi)} \Psi(\xi) - 1 \}}{\xi - \xi e^{-\lambda t}} \rightarrow -\frac{a\xi F(\xi)}{\lambda}$$

as $t \to 0$; i.e.

$$F'(\xi) = -a\xi\lambda^{-1}F(\xi).$$

Hence

$$F(\xi) = Ce^{-\frac{1}{2}a\xi^2/\lambda},$$

and $C = F(0) = (2\pi)^{-\frac{1}{2}}$. Hence

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{-\frac{1}{4}a\xi^{2}/\lambda - i\xi x} d\xi = \left(\frac{\lambda}{2\pi a}\right)^{\frac{1}{2}} e^{-\frac{1}{4}\lambda x^{2}/a},$$

i.e. the distribution is 'Maxwellian'.

Hence also

$$\Psi(\xi) = \frac{1}{\sqrt{(2\pi)}} \frac{F(\xi)}{F(\xi e^{-\lambda l})} = \frac{1}{\sqrt{(2\pi)}} \exp\left\{-\frac{a\xi^2}{2\lambda}(1 - e^{-2\lambda l})\right\},$$

and hence

$$\psi(x) = \left\{\frac{\lambda}{2\pi a(1-e^{-2\lambda l})}\right\}^{\frac{1}{2}} \exp\left\{-\frac{\lambda x^2}{2a(1-e^{-2\lambda l})}\right\}.$$

BOOKS AND MONOGRAPHS

- H. BATEMAN, Partial Differential Equations of Mathematical Physics, Cambridge, 1932.
- S. Bochner, Vorlesungen über Fouriersche Integrale, Leipzig, 1932.
- H. BURKHARDT, Trigonometrische Reihen und Integrale, Encyklopädie der Math. Wiss., II, 1 (ii), 819-1354.
- G. A. CAMPBELL and R. M. FOSTER, Fourier Integrals for Practical Applications, Bell Telephone System Technical Publications, U.S.A., 1931.
- H. S. Carslaw, Fourier's Series and Integrals, 3rd ed., London, 1930.
- H. S. Carslaw, Mathematical Theory of the Conduction of Heat in Solids, 2nd ed., London, 1921.
- R. COURANT and D. HILBERT, Methoden der mathematischen Physik, I, 2nd ed., Berlin, 1931; II, Berlin, 1937.
- J. Fourier, The Analytical Theory of Heat, translated by A. Freeman, Cambridge, 1878.
- R. H. FOWLER, Statistical Mechanics, 2nd ed., Cambridge, 1936.
- G. H. HARDY, J. E. LITTLEWOOD, and G. PÓLYA, *Inequalities*, Cambridge, 1934.
- E. W. Hobson, The Theory of Functions of a Real Variable, 2nd ed., Cambridge, 1925.
- E. Hopf, Mathematical Problems of Radiative Equilibrium, Cambridge tracts, No. 31, 1933.
- E. L. Ince, Ordinary Differential Equations, London, 1927.
- H. Jeffreys, Operational Methods in Mathematical Physics, Cambridge tracts, No. 23, 2nd ed., 1931.
- H. LAMB, Hydrodynamics, 6th ed., Cambridge, 1932.
- H. Lebesgue, Leçons sur les séries trigonométriques, Paris, 1906.
- S. Mandelbrojt, Séries de Fourier et classes quasi-analytiques de fonctions, Paris, 1935.
- R. E. A. C. Paley and N. Wiener, Fourier Transforms in the Complex Domain, American Math. Soc. Colloquium Publications, XIX, New York, 1934.
- M. H. Stone, Linear Transformations in Hilbert Space, and their Applications to Analysis, American Math. Soc. Colloquium Publications, XV, New York. 1932.
- E. C. TITCHMARSH, The Theory of Functions, Oxford, 1932.
- L. Tonelli, Serie trigonometriche, Bologna, 1928.
- C. DE LA VALLÉE-POUSSIN, Cours d'analyse infinitésimale, 3rd ed., Louvain, 1914.
- G. N. Watson, Theory of Bessel Functions, Cambridge, 1922. (Referred to in the text as 'Watson'.)
- H. Weber, Die partiellen Differential-Gleichungen der mathematischen Physik, nach Riemann's Vorlesungen, Braunschweig, 1900. (Referred to in the text as 'Riemann-Weber'.)

- E. T. WHITTAKER and G. N. WATSON, Modern Analysis, 4th ed., Cambridge, 1927.
- N. WIENER, The Fourier Integral, Cambridge, 1932.
- A. Zygmund, Trigonometrical Series, Warsaw, 1935.

ORIGINAL MEMOIRS REFERRED TO IN THE TEXT

R. BADESCU

(1) Sur une équation intégrale, Mathesis, 42 (1928), 414-16.

W. N. BAILEY

- (1) Some classes of functions which are their own reciprocals in the Fourier-Bessel integral transform, *Journal London Math. Soc.* 5 (1930), 258-65.
- (2) On the solutions of some integral equations, Journal London Math. Soc. 6 (1931), 242-7.
- (3) On a function which is its own reciprocal in the Fourier-Bessel integral transform, Journal London Math. Soc. 5 (1930), 92-5.
- (4) A note on an integral due to Ramanujan, Journal London Math. Soc. 6 (1931), 216-17.

E. W. BARNES

- (1) A new development of the theory of the hypergeometric functions, *Proc. London Math. Soc.* (2) 6 (1908), 141-77.
- (2) A transformation of generalized hypergeometric series, Quart. J. of Math. 41 (1910), 136-40.

H. BATEMAN

- (1) The inversion of a definite integral, Proc. London Math. Soc. (2) 4 (1906), 461-98.
- (2) Report on the history and present state of the theory of integral equations, Reports of the British Association (1910), 354-424.
- (3) The solution of partial differential equations by means of definite integrals, Messenger of Math. 41 (1911), 94-101.
- (4) Some simple definite integrals derived from the formulae of Fourier and Abel, Messenger of Math. 41 (1911), 180-4,
- (5) The solution of linear differential equations by means of definite integrals, *Trans. Camb. Phil. Soc.* 21 (1912), 171-96.
- (6) An integral equation occurring in a mathematical theory of retail trade, Messenger of Math. 49 (1920), 134-7.
- (7) The equation for the transverse vibrations of thin rods, Messenger of Math. 57 (1928), 145-54.

F. Bernstein

- (1) Über das Fourierintegral $\int_{0}^{\infty} e^{-x^{4}} \cos tx \, dx$, Math. Annalen, 79 (1919), 265–8.
- (2) Die Integralgleichung der elliptischen Thetanullfunktion, Berliner Sitzungsberichte (1920), 735-47.

F. Bernstein and G. Doetsch

(1) Probleme aus der Theorie der Wärmeleitung I, Math. Zeitschrift, 22 (1925), 285-92.

A. C. Berry

- (1) The Fourier transform identity theorem, Annals of Math. (2) 32 (1931), 227-32.
- (2) Necessary and sufficient conditions in the theory of Fourier transforms, ibid. 830-8.

R. P. Boas

(1) Some theorems on Fourier transforms and conjugate trigonometrical integrals, *Trans. Amer. Math. Soc.* 40 (1936), 287-308.

S. BOCHNER

- (1) Darstellung reellvariabler und analytischer Funktionen durch verallgemeinerte Fourier- und Laplace-Integrale, *Math. Annalen*, 97 (1927), 635-62.
- (2) Über eine Klasse singulärer Integralgleichungen, Berliner Sitzungsberichte, 22 (1930), 403-11.
- (3) Inversion formulae and unitary transformations, Annals of Math. 35 (1934), 111-15.

S. BOCHNER and G. H. HARDY

(1) Notes on two theorems of Norbert Wiener, Journal London Math. Soc. 1 (1926), 240-4.

L. S. BOSANQUET

(1) On Abel's integral equation and fractional integrals, *Proc. London Math. Soc.* (2) 31 (1930), 134-43.

F. W. BRADLEY

(1) An extension of the Fourier-Bessel integral, Proc. London Math. Soc.(2) 41 (1936), 209-14.

T. J. I'A. Bromwich

(1) Normal coordinates in dynamical systems, Proc. London Math. Soc.
(2) 15 (1915), 401-48.

T. A. Brown

(1) Fourier's integral, Proc. Edin. Math. Soc. 34 (1915), 3-10.

P. J. Browne

(1) Sur une formule directe pour la solution d'une équation intégrale d'Abel, Comptes rendus, 158 (1914), 1562-5.

J. C. Burkill

- (1) On Mellin's inversion formula, Proc. Camb. Phil. Soc. 23 (1926), 356-60.
- (2) The expression in Stieltjes integrals of the inversion formulae of Fourier and Hankel, *Proc. London Math. Soc.* (2) 25 (1926), 513-24.

I. W. BUSBRIDGE

(1) On general transforms with kernels of the Fourier type, *Journal London Math. Soc.* 9 (1934), 179-87.

E. CAHEN

 Sur la fonction ζ(s) de Riemann et sur des fonctions analogues, Ann. de l'Éc. Norm. (3) 11 (1894), 75-164.

B. H. CAMP

(1) Multiple integrals over infinite fields, and the Fourier multiple integral, Amer. Journal of Math. 39 (1917), 311-34.

G. CANTOR

(1) Über trigonometrische Reihen, Math. Annalen, 4 (1871), 139-43. (Werke, 87-91.)

F. CARLSON

(1) Sur une classe de séries de Taylor, Thèse, Upsala, 1914.

H. S. CARSLAW

(1) Gibbs's phenomenon in the sum (C, r) for r > 0 of Fourier's integral, Journal London Math. Soc. 1 (1926), 201-4.

L. Casper

(1) Über eine Erweiterung der Fourierschen Integralformel, Jahresbericht D.M.V. 37 (1928), 251-3.

A. L. CAUCHY

- (1) Mémoire sur la théorie des ondes, Œuvres, Sér. 1, 1, 5-318; in particular Note VI, 133-9, and Note XIX, Sur les fonctions réciproques, 300-3.
- (2) Mémoire sur l'intégration des équations linéaires, Œuvres, Sér. 2, 1, 275-357.

R. G. COOKE

- (1) The inversion formulae of Hardy and Titchmarsh, Proc. London Math. Soc. (2) 24 (1925), 381-420.
- (2) Gibbs's phenomenon in Fourier-Bessel series and integrals, Proc. London Math. Soc. (2) 27 (1928), 171-92.

J. Cossar

(1) A note on Fourier integrals, Journal London Math. Soc. 7 (1932), 38-46.

L. CRIJNS

(1) Over een eigenfunctie bij een integraalvergelijking, Handel. Nederl. Natuur- en Geneesk. Congr. 19 (1923), 116-17. (Fortschritte der Math. 49 (1923), 296-7.)

D. P. DALZELL

(1) Heaviside's operational method, Proc. Phys. Soc. 42 (1930), 75-81.

A. C. DIXON

- (1) Some limiting cases in the theory of integral equations, Proc. London Math. Soc. (2) 22 (1923), 201-22.
- (2) On the solving nuclei of certain integral equations whose nuclei are homogeneous and of degree -1, and the solution of a class of linear functional equations, *Proc. London Math. Soc.* (2) 27 (1926), 233-72.

A. L. DIXON and W. L. FERRAR

(1) Infinite integrals of Bessel functions, Quart. J. of Math. (Oxford), 6 (1935), 161-74.

G. DOETSCH

- (1) Die Integrodifferentialgleichungen vom Faltungstypus, Math. Annalen, 89 (1923), 192-207.
- (2) Bemerkung zu der Arbeit von V. Fock, Math. Zeitschrift, 24 (1925), 785-91.
- (3) Probleme aus der Theorie der Wärmeleitung II, Math. Zeitschrift, 22 (1925), 293-306.
- (4) Die Anwendung von Funktionaltransformationen in der Theorie der Differentialgleichungen und die symbolische Methode (Operatorenkalkül), Jahresbericht D.M.V., 43 (1934), 238-51.

W. Dorn

(1) Fouriersche Integrale als Grenzwerte Fourierscher Reihen, Sitz. Wiener Akad. der Wissensch. 135 (1926), 127-47.

H. EYRAUD

(1) La sommation des intégrales divergentes dans la théorie des spectres, Comptes rendus, 190 (1930), 254-6.

W. L. FERRAR

(1) Summation formulae and their relation to Dirichlet's series, Compositio Math. 1 (1935), 344-60.

V. Fock

(1) Über eine Klasse von Integralgleichungen, Math. Zeitschrift, 21 (1924), 161-73.

R. H. FOWLER

A simple extension of Fourier's integral theorem, and some physical applications, in particular to the theory of quanta, Proc. Royal Soc.

 (A) 99 (1921), 462-71.

C. Fox

- (1) A generalization of the Fourier-Bessel transform, *Proc. London Math. Soc.* (2) 29 (1929), 401-52.
- (2) A note on Hankel's theorem, *Proc. London Math. Soc.* (2) 30 (1929), 18-22.
- (3) Applications of Mellin's transformation to integral equations, *Proc. London Math. Soc.* (2) 38 (1935), 495-502.

R. GIBRAT

(1) Sur un type assez général d'équations intégrales singulières, Comptes rendus, 197 (1933), 1571-4.

S. GOLDSTEIN

- (1) Note on the operational solution of an integral equation, Journal London Math. Soc. 6 (1931), 262-8.
- (2) Some two-dimensional diffusion problems with circular symmetry, *Proc. London Math. Soc.* (2) 34 (1932), 51-88.
- (3) Operational representations of Whittaker's confluent hypergeometric function and Weber's parabolic cylinder function, *Proc. London Math. Soc.* (2) 34 (1932), 103-25.

G. Graziani

- (1) Sulla formula integrale di Fourier, Rom. Accad. L. Rend. (5) 18₂ (1909), 169-72.
- (2) Funzioni rappresentabili con la formula integrale di Fourier, ibid. 596-601.

M. E. GRIMSHAW

- (1) A case of distinction between Fourier integrals and Fourier series, *Proc. Camb. Phil. Soc.* 23 (1927), 755-67.
- (2) Summation of the integral conjugate to the Fourier integral of finite type, ibid. 871-81.
- (3) A contribution to the theory of uniqueness of representation by trigonometrical integrals, *Proc. London Math. Soc.* (2) 29 (1929), 243-56.

H. HAHN

- (1) Über die Methode der arithmetischen Mittel in der Theorie der verallgemeinerten Fourierintegrale, Sitz. Wiener Akad. der Wissensch. (1925), 449-70.
- (2) Über eine Verallgemeinerung der Fourierschen Integralformel, Acta Math. 49 (1927), 301-53.

G. H. HARDY

- (1) Note on the function $\int_{x}^{\infty} e^{\frac{1}{2}(x^{2}-t^{2})} dt$, Quart. J. of Math. 35 (1904), 193-207.
- (2) On an integral equation, *Proc. London Math. Soc.* (2) 7 (1909), 445-72.
- (3) Some cases of the inversion of the order of integration, Messenger of Math. 41 (1911), 102-9.
- (4) Further researches in the theory of divergent series and integrals, *Trans. Camb. Phil. Soc.* 21 (1912), 1-48.
- √(5) Fourier's double integral and the theory of divergent integrals, ibid. 427-51.
 - (6) On Weierstrass's singular integral, and on a theorem of Lerch, Messenger of Math. 46 (1916), 43-8.
 - (7) On Stieltjes' 'problème des moments', Messenger of Math. 46 (1917), 175-82, and 47 (1917), 81-8.
 - (8) On Mellin's inversion formula, Messenger of Math. 47 (1918), 178-84.
 - (9) On some definite integrals considered by Mellin, Messenger of Math. 49 (1919), 85-91.
 - (10) Further notes on Mellin's inversion formulae, Messenger of Math. 50 (1920), 165-71.
 - (11) On Fourier's series and Fourier's integral, Messenger of Math. 52 (1922), 49-53.
 - (12) On Fourier transforms, Messenger of Math. 53 (1923), 135-42.
 - (13) Some formulae in the theory of Bessel functions, London Math. Soc. Records, June 12, 1924; Proc. London Math. Soc. (2) 23 (1924), lxi-lxiii.

G. H. HARDY (cont.)

- (14) On Hilbert transforms, Messenger of Math. 54 (1924), 20-7 and 81-8.
- (15) The lattice points of a circle, *Proc. Roy. Soc.* (A) 107 (1925), 623-35,
- (16) Some further applications of Mellin's inversion formula, Messenger of Math. 56 (1926), 186-92.
- (17) A discontinuous integral, Messenger of Math. 57 (1928), 113-20.
- (18) Summation of a series of polynomials of Laguerre, Journal London Math. Soc. 7 (1932), 138-9.
- (19) A theorem concerning Fourier transforms, Journal London Math. Soc. 8 (1933), 227-31.
- (20) The resultant of two Fourier kernels, Proc. Camb. Phil. Soc. 31 (1935), 1-6.

G. H. HARDY and E. LANDAU

(1) The lattice points of a circle, *Proc. Royal Soc.* (A) 105 (1925), 244-58.

G. H. HARDY and J. E. LITTLEWOOD

- (1) Some new properties of Fourier constants, Math. Annalen, 97 (1926), 159-209.
- (2) A point in the theory of conjugate functions, Journal London Math. Soc. 4 (1929), 242-5.
- (3) Some new properties of Fourier constants, Journal London Math. Soc. 6 (1931), 3-9.
- (4) Some more theorems concerning Fourier series and Fourier power series, Duke Math. Journal, 2 (1936), 354-82.

G. H. HARDY and E. C. TITCHMARSH

- (1) Solutions of some integral equations considered by Bateman, Kapteyn, Littlewood, and Milne, *Proc. London Math. Soc.* (2) 23 (1924), 1-26.
- (2) Additional note on certain integral equations. Proc. London Math. Soc. (2) 30 (1929), 95-106.
- (3) Solution of an integral equation, Journal London Math. Soc. 4 (1929), 300-4.
- (4) Self-reciprocal functions, Quart. J. of Math. (Oxford), 1 (1930), 196-231.
- (5) A note on Parseval's theorem for Fourier transforms, Journal London Math. Soc. 6 (1931), 44-8.
- (6) Formulae connecting different classes of self-reciprocal functions, *Proc. London Math. Soc.* (2) 33 (1931), 225-32.
- (7) An integral equation, Proc. Camb. Phil. Soc. 28 (1932), 165-73.
- (8) A class of Fourier kernels, *Proc. London Math. Soc.* (2) 35 (1933), 116-55.
- (9) New solution of an integral equation, Proc. London Math. Soc. (2) 41 (1936), 1-15.

F. HAUSDORFF

(1) Eine Ausdehnung des Parsevalschen Satzes über Fourierreihen, Math. Zeitschrift, 16 (1923), 163-9.

A. E. HEINS

(1) Applications of the Fourier transform theorem, J. Math. Phys. Mass. Inst. Tech. 14 (1935), 137-42.

E. HILB

(1) Über Integraldarstellung willkürlicher Funktionen, Math. Annalen, 66 (1909), 1-66.

E. HILLE

- (1) A class of reciprocal functions, Annals of Math. (2), 27 (1926), 427-64.
- (2) On Laplace integrals, 8. Skand. Mat. Kongr. (1935), 216-27.

E. HILLE and J. D. TAMARKIN

- (1) On the theory of Laplace integrals, Proc. Nat. Acad. Sci. U.S.A. 19 (1933), 908-12.
- (2) On a theorem of Paley and Wiener, Annals of Math. (2), 34 (1933), 606-14.
- (3) On the theory of Fourier transforms, Bull. Amer. Math. Soc. 39 (1933), 768-74.
- (4) A remark on Fourier transforms and functions analytic in a half-plane, Compositio Math. 1 (1934), 98-102.
- (5) On the absolute integrability of Fourier transforms, Fundamenta Math. 25 (1935), 329-52.

E. HILLE, A. C. OFFORD, and J. D. TAMARKIN

(1) Some observations on the theory of Fourier transforms, Bull. Amer. Math. Soc. 41 (1935), 427-36.

E. W. Hobson

(1) On a general convergence theorem, and the theory of the representation of a function by series of normal functions, *Proc. London Math. Soc.* (2) 6 (1908), 349-95 (esp. 367-70).

E. Hopf

- (1) Über lineare Integralgleichungen mit positivem Kern, Sitz. Berlin. Akad. Wiss. 1928, No. 18.
- (2) Über eine Klasse singulärer Integralgleichungen; Bemerkungen zur Methode von Hardy und Titchmarsh, Journal London Math. Soc. 4 (1929), 23-7.
- (3) Mathematisches zur Strahlungsgleichgewichtstheorie der Fixsternatmosphären, *Math. Zeitschrift*, 33 (1931), 109.

P. Humbert

(1) Sur les intégrales de Fresnel, Mathematica, 10 (1935), 32-6 (Zentralblatt für Math. 10, 398).

A. E. Ingham

(1) A note on Fourier transforms, Journal London Math. Soc. 9 (1934), 29-32.

S. IZUMI

- (1) Remarks on the Fourier's inversion formulas, Tôhoku Math. Journal 29 (1928), 268-77.
- (2) Über die Summierbarkeit der Fourierschen Integralformel, Tôhoku Math. Journ. 30 (1929), 96-110.
- (3) On the Cahen-Mellin inversion formula, Tôhoku Math. Journal, 30 (1929), 111-14.
- (4) On Hankel transforms, Proc. Acad. Tokyo, 5 (1929), 113-15.
- (5) Hankel transforms, Tôhoku Math. Journal, 32 (1929), 54-8.
- (6) Remarks on Fourier integrals, Science Reports of the Tôhoku Imperial University, (1) 23 (1934), 484-90.
- (7) On the generalized Fourier integrals, Science Reports of the Tôhoku Imperial University, 23 (1935), 880-919.
- (8) On Wiener's proof of Plancherel's theorem, Science Reports Tôhoku, 24 (1935), 55-62.
- (9) A remark on an integral equation, Proc. Imp. Acad. Tôkyô, 11 (1935), 210-11.
- (10) On the Wiener's formula, Science Reports Tohoku, 24 (1935), 63-76.

S. Izumi and T. Kitagawa

- (1) On some integral equations I, Jap. Journal of Math. 12 (1935), 81-94.
- (2) On some integral equations II, Science Reports of the Tôhoku Imperial University (1) 25 (1936), 50-5.
- (3) On the linear operations, Tôhoku Math. Journal, 42 (1936), 54-64.

 M. JACOB

(1) Über die Cesàro'sche Summierbarkeit des Fourier'schen Integrales, Bull. Acad. Polonaise (1926), 41-74.

- (2) Über den Eindeutigkeitssatz in der Theorie der trigonometrischen Integrale, Math. Annalen, 97 (1927), 663-74.
- (3) Über den Eindeutigkeitssatz in der Theorie der verallgemeinerten trigonometrischen Integrale, Math. Annalen, 100 (1928), 278-94.
- (4) Über die Summierbarkeit von Fourierschen Reihen und Integralen, Math. Zeitschrift, 29 (1928), 20-33.

S. KACZMARZ

(1) Note on general transforms, Studia Math. 4 (1933), 146-51.

T. KAMEDA

(1) Theory of generating functions and its application to the theory of probability, Journal of the Faculty of Science, Tokyo, 1 (1925), 1-62.

L. V. KING

(1) On the acoustic radiation pressure on circular discs, inertia and diffraction corrections, *Proc. Royal Soc.* (A), 153 (1935), 1-16.

N. Koshliakov

(1) On a general summation formula and its applications, C. R. Acad. Sci. U.S.S.R. 4 (1934), 187-90.

B. KUTTNER

(1) Divisor functions; Fourier integral theorems, Proc. London Math. Soc. (2) 37 (1934), 161-208.

H. LEBESGUE

(1) Sur les intégrales singulières, Annales de Toulouse (3), 1 (1909), 25-118 (esp. 90-2).

N. LEVINSON

(1) On a class of non-vanishing functions, *Proc. London Math. Soc.* (2), 41 (1936), 393-407.

P. LÉVY

- (1) Sur le rôle de la loi de Gauss dans la théorie des erreurs, Comptes rendus, 174 (1922), 855-7.
- (2) Sur une application de la dérivée d'ordre non entier au calcul des probabilités, Comptes rendus, 176 (1923), 1118-20.
- (3) Sur une équation intégrale considerée par M. Picard, Bull. Sc. Math. (2), 52 (1928), 156-60.

E. H. LINFOOT

(1) A sufficiency condition for Poisson's formula, *Journal London Math.* Soc. 4 (1928), 54-61.

J. E. LITTLEWOOD

(1) On a theorem of Kolmogoroff, Journal London Math. Soc. 1 (1926), 229-31.

H. M. MACDONALD

(1) Some applications of Fourier's theorem, Proc. London Math. Soc. (1), 35 (1903), 428-43.

M. S. Macphail and E. C. Titchmarsh

(1) The summability of Fourier's integral, Journal London Math. Soc. 11 (1936), 313-18.

M. MATHIAS

(1) Über positive Fourier-Integrale, Math. Zeitschrift, 16 (1923), 103-25.

A. MAYER and E. LEONTOVIČ

(1) On some inequality relating to Fourier's integral, C. R. Acad. Sci. U.S.S.R. 4 (1934), 353-60 (Zentralblatt für Math. 10, 349).

B. M. MEHROTRA

- (1) Some theorems on self-reciprocal functions, *Proc. London Math. Soc.* (2), 34 (1932), 231-40.
- (2) On some self-reciprocal functions, Bull. Calcutta Math. Soc. 25 (1933), 167-72.
- (3) A few self-reciprocal functions, Proc. Physico-Math. Soc. of Japan (3), 16 (1934), 273-4.
- (4) A list of self-reciprocal functions, Journal Indian Math. Soc. (New Series), 1 (1934), 93-104.
- (5) Some definite integrals involving self-reciprocal functions, Bull. Amer. Math. Soc. 40 (1934), 265-6.

B. M. MEHROTRA (cont.)

- (6) Theorems connecting different classes of self-reciprocal functions, *Proc. Edinburgh Math. Soc.* (2), 4 (1934), 53-6.
- (7) An integral representing self-reciprocal functions, Bull. Calcutta Math. Soc. 26 (1934), 35-8.
- (8) Self-reciprocal functions, Tôhoku Math. Journal, 40 (1935), 451-85.
- (9) A brief history of self-reciprocal functions, Journal Indian Math. Soc. (New Series), 1 (1934), 209-27.

H. MELLIN

- (1) Ueber die fundamentale Wichtigkeit des Satzes von Cauchy für die Theorien der Gamma- und der hypergeometrischen Funktionen, Acta Soc. Fennicae, 21 (1896), No. 1, 1-115.
- (2) Über den Zusammenhang zwischen den linearen Differential- und Differenzengleichungen, Acta Math. 25 (1902), 139-64.
- (3) Eine Formel für den Logarithmus transcendenter Functionen von endlichem Geschlecht, *Acta Math.* 25 (1902), 165-83.
- (4) Abriss einer einheitlichen Theorie der Gamma- und der hypergeometrischen Funktionen, Math. Annalen, 68 (1910), 305-37.

A. MILNE

(1) On the equation of the parabolic cylinder function, Proc. Edin. Math. Soc. 32 (1914), 2-14.

E. A. MILNE

- (1) Radiative equilibrium in the outer layers of a star, Monthly Notices of the R.A.S. 81 (1921), 361-75.
- (2) Maxwell's law, and the absorption and emission of radiation, *Proc. Camb. Phil. Soc.* 23 (1926), 465-71.

S. C. MITRA

(1) An integral equation satisfied by the square of Weber's parabolic cylinder function, *Journal London Math. Soc.* 11 (1936), 252-6.

L. J. MORDELL

- (1) Poisson's summation formula and the Riemann zeta-function, Journal London Math. Soc. 4 (1928), 285-91.
- (2) The zeta-functions arising from quadratic forms, and their functional equations, Quart. J. of Math. (Oxford), 1 (1930), 77-101.

G. W. MORGAN

- (1) A note on Fourier transforms, Journal London Math. Soc. 9 (1934), 187-92.
- (2) On the convergence of transform integrals, Proc. London Math. Soc. (2) 41 (1936), 199-208.

N. J. MUSCHELOV

(1) Auflösung einer Integralgleichung, die in der Theorie der schwarzen Strahlung vorkommt. See Fortschitte der Math. 50 (1924), 275.

J. NEUFELD

(1) On the operational solution of linear mixed difference-differential equations, *Proc. Camb. Phil. Soc.* 30 (1934), 389-91.

(2) On multiplication of operational expressions, *Tôhoku Math. Journal*, 40 (1935), 449-50.

F. NEVANLINNA

(1) Über die Summation der Fourierschen Reihen und Integrale, Finska Vet. Soc. Förh. (A), 64 (1922), No. 3.

J. W. NICHOLSON

- (1) Generalisation of a theorem due to Sonine, Quart. J. of Math. 48 (1920), 321-9.
- (2) A problem in the theory of heat conduction, *Proc. Royal Soc.* (A), 100 (1921), 226-40.

A. C. Offord

- (1) On Fourier transforms, *Proc. London Math. Soc.* (2), 28 (1934), 197-216.
- (2) On Hankel transforms, *Proc. London Math. Soc.* (2), 39 (1935), 49-67.
- (3) The uniqueness of a certain trigonometrical integral, *Proc. Camb. Phil. Soc.* 31 (1935), 382-9.
- (4) On Fourier transforms II, *Proc. London Math. Soc.* (2), 40 (1935), 281-9.
- (5) On Fourier transforms III, Trans. Amer. Math. Soc. 38 (1935), 250-66.
- (6) Note on the uniqueness of the representation of a function by a trigonometric integral, *Journal London Math. Soc.* 11 (1936), 171-4.
- (7) On the uniqueness of the representation of a function by a trigonometric integral, *Proc. London Math. Soc.* (2) 42 (1937), 422-80.

A. Oppenheim

(1) Some identities in the theory of numbers, *Proc. London Math. Soc.* (2), 26 (1927), 295-350.

L. Orlando

- (1) Sulla formula integrale di Fourier, *Rom. Accad. L. Rend.* (5), 17₂ (1908), 367-71.
- (2) Nuove osservazioni sulla formula integrale di Fourier, Rom. Accad. L. Rend. (5), 18₂ (1909), 343-8.
- (3) Sopra un nuovo aspetto della formula integrale di Fourier, Rom. Accad. L. Rend. (5), 22₁ (1913), 65-6.

W. M'F. ORR

(1), (2) Extensions of Fourier's and the Bessel-Fourier theorems, Proc. Royal Irish Acad. 27 (1909), Sect. A, 205-48; and ibid. 29 (1911), Sect. A, 10-32.

P. M. OWEN

(1) The Riemannian theory of Hankel transforms, Proc. London Math. Soc. (2), 39 (1935), 295-320.

R. E. A. C. PALEY and N. WIENER

(1)-(7) Notes on the theory and application of Fourier transforms. Notes I-II, Trans. Amer. Math. Soc. 35 (1933), 348-55; Notes III-VII, ibid. 761-91.

E. G. PHILLIPS

(1) Note on a problem of Ramanujan, Journal London Math. Soc. 4 (1929), 310-13.

E. PICARD

(1) Sur un exemple simple d'une équation singulière de Fredholm où la nature analytique de la solution dépend du second membre, Ann. de l'Éc. Norm. (3) 28 (1911), 313-24.

S. Pincherle

(1) Studi sopra alcune operazioni funzionali, Mem. di Bologna (4), 7 (1886), 393-442.

M. PLANCHEREL

- (1) Contribution à l'étude de la représentation d'une fonction arbitraire par des intégrales définies, *Rend. di Palermo*, 30 (1910), 289-335.
- (2) Zur Konvergenztheorie der Integrale $\lim_{z \to \infty} \int_{a}^{z} f(x) \cos xy \, dx$, Math. Annalen, 74 (1913), 573-8.
- (3) Sur la convergence et sur la sommation par les moyennes de Cesàro de $\lim_{z\to\infty} \int_a^z f(x)\cos xy \ dx$, Math. Annalen, 76 (1915), 315-26.
- (4) Sur les formules d'inversion de Fourier et de Hankel, Proc. London Math. Soc. (2), 24 (1925), 62-70.
- (5) Formule de Parseval et transformations fonctionelles orthogonales, Commentarii math. Helvetici, 1 (1929), 273-88.
- (6) Sur les formules de réciprocité du type de Fourier, Journal London Math. Soc. 8 (1933), 220-6.

A. PLESSNER

- (1) Zur Theorie der konjugierten trigonometrischen Reihen, Dissertation, Giessen, 1923.
- (2) Über Konvergenz von trigonometrischen Reihen, Journal für Math. 155 (1925), 15-25.

H. Poincaré

(1) Sur la théorie des quanta, Journal de physique (5), 2 (1912), 5-34.

B. von der Pol

(1) On the operational solution of linear differential equations and an investigation of the properties of these solutions, *Phil. Mag.* (7), 8 (1929), 861-98.

S. POLLARD

- (1) The evaluation of certain definite integrals involving trigonometrical functions by means of Fourier's integral theorem, Messenger of Math. 50 (1920), 151-6.
- (2) The summation of a Fourier integral of finite type, Proc. Camb. Phil. Soc. 23 (1926), 373-82.
- (3) Identification of the coefficients in a trigonometrical integral, Proc. London Math. Soc. (2), 25 (1926), 451-68.
- (4) On Fourier's integral, Proc. London Math. Soc. (2) 26 (1927), 12-24.

G. Pólya

- (1) On the zeros of an integral function represented by Fourier's integral, Messenger of Math. 52 (1923), 185-8.
- (2) Herleitung des Gauss'schen Fehlergesetzes aus einer Funktionalgleichung, *Math. Zeitschrift*, 18 (1923), 96-108.
- (3) On the zeros of certain trigonometric integrals, Journal London Math. Soc. 1 (1926), 98-9.
- (4) Über trigonometrische Integrale mit nur reellen Nullstellen, Journal für Math. 158 (1927), 6-18.

H. S. A. POTTER

(1) Approximate equations for the Epstein zeta-function, *Proc. London Math. Soc.* (2), 36 (1933), 501-15.

B. N. PRASAD

(1) A new theorem on the representation of a function by Fourier's single integral, *Journal London Math. Soc.* 7 (1932), 36-8.

A. Pringsheim

- (1) Über neue Gültigkeitsbedingungen für die Fouriersche Integralformel, Math. Annalen, 68 (1910), 367-408.
- (2) Nachtrag zu der Abhandlung: Über neue..., Math. Annalen, 71 (1911), 289-98.

S. RAMANUJAN

- (1) Some definite integrals, Messenger of Math. 44 (1915), 10-18.
- (2) Some definite integrals connected with Gauss's sums, Messenger of Math. 44 (1915), 75-85.
- (3) New expressions for Riemann's functions $\xi(s)$ and $\Xi(t)$, Quart. J. of Math. 46 (1915), 253-60.
- (4) Some definite integrals, *Proc. London Math. Soc.* (2), 17 (1918), Records for Jan. 17, 1918.
- (5) Some definite integrals, Journal Indian Math. Soc. 11 (1919), 81-7.
- (6) A class of definite integrals, Quart. J. of Math. 48 (1920), 294-310.

LORD RAYLEIGH

(1) On the character of the complete radiation at a given temperature, *Phil. Mag.* (5), 27 (1889), 460-9.

B. RIEMANN

(1) Ueber die Anzahl der Primzahlen unter einer gegebener Grösse, Werke (ed. 1876), 136-44.

F. Riesz

- (1) Über eine Verallgemeinerung der Parsevalschen Formel, Math. Zeitschrift, 18 (1923), 117-24.
- (2) Sur la formule d'inversion de Fourier, Acta Szeged, 3 (1927), 235-41.

M. RIESZ

- (1) Les fonctions conjuguées et les séries de Fourier, Comptes rendus, 178 (1924), 1464-7.
- (2) Sur les fonctions conjuguées, Math. Zeitschrift, 27 (1927), 218-44.

F. SBRANA

- (1) Sopra alcune formule di risoluzione di certe equazioni integrali di Volterra, Rom. Accad. L. Rend. (5), 31, (1922), 454-6.
- (2) Sopra un gruppo di operatori funzionali che interessano la fisica, Annali di Mat. (4), 7 (1929), 33-45.
- (3) Sulla integrazione delle equazioni lineari alle derivate parziali a coefficienti costanti, Atti Soc. Liquet. Sci. Geneva, 13 (1934), 177-211.

E. SCHMIDT

(1) Über eine Klasse linearer funktionaler Differentialgleichungen, Math. Annalen, 70 (1911), 499-524.

F. SCHÜRER

(1) Eine gemeinsame Methode zur Behandlung gewisser Funktionalgleichungsprobleme, Leipziger Berichte, 70 (1918), 185-240.

A. Schuster

(1) On interference phenomena, Phil. Mag. (5), 37 (1894), 509-45.

A. SOMMERFELD

(1) Die willkürlichen Funktionen in der mathematischen Physik, Inaug. Diss., Königsberg, 1901.

P. L. SRIVASTAVA

(1) Sur une classe de séries de Taylor et les fonctions entières associées, Bulletin S.M.F. 57 (1929), 160-73.

S. W. P. STEEN

- (1) Divisor functions, *Proc. London Math. Soc.* (2), 31 (1930), 47-80; and ibid. 32 (1931), 356-68.
- (2) Note on transforms, Journal London Math. Soc. 10 (1935), 151-60.

S. SZIDON

(1) Reihentheoretische Sätze und ihre Anwendungen in der Theorie der Fourierschen Reihen, Math. Zeitschrift, 10 (1921), 121-7.

T. TAKAHASHI

(1) On the conjugate function of an integrable function and Fourier series and Fourier transform, Sci. Rep. Tôhoku Univ. (1), 25 (1936), 56-78.

J. D. TAMARKIN

(1) On Laplace's integral equations, Trans. Amer. Math. Soc. 28 (1926), 417-25.

E. C. TITCHMARSH

- (1) Hankel transforms, Proc. Camb. Phil. Soc. 21 (1923), 463-73.
- (2) A contribution to the theory of Fourier transforms, Proc. London Math. Soc. (2), 23 (1923), 279-89.
- (3) A pair of inversion formulae, London Math. Soc. Records, May 1923; Proc. London Math. Soc. (2) 22 (1923), xxxiv-xxxv.
- (4) Extensions of Fourier's integral formula to formulae involving Bessel functions, London Math. Soc. Records, Jan. 1924; Proc. London Math. Soc. (2) 23 (1924), xxii-xxiv.
- (5) Conjugate trigonometrical integrals, Proc. London Math. Soc. (2), 24 (1924), 109-30.

- (6) An inversion formula involving Bessel functions, London Math. Soc. Records, Nov. 1924; Proc. London Math. Soc. (2) 24 (1925), vi-vii.
- (7) Reciprocal formulae involving series and integrals, Math. Zeitschrift, 25 (1926), 321-47, and Correction, ibid. 26 (1927).
- (8) The zeros of certain integral functions, Proc. London Math. Soc. (2), 25 (1926), 283-302.
- (9) A theorem on infinite products, Journal London Math. Soc. 1 (1926), 35-7.
- (10) A note on Hankel transforms, Journal London Math. Soc. 1 (1926), 195-6.
- (11) Some integrals involving Bessel functions, Journal London Math. Soc. 2 (1927), 97-9.
- (12) A note on Fourier transforms, Journal London Math. Soc. 2 (1927), 148-50.
- (13) On conjugate functions, Proc. London Math. Soc. (2), 29 (1928), 49-80.
- (14) Additional note on conjugate functions, *Journal London Math. Soc.* 4 (1930), 204-6.
- (15) A proof of a theorem of Watson, Journal London Math. Soc. 8 (1933), 217-20.

F. TRICOMI

- (1) Trasformazione di Laplace e polinimi di Laguerre I, Rend. Lincei (6), 21 (1935), 232-9.
- (2) Sulla trasformazione di Laplace, Conferenze di fisica e matematica, Torino (1935), 1-15.
- (3) Autovalori e autofunzioni del nucleo di Hankel, Reale Acc. delle Scienze di Torino, 71 (1935), 1-7.
- (4) Sulla trasformazione e il teorema di reciprocita di Hankel, Rend. Lincei (6), 22 (1935), 564-71.
- (5) Un teorema abeliano per la trasformazione di Hankel e alcune nuove applicazioni di una formula sulle funzioni di Bessel, *Rend. Lincei* (6), 22 (1935), 572-6.
- (6) Sulle trasformazioni funzionali lineari commutabili con la derivazione, Commentarii Math. Helvetici, 8 (1935), 70-87.
- (7) Über Doetschs Umkehrformel der Gauss-Transformation und eine neue Umkehrung der Laplace-Transformation, *Math. Zeitschrift*, 40 (1936), 720-6.

R. S. VARMA

(1) Some functions which are self-reciprocal in the Hankel transform, *Proc. London Math. Soc.* (2), 42 (1936), 9-17.

S. Verblunsky

- (1) Some theorems on moment constants and functions, *Proc. London Math. Soc.* (2), 37 (1934), 338-82.
- (2) Trigonometric integrals and harmonic functions, *Proc. London Math. Soc.* (2), 38 (1934), 1-48.

S. VERBLUNSKY (cont.)

(3) On positive harmonic functions in a half-plane, Proc. Camb. Phil. Soc. 31 (1935), 482-507.

A. WALFISZ

- (1) Über die summatorischen Funktionen einiger Dirichletscher Reihen Dissertation, Göttingen, 1922.
- (2) Über zwei Gitterpunktprobleme, Math. Annalen, 95 (1926), 69-83.

G. N. WATSON

- (1) Some self-reciprocal functions, Quart. J. of Math. (Oxford), 2 (1931), 298-309.
- (2) General transforms, Proc. London Math. Soc. (2), 35 (1933), 156-99.
- (3) Notes on generating functions of polynomials: (2) Hermite polynomials, Journal London Math. Soc. 8 (1933), 194-9.
- (4) An integral equation for the square of a Laguerre polynomial, Journal London Math. Soc. 11 (1936), 256-61.
- (5) Ramanujan's integrals and Gauss's sums, Quart. J. of Math. (Oxford), 7 (1936), 175-83.

H. A. WEBB

(1) On the solution of linear difference equations by definite integrals, Messenger of Math. 34 (1904), 40-5.

K. WEIERSTRASS

(1) Ueber die analytische Darstellbarkeit sogenannter willkürlicher Functionen reeller Argumente, Berliner Sitzungsberichte (1885), 633-9, 789-805 (Werke, iii. 1-37).

H. WEYL

(1) Singuläre Integralgleichungen, Math. Annalen, 66 (1908), 273-324.

E. T. WHITTAKER

(1) On the numerical solution of integral equations, *Proc. Royal Soc.* (A), 94 (1917), 367-83.

D. V. WIDDER

- (1) The inversion of the Laplace integral and the related moment problem, Trans. Amer. Math. Soc. 36 (1934), 107-201.
- (2) An application of Laguerre polynomials, Duke Math. Journal, 1 (1935), 126-36.

N. WIENER

- (1) The operational calculus, Math. Annalen, 95 (1916), 557-84.
- (2) On the representation of functions by trigonometrical integrals, Math. Zeitschrift, 24 (1925), 575-616.
- (3) Hermitian polynomials and Fourier analysis, J. Math. Phys. Mass. Inst. Tech. 7 (1928), 161-84.

N. WIENER and E. HOPF

(1) Über eine Klasse singulärer Integralgleichungen, Sitz. Berlin. Akad. Wiss. (1931), 696-706.

B. M. Wilson

(1) On an extension of Milne's integral equation, Messenger of Math. 53 (1923), 157-60.

F. WOLF

(1) On the (C, k) summability of a trigonometrical integral, *Proc. London Math. Soc.* (2), 40 (1936), 502-23.

F. M. Wood

(1) Reciprocal integral formulae, Proc. London Math. Soc. (2), 29 (1929), 29-48.

D. M. WRINCH and J. W. NICHOLSON

(1) A class of integral equations occurring in physics, *Phil. Mag.* (7), 4 (1927), 531-60.

W. H. YOUNG

- (1) On Fourier's repeated integral, *Proc. Royal Soc. Edin.* 31 (1912), 559-86.
- (2) On Sommerfeld's form of Fourier's repeated integrals, *Proc. Royal Soc. Edin.* 31 (1912), 587-603.

A. ZYGMUND

- Über die Beziehung der Eindeutigkeitsfragen in den Theorien der trigonometrischen Reihen und Integrale, Math. Annalen, 99 (1928), 562-89.
- (2) A remark on Fourier transforms, Proc. Camb. Phil. Soc. 32 (1936), 321-7.

ADDITIONAL REFERENCES

BOOKS

- H. S. CARSLAW and J. C. JAEGER, Operational Methods in Applied Mathematics, Oxford, 1941.
- G. Doetsch, Theorie und Anwendung der Laplace-Transformation, Berlin, 1937.
- N. W. McLachlan, Complex Variable and Operational Calculus with Technical Applications, Cambridge, 1939.

ORIGINAL MEMOIRS

R. P. Boas

(2) Poisson's summation formula in L^2 , Journal London Math. Soc. 21 (1946), 102-5.

S. Borgen

- (1) Note on the summability of Poisson's formula, Journal London Math. Soc. 19 (1944), 100-5.
- (2) Note on Poisson's formula, Journal London Math. Soc. 19 (1944), 213-19.

I. W. BUSBRIDGE

- (2) Dual integral equations, *Proc. London Math. Soc.* (2), 44 (1938), 115-29.
- (3) A theory of general transforms for functions of the class $L^p(0, \infty)$ (1 , Quart. J. of Math. (Oxford), 9 (1938), 148-60.
- (4) An additional note on general transforms of the class $L^p(0, \infty)$ (1 , Quart. J. of Math. (Oxford), 10 (1939), 11-12.
- (5) The theory of general transforms for functions of the class $L^p(0, \infty)$ (1 II, Quart. J. of Math. (Oxford), 10 (1939), 13-27.
- (6) On the solution of an integral equation, Journal London Math. Soc. 14 (1939), 128-32.

J. A. CARROLL

(1) An application of Fourier transforms to the evaluation of a class of repeated integrals arising in calculations of probability, Quart J. of Math. (Oxford), 9 (1938), 176-81.

J. L. B. COOPER

- (1) An integral equation, Quart. J. of Math. (Oxford), 9 (1938), 263-73.
- (2) The absolute Cesàro summability of Fourier integrals, *Proc. London Math. Soc.* (2), 45 (1939), 425–39.
- (3) A mixed boundary-value problem, Journal London Math. Soc. 14 (1939), 124-8.
- (4) The uniqueness of trigonometrical integrals, *Proc. London Math. Soc.* (2), 48 (1944), 292-309.

E. T. Copson

- (1) On an integral equation arising in the theory of diffraction, Quart. J. of Math. (Oxford), 17 (1946), 19-34.
- (2) On the problem of the electrified disc, Proc. Edinburgh Math. Soc. (3) 8 (1947), 14-19.

J. Cossar

(2) On conjugate functions, *Proc. London Math. Soc.* (2), 45 (1939), 369-81.

M. M. CRUM

- (1) Some inversion formulae, Quart. J. of Math. (Oxford), 11 (1940), 49-52.
- (2) On the resultant of two functions, Quart. J. of Math. (Oxford), 12 (1941), 108-11.

G. Doetsch

(5) Zerlegung einer Funktion in Gaussche Fehlerkurven und zeitliche Zurückverfolgung eines Temperaturzustandes, Math. Zeitschrift, 41 (1936), 283-318.

- (6) Beitrag zu Watson's 'General Transforms', Math. Annalen, 113 (1937), 226-41.
- (7) Zur Theorie der involutorischen Transformationen (General Transforms) und der selbstreziproken Funktionen, *Math. Annalen*, 113 (1937), 665-76.
- (8) Bedingungen für die Darstellbarkeit einer Funktion als Laplace-Integral und eine Umkehrformel für die Laplace-Transformation, *Math. Zeitschrift*, 42 (1937), 263–86.

SHEILA M. EDMONDS

- (1) On the Parseval formulae for Fourier transforms, *Proc. Camb. Phil. Soc.* 38 (1942), 1-19.
- (2) The Parseval formulae for monotonic functions I, Proc. Camb. Phil. Soc. 43 (1947), 289-306.

A. Erdélyi

- (1) On certain Hankel transforms, Quart. J. of Math. (Oxford), 9 (1938), 196-8.
- (2) On the connection between Hankel transforms of different order, Journal London Math. Soc. 16 (1941), 113-17.

W. L. FERRAR

(2) Summation formulae and their relation to Dirichlet's series II, Compositio Math. 4 (1937), 394-405.

C. Fox

(4) A class of Fourier kernels, Journal London Math. Soc. 14 (1939), 278-81.

A. P. GUINAND

- Summation formulae and self-reciprocal functions, Quart. J. of Math. (Oxford), 9 (1938), 53-67.
- (2) Finite summation formulae, Quart. J. of Math. (Oxford), 10 (1939), 38-44.
- (3) Summation formulae and self-reciprocal functions II, Quart. J. of Math. (Oxford), 10 (1939), 104-18.
- (4) General transformations and the Parseval theorem, Quart. J. of Math. (Oxford), 12 (1941), 51-6.
- (5) On Poisson's summation formula, *Annals of Math.* 42 (1941), 591–603.
- (6) Summation formulae and self-reciprocal functions III, Quart. J. of Math. (Oxford), 13 (1942), 30-9.
- (7) Simple Fourier transformations, Quart. J. of Math. (Oxford), 13 (1942), 153-8.
- (8) Some Fourier transforms in prime-number theory, Quart. J. of Math. (Oxford), 18 (1947), 53-64.

G. H. HARDY

(21) Ramanujan and the theory of Fourier transforms, Quart. J. of Math. (Oxford), 8 (1937), 245-54.

E. HILB

(2) Zur Theorie der linearen funktionalen Differentialgleichungen, Math. Annalen, 78 (1918), 137-70.

M. KAC

(1) Quelques remarques sur les zéros des intégrales de Fourier, *Journal London Math. Soc.* 13 (1938), 128–30.

T. KITAGAWA

(1) On the theory of linear translatable functional equation and Cauchy's series, *Japanese Journ. Math.* 13 (1937), 233-332.

H. KOBER

- (1) Involutorische Transformationen und selbstreziproke Funktionen, Proc. London Math. Soc. (2), 44 (1938), 453-65.
- (2) Unbeschränkte Operatoren, die eine Verallgemeinerung der Fouriertransformationen sind, Quart. J. of Math. (Oxford), 9 (1938), 41-52.
- (3) Wurzeln aus der Hankel-, Fourier-, und aus anderen stetigen Transformationen, Quart. J. of Math. (Oxford), 10 (1939), 45-59.
- (4) Beziehungen zwischen Systemen selbstreziproker Funktionen, Proc. London Math. Soc. (2), 45 (1939), 229-42.
- (5) On Dirichlet's singular integral and Fourier transforms, Quart. J. of Math. (Oxford), 12 (1941), 78-85.
- (6) A note on Hilbert transforms, Quart. J. of Math. (Oxford), 14 (1943), 49-54.
- (7) A note on Hilbert transforms, Journal London Math. Soc. 18 (1943), 66-71.
- (8) A note on Fourier transforms, Journal London Math. Soc. 19 (1944), 144-52.

B. Mohan (= B. M. Mehrotra)

(10) On self-reciprocal functions, Quart. J. of Math. (Oxford), 10 (1939), 252-60.

P. M. OWEN

(2) Parseval's theorem for Hankel transforms, Proc. London Math. Soc. (2), 45 (1939), 458-74.

M. PLANCHEREL

(7) Quelques remarques à propos d'une note de G. H. Hardy: The resultant of two Fourier kernels, *Proc. Camb. Phil. Soc.* 33 (1937), 413-18.

I. N. SNEDDON

(1) The distribution of stress in the neighbourhood of a crack in an elastic solid, *Proc. Royal Soc.* (A) 187 (1946), 229-60.

E. C. TITCHMARSH

(16) Solutions of some functional equations, Journal London Math. Soc. 14 (1939), 118-24.

F. TRICOMI

(8) Les transformations de Fourier, Laplace et Gauss, et leurs applications au calcul des probabilités et à la statistique, Annales de l'Institut H. Poincaré, 8, 111-49.



INDEX

Abel, 331. Allied integral, 119, 147. Analytic functions, 44, 252, 259.

Barnes, 194, 195.
Bateman, 216, 297, 328, 349.
Bernstein, F., 210.
Bernstein, S., 115.
Bessel functions, 64, 65, 121, 178, 182-4, 190, 196-210, 212, 214-20, 231-2, 240-4, 264, 269, 274, 279, 288, 297-8, 314, 329-31, 334, 339, 352-8.
Bochner, 73, 307.
Bosanquet, 332, 359.
Bromwich, 275.
Browne, 311.

Cahen, 7.
Campbell and Foster, 177.
Cantor, 160.
Carlson, 9.
Carslaw, 281, 284, 289.
Cauchy, 1-4, 30, 124, 128.
Constant in Fourier's formula, 24.
Cooke, 216.
van der Corput, 22.

Busbridge, 222, 225, 248, 335.

Dini's test, 13–14.
Dirichlet series, 7, 43.
Dixon, A. C., 309, 342.
Dixon, A. L., 217.
Doetsch, 312.

Fatou, 89, 134.
Fejér, 29, 38.
Ferrar, 61, 62.
Fock, 312, 314.
Fourier, 1.
Fourier kernel, 212, 229.
Fourier's single-integral formula, 3, 25, 83.
Fourier transforms, 3, 69-118, 148, 169, 174.
Fowler, R. H., 367.
Fox, 215, 332.

Generalized integrals, 4, 43, 257, 282, 293, 299–302, 306–7, 340–3, 347. Goldstein, 287, 329. Goursat, 313.

Hahn, 14, 20.
Hankel, 214, 240, 242, 245, 266, 335.
Hardy, 29, 30, 31, 36, 43, 47, 51, 54, 56, 93, 96, 102, 107, 110, 113, 114, 122, 174, 180, 208, 210, 212, 216, 217, 227, 229, 232, 248, 265, 267, 271, 310, 318, 320, 346, 349, 352, 358.
Hausdorff, 100.
Hermite polynomials, 76–82, 261.
Hille, 148.
Hobson, 25, 31, 238.
Hopf, 339, 344, 346.
Humbert, 210.

Ingham, 174.

Integrated form of Fourier's formula, 41.

Jacob, 157. Jeffreys, 277, 286, 296. Jordan's test, 13.

Kapteyn, 352, 355. King, 337. Kolmogoroff, 143. Kuttner, 217.

L-class, 10.

Macphail, 36.

£-class, 94. L-function, 66, 262-3, 265. Lamb, 296. Landau, 265, 267. Laplace, 6, 48, 206, 316. Lattice-points, 265. Lebesgue, 31. Linfoot, 61. Lipschitz condition, 115, 145. Littlewood, 96, 102, 107, 110, 113, 132, 143.

Mehrotra, 248, 264, 267, 272.

Mellin, 7, 46, 52, 60, 94, 118, 190, 196, 222.

Milne, A., 261.

Milne, E. A., 344, 346, 367.

Mitra, 262.

Monotonic functions, 15.

Mordell, 61, 266.

Morgan, 174, 227.

Neumann series, 352. Nicholson, 243, 287. Notation, 9. 394 INDEX

Offord, 157, 165, 166. Oppenheim, 267. Oscillating functions, 22.

Paley, 174, 218, 313, 317, 318, 323, 339.

Parseval formula, 50, 54, 56, 58, 94, 105, 110, 150, 166, 180.

Periodic factors, 17.

Perron, 43.

Phillips, E. 3. 267.

Phragmén and Lindelöf, 175, 325.

Picard, 309.

Plancherel, 69, 84–6, 223.

Plessner, 87, 122, 132.

Poisson, 60.

Pollard, 157.

Pólya, 96, 363.

Ramanujan, 51, 62, 184, 186. Rayleigh, 51. Resultant, 51, 52, 59, 90, 106. Riemann, 7. Riemann-Lebesgue theorem, 11. Riemann-Weber, 281, 284, 294, 334. Riesz, F., 75, 102. Riesz, M., 43, 105, 133, 138.

Schmidt, 300. Sommerfeld, 30.

Potter, 266.

Privaloff, 145.

Prasad, B. N., 25. Pringsheim, 15, 17, 25. Sonine, 242, 260. Steen, 217. Stieltjes, 317, 320. Struve's function **H**_v, 121, 179, 182, 197, 212, 215. Summability, 26-41, 45, 83-4. Szasz, 115. Szidon, 170.

Tamarkin, 148.
Titchmarsh, 36, 56, 70, 96, 111, 115, 121, 133, 143, 145, 172, 212, 215, 216, 217, 223, 227, 232, 243, 248, 265, 267, 310, 323, 325, 346, 349, 352, 358.

de la Vallée-Poussin, 153. Varma, 262. Verblunsky, 167.

Walfisz, 267.
Watson, 77, 184, 212, 222, 262, 274.
Weierstrass, 31, 76, 282.
Whittaker, E. T., 313.
Whittaker and Watson, 42, 194.
Widder, 317.
Wiener, N., 76, 174, 218, 313, 317, 318, 323, 339.
Wilson, B. M., 261.

Young, W. H., 97, 98, 100.

zeta-function, 63, 65, 262, 265, 273. Zygmund, 98, 105, 109, 118, 143, 145

PRINTED IN
GREAT BRITAIN
AT THE
UNIVERSITY PRESS
OXFORD
BY
CHARLES BATEY
PRINTER
TO THE

UNIVERSITY